Nutrition and Health Series Editor: Adrianne Bendich

Connie Watkins Bales Julie L. Locher Edward Saltzman *Editors*

Handbook of Clinical Nutrition and Aging

Third Edition



NUTRITION AND HEALTH

Adrianne Bendich, Ph.D., FASN, FACN, SERIES EDITOR

For further volumes: http://www.springer.com/series/7659

Connie Watkins Bales • Julie L. Locher Edward Saltzman Editors

Handbook of Clinical Nutrition and Aging

Third Edition

🔆 Humana Press

Editors Connie Watkins Bales Geriatric Research, Education, and Clinical Center Durham VA Medical Center and Department of Medicine Duke University Medical Center Durham, NC, USA

Edward Saltzman Jean Mayer USDA Human Nutrition Research Center on Aging Tufts University Boston, MA, USA Julie L. Locher Departments of Medicine and Health Care Organization and Policy Translational Nutrition and Aging Program University of Alabama at Birmingham Birmingham, AL, USA

ISBN 978-1-4939-1928-4 ISBN 978-1-4939-1929-1 (eBook) DOI 10.1007/978-1-4939-1929-1 Springer New York Heidelberg Dordrecht London

Library of Congress Control Number: 2014953584

© Springer Science+Business Media New York 2004, 2009, 2015

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. Exempted from this legal reservation are brief excerpts in connection with reviews or scholarly analysis or material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work. Duplication of this publication or parts thereof is permitted only under the provisions of the Copyright Law of the Publisher's location, in its current version, and permission for use must always be obtained from Springer. Permissions for use may be obtained through RightsLink at the Copyright Clearance Center. Violations are liable to prosecution under the respective Copyright Law.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

While the advice and information in this book are believed to be true and accurate at the date of publication, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed on acid-free paper

Humana Press is a brand of Springer Springer is part of Springer Science+Business Media (www.springer.com) Dr. Bales dedicates this book in honor of her brand new grandchildren, Ada Elizabeth and William Thomas Britton. Thank you for reminding us that tomorrow brings new life, fresh beginnings, and hope for a better world!

Dr. Locher dedicates this work to those in health care and community settings who are working together to comprehensively address nutrition risk and food insecurity among older adults.

Dr. Saltzman: To my mentors and those who have served as role models, whose contributions to my success have been incalculably great and for which I am more grateful than words can express.

Foreword

This new edition of the *Handbook of Clinical Nutrition and Aging*, edited by Connie Bales, Edward Saltzman, and Julie Locher, is a most timely comprehensive overview of relevant new findings for practitioners, investigators, and policy makers. The individual chapters share knowledge and expertise from authors who are preeminent leaders in the field.

As the twenty-first century unfolds, the world will confront many challenges in caring for aged adults that far exceed any ever experienced previously. The US population aged 65 years and older has increased by 21 % over the past two decades. In the coming years, it will grow even more rapidly, increasing more than twofold by 2060. Even more rapid change is projected for the oldest-old; those 85 years and older are expected to triple between 2012 and 2040. Such unprecedented increases in the numbers of adults reaching old age will produce major changes in population age distributions, not only in the Western world but around the globe, i.e., in developed as well as developing nations. These dramatic changes will have an indelible impact on health care and the practice of clinical nutrition.

To some extent, this is old news. These predictions have been on the radar for at least two decades, prompting efforts to consider new comprehensive systems of healthcare delivery for older persons. However, as we now engage the aging cohort of the "baby boomer" generation, the future has arrived, and we must face the realities of shifting demographics and limited resources. What it means to be an older person has also continued to evolve. Who imagined even 100 years ago that life expectancies would rise to the current degree? Who imagined that, for some, age 70 would functionally become the new age 60? Who imagined that so many older persons would have such active, rich, and fulfilling lives? Unfortunately a substantial number of older persons will not have these opportunities because they suffer aging-associated conditions like chronic disease, depression, dementia, poor oral health, sarcopenia, functional decline, and frailty. Malnutrition is associated with many of these conditions, both as a precipitating factor and as an adverse outcome that blunts the success of medical therapies and contributes to decline. Obesity has also become common among older persons and is strongly associated with disease burden and functional limitation. Sarcopenic obesity is especially common and may contribute to functional impairment. Paradoxically, many obese older persons consume poor quality diets and also suffer micronutrient deficiencies. Therefore, as highlighted in this book, practitioners will confront a complex spectrum of nutritional concerns among older persons, everything from malnutrition in the setting of classic starvation to malnutrition in the setting of disease or injury to malnutrition in the setting of obesity.

In most Western countries, including the United States, diet-related chronic diseases are the single largest cause of morbidity and mortality. Optimally implemented, preventive nutrition and medical nutrition therapies have the potential to be major determinants of the ultimate impact of aging on world populations. Sound nutrition offers opportunities to favorably alter health-related quality of life as well as healthcare resource use by older persons. With respect to related aging policies, we are witnessing two evolving trends that bring nutritional concerns of older persons to the forefront. First, there is a growing movement to shift the emphasis of long-term care from institutional settings to home- and community-based settings. Central to this movement is the provision of nutrition services to older adults residing in the community who are at nutritional risk, especially those lacking resources (both financial and social), and those who experience chronic disease burden and functional decline that increase their likelihood of being hospitalized or placed in a skilled nursing facility. Second, there are increasing financial penalties to hospital readmission. Malnutrition is strongly associated with increased risk for such undesirable outcomes. Connecting medical and social services that address nutritional needs is vital in caring for older persons. An important new feature of this edition of this book is a focus on elders residing in the community and how best, both effectively and efficiently, to meet their nutritional needs.

Gordon L. Jensen, M.D., Ph.D.

Preface

It has been a decade since the first edition of this *Handbook* was created. The intervening time has indeed brought a mixture of blessings and concerns. Rapid scientific progress and advances in information transfer have fostered medical discoveries and dissemination of health-related information at an unprecedented rate. Consumers are more aware of the value of health-promoting behaviors than ever before, and older adults are paying close attention. For example, older persons are the greatest consumers of dietary supplements. Many employers have introduced wellness in the workplace initiatives, including to their retirees. Yet, the stark economic difficulties of the past few years have added unanticipated challenges for many individuals and families, often making it difficult to take advantage of these advances. The stress of limited time and resources and the confusion resulting from mixed messages in the media can be particularly challenging for older adults. Many are living on a fixed income and coping with a number of comorbidities as they make choices about health-related behaviors, including nutrition. As a result, they may not rank nutrition highly amongst the complex array of health-related recommendations they are offered, but we believe that they should!

In later life, as at any age, a poor diet and related factors like inactivity and obesity increase the risk of developing a long list of chronic conditions, such as cardiovascular disease, type 2 diabetes, metabolic syndrome, gallbladder disease, and cancer. Conversely, adherence to medical nutrition therapies for age-associated conditions such as renal disease and chronic heart failure can foster important improvements in health-related quality of life by preserving organ function and reducing problematic symptoms. With this in mind, we sought in this third edition of the Handbook to emphasize newer topics that have received limited attention heretofore, as well as mainstay topics for which assessment and therapies have continued to evolve and improve. For example, the increased recognition of physical function as a determinant of independence and well-being in later life is supported by three relevant chapters (Chaps. 6, 7, and 22). Two chapters (Chaps. 8 and 18) deal with aspects of diet and cognition, a current "hot topic" in geriatric nutrition. Chapters on food insecurity (Chap. 9), pressure ulcers (Chap. 14), feeding in late dementia (again, Chap. 18), and end of life issues (Chap. 19) highlight nutritional concerns in highly vulnerable populations. As in past editions, our "menu" of chapters on clinical conditions includes obesity, diabetes, heart failure, cancer, kidney disease, and osteoporosis. We are also pleased to include a new chapter on nutrition and anti-aging (Chap. 20) and the latest information on the ever-challenging topic of dietary supplements (Chap. 22), as well as a number of additional chapters relating to new knowledge about the role of nutrition in aging on a broader scope. The contents of these chapters reflect an optimistic outlook. Clearly, the authors contributing chapters to the Handbook support the idea that optimal nutrition benefits all older adults and they offer specific suggestions for accomplishing these benefits.

As with the first two editions, this book is a uniquely comprehensive and current resource on the topic of secondary disease prevention and medical nutrition therapy for older adults. We hope that it will be a valuable guide to clinical nutritionists/dietitians, physicians, nurses, and therapists in speechlanguage and occupational therapy, as well as many other health professionals, including social workers and case managers, who provide care for this high-risk population. We are indebted to many individuals who contributed as we put together this edition of the *Handbook*. In particular, we sincerely thank Diane Lamsback, Developmental Editor at Springer. With the consuming demands on the time of both editors and authors, her ongoing support was invaluable and truly helped to make this edition become a reality. CWB thanks Sarah B. Rose, MS, RD, who assisted with all the initial chapter reviews and supported the Duke Center for Aging work on the *Handbook*. CWB also thanks Amanda Quinn, who was with Springer until recently, for support and guidance as three editions of this *Handbook* have come together. We all as editors especially acknowledge the encouragement from our series editor, Dr. Adrianne Bendich, to explore the critical clinical issues in geriatric nutrition through this collaboration with the gifted and dedicated scientists who study them. It is these scientist-authors who ultimately made the book a successfully reality and we thank them all.

Durham, NC, USA Birmingham, AL, USA Boston, MA, USA Connie Watkins Bales, Ph.D., R.D. Julie L. Locher, Ph.D. Edward Saltzman, M.D.

Series Editor Page

The great success of the Nutrition and Health Series is the result of the consistent overriding mission of providing health professionals with texts that are essential because each includes: (1) a synthesis of the state of the science, (2) timely, in-depth reviews by the leading researchers and clinicians in their respective fields, (3) extensive, up-to-date fully annotated reference lists, (4) a detailed index, (5) relevant tables and figures, (6) identification of paradigm shifts and the consequences, (7) virtually no overlap of information between chapters, but targeted, interchapter referrals, (8) suggestions of areas for future research, and (9) balanced, data-driven answers to patient as well as health professionals questions which are based upon the totality of evidence rather than the findings of any single study.

The series volumes are not the outcome of a symposium. Rather, each editor has the potential to examine a chosen area with a broad perspective, both in subject matter as well as in the choice of chapter authors. The international perspective, especially with regard to public health initiatives, is emphasized where appropriate. The editors, whose trainings are both research and practice oriented, have the opportunity to develop a primary objective for their book; define the scope and focus; and then invite the leading authorities from around the world to be part of their initiative. The authors are encouraged to provide an overview of the field, discuss their own research, and relate the research findings to potential human health consequences. Because each book is developed de novo, the chapters are coordinated so that the resulting volume imparts greater knowledge than the sum of the information contained in the individual chapters.

Handbook of Clinical Nutrition and Aging, Third Edition, edited by Connie W. Bales Ph.D., RD, Julie L. Locher Ph.D., MSPH and Edward Saltzman, M.D. is a very welcome addition to the Nutrition and Health Series and fully exemplifies the Series' goals. The first volume was published in 2004 and the Second Edition was published in 2009. Both volumes were given excellent evidence-based reviews by health professionals in both the geriatrics and clinical nutrition professional communities. Over the past 5 years, there have been major advances in the treatment of senior individuals, especially those with chronic diseases who require multiple medications, advanced treatments, and/or surgery. Likewise, research on the nutritional requirements of geriatric patients has expanded and reflects the changes in demographics, technical advances, and further emphasis on the interactions among a number of disease states that are at increased risk with aging. This Third Edition is therefore especially timely as almost 10 % of the US adult population is over 65 years of age and the number of seniors around the world is increasing. At the same time, major adverse health risks including obesity and diabetes continue to increase in this already at risk population.

The three editors of this volume, Connie W. Bales, Julie L. Locher, and Edward Saltzman, are internationally recognized leaders in the fields of clinical nutrition and geriatrics research, treatment and management. Each has extensive experience in academic medicine and collectively, they have over 500 peer-reviewed publications and numerous awards for their efforts to improve the care of the

geriatric population. The editors are excellent communicators and they have worked tirelessly to develop a book that is destined to continue as the benchmark in the field of nutrition and aging. Over the past 5 years, the editors have grown in their prominence in their fields and it is of benefit to the reader that these practice and research-oriented editors are committed to providing readers with the most up-to-date objective and balanced reviews as found in the last two excellent editions.

Dr. Connie Bales is a Professor in the Division of Geriatrics in the Department of Medicine and Senior Fellow in the Center for the Study of Aging and Human Development at Duke University Medical Center. She also serves as Associate Director for Education and Evaluation of the Geriatrics Research, Education, and Clinical Center at the Durham VA Medical Center. Dr. Bales received her doctorate in Nutritional Sciences at the University of Tennessee-Knoxville and held an appointment in the Graduate Nutrition Division at the University of Texas-Austin before joining the faculty at Duke. She maintains a clinical research laboratory at Duke's Center for Aging and is a well-recognized expert in the field of nutrition and aging. She has published broadly on such topics as obesity in later life, nutritional frailty, nutritional interventions for chronic diseases in aging, and calorie restriction as a modifier of the aging process. She is the recipient of numerous awards, including the Grace Goldsmith Award and Max K. Horwitt Distinguished Lectureship and has served on numerous grant review panels and editorial boards. She currently serves the American Society for Nutrition as Chair of the Medical Nutrition Council and is a past-president of the American College of Nutrition. Dr. Bales also serves as the Editor of the *Journal of Nutrition in Gerontology and Geriatrics*.

Dr. Julie L. Locher is a medical sociologist and health services researcher. She is Professor of Medicine, Public Health, and Medical Sociology at the University of Alabama at Birmingham (UAB). Her primary area of research focuses on social and environmental factors (including especially the roles of social support and community and healthcare practices and policies) that affect eating behaviors and nutrition-related health outcomes in different populations, especially older adults (particularly those who are frail and those who are transitioning from hospitals to home) and in cancer patients and survivors. She is the recipient of an Academic Career Leadership Award from the National Institute on Aging to build the UAB Program in Translational Nutrition and Aging.

Dr. Edward Saltzman is a physician specializing in clinical nutrition. He currently serves as Chair of the Department of Nutrition Sciences at the Friedman School of Nutrition Science and Policy at Tufts University and is a scientist in the Energy Metabolism Laboratory of the Jean Mayer USDA Human Nutrition Research Center at Tufts University. He also serves as Chief of the Division of Clinical Nutrition at Tufts Medical Center. Dr. Saltzman serves on the editorial board of the American Journal of Clinical Nutrition as well as other journals, and is a member of the board of the directors of the American Society for Nutrition.

This updated text, containing 23 chapters that integrate clinical practice and the underlying science, continues to include many unique features, such as highly relevant descriptions of case studies that help to illustrate the complexity of treating the aging patient. The volume is relevant for practicing as well as research-centered healthcare professionals as there are in-depth discussions of the basic assessment tools; demographics of the different diseases and disease conditions that affect the aging population preferentially, such as age-related macular degeneration and alterations in muscle metabolism that are often adversely affected by the aging process. There are also clear, concise recommendations about dietary intakes and use of drugs and supplements across the stages of aging. Thus, this volume provides a broad base of knowledge concerning the changes in anatomy, physiology, and pathology associated with growing older. This volume is appropriate for health professionals, students, and faculty who have an interest in the latest, up-to-date information on the consequences of loss of certain body functions, treatment of infectious, as well as the chronic diseases of aging and disease-related morbidity and mortality.

This volume serves a dual purpose of providing in-depth focus on the nutritional aspects of treating elderly individuals throughout the last 30 or more years of their lifespan who have lost some of their organ/tissue functions, as well as examining the current clinical modalities used in treating a number

of the chronic diseases of aging and the consequences of the treatments on nutritional status. The book is organized as a stand-alone resource text that provides the basics of nutritional assessment of the patient, intervention in healthy elderly and patients, and reflects upon the necessity of medical nutrition support as a keystone for disease management. The volume includes extensive, in-depth chapters covering the most important aspects of the complex interactions between diet, obesity, cardiovascular disease, diabetes, and loss of cognitive functions, development of sarcopenia, as examples, and the impact of loss of certain functions on nutritional status.

Handbook of Clinical Nutrition and Aging, Third Edition is organized into four relevant sections. The three introductory chapters in the first part, entitled "Overarching Issues for Nutritional Well-Being in Later Life" provide readers with the basics so that the more clinically related chapters can be easily understood. The first chapter describes the process of aging as a time in the life cycle where there are progressive physical, cognitive, and psychosocially related functional declines resulting in disproportionately high healthcare utilization especially in the final years of life. The chapter, that includes over 100 references, uses a multilevel approach to describe factors influencing the nutritional status and food intake of older adults, explores major concepts in the planning, implementation, and evaluation of nutrition interventions specific to older adults, provides examples of current and emerging nutrition interventions, and discusses future directions while taking into consideration the heterogeneity and varied needs of aging individuals. The next chapter reviews the importance of using systematic review processes, such as meta-analyses, to better understand the totality of the evidence concerning nutritional issues and the aging population. The basic steps in conducting such reviews are examined. Unique issues specific to nutrition-related topics include baseline exposure, nutrient status, nutrient bioavailability, nutrient bioequivalence, biological stores, multiple biological functions, undefined nature of nutrient intervention, and uncertainties in assessing dose-response relationships are discussed. The third chapter reviews and recommends nutrition assessment tools for screening elderly at risk for undernutrition including the Mini-Nutritional Assessment-Short Form and the Determine Checklist. These tools are recommended for evaluation of patients upon admission to the hospital and long-term care settings, upon first home care visit, enrollment in community-based programs and during annual primary care visits. The chapter describes the assessment exam which is a comprehensive exam focused on assessment of weight changes, body composition status, physical examination, comorbid conditions, functional status, medication history, dietary intake, and psychosocial and economic status. Additionally, adult disease-related malnutrition diagnoses applicable to older adults including starvation, chronic disease-related malnutrition, and acute disease-related malnutrition are described in detail.

The next six chapters (Chaps. 4–9) describe the fundamentals of nutrition with regard to geriatric syndromes. The next chapter, containing over 100 up-to-date references, reviews the major effects of aging on the eye. Both individual essential nutrients and foods containing these nutrients are examined with regard to primary and secondary reduction in risk of cataracts and age-related macular degeneration. Chapter 5 identifies and tabulates the critical requirement of the evaluation of the nutritional status of the geriatric patient who has suffered from adverse changes in the oral cavity. The chapter includes extensive discussions of edentulism, dental caries, periodontitis, oral cancer, and their effects on consumption of certain nutrients and dietary patterns. Chapters 6 and 7 examine the importance of voluntary, skeletal muscle function in the aging population and also the effects of obesity on muscle status. In aging, there may be the loss of muscle mass and/or muscle strength. Sarcopenia refers to the process of age-related skeletal muscle mass loss, and dynapenia refers to the process of age-related muscle strength loss. The chapter reviews the well-designed longitudinal studies that show little or no association between sarcopenia and the risk of impaired physical function, falls, and mortality. Conversely, dynapenia is consistently associated with these outcomes. Either or both of these losses of muscle function can be seen in the obese geriatric patient. Unfortunately, weight loss is associated with both muscle mass and muscle strength loss. The chapter includes a discussion of the benefits of resistance exercise as it is one of the most promising approaches for

preventing and treating both sarcopenia and dynapenia in obese and non-obese older adults. Muscle physiology is further explored in the next chapter, containing over 100 relevant references, that contrasts the effects of different types of exercise on the aging skeletal muscle. The authors conclude that endurance exercise should be recommended for the elderly as this type of exercise increases both mitochondrial and myofibrillar protein synthesis. Chapter 8 examines the potential for certain nutrients and dietary constituents to reduce the risk of cognitive decline and reviews the findings from published survey studies and intervention trials. Substances reviewed include chocolate, omega-3 fatty acids, B vitamins, vitamin D, and the Mediterranean diet. The chapter includes over 180 targeted references. The final chapter in this section, Chap. 9, discusses the consequences of food insecurity for the elderly population. The chapter reviews the public health responses to dealing with food insecurity among older adults in the United States. There is an extensive discussion of the need for a multipronged approach, including continued support for government programs, such as the nutrition services provided for by the Older Americans Act and the Supplemental Nutrition Assistance Program, and private sector programs, such as food banks.

The third section, entitled "Common Clinical Conditions" contains ten practice-oriented chapters covering topics from pressure ulcers to end of life nutritional decisions.

Chapters 10 and 11 review the effects of obesity, diabetes, and the metabolic syndrome and the potential to develop strategies that can improve activities of daily living while reducing the risk of loss of skeletal muscle functions as described in an earlier chapter. We learn that there is an age-related increase in the prevalence of all of the components of the metabolic syndrome. Relative to those who are <35 years of age, the odds of having metabolic syndrome for those who are over 65 years are almost six times greater in men and about five times greater in women. Increased abdominal fat is independently associated with metabolic syndrome in adults aged 70-79 years. The prevalence of type 2 diabetes mellitus is highest in older persons. The age-related increase in visceral fat is associated with the increased prevalence of diabetes mellitus and insulin resistance in the elderly. Chapter 11 emphasizes the consequences of diabetes in the elderly and the potential for beneficial interventions. The aim of diabetes intervention is to prevent or delay the development of long-term complications of high blood glucose and related metabolic abnormalities and improve the quality of life. The chapter includes over 100 references that are reviewed with regard to diabetes-related comorbidities including diabetic retinopathy, cardiovascular disease, peripheral vascular disease, and congestive heart failure that can limit activities of daily living, including transportation, shopping for food, and ability to read food labels and restaurant menus. The next two chapters follow logically and examine the effects of aging on the heart and the potential for preventive cardiology. The authors explain that preventive cardiology focuses on the prevention of future cardiovascular disease by reducing the burden of known cardiovascular risk factors such as hypertension, dyslipidemia, glucose intolerance, and obesity. Special concerns that affect implementation of cardiac prevention strategies in older at-risk adults include complicated comorbidities; functional limitations; alterations in taste, smell, and appetite, difficulties with medication use/effectiveness; and limited financial, social, and/or caregiver resources. Preventive cardiology nutrition recommendations stress increasing the intake of fruits, vegetables, whole grains, and omega-3 and omega-6 fatty acids while substituting non-hydrogenated unsaturated fats for saturated and trans-fats and minimizing the intake of beverages and foods with added sugar. The next chapter examines a major consequence of cardiac disease in the elderly and the potential nutritional interventions for secondary prevention of further adverse effects. We learn that heart failure is the leading cause of hospitalization in the elderly. The prognosis for established heart failure in those over age 65 is poor with 5-year survival rates of less than 50 % in both men and women. Unintentional weight loss in heart failure is likely due to both increased energy utilization and decreased availability of fat, protein, and carbohydrates despite normal caloric intake. Recommendations include moderate dietary sodium restriction and supplementation with potassium, calcium, and/or magnesium if adequate amounts cannot be obtained from the diet. The authors, who have reviewed the literature extensively and cite almost 150 references, note that the importance of most vitamins and other micronutrients in the pathogenesis and treatment of chronic heart failure has not been well characterized.

Chapter 14 examines the relationship between nutritional status and the development and treatment of pressure ulcers that are frequently seen in elderly with limited mobility. There is a consistent epidemiological association between nutritional status and the incidence, progression, and severity of pressure sores. At present, however, there are inconsistent data from intervention trials with macroand/or micronutrients that may be linked to underlying diseases that result in the synthesis of cachexiainducing molecules.

Chapter 15 examines the conflicting data concerning the provision of nutrition support defined as the prescription and administration of enteral or parenteral nutrition to patients with incurable cancer at the last stages of their disease. The authors provide a sensitive analysis of this emotional as well as nutritionally related medical dilemma. Chronic kidney disease can be a serious adverse effect of diabetes and/or cardiovascular disease. Chapter 16 reviews the importance of nutritional support for the older patient with this disease that is often further complicated by other disease processes. There are discussions of both the development and progression of chronic kidney disease that is influenced by a number of dietary factors including salt, protein intake, and energy balance. High-protein intakes are detrimental to individuals with mild impairment of renal function. Specific dietary recommendations regarding protein level and intakes of sodium, phosphorus, potassium, and fluids are reviewed in detail. Osteoporosis is another chronic disease of aging that has a critical nutritional component that is reviewed in the next chapter (Chap. 17). The roles of calcium and vitamin D are discussed in detail and recommendations for supplementation levels are included. Bone physiology is stressed within the context of the balance between bone formation and bone loss that continues to favor bone loss throughout most of adult life. Chapter 18 provides practice-based recommendations for the nutrition management of dementia patients in long-term care facilities. The chapter concentrates on one of the major risk factors for experiencing nutritional problems in nursing homes which is cognitive impairment and/or severe dementia. Approximately 65 % of nursing home residents have documented moderate to severe dementia. The chapter provides a framework for reducing the risk of undernutrition due to mealtime difficulties. The final chapter in this section, Chap. 19, is related to the sensitive discussions in Chap. 15 concerning nutritional support in end-stage cancer patients. Chapter 19 examines the role of nutritional support at the end of life. The authors remind us that many of those emotionally attached to the terminal patient consider nutrition and fluid intake as an essential part of life and not a part of medical treatment. However, case law in the United States considers artificial nutrition (both enteral and parenteral) to be medical treatment and thus recognizes a patient's right to refuse artificial nutritional support. The chapter includes a discussion of feeding options as a key component of the overall goals of care for the patient, whether it is for curative treatment, rehabilitative treatment, or comfort-focused treatment.

The last section, that reviews contemporary diet-focused concerns, contains four insightful chapters. There has been a great deal of interest in the potential for calorie restriction to increase the lifespan. Chapter 20 looks at whether there are any antiaging effects of nutritional modification. The chapter reviews the data from earlier studies that examined the biochemical consequences of calorie restriction including reductions in oxidative biomarkers and insulin resistance. The chapter includes a detailed discussion of the ongoing intervention study, CALERIE, which is a randomized controlled trial that is testing the effects of prolonged calorie restriction in humans on biomarkers of aging and the rate of living theory hypothesis. Chapter 21 reviews the high-risk nutrients in the aging population. This comprehensive chapter that includes almost 150 references indicates that seniors have lower energy requirements and higher requirements for some nutrients. Population data indicate inadequate intakes of macronutrients including protein, n-3 fatty acids, and dietary fiber; vitamins B6, B12, D, E, and carotenoids; calcium, magnesium, and potassium. Individual supplements may not be as effective as obtaining the nutrients from whole foods. The exceptions are reviewed and include vitamin D and vitamin B12. Foods that are emphasized include fruit, vegetables, legumes, whole grains, nuts or seeds, fish, lean meat, poultry, and low-fat fluid dairy products. The final two chapters in this comprehensive volume examine the role of physical activity and dietary supplements. The rationale for including physical activity in the management of age-related chronic diseases is provided in Chap. 22. There are discussions of cardiovascular disease, cancer, osteoporosis, obesity, diabetes, age-related loss of physical function and frailty, as well as the importance of exercise for improving mood and reducing the risk of depression. Public health guidelines for physical activity are reviewed and several are described. Specific programs that include aerobic exercise, resistance training, flexibility and balance exercises for older adults to decrease the risk of falls are reviewed based upon data from evidence-based programs. There is an emphasis on community-based programs that include both physical activity and nutrition programs. The last chapter in the volume provides an encyclopedic examination of the dietary supplements available in the United States and their use in seniors. The chapter, containing almost 250 references, includes detailed discussions of the demographics of supplement use, the regulatory structure, the claims made on supplements, and the evidence of both benefit and risk of using dietary supplements by seniors in the United States.

The above description of the volume's 23 chapters attests to the depth of information provided by the 49 well-recognized and respected chapter authors. Each chapter includes complete definitions of terms with the abbreviations fully defined for the reader and consistent use of terms between chapters. The volume includes over 75 detailed tables and informative figures, an extensive, detailed index, and more than 2,300 up-to-date references that provide the reader with excellent sources of worthwhile information.

In conclusion, *Handbook of Clinical Nutrition and Aging, Third Edition*, edited by Connie W. Bales Ph.D., RD, Julie L. Locher Ph.D., MSPH, and Edward Saltzman, M.D. provides health professionals in many areas of research and practice with the most up-to-date, well-referenced volume on the importance of maintaining the nutritional status of the elderly patient regardless of cause. The volume's chapters carefully document the critical value of medical nutrition evaluation, treatment support, and management for patients with many of the chronic diseases of aging that are extensively reviewed. This volume will serve the reader as the benchmark in this complex area of interrelation-ships between diet, nutritional, and non-nutritional supplements and the specific products for maintaining the nutritions are clearly delineated so that students as well as practitioners can better understand the complexities of these interactions. Unique chapters that examine the effects of the aging process on critical organ systems such as the eye, bone, brain, and heart are included along with resources for enhancing behaviors that can increase patient adherence to nutritional therapies. The editors are applauded for their efforts to develop the most authoritative resource in the field to date and this excelent text is a very welcome addition to the Nutrition and Health Series.

Adrianne Bendich, Ph.D., F.A.C.N., F.A.S.N. Series Editor

About the Series Editor



Dr. Adrianne Bendich, Ph.D., F.A.S.N., F.A.C.N. has served as the "Nutrition and Health" Series Editor for over 15 years and has provided leadership and guidance to more than 120 volume editors who have developed the 60+ well respected and highly recommended volumes in the Series.

In addition to "Handbook of Clinical Nutrition and Aging, Third Edition" edited by Connie W. Bales Ph.D., RD, Julie L. Locher Ph.D., MSPH, and Edward Saltzman, M.D. major new editions in 2012–2014 include:

- Nutrition and Oral Medicine, Second Edition, edited by Dr. Riva Touger-Decker, Dr. Connie C. Mobley, and Dr. Joel B. Epstein, 2014
- 2. Fructose, High Fructose Corn Syrup, Sucrose and Health, edited by Dr. James M. Rippe, 2014
- Nutrition in Kidney Disease, Second Edition, edited by Dr. Laura D. Byham-Gray, Dr. Jerrilynn D. Burrowes, and Dr. Glenn M. Chertow, 2014
- 4. *Handbook of Food Fortification and Health, volume I* edited by Dr. Victor R. Preedy, Dr. Rajaventhan Srirajaskanthan, and Dr. Vinood B. Patel, 2013
- 5. *Handbook of Food Fortification and Health, volume II* edited by Dr. Victor R. Preedy, Dr. Rajaventhan Srirajaskanthan, and Dr. Vinood B. Patel, 2013
- 6. *Diet Quality: An Evidence-Based Approach, volume I* edited by Dr. Victor R. Preedy, Dr. Lan-Ahn Hunter, and Dr. Vinood B. Patel, 2013
- 7. *Diet Quality: An Evidence-Based Approach, volume II* edited by Dr. Victor R. Preedy, Dr. Lan-Ahn Hunter, and Dr. Vinood B. Patel, 2013

- 8. *The Handbook of Clinical Nutrition and Stroke*, edited by Mandy L. Corrigan, MPH, RD Arlene A. Escuro, MS, RD, and Donald F. Kirby, MD, FACP, FACN, FACG, 2013
- 9. *Nutrition in Infancy, volume I* edited by Dr. Ronald Ross Watson, Dr. George Grimble, Dr. Victor Preedy, and Dr. Sherma Zibadi, 2013
- 10. *Nutrition in Infancy, volume II* edited by Dr. Ronald Ross Watson, Dr. George Grimble, Dr. Victor Preedy, and Dr. Sherma Zibadi, 2013
- 11. Carotenoids and Human Health, edited by Dr. Sherry A. Tanumihardjo, 2013
- 12. *Bioactive Dietary Factors and Plant Extracts in Dermatology*, edited by Dr. Ronald Ross Watson and Dr. Sherma Zibadi, 2013
- 13. *Omega 6/3 Fatty Acids*, edited by Dr. Fabien De Meester, Dr. Ronald Ross Watson, and Dr. Sherma Zibadi, 2013
- 14. Nutrition in Pediatric Pulmonary Disease, edited by Dr. Robert Dumont and Dr. Youngran Chung, 2013
- 15. Magnesium and Health, edited by Dr. Ronald Ross Watson and Dr. Victor R. Preedy, 2012.
- 16. *Alcohol, Nutrition and Health Consequences*, edited by Dr. Ronald Ross Watson, Dr. Victor R. Preedy, and Dr. Sherma Zibadi, 2012
- 17. Nutritional Health, Strategies for Disease Prevention, Third Edition, edited by Norman J. Temple, Ted Wilson, and David R. Jacobs, Jr., 2012
- Chocolate in Health and Nutrition, edited by Dr. Ronald Ross Watson, Dr. Victor R. Preedy, and Dr. Sherma Zibadi, 2012
- 19. *Iron Physiology and Pathophysiology in Humans*, edited by Dr. Gregory J. Anderson and Dr. Gordon D. McLaren, 2012

Earlier books included Vitamin D, Second Edition edited by Dr. Michael Holick; "Dietary Components and Immune Function" edited by Dr. Ronald Ross Watson, Dr. Sherma Zibadi, and Dr. Victor R. Preedy; "Bioactive Compounds and Cancer" edited by Dr. John A. Milner and Dr. Donato F. Romagnolo; "Modern Dietary Fat Intakes in Disease Promotion" edited by Dr. Fabien De Meester, Dr. Sherma Zibadi, and Dr. Ronald Ross Watson; "Iron Deficiency and Overload" edited by Dr. Shlomo Yehuda and Dr. David Mostofsky; "Nutrition Guide for Physicians" edited by Dr. Edward Wilson, Dr. George A. Bray, Dr. Norman Temple, and Dr. Mary Struble; "Nutrition and Metabolism" edited by Dr. Christos Mantzoros; and "Fluid and Electrolytes in Pediatrics" edited by Leonard Feld and Dr. Frederick Kaskel. Recent volumes include: "Handbook of Drug-Nutrient Interactions" edited by Dr. Joseph Boullata and Dr. Vincent Armenti; "Probiotics in Pediatric Medicine" edited by Dr. Sonia Michail and Dr. Philip Sherman; "Handbook of Nutrition and Pregnancy" edited by Dr. Carol Lammi-Keefe, Dr. Sarah Couch, and Dr. Elliot Philipson; "Nutrition and Rheumatic Disease" edited by Dr. Laura Coleman; "Nutrition and Kidney Disease" edited by Dr. Laura Byham-Grey, Dr. Jerrilynn Burrowes, and Dr. Glenn Chertow; "Nutrition and Health in Developing Countries" edited by Dr. Richard Semba and Dr. Martin Bloem; "Calcium in Human Health" edited by Dr. Robert Heaney and Dr. Connie Weaver; and "Nutrition and Bone Health" edited by Dr. Michael Holick and Dr. Bess Dawson-Hughes.

Dr. Bendich is President of Consultants in Consumer Healthcare LLC, and is the editor of ten books including "*Preventive Nutrition: The Comprehensive Guide for Health Professionals, Fourth Edition*" co-edited with Dr. Richard Deckelbaum (www.springer.com/series/7659). Dr. Bendich serves on the Editorial Boards of the *Journal of Nutrition in Gerontology and Geriatrics*, and Antioxidants, and has served as Associate Editor for "Nutrition" the International Journal; served on the Editorial Board of the *Journal of Women's Health and Gender-based Medicine*, and served on the Board of Directors of the American College of Nutrition.

Dr. Bendich was Director of Medical Affairs at GlaxoSmithKline (GSK) Consumer Healthcare and provided medical leadership for many well-known brands including TUMS and Os-Cal. Dr. Bendich had primary responsibility for GSK's support for the Women's Health Initiative (WHI) intervention study. Prior to joining GSK, Dr. Bendich was at Roche Vitamins Inc. and was involved with the groundbreaking clinical studies showing that folic acid-containing multivitamins significantly reduced major classes of birth defects. Dr. Bendich has coauthored over 100 major clinical research studies in the area of preventive nutrition. She is recognized as a leading authority on antioxidants, nutrition and immunity and pregnancy outcomes, vitamin safety, and the cost-effectiveness of vitamin/mineral supplementation.

Dr. Bendich received the Roche Research Award, is a *Tribute to Women and Industry* Awardee, and was a recipient of the Burroughs Wellcome Visiting Professorship in Basic Medical Sciences. Dr. Bendich was given the Council for Responsible Nutrition (CRN) Apple Award in recognition of her many contributions to the scientific understanding of dietary supplements. In 2012, she was recognized for her contributions to the field of clinical nutrition by the American Society for Nutrition and was elected a Fellow of ASN. Dr Bendich is an Adjunct Professor at Rutgers University. She is listed in Who's Who in American Women.

About Volume Editors



Dr. Connie Watkins Bales is a Professor in the Division of Geriatrics in the Department of Medicine and Senior Fellow in the Center for the Study of Aging and Human Development at Duke University Medical Center. She also serves as Associate Director for Education and Evaluation of the Geriatrics Research, Education, and Clinical Center at the Durham VA Medical Center. Dr. Bales received her doctorate in Nutritional Sciences at the University of Texnessee-Knoxville and held an appointment in the Graduate Nutrition Division at the University of Texas-Austin before joining the faculty at Duke. She maintains a clinical research laboratory at Duke's Center for Aging and is a well-recognized expert in the field of nutrition and aging. She has published broadly on such topics as obesity in later life, nutritional frailty, nutritional interventions for chronic diseases in aging, and calorie restriction as a modifier of the aging process. She is the recipient of numerous awards, including the Grace Goldsmith Award and Max K. Horwitt Distinguished Lectureship, and has served on numerous grant review panels and editorial boards. She currently serves the American Society for Nutrition as Chair of the Medical Nutrition Council and is a past-president of the American College of Nutrition. Dr. Bales also edits the *Journal of Nutrition in Gerontology and Geriatrics*.



Dr. Julie L. Locher is a medical sociologist and health services researcher. She is Professor of Medicine, Public Health, and Medical Sociology at the University of Alabama at Birmingham. Her primary area of research focuses on social and environmental factors (including especially the roles of social support and community and healthcare practices and policies) that affect eating behaviors and nutrition-related health outcomes in different populations, especially older adults (particularly those who are frail and those who are transitioning from hospitals to home) and in cancer patients and survivors. She is the recipient of an Academic Career Leadership Award from the National Institute on Aging to build the UAB Program in Translational Nutrition and Aging. Dr. Locher's contributions were supported by Grant #K07AG043588 (Translational Nutrition and Aging Research Academic Career Leadership Award) from the National Institute on Aging.



Dr. Edward Saltzman is a physician specializing in clinical nutrition. He currently serves as Chair of the Department of Nutrition Sciences at the Friedman School of Nutrition Science and Policy at Tufts University and is a scientist in the Energy Metabolism Laboratory of the Jean Mayer USDA Human Nutrition Research Center at Tufts University. He also serves as Chief of the Division of Clinical Nutrition at Tufts Medical Center. Dr. Saltzman serves on the editorial board of the American Journal of Clinical Nutrition as well as other journals, and is a member of the board of the directors of the American Society for Nutrition.

Contents

Part I Overarching Issues for Nutritional Well-Being in Later Life

1	Nutrition Interventions for Aging Populations Jylana L. Sheats, Sandra J. Winter, and Abby C. King	3		
2	Systematic Reviews in the Field of Nutrition	21		
3	Nutrition Assessment Rose Ann DiMaria-Ghalili and Michi Yukawa	35		
Par	Part II Fundamentals of Nutrition and Geriatric Syndromes			
4	Nutrition and the Aging Eye Elizabeth J. Johnson	57		
5	Nutrition and Oral Health: A Two-Way Relationship Cristina Palacios and Kaumudi J. Joshipura	81		
6	Loss of Muscle Mass and Muscle Strength in Obese and Nonobese Older Adults Danielle R. Bouchard and Ian Janssen	99		
7	Muscle Metabolism, Nutrition, and Functional Status in Older Adults Douglas Paddon-Jones and Aaron P. Russell	113		
8	Nutrition in the Prevention and Treatment of Cognitive Decline Grace E. Giles, Kristen E. D'Anci, and Robin B. Kanarek	125		
9	Food Insecurity and Hunger Among Older Adults David R. Buys and Julie L. Locher	147		
Par	Part III Common Clinical Conditions			
10	Obesity in Older Adults and Strategies for Weight Management Dennis T. Villareal and Krupa Shah	163		
11	Nutrition and Lifestyle Change in Older Adults with Diabetes Mellitus and Metabolic Syndrome Barbara Stetson, Holly M. Knight, and Sri Prakash L. Mokshagundam	179		

12	Preventive Cardiology: Counseling Older At-Risk Adults on Nutrition	203
13	Chronic Heart Failure Justin M. Vader, Christopher L. Holley, and Michael W. Rich	215
14	The Relationship of Nutrition and Pressure Ulcers David R. Thomas	237
15	Nutrition Support in Solid Tumor Cancer Patients Krishna A. Patel and Aminah Jatoi	253
16	Nutrition and Chronic Kidney Disease Xiaorui Chen and Srinivasan Beddhu	261
17	Nutritional Concerns in Osteoporosis Bess Dawson-Hughes	273
18	Dementia-Related Mealtime Difficulties: Assessment and Management in the Long-Term Care Setting Melissa Batchelor-Aselage, Elaine J. Amella, Sarah Broome Rose, and Connie Watkins Bales	287
19	Nutrition at the End of Life Michi Yukawa and Christine Seel Ritchie	303
Par	t IV Contemporary Diet-Focused Concerns	
20	Anti-aging Effects of Nutritional Modification: The State of the Science on Calorie Restriction L. Anne Gilmore, Eric Ravussin, and Leanne M. Redman	315
21	High-Risk Nutrients in the Aging Population Katherine L. Tucker	335
22	Physical Activity and Exercise: Important Complements to Nutrition in Older Adults Anne O. Brady, Alison Clune Berg, Mary Ann Johnson, and Ellen M. Evans	355
23	Dietary Supplements in Older Adults Johanna T. Dwyer, Rebecca Costello, Joyce Merkel, and Paul M. Coates	375
Ind	ex	425

Contributors

Elaine J. Amella, Ph.D., R.N., F.A.A.N. College of Nursing, Medical University of South Carolina, Charleston, SC, USA

Connie Watkins Bales, Ph.D., R.D. Durham VA Medical Center and Duke University Medical Center, Durham, NC, USA

Melissa Batchelor-Aselage, Ph.D., R.N.-B.C., F.N.P.-B.C. School of Nursing, Duke University, Durham, NC, USA

Srinivasan Beddhu, M.D. Medical Service, Salt Lake Veterans Affairs Healthcare System and Division of Nephrology & Hypertension, University of Utah School of Medicine, Salt Lake City, UT, USA

Alison Clune Berg, B.S.F.C.S., M.S. Department of Foods and Nutrition, University of Georgia, Athens, GA, USA

Danielle R. Bouchard, Ph.D. Department of Kinesiology and Recreation Management, University of Manitoba, Winnepeg, MB, Canada

Anne O. Brady, B.S., M.S., Ph.D. Department of Kinesiology, University of North Carolina at Greensboro, Greensboro, NC, USA

David R. Buys, Ph.D. Department of Food Science, Nutrition, and Health Promotion State Health Specialist, MSU Extension Service Mississippi State University, MS, USA

Xiaorui Chen, M.S. Department of Nephrology, University of Utah, Salt Lake City, UT, USA

Paul M. Coates, Ph.D. Office of Dietary Supplements, National Institutes of Health, Bethesda, MD, USA

Rebecca Costello, Ph.D. Office of Dietary Supplements, National Institutes of Health, Bethesda, MD, USA

Kristen E. D'Anci, Ph.D. Department of Psychology, Salem State University, Salem, MA, USA Department of Psychology, Tufts University, Medford, MA, USA

Bess Dawson-Hughes, M.D. Bone Metabolism Laboratory, Jean Mayer USDA Human Nutrition Research Center on Aging, Tufts University, Boston, MA, USA

Rose Ann DiMaria-Ghalili, Ph.D., R.N., F.A.S.P.E.N., C.N.S.C. Doctoral Nursing Department, College of Nursing and Health Professions, Drexel University, Philadelphia, PA, USA

Johanna T. Dwyer, D.Sc. R.D. Office of Dietary Supplements, National Institutes of Health, Bethesda, MD, USA

Ellen M. Evans, Ph.D. Department of Kinesiology, University of Georgia, Athens, GA, USA

Grace E. Giles, M.S. Department of Psychology, Tufts University, Medford, MA, USA

L. Anne Gilmore, Ph.D. Department of Reproductive Endocrinology and Woman's Health, Pennington Biomedical Research Center, Baton Rouge, LA, USA

Christopher L. Holley, M.D., Ph.D. Cardiovascular Division, Department of Internal Medicine, Barnes-Jewish Hospital, St. Louis, MO, USA

Ian Janssen, Ph.D. Department of Community Health and Epidemiology, Queen's University, Kingston, ON, Canada

Aminah Jatoi, M.D. Department of Oncology, Mayo Clinic, Rochester, MN, USA

Gordon L. Jensen, M.D., Ph.D. Department of Nutritional Sciences, Penn State University, University Park, PA, USA

Mary Ann Johnson, Ph.D. Department of Foods and Nutrition, University of Georgia, Athens, GA, USA

Elizabeth J. Johnson, Ph.D. Jean Mayer USDA Human Nutrition Research Center on Aging, Tufts University, Boston, MA, USA

Kaumudi J. Joshipura, B.D.S., M.S., Sc.D. Center for Clinical Research and Health Promotion, School of Dental Medicine, University of Puerto Rico, San Juan, Puerto Rico

Robin B. Kanarek, Ph.D. Department of Psychology, Tufts University, Medford, MA, USA

Abby C. King, Ph.D. Health Research and Policy, and the Stanford Prevention Research Center, Stanford University School of Medicine, Stanford, CA, USA

Holly M. Knight, B.A. Department of Psychological and Brain Sciences, University of Louisville, Louisville, KY, USA

William E. Kraus, M.D. Division of Cardiology, Department of Medicine, Duke University School of Medicine, Durham, NC, USA

Alice H. Lichtenstein, D.Sc. Cardiovascular Nutrition Laboratory, Jean Mayer USDA HNRCA at Tufts University, Boston, MA, USA

Julie L. Locher, Ph.D. Translational Nutrition and Aging Program, Departments of Medicine and Health Care Organization, University of Alabama at Birmingham, Birmingham, AL, USA

Joyce Merkel, M.S., R.D. Office of Dietary Supplements, National Institutes of Health, Bethesda, MD, USA

Sri Prakash L. Mokshagundam, M.D. Division of Endocrinology and Metabolism, University of Louisville, Louisville, KY, USA

Douglas Paddon-Jones, Ph.D. Department of Nutrition and Metabolism, The University of Texas Medical Branch, Galveston, TX, USA

Cristina Palacios, L.N.D., M.Sc., Ph.D. Nutrition Program, Graduate School of Public Health, University of Puerto Rico, San Juan, Puerto Rico

Krishna A. Patel, M.D. Department of Internal Medicine, Mayo Clinic Rochester, Rochester, MN, USA

Julie D. Pruitt, M.S., R.D., L.D.N. Division of Cardiology, Department of Medicine, Duke University School of Medicine, Durham, NC, USA

Eric Ravussin, Ph.D. Department of Health and Human Performance, Pennington Biomedical Research Center, Baton Rouge, LA, USA

Leanne M. Redman, M.S., Ph.D. Department of Reproductive Endocrinology and Women's Health, Pennington Biomedical Research Center, Baton Rouge, LA, USA

Michael W. Rich, M.D. Washington University School of Medicine, St. Louis, MO, USA

Christine Seel Ritchie, M.D., M.S.P.H. Department of Medicine, University of California San Francisco, San Francisco, CA, USA

Sarah Broome Rose, M.S., R.D. Department of Molecular and Medical Genetics, Oregon Health and Science University, Portland, OR, USA

Aaron P. Russell, Ph.D. Centre for Physical Activity and Nutrition (C-PAN) Research, School of Exercise and Nutrition Sciences, Deakin University, Melbourne, VIC, Australia

Edward Saltzman, M.D. Jean Mayer USDA Human Nutrition Research Center on Aging, Tufts University, Boston, MA, USA

Krupa Shah, M.D., M.P.H. Department of Medicine and Orthopedics, Highland Hospital, University of Rochester School of Medicine and Dentistry, Rochester, NY, USA

Joseph R. Sharkey, Ph.D., M.P.H., R.D. Program for Research in Nutrition and Health Disparities, School of Rural Public Health, College Station, TX, USA

Jylana L. Sheats, Ph.D., M.P.H. Stanford Prevention Research Center, Stanford University School of Medicine, Palo Alto, CA, USA

Barbara Stetson, Ph.D. Department of Psychological and Brain Sciences, University of Louisville, Louisville, KY, USA

David R. Thomas, M.D., F.A.C.P., A.G.S.F., Division of Geriatric Medicine, Department of Internal Medicine, Saint Louis University Hospital, Saint Louis, MO, USA

Katherine L. Tucker, Ph.D. Department of Clinical Laboratory and Nutritional Sciences, University of Massachusetts, Lowell, MA, USA

Justin M. Vader, M.D. Cardiovascular Division, Department of Internal Medicine, Washington University School of Medicine, St. Louis, MO, USA

Dennis T. Villareal, M.D. Department of Medicine, Baylor College of Medicine and Michael E. DeBakey Va Medical Center, Houston, TX, USA

Sandra J. Winter, Ph.D. Stanford Prevention Research Center, Stanford University School of Medicine, Palo Alto, CA, USA

Michi Yukawa, M.D., M.P.H. Division of Geriatrics, Department of Medicine, San Francisco VA Medical Center, University of California San Francisco, San Francisco, CA, USA

Part I Overarching Issues for Nutritional Well-Being in Later Life

Chapter 1 Nutrition Interventions for Aging Populations

Jylana L. Sheats, Sandra J. Winter, and Abby C. King

Key Points

- Optimal nutrition is essential for successful and healthy aging, as advancing age increased the risk of chronic disease and other health concerns.
- Aging populations are heterogeneous and require tailored, theory-based interventions to help ensure that their nutritional challenges are appropriately met.
- Recognition of issues specific to the population and using a variety of intervention delivery channels and strategies, will assist in the development of appropriate and clear intervention goals, measures, and outcomes.
- Nutrition-related behaviors are complex; thus, interventions need to be evaluated before, during, and after implementation.
- Special populations such as racial/ethnic minorities, persons with low socio-economic status, those with physical or cognitive impairments or limited access to food stores and food-related services, as well as persons who live in rural communities, may present unique nutrition challenges.
- Technology-based health promotion and intervention strategies may help reduce the burden of nutrition-related disease.

Keywords Older adults • Nutrition • Diet • Intervention • Health promotion • Health behavior change • Theory • Ecological model • Technology

J.L. Sheats, Ph.D., M.P.H.

S.J. Winter, Ph.D.

Stanford Prevention Research Center, Stanford University School of Medicine, 1070 Arastradero Road, Suite 100, Palo Alto, CA 94304-1334, USA e-mail: sheats@stanford.edu

Stanford Prevention Research Center, Stanford University School of Medicine, 1070 Arastradero Road, Suite 300, Palo Alto, CA 94304, USA e-mail: sjwinter@stanford.edu

A.C. King, Ph.D.

Health Research and Policy, and the Stanford Prevention Research Center, Stanford University School of Medicine, 259 Campus Drive, HRP Redwood Building, Room T221, Stanford, CA 94305-5405, USA e-mail: king@stanford.edu

Introduction

Understanding Global Aging Trends, Successful Aging, and the Role of Nutrition

A major accomplishment of industrialized countries in the twentieth century has been vast improvements in human life expectancy [1]. In the United States it has been projected that the number of older adults, defined as individuals aged 65 years or older [2], will double from the current estimate of 40.2 million to 88.5 million [3]; and by 2030 the population of oldest-old Americans (i.e., aged 85 years or older) could grow to 19 million [3]. The "graying of America" is attributed to people living longer [1], the aging of the "baby boomer" cohort (those born between 1946 and 1964), and immigration [3]. It is also expected that there will be a demographic shift toward a more racially and ethnically diverse population of older Americans, with minorities (i.e., all races and ethnic groups other than non-Hispanic White) becoming the majority by 2042 [3]. As in the United States, it has been predicted that by 2036 the population of older adults in Canada will double [4]. Comparable trends are being experienced by other industrialized parts of the world, such as Europe [5], Australia [6], New Zealand, and Japan [1].

Aging can be characterized as a time in the life cycle where one experiences progressive physical, cognitive, and psychosocially related functional declines resulting in disproportionately high healthcare utilization in the final years of life [7]. However, there are older adults who live healthy lives full of vitality [8]. According to the U.S. Administration on Aging [9], 41.6 % of noninstitutionalized older adults perceive their heath as "excellent" or "very good" relative to 64.5 % for individuals aged 18–64 years. Further, many adults aged 85 years or older are able to engage in Independent Activities of Daily Living (IADLs) [10], such as the ability to perform household tasks such as housework and laundry, shopping, preparing food, and handling finances. Positive health behaviors practiced early in life, such as routinely eating a healthful diet, are essential for healthy, successful aging [8]. While definitions vary, essential characteristics of successful aging have included factors such as having high levels of physical and cognitive functioning, the absence of significant chronic disease and disability, and the ability to socially engage with others [11].

The primary goal for the nation's older adults, as documented by Healthy People 2020, is to "improve health, function, and quality of life" [12]. Given the approaching "silver tsunami" [13], optimal nutrition is key and has the ability to improve health outcomes, quality of life, and well-being [14]. Individuals are products of their environment, and the factors that influence food intake, the etiology of health conditions, and the support of positive health behaviors often involve factors beyond the individual, so a holistic multilevel approach is suggested—especially in regard to addressing the nutritional health of older adults [15–17]. Thus, researchers increasingly use ecological models that focus on individual factors, as well as factors within a broader environmental context that may contribute to positive or negative health behavior and/or health outcomes [18]. Ecological perspectives posit that intrapersonal (individual characteristics), interpersonal processes, institutional, community, and public policy have the potential to impact behavior [18]. This chapter will use a multilevel approach to describe factors influencing the nutritional status and food intake of older adults; explore major concepts in the planning, implementation, and evaluation of nutrition interventions specific to older adults; provide examples of current and emerging nutrition interventions; and discuss future directions while taking into consideration their heterogeneity and varied needs.

Nutritional Status of Older Adults

As individuals age, they experience various life transitions and are at increased risk for having health problems [19, 20]. Nutrition status is typically a self-managed condition wherein small self-care and lifestyle improvements can yield significant benefits [14, 21]. Yet, a pilot study conducted by the U.S.

Administration on Aging [22] found that 48 % and 43 % of respondents had moderate or high nutritional risk, respectively. With advancing age individuals may have lower energy intake requirements and require more nutrient-dense foods (i.e., foods that are high in nutrients, but low in calories) [23]. Subsequently, they may have higher requirements for foods and/or supplements rich in antioxidant nutrients, calcium, fiber, folic acid, zinc, and vitamins A, B₁₂, C, D, and E [8, 24] to address agerelated issues including, but not limited to, chronic disease, low bone density [8], poor vision [25], and cognitive impairment [26]. Studies have shown that older adults have problems meeting national recommendations for each food group-particularly, the grain group and dairy group, for example [27–30]. Further, of the six most frequently occurring health conditions reported among older adults (i.e., uncontrolled hypertension (34 %), diagnosed arthritis (50 %), all types of heart disease (32 %), any cancer (23 %), diabetes (19 %), and sinusitis (14 %)), four are nutrition-related chronic diseases. This is not surprising given that 80 % of older adults have at least one chronic disease diagnosis [9]; and 50 % have more than two [31]. The importance of a healthy, nutrientdense diet of appropriate quantity and quality is essential for health maintenance as well as the prevention and delay of chronic disease and disability for all sectors of the population, but particularly for older adults [32].

Factors Influencing Food Intake Among Older Adults

Food intake among older adults can be influenced not only by individual, intrapersonal, and social level factors, but also by the neighborhood environment, healthcare service, and policy context in which they reside.

Interpersonal and Intrapersonal Factors

At an interpersonal, or individual level, food intake and appetite may be influenced by life-long eating behaviors, poor health, disability, and altered sensory functions (e.g., sight, taste, and smell). Among potentially influential factors are taking multiple medications, drug–nutrient interactions and/or side effects, poor oral health and functionality, hormone-related changes related to the control of satiety, difficulty in preparing foods because of arthritic joints, inadequate portion control, lack of information related to individual dietary needs, and having a restrictive diet or complex dietary needs due to current health status or other factors [8, 14, 33]. Other factors to consider are poverty or limited financial resources, transportation issues affecting access to food stores and meal/nutrition programs, and lack of food storage and preparation facilities [34–38]. Intrapersonal factors cited in the literature include self-efficacy, perceived barriers, intentions, and social support [38, 39].

Social Factors

Food intake among older adults is also affected by social factors such as loss of role functions caused by changes in employment status (e.g., retirement), an altered family structure (e.g., "empty nest"), loss of independent living, loneliness and social isolation due to the death or departure of family members and/or friends, caregivers, socio-cultural and religious/spiritual influences [8, 34–36].

Neighborhood Environment, Healthcare Services, and Policy-Related Factors

Older adults may live in neighborhoods where the availability of food stores offering healthy foods is scarce [34]; or where there is limited availability of nutrition support programs and services [40]. Many of the agencies offering nutritional support report being unable to meet increased demand, so prioritizing those most in need of services is required [40]. In addition, healthcare services are increasingly being provided in ambulatory rather than acute care settings. This shift in healthcare service delivery, in combination with early hospital discharge policies, can strain the ability of sick or recovering older adults to obtain and prepare healthy food [40, 41]. Circumstances such as this place older adults at higher nutritional risk and have been associated with an increased prevalence of depression and food-related anxiety [41]. Ensuring that a nutrition component is part of hospital discharge planning may ameliorate this burden. Finally, it has been noted that policies that promote community gardens, locally grown produce, and farmers markets may be useful in improving access to more healthful foods [40].

In summary, an examination of factors influencing the diets of older adults is an area that is complex and not well understood [42]. Regardless of age, improved dietary intake and habits have the ability to improve nutritional status [21]. Yet, diet-related behavior change can be difficult and may take a significant amount of time and effort [21]. In addition to the factors described here, age-related health concerns and changing nutritional requirements present unique challenges for older adults and require the development of tailored nutrition interventions [43].

Helping Older Adults Meet Nutritional Challenges Through Intervention

Nutrition Intervention Planning and Design

Nutrition interventions are typically designed to improve and where appropriate increase intake and variety of foods and/or nutrients (including oral supplementation), provide nutrition education or counseling, and/or increase access and availability of healthful foods for individuals and/or communities. Successful nutrition interventions have the potential to reduce the burden of disease and improve quality of life and well-being [42]. Addressing nutrition challenges experienced by older adults through the development and implementation of effective nutrition interventions is thus important. For older adults, however, staying current with changes taking place in the field of nutrition promotion and recommendations can be confusing. Therefore, identifying and implementing evidence-based best practices specifically targeted and developed for this population are essential [40, 37, 44]. Recognition of issues specific to the target population will assist in the development of appropriate and clear intervention goals, measures, and outcomes.

Frameworks to Guide the Development and Implementation of Nutrition Interventions

In an effort to assist interventionists, Bartholomew and colleagues [45] identified a framework for mapping interventions, proposing six fundamental steps:

- 1. Conducting a needs assessment.
- 2. Creating change objectives based on behavioral determinants and environmental factors.
- 3. Selecting theory-based intervention methods and practical strategies.
- 4. Translating methods and theories into an organized program.
- 5. Planning for adoption, implementation, and sustainability of the program.
- 6. Creating an evaluation plan.

In utilizing this approach, a thorough review of the literature and formative research with the target population should be conducted. For older adults (i.e., community-dwelling, homebound, institutionalized), interventionists should consider the heterogeneity of the population and factors influencing food choice/selection and intake and preparation, as well as perceived motivators and barriers, nutrition risks, needs, preferences, and contextual factors [46]. An evaluation of these factors for both the older adult and their caregivers (who may also be aged) is essential for successful interventions [47, 48]. Assessment methods may include dietary recalls and other nutrition assessment and screening tools such as interviews, surveys, and structured group discussions/focus groups and home visits [41, 47]. Other effective intervention design principles have been identified by Freudenberg and colleagues [49], who suggest that successful interventions: be tailored to the target audience and setting; involve participants in planning, implementation, and evaluation; link participant health concerns to broader life concerns and society as whole; use existing environmental resources; build on the strengths of participants and their communities; advocate for policy changes to support the achievement of health objectives; prepare and empower participants; support diffusion of innovation; and institutionalize and replicate effective intervention components. These principles cut across different populations, settings, and strategies [49].

The use of health behavior change theory to inform and guide interventions has numerous benefits and is recommended [50]. Such benefits include obtaining a better understanding of possible determinants of both risks and helpful strategies, and identifying methods to help promote changes in the determinants, behavior, and environmental conditions surrounding the dietary areas being targeted [45]. Several health behavior change theories and models have been used, or are recommended to guide nutrition interventions for older adults, such as the Health Belief Model [51], Trans-Theoretical Model [52], Social Cognitive Theory [48], Theory of Planned Behavior [53], social support models [54], and ecological models [48, 54]. These theories contain constructs deemed highly applicable to aging populations. For example, social ecological models are particularly useful for nutrition interventions in that eating is often influenced by micro and macro-level factors [48]. The effectiveness of behavior change theory depends on its application as part of an intervention [55, 56]. Thus, there has been an expressed need for more guidance in the application of theory in practice [45].

Delivery Channels and Intervention Components

The type of nutrition intervention delivery channels used and the specific components introduced into a population can vary. It is critical, however, that the intervention mechanisms are preferred by the intended audience [45, 47]. Examples of preferred nutrition delivery channels for older adult populations include engaging videos, carefully designed PowerPoint presentations (e.g., with large, simple text, bright colors, use of images with elderly individuals, and generational music) and the provision of supplemental handouts, and newsletters or pamphlets designed to complement the education components of an intervention [57].

Nutrition intervention components designed for older adults that have resulted in positive health outcomes have included theory-based behavior modification practices; older adults' active involvement in determining intervention goals and having targeted, clear, simple, yet impactful and hard-to-ignore, practical and reinforced messages that are limited in number [15, 55]. Other effective means for promoting positive changes among older adults include integrating real-life scenarios into intervention content and processes; allowing for hands on, interactive learning to complement and promote an understanding of messages; fostering social support; providing access to relevant health-care professionals and resources; utilizing "geragogical" learning (i.e., guiding the learning of older adults in a way that is tailored to any special needs; for example, instructor-directed learning, supervised decision-making, person-centered activities); and applying relevant education and health behavior change theories [47, 55, 58–60]. Learning styles, abilities, and needs of older adults are often more varied compared to younger adults. To address this issue, the integration of multiple strategies can be used to enhance the quality and reach of nutrition interventions for older adults [21, 56, 57].

It is also important to focus on the robustness of intervention components to elicit behavior change and prevent Type II error (i.e., failing to find intervention effectiveness because the program is poorly designed or implemented) [45, 61]. Nutrition intervention design characteristics that increase the robustness of an intervention design highlighted by Kristal and Ollberding [62] include having a representative sample of potential participants; one or more measures of the evaluation outcome at the pre-intervention time point; one or more comparison groups (i.e., control group not receiving the intervention); randomized assignment to treatment (i.e. intervention group receiving the intervention); and one or more measures of the evaluation outcome at the post-intervention time point. While each component is important, all do not have to be present within an intervention, with the exception of the comparison group and post-intervention outcome measures [62].

Nutrition Interventions from an Ecological Perspective

In an assessment of behavioral health interventions Cutler [15] posits that individual behavior change cannot occur without environmental change (e.g., institutional, organizational, policy-focused interventions). Supporting this argument, Sahyoun and colleagues [55] advocate for the need to explore individual and environmental factors that can potentially impact older adults' nutritional health by utilizing ecological approaches. Here we provide examples of interpersonal, institutional, institutional, community, policy, and multilevel nutrition interventions as well as insights.

Interpersonal and Intrapersonal Nutrition Interventions

Person-centered interventions (interpersonal, intrapersonal) require individuals to proactively make lifestyle changes. To address the needs of older adults requiring nutrient dense meals, a number of interventions have been successful using individual-level methods, such as conducting a series of personalized nutrition sessions by a trained food and nutrition practitioner [63]; the use of theory-based behavior change education materials in the form of manuals and newsletters, expert system assessment reports to personalize feedback, and telephone coaching [52, 56, 64]; and home visits, bi-weekly telephone contacts, and monthly newsletters that incorporate behavioral modification techniques; goal-setting [53]; and tailoring recommendations to participants' dietary patterns and lifestyle [65]. The usefulness of methods mentioned above in the design of nutrition interventions for older adults, such as tailoring to the individual learning needs of participants, having multiple in-person or phone-based contacts, and being theory-based, have been documented throughout the literature [38, 50, 65, 66]. Bernstein and colleagues [65] further note that in such interventions, compliance with protocols can be encouraged through participants' record keeping as well as continual monitoring and positive reinforcement from the research team. Future interventions should aim to increase the

dose of the treatment (i.e., intervention). Although longer-term interventions can be costly, it has been found that longitudinal health promotion activities can lengthen an older adults' number of healthy years [67]. In addition, investigations of older adults' perceived motivations and barriers to the use of programs and services should be examined, especially in view of the global economic downturn and the emergence of the "new poor" who may not be used to accessing nutrition support programs and services. To date, few randomized controlled intervention trials have included theoretically based behaviorally focused nutrition interventions targeted specifically at older adults. Randomized controlled trials with suitably representative sample populations and appropriate sample sizes are preferable to observational, cross-sectional, or quasi-experimental design because the randomization process reduces confounding from known or unknown factors; and allows researchers to posit causal associations. A major drawback of the sole use of a person-centered perspective is that the achievement of behavioral targets is ultimately controlled by the individual without recognition of external factors; changes in behavior are often short-term and not sustainable over time; and there are limited public health implications for larger segments of the population [53, 55]—further highlighting the need for an approach at multiple levels of influence.

Institutional and Community-Based Nutrition Interventions

Older adults transitioning from hospital to home have been found to lack in their consumption of fresh fruits, vegetables, and meat; and experience difficulty shopping for and preparing food [41]. Therefore, ensuring that a nutrition component is part of discharge planning is recommended. DiMaria-Ghalili and Amella [68] suggest that an assessment of the location in which meals are served is a key component to implementing environmental change. For example, the creation of a "home-like," environment as opposed to a traditional institution-like environment in long-term care hospital units has been found to optimize energy intake by older adults with cognitive impairments [69]. Other examples of changes that can take place in the food-service environments of institutionalized older adults are improved-lighting; the observation of rituals and practices prior to institutionalization (e.g., blessing food, washing hands before a meal); and the stimulation of senses via preparation of food in close proximity to the dining area [69]. In terms of a geographic, or locally-based community, Abusabha et al. [70] implemented an innovative, community-based study to increase access and the availability of produce. Their project was unique in that a *Veggie Mobile* (i.e., van) traveled to low-income senior housing sites to sell affordable fruits and vegetables.

Utilization of public health, or population-based, approaches to behavior change through the provision of information and environmental change (e.g., policy and use of media/mass communication) have proven successful [15, 56]. Further, collaborations with multidisciplinary institutional and organizational networks may help eliminate or reduce gaps in service, enhance the types of services available, reach more people, and increase awareness and the use of available community resources by older adults, their families, and caregivers [8].

Multilevel and Policy Nutrition Interventions

Person-centered interventions are more frequently conducted and assessed than environmental or policy interventions. In a review of health promotion interventions, Golden [71] found that of 62 nutrition interventions, 95 % focused on individual activities, 71 % on interpersonal activities, and only 3 % focused on the policy environment [71]. This review also showed that, compared to other health behavior change interventions, such as the promotion of physical activity and reductions in
smoking, nutrition interventions were significantly more likely to target three or more levels of intervention [71]. Cross-sectional studies have shown that, nutrition-related knowledge [72], selfefficacy, family support [73], and environmental factors are all associated with the eating behaviors of mid-life to older adults [74]. Therefore a variety of strategies implemented across multiple settings may be most promising to ensure successful nutrition interventions [55]. An example of multilevel intervention strategies successfully used by Johnson [75] to increase consumption of fruits and vegetables in an older adult population included: using serving size guides (individual level) and community volunteers (Meals-on-Wheels drivers) (interpersonal level); creating partnerships with community organizations and universities (community level); using existing infrastructures with similar missions (organizational level) as well as existing federal programs available throughout the United States (institutional level) [75]. Multilevel approaches such as this could potentially extend the reach of interventions into the population; set the stage for enhanced acceptance and diffusion of innovation by the target population and decision-makers alike; and have implications for policy [75]. To further increase the chance of success, a thorough understanding of older adults' social support systems and networks, or lack thereof, is essential in the development and implementation of multilevel nutrition interventions [55]. Within the policy realm, and relevant to both farmer's markets and the sale of reduced priced produce, a recent review of food subsidization programs in seven countries, including the United States, found that the subsidization of healthy foods (e.g., through supermarkets, restaurants, and farmers markets) can be effective in changing dietary behaviors [76]. The federal Senior Farmer's Market Nutrition Program (SFMNP) is an example of such a program. The SFMNP utilizes a coupon distribution system so that older adults can purchase locally grown, fresh fruits and vegetables [75], has the potential to help reduce food insecurity. Similar to person-centered interventions, robust assessments and evaluation methods beyond for example, cross-sectional and quasiexperimental designs, should be considered.

As a means to advance the science of multilevel intervention design, interventionists and nutrition practitioners may find it beneficial to consult with research scientists to assess proposed intervention methods. While consultation services may exceed the budgets of nonprofit and government programs, such efforts will help assess the efficacy, effectiveness, and validity of intervention findings [62]. Partnering with academic institutions and/or other organizations may help alleviate the burden of associated costs. To better inform policy, research studies should aim to identify predictive factors and interventions that can reduce food insecurity among older populations. Researchers should be encouraged to ensure that research budgets include appropriations to fund the translation and dissemination of research findings to the targeted populations. An increased effort to monitor and evaluate the impact and efficacy of food assistance programs is critical for the development of policy. Increased advocacy that involves multilevel collaborations focusing on reducing food insecurity, removing barriers to healthy eating, and improving the quality of older adults' diets will also inform program and service design as well as future nutrition policies.

Nutrition Intervention Evaluation and Assessment

There has been a call for the evaluation of well-designed intervention studies and methods to determine their effectiveness on nutrition outcomes [77]. Evaluating a wide range of nutrition intervention outcomes is particularly important because nutrition behavior is complex, individuals' self-report data are not always reliable, improving nutrition knowledge does not always lead to changes in behavior, and short-term programs do not necessarily result in long-term changes [78]. Thus, it is imperative that evaluations are conducted before, during, and after the implementation of interventions [45]. There are three types of evaluations: (1) formative evaluations, (2) process evaluations, and (3) outcome evaluations. Formative evaluation helps to ensure that the intervention is appropriate for both the target audience and setting. Process evaluation helps to ensure that the intervention will be implemented as planned. Outcome evaluation helps to ensure that the desired objectives of the intervention have been met [62]. Formative evaluation methods used in nutrition interventions include interviews, focus groups, pilot testing, and, in some cases, participant-completed surveys [79, 80]. Process evaluation methods include post-intervention participant satisfaction questionnaires, evaluation of intervention fidelity (i.e., how well intervention personnel have followed intervention procedures and protocols), participation metrics (e.g., participant involvement in intervention activities and adherence to intervention components), and interviews of both participants and intervention personnel relating to intervention delivery and uptake [50]. Staff- and participant-maintained program activity notes, resource allocation documents, and video or audio recordings of the intervention processes can be reviewed to ensure the fidelity of interventions to protocols [81]. Outcome evaluations—the most frequently conducted type of intervention evaluation, can include a variety of outcomes, for example:

- Physiologic measures such as weight, waist circumference, body mass index, blood pressure, and serum cholesterol and glucose levels [82].
- Knowledge outcomes, including declarative knowledge (knowledge about facts and events) and procedural knowledge (knowing how to apply declarative knowledge) [83] for example, knowing that an adequate dietary intake of protein is important for older adults [84], and how to incorporate an adequate amount of protein into the diet.
- Psychological outcomes such as movement through motivational stages of behavior change, and changes in behavior-specific outcome expectations and self-efficacy [83].
- Behavior change outcomes such as reduction in daily percent energy consumption from fat, or increase in daily consumption of fruits and vegetables [55].

Interventions for all populations, including aging populations, must be carefully designed. Evidence- and theory-based strategies should be used in the planning, implementation, and evaluation phases of tailored nutrition interventions. Formative, process, and outcome evaluations should be conducted to tailor interventions and assess their quality of delivery and effectiveness. It has been suggested that complex interventions having multiple components consider utilizing a phased approach to intervention delivery as well as both quantitative and qualitative evaluation methods [85]. Further, if an ecological intervention approach is used that targets not only individual level characteristics but also social, environmental, and policy determinants, then an evaluation should be conducted for each level of intervention.

In summary, there is currently a need for more comprehensive evaluations of nutrition interventions that include: (1) formative, process, and outcome assessments, (2) a program or behavioral maintenance component to help individuals and communities sustain nutritional gains over the longerterm, and (3) a multilevel perspective to accommodate a social-ecological perspective.

Tailoring Interventions for Special Populations

As Americans increase in age, number, and diversity, nutritional interventions tailored to meet the needs of special populations will increasingly be required. Racial and ethnic minorities—the fastest growing population group in the nation—often have unique dietary patterns and differential disease risk profiles which require culturally tailored nutritional programs and policies [86, 87]. Nutrition screening and assessment tools, educational material, and interventions should be designed and validated for ethnic-specific diets [36], and interventions should be available in multiple languages and literacy levels [40].

Living in rural areas has been associated with higher rates of chronic disease, obesity, and poorer health [88]. A greater number of older adults live in rural areas than urban areas [89]. Social isolation, lower incomes, lack of transportation, increased distances to purchase food or utilize nutrition programs, low quality diets types of "country cooking" eating patterns that emphasize higher fat and sodium-based methods of food preparation; reduced availability of healthy foods and higher rates of physical inactivity may all be contributing factors to consider when tailoring interventions for this population [37, 88, 89]. Policies that support the operation of rural food stores and improvements in rural transportation networks are needed to address some of these issues [90].

Older adults with physical impairments, as well as those with difficulties traveling, have been shown to have reduced access to food sources and nutrition support programs, which increases risk of poor nutrition and health outcomes [91]. Older adults at risk for, or suffering from, chronic or degenerative conditions have also been shown to be at increased risk of poor nutrition [34, 35] and to benefit from tailored medical nutrition therapy (MNT) services [34, 92, 93] MNT is provided by a registered dietitian, focuses on disease management, and involves detailed and individualized nutrition diagnostic, therapeutic, and counseling services [94].

Individuals with fewer financial resources have been shown to be more likely to choose foods that are highly processed with increased calories and sugar content. These foods are generally less expensive [40]. Alarmingly, food insecurity issues have increased among older adults in the United States since 2007 [95]. Older adults are at higher risk for negative physical and mental health outcomes. In 2009, 15.6 million adults aged 50 years or older faced the threat of hunger (i.e., were marginally food insecure), 8.8 million faced the risk of hunger (i.e., were food insecure), and 3.5 million experienced hunger (i.e., were very low food secure) [95]. There are various government (e.g., Older American's Act Elderly Nutrition Program, Nutrition Services Incentive Program, Supplemental Nutrition Assistance Program, Senior Farmers' Market Program, and Child and Adult Care Food Program) and nongovernmental food assistance programs either targeted specifically to community-dwelling older adults or that allow their participation [32]. Unfortunately, a lack of awareness of these resources may result in less than optimal participation rates [40]. Efforts should be made to ensure that older adults and their caregivers are aware of such programs.

Implementing flexible programs can help broaden their reach to better accommodate subgroups within the older adult population. For example, it has been suggested that the traditional model of home-delivery programs be reevaluated. There is a need to target specific subgroups of recipients who may have differing nutrition needs, as, for example, food insufficient individuals who need more nutrient dense meals [96, 97]. Other examples include providing programs at various sites—including private households and institutional settings such as senior housing sites, assisted living and extended care facilities, and nursing homes [34], and using voucher programs that enable participants to obtain food from restaurants, grocery stores, homeless shelters, and hospital cafeterias [40].

In summary, while they are diverse in their purpose and strategies, nutrition interventions may be a potentially useful gateway through which special populations of older adults can have improved health and access to other needed social and health programs and services [34, 36].

Health-Promoting Information Technology: Emerging Interventions

Older adults have a range of barriers that may prohibit participation in community health promotion programs (e.g., poor health, physical mobility, transportation difficulties, and social factors) [98]. The development and design of technology-based health promotion has been described as a promising intervention strategy [99] and may be a viable option for meeting the demands of aging populations [100–105]. Older adults are increasingly using digital communication technology for a range of purposes. A recent nationwide survey conducted by the Pew Research Center's Internet and American

Life Project reports that 53 % of U.S. adults aged 65 years or older use the Internet or e-mail, 69 % have a mobile phone, and 33 % use social networking sites [106]. The use of technology-based health promotion using computers (traditional, touchscreen, kiosk), electronic-tablets, smartphones, Personal Digital Assistants (PDAs), and automated telephone systems provide opportunities to implement individually tailored, real time, convenient, cost-effective, and widely distributed evidence and theoretically based interventions. Nutritional interventions using PDAs [100, 107] the Internet [108, 109], and telephones [110] have been tested in older adult populations and have been successfully used to monitor changes in health, promote and support positive health behavior change, and provide personal feedback [111, 112]. Given that technology-based interventions do not require face-to-face interaction, interventionists may be able to reach more individuals compared to traditional intervention and health promotion strategies. Increasingly, innovative applications have been used to encourage health behavior change and adherence (e.g., for diet, physical activity, medication) though smartphones and wearable technologies, some of which utilize point-of-decision prompts [113, 114]. These innovative approaches warrant additional systematic evaluation to determine the effectiveness of such approaches [99, 115]. This is especially important for interventions designed for older adult populations [99, 116]. In designing electronic technologies for use by older adults, special considerations should be made to account for the sensory, motor skills, computer literacy, and cognitive abilities of older adults, for example, by creating simple user interfaces with large clear typefaces and easy navigation elements [117–119].

In summary, studies have been conducted to assess the use of technology-based and/or web-based interventions among older adults [99], but more research needs to be conducted to determine their efficacy, especially in view of the increasing use of the Internet and information technology by older populations.

Conclusion and Recommendations

With the global increase in persons aged 65 years or older, the translation and application of nutrition research findings into tailored interventions that are evidence-based and have a broad reach is of utmost importance to promote successful aging. Improvement in health outcomes will most often be experienced following improvements in diet and related behaviors, particularly after such behavior changes have been sustained for long periods of time [50]. The heterogeneity of older adult populations and the varied influential factors on diet and related behaviors may impact the length of time in which objective and subjective changes in health and well-being are observed. Over 20 years ago, researchers identified three categories of older adults based on their health status. Aging categories created by Rowe and Kahn [120] were "Successful Agers" (i.e., absence of chronic disease, minimal age-related physiological changes) and "Usual Agers" (i.e., the presence of chronic age-related medical conditions and disability), with "High-Risk Agers" (experiencing the heaviest burden of chronic disease and ability) subsequently described by Harris and Feldman [121]. Conceptualizing older adults in this way may help interventionists, and food, nutrition, and/or healthcare practitioners prioritize and tailor nutrition interventions accordingly. Other areas of prioritization and focus for future work should include:

- Meeting the needs of special populations such as the older-old (i.e., aged >75 years) [27], racial/ ethnic minorities, people living in rural areas, and those experiencing economic hardships.
- Creating programs and services that address the needs of older adults along the continuum of weight status, i.e., those who are underweight and those who are obese or overweight.
- Explicitly targeting both nutritional and physical activity behaviors together in developing programs for aging adults, in light of the physiological and behavioral synergies accompanying the optimization of both of these key health behaviors [113].

- Creating urban and rural environments that support healthy behaviors through infrastructures that facilitate access to healthful food and physical activity opportunities;
- Promoting collaborative community partnerships (e.g., food banks, nonprofits, and faith-based institutions) specifically targeted toward older adults (with a dual advantage of meeting both nutritional and social engagement needs).
- Developing and using communication and technology-based interventions specifically targeted to the needs and preferences of older adults to promote positive health behaviors among older populations in an effort to reduce the burden of disease and help decrease the digital divide.

The increasing prevalence of obesity on a global scale, including among aging populations [122], emphasizes the need for evidence and theory-based nutrition interventions that adopt a life-course perspective—and take into account the social and environmental contexts in which people live. Such approaches have the potential to improve the health status and quality of life of many individuals in a cost-effective, yet efficient, manner. While much work remains to be done in this area, focusing scientific as well as public health energy and resources on the types of areas described above may yield important insights that can advance the field. Given current population trends in the aging field, the United States and other nations have much to gain from such investments, which can help older adults reach their full potential as individuals and as citizens, maximizing the potential of an aging population through shaping an active and vital old age [123].

Acknowledgments Dr. King was supported by U.S. Public Health Service grants R01 HL089694, RC1 HL099340, and U01 AG022376. Drs. Sheats and Winter were supported by U.S. Public Health Service grant T32 HL007034 From the National Heart, Lung, and Blood Institute.

References

- 1. Christensen K, Doblhammer G, Rau R, Vaupel JW. Aging populations: the challenges ahead. Lancet. 2009; 374:1196–208.
- 2. World Health Organization. Definition of an older or elderly person [Internet]. 2013 [cited 2012 Dec 15]. Available from: http://www.who.int/healthinfo/survey/ageingdefnolder/en/index.html
- Vincent GK, Velkoff VA. The next four decades—the older population in the United States: 2010 to 2050 population estimates and projections. Census report on the latest (2008) projections of the older population to 2050 [Internet]. 2010 May [updated 2010 Jun 23; cited 2013 Dec 15]. Available from: http://www.aoa.gov/aoaroot/ aging_statistics/future_growth/future_growth.aspx
- Statistics Canada. Population projections for Canada, provinces and territories, 2009 to 2036 [Internet]. 2009 [updated 2011 Jul 5; cited 2013 Dec 19]. Available from: http://www.statcan.gc.ca/daily-quotidien/100526/ dq100526b-eng.htm
- 5. Giannakouris K. Ageing characterises the demographic perspectives of the European societies. Eurostat, Statistics in Focus 2008;72.
- Australian Bureau of Statistics. Population Projections, Australia, 2006 to 2101, 3222.0 [Internet]. 2008 [2011 Jul 6; cited 2013 Dec 15]. Available from: http://www.abs.gov.au/AUSSTATS/abs@.nsf/DetailsPage/3222.02006%20 to%202101
- 7. Jeste DV, Savla GN, Thompson WK, Vahia IV, Glorioso DK, Palmer BW, et al. Association between older age and more successful aging: critical role of resilience and depression. Am J Psychiatry. 2013;170(2):188–96.
- 8. Bernstein M, Munoz N. Position of the Academy of Nutrition and Dietetics: food and nutrition for older adults: promoting health and wellness. J Acad Nutr Diet. 2012;112(8):1255–77.
- U.S. Department of Health and Human Services Administration on Aging. Profile of older Americans [Internet]. 2010 [cited 2012 Dec 29]. Available from: http://www.aoa.gov/aoaroot/aging_statistics/Profile/2010/docs/ 2010profile.pdf
- American Dietetic Association. Position of the American Dietetic Association: nutrition, aging, and the continuum of care. J Am Diet Assoc. 2000;100(5):580–95.
- Pruchno RA, Wilson-Genderson M, Cartwright F. A two-factor model of successful aging. J Gerontol B Psychol Sci Soc Sci. 2010;65B(6):671–9. doi:10.1093/geronb/gbq051.

- U.S. Department of Health and Human Services. Office of Disease Prevention and Health Promotion. Healthy People 2020 [Internet]. 2010 [updated 2013 Jan 25; cited 2013 Dec 28]. Available from: http://www.healthypeople. gov/2020/default.aspx
- 13. Delafuente JC. The silver tsunami is coming: will pharmacy be swept away with the tide? Am J Pharm Educ. 2009;73(1):1.
- Schlenker ED. Healthy aging: nutrition concepts for older adults. In: Wilson B, Bray G, Temple N, Strubble M, editors. Nutrition guide for physicians. New York: Humana Press; 2010.
- Cutler D. Behavioral health interventions: what works and why? In: Bulatao R, Anderson N, editors. Understanding racial and ethnic differences in health in late life: a research agenda. Washington, DC: National Academies Press; 2004.
- McLeroy KR, Bibeau D, Steckler A, Glanz K. An ecological perspective on health promotion programs. Health Educ Behav. 1988;15(4):351–77.
- 17. Pentz MA. Institutionalizing community-based prevention through policy change. J Community Psychol. 2000;28(3):257–70.
- Richard L, Gauvin L, Raine K. Ecological models revisited: their uses and evolution in health promotion over two decades. Annu Rev Public Health. 2011;32:307–26.
- Edstrom KM, Devine CM. Consistency in women's orientations to food and nutrition in midlife and older age: a 10-year qualitative follow-up. J Nutr Educ. 2001;33(4):215–23.
- Kremers SPJ, Visscher TLS, Brug J, Paw MCA, Schouten EG, Schuit AJ, et al. Netherlands research programme weight gain prevention (NHF-NRG): rationale, objectives and strategies. Eur J Clin Nutr. 2005;59(4):498–507.
- Higgins M, Barkley M. Important nutrition education issues and recommendations related to a review of the literature on older adults. J Nutr Elder. 2003;22(2):65–78.
- 22. U.S. Department of Health and Human Services Administration on Aging. Highlights from the pilot study: first national survey of older Americans Act Title III service recipients [Internet]. 2004 [cited 2013 Dec 28]. Available from: http://nutritionandaging.fiu.edu/suann/cd_contents/5%20Using%20Management%20Tools%20&%20Evaluation% 20Data/POMP%20highlights.pdf
- Lichtenstein AH, Rasmussen H, Yu WW, Epstein SR, Russell RM. Modified MyPyramid for older adults. J Nutr. 2008;138:5–11.
- Pettigrew S, Pescud M, Donovan RJ. Older people's diet-related beliefs and behaviours: intervention implications. Nutr Diet. 2012. doi:10.1111/j.1747-0080.2012.01602.x.
- Sin HP, Liu DT, Lam DS. Lifestyle modification, nutritional and vitamins supplements for age-related macular degeneration. Acta Ophthalmol. 2013;91(1):6–11.
- Devore EE, Kang JH, Stampfer MJ, Grodstein F. Total antioxidant capacity of diet in relation to cognitive function. Am J Clin Nutr. 2010;92:1157–64.
- Weeden AM, Remig VM. Food intake of Kansans over 80 years of age attending congregate meal sites. Nutrients. 2010;2:1297–307.
- 28. Ervin RB. Healthy eating index scores among adults, 60 years of age and over, by sociodemographic and health characteristics: United States, 1999–2002. Adv Data. 2008;395:1–16.
- 29. Vitolins MZ, Quandt SA, Bell RA, Arcury TA, Case LD. Quality of diets consumed by older rural adults. J Rural Health. 2002;18:49–56.
- Foote JA, Giuliano AR, Harris RB. Older adults need guidance to meet nutritional recommendations. J Am Coll Nutr. 2000;19:628–40.
- Centers for Disease Control and Prevention. Healthy aging at a glance [Internet]. 2011 [updated 2011 May 11; cited 2012 Dec 28]. Available from: http://www.cdc.gov/chronicdisease/resources/publications/AAG/aging.htm
- 32. Kamp B. Position of the American Dietetic Association, American Society for Nutrition, and Society for Nutrition Education: food and nutrition programs for community-residing older adults. J Am Diet Assoc. 2010;110(3): 463–72.
- 33. Dorner B, Friedrich E, Posthauer M. Position of the American Dietetic Association: individualized nutrition approaches for older adults in health care communities. J Am Diet Assoc. 2010;110(10):1549–53.
- Haughton B, Stang J. Population risk factors and trends in health care and public policy. J Acad Nutr Diet. 2012;112 Suppl 1:s35–46.
- 35. Chen CC-H, Schilling LS, Lyder CH. A concept analysis of malnutrition in the elderly. J Adv Nurs. 2001;36(1):131–42.
- Buchowski MS, Sun M. Nutrition in minority elders: current problems and future directions. J Health Care Poor Underserved. 1996;7(3):184–209.
- 37. Krondl M, Coleman P, Lau D. Helping older adults meet nutritional challenges. J Nutr Elder. 2008; 27(3-4):205-20.
- Sahyoun NR, Zhang XL. Dietary quality and social contact among a nationally representative sample of the older adult population in the United States. J Nutr Health Aging. 2005;9(3):189–93.

- Nestle M, Wing R, Birch L, DiSogra L, Drewnowski A, Middleton S, Sobal J, Winston M, Economos C. Behavioral and social influences on food choice. Nutr Rev. 1998;56(5 Pt 2):S50–64.
- 40. Institute of Medicine. Nutrition and healthy aging in the community: workshop summary. Washington, DC: The National Academies Press; 2012.
- 41. Anyanwu UO, Sharkey JR, Jackson RT, Sahyoun NR. Home food environment of older adults transitioning from hospital to home. J Nutr Gerontol Geriatr. 2011;30(2):105–21.
- 42. McNaughton SA, Crawford D, Ball K, Salmon J. Understanding determinants of nutrition, physical activity and quality of life among older adults: the wellbeing, eating and exercise for a long life (WELL) study. Health Qual Life Outcomes. 2012;10(1):109.
- 43. Chernoff R. Nutrition and health promotion in older adults. J Gerontol A Biol Sci Med Sci. 2000;56A:47-53.
- Higgins MM. Food and nutrition professionals can help older adults improve dietary practices. J Am Diet Assoc. 2007;107(5):806–7. doi: http://dx.doi.org/10.1016/j.jada.2007.02.009.
- Bartholomew LK, Parcel GS, Kok G, Gottlieb NH, Fernandez ME. Planning health promotion programs: an intervention mapping approach. 3rd ed. San Francisco: Jossey-Bass; 2011.
- Locher JL, Ritchie CS, Roth DL, Sen B, Vickers Douglas K, Vailas LI. Food choice among homebound older adults: motivations and perceived barriers. J Nutr Health Aging. 2009;13(8):659–64.
- Higgins MM, Barkley MC. Tailoring nutrition education intervention programs to meet needs and interests of older adults. J Nutr Elder. 2003;23(1):59–79.
- Locher JL, Bales CW, Ellis AC, Lawrence JC, Newton L, Ritchie CS, Roth D, Buys D, Vickers KS. A theoretically based behavioral nutrition intervention for community elders at high risk: the B-NICE randomized controlled clinical trial. J Nutr Gerontol Geriatr. 2011;30(4):384–402.
- Freudenberg N, Eng E, Flay B, Parcel G, Rogers T, Wallerstein N. Strengthening individual and community capacity to prevent disease and promote health: in search of relevant theories and principles. Health Educ Behav. 1994;22(3):290–306.
- Burke L, Froehlich R, Zheng Y, Glanz K. Current theoretical bases for nutrition intervention and their uses. In: Coulston A, Boushey CJ, Ferruzzi M, editors. Nutrition in the prevention and treatment of disease. 3rd ed. New York: Elsevier; 2013.
- 51. Shikany JM, Bragg CS, Ritchie CS. Behavioral theories applied to nutritional therapies for chronic diseases in older adults. In: Bales C, Ritchie C, editors. Handbook of clinical nutrition and aging. New York: Springer; 2009.
- Clark PG, Blissmer BJ, Greene GW, Lees FD, Riebe DA, Stamm KE. Maintaining exercise and healthful eating in older adults: the SENIOR project II: study design and methodology. Contemp Clin Trials. 2011;32(1):129–39.
- Kelley K, Abraham C. RCT of a theory-based intervention promoting healthy eating and physical activity amongst out-patients older than 65 years. Soc Sci Med. 2004;59(4):787–97.
- Campbell MK, Demark-Wahnefried W, Symons M, Kalsbeek WD, Dodds J, Cowan A, et al. Fruit and vegetable consumption and prevention of cancer: the Black Churches United for Better Health project. Am J Public Health. 1999;89(9):1390–6.
- Sahyoun NR, Pratt CA, Anderson AMY. Evaluation of nutrition education interventions for older adults: a proposed framework. J Am Diet Assoc. 2004;104(1):58.
- Clark PG, Nigg CR, Greene G, Riebe D, Saunders SD. The study of exercise and nutrition in older Rhode islanders (SENIOR): translating theory into research. Health Educ Res. 2002;17(5):552–61.
- 57. Parker S, Powell L, Hermann J, Phelps J, Brown B. Preferred educational delivery strategies among limited income older adults enrolled in community nutrition education programs. J Ext. 2011;49(1):n1.
- Doerksen SE, Estabrooks PA. Brief fruit and vegetable messages integrated within a community physical activity program successfully change behaviour. Int J Behav Nutr Phys Act. 2007;4(1):12.
- 59. Schuetz J. Geragogy: instructional programs for elders. Commun Educ. 1982;31(4):339-47.
- 60. Maderer P, Skiba A. Integrative geragogy: Part 1: theory and practices based model. Educ Gerontol. 2006; 32(2):125–45.
- 61. Green J. The role of theory in evidence-based health promotion practice. Health Educ Res. 2000;15(2):125-9.
- 62. Kristal A, Ollberding N. Evaluation of nutrition interventions. In: Coulston A, Boushey CJ, Ferruzzi M, editors. Nutrition in the prevention and treatment of disease. 3rd ed. New York: Elsevier; 2013.
- 63. Endevelt R, Lemberger J, Bregman J, Kowen G, Berger-Fecht I, Lander H, et al. Intensive dietary intervention by a dietitian as a case manager among community dwelling older adults: the edit study. J Nutr Health Aging. 2011; 15(8):624–30.
- 64. Greene GW, Fey-Yensan N, Padula C, Rossi SR, Rossi JS, Clark PG. Change in fruit and vegetable intake over 24 months in older adults: results of the SENIOR project intervention. Gerontologist. 2008;48(3):378–87.
- Bernstein MA, Nelson ME, Tucker KLL, Layne J, Johnson E, Nuernberger A, et al. A home-based nutrition intervention to increase consumption of fruits, vegetables, and calcium-rich foods in community dwelling elders. J Am Diet Assoc. 2002;102(10):1421–7.

- 66. Bandayrel K, Wong S. Systematic literature review of randomized control trials assessing the effectiveness of nutrition interventions in community-dwelling older adults. J Nutr Educ Behav. 2011;43(4):251–62.
- 67. Chernoff RJ. Nutrition and health promotion in older adults. Gerontol A Biol Sci Med Sci. 2001;56 Suppl 2:47–53.
- DiMaria-Ghalili RA, Amella E. Nutrition in older adults: intervention and assessment can help curb the growing threat of malnutrition. Am J Nurs. 2005;105(3):40.
- 69. Desai J, Winter A, Young KW, Greenwood CE. Changes in type of foodservice and dining room environment preferentially benefit institutionalized seniors with low body mass indexes. J Am Diet Assoc. 2007;107(5): 808–14.
- AbuSabha R, Namjoshi D, Klein A. Increasing access and affordability of produce improves perceived consumption of vegetables in low-income seniors. J Am Diet Assoc. 2011;111(10):1549–55.
- Golden SD, Earp JAL. Social ecological approaches to individuals and their contexts twenty years of health education & behavior health promotion interventions. Health Educ Behav. 2012;39(3):364–72.
- Dallongeville J, Marecaux N, Cottel D, Bingham A. Association between nutrition knowledge and nutritional intake in middle-aged men from northern France. Public Health Nutr. 2001;4(1):27–33.
- Hermstad AK, Swan DW, Kegler MC, Barnette JK, Glanz K. Individual and environmental correlates of dietary fat intake in rural communities: a structural equation model analysis. Soc Sci Med. 2010;71(1):93–101.
- 74. Sharkey JR, Johnson CM, Dean WR. Food access and perceptions of the community and household food environment as correlates of fruit and vegetable intake among rural seniors. BMC Geriatr. 2010;10:32.
- 75. Johnson D, Beaudoin S, Smith L, Beresford S, LoGerfo J. Increasing fruit and vegetable intake in homebound elders: the Seattle Senior Farmers' Market Nutrition Pilot Program. Prev Chronic Dis. 2004;1(1):1–9.
- 76. An R. Effectiveness of subsidies in promoting healthy food purchases and consumption: a review of field experiments. Public Health Nutr. 2012;1(1):1–14.
- McCormack LA, Laska MN, Larson NI, Story M. Review of nutritional implications of farmer's markets and community gardens: a call for evaluation and research efforts. J Am Diet Assoc. 2010;110(3):399–408.
- Higgins MM, Barkley MC. Evaluating outcomes and impact of nutrition education programs designed for older adults. J Nutr Elder. 2003;22(4):69–81.
- Buta B, Brewer L, Hamlin DL, Palmer MW, Bowie J, Gielen A. An innovative faith-based healthy eating program from class assignment to real-world application of PRECEDE/PROCEED. Health Promot Pract. 2011;12(6): 867–75.
- Vereecken C, Covents M, Maes L, Moyson T. Formative evaluation of the feedback component of Children's and Adolescents' Nutrition Assessment and Advice on the Web (CANAA-W) among parents of schoolchildren. Public Health Nutr. 2012;16:15–26.
- Bellg AJ, Borrelli B, Resnick B, Hecht J, Minicucci DS, Ory M, et al. Enhancing treatment fidelity in health behavior change studies: best practices and recommendations from the NIH Behavior Change Consortium. Health Psychol. 2004;23(5):443.
- 82. Wunderlich S, McKinnon C, Piemonte J, Ahmad ZN. Measuring the impact of nutrition education and physical activity on older adults participating in government sponsored programs. J Nutr Elder. 2009;28(3):255–71.
- Miller CK, Edwards L, Kissling G, Sanville L. Evaluation of a theory-based nutrition intervention for older adults with diabetes mellitus. J Am Diet Assoc. 2002;102(8):1069–74. 79–81.
- 84. Chernoff R. Protein and older adults. J Am Coll Nutr. 2004;23 Suppl 6:6275–305.
- Campbell M, Fitzpatrick R, Haines A, Kinmonth AL, Sandercock P, Spiegelhalter D, et al. Framework for design and evaluation of complex interventions to improve health. BMJ. 2000;321(7262):694.
- Weddle D, Wilson FL, Berkshire SD, Heuberger R. Evaluating nutrition risk factors and other determinants of use of an urban congregate meal program by older African Americans. J Nutr Gerontol Geriatr. 2012;31(1):38–58.
- Arandia G, Nalty C, Sharkey JR, Dean WR. Diet and acculturation among Hispanic/Latino older adults in the United States: a review of literature and recommendation. J Nutr Gerontol Geriatr. 2012;31(1):6–37.
- Befort CA, Nazir N, Perri MG. Prevalence of obesity among adults from rural and urban areas of the United States: findings from NHANES (2005–2008). J Rural Health. 2012;28(4):392–7.
- Johnson JE. Informal social support networks and the maintenance of voluntary driving cessation by older rural women. J Commun Health Nurs. 2008;25(2):65–72.
- Durazo E, Jones M, Wallace S, Van Arsdale J, Aydin M, Stewart C. The health status and unique health challenges of rural older adults in California. Policy Brief UCLA Cent Health Policy Res. 2011;PB2011-7:1–8.
- Satariano WA, Guralnik JM, Jackson RJ, Marottoli RA, Phelan EA, Prohaska TR. Mobility and aging: new directions for public health action. Am J Public Health. 2012;102(8):1508–15.
- Racine E, Troyer JL, Warren-Findlow J, McAuley WJ. The effect of medical nutrition therapy on changes in dietary knowledge and DASH diet adherence in older adults with cardiovascular disease. J Nutr Health Aging. 2011;15(10):868–76.

- Mooradian AD, Haas MJ, Wehmeier KR, Wong NC. Obesity-related changes in high-density lipoprotein metabolism. Obesity. 2012;16(6):1152–60.
- 94. Daly A, Michael P, Johnson EQ, Harrington CC, Patrick S, Bender T. Diabetes white paper: defining the delivery of nutrition services in Medicare medical nutrition therapy vs. Medicare diabetes self-management training programs. J Am Diet Assoc. 2009;109(3):528–39. doi: http://dx.doi.org/10.1016/j.jada.2008.11.004.
- 95. Ziliak JP, Gundersen C. Senior hunger in America 2010: an annual report. Alexandria, VA: Meals on Wheels Research Foundation; 2012.
- 96. Sharkey JR. Risk and presence of food insufficiency are associated with low nutrient intakes and multimorbidity among homebound older women who receive home-delivered meals. J Nutr. 2003;133(11):3485–91.
- Sharkey JR, Branch LG, Zohoori N, Giuliani C, Busby-Whitehead J, Haines PS. Inadequate nutrient intakes among homebound elderly and their correlation with individual characteristics and health-related factors. Am J Clin Nutr. 2002;76(6):1435–45.
- 98. Schutzer KA, Graves BS. Barriers and motivations to exercise in older adults. Prev Med. 2004;39:1056-61.
- Ammann R, Vandelanotte C, de Vries H, Mummery K. Can a website-delivered computer-tailored physical activity intervention be acceptable, usable, and effective for older people? Health Educ Behav. 2012. doi:10.1177/ 1090198112461791.
- Atienza AA, King AC, Oliveira BM, Ahn DK, Gardner CD. Using hand-held computer technologies to improve dietary intake. Am J Prev Med. 2008;34(6):514–8. doi:10.1016/j.amepre.2008.01.034.
- 101. Coughlin JF, Pope J. Innovations in health, wellness, and aging-in-place. Eng Med Biol Mag. 2008;27(4):47–52.
- 102. King AC, Friedman R, Marcus B, Castro C, Napolitano M, Ahn D, et al. Ongoing physical activity advice by humans versus computers: the Community Health Advice by Telephone (CHAT) trial. Health Psychol. 2007;26(6): 718–27.
- 103. Atienza AA, Oliveira B, Fogg BJ, King AC. Using electronic diaries to examine physical activity and other health behaviors of adults age 50+. J Aging Phys Act. 2006;14(2):192–202.
- 104. Dishman E. Healthcare revolution: technologies enable seniors to age gracefully at home. Technology@ Intel Mag. 2004; pp. 1–11.
- 105. Coughlin JF. Technology needs of aging boomers. Issues Sci Technol. 1999;16(1):53-60.
- 106. Zickuhr K, Madden M. Older adults and internet use. Report of the Pew Research Center's Internet & American Life Project [Internet]. 2012 [updated 2012 June 6; cited 2013 Jan 6]. Available from: http://pewinternet.org/ Reports/2012/Older-adults-and-internet-use
- 107. Welch JL, Siek KA, Connelly KH, Astroth KS, McManus MS, Scott L, et al. Merging health literacy with computer technology: self-managing diet and fluid intake among adult hemodialysis patients. Patient Educ Couns. 2010;79(2):192–8.
- McCoy MR, Couch D, Duncan ND, Lynch GS. Evaluating an internet weight loss program for diabetes prevention. Health Promot Int. 2005;20(3):221–8.
- Pintauro SJ, Regimbald S, Burczy SA, Nickerson A, Buzzell PR, Berlin L. Evaluation of a nutrition and health educational online computer program for older adults. J Nutr Food Sci. 2011;1(5):117.
- 110. Lee AH, Jancey J, Howat P, Burke L, Kerr DA, Shilton T. Effectiveness of a home-based postal and telephone physical activity and nutrition pilot program for seniors. J Obes. 2011. doi:10.1155/2011/786827.
- 111. Piette JD. Interactive behavior change technology to support diabetes self-management: where do we stand? Diabetes Care. 2007;30(10):2425–32.
- 112. Czaja S, Lee CC. Information technology and older adults. In: Sears A, Jacko J, editors. The human-computer interaction handbook. 2nd ed. New York: CRC Press; 2007.
- 113. Silva JM, Mouttham A, El Saddik A. UbiMeds: a mobile application to improve accessibility and support medication adherence. Proceedings of the 1st ACM SIGMM International Workshop on Media Studies and Implementations that Help Improving Access to Disabled Users; 2009 October 23; Beijing China; 2009, pp. 71–78.
- 114. King AC, Castro CM, Buman MP, Hekler EB, Urizar GG, Ahn DK. Behavioral impacts of sequentially versus simultaneously delivered dietary plus physical activity interventions: the CALM trial. Ann Behav Med. 2013; 46(2):157–68.
- 115. Kroeze W, Werkman A, Brug J. A systematic review of randomized trials on the effectiveness of computer-tailored education on physical activity and dietary behaviors. Ann Behav Med. 2006;31(3):205–23.
- Wantland DJ, Portillo CJ, Holzemer WL, Slaughter R, McGhee EM. The effectiveness of web-based vs. non-webbased interventions: a meta-analysis of behavioral change outcomes. J Med Internet Res. 2004;6(4):e40.
- 117. Pak R, Price MM, Thatcher J. Age-sensitive design of online health information: comparative usability study. J Med Internet Res. 2009;11(4):e45.
- 118. Kaufman DR, Pevzner J, Hilliman C, Weinstock RS, Teresi J, Shea S, Starren J. Redesigning a telehealth diabetes management program for a digital divide seniors population. Home Health Care Manag Pract. 2006;18(3): 223–34.

- 119. Sheats JL, Winter SJ, Padilla-Romero P, King AC.. FEAST (Food Environment Assessment using the Stanford Tool): Development of a mobile application to crowd-source resident interactions with the food environment. Poster presented at the annual meeting of the Society of Behavioral Medicine, Philadelphia, PA. April, 2014.
- 120. Rowe JW, Kahn RL. Human aging: usual and successful. Science. 1987;237:143-9.
- 121. Harris TB, Feldman JJ. Implications of health status in analysis of risk in older persons. J Aging Health. 1991;3:262-84.
- 122. Villareal DT, Apovian CM, Kushner RF, Klein S. Obesity in older adults: technical review and position statement of the American Society for Nutrition and NAASO. The Obesity Society. Obesity Res. 2012;13(11):1849–63.
- 123. King AC, Guralnik JM. Maximizing the potential of an aging population. JAMA. 2010;304(17):1944–5.

Chapter 2 Systematic Reviews in the Field of Nutrition

Alice H. Lichtenstein

Key Points

- Systematic reviews are valuable tools for staying abreast of evolving nutrition and aging-related topics, formulating dietary guidelines, establishing nutrient reference intakes, formulating clinical practice guidance, evaluating health claims, and setting research agendas.
- Basic steps of conducting a systematic review include identifying a review team, developing an analytic framework, formulating key questions, selecting inclusion/exclusion criteria, identifying search terms, searching the literature, selecting publications for inclusion, extracting and summarizing data, rating the methodological quality of the included studies, and if adequate data is available, conducting a meta-analysis.
- Unique issues specific to nutrition-related topics include baseline exposure, nutrient status, nutrient bioavailability, nutrient bioequivalence, biological stores, multiple biological functions, undefined nature of nutrient intervention, and uncertainties in assessing dose–response relationships
- Conclusions of systematic reviews or meta-analysis are helpful tools that can contribute to decisions but do not in themselves establish guidelines or research agendas.

Keywords Nutrition • Diet • Systematic review • Key questions • Analytic framework • Evidence tables • Meta-analysis • Bioavailability • Bioequivalence • Dietary supplements • Older adults

The amount of scientific literature published each year increases exponentially. Very few individuals have the time to keep abreast of the latest findings, and even fewer have the time to integrate the latest findings into prior work. For topics unaligned with one's primary area of expertise or research focus the task can become insurmountable. Under these circumstances, review articles can serve an important function.

There are two general types of reviews; narrative and systematic. Both summarize the literature on a specific topic. For the most part, narrative reviews do not use a consistent methodological approach to accomplish this task. Unclear are the criteria used to include or exclude studies, making it difficult to determine whether the review is a comprehensive and unbiased evaluation of the literature.

A.H. Lichtenstein, D.Sc.

Cardiovascular Nutrition Laboratory, Jean Mayer USDA HNRCA at Tufts University, 711 Washington Street, Boston, MA 02111, USA e-mail: alice.lichtenstein@tufts.edu

A systematic review specifies inclusion/exclusion criteria and follows a relatively standard format designed to provide an impartial and complete assessment of all the literature available on the topic. This chapter will focus on the latter type, systematic review, and the unique considerations that should be addressed when performing a review of nutrition-related topics.

Background

Systematic review methodology was first developed in the field of medicine as an approach to formulate clinical practice guidelines [1-3]. The approach has been used to not only establish guidelines [4, 5] but also set research agendas [6] and formulate scientific consensus statements [7, 8]. More recently systematic review methodology has been applied to the field of nutrition [9-11]. In addition to helping professionals stay abreast of evolving topics, the reviews also serve as the basis for formulating dietary guidelines [12], establishing nutrient reference intakes [13], formulating clinical practice guidance [14], evaluating health claims [15], and setting research agendas [16].

Steps to Conducting Systematic Reviews

There are basic components that are integral to the systematic review process [17–19]. Critical is the ability to identify and synthesize all the available data in an unbiased manner to answer prespecified questions. A key component of conducting a systematic review is to thoroughly document all aspects of the process. This latter point is important not only to ensure validity and reproducibility but to also facilitate cost-effective periodic updates as new data emerge. Cochrane Collaboration [20] and the Agency for Healthcare Research and Quality [21] are two organizations that have developed guide-lines for conducting systematic reviews.

Review Team

Prior to commencing work on a systematic review, it is important to identify the individuals who will be working on the project and their roles. The size of the team will depend on expertise of those involved, the range, number, and detailed nature of research questions to be addressed, and the scope of available literature. In addition to those individuals who will be responsible for conducting the actual systematic review, there may be others involved. These individuals are domain experts, commonly referred to as a technical expert panel. Particularly in the field of nutrition, it is important to include domain experts who represent multiple disciplines. Domain experts provide guidance in terms of refining the research questions and identifying search terms. They typically are not involved in the actual review process so as to ensure independence of those conducting the systematic review and to avoid potential bias or the appearance of bias according to prior publications or public statements. Once the systematic review is completed it is not uncommon for the technical expert panel to serve as reviewers.

Analytic Framework

An analytic framework provides a graphic representation of the organizational structure of a systematic review and includes the basic elements of the review: exposure(s), indicator marker(s), surrogate outcome(s), and clinical outcome(s) of interest. A generic analytic framework is depicted in Fig. 2.1 [11].



Curve 1. Association of exposure with clinical outcomes of interest Curve 2. Association of exposure with surrogate outcomes or intermediate markers (with good or possible

evidence for linkage with clinical outcomes)

Curve 3. Association of indicator markers to clinical outcomes

Curve 4. Association between exposure and indicator markers

Curve 5. Association of indicator markers to surrogate outcomes or intermediate markers (with good or possible evidence for linkage with clinical outcomes)

Curve 6. Association between surrogate outcomes and clinical outcomes ("good" evidence for linkage)

6a. Association between intermediate markers and clinical outcomes (uncertain linkage)

Fig. 2.1 Generic analytic framework of dietary reference intakes [22]. Reprinted with permission of the Agency for Healthcare Research and Quality from M. Chung, et al. Vitamin D and Calcium: A Systematic Review of Health Outcomes, p. 23

In general, solid lines indicate established paths between exposure, and indicator markers, surrogate markers, or clinical outcomes. The dashed lines indicate possible or yet to be validated relationships. Some component of each research question should be included in the analytical framework. The complexity of the analytic framework depends on the breadth of the topic and the number of questions to be addressed. For example, Fig. 2.2 depicts an analytical framework developed to address five key questions involving two nutrients, vitamin D and calcium [22]. This analytic framework was developed jointly by the methodologists and domain experts, prior to starting the systemic literature review.

Research (Key) Questions

The initial research questions, henceforth referred to as key questions, are drafted by either the sponsor or researchers. These questions are carefully reviewed, and frequently modified, by the research team to improve clarity or increase specificity. For example, a single question may be divided into multiple questions, each defined by specific criteria. The essential components of each key question are summarized by the acronym PICO. PICO is an abbreviation for population/patient (P), intervention/independent variable (I), comparator (C), and outcome (O). In some cases, PICO-D is used, with the D representing study design. Key questions must be carefully formulated to match the intent of the systemic review because they will define the literature search terms, inclusion criteria, and exclusion criteria. Prior to finalizing the key questions it is helpful to evaluate the available resources for the project and potential extent of the literature (horizon scan). This information may lead to a reassessment of the project scope and/or modifications in the PICO criteria.



Fig. 2.2 Analytic framework for vitamin D and/or calcium estimated average requirement [22]. Reprinted with permission of the Agency for Healthcare Research and Quality from M. Chung, et al. Vitamin D and Calcium: A Systematic Review of Health Outcomes, p. 23

Inclusion/Exclusion Criteria

After the key questions are finalized, the next step in the systematic review process is to define the inclusion/exclusion criteria. These decisions should be made on the basis of the PICO components. The population will be defined by specific characteristics, e.g., age range, sex, health status. The intervention will be defined by the specific dietary component(s) (in the case of nutrition-related systematic reviews), dose, chemical form, and duration of intervention. For example, if the area of interest is omega-3 fatty acids, the intervention could be all omega-3 fatty acids, alpha-linolenic acid, eicosapentaenoic acid or docosahexaenoic acid, fish oil or fish, or some combination thereof. The dose could have minimum and maximum daily limits and the duration could be limited in terms of minimum/ maximum length of time. To carry the omega-3 fatty acids example forward, the comparator(s) could be a placebo, such as another vegetable oil or, if the intervention is fish, an alternate food that has an equivalent amount of protein. The outcomes could be broad, e.g., all-cause mortality; narrow, e.g., coronary heart disease and stroke; or limited to indicator biomarkers, e.g., plasma triglyceride or C-reactive protein concentrations. For some topics if the analytical methodologies for the indicator biomarker have changed over time or the number of publications that appear to address the key questions is large, a cut-off publication date can be used as an inclusion criterion.

Search Terms

Search terms are identified on the basis of the key question components. Both the methodologists and domain experts should participate in identifying the search terms to ensure they are comprehensive enough to adequately capture all the available literature and specific enough to minimize identifying extraneous literature that does not address the key questions. Filtering out extraneous literature can be a time-consuming burden.

Literature Search

At this point the domain experts step back from the systematic review process and the remaining members of the research team should conduct the literature search. This division of labor is necessary to avoid potential bias or the appearance of bias during the review process. To ensure that all literature meeting the inclusion criteria is identified, multiple databases should be searched (e.g., Medline, CAB Abstracts, and Cochrane Library Central) and citations of recent publications should be reviewed to ensure the most relevant literature was identified by the search strategy. If more than a few relevant articles are identified using the latter approach the search terms should be reassessed by the research team. At all stages of the literature search it is critical to meticulously document the methodology.

Literature Selection

As a general rule, the majority of publications identified during the literature search will not meet the inclusion criteria. Screening is initially done on the basis of abstracts. Reasons for excluding publications are dictated by the inclusion/exclusion criteria. Typical reasons for exclusion include the type of article (e.g., editorials, reviews, case studies), study design (e.g., inadequate statistical analysis, lack of control group), and not meeting the inclusion criteria (relevant exposure or outcome measures, inadequate intervention period, study subject characteristics out of range). During the screening process it is important to generate a flow diagram that documents information for the total number of publications identified, and the number remaining during the preliminary screening process, with particular attention to documenting publications filtered out. The next step is to retrieve full texts of the remaining publications. As for the prior screening step, the reasons for screening out studies should be fully documented. One example of a flow diagram is shown in Fig. 2.3 [23].

Extract and Summarize Data

A system must be developed to accurately and comprehensively extract data from the included publications. This is assisted by the development of an evidence table template. The format can be unique to each systematic review. Critical components may include participant characteristics, study design, intervention, methods used to measure compliance, results, statistical significance, and outcomes. It is important to include all the information necessary to answer the key questions in a format that will facilitate comparisons among studies. Data extraction is time consuming and every effort should be made to avoid having to re-extract data due to an omission. Some systematic reviews include publications of studies with very different designs, such as clinical interventions and observation cohorts. In



Fig. 2.3 Flow diagram (PRISMA) [23]

those cases separate evidence tables may be constructed to ensure the critical components of each type of study design is adequately captured. Evidence tables tend to be large in size and cumbersome to work with. Comparison among studies is facilitated by using the evidence tables to construct summary evidence tables that capture the specific elements necessary to synthesize the data.

Methodological Quality and Applicability of Studies

Some type of quality assessment is usually part of a systematic review. Table 2.1 proves a simple example with some common elements [17]. A number of quality assessment tools are available, for example, Cochrane Collaboration [20], the U.S. Preventive Services Task Force [21] and the Grading Recommendations Assessment, Development and Evaluation working group [24]. The choice of which to use depends on the topic and characteristics of the review.

Table 2.1 One approach to assessing methodological quality and applicability of studies^a

Methodological quality

- A Least bias; results are valid.
- B Susceptible to some bias, but not sufficient to invalidate the results.
- C Significant bias that may invalidate the results.

Applicability

- I Sample is representative of the target population. It should be sufficiently large to cover both sexes, a wide age range, and other important features of the target population (e.g., diet).
- II Sample is representative of a relevant subgroup of the target population, but not the entire population.
- III Sample is representative of a narrow subgroup of subjects only, and is of limited applicability to other subgroups.

Overall effect

- ++ Clinically meaningful benefit demonstrated.
- + A clinically meaningful beneficial trend exists but is not conclusive.
- 0 Clinically meaningful effect not demonstrated or is unlikely.
- Harmful effect demonstrated or is likely.

^aAdapted from [17]

Meta-analysis

Meta-analysis use statistical approaches to combine the data from multiple studies to derive a conclusion about the totality of the data. Frequently the result of a meta-analysis is to calculate an overall effect estimate. In some cases, a meta-regression can be performed to evaluate discrepancies among studies and to explore factors such as dose–response relationships. The major concern when performing and interpreting a meta-analysis is the appropriateness of combining studies that did not use identical experimental protocols. Decisions on whether it is appropriate to combine studies must be made on a case by case basis. Meta-analysis should not be confused with a pooled analysis. For a pooled analysis primary data from multiple individual studies are combined and analyzed. For a metaanalysis mean data derived from published studies are combined and analyzed.

Unique Issues Related to Nutrition-Related Systematic Reviews

Methodological approaches to performing systematic reviews were first developed to address questions in the field of medicine [25]. The literature on pharmaceutical interventions tends to be more straightforward than that related to nutrition. There are a number of unique factors that must be accounted for when performing systematic reviews of nutrition-related topics (Table 2.2). It is important to be aware of these prior to starting the review process so that accommodations, when necessary, can be made.

Baseline Exposure

There is a baseline exposure for all essential nutrients. This exposure can be the variable of interest, as for observational studies, or a starting point to be accounted for in interventional studies. For interventional studies this information is used as a starting point from which to increase or decrease intake. Inaccurate accounting for potential differences in baseline exposure among study subjects can alter the conclusions drawn from the data. Baseline exposure most commonly comes from foods and

 Table 2.2
 Unique issues related to nutrition-related systematic reviews

- Baseline exposure
- Nutrient status at baseline
- Nutrient bioavailability
- Nutrient bioequivalence
- Biological stores
- Multiple and interrelated biological functions
- Undefined nature of nutrient/food intervention
- Uncertainty in assessing dose–response relationships

beverages, nutrient and herbal supplements, endogenous synthesis (e.g., vitamin D, vitamin K), and/ or environmental contamination (e.g., iron cooking pot). Limitations in accurately quantifying baseline nutrient exposure include factors such as the subjective nature of self-reported food and beverage intake, incomplete data bases with which to analyze the data, subject burden associated with collecting intake data, and inadequate methodology for quantifying in vivo synthesis. Currently, few biomarkers to characterize baseline nutrient exposure are available. Information on how the baseline food/nutrient intake is assessed should be included in the systematic review methodology section.

Nutrient Status

The nutrient status of an individual or population at baseline will impact on the ability to assess potential relationships between intake and outcomes within a cohort, and the response to altering the intake after an intervention. There is a wide range of approaches used to determine nutrient status, including, but not limited to concentrations in blood-borne cells, hard tissue concentrations (hair, nails), activity of enzymes for which the nutrient is a cofactor, saturation of carrier proteins, and nutrient or metabolite excretion. Homeostatic mechanisms, trafficking of nutrients to different tissues and storage capacity for the nutrient of interest precludes, with rare exceptions, the use of plasma concentrations as an indicator of nutrient status.

Nutrient Bioavailability

Not all forms of a nutrient are absorbed to the same degree of efficiency, otherwise referred to as bioavailability. Many factors determine nutrient bioavailability. These include, but are not limited to ionic state (e.g., ferric $[Fe^{3+}]$ /ferrous $[Fe^{2+}]$); chemical form (e.g., folic acid/folate); nutrient–nutrient interactions (e.g., vitamin C and non-heme iron); nutrient–food interactions (e.g., dietary fat facilitates fat-soluble vitamin absorption, phytic acid/oxalic acid-containing foods impedes zinc absorption); nutrient–drug interactions (e.g., isoniazid and vitamin B₆, coumadin and vitamin K, folate and metformin); mineral salts (e.g., calcium carbonate, calcium citrate, calcium malate); single versus multiple daily doses (e.g., calcium, iron), and habitual intake (e.g., iron, vitamin C). Additional factors that may alter nutrient bioavailability include biological status (e.g., iron and pregnancy, vitamin B₁₂ and achlorhydria), food processing (e.g., particle size and dietary fiber, lye-treatment [corn] and tryptophan, heat treatment and carotenoids), and for nutrient supplements completeness or rate of release (e.g., coatings, excipients, surfactants).

Nutrient Bioequivalence

Many nutrients occur in multiple forms that have different biological activity, commonly referred to as bioequivalence. The general approach used to address this issue is to calculate nutrient equivalents as was done when setting the recommended dietary allowances for vitamin A (preformed vitamin A and carotenoids), folate (folate and folic acid), vitamin K (phylloquinone and menaquinone), and niacin (niacin and tryptophan) [26–28]. The challenge of determining accurate conversion factors for the calculation of nutrient equivalents has recently been demonstrated for beta-carotene [27].

Nutrient Availability from Biological Stores

In humans, release of nutrients from storage depots can be unrelated to biological need. The release of vitamin A from hepatic stores is dependent on protein nutriture. The release or deposition in adipocytes of fat-soluble vitamins can be determined by changes in tissue mass rather than biological need for the nutrients.

Multiple and Interrelated Biological Functions of Nutrients

Most nutrients have multiple biological functions. The key questions define the nutrient-specific scope of the systematic review. This is often accomplished by narrowing the range of the work (e.g., vitamin D and bone health or breast cancer risk, vitamin A and infectious disease or xerophthalmia). Some biological functions of nutrients are dependent on the biological status of other nutrients (e.g., folate/vitamin B_{12} /vitamin B_6 , vitamin D/calcium) and must be accounted for.

Undefined Nature of Nutrient Intervention

Food-based interventions, in contrast to supplement-based interventions, present unique challenges in accurately attributing a cause and effect to an individual nutrient or group of nutrients. For example, one approach to increasing very long-chain n-3 fatty acid (eicosapentaenoic acid [EPA] and docosahexaenoic acid [DHA]) intake is to provide or instruct study participants to increase fish intake. However, there is considerable variability in the levels of EPA and DHA in different fish, within species of the same type of fish [29], season the fish was caught and animal husbandry practices for farm-raised fish. In addition, there are other compounds in fish, in addition to EPA and DHA, that could alter the biological outcomes assessed. Evaluating the effect of EPA and DHA using nutrient supplements is not without similar challenges due to the wide range of fatty acid profiles in available fish oil, potential changes in absolute levels during prolonged storage or heat exposure, and different chemical forms of the fatty acids (e.g., triglyceride, ethyl esters). Documentation of nutrient intake assessment is important to record.

Uncertainties in Assessing Dose-Response Relationships

It is difficult to accurately assess food, hence, nutrient intakes. In some cases systematic bias is unavoidable using currently available dietary instruments. This can be particularly important for systematic reviews designed to assess absolute rather than relative dose–response relationships. In general, food frequency questionnaires underestimate energy and protein intakes with greater biases than 24-h recall [30]. Potential biases for other nutrient intake estimates are not adequately documented but likely exist. Assay procedures for biomarkers of nutritional status can also significantly affect the mean and distribution of reported values and need to be factored into data interpretation [31, 32].

Strengths and Limitations of Systematic Review Approach for Nutrition Applications

Applying the systematic review approach to nutrition applications has both strengths and limitations (Table 2.3).

Strengths

Strengths of the systematic review process include representing an objective process while avoiding the appearance of bias, summarizing a comprehensive assessment of the available literature, defining the scope of the review as reflected in key questions, documenting the search strategy in a detailed manner, allowing for flexibility to customize searches by each topic without compromising rigorousness, generating a report that can be updated in a cost effective manner, identifying a transparent process of review, enhancing statistical power by simultaneously assessing multiple and frequently small studies, incorporating some ability to detect publication bias, providing an assessment of methodological quality, applicability and overall effect, and facilitating identification of research gaps.

Table 2.3 Strengths and limitations of the systematic review process

Strengths

- · Represents an objective process and avoids the appearance of bias
- · Summarizes a comprehensive assessment of the available literature
- Defines scope of the review as reflected in key questions
- · Documents search strategy in a detailed manner
- · Allows for flexibility to customize searches by each topic without compromising rigorousness
- · Generates a report that can be updated in cost effective manner
- · Identifies a transparent process of review
- · Enhances statistical power by simultaneously assessing multiple and frequently small studies
- · Incorporates some ability to detect publication bias
- · Provides an assessment of methodological quality, applicability and overall effect
- · Facilitates identification of research gaps

Limitations

- Confines imposed by key questions in terms of population, intervention, comparator and outcome measures
- · Restrictions impact on the number of studies meeting the inclusion criteria
- Limitations imposed by the quality and scope of available data (e.g., poor study design, missing data, publication bias)

The detailed level of documentation is an integral part of the systematic review process. A beneficial byproduct of the systematic reviews is the ability to identify needed improvements in the quality and nature of data reporting. This is particularly valuable in the field of nutrition where the number of studies available to answer a key question is often limited.

Limitations

Limitations of the systematic review process include confines imposed by the key questions in terms of population, intervention, comparator and outcome measures, restrictions that impact on the number of studies meeting the inclusion criteria, and limitations imposed by the quality and scope of available data (e.g., poor study design, missing data, publication bias).

Confusion can be caused when multiple systematic reviews addressing what appear to be on the same topic come to different conclusions [9, 33, 34]. Such discrepancies can usually be attributed to differences in the key questions, study inclusion/exclusion criteria, evolution of available data, and changes in the analytical approaches used to generate the data or quality of the data. Meticulous documentation of the process can avoid confusion.

Contribution of Systematic Reviews to Nutritional Status of Older Adults

It is difficult to isolate an independent contribution of systematic reviews in defining the nutritional status specifically of older adults or understanding their unique nutritional challenges. However, the contribution of systematic reviews to the understanding of health outcomes, particularly in terms of chronic disease risk, has had a marked effect.

In 2013, it was reported that the prevalence of multi-morbidity, the concurrent existence of two or more chronic disorders, is increasing in the United States, paralleling the increased life-span and demographic shift in the population to the older aged groups [35]. The rates of multi-morbidity in individuals aged <65 years, 65–74 years, 75–84 years, and >85 years is 50 %, 62 %, 76 %, and 82 %, respectively. These data were corroborated using previously published systematic reviews [35]. It can be difficult to address the nutritional needs of individuals with multi-morbidities. To be most efficacious intervention approaches may need to be tailored to different age groups. However, a systematic review of nutrition and health literacy literature concluded that health literacy was a stronger predictor of health status than age, income, employment, education, and race [36]. This type of information is critical when designing intervention programs to improve health status of older adults.

Evaluating the nutritional status of older adults is an important component when assessing health status. A challenge is determining which nutritional screening tool is the best predictor of status for older adults and most likely to provide useful information to optimize health outcomes. A systematic review of the instruments available was used to address this issue [37]. In addition to providing valuable information on the relative strengths of each instrument for which data were available, research gaps that would benefit from being addressed were also identified. The most recent iteration of the Dietary Reference Intakes for calcium and vitamin D relied heavily on a systemic review of the evidence [13, 22]. The conclusions of the review resulted in changes in the vitamin D dietary reference intakes for most age groups, including individuals >70 years [22].

Multivitamins use is common in older adults [38, 39]. Strikingly, use is most prevalent among those with the most nutrient adequate diets [40, 41]. Conclusions of a systematic review indicate the most prominent motivator for nutrient supplement use among all U.S. adults is desire to improve overall health [42]. However, for older adults this is not the case, their motivations stem more from

specific health concerns and nutrient supplement use parallels engaging in more favorable health and lifestyle behaviors [42]. Recently, concern has been raised that multivitamin–multimineral supplement use in older adults is associated with increased mortality [43–45]. A systematic review was conducted to address this issue. The totality of the data indicated that there was no trend of multivitamin–multimineral supplement use with all-cause mortality, and there was a trend for reduced mortality from vascular and cancer causes [46]. In addition, no statistically significant evidence for heterogeneity or publication bias in the data was identified by the systematic review. Only an unbiased approach that considered all the data together could have adequately addressed this issue.

Conclusion

Systematic reviews are by their very nature an objective assessment of the available literature on a defined topic. The availability of systematic reviews that have addressed nutrition-related questions is extremely useful. The process is flexible, allowing for a wide range of topics to be addressed and facilitates periodic updates as new data emerge. However, it is important to note that by definition, systematic reviews are limited in scope as defined by the key question and restricted to what is frequently a less than complete dataset. The key questions answered are discrete units rather than interrelated entities. The conclusions of systematic reviews or meta-analysis do not in themselves establish guidelines or research agendas. They are an important tool that can contribute to decisions in these areas. A well-documented transparent process is critical for reviewing the data, and if inadequate data are available and expert opinion impacts on the final recommendations, there is a clear indication of when that occurred and within what context.

References

- 1. Chalmer I, Atlamn DG. Systematic reviews. London: BMJ Publishing Group; 1995.
- West S, King V, Carey TS, Lohr KN, McKoy N, Sutton SF, et al. Systems to rate the strength of scientific evidence. Evid Rep Technol Assess (Summ). 2002;47:1–11.
- 3. Harris RP, Helfand M, Woolf SH, Lohr KN, Mulrow CD, Teutsch SM, et al. Current methods of the US Preventive Services Task Force: a review of the process. Am J Prev Med. 2001;20(3 Suppl):21–35.
- Guirguis-Blake J, Calonge N, Miller T, Siu A, Teutsch S, Whitlock E, et al. Current processes of the U.S. Preventive Services Task Force: refining evidence-based recommendation development. Ann Intern Med. 2007;147(2): 117–22.
- Briss PA, Zaza S, Pappaioanou M, Fielding J, Wright-De Aguero L, Truman BI, et al. Developing an evidencebased guide to community preventive services-methods. The Task Force on Community Preventive Services. Am J Prev Med. 2000;18(1 Suppl):35–43.
- Oxman AD, Sackett DL, Guyatt GH. Users' guides to the medical literature. I. How to get started. The Evidence-Based Medicine Working Group. JAMA. 1993;270:2093–5.
- Ratko TA, Burnett DA, Foulke GE, Matuszewski KA, Sacher RA. Recommendations for off-label use of intravenously administered immunoglobulin preparations. University Hospital Consortium Expert Panel for off-label use of polyvalent intravenously administered immunoglobulin preparations. JAMA. 1995;273(23):1865–70.
- Jassal SV, Roscoe JM, Zaltzman JS, Mazzulli T, Krajden M, Gadawski M, et al. Clinical practice guidelines: prevention of cytomegalovirus disease after renal transplantation. J Am Soc Nephrol. 1998;9(9):1697–708.
- Wang C, Harris WS, Chung M, Lichtenstein AH, Balk EM, Kupelnick B, et al. n-3 Fatty acids from fish or fish-oil supplements, but not alpha-linolenic acid, benefit cardiovascular disease outcomes in primary- and secondaryprevention studies: a systematic review. Am J Clin Nutr. 2006;84(1):5–17.
- Balk EM, Tatsioni A, Lichtenstein AH, Lau J, Pittas AG. Effect of chromium supplementation on glucose metabolism and lipids: a systematic review of randomized controlled trials. Diabetes Care. 2007;30(8):2154–63.
- Russell R, Chung M, Balk EM, Atkinson S, Giovannucci EL, Ip S, et al. Opportunities and challenges in conducting systematic reviews to support the development of nutrient reference values: vitamin A as an example. Am J Clin Nutr. 2009;89:728–33.

- 2 Systematic Reviews in the Field of Nutrition
- 12. Dietary Guidelines for Americans. 2010. http://www.cnpp.usda.gov/DGAs2010-DGACReport.htm
- IOM. Dietary reference intakes for calcium and vitamin D. 2010. http://www.iomedu/Reports/2010/ Dietary-Reference-Intakes-for-Calcium-and-Vitamin-Daspx
- Stang J, Bayerl CT. Position of the American Dietetic Association: child and adolescent nutrition assistance programs. J Am Diet Assoc. 2010;110:791–9.
- Trumbo PR, Ellwood KC. Lutein and zeaxanthin intakes and risk of age-related macular degeneration and cataracts: an evaluation using the Food and Drug Administration's evidence-based review system for health claims. Am J Clin Nutr. 2006;84(5):971–4.
- Balk EM, Horsley TA, Newberry SJ, Lichtenstein AH, Yetley EA, Schachter HM, et al. A collaborative effort to apply the evidence-based review process to the field of nutrition: challenges, benefits, and lessons learned. Am J Clin Nutr. 2007;85(6):1448–56.
- Lichtenstein AH, Yetley EA, Lau J. Application of systematic review methodology to the field of nutrition. J Nutr. 2008;138:2297–306.
- Chung M, Balk EM, Ip S, Lee J, Terasawa T, Raman G, et al. Systematic review to support the development of nutrient reference intake values: challenges and solutions. Am J Clin Nutr. 2010;92:307–14.
- Trikalinos TA, Moorthy D, Chung M, Yu WW, Lee J, Lichtenstein AH, et al. Concordance of randomized and nonrandomized studies was unrelated to translational patterns of two nutrient-disease associations. J Clin Epidemiol. 2012;65:16–25.
- Higgins JP, Green S. Cochrane handbook for systematic reviews of interventions Version 5.0.0. [updated February 2008; cited 2008 Aug]. Available from: www.cochrane-handbook.org
- 21. AHRQ. Agency for Healthcare Research and Quality Rockville, MD: U.S. Department of Health & Human Services. [cited 2008 August]. Available from: http://www.ahrq.gov
- 22. Chung M, Balk E, Bendel M, Ip S, Lee J, Lichtenstein AH, et al. Vitamin D and calcium: a systematic review of health outcomes. AHRQ Publication No. 09-E015. Rockville, MD: Agency for Healthcare Research and Quality; 2009.
- 23. PRISMA. http://www.prisma-statement.org/statement.htm
- 24. GRADE. GRADE Working Group [homepage on Internet]. Available from: http://www.gradeworkinggroup.org/
- Moher D, Tricco AC. Issues related to the conduct of systematic reviews: a focus on the nutrition field. Am J Clin Nutr. 2008;88:1191–9.
- IOM. Dietary reference intakes. Thiamin, riboflavin, niacin, vitamin B6, folate, vitamin B12, pantothenic acid, biotin and choline. Washington, DC: National Academy of Sciences; 1998.
- 27. IOM. Dietary reference intakes. Vitamin C, vitamin E, selenium and carotenoids. Washington, DC: National Academy of Sciences; 2000.
- IOM. Dietary reference intakes. Vitamin A, vitamin K, arsenic, boron, chromium, copper, iodine, iron, manganese, molybdenum, nickel, silicon, vanadium and zinc. Washington, DC: National Academy of Sciences; 2001.
- 29. USDA. http://www.nalusdagov/fnic/foodcomp/Data/
- 30. Schatzkin A, Kipnis V, Carroll RJ, Midthune D, Subar AF, Bingham S, et al. A comparison of a food frequency questionnaire with a 24-hour recall for use in an epidemiological cohort study: results from the biomarker-based Observing Protein and Energy Nutrition (OPEN) study. Int J Epidemiol. 2003;32(6):1054–62.
- National Center for Health Statistics. Questionnaire for NHANES 2001–02. Internet: http://www.cdcgov/nchs/data/ nhanes/nhanes_01_02/sp_dsqpdf. Accessed 7 May 2008.
- 32. Official Journal of the European Communities. Directive 2002/46/EC of the European Parliament and of the Council of 10 June 2002 on the approximation of the laws of the Member States relating to food supplements [Internet].
- Hooper L, Thompson RL, Harrison RA, Summerbell CD, Ness AR, Moore HJ, et al. Risks and benefits of omega 3 fats for mortality, cardiovascular disease, and cancer: systematic review [see comment]. BMJ. 2006;332(7544): 752–60.
- 34. Kwak SM, Myung SK, Lee YJ, Seo HG, Korean Meta-analysis Study Group. Efficacy of omega-3 fatty acid supplements (eicosapentaenoic acid and docosahexaenoic acid) in the secondary prevention of cardiovascular disease: a meta-analysis of randomized, double-blind, placebo-controlled trials. Arch Intern Med. 2012;172:638–94.
- 35. Salive ME. Multimorbidity in older adults. Epidemiol Rev. 2013;35:75-83.
- Carbone ET, Zoellner JM. Nutrition and health literacy: a systematic review to inform nutrition research and practice. J Acad Nutr Diet. 2012;112:254–66.
- Dent E, Visvanathan R, Piantadosi C, Chapman I. Nutritional screening tools as predictors of mortality, functional decline, and move to higher level care in older people: a systematic review. J Nutr Gerontol Geriatr. 2012; 31:97–145.
- Nahin RL, Pecha M, Welmerink DB, Sink K, DeKosky ST, Fitzpatrick AL, et al. Concomitant use of prescription drugs and dietary supplements in ambulatory elderly people. J Am Geriatr Soc. 2009;57:1197–205.
- Bailey RL, Fulgoni VL, Keast DR, Dwyer JT. Dietary supplement use is associated with higher intakes of minerals from food sources. Am J Clin Nutr. 2011;94:1376–81.

- Bailey RL, Fulgoni VL, Keast DR, Dwyer JT. Examination of vitamin intakes among US adults by dietary supplement use. J Acad Nutr Diet. 2012;112:657–63.
- 41. Morgan TK, Williamson M, Pirotta M, Stewart K, Myers SP, Barnes J. A national census of medicines use: a 24-hour snapshot of Australians aged 50 years and older. Med J Aust. 2012;196:50–3.
- Bailey RL, Gahche JJ, Miller PE, Thomas PR, Dwyer JT. Why US adults use dietary supplements. JAMA Intern Med. 2013;4:1–7. doi:10.1001/jamainternmed.2013.2299.
- Watkins ML, Erickson JD, Thun MJ, Mulinare J, Heath CWJ. Multivitamin use and mortality in a large prospective study. Am J Epidemiol. 2000;152:149–62.
- 44. Park SY, Murphy SP, Wilkens LR, Henderson BE, Kolonel LN. Multivitamin use and the risk of mortality and cancer incidence: the multiethnic cohort study. Am J Epidemiol. 2001;173:906–14.
- Mursu J, Robien K, Harnack LJ, Kyong Park K, Jacobs Jr DR. Dietary supplements and mortality rate in older women: Iowa Women's Health Study. Arch Intern Med. 2011;171:1625–33.
- Macpherson H, Pipingas A, Pase MP. Multivitamin-multimineral supplementation and mortality: a meta-analysis of randomized controlled trials. Am J Clin Nutr. 2013;97:437–44.

Chapter 3 Nutrition Assessment

Rose Ann DiMaria-Ghalili and Michi Yukawa

Key Points

- Older adults are at risk for malnutrition (undernutrition) across the care continuum.
- Nutrition screening should be conducted upon admission to the hospital and long-term care settings, upon first home care visit, enrollment in community-based programs, and during annual primary care visits.
- Recommended nutrition screening tools are the Mini-Nutritional Assessment-Short Form and the Determine Checklist.
- Nutrition assessment should be performed for any older adult at nutrition risk.
- Nutrition assessment is a comprehensive exam focused on weight changes, body composition status, physical examination, comorbid conditions, functional status, medication history, dietary intake, and psychosocial and economic status.
- Adult disease-related malnutrition diagnosis applicable to older adults include: starvation, chronic disease-related malnutrition, and acute disease-related malnutrition.
- All health professionals (physicians, nurses, dietitians, pharmacists, dentists, speech pathologists, and physical therapists) should be knowledgeable about appropriate approaches for the nutrition screening and assessment of older adults.

Keywords Nutrition assessment • Nutrition screening • Malnutrition • Weight loss • Unintentional weight loss • MNA-SF • DETERMINE checklist

R.A. DiMaria-Ghalili, Ph.D., R.N., F.A.S.P.E.N., C.N.S.C. (🖂)

Doctoral Nursing Department, College of Nursing and Health Professions, Drexel University, 245 N. 15th Street, Mail Stop 1030, Bellet Building Room 518, Philadelphia, PA 19102, USA e-mail: rad83@drexel.edu

M. Yukawa, M.D., M.P.H. Division of Geriatrics, Department of Medicine, San Francisco VA Medical Center, University of California San Francisco, 4150 Clement St. 181 NH, San Francisco, CA 94121, USA e-mail: Michi.Yukawa@va.gov

Introduction

Nutrition plays an important role in the promotion and maintenance of optimal physical functioning and quality of life in older adults [1]. Nutritional status is the balance of nutrient intake, physiological demands, and metabolic rate. Malnutrition includes both undernutrition (protein and/or calorie deficiency) and overnutrition (obesity); in older adults malnutrition more commonly refers to undernutrition. Obesity (BMI \geq 30 kg/m²) is a key indicator of well-being in older adults, and 38 % of older Americans were obese in 2009–2010 [2]. While obesity is often easy to identify, unintentional weight loss and malnutrition are not. Malnutrition is a common geriatric syndrome [3] and 12–72 % of older adults are malnourished or at risk for malnutrition [4–9]. Malnutrition can impact quality of life [10, 11] and is associated with measurable adverse outcomes which include infections, pressure ulcers, falls, increased length of hospital stay, institutionalization, increased cost of care, and death [9, 12–19]. Identification of malnourished older adults through nutrition screening and assessment can lead to prompt interventions to promote or maintain nutritional health and improve health outcomes.

Nutrition Screening

Meeting the nutritional needs of the older adults includes screening for nutrition risk, performing a nutrition assessment, implementing nutrition interventions, monitoring responses to interventions, rescreening, and when appropriate diagnosing malnutrition (Table 3.1). Nutrition screening is the first step [20] and is defined as "a process to identify an individual who is malnourished or who is at risk for malnutrition to determine if a detailed nutrition assessment is indicated" [21]. Nutrition screening and rescreening can be performed by any health care professionals in any practice setting with valid and reliable tools for older adults [22]. Since nutrition screening can take place in a variety of care settings (primary care, ambulatory care, acute care, home care, community, and long-term care) the selection of the screening tool and timing of screening should be appropriate to the setting [22].

Nutrition Screening Regulations

Hospital or Acute Care Settings

In the hospital setting the Joint Commission mandates that every patient admitted to the hospital or acute care setting have a nutrition screen performed within 24 h of admission [23]. The standard is not different for older patients and the elements to be included in the nutrition screen are not specified. Nutrition screening on admission identifies those patients at risk and who would benefit from in-depth nutrition assessment; it triggers follow-up by a dietitian at the beginning of hospital treatment. Nutrition status can deteriorate during the course of admission, and an older adult may be at increased risk for malnutrition at discharge than upon admission. Presently there is no regulatory requirement for continued rescreening of nutrition status during hospitalization, and most importantly, upon discharge.

Long-Term Care

Federal regulations require nutrition screening for new residents within 14 days of admission to the skilled care facility (nursing home). Nutritional status and unintentional weight loss in long-term care residents are monitored very closely. In fact, one of the quality measures mandated by the Centers for



Medicare and Medicaid Services (CMS) is "significant weight loss" in residents of a long-term care facility. The MDS (Minimum Data Set) 3.0 tracks weights on nursing home residents, and CMS defines significant weight loss as weight loss of $\geq 5 \%$ in 1 month or $\geq 10 \%$ weight loss in the last two quarters in residents who were not on prescribed weight-loss regimen (www.CMS.gov/medicare/Quality-Initiative-Patient-Assessment-Instruments). The MDS nurse coordinator contacts the physician when a resident has lost significant weight.

Primary Care

According to the Joint Commission, ambulatory clinics, office practices, and primary care settings should have an organizational policy that details when nutritional screenings and assessments are performed [24]. The Joint Commission recommends "nutritional screening may be performed at the first visit" [24], and rescreening and assessment are needed only "when appropriate to the reason the patient is presenting for care" [24]. If a primary care practice is not accredited by the Joint Commission there is no specific regulation mandating monitoring of weight or nutrition screening. However, experts advocate for the monitoring of weight loss in geriatric patients [25–27].

Community

Significant weight loss in community-dwelling older adults is defined as ≥ 5 % weight loss in 1 year [26, 27]. The US Preventive Services Task Force does not provide a recommendation on monitoring weight changes or nutrition screening in older adults. However, the Academy of Nutrition and Dietetics' evidence-based nutrition practice guideline on unintended weight loss in older adults strongly recommends "older adults are screened for unintended weight loss regardless of setting" [28]. The Administration on Aging does require nutrition screening with the DETERMINE checklist [29, 30] for all community-dwelling older adults who receive home-delivered meals or take part in congregate meal programs through the Title III Older American Act Nutrition Programs [31].

Homecare

Nutrition screening is not included in the Outcomes and Assessment Information Set (OASIS) for Medicare home health agencies [32]. However, questions are asked regarding whether patient can perform certain activities for oneself, including feeding, preparing meals, and food shopping. The Joint Commission requires that a home care agency defines in writing which nutrition status information it collects in the patient's assessment and reassessment [33].

Types of Screening Tools

Nutrition screening tools identify those who have, or are at risk for, malnutrition [22]. In the United States, there is no standardized tool to screen for nutrition risk in older adults. Using different tools across practice settings and institutions limits the ability to benchmark nutrition risk or malnutrition and to determine the yearly prevalence of malnutrition. One literature review identified 21 different nutrition screening and assessment tools for use with older adults [34]. However, many of these tools are not fully developed and the validity, reliability, sensitivity, and specificity are not established [34]. The Mini-Nutritional Assessment Short-Form (MNA-SF), and the DETERMINE checklist are nutrition screening instruments recommended for use in older adults.

Mini-Nutrition Assessment-Short Form (MNA-SF)

The original or full MNA (www.mna-elderly.com) was developed in 1991 and consists of 18 items derived from four parameters of assessment: anthropometric, general, dietary, and subjective [35]. The full MNA has two components: a nutrition screen and a complete nutrition assessment. The full MNA is available in 24 languages and has been used in over 122 international studies in 30,000 older adults in different settings [35]. The reliability and validity of the full MNA is well established across care settings and the tool has a reported sensitivity of 96 % and specificity of 98 % [35]. The MNA-SF is a screening tool to identify older adults who are malnourished or at risk of malnutrition. The most recent version of the MNA-SF consists of six questions on food intake, weight loss, mobility, psychological stress or acute disease, presence of dementia or depression, and body mass index (BMI). When BMI is not available, then calf circumference is measured and used in place of BMI. The MNA-SF has a sensitivity of 89 %, specificity of 82 %, and a strong positive predictive value (Youden Index = 0.70) [36]. While the MNA-SF takes about 5 min to complete and is a recommended evidence-based screening tool for use in older adults [20, 22, 38].

DETERMINE Checklist

The DETERMINE checklist was developed by the Nutrition Screening Initiative to assess nutrition risk in older adults. The tool was designed to be self-administered, though health professionals can administer the screen. The DETERMINE checklist consists of ten items that assesses eating and drinking habits, mouth problems, income, eating alone, medications, weight loss/gain, and ability to shop/cook/feed self [29, 30]. The tool has established validity [34], but it has a high positive screen rate [39]. While the DETERMINE checklist is recommended for use in older adults as an evidence-based nutrition risk screen [38], the tool is most appropriate for use in community-dwelling older adults [31] and in the primary care setting.

Developing Screening Programs

Nutrition risk screening can be easily incorporated into the routine care of older adults across practice settings. The DETERMINE checklist and the Self-MNA (www.mna-elderly.com) nutrition risk questionnaires can be completed by older adults in a physician waiting room and the results can be discussed during the office visit. Nutrition risk screening tools (MNA-SF) can be incorporated into the nursing admission assessment for older adults admitted to the hospital, extended care facility, long-term care, or home health. While nutrition screening programs identify those older adults at nutritional risk who need a more in-depth nutrition assessment and/or nutrition intervention, screening only provides a snapshot of nutrition risk at a precise point in time. While there are no evidencebased criteria for rescreening, any changes in the medical history, psychological status, social status, or functional status can impact the level of nutrition screen upon discharge to identify nutrition concerns during the transition from hospital to home [40, 41]. Institutionalized older adults should be rescreened every 3 months, and normally nourished community-dwelling older adults should be rescreened annually [42] (Table 3.1).

Nutrition Assessment

Nutrition assessment is a "comprehensive approach to diagnosing nutrition problems that uses a combination of the following: medical, nutrition, and medication histories; physical examination; anthropometric measurements; and laboratory data" [21]. Psychosocial status, economic factors, and functional status are also important components of a comprehensive nutrition assessment for older adults.

Anthropometrics

Weight

Body weight is the mass of an individual and represents the sum of individual body compartments. Body weight is characterized at the atomic (oxygen, carbon, hydrogen, other), molecular (water, lipid, protein, other), cellular (cell mass, extracellular fluid, extracellular solids), tissue (skeletal muscle, adipose tissue, bone, blood, other), and whole body (head, neck, trunk, lower extremities, upper extremities) levels [43]. Change in one of these compartments can cause change in weight. On the molecular level, body weight is the sum of water (60 % of body weight in healthy individual is water) and dry body weight (minerals, protein, and lipid) [43]. Lean body mass and fat free mass are essentially the same—both include water, protein, and minerals, while lean body mass does not include the relatively small amount of essential lipids [43]. Body weight is dynamic as it represents the balance of energy intake and energy expended in the form of physical activity and metabolism.

Weight is a critical vital sign in older adults. Weight should be measured without shoes or heavy clothing. Bed scales and chair scales can be used for older adults who cannot stand erect. The percentage of weight loss should be calculated as follows: usual weight – current weight/usual weight $\times 100 \%$) [46]. If an individual cannot recall whether or not they have lost weight, ask about changes in clothing size, examination of the belt, or ask significant others [44]. Daily fluctuations in total body weight are minimal in healthy adults (i.e., less than ±0.5 kg) [45], while an increase in 1–2 kg/day indicates fluid retention [44] or fluid overload in individuals with heart failure [46]. Weight gain and/or weight loss over time reflect changes in body composition [47]; furthermore, body composition can change over time (i.e., increase or decrease in fat or lean mass) *without* a change in body weight [47, 48].

An acceleration of weight loss is not part of normal aging. Body weight changes throughout adulthood, generally increasing until age 60 [48], stabilizing until about age 70 [49] and then slowly decreasing after age 70–75 [50]. Studies in very healthy elderly subjects indicate that only very small decrements in weight (0.1–0.2 kg/year) appear to occur in association with normal aging [49]. Further research needs to be conducted to understand the normal weight trajectory in those 80 years of age and older.

Body weight may decrease 50–60 % of ideal weight during chronic semi-starvation [45]. Physiological function is significantly impaired when more than 20 % of body protein is lost [51]. Hill's early work demonstrated that a weight loss of >10 % of body weight is associated with physiological impairment [51] and protein loss is usually clinically significant when more than 15 % of body weight has been lost [51]; however, these studies were not performed exclusively in the elderly. While a weight loss of 4 % of usual body weight in 1 year was reported to be the most sensitive and specific cut-point to define clinically important weight loss [25, 52], a weight loss of 5 % of usual body weight over 6–12 months is the most widely accepted definition for clinically important weight loss in older adults [44].

Intentional Versus Unintentional Weight Loss

The intentionality of weight change should be considered when assessing weight loss. Voluntary weight loss is due to either intentional planned restrictions or modifications in dietary intake (i.e., dieting), increase in physical activity, or both. Involuntary or unintentional weight loss is a decrease in weight over a defined period of time without a desire or intention of the individual to reduce body weight. Unintentional weight loss is considered unexplained when a definitive cause of the weight loss cannot be made [53]. Unexplained, ongoing, weight loss in persons without a history of weight fluctuations (e.g., "yo yo" dieting) is cause for concern [44]. Unintentional weight loss can occur in isolation [50] or as component of geriatric syndromes such as malnutrition (undernutrition) [54] and frailty [55], as well as secondary to sarcopenia [56], cachexia [56], and inflammatory conditions [56]. Some patients may ignore or minimize their unintentional weight loss [44] and not report it to the physician [57]. Older adults who lose weight unintentionally may report being satisfied with weight loss after years of struggling to lose weight and indicate that they do not want to regain the weight they lost [57]. One study found cancer patients who lost up to 60 lb of their pre-cancer weight were pleased with the unintentional weight loss despite the impact on functional status [58]. Unintentional weight loss may go untreated if clinicians do not monitor weight changes in older adults over time. Unintentional weight loss in older adults is associated with increased morbidity and mortality, primarily due to the loss of muscle mass [48]. Clinicians should consider weight in older adults as an important vital sign.

Adverse Effects of Unintentional Weight Loss

Any weight loss in older adults is associated with a loss of lean mass [48, 59] and accelerates the development of sarcopenia [59], resulting in functional decline [60, 61]. Unintentional weight loss is associated with frailty [55], lower levels of self-reported physical health [62], hospital readmission [63, 64], and falls [65, 66]. Unintentional weight loss is a risk factor for malnutrition (i.e., undernutrition) [67] and increases the risk for death [52, 68–70]. If unintentional weight loss is from total starvation, death occurs when the maximal weight loss is approximately 30 % of initial body weight [45].

Height. Since older adults may lose height as part of normal aging [71], it is important to measure height (without shoes) at least on a yearly basis. Alternatives to standing height include demi-span measurement [71] or knee-height with a caliper when an older adult has difficulty standing erect [71, 72].

Body Mass Index. Body mass index is an indicator of body fatness. Compared to young adults of the same BMI, older adults have more body fat, and BMI may overestimate health risks with higher BMIs in the elderly [48]. BMI is the best metric to determine if weight for height is within normal limits since ideal body weight tables are not norm-based for older adults [73, 74]. However, the recommendations for normal BMI for older adults are conflicting (Table 3.2).

Body Composition

Important changes in body composition occur with normal aging. The decline in lean mass, mostly muscle, with and without intentional weight loss [48], starts in the third decade at a rate of 0.3 kg/year [50]. By the age of 70 there is an estimated loss of 40 % of lean mass since age 30 [75]. Loss of lean mass is attributed in part to physiological aging from physical inactivity, age-related declines in anabolic hormones, adverse effects of free radicals, increased cytokines, and effects of acute illness [50]. Both men and women lose lean body mass with aging [76], although women lose lean mass to a lesser

Source	Normal BMI (kg/m ²) for older adults
Centers for Medicare and Medicaid Services [101]	≥23 and <30
Medline Plus [102]	25–27
National Heart, Lung, and Blood Institute [103] ^a	18.5–24.9
Nutrition Screening Initiative [30]	22–27

Table 3.2 BMI recommendations for older adults

^aNational Heart, Lung, and Blood Institute Guidelines do not differentiate BMI recommendations for young, middle-aged, and older adults

extent than men [76]. Individuals \geq 75 years of age lose more lean mass over time (men: 0.4 kg/year; women: 0.2 kg/year) than those aged 65–74 years [76].

Lean mass is replaced by fat until around age 65–70 [50] and the distribution of fat changes with aging. Subcutaneous fat increases up to around age 60 [77] and lower circumferences in arms and legs are observed in older versus younger adults [77]. In those over 65 years of age there is an increase in waist diameters [77], higher waist to hip or waist to thigh ratios [76, 77], and an increase in abdominal adiposity [76, 77]. Changes in body composition with aging are also dependent on other age-related factors, such as acute and chronic illness, physical activity, medications, hormone, and nutrient intake [76]. Increased physical activity during aging may attenuate the loss of lean mass [76]. Most experts agree that weight loss should not be dismissed as part of normal aging [48, 50, 68], and that measurement of weight loss cannot distinguish between alterations in lean versus fat mass [68].

A variety of methods are available to measure body composition in older adults. Specialized measurements such as whole-body immersion, air-displacement plethysmography, neutron activation analysis, and dilution techniques are accurate and precise, but are available only in specialized research facilities [78]. Furthermore, frail older adults may not be able to withstand the testing procedures. Computed tomography, MRI scanning, and dual energy X-ray absorptiometry (DXA) are also used to assess body composition. Whole-body DXA is considered the gold standard in epidemiologic cohort studies on aging, is generally well tolerated, clinically applicable, and has minimal radiation exposure [78]. Bioelectrical impedance analysis (BIA) is relatively inexpensive, safe, portable, and easy to perform in the clinical setting [78]. Anthropometric measurements are the most inexpensive body composition measures and are easy to obtain in clinical practice. These measurements include skinfold thickness (estimate of subcutaneous body fat) as well as upper arm and calf circumference measurements (estimates muscle mass). While researchers continue to advance the field of body composition, these methods are not routinely integrated into clinical practice. Normal ranges need to be developed for the elderly. When suitable, body composition measurements should be performed over time to assess changes in fat and muscle mass by appropriately trained individuals.

Physical Exam

Physical signs of malnutrition should be noted during a physical exam: loss of subcutaneous fat (orbital, triceps over rib cage); muscle loss (wasting of the temples, clavicles, shoulders, interosseous muscles, scapula, and calf); and fluid accumulation (extremities, vulvar/scrotal edema, or ascites) [67]. Diminished immune response, cognitive dysfunction, osteopenia, and anemia have also been associated with protein malnutrition [79, 80]. Protein malnourished individuals may develop anasarca or lower extremity pitting edema due to loss of oncotic pressure to maintain intravascular volume, as well as dry, atrophic skin and dull hypopigmented fragile hair. Protein malnutrition may be physically indistinguishable from cachexia and sarcopenia. Starvation or chronic protein malnutrition can be reversed with refeeding, while cachexia and sarcopenia are not easily treated with increased calorie intake [56].

	Physical findings	
Fat-soluble vitamin	deficiency	
A [105]	Night blindness, xerophthalmia, decrease immune function, dry skin, follicular hyperkeratosis, poor wound healing	
E [106]	Spinocerebellar ataxia, areflexia or decreased deep tendon reflexes, peripheral neuropathy, myopathy, impairment of proprioception and vibratory sense, retinopathy, extraocular muscle paresis	
D	Osteomalacia, osteoporosis, frequent falls, increased risk for vertebral and nonvertebral fractures	
K	Increased risk for bleeding	
Water-soluble vitam	in deficiency	
B1 (thiamine)	Wernicke's encephalopathy (confusion, ataxia and nystagmus), Korsakoff's syndrome (retrograde amnesia, confabulation, inability to learn new things), beriberi, foot-drop or wrist-drop from nerve damages	
B2 (riboflavin)	Corneal neovascularization, cheilosis, glossitis	
B3 (niacin)	Diarrhea, cognitive impairment (delirium, memory loss, depression, mania and paranoia), dermatitis (bullae, vesicles, intertrigo, or chronic hypertrophy with hyperpigmentation)	
B6 (pyridoxine)	Stomatitis, glossitis, angular cheilosis, sideroblastic anemia	
B12	Peripheral neuropathy, depression, cognitive dysfunction, megaloblastic anemia	
C [107]	Poor wound healing, reduced collagen cross-linking, hemarthrosis, perifollicular petechiae, hypertrophic gingivitis, gingival bleeding, ecchymoses, petechiae, nail bed splinter hemorrhages	
Folate	Megaloblastic anemia	
Minerals		
Copper [108]	Muscle weakness, ataxia, neuropathy, cognitive deficits, abnormally formed hair, depigmentation of the skin	
Iodine [109]	Goiter, hypothyroidism	
Iron	Microcytic, hypochromic anemia, lethargy	
Manganese [84]	Scaly dermatitis, dyslipidemia	
Selenium [100, 110]	Cardiomyopathy, skeletal muscle dysfunction, mood disorders, decreased immune function	
Zinc [111, 112]	Poor wound healing, decreased immune function, impotence, hypogonadism, decreased taste, alopecia, acrodermatitis enteropathica	

Table 3.3 Physical findings of micronutrient deficiencies [104]

Older adults with protein calorie malnutrition may also experience micronutrient deficiencies. Physical findings associated with vitamin and trace mineral deficiencies are detailed in Table 3.3. While vitamin D is most notable for its impact on calcium and bone physiology, research on Vitamin D deficiency has been expanding due to its pleiotropic clinical effects. Vitamin D deficiencies have been linked to increased risk for arterial disease, heart failure, insulin resistance, poor glycemic control, and even cancer [81–83]. However, most notable physiological signs of vitamin D deficiency are osteomalacia, osteoporosis, and increased risk for fall, balance problems, and fractures. Although deficiency in manganese is very rare, one clinical study showed that it is associated with dyslipidemia and scaly dermatitis [84].

Visceral Proteins

While serum albumin, transferrin, and prealbumin are visceral protein measures, they lack sensitivity and specificity as indicators of nutrition status during inflammation [67, 72]. In order to determine if the depleted measures of albumin and prealbumin reflect malnutrition, it is recommended to obtain a C-reactive protein level [72]. If C-reactive protein is elevated and albumin and prealbumin are low then the changes in albumin and prealbumin are due to an underlying inflammation [72].

Dietary Intake

Dietary intake can be assessed with a modified diet history or 24-h food recall [72], though food recall may be inaccurate especially in older adults with cognitive impairment. If the older adult is in a health care setting, a 3-day calorie count [71] can help determine current intake from total calories as well as protein, fat, and carbohydrates. Older adults may have diminished dietary intake due to loss of appetite related to side effects of medications, depression, or anorexia of aging [27] (see also Chap. 18). Also chronic conditions that lead to an impairment in ADL or IADL may limit the ability to physically shop, cook, or feed self [71]. Dysphagia, difficulty swallowing, and poor oral health (cavities, gum disease, missing teeth) can limit dietary intake [71]. Poor oral health is a special concern in older adults since dental coverage is not provided under Medicare and many older adults may not have the money to visit the dentist (see also Chap. 5).

Functional Status

Physical performance is a component of a geriatric assessment. Unintentional weight loss and malnutrition can negatively impact muscle strength and muscle mass leading to frailty and physical disability. Hand-grip strength, an indicator of upper body muscle function, is used as an outcome variable in nutrition intervention studies as well as a marker of nutritional status [85]. Hand-grip strength measurements are noninvasive, portable, and can easily be measured at the bed side [86]. Ergonomic characteristics vary with each dynamometer, and the Jamar is considered as the "gold standard" [86]. The American Society for Parenteral and Enteral Nutrition and the Academy of Nutrition and Dietetics recommend reduced hand-grip strength as a clinical indicator of malnutrition [67]. However, there is a need to develop standardized measurement protocols as well as normative data to widely translate the use of hand-grip measurements to clinical practice [67, 85].

Medications

Side effects of medications can contribute to weight loss in geriatric patients [87, 88]. Common prescription and over-the-counter medications associated with unintentional weight loss are listed in Table 3.4. Some medications can cause dry mouth, and alter taste or smell, which can lead to decreased appetite. Other medications can slow gastric motility causing a sensation of persistent fullness leading to decreased food intake. Reviewing all medications, including over-the-counter medications, vitamins, and prescription medications in patients who are losing weight is recommended.

Psychosocial and Economic Factors

Poverty or limited income can impact shopping patterns; for example, financial limitations and/or lack of transportation may reduce availability of fresh fruits or vegetables and high quality protein foods [71]. Older adults with reduced income have also reported limiting the number of meals eaten each day for economic reasons [71]. Older adults who are primary caretakers of their grandchildren are more likely to have food insecurity [89]. Psychosocial factors can contribute to unintentional weight loss by either decreasing appetite, or reducing dietary intake. Older adults who live alone may be

Nutritional alteration	Medications
Anorexic	Amantadine, amphetamine, benzodiazepines, digoxin, gold, levodopa, metformin, nicotine, opiates, SSRIs, theophylline
Altered taste or smell	ACE inhibitors, calcium channel blockers, spironolactone, iron, levodopa, pergolide, selegiline, opiates, gold, allopurinol
Dry mouth	Anticholinergic, diuretics, antihistamines
Nausea/vomiting	Antibiotics (atovaquone, ciprofloxacin clarithromycin, doxycycline, ethambutol, metronidazole, ofloxacin, pentamidine, rifabutin, tetracycline)
	Bisphosphonates, digoxin, levodopa, opiates, tricyclic antidepressants, SSRI
Slow gastric motility	Antibiotic (erythromycin)

 Table 3.4
 Nutrition-related side effects of medications [87, 88, 96, 113]

SSRI selective serotonin reuptake inhibitors, ACE angiotensin converting enzyme

lonely and lose the desire to eat [71]. Grief due to loss of a spouse can impact desire to eat or enjoyment in eating [71]. If the wife was the primary caretaker of the home, older men who lose their spouse may not be accustomed to shopping for food, preparing meals, or cooking. Depression is a major contributor to decreased appetite in older adults [71].

Unintentional weight loss is a classic sign of Alzheimer's disease (AD) [90–92], occurring in 25–90 % of patients across the stages of the disease [52, 92, 93]. AD is considered a pro-inflammatory condition and elevations in cytokines can contribute to weight loss and wasting associated with AD [94]. While the exact mechanism is unclear, unintentional weight loss starts in the preclinical stage [90]. Neuropathological changes in the brain contributing to weight loss in AD include changes in feeding behavior and memory [90–92], disturbed appetite signaling [90], lower orexigenic factor concentrations [90], volitional swallowing disorders [91], and alterations in taste and smell [91]. Also, side effects of medications used to treat AD can impact appetite or ability to eat [90, 92]. Behavioral changes at each stage of the AD can profoundly impact dietary intake and contribute to unintentional weight loss.

Comorbid Conditions

A review of the medical history can identify comorbid conditions which are associated with unintentional weight loss and malnutrition. Medical conditions associated with unintentional weight loss are listed in Table 3.5. Involuntary weight loss can be categorized into three causes: malignant, nonmalignant, and idiopathic. The landmark studies on unintentional weight loss patients [95, 96] from the 1980s showed that 19–36.6 % of patients had a malignancy, 40.5 % had nonmalignant disease, and in 23.2–35 % no underlying causes could be found despite extensive work-up. The most common malignant diseases were gastrointestinal in origin (pancreatic, hepatic, esophageal, and colorectal) followed by genitourinary, hematologic, lung, breast, and brain [96]. For nonmalignant causes of weight loss, gastrointestinal disorders were frequently noted (26 %) and psychiatric disorders accounted for 10.3 % of involuntary weight loss have underlying neoplasms; GI malignancies accounted for majority of the cases [87, 97, 98]. Twenty-three percent of patients had underlying psychiatric illness and nonmalignant GI causes were also more commonly seen in patients with unintentional weight loss. In approximately 11–25 % of cases, the cause of weight loss was not determined. For geriatric patients, underlying dementia could lead to unintentional weight loss, as previously noted.

	X7 11
Malignant causes	Nonmalignant causes
Gastrointestinal malignancy (esophageal, pancreatic, gastric cancer, colorectal, hepatic, biliary tract)	GI disorders (motility or swallow dysfunction, mesenteric ischemia, peptic ulcers, inflammatory bowel disease, chronic liver disease, chronic pancreatitis)
Lung	Congestive heart failure
Lymphoma/leukemia	Dementia
Prostate	COPD
Ovarian	Endocrine disorder (diabetes mellitus, hyperthyroidism, hyperparathyroidism)
Bladder	Stroke
Breast	End-stage renal failure
Brain	Infectious disease (tuberculosis, HIV)
	Alcoholism
	Connective tissue diseases (rheumatoid arthritis, polymalgia rheumatica)
	Oral or dental problems
Psychiatric causes	Unknown causes
Depression	
Anxiety disorder	

Table 3.5 Comorbid conditions contributing to involuntary weight loss in older adults [87, 88, 96, 113]

Diagnosing Malnutrition

The purpose of nutrition screening and assessment is to identify findings associated with malnutrition (undernutrition). Over the years, researchers and clinicians used a variety of clinical criteria to define malnutrition, as well as a variety of terminologies. Marasmus, kwashiorkor, and mixed-marasmus kwashiorkor were terms developed to classify macronutrient deficiencies in children during famine, and subsequently became used to characterize and diagnose disease-related malnutrition in adults. Leaders from the American Society for Parenteral and Enteral Nutrition (ASPEN) and the European Society for Parenteral and Enteral Nutrition and disease-related malnutrition [99]. Inflammation has long been recognized as having a negative impact on nutrition status due to breakdown of skeletal muscle [99] and is now the cornerstone of the new adult malnutrition classification. ASPEN and the Academy of Nutrition and Dietetics (The Academy) published a consensus statement in 2012 on characteristics for the identification and documentation of adult malnutrition (undernutrition) [67], which is also applicable to older adults.

The three types of adult malnutrition are: starvation-related malnutrition or malnutrition in the context of social or environmental circumstances (i.e., pure chronic starvation, anorexia nervosa), chronic disease-related malnutrition (inflammation of a mild to moderate degree, for example, conditions such as organ failure, pancreatic cancer, rheumatoid arthritis, sarcopenic obesity), and acute disease-related malnutrition (acute inflammation with a marked inflammatory response, for example, major infection, burns, trauma, closed head injury) [67]. Each of the three types of malnutrition is further characterized as nonsevere/moderate and severe. Clinical characteristics include: decreased energy intake, weight loss, loss of body fat and muscle mass, fluid accumulation, and reduced grip strength [67]. A summary of the clinical characteristics to diagnosis malnutrition and associated ICD-9 codes is found in Table 3.6. ASPEN and The Academy recommend for the diagnosis of severe or nonsevere malnutrition (of either type, namely, starvation-related malnutrition, chronic disease-related malnutrition, or acute disease-related malnutrition) at least two of the six characteristics should be present [67]. These recommended clinical characteristics of adult malnutrition (see Table 3.6) will
Clinical ch	aracteristics ^a						ICD	-9 Coo	les	
	Weight (amount weight loss/time)	Estimated energy intake required	Subcutaneous fat	Muscle mass	Edema	Hand grip strength	260	261	262	263.0
Starvation	(malnutrition in the	e context of so	cial or environm	ental circu	mstances	;)				
Moderate	5 %/1 month 7.5 %/3 months 10 %/6 months 20 %/1 year	<75 % for ≥ 3 months	↓	ţ	+	NA	NA	NA	No	Yes
Severe	>5 %/1 month >7.5 %/3 months >10 %/6 months >20 %/1 year	$\leq 50 \%$ for ≥ 1 month	ţţŢ	↓↓↓	+++	Reduced	NA	NA	Yes	No
Chronic di	sease-related malni	utrition								
Moderate	5 %/1 month 7.5 %/3 months 10 %/6 months 20 %/1 year	<75 % for ≥ 1 month	ţ	Ţ	+	NA	NA	NA	No	Yes
Severe	>5 %/1 month >7.5 %/3 months >10 %/6 months >20 %/1 year	$\leq 75 \%$ for ≥ 1 month	†††	ţ↓↓	++	Reduced	NA	NA	Yes	No
Acute dise	ase-related malnutr	ition								
Moderate	1–2 %/1 week 5 %/1 month 7.5 %/3 months	<75 % for >7 days	Ţ	ţ	+	NA	NA	NA	No	Yes
Severe	>2 %/1 week >5 %/1 month >7.5 %/3 months	≤50 % for ≥5 days	↓↓	↓↓	++	Reduced	NA	NA	Yes	No

Table 3.6 Clinical characteristics and associated reimbursement codes for adult malnutrition

Adapted from [67]

NA, not applicable; \downarrow , mild decrease; $\downarrow\downarrow$, moderate decrease; $\downarrow\downarrow\downarrow\downarrow$, severe decrease; +, present to a mild degree; ++, present to a moderate degree; +++, present to a severe degree

^aA minimum of two of the six characteristics is recommended for the diagnosis of either severe or nonsevere malnutrition

continually be updated as new evidence becomes available [67]. It should be noted that acute phase proteins are no longer considered valid markers of nutrition status and instead are associated with the degree of inflammation [99].

Conclusion: Nutrition Screening and Assessment Across the Health Professions

Nutrition education and training in many health professions programs is inadequate, especially when it comes to geriatric nutrition. The Institute of Medicine noted, "Poor nutrition is prevalent among seniors, but most professionals are still not trained in the nutritional needs of older adults" [3]. Recently, the National Heart Lung and Blood Institute of the National Institutes of Health convened a "Working Group on Future Directions for Implementing Nutrition Across the Continuum of Medical and Health Professions Education and Training, and Research" (http://www.nhlbi.nih.gov/meetings/workshops/nutrition.htm).

The guiding principles summarized in the executive summary of this 2-day meeting are applicable to nutrition screening and nutrition assessment of older adults:

- Health professionals should implement recommended nutrition practices and promote current Dietary Guidelines for the prevention and treatment of disease with all patients.
- Health professionals should understand the fundamentals of the Nutrition Care Process, including assessing nutritional status and dietary intakes, diagnosing nutrition-related problems, and implementing, and monitoring and evaluating nutrition care plans. The roles and responsibilities of different health professionals—when to refer, how to interact, and how multidisciplinary teams work—need to be understood.
- The dietitian plays an important role on the health-care team and can help provide training for other health professionals as well as partner with physicians in the nutrition assessment, therapeutic recommendations, and joint follow-up of patients.
- Inter-professional nutrition education is critical to instill a team approach to teaching, training and learning and to patient care.
- Health professionals should know where to access nutrition education and advocacy resources and should use innovative approaches, including online resources and case-based approaches to enhance learning.
- Nutrition education in medical/professional schools should strive for a longitudinal integrated approach rather than single courses and involve a collaborative effort of multiple stakeholders.
- Research is needed to identify and validate strategies for providing nutrition education to health professional students and practicing clinicians.
- Health professionals should understand the role of evidence-based research in the development of diet and nutrition guidelines for public health and must be able to apply that knowledge toward better patient outcomes.

Nutrition screening and assessment is a responsibility of all health professionals (physicians, dietitians, nurses, pharmacists, physical therapists, occupational therapists, speech language pathologists) who provide care to older adults. Nutritional interventions should be implemented in a timely fashion when an older adult is malnourished. Under optimal circumstances, the older adult should be referred to a registered dietitian (RD) for an in-depth nutrition assessment and nutritional plan of care. However, work-force and policy issues, such as limited reimbursement for medical nutrition therapy provided by RDs in independent practice in primary care, home care, or community settings, can lead to long waits for consultation with an RD outside of the hospital setting. Therefore, all health professionals who provide care for older adults should develop competence in nutrition screening and assessment, and be knowledgeable of evidence-based diet and nutrition guidelines for older adults. Furthermore, there is an urgent need to increase efforts to advocate for expanded reimbursement for medical nutrition therapy for RDs who specialize in geriatrics.

Clinical recommendations include:

- Nutrition screening should be conducted upon admission to the hospital and long-term care settings, upon first home care visit, enrollment in community-based programs and during annual primary care visits.
- Recommended nutrition screening tools are the Mini-Nutritional Assessment-Short Form and the Determine Checklist.
- Nutrition assessment should be performed for any older adult at nutrition risk.
- A weight loss of 5 % of usual body weight over 6–12 months is the most widely accepted definition for clinically important weight loss in older adults.
- Nutrition assessment is a comprehensive exam focused on weight changes, body composition status, physical examination, comorbid conditions, functional status, medication history, dietary intake, and psychosocial and economic status.

References

- Kamp BJ, Wellman NS, Russell C, American Dietetic Association, American Society for Nutrition, and Society for Nutrition Education. Position of the American Dietetic Association, American Society for Nutrition, and Society for Nutrition Education: food and nutrition programs for community-residing older adults. J Am Diet Assoc. 2010;110(3):463–72.
- Federal Interagency Forum on Aging-Related Statistics. Older Americans 2012: key indicators of well-being. Washington, DC: U.S. Government Printing Office; 2012.
- 3. Institute of Medicine. Retooling for an aging America: building the health care workforce. Washington, DC: National Academies Press; 2008.
- Kaiser MJ, Bauer JM, Ramsch C, Uter W, Guigoz Y, Cederholm T, et al. Frequency of malnutrition in older adults: a multinational perspective using the mini nutritional assessment. J Am Geriatr Soc. 2010;58(9):1734–8. PubMed PMID: 20863332. Epub 2010/09/25. eng.
- Edington J, Boorman J, Durrant ER, Perkins A, Giffin CV, James R, et al. Prevalence of malnutrition on admission to four hospitals in England. The Malnutrition Prevalence Group. Clin Nutr. 2000;19(3):191–5. PubMed PMID: 10895110.
- Singh H, Watt K, Veitch R, Cantor M, Duerksen DR. Malnutrition is prevalent in hospitalized medical patients: are housestaff identifying the malnourished patient? Nutrition. 2006;22(4):350–4. PubMed PMID: 16457988. Epub 2006/02/07. eng.
- Westergren A, Unosson M, Ohlsson O, Lorefalt B, Hallberg IR. Eating difficulties, assisted eating and nutritional status in elderly (>or =65 years) patients in hospital rehabilitation. Int J Nurs Stud. 2002;39(3):341–51. PubMed PMID: 11864657.
- Kagansky N, Berner Y, Koren-Morag N, Perelman L, Knobler H, Levy S. Poor nutritional habits are predictors of poor outcome in very old hospitalized patients. Am J Clin Nutr. 2005;82(4):784–91. quiz 913–4. PubMed PMID: 16210707. Epub 2005/10/08. eng.
- Heersink JT, Brown CJ, Dimaria-Ghalili RA, Locher JL. Undernutrition in hospitalized older adults: patterns and correlates, outcomes, and opportunities for intervention with a focus on processes of care. J Nutr Elder. 2010; 29(1):4–41.
- Keller HH. Nutrition and health-related quality of life in frail older adults. J Nutr Health Aging. 2004;8(4):245–52. PubMed PMID: 15316589. Epub 2004/08/19. eng.
- 11. Keller HH, Ostbye T, Goy R. Nutritional risk predicts quality of life in elderly community-living Canadians. J Gerontol A Biol Sci Med Sci. 2004;59(1):68–74. PubMed PMID: 14718488. Epub 2004/01/14. eng.
- Agarwal N, Acevedo F, Leighton LS, Cayten CG, Pitchumoni CS. Predictive ability of various nutritional variables for mortality in elderly people. Am J Clin Nutr. 1988;48(5):1173–8. PubMed PMID: 3189202.
- Constans T, Bacq Y, Brechot JF, Guilmot JL, Choutet P, Lamisse F. Protein-energy malnutrition in elderly medical patients. J Am Geriatr Soc. 1992;40(3):263–8. PubMed PMID: 1538047.
- 14. Haydock DA, Hill GL. Impaired wound healing in surgical patients with varying degrees of malnutrition. J Parenter Enteral Nutr. 1986;10(6):550–4. PubMed PMID: 3098996.
- Klidjian AM, Foster KJ, Kammerling RM, Cooper A, Karran SJ. Relation of anthropometric and dynamometric variables to serious postoperative complications. Br Med J. 1980;281(6245):899–901. PubMed PMID: 7427501.
- 16. Dwyer J, editor. Screening older Americans nutritional health: current practices and future possibilities. Washington, DC: Nutrition Screening Initiative; 1991.
- Ek AC, Larsson J, von Schenck H, Thorslund S, Unosson M, Bjurulf P. The correlation between anergy, malnutrition and clinical outcome in an elderly hospital population. Clin Nutr. 1990;9(4):185–9. PubMed PMID: 16837354. Epub 1990/08/01. eng.
- Langkamp-Henken B, Hudgens J, Stechmiller JK, Herrlinger-Garcia KA. Mini nutritional assessment and screening scores are associated with nutritional indicators in elderly people with pressure ulcers. J Am Diet Assoc. 2005;105(10):1590–6. PubMed PMID: 16183360.
- van Venrooij LM, Verberne HJ, de Vos R, Borgmeijer-Hoelen MM, van Leeuwen PA, de Mol BA. Postoperative loss of skeletal muscle mass, complications and quality of life in patients undergoing cardiac surgery. Nutrition. 2012;28(1):40–5. PubMed PMID: 21621393. Epub 2011/05/31. eng.
- Mueller C, Compher C, Ellen DM, American Society for Parenteral and Enteral Nutrition Board of A.S.P.E.N Directors. A.S.P.E.N clinical guidelines: nutrition screening, assessment, and intervention in adults. J Parenter Enteral Nutr. 2011;35(1):16–24. PubMed PMID: 21224430.
- A.S.P.E.N. Board of Directors and Clinical Practice Committee. Definition of terms, style, and conventions used in A.S.P.E.N. Board of Directors-approved documents, 2010 January 2, 2013.
- 22. Skipper A, Ferguson M, Thompson K, Castellanos VH, Porcari J. Nutrition screening tools: an analysis of the evidence. J Parenter Enteral Nutr. 2012;36(3):292–8. PubMed PMID: 22045723.

- Joint Commission on Accreditation of Healthcare Organizations. Comprehensive accreditation manual for hospitals. Chicago, IL: Joint Commission on Accreditation of Healthcare Organizations; 2007.
- 24. Joint Commission on Accreditation of Healthcare Organizations. Standards frequently asked questions: nutritional, functional, and pain assessment and screens. http://www.jointcommissioni.org/standards_information/ jcfaqdetails.aspx? Accessed 2 Jan 2013.
- 25. Wallace JI, Schwartz RS. Involuntary weight loss in elderly outpatients: recognition, etiologies, and treatment. Clin Geriatr Med. 1997;13(4):717–35. PubMed PMID: 9354751.
- Payette H, Coulombe C, Boutier V, Gray-Donald K. Weight loss and mortality among free-living frail elders: a prospective study. J Gerontol A Biol Sci Med Sci. 1999;54(9):M440–5. PubMed PMID: 10536646. Epub 1999/ 10/28. eng.
- Morley JE. Anorexia and weight loss in older persons. J Gerontol A Biol Sci Med Sci. 2003;58(2):131–7. PubMed PMID: 12586850. Epub 2003/02/15. eng.
- 28. American Dietetic Association. Unintended weight loss (UWL) in older adults evidence-based nutrition practice guideline Chicago, IL: American Dietitic Association. 2009 [cited 2013 1/06]. Available from: www.guideilnes.gov.
- Posner BM, Jette AM, Smith KW, Miller DR. Nutrition and health risks in the elderly: the nutrition screening initiative. Am J Public Health. 1993;83(7):972–8. PubMed PMID: 8328619.
- White JV, Dwyer JT, Posner BM, Ham RJ, Lipschitz DA, Wellman NS. Nutrition screening initiative: development and implementation of the public awareness checklist and screening tools. J Am Diet Assoc. 1992;92(2):163–7. PubMed PMID: 1737899.
- National Resource Center on Nutrition, Physical Activity & Aging. Older Americans act nutrition programs toolkit. Miami, FL: Florida International University; 2005.
- Centers for Medicare & Medicaid Services. Outcome and assessment information set. Available from: http://www. cms.gov/Medicare/Quality-Initiatives-Patient-Assessment-Instruments/OASIS/index.html.
- 33. The Joint Commission. Comprehensive accreditation manual for home care. Chicago, IL: The Joint Commission; 2012.
- Green SM, Watson R. Nutritional screening and assessment tools for older adults: literature review. J Adv Nurs. 2006;54(4):477–90. PubMed PMID: 16671977.
- 35. Guigoz Y. The Mini Nutritional Assessment (MNA) review of the literature—what does it tell us? J Nutr Health Aging. 2006;10(6):466–85. discussion 85–7. PubMed PMID: 17183419. Epub 2006/12/22. eng.
- 36. Kaiser MJ, Bauer JM, Ramsch C, Uter W, Guigoz Y, Cederholm T, et al. Validation of the Mini Nutritional Assessment short-form (MNA-SF): a practical tool for identification of nutritional status. J Nutr Health Aging. 2009;13(9):782–8. PubMed PMID: 19812868. Epub 2009/10/09. eng.
- Skates JJ, Anthony PS. Identifying geriatric malnutrition in nursing practice: the Mini Nutritional Assessment (MNA(R))-an evidence-based screening tool. J Gerontol Nurs. 2012;38(3):18–27. quiz 8–9. PubMed PMID: 22329392. Epub 2012/02/15. eng.
- 38. National Guideline Clearinghouse. Unintended weight loss in older adults (UWL) evidence-based nutrition practice guideline. Rockville, MD: Agency for Healthcare Research and Quality (AHRQ). [1/9/2013]. Available from: http://www.guidelines.gov/content.aspx?id=15436&search=weight+loss+and+older+adult.
- 39. Thomas DR. Nutrition assessment in long-term care. Nutr Clin Pract. 2008;23(4):383-7. PubMed PMID: 18682589.
- 40. Locher JL, Wellman NS. "Never the Twain Shall Meet": dual systems exacerbate malnutrition in older adults recently discharged from hospitals. J Nutr Gerontol Geriatr. 2011;30(1):24–8.
- Sahyoun NR, Anyanwu UO, Sharkey JR, Netterville L. Recently hospital-discharged older adults are vulnerable and may be underserved by the Older Americans Act Nutrition Program. J Nutr Elder. 2010;29(2):227–40. Epub 2010/05/18. eng.
- 42. DiMaria-Ghalili R, Amellea E. Assessing nutrition in older adults. New York University, College of Nursing: Hartford Institute for Geriatric Nursing. 2012 [cited 2013 01-06]. Issue Number 9. Available from: www. ConsultGeriRN.org.
- Wang ZM, Pierson Jr RN, Heymsfield SB. The five-level model: a new approach to organizing body-composition research. Am J Clin Nutr. 1992;56(1):19–28. PubMed PMID: 1609756. Epub 1992/07/11. eng.
- Vanderschueren S, Geens E, Knockaert D, Bobbaers H. The diagnostic spectrum of unintentional weight loss. Eur J Intern Med. 2005;16(3):160–4. PubMed PMID: 15967329. Epub 2005/06/22. Eng.
- Gibson RS. Anthropometric assessment of body size. In: Gibson RS, editor. Principles of nutritional assessment. New York, NY: Oxford University Press; 2005.
- Lindenfeld J, Albert NM, Boehmer JP, Collins SP, Ezekowitz JA, Givertz MM, et al. HFSA 2010 comprehensive heart failure practice guideline. J Card Fail. 2010;16(6):e1–194. PubMed PMID: 20610207. Epub 2010/07/09. eng.
- Baumgartner RN, Heymsfield SB, Roche AF. Human body composition and the epidemiology of chronic disease. Obes Res. 1995;3(1):73–95. PubMed PMID: 7712363. Epub 1995/01/01. eng.
- Miller SL, Wolfe RR. The danger of weight loss in the elderly. J Nutr Health Aging. 2008;12(7):487–91. PubMed PMID: 18615231. Epub 2008/07/11. eng.

- Chumlea WC, Garry PJ, Hunt WC, Rhyne RL. Distributions of serial changes in stature and weight in a healthy elderly population. Hum Biol. 1988;60(6):917–25. PubMed PMID: 3235081. Epub 1988/12/01. eng.
- Wallace JI, Schwartz RS. Epidemiology of weight loss in humans with special reference to wasting in the elderly. Int J Cardiol. 2002;85(1):15–21. PubMed PMID: 12163206. Epub 2002/08/07. eng.
- Hill GL. Jonathan E. Rhoads Lecture. Body composition research: implications for the practice of clinical nutrition. JPEN J Parenter Enteral Nutr. 1992;16(3):197–218. PubMed PMID: 1501350. Epub 1992/05/01. eng.
- Wallace JI, Schwartz RS, LaCroix AZ, Uhlmann RF, Pearlman RA. Involuntary weight loss in older outpatients: incidence and clinical significance. J Am Geriatr Soc. 1995;43(4):329–37. PubMed PMID: 7706619.
- Wu JM, Lin MH, Peng LN, Chen LK, Hwang SJ. Evaluating diagnostic strategy of older patients with unexplained unintentional body weight loss: a hospital-based study. Arch Gerontol Geriatr. 2011;53(1):e51–4. PubMed PMID: 21071102. Epub 2010/11/13. eng.
- 54. Institute of Medicine. Retooling for an aging America: building the health care workforce. Washington, DC: National Academies Press; 2010.
- Fried LP, Tangen CM, Walston J, Newman AB, Hirsch C, Gottdiener J, et al. Frailty in older adults: evidence for a phenotype. J Gerontol A Biol Sci Med Sci. 2001;56(3):M146–56. PubMed PMID: 11253156.
- 56. Thomas DR. Unintended weight loss in older persons. Aging Health. 2008;4(2):191-200.
- Thompson Martin C, Kaiser Jones J, Stotts NA, Sivarajan Froelicher E. Community-living elder's views on normal and low weight. J Nutr Health Aging. 2008;12(1):45–8. PubMed PMID: 18165844. Epub 2008/01/01. eng.
- Locher JL, Robinson CO, Bailey FA, Carroll WR, Heimburger DC, Magnuson JS, et al. The contribution of social factors to undereating in older adults with cancer. J Support Oncol. 2009;7(5):168–73. PubMed PMID: 19831160. Epub 2009/10/17. eng.
- Newman AB, Lee JS, Visser M, Goodpaster BH, Kritchevsky SB, Tylavsky FA, et al. Weight change and the conservation of lean mass in old age: the Health, Aging and Body Composition Study. Am J Clin Nutr. 2005;82(4):872–8. quiz 915–6. PubMed PMID: 16210719. Epub 2005/10/08. eng.
- 60. Lee JS, Kritchevsky SB, Tylavsky F, Harris T, Simonsick EM, Rubin SM, et al. Weight change, weight change intention, and the incidence of mobility limitation in well-functioning community-dwelling older adults. J Gerontol A Biol Sci Med Sci. 2005;60(8):1007–12. PubMed PMID: 16127104.
- Janssen I, Baumgartner RN, Ross R, Rosenberg IH, Roubenoff R. Skeletal muscle cutpoints associated with elevated physical disability risk in older men and women. Am J Epidemiol. 2004;159(4):413–21. PubMed PMID: 14769646. Epub 2004/02/11. eng.
- DiMaria-Ghalili RA. Changes in nutritional status and postoperative outcomes in elderly CABG patients. Biol Res Nurs. 2002;4(2):73–84. PubMed PMID: 12408213. Epub 2002/11/01. eng.
- 63. DiMaria-Ghalili RA. Changes in body mass index and late postoperative outcomes in elderly coronary bypass grafting patients: a follow-up study. Biol Res Nurs. 2004;6(1):24–36. PubMed PMID: 15230244.
- Friedmann JM, Jensen GL, Smiciklas-Wright H, McCamish MA. Predicting early nonelective hospital readmission in nutritionally compromised older adults. Am J Clin Nutr. 1997;65(6):1714–20. PubMed PMID: 9174465.
- Zoltick ES, Sahni S, McLean RR, Quach L, Casey VA, Hannan MT. Dietary protein intake and subsequent falls in older men and women: the Framingham Study. J Nutr Health Aging. 2011;15(2):147–52. PubMed PMID: 21365169. Pubmed Central PMCID: 3136106. Epub 2011/03/03. eng.
- Bales CW, Ritchie CS. Sarcopenia, weight loss, and nutritional frailty in the elderly. Annu Rev Nutr. 2002;22:309– 23. PubMed PMID: 12055348.
- 67. White JV, Guenter P, Jensen G, Malone A, Schofield M. Consensus statement of the Academy of Nutrition and Dietetics/American Society for Parenteral and Enteral Nutrition: characteristics recommended for the identification and documentation of adult malnutrition (undernutrition). J Acad Nutr Diet. 2012;112(5):730–8. PubMed PMID: 22709779. Epub 2012/06/20. eng.
- Lee CG, Boyko EJ, Nielson CM, Stefanick ML, Bauer DC, Hoffman AR, et al. Mortality risk in older men associated with changes in weight, lean mass, and fat mass. J Am Geriatr Soc. 2011;59(2):233–40. PubMed PMID: 21288234. Pubmed Central PMCID: 3403719. Epub 2011/02/04. eng.
- 69. Locher JL, Roth DL, Ritchie CS, Cox K, Sawyer P, Bodner EV, et al. Body mass index, weight loss, and mortality in community-dwelling older adults. J Gerontol A Biol Sci Med Sci. 2007;62(12):1389–92. PubMed PMID: 18166690. Pubmed Central PMCID: 2750037. Epub 2008/01/02. eng.
- Newman AB, Yanez D, Harris T, Duxbury A, Enright PL, Fried LP. Weight change in old age and its association with mortality. J Am Geriatr Soc. 2001;49(10):1309–18. PubMed PMID: 11890489.
- DiMaria-Ghalili RA, Amella E. Nutrition in older adults. Am J Nurs. 2005;105(3):40–50. quiz 50–1. PubMed PMID: 15729041. Epub 2005/02/25. eng.
- Jensen GL, Hsiao PY, Wheeler D. Adult nutrition assessment tutorial. J Parenter Enteral Nutr. 2012;36(3):267–74. PubMed PMID: 22402644. Epub 2012/03/10. eng.
- 73. Pai MP, Paloucek FP. The origin of the "ideal" body weight equations. Ann Pharmacother. 2000;34(9):1066–9. PubMed PMID: 10981254. Epub 2000/09/12. eng.

- 74. Shah B, Sucher K, Hollenbeck CB. Comparison of ideal body weight equations and published height-weight tables with body mass index tables for healthy adults in the United States. Nutr Clin Pract. 2006;21(3):312–9. PubMed PMID: 16772549. Epub 2006/06/15. eng.
- 75. Gallagher D, Ruts E, Visser M, Heshka S, Baumgartner RN, Wang J, et al. Weight stability masks sarcopenia in elderly men and women. Am J Physiol Endocrinol Metab. 2000;279(2):E366–75. PubMed PMID: 10913037. Epub 2000/07/27. eng.
- Genton L, Karsegard VL, Chevalley T, Kossovsky MP, Darmon P, Pichard C. Body composition changes over 9 years in healthy elderly subjects and impact of physical activity. Clin Nutr. 2011;30(4):436–42. PubMed PMID: 21324569. Epub 2011/02/18. eng.
- 77. Hughes VA, Roubenoff R, Wood M, Frontera WR, Evans WJ, Fiatarone Singh MA. Anthropometric assessment of 10-y changes in body composition in the elderly. Am J Clin Nutr. 2004;80(2):475–82. PubMed PMID: 15277173. Epub 2004/07/28. eng.
- Woodrow G. Body composition analysis techniques in the aged adult: indications and limitations. Curr Opin Clin Nutr Metab Care. 2009;12(1):8–14.
- Di Francesco V, Fantin F, Omizzolo F, Residori L, Bissoli L, Bosello O, et al. The anorexia of aging. Dig Dis. 2007;25(2):129–37. PubMed PMID: 17468548. Epub 2007/05/01. eng.
- Rodondi A, Ammann P, Ghilardi-Beuret S, Rizzoli R. Zinc increases the effects of essential amino acids-whey protein supplements in frail elderly. J Nutr Health Aging. 2009;13(6):491–7. PubMed PMID: 19536417. Epub 2009/06/19. eng.
- 81. de Boer IH, Levin G, Robinson-Cohen C, Biggs ML, Hoofnagle AN, Siscovick DS, et al. Serum 25-hydroxyvitamin D concentration and risk for major clinical disease events in a community-based population of older adults: a cohort study. Ann Intern Med. 2012;156(9):627–34. PubMed PMID: 22547472. Epub 2012/05/02. eng.
- van de Luijtgaarden KM, Voute MT, Hoeks SE, Bakker EJ, Chonchol M, Stolker RJ, et al. Vitamin D deficiency may be an independent risk factor for arterial disease. Eur J Vasc Endovasc Surg. 2012;44(3):301–6. PubMed PMID: 22841360. Epub 2012/07/31. eng.
- George PS, Pearson ER, Witham MD. Effect of vitamin D supplementation on glycaemic control and insulin resistance: a systematic review and meta-analysis. Diabet Med. 2012;29(8):e142–50. PubMed PMID: 22486204. Epub 2012/04/11. eng.
- Friedman BJ, Freeland-Graves JH, Bales CW, Behmardi F, Shorey-Kutschke RL, Willis RA, et al. Manganese balance and clinical observations in young men fed a manganese-deficient Diet. J Nutr. 1987;117(1):133–43. PubMed PMID: 3819860. Epub 1987/01/01. eng.
- Norman K, Stobaus N, Gonzalez MC, Schulzke JD, Pirlich M. Hand grip strength: outcome predictor and marker of nutritional status. Clin Nutr. 2011;30(2):135–42. PubMed PMID: 21035927.
- Guerra RS, Amaral TF. Comparison of hand dynamometers in elderly people. J Nutr Health Aging. 2009;13(10):907–12. PubMed PMID: 19924352.
- Alibhai SM, Greenwood C, Payette H. An approach to the management of unintentional weight loss in elderly people. CMAJ. 2005;172(6):773–80. PubMed PMID: 15767612. Pubmed Central PMCID: 552892. Epub 2005/03/16. eng.
- Stajkovic S, Aitken EM, Holroyd-Leduc J. Unintentional weight loss in older adults. CMAJ. 2011;183(4):443–9. PubMed PMID: 21324857. Pubmed Central PMCID: 3050948. Epub 2011/02/18. eng.
- IOM (Institute of Medicine). Nutrition and healthy aging in the community: workshop summary. Washington, DC: The National Academies Press; 2012.
- Smith KL, Greenwood CE. Weight loss and nutritional considerations in Alzheimer disease. J Nutr Elder. 2008;27(3–4):381–403. PubMed PMID: 19042581. Epub 2008/12/02. eng.
- Inelmen EM, Sergi G, Coin A, Girardi A, Manzato E. An open-ended question: Alzheimer's disease and involuntary weight loss: which comes first? Aging Clin Exp Res. 2010;22(3):192–7. PubMed PMID: 19940557. Epub 2009/11/27. eng.
- Hansen ML, Waldorff FB, Waldemar G. Prognostic factors for weight loss over 1-year period in patients recently diagnosed with mild Alzheimer Disease. Alzheimer Dis Assoc Disord. 2011;25(3):269–75. PubMed PMID: 21285857. Epub 2011/02/03. eng.
- Nordenram G, Ljunggren G, Cederholm T. Nutritional status and chewing capacity in nursing home residents. Aging (Milano). 2001;13(5):370–7. PubMed PMID: 11820710. Epub 2002/02/01. eng.
- 94. Jensen GL. Inflammation as the key interface of the medical and nutrition universes: a provocative examination of the future of clinical nutrition and medicine. J Parenter Enteral Nutr. 2006;30(5):453–63. PubMed PMID: 16931617.
- Marton KI, Sox Jr HC, Krupp JR. Involuntary weight loss: diagnostic and prognostic significance. Ann Intern Med. 1981;95(5):568–74. PubMed PMID: 7294545. Epub 1981/11/01. eng.
- Rabinovitz M, Pitlik SD, Leifer M, Garty M, Rosenfeld JB. Unintentional weight loss. A retrospective analysis of 154 cases. Arch Intern Med. 1986;146(1):186–7.

- Hernandez JL, Riancho JA, Matorras P, Gonzalez-Macias J. Clinical evaluation for cancer in patients with involuntary weight loss without specific symptoms. Am J Med. 2003;114(8):631–7. PubMed PMID: 12798450. Epub 2003/06/12. eng.
- Bilbao-Garay J, Barba R, Losa-Garcia JE, Martin H, Garcia de Casasola G, Castilla V, et al. Assessing clinical probability of organic disease in patients with involuntary weight loss: a simple score. Eur J Intern Med. 2002;13(4):240–5. PubMed PMID: 12067819. Epub 2002/06/18. Eng.
- 99. Jensen GL, Mirtallo J, Compher C, Dhaliwal R, Forbes A, Grijalba RF, et al. Adult starvation and disease-related malnutrition: a proposal for etiology-based diagnosis in the clinical practice setting from the International Consensus Guideline Committee. J Parenter Enteral Nutr. 2010;34(2):156–9. PubMed PMID: 20375423.
- Hawkes WC, Hornbostel L. Effects of dietary selenium on mood in healthy men living in a metabolic research unit. Biol Psychiatry. 1996;39(2):121–8. PubMed PMID: 8717610. Epub 1996/01/15. eng.
- 101. Centers for Medicare & Medicaid Services. 2012 Physician quality reporting system measures specification manual for claims and registry: reporting of individual measures. Chicago, IL: American Medical Association. 2011 [cited 2012 Nov 8]. Available from: http://www.aan.com/globals/axon/assets/9111.pdf.
- National Library of Medicine. Body mass index. Available from: http://www.nlm.nih.gov/medlineplus/ency/article/007196.htm.
- 103. National Heart, Lung, and Blood Institute Clinical guidelines on the identification, evaluation, and treatment of overweight and obesity in adults: the evidence report 1998. Available from: http://www.nhlbi.nih.gov/guidelines/ obestiy/ob_gdlns.pdf.
- Johnson KA, Bernard MA, Funderburg K. Vitamin nutrition in older adults. Clin Geriatr Med. 2002;18(4):773–99. PubMed PMID: 12608503. Epub 2003/03/01. eng.
- 105. Tang G, Qin J, Dolnikowski GG, Russell RM, Grusak MA. Golden Rice is an effective source of vitamin A. Am J Clin Nutr. 2009;89(6):1776–83. PubMed PMID: 19369372. Pubmed Central PMCID: 2682994. Epub 2009/04/17. eng.
- 106. Ueda N, Suzuki Y, Rino Y, Takahashi T, Imada T, Takanashi Y, et al. Correlation between neurological dysfunction with vitamin E deficiency and gastrectomy. J Neurol Sci. 2009;287(1–2):216–20. PubMed PMID: 19709675. Epub 2009/08/28. eng.
- 107. Raynaud-Simon A, Cohen-Bittan J, Gouronnec A, Pautas E, Senet P, Verny M, et al. Scurvy in hospitalized elderly patients. J Nutr Health Aging. 2010;14(6):407–10. PubMed PMID: 20617280. Epub 2010/07/10. eng.
- Kumar N, Gross Jr JB, Ahlskog JE. Copper deficiency myelopathy produces a clinical picture like subacute combined degeneration. Neurology. 2004;63(1):33–9. PubMed PMID: 15249607. Epub 2004/07/14. eng.
- Hetzel BS, Potter BJ, Dulberg EM. The iodine deficiency disorders: nature, pathogenesis and epidemiology. World Rev Nutr Diet. 1990;62:59–119. PubMed PMID: 2180217. Epub 1990/01/01. eng.
- 110. Rayman MP. The importance of selenium to human health. Lancet. 2000;356(9225):233–41. PubMed PMID: 10963212. Epub 2000/08/30. eng.
- 111. Langemo D, Anderson J, Hanson D, Hunter S, Thompson P, Posthauer ME. Nutritional considerations in wound care. Adv Skin Wound Care. 2006;19(6):297–8, 300, 3. PubMed PMID: 16885642. Epub 2006/08/04. eng.
- 112. Macdonald JB, Connolly SM, DiCaudo DJ. Think zinc deficiency: acquired acrodermatitis enteropathica due to poor diet and common medications. Arch Dermatol. 2012;148(8):961–3. PubMed PMID: 22911205. Epub 2012/08/23. eng.
- 113. Hernandez JL, Matorras P, Riancho JA, Gonzalez-Macias J. Involuntary weight loss without specific symptoms: a clinical prediction score for malignant neoplasm. QJM. 2003;96(9):649–55. PubMed PMID: 12925720. Epub 2003/08/20. eng.

Part II Fundamentals of Nutrition and Geriatric Syndromes

Chapter 4 Nutrition and the Aging Eye

Elizabeth J. Johnson

Key Points

- There is growing interest in the role that nutrition plays in modifying the development and/or progression of vision disorders in older persons, including age-related cataract and macular degeneration.
- Available evidence to date supports a possible protective role of several nutrients, including vitamins C and E, the carotenoids lutein and zeaxanthin, and the omega-3 fatty acids.
- Due to inconsistencies among the findings of currently available studies regarding doses and combinations of nutrients, it may be most practical to recommend specific natural diet choices rich in vitamins C and E, lutein and zeaxanthin, omega-3 fatty acids and zinc, which would also provide potential benefits from other components of these natural food sources.

Keywords Cataract • Macular degeneration • Eye disease • Retina

Introduction

Vision loss among the elderly is an important health problem. Approximately one person in three has some form of vision-reducing eye disease by the age of 65 [1]. Age-related cataract and age-related macular degeneration (AMD) are the major causes of visual impairment and blindness in the aging US population. Approximately 50 % of the 30–50 million cases of blindness worldwide result from unoperated cataracts [2, 3]. A clinically significant cataract is present in about 5 % of Caucasian Americans aged 52–64 years and rises to 46 % in those aged 75–85 years [4]. In the United States, cataract extraction accompanied by ocular lens implant is currently the most common surgical procedure done in Medicare beneficiaries [5]. Lens implantation is highly successful in restoring vision. However, the procedure is costly, accounting for 12 % of the Medicare budget and accounts for more than \$3 billion in annual health expenditures [5, 6]. For these reasons, there is much interest in the prevention of cataract as an alternative to surgery.

E.J. Johnson, Ph.D. (🖂)

Jean Mayer USDA Human Nutrition Research Center on Aging, Tufts University, 711 Washington Avenue, Boston, MA 02111, USA e-mail: Elizabeth.Johnson@tufts.edu

The prevalence of AMD also increases dramatically with age. Nearly 30 % of Americans over the age of 75 have early signs of AMD and 7 % have late stage disease, whereas the respective prevalence among people 43–54 years are 8 and 0.1 % [4]. Dry AMD is the most common form of AMD in its early or intermediate stages. It occurs in about 90 % of the people with the condition. Wet AMD is more severe than the early and intermediate stages of the dry form. Wet AMD occurs when abnormal blood vessels behind the retina start to grow under the macula. These new blood vessels can be fragile and leak blood and fluid. The blood and fluid cause the macula to swell and damage occurs rapidly. The damage may also cause scarring of the retina.

AMD is the leading cause of blindness among the elderly in industrialized countries. Because there are currently no effective treatment strategies for most patients with AMD, attention has focused on efforts to stop the progression of the disease or to prevent the damage leading to this condition [7].

Cataract and AMD share common modifiable risk factors, such as light exposure and smoking [7, 8]. Of particular interest is the possibility that nutritional counseling or intervention might reduce the incidence or retard the progression of these diseases. The components of the diet that may be important in the prevention of cataract and AMD are vitamins C and E, the carotenoids, lutein and zeaxanthin, the omega-3 polyunsaturated fatty acids eicosapentaenoic acid (EPA) and docosahexaenoic acid (DHA), and zinc.

Both vitamins C and E are found in the lens [9–11]. Omega-3 fatty acids, lutein, and zinc are highly concentrated in the eye [12-15]. Given that the lens and retina suffer oxidative damage, some of these nutrients are thought to be protective through their role as antioxidants. Vitamin E and carotenoids are lipid-soluble oxidant scavangers that protect biomembranes. Vitamin C is an important water-soluble antioxidant and also promotes the regeneration of vitamin E. Of the 20-30 carotenoids found in human blood and tissues [16], only lutein and zeaxanthin are found in the lens and retina [10, 17]. Lutein and zeaxanthin are concentrated in the macula or central region of the retina and are referred to as macular pigment. In addition to their role as antioxidants, lutein and zeaxanthin are believed to limit retinal oxidative damage by absorbing incoming blue light and/or quenching reactive oxygen species. Many putative risk-factors for AMD have been linked to a lack of macular pigment, including female gender, lens density, smoking, light iris color, obesity, and reduced visual sensitivity [7, 18]. Given that inflammation is also a component to the etiology of age-related eye disease [13, 19], the omega-3 fatty acids are thought to be protective. It has been suggested that atherosclerosis of the blood vessels that supply the retina contributes to the risk of AMD, analogous to the mechanism underlying coronary heart disease [20]. Long-chain omega-3 fatty acids may have a special role in the function of the retina in addition to their antithrombotic and hypolipidemic effects on the cardiovascular system. DHA is the omega-3 fatty acid of key interest. DHA is a major fatty acid found in the retina [21]. Rod outer segments of vertebrate retina have a high DHA content [21, 22]. Since photoreceptor outer segments are constantly being renewed, a constant supply of DHA may be required for proper retinal function and a marginal depletion may impair retinal function and influence the development of AMD.

Physiological Basis of Cataracts and AMD

The role of the lens is to transmit and focus light on the retina. Therefore, for optimal performance the lens must be transparent. The lens is an encapsulated organ without blood vessels or nerves (Fig. 4.1). The anterior hemisphere is covered by a single layer of epithelial cells containing subcellular organelles. At the lens equator the epithelial cells begin to elongate and differentiate to become fiber cells. Fully differentiated fiber cells have no organelles but are filled with proteins called crystallins, organized in a repeating lattice. The high density and repetitive spatial arrangement of crystallins produce a medium of nearly uniform refractive index with dimensions similar to light wavelengths [23].



Fig. 4.1 The human eye showing typical organization and terminology. Courtesy of National Eye Institute, National Institute of Health

Cataracts result when certain events, e.g., light exposure, cause a loss of order and results in abrupt fluctuations in refractive index causing increased light scattering and loss in transparency in the lens. It is proposed that lens opacity results from damage to lens enzymes, proteins, and membranes by activated oxygen species, e.g., hydrogen peroxide, superoxide anion, and hydroxyl free radicals, which are the results of exposure to light and other types of radiation. There are three types of cataract, defined by their location in the lens. Nuclear cataract occurs in the center, or nucleus of the lens. Cortical cataract begins at the outer rim of the lens, which is referred to as the cortex, and progresses towards the center. Posterior subcapsular cataract (PSC) occurs in the central posterior cortex, just under the posterior capsule, the membrane that envelops the lens. Nuclear cataract is the most common type of cataract [1]. It interferes with a person's ability to see distant objects and is usually the result of advancing age. Cortical cataract is most commonly seen in patients who have diabetes. PSC can be present in younger individuals and progresses more rapidly resulting in glare and blurriness [24]. This type of cataract is usually seen in patients who use steroids, or who suffer from diabetes or extreme nearsightedness.

AMD is a disease affecting the central area of the retina (macula) (see Fig. 4.1) resulting in loss of central vision. In the early stages of the disease, lipid material (drusen) accumulates in deposits underneath the retinal pigment epithelium (RPE). This is believed to arise after failure of the RPE to perform its digestive function adequately and can be seen as pale yellow spots on the retina. The presence of a few small, hard drusen is normal with advancing age. However, the presence of indistinct, soft, larger, reticular, and more numerous drusen in the macula is a common early sign of AMD [25]. The pigment of the RPE may become disturbed with areas of hyperpigmentation and hypopigmentation. In the later stages of the disease, the RPE may atrophy completely. This loss can occur in small focal areas or can be widespread. In some cases, new blood vessels grow under the RPE and occasionally into the subretinal space (exudative or neovascular AMD). Hemorrhage can occur which often results in increased scarring of the retina. The early stages of AMD are in general asymptomatic. In the later stages there may be considerable distortion of vision and complete loss of visual function, particularly in the central area of vision [7]. Although the specific pathogenesis of AMD is still unknown, chemical and light-induced oxidative damage to the photoreceptors is thought to be important in the dysfunction of the RPE. The retina is particularly susceptible to oxidative stress because of its high consumption of oxygen, its high proportion of polyunsaturated fatty acids, and its exposure to visible light. Currently, there is no treatment which can restore vision in AMD. Therefore, efforts have focused on its prevention or delay in progression.

Given the role of light damage in the etiology of cataracts and AMD, many antioxidant nutrients may prevent damage in the lens by reacting with free radicals produced in the process of light absorption as well as decrease oxidative stress in photoreceptors in the retina.

Epidemiological Studies on Dietary Intake and Blood Levels of Nutrients and Eye Disease

Studies with human subjects provide information on the strength of associations between nutritional factors and the frequency of a disease. Such studies can be a valuable means of identifying and evaluating risk factors. Although there are limitations to such studies, consistency of findings among studies lends credibility to the role that nutrition can play in age-related eye health.

Cataract

Vitamin C

Several studies have found a relationship between increased dietary vitamin C and decreased risk of cataract (Table 4.1). For example, it was observed that the prevalence of nuclear cataract was lower for men with total vitamin C intakes in the highest quintile category relative to the lowest intake quintile [26]. It has also been observed that the prevalence of cataract was about 75 % lower in persons with vitamin C intakes >490 mg/day than in those with intakes <125 mg/day [27]. Others have reported that vitamin C intake had no effect on cortical cataract, posterior subcapsular (PSC) and cataract extraction [26, 28–32].

Compared to diet, plasma concentrations of a nutrient are considered to be a better measure of nutrient status. There are many reports of an inverse relationship between plasma ascorbic acid concentrations and the prevalence of cataract [27, 33–36]. For example, the National Health and Nutrition Examination Survey II (NHANES II) correlated a 1 mg/dL increase in plasma ascorbate with a 26 % decreased risk of cataracts in older (62–70 years) Americans [33]. However, others have observed that plasma vitamin C concentrations were not associated with risk of nuclear or cortical cataract [30, 37]. Furthermore, the India-United States case control study of age-related cataracts found an increased prevalence of (PSC) and nuclear cataracts with increased plasma vitamin C concentrations [38]. In this study, however, when vitamin C status was combined with other indicators of antioxidant status (glutathione peroxidase, vitamin E, and glucose-6-phosphate dehydrogenase) the relationship became significantly protective (Table 4.2). This may reflect the synergistic actions of antioxidant systems.

Vitamin E

A protective effect of dietary vitamin E has been observed in several studies (Table 4.3). Persons in the highest quintile for vitamin E intake were reported to be 50 % less likely to undergo cataract extraction compared to those in the lowest quintile for vitamin E intake [31]. Mares-Perlman et al. [26] observed a lower prevalence of nuclear cataract in men in the highest quintile category of total vitamin E intake relative to those in the lowest vitamin E intake. Similarly, Leske et al. [39] reported that

Data analysis method	Result	Reference
Positive outcome		
Highest vs. lowest quintile (men, 104 vs. 33 mg/day)	Lower prevalence of cataract in highest quintile	Mares-Perlman et al. [26]
>490 mg/day vs. <125 mg/day	75 % lower prevalence of cataract with high intake	Jacques et al. [27]
Case-control	Lower dietary vit C in nuclear cataract	Leske et al. [39]
Highest vs. lowest quintile (767 vs. 101 mg/day)	45 % lower incidence of nuclear cataract	Tan et al. [32]
Null outcome		
Highest vs. lowest quintile (women, 171 vs. 34 mg/day)	No difference in prevalence of cataract between groups	Mares-Perlman et al. [26]
Highest vs. lowest quintile (705 vs. 70 mg/day) (women)	No difference in prevalence of cataract between groups	Hankinson et al. [29]
Highest vs. lowest quintile	No difference in prevalence of cataract extraction between groups	Tavani et al. [31]
Highest vs. lowest quartile (261.1 vs. 114.4 mg/day)	No difference in prevalence or nuclear or cortical cataract between groups	Vitale et al. [30]
Multiple logistic regression	No association of dietary vit C with nuclear or cortical cataract	Italian American Cataract Study Group [28]
Highest vs. lowest quintile (767 vs. 101 mg/day)	No difference in incidence of cortical and PSC cataract or cataract surgery	Tan et al. [32]

Table 4.1 Summary of epidemiologic studies of dietary vitamin C^a and cataract

^aRDA for adults: 60 mg/day

Table 4.2 Summary of epidemiologic studies of plasma vitamin Ca and cataract

Data analysis method	Result	Reference
Positive outcome		
Multiple logistic regression	Serum vit C inversely associated with prevalence of cataract $(p=0.03)$	Simon et al. [33]
>90 vs. <40 µmol/L	Lower prevalence of cataract with high plasma levels	Jacques et al. [27]
Increase of 1 mg/mL plasma vitamin C	26 % decrease risk of cataracts	Valero et al. [34]
Case-control	Mean plasma vit C significantly lower in cataract patients than in controls (0.96 vs. 1.12 mg/dL, respectively)	Jalal et al. [36]
Highest vs. lowest tertile >15 vs. <6.3 µmol/L	Serum vit C inversely related risk of cataract	Dherani et al. [35]
Null outcome		
Highest vs. lowest quartile	Plasma vit C levels were not associated with risk of cortical or nuclear cataract	Vitale et al. [30]
Multivariate logistic regression	No relationship between plasma vit C and cataract	Ferrigno et al. [37]
Negative outcome		
2	Higher prevalence of cataract with increased plasma vit C	Mohan et al. [38]
^a Reference range: 23–125 µmol/L	(Nutrition Evaluation Laboratory, Tufts University, 2001)	

persons with vitamin E intakes in the highest quintile category had an approximately 40 % lower prevalence of cataract relative to persons with intakes in the lowest quintile category. Jacques and Chylack [27] found that although persons with vitamin E intake greater than 35.7 mg/day had a 55 % lower prevalence of cataract than did persons with intakes less than 8.4 mg/day, a significant difference was not found. Two other studies also reported no difference in cataract prevalence between persons with high and low vitamin E intake [26, 29]. In the first of these two studies, the null

Data analysis method	Result	Reference	
Positive outcome			
Highest vs. lowest quintile	50 % lower cataract extraction in highest quintile	Tavani et al. [31]	
Highest vs.	Lower prevalence of nuclear cataract	Mares-Perlman et al. [26]	
Lowest quintile (men, 12.8 vs. 4.0 mg/day)	Cataract in highest quintile		
Highest vs. lowest quintile	40 % lower prevalence of cataract in highest quintile	Leske et al. [39]	
Null outcome			
>35.7 mg/day vs. <8.4 mg/day	No difference between groups	Jacques et al. [27]	
Highest vs. lowest quintile (women, 19.9 vs. 5 mg/day)	No difference between groups	Mares-Perlman et al. [26]	
	No relationship between dietary vit E and cataract extraction in women	Hankinson et al. [29]	

Table 4.3 Summary of epidemiologic studies of dietary vitamin E^a and cataract

aRDA: 8 mg/day and 10 mg/day, women and men, respectively

relationship was in women only (in men, dietary vitamin E was protective). In the second study only women were studied.

Results from studies reporting relationships between plasma vitamin E and cataract have been mixed. In five of eight studies examining this issue, increased plasma vitamin E was observed to be protective against the risk of cataract [30, 40–43] (Table 4.4). However, one study observed that the prevalence of cataract was not related to plasma vitamin E concentrations [28] and one study observed that the prevalence of cortical cataract did not differ between those with high (>30 μ mol/L) and low (<19 μ mol/L) plasma vitamin E concentrations [30]. Mares-Perlman et al. reported a significantly increased prevalence of nuclear cataract among women and men in the highest serum vitamin E quintile relative to those in the lowest quintile. Several other studies have found increased levels of plasma alpha-tocopherol to be a risk factor for nuclear cataract [44], cortical [28, 37], and PSC cataracts [28]. These results are contrary to what would be expected. Because these are cross-sectional observational studies, there is the possibility that unadjusted confounding affected the findings.

Lutein and Zeaxanthin

The majority of the epidemiological data suggest that dietary lutein and zeaxanthin function in cataract prevention. The Melbourne Visual Impairment Study, an observational study of >3,000 over the age of 40 years, reported an inverse correlation between high lutein and zeaxanthin intake and the risk of nuclear cataracts [45].

Lower intake of lutein-rich foods, e.g., green leafy vegetables, was reported to be related to an increased rate of cataract extraction [29]. Similarly, it was observed that women in the highest quintile of lutein intake had a 27 % lower prevalence of nuclear cataract than those in the lowest lutein intake quintile [26]. In this study, this was not observed in men. In the Nurses' Health Study, those with the highest quintile of intake of lutein and zeaxanthin had a 22 % decreased risk of cataract surgery and nuclear cataract compared with those in the lowest quintile after adjusting for other risk factors [46]. The intake of other dietary carotenoids was not associated with nuclear cataract. Similar findings were observed in the US male Health Professional's prospective study, in which men in the highest fifth of lutein and zeaxanthin intake had a 19 % lower risk of cataract surgery relative to men in the lowest fifth. There was no relationship observed with the other carotenoids [47]. The Beaver Dam Eye Study found that the incidence of nuclear cataract in subjects in the highest quintile of lutein intake was half

Data analysis method	Result	Reference
Positive outcome		
Highest vs. lowest quartile	Decrease in cortical cataract progression in highest quintile	Rouhiainen et al. [42]
>30 μmol/L vs. <19 μmol/L	Less nuclear cataract in high plasma vit E group	Vitale et al. [30]
>20 µmol/L vs. <20 µmol/L	Higher plasma levels of vit E had one-half the amount of cataract surgery	Knekt et al. [40]
Highest vs. lowest quintile	Lower prevalence of nuclear cataract in highest quintile	Leske et al. [41]
Regression model	High plasma vit E related to decreased prevalence of nuclear cataract	Leske et al. [43]
Multivariate logistic regression	Plasma vit E significantly lower in patients with cataract than in controls (9.16 vs. 13.26 mg/ mL, respectively)	Nourmohammadi et al. [113]
Null outcome		
>30 μmol/L vs. <19 μmol/L	No difference in cortical cataract progression	Vitale et al. [30]
Multivariate logistic regression	No relationship between plasma vit E and cataract	Italian-American Cataract Study Group [28]
Negative outcome		
Highest vs. lowest quintile men, 37.8 vs. 16.9 μmol/L women, 46.5 vs. 18.2 μmol/L	Increased prevalence of nuclear cataract in highest	Mares-Perlman [26]
Multivariate logistic regression	Higher plasma vit E associated with increase prevalence of cortical, PSC or any cataract	Ferrigno et al. [37]

Table 4.4 Summary of epidemiologic studies of plasma vitamin E^a and cataract

^aReference range: 12.0–43.2 mol/L (Nutrition Evaluation Laboratory, Tufts University, 2001)

of that found in those in the lowest quintile [48]. Data from an observational study of 372 volunteers found that the risk of PSC cataract was lowest in those with the highest plasma concentrations of lutein [49]. There was no relationship between plasma lutein and zeaxanthin and cortical and PSC cataract. In a prospective observational study higher intakes of lutein and zeaxanthin from both food and supplements were significantly related to a 20 % decreased risk of cataract [50]. Delcourt et al. [51] observed a 75 % lower risk of nuclear cataracts among individuals with high plasma zeaxanthin, but not lutein, in a prospective study of 899 adults aged ≥ 60 years. A summary of studies evaluating the relationship between lutein and cataract is found in Table 4.5.

Omega-3 Fatty Acids

Cataract formation is associated with perturbations of lens membrane composition, structure, and function [52–54], as well as changes in fatty acid composition [55]. Studies in rats find that high intake of polyunsaturated fatty acids delays the cataract formation [56, 57]. High intakes of dietary omega-3 polyunsaturated fatty acids were shown to reduce the incidence of nuclear cataract in the Blue Mountains Eye Study cohort [58]. In a prospective study examining the relationship between dietary fat and cataract extraction in women (n=71,083, 16-years follow-up), women in the highest quintile of long-chain omega-3 fatty acids (0.21 % of energy) had a 12 % lower risk of cataract extraction compared to those in the lowest quintile (0.03 % of energy) (relative risk=0.88, 95 % CI: 0.79–0.98, p for trend=0.02) [59]. Both Arnarsson et al. [60] and Cumming et al. [61] report that there was no association between the intake of foods or oils containing omega-3 fatty acids and age-related cataract prevalence.

Data analysis method	Result	Reference
Positive outcome		
Relationship with lutein/zeaxanthin intake	Inverse relationship with risk of nuclear cataract (nonsmoking women)	Vu et al. [45]
Consumption ≥5×/week vs. <1×/month spinach (women)	29 % decrease risk of cataract extraction	Hankinson et al. [29]
Highest vs. lowest quintile lutein/zeaxanthin intake (women)	27 % lower prevalence of nuclear cataract	Mares-Perlman et al. [26]
Highest vs. lowest quintile lutein/zeaxanthin intake (women)	22 % decrease risk of cataract extraction	Chasen-Taber et al. [46]
Highest vs. lowest quintile lutein/zeaxanthin intake (men)	19 % decrease risk of cataract extraction	Brown et al. [47]
Highest vs. lowest quintile lutein/zeaxanthin intake (women)	14 % decrease risk of incident cataract	Christen et al. [50]
Highest vs. lowest quintile lutein/zeaxanthin intake (1,245 vs. 298 µg/18 MJ)	Persons in the highest quintile were half as likely to have incident cataract as persons in the lowest quintile	Lyle et al. [48]
Logistic regression analysis of plasma carotenoids	Risk of posterior subcapsular cataract was lowest in those with higher concentrations of lutein	Gale et al. [49]
Relationship with serum carotenoids	75 % decreased risk of nuclear cataracts with increased serum zeaxanthin	Delcourt et al. [51]
Null outcome		
Risk estimated for tertiles 2 and 3 relative to tertile 1 using odds ratios	Serum lutein not associated with nuclear cataract	Lyle et al. [114]
Highest vs. lowest quintile lutein intake (men)	No difference in prevalence of nuclear cataract	Mares-Perlman et al. [26]

Table 4.5 Summary of epidemiologic studies of lutein/zeaxanthin status and cataract

AMD

Vitamin C

Seddon et al. [62] observed that persons in the highest and lowest intake quintiles for vitamin C had the same prevalence of advanced AMD. No relationship between dietary intake of vitamin C and cataract risk was observed in the Age-Related Eye Disease Study [63]. However, in that study results examining relationships between plasma levels of vitamin C and AMD suggests that increased plasma vitamin C may decrease the risk of AMD. West et al. [64] reported that individuals with plasma vitamin C concentrations >80 μ mol/L had a 45 % lower prevalence of AMD compared with individuals who had concentrations <60 μ mol/L. Others have reported that individuals with serum vitamin C concentrations \geq 91 μ mol/L had a 30 % lower prevalence of neovascular AMD compared with those who had concentrations <40 μ mol/L [65] (Table 4.6).

Vitamin E

A protective effect of increased diet and plasma vitamin E against AMD has been found in some [62, 64, 65], but not in others [44, 66] (Table 4.7).

Data analysis method	Result	Reference
Dietary vitamin C		
Highest vs. lowest quintile for vit C intake (1,039 vs. 65 mg/day)	No difference in prevalence of advanced AMD between groups	Seddon et al. [62]
Vit C intake energy adjusted, classified by quintiles. Relationship between diet and AMD assessed with logistic regression analyses.	Vit C intake not associated with AMD	AREDS [63]
Plasma vitamin C		
>80 µmol/L vs. <60 µmol/L	45 % lower prevalence of AMD in high plasma vit C group	West et al. [64]
>91 μmol/L vs. <40 μmol/L	Lower prevalence of neovascular AMD in high plasma vit C group	EDCCSG [65]

Table 4.6 Summary of epidemiologic studies of dietary and plasma vitamin C^a with AMD

^aRDA for adults: 60 mg/day. Reference range 23–125 µmol/L (Nutrition Evaluation Laboratory, Tufts University, 2001)

 Table 4.7
 Summary of epidemiologic studies of dietary and plasma vitamin E^a with AMD

Data analysis method	Result	Reference
Dietary vitamin E		
Highest vs. lowest quintile for vit E intake (405 vs. 3.4 mg/day)	No difference in prevalence of advanced AMD between groups	Seddon et al. [62]
Plasma vitamin E		
>30 µmol/L vs. <19 µmol/L	Lower prevalence of AMD with high plasma vit E	West et al. [64]
>43 µmol/L vs. <25 µmol/L	Lower prevalence of AMD with high plasma vit E	EDCCSG [65]
>23 µmol/L vs. <23 µmol/L	No difference between groups in prevalence of AMD	Mares-Perlman J et al. [44]
Patients with AMD vs. age-, sex-matched controls	No difference in plasma concentration of vit E between groups	Sanders et al. [66]

^aRDA: 8 mg/day and 10 mg/day, women and men, respectively. Reference range: 12.0–43.2 µmol/L (Nutrition Evaluation Laboratory, Tufts University)

Lutein and Zeaxanthin

Findings from several case control studies suggest that high intakes of carotenoids, particularly lutein and zeaxanthin, are related to lower risk of advanced neovascular AMD [62, 65, 67]. A relationship between carotenoid status and age-related eye disease risk was evaluated in the Carotenoids in Age-Related Eye Disease Study (CAREDS), a cross sectional study of 1,678 women (54–86 years) [68]. In CAREDS, high dietary lutein and zeaxanthin were related to a decreased risk of intermediate AMD in women <75 years, but not in women \geq 75 years [68]. Similarly, the Blue Mountain Eye Study found that high intake of lutein/zeaxanthin was related to a reduced risk of incident neovascular AMD and indistinct soft or reticular drusen over 5 and 10 years [69]. In the Age-Related Eye Diseases (AREDS) study, lutein/zeaxanthin intake was associated with a decreased risk of prevalent neovascular AMD and large or extensive intermediate drusen when comparing the highest and lowest quintiles of intakes [63]. However, a nested case control study from the Beaver Dam Eye Study found no difference in serum levels of lutein/zeaxanthin between early AMD and age-, sex-, and smoking-matched controls [44].

Evidence that high macular pigment concentrations may lower the risk of AMD comes from analysis of retinas from donors with AMD and controls for lutein and zeaxanthin [70]. Concentrations of lutein and zeaxanthin in retinal tissue were less, on average, for the AMD donors than for the controls.

Study design	Result	Reference
Positive outcome		
Case-control	Dietary lutein/zeaxanthin was significantly associated with a decreased risk for AMD	Seddon et al. [62]
Case-control	Risk of neovascular AMD decreased by half in the >20th to 80th percentile of lutein/zeaxanthin intake group compared to the low percentile group (≤20th percentile)	EDCCSG [65]
Case-control	Prevalence of AMD in subjects with low lutein intake was twice as high as that in subject with high intake	Snellen et al. [67]
Cross-sectional	Prevalence of intermediate AMD was not different between high and low lutein/zeaxanthin intake groups. Lower odds ratio in women <75 years at risk for diet changes compared to women >75 years	Moeller et al. [68]
Prospective	Subjects in the top tertile for dietary lutein/zeaxanthin were significantly less likely to develop neovascular AMD and indistinct soft drusen	Tan et al. [69]
Case-control	High lutein/zeaxanthin intake associated with lower odds of neovascular AMD, geographic atrophy and large or extensive intermediate drusen	AREDS [63]
Case-control	Higher quartile of retinal lutein/zeaxanthin associated with 82 % decreased risk of AMD compared to lowest quartile	Bone et al. [70]
Healthy eyes at risk for AMD healthy eye not at risk	Healthy eyes at risk had significantly less macular pigment	Beatty et al. [72]
Case-control	32 % lower macular pigment in AMD patients than in controls	Bernstein et al. [73]
Prospective	Higher macular pigment significantly related to decreased risk of late AMD	Schweitzer et al. [74]
Null outcome		
Case-control	No association between serum lutein/zeaxanthin and AMD risk	Mares-Perlman et al. [44]
Cross-sectional	Risk for AMD among women in the highest and lowest quintile for macular pigment not different	LaRowe et al. [75]

Table 4.8 Summary of studies of lutein/zeaxanthin status and AMD

Macular pigment can be measured noninvasively in vivo [71]. In subjects (21–81 year) with healthy eyes and in healthy eyes of subjects at high-risk of AMD due to advanced disease in the fellow eye, at-risk eyes had significantly less macular pigment than eyes not at risk [72]. In another study, macular pigment in the eyes of AMD patients were 32 % lower in than that in healthy eyes [73]. In a prospective study in patients with early and late AMD, their children, and normal controls, it was found that macular pigment density was reduced in late AMD [74]. In contrast, in CAREDS, macular pigment density was not related to AMD [75]. Inconsistencies among studies may be due to the cross-sectional study design, suggesting that prospective studies are needed to determine if there is a relationship between AMD risk and macular pigment density. A summary of studies evaluating the relationship between lutein/zeaxanthin status and AMD risk is found in Table 4.8.

Omega-3 Fatty Acids

A meta-analysis, including nine prospective studies and three randomized clinical trials, reported that a high dietary intake of omega-3 fatty acids was associated with a 38 % reduction in the risk of late AMD [76]. Fish intake at least twice a week was associated with a 24 % reduced risk for early AMD and 33 % reduced risk for late AMD (pooled OR, 0.67; 95 % CI, 0.53–0.85). In a prospective

follow-up study of the Nurses' Health Study and the Health Professionals Follow-up Study, odds of AMD decreased with increased DHA intake. Consumption of >4 servings of fish per week was associated with a 35 % lower risk of AMD compared with \leq 3 servings per month in pooled multivariate analysis [77]. Of the individual fish types examined, a significant inverse association was found only with tuna intake.

The Dietary Ancillary Study of the Eye Disease Case–Control Study [65] reported results for participants with neovascular AMD and control subjects without AMD [78]. In demographically adjusted analyses, increasing intake of linoleic acid (an unsaturated n-6 fatty acid) was significantly associated with higher prevalence of AMD (p for trend=0.004). This association remained in multivariate analyses (p for trend=0.02). However, intake of omega-3 fatty acids was not associated with AMD after controlling for confounding variables. When the study population was stratified by linoleic acid intake (≤ 5.5 or ≥ 5.6 g/day), the risk for AMD significantly decreased with high intake of omega-3 fatty acids only among those with low linoleic acid intake (p for trend=0.05; p for continuous variable=0.03). A prospective cohort study reported similar findings. In this study, 261 persons aged 60 years or older at baseline with an average follow-up of 4.6 years were evaluated for progression to late AMD. Higher fish intake was associated with a 64 % lower risk of progression to advanced AMD among subjects with lower linoleic acid intake [79]. Others have also observed a benefit of a high omega-3 fatty acid intake with low intake of linoleic acid [80]. These findings indicate an interaction or competition between omega-3 and omega-6 fatty acids such that both the levels of omega-3 fatty acids and its ratio to the n-6 acids may be important when considering the risk/benefit to AMD.

The Blue Mountain Eye Study was a population-based survey of vision, in which patients (\geq 49 years) were evaluated for AMD at baseline, 5 and 10 years after initial enrollment and energy-adjusted nutrient intakes were assessed for relationships with AMD risk [81]. In the 2,915 subjects, more frequent consumption of fish was related to a decreased risk of late AMD after adjusting for age, sex, and smoking. The protective effect of fish intake was observed at a relatively low consumption (1–3 times per month compared with intake <1 time per month (OR=0.23, 95 % CI: 0.08–0.63 after adjusting for age, sex, and smoking). Although the OR was approximately 0.5 at higher intakes, this was not significant. There was little evidence of a decreased risk for early AMD with increased fish intake. A prospective cohort study from the AREDS trial found that increased intakes of omega-3 fatty acids reduced the risk by 30 % of developing central geographic atrophy and neovascular AMD [82].

In the Beaver Dam Eye Study, a retrospective population-based study, a relationship between fish intake, neovascular AMD, or geographic atrophy was not found. A null finding may be due to the observation that fish intakes were very low [83] and not varied enough to measure a difference in risk for AMD. In the NHANES III study, consuming fish more than once a week compared with once a month or less was not associated with either early or late AMD [84]. A summary of the studies that have evaluated the relationship between omega-3 fatty acids and AMD can be found in Table 4.9.

AREDS 2 clinical trial evaluated the benefit of 1 g eicosapentaenoic acid+DHA (2:1) on the progression to late AMD [85]. Addition of DHA+EPA to the AREDS formulation in primary analyses did not further reduce risk of progression to advanced AMD.

Zinc

In a study by Newsome et al. [85], zinc supplementation given to elderly people with early stages of AMD resulted in better maintenance of visual acuity than in those receiving placebo; scientific evidence that zinc intake is associated with the development or progression of AMD is limited. A recent study found zinc levels in the RPE and choroid to be reduced significantly by 24 % in AMD eyes compared to normal eyes, lending support to the possible importance of zinc in AMD [87].

Study	Sample design	Exposure (high vs. low)	Outcome	Cases	OR	95 % CI	Reference
NHS/HPFU	Prospective	LCPUFA	NV AMD	9	0.4	0.2-1.2	[77]
EDCCS	Case-control	LCPUFA	NV AMD	349	0.6	0.3-1.4	[78]
BDES	Population-based	Fish	Late ARM	30	0.8	0.2-1.5	[83]
NHANES	National Survey	Fish	Late ARM	9	0.4	0.2-1.2	[84]
BMES	Population-based	Fish	Late ARM	46	0.5	0.2-1.2	[81]
Seddon et al.	Prospective	Fish	Late AMD	261	0.4	0.1-0.95	[79]

Table 4.9 Multivariable odds ratios for neovascular AMD and late ARM and omega-3 fatty acid intake

Adapted from [13]

Abbreviations: BDES Beaver Dam Eye Study, BMES Blue Mountains Eye Study, NHS/HPFU Nurses' Health Study/ Health Professionals Follow-Up, NHANES National Health and Nutrition Survey, EDCCS Eye Disease Case-control Study, AREDS Age-related Eye Disease Study, Late ARM biographic atrophy or neovascular age-related macular degeneration, NV AMD neovascular age-related macular degeneration

In the Blue Mountain Eye Study, the relative risk comparing the top decile of zinc intake (\geq 15.8 mg/ day) with the remaining population was 0.56 (95 % CI, 0.32–0.97) for all forms of AMD and 0.54 (95 % CI, 0.30–0.97) for early AMD. However, in two large prospective studies (Nurses' Health Study and the Health Professionals Follow-up Study) involving men and women (n=72,489) with no diagnosis of AMD at baseline, zinc intake, either in food or in supplements, was not associated with a reduced risk of AMD [88]. Furthermore, no association was observed between zinc intake and prevalence of AMD-related drusen in the cohort of the Nurses' Health Study who also participated in the Nutrition and Vision project [89]. A protective effect of zinc is in contrast to the findings of high concentrations in sub RPE deposits, with levels being particularly high in eyes with AMD [14].

In summary, the studies examining dietary intakes of certain nutrients and eye disease relationships are not entirely consistent. Methodology differences among studies may, in part, explain the inconsistencies. Also, there are limitations to such studies that examine relationships between a nutrient and disease because calculations from dietary recall may not always accurately estimate nutrient intakes due to limitations of the database or recall abilities of the subjects. Furthermore, a single blood value for a nutrient may not always be an accurate indicator of long-term status. In addition, the high degree of correlation in intake among the various dietary micronutrients makes it difficult to determine which specific nutrient or nutrients are related to the observed relationships. Despite these drawbacks, a possible protective role of certain nutrients cannot be dismissed given the number of studies that found a protective effect and the comparatively fewer studies that found a negative effect. In some cases, it may be difficult to measure an outcome if nutrient intake levels are at those found in diet alone. That is, dietary and plasma levels may not be sufficiently high to see an effect. In this regard, review of studies that have examined the relationship between supplemental nutrient intake with cataract and AMD risk may be useful.

The Effect of Nutrient Supplements on Eye Disease Risk

Supplemental vitamins C and E have been long available to the general public. Currently, there are a variety of supplement products available in health food stores that contain lutein in amounts of 6-25 mg/capsule. At this point lutein can be found in a few multi-vitamin products. Centrum was the first multivitamin supplement with lutein and contains 250 µg per capsule. Alcon Laboratories and Bausch & Lomb have made available multivitamin supplements formulated for eye care. These products contain lutein in higher amounts.

Data analysis method	Result	Reference	
Positive outcome			
Usage ≥10 year vs. usage <10 year	Decrease in cataract in long-term users of vit C supplements	Jacques et al. [90]	
Usage ≥10 year vs. usage <10 year	Decrease in cataract surgery with increased usage of vit C supplement	Hankinson et al. [29]	
>300 mg/day vs. nonusers of vitamin C supplements	Lower prevalence of cataract in vit C supplement users	Robertson et al. [91]	
Null outcome			
Usage ≥10 year vs. nonusers	No difference between groups	Chasen-Taber et al. [92]	
Logistic regression analysis	No relationship between serum vit C and cortical and nuclear cataract	Vitale et al. [30]	
Regression models	No relationship between vit C intake and cataract extraction	Tavani et al. [31]	
Highest vs. lowest quintile for intake (705 vs. 70 mg/day)	No relationship between vit C intake and cataract extraction	Hankinson et al. [29]	
Case-control	No relationship between vit C intake and cataract	Italian-American Cataract Study [28]	
Negative outcome			
Usage 5-9 years vs. 1-4 years	Increase risk of cortical cataract	Gritz et al. [93]	
1,000 mg/day vit C vs. multivitamin use	Increase risk of cataract	Rautianen et al. [94]	

Table 4.10 Summary of epidemiologic studies of supplemental vitamin C^a and cataract

^aRDA for adults: 60 mg/day

Cataract

Jacques et al. [90] observed a >75 % lower prevalence of early opacities in women who used vitamin C supplement for ≥ 10 years (Table 4.10). None of the 26 women who used vitamin C supplements for \geq 10 years had more advanced nuclear cataract. Hankinson et al. [29] observed that women who reported use of vitamin C supplement for ≥ 10 years had a 45 % reduction in rate of cataract surgery. The study of Robertson et al. [91] observed that the prevalence of cataract in persons who consumed vitamin C supplement of >300 mg/day was approximately one-third the prevalence in persons who did not consume vitamin C supplements. However, Chasen-Tabar et al. [92] prospectively examined the association between vitamin supplement intake and the incidence of cataract extraction during 12 years of follow-up in a cohort of 73,956 female nurses. After adjusting for cataract risk factors, including cigarette smoking, body mass index, and diabetes mellitus, there was no difference in the incidence of cataract between users of vitamin C supplements for 10 years or more and nonusers. There was an increased risk of cortical cataract in older women (≥ 60 years) who consumed vitamin C supplements for 5–9 years compared to the those taking vitamin C for 1–4 years, ≥ 10 years and those who never used vitamin C supplements [93]. Similarly, in a population-based, prospective study involving 24,593 women (≥ 65 years), a significantly higher risk of cataract was observed in those consuming ~1,000 mg vitamin C supplements but not among those consuming a multivitamin, which contain much lower levels of vitamin C [94]. However, such relationship was not always observed [28-31].

Nadalin et al. [95] cross-sectionally examined the association between prior supplementation with vitamin E and early cataract changes in volunteers (Table 4.11). Of 1,111 participants 26 % reported prior supplementation with vitamin E. Only 8.8 % of these participants took supplementation greater than the recommended daily intake (10 mg/day). A statistically significant association was found between prior supplementation and the absence of cortical opacity, after adjusting for age. However, the levels of nuclear opacity were not statistically different between those who reported intake and those with no prior vitamin E supplementation. Leske et al. [43] examined the association of

Data analysis method	Result	Reference	
Positive outcome			
Vit E supplement users vs. nonusers	Decrease in cortical cataract in users of vit E supplements	Nadalin et al. [95]	
Vit E supplements users vs. nonusers	Decrease in nuclear cataract in users	Leske et al. [41]	
Vit E supplements users vs. nonusers	Decrease in cataract in users	Robertson et al. [91]	
Null outcome			
Vit E supplement users vs. nonusers	No difference in nuclear cataract	Nadalin et al. [95]	
Vit E supplement users vs. nonusers	No difference in prevalence of cataract between users and nonusers	Hankinson et al. [29]	

Table 4.11 Summary of epidemiologic studies of supplemental vitamin E^a and cataract

^aRDA: 810 mg/day and 10 mg/day, women and men, respectively

antioxidant nutrients and risk of nuclear opacification in a longitudinal study. The risk of nuclear opacification at follow-up was decreased in regular users of multivitamin supplements, vitamin E supplements, and in persons with higher plasma levels of vitamin E. The investigators concluded that in regular users of multivitamin supplements, the risk of nuclear opacification was reduced by one-third. They also reported that in regular users of vitamin E supplement and persons with higher plasma levels of vitamin E, the risk was reduced by approximately half. These results are confirmed by Robertson et al. [91] who reported that the prevalence of cataract was 56 % lower in persons who consumed vitamin E supplement than in persons not consuming supplements. One study observed no relation between risk of cataract and vitamin E supplements [29, 95].

To date, there are few data from intervention trials of vitamins and cataract risk. There has been only one randomized, double-masked, placebo controlled trial evaluating a vitamin E intervention and cataract prevention. Pharmacological doses of vitamin E (500 IU) for 4 years was reported to reduce the incidence or progression of nuclear, cortical, or PSC cataracts in subjects aged 55–80 years [96]. This may provide the best evidence for a role for vitamin E in cataract prevention.

It was reported that a high dose combination of antioxidants (vitamins C and E, beta-carotene, and zinc) had no significant effect on the development or progression of cataract [97]. The LINXIAN trial [98] examined the role of antioxidants in prevention of cataract, and effect is not clear. The intervention was a combination dose of 14 vitamins and 12 minerals. Therefore, a specific role of any one nutrient could not be accurately evaluated. The multivitamin component demonstrated that nutrition can modify the risk of nuclear cataract, but specific nutrients were not evaluated. Also, the population examined had suboptimal nutritional intakes at the study start and the effect may have been due to a correction of certain nutrient deficiencies.

The Roche European-American Anticataract Trial (REACT) was carried out to examine if a mixture of oral antioxidant micronutrients (beta-carotene, 18 mg/day; vitamin C, 750 mg/day; vitamin E, 600 mg/day) would modify the progression of age-related cataract [99]. This was a multi-center prospective double-masked randomized placebo-controlled 3-year trial in 445 patients with early agerelated cataract. REACT demonstrated a statistically significant positive treatment effect after 2 years for US patients and for both subgroups (US, UK) after 3 years, but no effect for the UK patients alone. The conclusion from this study was that daily supplementation with these nutrients for 3 years produced a small deceleration in progression of age-related cataract.

AMD

A recent study reported that high levels of antioxidants and zinc significantly reduce the risk of AMD and its associated vision loss [97]. In the AREDS study it was found that people at high risk for developing advanced stages of AMD (people with intermediate AMD or advanced AMD in one eye but not the

Data analysis method	Result	Reference
Positive outcome		
Vitamins C, E, β-carotene zinc, copper (AREDS formulation)	25 % decrease in risk for late AMD, in patients with intermediate or advanced AMD in one eye	Age-Related Eye Disease Study [97]
Zinc vs. placebo	Better visual acuity with zinc supplementation in patients with early AMD	Newsome et al. [86]
Zinc (50 mg/day, 6 months) vs. placebo	Improved macular function in dry AMD	Newsome et al. [106]
Vitamins C, E, β-carotene zinc, copper (AREDs formulation plus lutein (10 mg/day)) and zeaxanthin (2 mg/day)	Those with low dietary intake of lutein/zeaxanthin at the start of the study, but took AREDS formulation with lutein+zeaxanthin during, were ~25 % less likely to develop advanced AMD compared with those with similar dietary intake who did not take lutein and zeaxanthin	Age-Related Eye Disease Study [85]
Null outcome		
Vitamin C supplement users (>2 years) vs. nonusers	No difference between groups in AMD prevalence	EDCCSG [65]
Vitamin E supplement users (>2 years) vs. nonusers	No difference between groups in AMD prevalence	Seddon et al. [101]
Vitamin E (600 IU/day, 10 year) vs. placebo (women)	No effect on risk of AMD	Christen et al. [102]
Vitamin E (500 IU, 4 years) vs. placebo	No effect on risk of AMD	Taylor et al. [103]
Vitamin E (50 mg/day 5–8 years) vs. placebo	No effect on risk of AMD	Teikari et al. [104]

Table 4.12 Summary of epidemiologic studies of supplement use and AMD

other eye) lowered their risk by about 25 % when treated with a high-dose combination of vitamins C and E, beta-carotene, and zinc. In the same high risk group the nutrients reduced the risk of vision loss caused by advance AMD by about 19 % [97] (Table 4.12). For those subjects who had either no AMD or early AMD, the nutrients did not provide a measured benefit. Because single nutrients were not evaluated, specific effects could not be determined. AREDS 2 aimed to refine the findings of AREDS by including the xanthophylls lutein and zeathanxin as well as omega-3 fatty acids into the test formulation [100]. In secondary analysis, lutein and zeaxanthin supplements on top of the AREDS supplement lowered the progression to advanced AMD in persons with low dietary lutein and zeaxanthin [85].

It has been reported that the prevalence of AMD in persons who consumed vitamin C supplement for >2 years was similar to those who never took vitamin C supplements [65] (see Table 4.12). In a study conducted by Seddon et al. [101] the prevalence of AMD was also similar between those who took vitamin E supplement for >2 years and those who never took vitamin E supplements. The results of three double-masked, placebo-controlled primary prevention studies involving high doses of vitamin E and AMD found that there was no association of treatment group with any sign of AMD [102–104]. The Alpha-Tocopherol, Beta-Carotene trial evaluated the effect of nutritional antioxidants on AMD. Overall there were 728 people randomized to any antioxidant (alpha-tocopherol and/or β -carotene) and 213 to placebo. The results of this study found that there was no association of treatment group with any sign of maculopathy. There were 216 cases of the disease in the antioxidant groups and 53 in the placebo group. The majority of these cases were early age-related maculopathy. There was no association with the treatment group and development of early stages of the disease. The findings are similar when each of the antioxidant groups-alpha-tocopherol, beta-carotene, alphatocopherol + beta-carotene—are compared with placebo. Although this was a large, high quality study there were few cases of late AMD (14 cases in total) which means that the study had limited power to address the question as to whether supplementation prevents late AMD. Furthermore, the 5-8-year intervention with alpha-tocopherol and/or beta-carotene may not have been a long enough intervention period to assess a difference in the prevalence of late AMD. This study was conducted in Finnish male smokers and caution must be taken when extrapolating the findings to other geographical areas, to people in other age-groups, to women, and to nonsmokers. However, the incidence of AMD, particularly neovascular disease, is likely to be higher in smokers [105], which means that they provide a good population to demonstrate any potential protective effects of antioxidant supplementation.

As mentioned above, The AREDS 2 trial found that lutein and zeaxanthin supplements lowered the progression to advanced AMD in persons with low dietary lutein and zeaxanthin.

Zinc supplementation (100 mg zinc sulfate for 12–24 months) given to elderly people with early stages of AMD resulted in better maintenance of visual acuity than in those receiving placebo [86]. A more recent randomized, placebo-controlled study showed supplementation with 50 mg/day for 6 months of zinc monocysteine significantly improved macular function in persons with dry AMD [106]. However, patients with wet AMD in one eye had no positive effect on the other eye diagnosed with drusen after 200 mg/day zinc supplementation for 2 years [107].

In summary, of the studies that have examined nutrient supplement use vs. the risk of eye disease, it is difficult to determine if supplements provide any added protection against eye disease. The number of studies reporting a positive outcome, i.e., a decreased risk, was about the same as the number of null outcomes. Further, in a recent meta-analysis it was concluded that there is insufficient evidence to support the role of dietary antioxidants including the use of dietary antioxidant supplements for the primary prevention of AMD [108]. However, AREDS 1 and 2 strongly support the beneficial effects of supplementation with a combination of antioxidants on delaying the progression from intermediate to late AMD.

Clinical Recommendations/Treatment Guidelines

The inconsistencies among studies in terms of the amount of nutrient required for protection against eye disease makes it difficult to make specific recommendations for dietary intakes of these antioxidants. Therefore, it may be more practical to recommend specific food choices rich in vitamins C and E, lutein and zeaxanthin, omega-3 fatty acids and zinc, thereby benefiting from possible effects of the components in food that may also be important. This necessitates an awareness of dietary sources of nutritional antioxidants for both the patient and clinician. Good sources of vitamin C include citrus fruit, berries, tomatoes, and broccoli (Table 4.13). Good sources of vitamin E are vegetable oils, wheat germ, whole grain cereals, nuts, and legumes (Table 4.14). The two foods that were found to have the highest amount of lutein and zeaxanthin are kale and spinach (Table 4.15). Other major sources include broccoli, peas, and brussels sprouts. Fish oils are the primary source of omega-3 fatty acids (Table 4.16)

A healthy diet including a variety of fresh fruit and vegetables, legumes, fish, and nuts, will have many benefits, will not do any harm, and will be a good source of the antioxidant vitamins and minerals implicated (but not proven) in the etiology of cataract and age-related macular degeneration. There is no evidence that nutrient-dense diets high in these foods, which provide known and unknown antioxidant components, are harmful. In fact, intake of fruits and vegetables is associated with reduced risk of death due to cancer, cardiovascular disease, and all causes [109]. Thus, recommendations such as consuming a more nutrient-dense diet, i.e., lower in sweets and fats, and increasing levels of fruit and vegetable intake do not appear to be harmful and may have other benefits despite their unproven efficacy in preventing or slowing disease. Until the efficacy and safety of taking supplements containing nutrients can be determined, current dietary recommendations [110] are advised.

In addition to antioxidant vitamins, patients ask about a wide variety of unproven and often untested nutritional supplements. These include bilberries, shark cartilage, and Ginko biloba extract. Unfortunately, little is known about the effect of these products on cataract or AMD: no clinical trials

Food	Amount	Milligrams
Orange juice	1 cup	12
Green peppers	1/2 cup	96
Grapefruit juice	1 cup	94
Papaya	1/2 med	94
Brussels sprouts	4 sprouts	73
Broccoli, raw	1/2 cup	70
Orange	1 medium	70
Cantaloupe	1/4 melon	70
Turnip greens, cooked	1/2 cup	50
Cauliflower	1/2 cup	45
Strawberries	1/2 cup	42
Grapefruit	1/2 medium	41
Tomato juice	1 cup	39
Potato, boiled with peel	2 1/2" diam.	19
Cabbage, raw, chopped	1/2 cup	15
Blackberries	1/2 cup	15
Spinach, raw, chopped	1/2 cup	14
Blueberries	1/2 cup	9

Table 4.13 Vitamin C content of foods^a

Table created with data from [115]

^aEdible portion

Table 4.14Vitamin E content of foods^a

Food	Amount	Milligrams (alpha-tocopherol equivalents)
Wheat germ oil	1 tb	26.2
Sunflower seeds	1/4 cup	16.0
Almonds	1/4 cup	14.0
Safflower oil	1 tb	4.7
Peanuts	1/4 cup	4.2
Corn oil	1 tb	2.9
Peanut butter	2 tb	4.0
Soybean oil	1 tb	2.0
Pecan, halves	1/4 cup	2.0
Table created with data from [115]		

^aEdible portion

Table 4.15 Lutein/zeaxanthin content of foods^a

Food	Amount	Milligrams
Kale, cooked	1/2 cup	8.7
Spinach, raw	1/2 cup	6.6
Spinach, cooked	1/2 cup	6.3
Broccoli, cooked	1/2 cup	2.0
Corn, sweet, cooked	1/2 cup	1.5
Peas, green, cooked	1/2 cup	1.1
Brussels sprouts, cooked	1/2 cup	0.9
Lettuce, raw	1/2 cup	0.7

Table created with data from [116]

^aEdible portion

	EPA+DHA, g/3-oz (85 g)		
Fish	serving (edible portion)		
Trout			
Farmed	0.15		
Wild	0.20		
Crab, Alaskan King	0.35		
Flounder/sole	0.42		
Haddock	0.20		
Halibut	0.40-1.00		
Herring			
Atlantic	1.71		
Pacific	1.81		
Mackerel	0.34-1.57		
Salmon			
Atlantic, farmed	1.09-1.83		
Atlantic, wild	0.90-1.56		
Chinook	1.48		
Sockeye	0.68		
Sardines	0.98-1.70		
Shrimp, mixed species	0.27		
Tuna			
Fresh	0.24-1.28		
White, canned in water, drained	0.73		
Created with data from [117]			

 Table 4.16
 EPA + DHA content in fish

Created with data from [117]

have been conducted. Patients with eye disease who are offered these often expensive and sometimes risky treatments are given little information as to their benefit or risk. Patients should be advised to avoid unproven treatments.

Conclusion

The hypothesis that antioxidant nutrients may protect against the cataract and AMD is a plausable one given the role of oxidative damage in the etiology of these diseases. It is not known at what stage the protective effect may be important. The question that needs to be addressed is whether people who begin to consume antioxidant vitamins in their 60s and 70s alter their risk of age-related macular degeneration. Although data regarding the use of nutrient supplements suggest protection in cataract, the data are less convincing for AMD. The research to date has not sufficiently evaluated the effectiveness vs. safety of nutrient supplements. But advocating the use of nutrient supplementation must be done with a cautionary note given that there have been trials which have suggested that supplementation with beta-carotene may have an adverse effect on the incidence of lung cancer in smokers and workers exposed to asbestos [111, 112]. Clearly further trials are warranted to address the usefulness of nutrient supplementation in eye disease prevention.

It is likely that cataract and AMD develops over many years and the etiology of these diseases is due to many factors. There are likely to be differences in the potential protective effect of antioxidant supplementation depending on the stage of the disease. Future research needs to take into account the stage at which oxidative damage, and therefore antioxidant supplementation, may be important.

References

- 1. Congdon N, O'Colmain B, Klaver CCW, Klein R, Munoz B, Friedman DS, et al. Causes and prevalence of visual impairment among adults in the United States. Arch Ophthalmol. 2004;122(4):477–85.
- 2. World Health Organization. Use of intraocular lenses in cataract surgery in developing countries. Bull World Health Organ. 1991;69:657–66.
- 3. Thylefors B, Negrel AD, Pararajasegaram R, Dadzie KY. Global data on blindness. Bull World Health Organ. 1995;69:115–21.
- 4. Prevalence of Blindness Data Tables (NEI Statistics and Data). National Eye Institute, National Institute of Health. 2004 [cited 7 June 2012]. Available from: http://www.nei.nih.gov/eyedata/
- 5. Javitt JC. Who does cataract surgery in the United States? Arch Ophthalmol. 1993;111:1329.
- Steinberg EP, Javitt JC, Sharkey PD, Zuckerman A, Legro MW, Anderson GF, et al. The content and cost of cataract surgery. Arch Ophthalmol. 1993;111:1041–9.
- 7. Snodderly DM. Evidence for protection against age-related macular degeneration by carotenoids and antioxidant vitamins. Am J Clin Nutr. 1995;62(6 Suppl):1448S–61.
- 8. Taylor HR. Epidemiology of age-related cataract. Eye. 1999;13:445-8.
- Taylor A, Jacques PF, Nadler S, Morrow F, Sulsky SI, Shepard D. Relationship in humans between ascorbic acid consumption and levels of total and reduced ascorbic acid in lens, aqueous humor and plasma. Curr Eye Res. 1991;10:751–9.
- Yeum KJ, Taylor A, Tang G, Russell RM. Measurement of carotenoids, retinoids, and tocopherols in human lenses. Invest Ophthalmol Vis Sci. 1995;36:2756–61.
- 11. Yeum KJ, Shang F, Schalch W, Russell RM, Taylor A. Fat-soluble nutrient concentrations in different layers of human cataractous lens. Curr Eye Res. 1999;19:502–5.
- 12. Newsome DA, Oliver PD, Deupree DM, Miceli MV, Diamond JO. Zinc uptake by primate retinal pigment epithelium and choroid. Curr Eye Res. 1992;11:213–7.
- 13. SanGiovanni JP, Chew EY. The role of omega-3 long-chain polyunsaturated fatty acids in health and disease of the retina. Prog Retin Eye Res. 2005;24:87–138.
- 14. Lengyel I, Flinn JM, Peto T, Linkous DH, Cano K, Bird AC, et al. High concentration of zinc in sub-retinal pigment epithelial deposits. Exp Eye Res. 2007;84(4):772–80.
- 15. Bone RA, Landrum JT. Macular pigment in Henle fiber membranes: a model for Haidinger's brushes. Vision Res. 1984;24:103–8.
- 16. Parker RS. Bioavailability of carotenoids. Eur J Clin Nutr. 1997;51:S86-90.
- 17. Bone RA, Landrum JT, Tarsis SL. Preliminary identification of the human macular pigment. Vision Res. 1985;25(11):1531–5.
- Johnson EJ. Obesity, lutein metabolism, and age-related macular degeneration: a web of connections. Nutr Rev. 2005;63(1):9–15.
- Hodge WG, Whitcher JP, Satariano W. Risk factors for age-related cataracts. Epidemiol Rev. 1995;17(2):336–46.
- Sarks SH, Sarks JP. Age-related macular degeneration: atrophic form. In: Ryan SJ, Schachat SP, Murphy RM, editors. Retina. St. Louis: Mosby, Inc.; 1994. p. 149–73.
- Fliesler SJ, Anderson RE. Chemistry and metabolism of lipids in the vertebrate retina. Prog Lipid Res. 1983;22(2):79–131.
- Bazan NG, Reddy TS, Bazan HEP, Birkle DL. Metabolism of arachidonic and docosahexaenoic acids in the retina. Prog Lipid Res. 1986;25:595–606.
- 23. Benedek GB. Theory of transparency of the eye. Appl Opt. 1971;10:459-73.
- 24. Asbell PA, Dualan I, Mindel J, Brocks D, Ahmad M, Epstein S. Age-related cataract. Lancet. 2005;365(9459):599–609.
- 25. Facts about Age-Related Macular Degeneration. 2009 [cited 22 November 2011]. Available from: http://www.nei. nih.gov/health/maculardegen/armd_facts.asp#1
- Mares-Perlman JA, Brady WE, Klein BE, Klein R, Haus GJ, Palta M, et al. Diet and nuclear lens opacities. Am J Epidemiol. 1995;141(4):322–34.
- 27. Jacques PF, Chylack Jr LT. Epidemiologic evidence of a role for the antioxidant vitamins and carotenoids in cataract prevention. Am J Clin Nutr. 1991;53:3538–5.
- The Italian-American Cataract Study Group. Risk factors for age-related cortical, nuclear, and posterior subcapsular cataracts. Am J Epidemiol. 1991;133:541–53.
- Hankinson SE, Stampfer MJ, Seddon JM, Colditz GA, Rosner B, Speizer FE, et al. Nutrient intake and cataract extraction in women: a prospective study. BMJ. 1992;305:244–51.
- Vitale S, West S, Hallfrisch J, Alston CM, Wang F, Moorman C, et al. Plasma antioxidants and risk of cortical and nuclear cataract. Epidemiology. 1993;4:195–203.

- 31. Tavani A, Negri E, LaVeccia C. Food and nutrient intake and risk of cataract. Ann Epidemiol. 1996;6:41-6.
- 32. Tan AG, Mitchell P, Flood VM, Burlutsky G, Rochtchina E, Cumming RG, et al. Antioxidant nutrient intake and the long-term incidence of age-related cataract: the Blue Mountains eye study. Am J Clin Nutr. 2008;87(6): 1899–905.
- Simon JA, Hudes ES. Serum ascorbic acid and other correlates of self-reported cataract among older Americans. J Clin Epidemiol. 1999;52:1207–11.
- Valero MP, Fletcher AE, DeStavola BL, Vioque J, Alepuz VC. Vitamin C is associated with reduced risk of cataract in a Mediterranean population. J Nutr. 2002;132:1299–306.
- Dherani M, Murthy GVS, Gupta SK, Young IS, Maraini G, Camparini M, et al. Blood levels of vitamin C, carotenoids and retinol are inversely associated with cataract in a North Indian population. Invest Ophthalmol Vis Sci. 2008;49(8):3328–35.
- Jalal D, Koorosh F, Fereidoun H. Comparative study of plasma ascorbic acid levels in senile cataract patients and in normal individuals. Curr Eye Res. 2009;34(2):118–22.
- 37. Ferrigno L, Aldigeri R, Rosmini F, Sperduto RD, Maraini G, Italian-American Cataract Study Group. Associations between plasma levels of vitamins and cataract in the Italian-american clinical trial of nutritional supplements and age-related cataract (CTNS): CTNS Report #2. Ophthalmic Epidemiol. 2005;12(2):71–80.
- Mohan M, Sperduto RD, Angra SK, Milton RC, Mather RL, Underwood BA, et al. Indian-US case-control study of age-related cataracts. India-US case-control Study Group. Arch Ophthalmol. 1989;107:670–6.
- Leske MC, Chylack Jr LT, Wu SY. The lens opacities case-control study. Risk factors for cataract. Arch Ophthalmol. 1991;109(2):244–51.
- 40. Knekt P, Heliovaara M, Rissenen A, Aronaa A, Aaran R. Serum antioxidant vitamins and risk of cataract. BMJ. 1992;304:1392–4.
- 41. Leske MC, Wu SY, Hyman L, Sperduto R, Underwood BA, Chylack LT, et al. Biochemical factors in the lens opacities case-control study. Arch Ophthalmol. 1995;113:1113–9.
- Rouhiainen P, Rouhiainen H, Saloneen JT. Association between low plasma vitamin E concentrations and progression of early cortical lens opacities. Am J Epidemiol. 1996;114:496–500.
- Leske MC, Chylack LT, He Q, Wu SY, Schofield F, Friend J, et al. Antioxidant vitamins and nuclear opacities: the longitudinal study of cataract. Ophthalmology. 1998;105:831–6.
- 44. Mares-Perlman JA, Brady WE, Klein R, Klein BE, Bowen P, Stacewicz-Sapuntzakis M, et al. Serum antioxidants and age-related macular degeneration in a population-based case-control study. Arch Ophthalmol. 1995;113(12): 1518–23.
- 45. Vu HTV, Robman L, Hodge A, McCarty CA, Taylor HR. Lutein and zeaxanthin and the risk of cataract: the Melbourne visual impairment project. Invest Ophthalmol Vis Sci. 2006;47(9):3783–6.
- 46. Chasen-Taber L, Willett WC, Seddon JM, Stamper MJ, Rosner B, Colditz GA, et al. A prospective study of carotenoid and vitamin A intakes and risk of cataract extraction in US women. Am J Clin Nutr. 1999;70:517–24.
- Brown L, Rimm EB, Seddon JM, Giovanucci EL, Chasen-Taber L, Speigelman D, et al. A prospective study of carotenoid intake and risk of cataract extraction in US men. Am J Clin Nutr. 1999;70:517–24.
- 48. Lyle BJ, Mares-Perlman JA, Klein BE, Klein R, Greger JL. Antioxidant intake and risk of incident age-related nuclear cataracts in the Beaver Dam eye study. Am J Epidemiol. 1999;149(9):801–9.
- 49. Gale CR, Hall NF, Phillips DIK, Martyn CN. Plasma antioxidant vitamins and carotenoids and age-related cataract. Ophthalmology. 2001;108:1992–8.
- 50. Christen WG, Liu S, Glynn RJ, Gaziano JM, Buring JE. Dietary carotenoids, vitamins C and E, and risk of cataract in women: a prospective study. Arch Ophthalmol. 2008;126(1):102–9.
- Delcourt C, Carriere I, Delage M, Barberger-Gateau P, Schalch W, POLA Study Group. Plasma lutein and zeaxanthin and other carotenoids as modifiable risk factors for age-related maculopathy and cataract: the POLA Study. Invest Ophthalmol Vis Sci. 2006;47(6):2329–35.
- 52. Kistler J, Bullivant S. Structural and molecular biology of the eye lens membranes. Crit Rev Biochem Mol Biol. 1989;24:151–81.
- Borchman D, Cenedella RJ, Lamba OP. Role of cholesterol in the structural order of lens membrane lipids. Exp Eye Res. 1996;62:191–7.
- 54. Simonelli F, Libondi T, Romano N, Nunziata G, D'Aloia A, Rinaldi E. Fatty acid composition of membrane phospholipids of cataractous human lenses. Ophthalmic Res. 1996;28:101–4.
- 55. Rosenfeld L, Spector A. Comparison of polyunsaturated fatty acid levels in normal and mature cataractous human lenses. Exp Eye Res. 1982;35:69–75.
- Hatcher H, Andrews JS. Changes in lens fatty acid composition during galactose cataract formation. Invest Ophthalmol. 1970;9:801–6.
- 57. Hutton JC, Schofield PH, Williams JF, Regtop HL, Hollows FC. The effect of an unsaturated-fat diet on cataract formation in streptozotocin-induced diabetic rats. Br J Nutr. 1976;36:161–7.

- 4 Nutrition and the Aging Eye
- 58. Townend BS, Townend ME, Flood V, Burlutsky G, Rochtchina E, Wang JJ, et al. Dietary macronutrient intake and five-year incident cataract: the Blue Mountains eye study. Am J Ophthalmol. 2007;143(6):932–9.
- Lu M, Cho E, Taylor A, Hankinson SE, Willett WC, Jacques PF. Prospective study of dietary fat and risk of cataract extraction among US women. Am J Epidemiol. 2005;161(10):948–59.
- Arnarsson A, Jonasson F, Sasaki H, Ono M, Jonsson V, Kojima M, et al. Risk factors for nuclear lens opacification: the Reykjavik eye study. Dev Ophthalmol. 2002;35:12–20.
- 61. Cumming RG, Mitchell P, Smith W. Diet and cataract: the Blue Mountains eye study. Ophthalmology. 2000; 107(3):450–6.
- 62. Seddon JM, Ajani UA, Sperduto RD, Hiller R, Blair N, Burton TC, et al. Dietary carotenoids, vitamins A, C, and E, and advanced age-related macular degeneration. Eye Disease Case-Control Study Group. JAMA. 1994;272(18): 1413–20.
- 63. Age-Related Eye Disease Study Research Group, SanGiovanni JP, Chew EY, Clemons TE, Ferris III FL, Gensler G, et al. The relationship of dietary carotenoid and vitamin A, E, and C intake with age-related macular degeneration in a case-control study: AREDS Report No. 22. [see comment]. Arch Ophthalmol. 2007;125(9):1225–32.
- West S, Vitale S, Hallfrisch J, Munoz B, Muller D, Bressler S, et al. Are antioxidants or supplements protective for age-related macular degeneration? Arch Ophthalmol. 1994;112(2):222–7.
- 65. Eye Disease Case-Control Study Group, (EDCCSG). Antioxidant status and neovascular age-related macular degeneration. Arch Ophthalmol. 1993;111:104–9.
- 66. Sanders TAB, Haines AP, Wormald R, Wright LA, Obeid O. Essential fatty acids, plasma cholesterol, and fatsoluble vitamins in subjects with age-related maculopathy and matched control subjects. Am J Clin Nutr. 1993; 57:428–33.
- 67. Snellen EL, Verbeek AL, Van Den Hoogen GW, Cruysberg JR, Hoyng CB. Neovascular age-related macular degeneration and its relationship to antioxidant intake. Acta Ophthalmol Scand. 2002;80(4):368–71.
- 68. Moeller SM, Parekh N, Tinker L, Ritenbaugh C, Blodi B, Wallace RB, et al. Associations between intermediate age-related macular degeneration and lutein and zeaxanthin in the carotenoids in age-related Eye disease study (CAREDS): ancillary study of the Women's health initiative. Arch Ophthalmol. 2006;124(8):1151–62.
- 69. Tan JSL, Wang JJ, Flood V, Rochtchina E, Smith W, Mitchell P. Dietary antioxidants and the long-term incidence of age-related macular degeneration: the Blue Mountains eye study. Ophthalmology. 2008;115(2):334–41.
- Bone RA, Landrum JT, Mayne ST, Gomez CM, Tibor SE, Twaroska EE. Macular pigment in donor eyes with and without AMD: a case-control study. Invest Ophthalmol Vis Sci. 2001;42(1):235–40 [Erratum appears in Invest Ophthalmol Vis Sci. 2001;42(3):548].
- Wooten BR, Hammond BR, Land RI, Snodderly DM. A practical method for measuring macular pigment optical density. Invest Ophthalmol Vis Sci. 1999;40:2481–9.
- Beatty S, Murray IJ, Henson DB, Carden D, Koh H, Boulton ME. Macular pigment and risk for age-related macular degeneration in subjects from a Northern European population. Invest Ophthalmol Vis Sci. 2001;42(2): 439–46.
- Bernstein PS, Shao D-Y, Wintch SW. Resonance Raman measurement of macular carotenoids in normal subjects and in age-related macular degeneration patients. Ophthalmology. 2002;109:1780–7.
- 74. Schweitzer D, Lang GE, Remsch H, Beuerman B, Hammer M, Thamm E, et al. Age-related maculopathy. Comparative studies of patients, their children and healthy controls. (German). Ophthalmologe. 2000;97:84–90.
- 75. LaRowe TL, Mares JA, Snodderly DM, Klein ML, Wooten BR, Chappell R, et al. Macular pigment density and age-related maculopathy in the carotenoids in age-related eye disease study. An ancillary study of the women's health initiative. Ophthalmology. 2008;115(5):876–83.e1.
- 76. Chong EWT, Kreis AJ, Wong TY, Simpson JA, Guymer RH. Dietary omega-3 fatty acid and fish intake in the primary prevention of age-related macular degeneration. Arch Ophthalmol. 2008;126(6):826–33.
- 77. Cho E, Hung S, Willett WC, Spiegelman D, Rimm EB, Seddon JM, et al. Prospective study of dietary fat and the risk of age-related macular degeneration. Am J Clin Nutr. 2001;73:209–18.
- Seddon JM, Rosner B, Sperduto RD, Yannuzzi L, Haller JA, Blair NP, et al. Dietary fat and risk for advanced age-related macular degeneration. Arch Ophthalmol. 2001;119:1191–9.
- Seddon JM, Cote J, Rosner B. Progression of age-related macular degeneration. Association with dietary fat, trans unsaturated fat, nuts and fish intake. Arch Ophthalmol. 2003;121:1728–37.
- Tan JSL, Wang JJ, Flood V, Mitchell P. Dietary fatty acids and the 10-year incidence of age-related macular degeneration: the Blue Mountains eye study. Arch Ophthalmol. 2009;127(5):656–65.
- Smith W, Mitchell P, Leeder SR. Dietary fish and fish intake and age-related maculopathy. Arch Ophthalmol. 2000;118:401–4.
- 82. Sangiovanni JP, Agron E, Meleth AD, Reed GF, Sperduto RD, Clemons TE, et al. {omega}-3 Long-chain polyunsaturated fatty acid intake and 12-y incidence of neovascular age-related macular degeneration and central geographic atrophy: AREDS report 30, a prospective cohort study from the age-related eye disease study. Am J Clin Nutr. 2009;90(6):1601–7.

- Mares-Perlman JA, Brady WE, Klein R, VandenLangenberg GM, Klein BE, Palta M. Dietary fat and age-related maculopathy. Arch Ophthalmol. 1995;113:743–8.
- Heuberger RA, Mares-Perlman JA, Klein R, Klein BE, Millen AE, Palta M. Relationship of dietary fat to agerelated maculopathy in the Third National Health and Nutrition Examination Survey. Arch Ophthalmol. 2001; 119(12):1833–8.
- 85. Age-Related Eye Disease Study 2 Research Group. Lutein+zeaxanthin and omega-3 fatty acids for age-related macular degeneration: the age-related eye disease study 2 (AREDS2) randomized clinical trial. JAMA. 2013;309(19):2005–15.
- Newsome DA, Swartz M, Leone NC, Elston RC, Miller E. Oral zinc in macular degeneration. Arch Ophthalmol. 1988;106(2):192–8.
- Erie JC, Good JA, Butz JA, Pulido JS. Reduced zinc and copper in the retinal pigment epithelium and choroid in age-related macular degeneration. Am J Ophthalmol. 2009;147(2):276–82.e1.
- Cho E, Stampfer MJ, Seddon JM, Hung S, Spiegelman D, Rimm EB, et al. Prospective study of zinc intake and the risk of age-related macular degeneration. Ann Epidemiol. 2001;11:328–36.
- Morris MS, Jacques PF, Chylack LT, Hankinson SE, Willett WC, Hubbard LD, et al. Intake of zinc and antioxidant micronutrients and early age-related maculopathy lesions. Ophthalmic Epidemiol. 2007;14(5):288–98.
- 90. Jacques PF, Taylor A, Hankinson SE, Lahav M, Mahnken B, Lee Y, et al. Long-term vitamin C supplement and prevalence of age-related opacities. Am J Clin Nutr. 1997;66:911–6.
- 91. Robertson JM, Donner AP, Trevithick JR. Vitamin E intake and risk of cataracts in humans. Ann N Y Acad Sci. 1989;570:372–82.
- Chasan-Taber L, Willett WC, Seddon JM, Stamper MJ, Rosner B, Colditz GA. A prospective study on vitamin supplement intake and cataract extraction among US women. Epidemiology. 1999;10:679–84.
- Gritz DC, Srinivasan M, Smith SD, Kim U, Lietman TM, Wilkins JH, et al. The antioxidants in prevention of cataracts study: effects of antioxidant supplements on cataract progression in South India. Br J Ophthalmol. 2006;90(7):847–51.
- 94. Rautiainen S, Lindblad BE, Morgenstern R, Wolk A. Vitamin C supplements and the risk of age-related cataract: a population-based prospective cohort study in women. Am J Clin Nutr. 2010;91(2):487–93.
- Nadalin G, Robman LD, McCarty CA, Garrett SK, McNeil JJ, Taylor HR. The role of past intake of vitamin E in early cataract changes. Ophthalmic Epidemiol. 1999;6:105–12.
- McNeil JJ, Robman L, Tikellis G, Sinclair MI, McCarty CA, Taylor HR. Vitamin E supplementation and cataract: randomized controlled trial. Ophthalmology. 2004;111(1):75–84.
- 97. Age-Related Eye Disease Study Research Group. A randomized, placebo-controlled, clinical trial of high-dose supplementation with vitamins C and E, beta carotene, and zinc for age-related macular degeneration and vision loss: AREDS report no. 8. Arch Ophthalmol. 2001;119(10):1417–36 [Erratum appears in Arch Ophthalmol. 2008;126(9):1251].
- Sperduto RD, Hu TS, Milton RC, Zhao JL, Everett DF, Cheng QF, et al. The Linzian cataract studies. Two nutrition intervention trials. Arch Ophthalmol. 1993;111:1246–53.
- 99. Chylack LTJ, Brown NB, Bron A, Hurst M, Kopcke W, Thien U, et al. The Roche European American cataract trial (REACT): a randomized clinical trial to investigate the efficiency of a antioxidant micronutrient mixture to slow progression of age-related cataract. Ophthalmic Epidemiol. 2002;9:49–80.
- 100. Age-Related Eye Disease Study 2 (AREDS2). [cited 9 May 2013]. Available from: www.nei.nih.gov/areds2
- 101. Seddon JM, Christen WG, Manson JE, LaMotte FS, Glynn RJ, Buring JE, et al. The use of vitamin supplements and the risk of cataract among US male physicians. Am J Public Health. 1994;84(5):788–92.
- 102. Christen WG, Glynn RJ, Chew EY, Buring JE. Vitamin E and age-related macular degeneration in a randomized trial of women. Ophthalmology. 2010;117(6):1163–8.
- Taylor HR, Tikellis G, Robman LD, McCarty CA, McNeil JJ. Vitamin E supplementation and macular degeneration: randomised controlled trial. BMJ. 2002;325(7354):11.
- 104. Teikari JM, Laatikaineen L, Virtamo J, Haukka J, Rautalahti M, Liesto K, et al. Six-year supplementation with alpha-tocopherol and beta-carotene and age-related maculopathy. Acta Ophthalmol Scand. 1998;76:224–9.
- 105. Solberg Y, Posner M, Belkin M. The association between cigarette smoking and ocular diseases (review). Surv Ophthalmol. 1998;42:535–47.
- 106. Newsome DA. A randomized, prospective, placebo-controlled clinical trial of a novel zinc-monocysteine compound in age-related macular degeneration. Curr Eye Res. 2008;33(7):591–8.
- 107. Stur M, Tittl M, Reitner A, Meisinger V. Oral zinc and the second eye in age-related macular degeneration. Invest Ophthalmol Vis Sci. 1996;37(7):1225–35.
- Chong EWT, Wong TY, Kreis AJ, Simpson JA, Guymer RH. Dietary antioxidants and primary prevention of age related macular degeneration: systematic review and meta-analysis. [see comment]. BMJ. 2007;335(7623):755.

4 Nutrition and the Aging Eye

- 109. Kris-Etherton PM, Hecker KD, Bonanome A, Coval SM, Binkoski AE, Hilpert KF, et al. Bioactive compounds in foods: their role in the prevention of cardiovascular disease and cancer. Am J Med. 2002;113(Suppl 9B):71S–88.
- 110. USDA. Dietary Guidelines for Americans 2005. http://www.health.gov/dietaryguidelines/dga2005/document/
- 111. The Alpha-Tocopherol Beta-Carotene Cancer Prevention Study Group. The effect of vitamin E and beta-carotene on the incidence of lung cancer and other cancers in male smokers. N Engl J Med. 1994;330:1029–35.
- 112. Omenn GS, Goodman GE, Thornquist MD, Balmes J, Cullen MR, Glass A, et al. Risk factors for lung cancer and for intervention effects in CARET, the beta-carotene and retinol efficiency trial. J Natl Cancer Inst. 1996; 88:1550–9.
- 113. Nourmohammadi I, Modarress M, Khanaki K, Shaabani M. Association of serum alpha-tocopherol, retinol and ascorbic acid with the risk of cataract development. Ann Nutr Metab. 2008;52(4):296–8.
- 114. Lyle BJ, Mares-Perlman JA, Klein BE, Klein R, Palta M, Bowen PE, et al. Serum carotenoids and tocopherols and incidence of age-related nuclear cataract. Am J Clin Nutr. 1999;69(2):272–7.
- 115. USDA ARS Nutrient Data Laboratory. National Nutrient Database for Standard Reference. 2010 [cited 1 November 2010]. Available from: www.nal.usda.gov/fnic/foodcomp/search/
- 116. US Department of Agriculture. USDA-NCC Carotenoid Database for US Foods—1 Nov 1998. 2012. Available from: www.ars.usda.gov/SP2UserFiles/Place/12354500/Data/SR25/nutrlist/sr25a338.pdf
- 117. Kris-Etherton PM, Taylor DS, Yu-Poth S, Huth P, Moriarty K, Fishell V, et al. Polyunsaturated fatty acids in the food chain in the United States. Am J Clin Nutr. 2000;71:179S–88.

Chapter 5 Nutrition and Oral Health: A Two-Way Relationship

Cristina Palacios and Kaumudi J. Joshipura

Key Points

- Tooth loss reduces the ability to consume a more varied diet and increases systemic disease risk.
- The consumption of fermentable carbohydrates, along with increased use of xerogenic medications, places older adults at high risk for dental caries.
- Recent studies suggest an important role for calcium, vitamin D, and possibly vitamin C in reducing periodontitis risk.
- Diets high in fruits and vegetables have consistently been shown to reduce oral cancer risk.
- Tooth loss is associated with increased tendency among older adults to consume poorer quality diets, which in turn could increase cardiovascular risk.

Keywords Edentulism • Dental caries • Periodontitis • Oral cancer • Nutrients • Dietary patterns • Foods

Introduction: Oral Health Status in Older Adults

Oral health contributes greatly to quality of life in older adults. Poor oral health can hinder a person's ability to sustain a satisfying diet, participate in interpersonal relationships, and maintain a positive self-image [1-3]. Oral health problems may lead to chronic pain, discomfort, and alterations in diet that may adversely impact systemic disease.

Dental caries and chronic periodontitis are by far the most common oral diseases in the elderly [4]. These two diseases are the major causes of tooth loss and thus the major cause of dental morbidity. Caries, periodontal disease, and tooth loss still continue to be significant public health problems in the United States. Other oral diseases or conditions are relatively rare, of lesser importance from a public

C. Palacios, L.N.D., M.Sc., Ph.D. (🖂)

Nutrition Program, Graduate School of Public Health, University of Puerto Rico, Medical Sciences Campus, Main Building, Office B-450, 4th floor, San Juan 00935, Puerto Rico e-mail: cristina.palacios@upr.edu

K.J. Joshipura, B.D.S., M.S., Sc.D.

Center for Clinical Research and Health Promotion, School of Dental Medicine, University of Puerto Rico, Medical Sciences Campus, San Juan 00935, Puerto Rico e-mail: Kaumudi.joshipura@upr.edu

health perspective, and less related to nutrition, even though they may predominantly affect the elderly (e.g., denture stomatitis and other soft-tissue lesions). We will therefore focus on dental caries, chronic periodontitis, and tooth loss as the most common oral health problems in the elderly. In addition, we will also discuss oral cancer, which although rare, is often fatal and thus of high public health importance.

Common Oral Conditions in Older Adults

Dental Caries

Among dentate adults aged 60 years or older, 93 % had caries experience, which was higher among whites (93.3 %) compared to blacks (84.6 %) and Mexican Americans in 1999–2002 (83.5 %) [4]. In addition, 18.6 % had untreated tooth decay and there was a mean of 9.1 decayed and filled permanent teeth. A more recent statistic shows that almost 20 % of dentate adults 65 years of age or older had at least one tooth with untreated tooth decay in 2005–2008 [5]. On average, these seniors had about 18 decayed, missing, or filled teeth.

Root Caries

Root caries is also an important problem in older adults. Data from the National Health and Nutritional Examination Survey (NHANES) 1988–1994 and 1999–2002 shows that the prevalence of root caries is 31.6 % among those aged >60 years [4]. Overall, the prevalence of untreated root caries decreased in this age group (from 20 to 12.8 %). In addition, approximately 22 % of those aged >60 years had one or more filled roots (restored), which is higher compared to those aged 40–59 years (8.0 %) and those aged 20–39 years (1.7 %).

Periodontal Disease

Data from NHANES 2009–2010 show that the prevalence of periodontitis was 47 % in adults aged 30 years and older; and in those aged 65 years and older, 64 % had either moderate or severe periodontitis [6]. Furthermore, this report also shows that periodontitis was highest among men, Mexican Americans, current smokers and among those with less than a high school education and below 100 % of Federal Poverty Levels.

Tooth Loss

In 2009–2010, 15 % of adults aged 65–74 and 23 % of seniors aged 65 and over were edentulous (i.e., had lost all their natural teeth) [5]. The good news is that data from several representative health surveys in the United States clearly demonstrate a steady decline in the prevalence of edentulism, caries, and periodontitis over the past several decades beginning in the 1960s. However, not all segments of the population have benefited equally from this trend and tremendous heterogeneity and disparities exist between socioeconomic and racial/ethnic groups. For example, more non-Hispanic black adults aged 65 and over were edentulous (32 %) compared with non-Hispanic white adults (22 %), whereas only 16 % of Mexican-American adults had complete tooth loss [5]. Also, data from the National Health Interview Surveys from 1999 to 2008 shows that there was a significant reduction in edentulism rates between these years, with the highest rate among Native Americans (24 %), followed by African Americans (19 %), Caucasians (17 %), Asians (14 %), and Hispanics (14 %) [7].

Also, the prevalence of complete edentulism was more than twice as high for those living at or below 100 % of the federal poverty level (34 %), compared to those above this level (13 %) among adults aged 65–74 years but not among those aged 75 and over.

In summary, dental caries and periodontitis remain highly prevalent in the elderly. Although in most industrialized countries the prevalence of caries has declined over the past decades, dental caries and periodontitis and the associated tooth loss continue to be major public health problems.

Oral Cancer

The most serious and potentially fatal oral condition among older adults is oral cancer. By far the most common form of malignant oral cancer is squamous cell carcinoma. Squamous cell carcinoma of the oral cavity (including the gums, tongue and floor of mouth and other oral cavity), the oropharynx, hypopharynx, and larynx is frequently called "head and neck squamous cell carcinoma" or "head and neck cancer."

More than 7,800 people, mostly older Americans, die from oral and pharyngeal cancers each year [8]. In 2007, there were 34,360 newly diagnosed cases of oral and pharyngeal cancers (24,180 in males, 10,180 in females) and 7,550 oral and pharyngeal cancer deaths (5,180 males, 2,370 females) in the United States [8]. This increased in 2012 to 40,250 newly diagnosed cases of oral (28,540 in males, 11,710 in females) and 7,850 oral and pharyngeal cancer deaths (5,440 in males and 2,410 in females) [9]. In 2009, the oral cancer incidence increased from 10.5 cases per 100,000 individuals in 2000–2004 to 10.9 cases, with increasing rates in men (16.1/100,000) and in women (6.2/100,000) [10]. However, this report also showed that mortality from oral and pharyngeal cancer decreased to 2.4/100,000 individuals compared to 2000–2004 (males: 3.7/100,000; females: 1.3/100,000). Oral and pharyngeal cancer incidence and mortality are the highest among persons 65 years and older. In 2005–2009, the age-standardized incidence rate for this age group was 39.7/100,000 (males: 59.6/100,000), females: 24.9/100,000) and the mortality rate was 11.8/100,000 (males: 17.7/100,000, females: 7.4/100,000).

There are major racial/ethnic disparities in oral cancer incidence and particularly in oral cancer mortality and survival. For example, mortality rates per 100,000 were 5.7 and 1.4 for Black males and females, compared to 3.6 and 1.4 for White males and females, respectively [10]. The 5-year survival rate for oral and pharyngeal cancer diagnosis slightly improved from 60.4 % in 1999–2001 to 64.6 % in 2002–2008. This improvement was observed in Whites (55 % in 1975–1977 to 66.5 % in 2002–2008), as in African Americans (36.1 % in 1975–1977 to 45.2 % in 2002–2008). Although the oral cavity and pharynx are easily accessible for inspection, only 33 % of oral and pharyngeal cancers are diagnosed in a localized stage (35 % among Whites, 21 % among Blacks). The 5-year relative survival rate is highly dependent on stage of diagnosis. More than 80 % of patients with localized disease survive 5 years, compared to 57.3 and 34.9 % with regional and distant metastases, respectively. However, even if diagnosed at the same stage, Blacks have lower 5-year relative survival rates than Whites.

Impact of Nutritional Status on Oral Health

Plaque and Calculus Formation

Bacteria in the mouth, or oral flora, form a complex community or biofilm that adheres to teeth and is called plaque. These bacteria ferment sugars and carbohydrates and generate acid, which can in turn dissolve minerals in tooth enamel and dentin and lead to dental caries. Furthermore, bacterial products and components elicit an inflammatory immune response in the gingival epithelium and underlying

connective tissues (gingivitis) that may lead to periodontitis in susceptible individuals. Although the presence of plaque itself is not sufficient to cause either caries or periodontitis, the current understanding of the pathogenesis views bacterial plaque as a necessary cause for both diseases. Hence, oral hygiene measures that aim to remove or reduce bacterial plaque are a key strategy for the prevention of both caries and periodontitis.

Plaque can be present in subjects who do not consume carbohydrates, but is more prolific and produces more acid in individuals who eat sucrose-rich food. Frequency of carbohydrate consumption, physical characteristics of food (e.g., softness and stickiness), and timing of food intake all contribute to plaque formation and composition [11]. Plaque on tooth surfaces mineralizes to form calculus or tartar, which is often covered by unmineralized biofilm [12]. Hence, dietary factors are important determinants of plaque quantity and quality and are therefore important in the pathogenesis of dental disease, in particular dental caries (see below).

Dental Caries

Dental caries are characterized by the demineralization of dental hard tissues (enamel and dentin) by acids produced by plaque bacteria. Many factors that influence the quantity and timing of bacterial acid production ultimately determine the risk of caries. Here, diet plays an important role in caries occurrence and progression. Foods containing fermentable carbohydrates result in acid production by cariogenic plaque bacteria. The production of organic acids by sugar metabolizing bacteria then leads to significant decreases in plaque pH. If plaque pH falls below 5.5 for an appreciable period of time, demineralization of dental enamel occurs. As the plaque pH varies according to the availability of fermentable carbohydrates to plaque bacteria, de-mineralization and remineralization processes occur in a dynamic process. Factors other than diet that affect this dynamic process include, for example, the fluoride concentration. If de-mineralization is not compensated by remineralization, a breakdown of the enamel surface and formation of a cavity that can extend through the dentin (the part of the tooth located under the enamel) to the pulp tissue will result.

Because of the complexity of the demineralization/remineralization processes, the effects of fermentable sugars and other carbohydrates are not just determined by their amount. Most importantly, the frequency of sugar intake (e.g., eating sweets with main meals or as snacks at multiple occasions between meals) has been clearly shown to be a major determinant of caries risk [13]. Consumption of soft drinks containing sugar is also related to increased caries prevalence and incidence [14].

Furthermore, the effect of dietary intake of sugars or other carbohydrates is modified by other factors, primarily fluoride intake and oral hygiene. Fluorides (e.g., in toothpastes) have become highly abundant over the past decades and dietary factors may be less important in subjects with good oral hygiene and regular fluoride exposure [15]. Artificial sweeteners such as aspartame and saccharin and sugar alcohols such as sorbitol, mannitol, and xylitol were shown to be non-cariogenic in clinical trials [16]. Chewing sugar-free gum with added xylitol or sorbitol has been shown in clinical trials to be effective in reducing caries [17], although this caries-preventive effect of chewing gums with sugar alcohol could be related to stimulation of the salivary flow. Indeed, saliva also contains components that can directly attack cariogenic bacteria and contains calcium and phosphates that help remineralize tooth enamel.

Studies have also found the consumption of dairy products is non-cariogenic. A high yogurt intake was significantly associated with a lower prevalence of dental caries consumption compared with low yogurt consumption in Japanese children [18]. In addition, cheese consumption was inversely associated with caries experience in children from Spain [19]. A longitudinal study in 600 Japanese aged 70 years or more also found that intake of milk and other dairy products was protective of root caries [20].

Chronic Periodontitis

Bacterial plaque is considered a necessary cause of periodontitis, as bacterial components and products elicit an inflammatory response in the periodontal host tissues. In susceptible individuals, this inflammatory response leads to the resorption of periodontal ligament and alveolar bone. Susceptibility to periodontitis is determined by environmental and genetic host factors. For example, smoking and diabetes are established as major risk factors for periodontitis and tooth loss [21, 22]. Genetic risk factors that increase periodontitis susceptibility have been proposed, although to date data on specific genetic factors remain equivocal [23, 24], except for vitamin D receptor (VDR) polymorphism (discussed later in the chapter).

Diet and nutrients could affect periodontitis risk by influencing plaque quality and quantity, but may also, and perhaps more importantly, affect the inflammatory response and thus affect periodontitis susceptibility. It is important to note that research in the nutritional determinants of periodontitis risk is scant and most of the currently available data are from cross-sectional surveys, particularly from the NHANES III. In NHANES III, inverse associations between the intake of calcium [25, 26] and dairy products [26]—which is highly correlated with calcium intakes in the United States—and periodontitis prevalence have been reported. In addition, a study in more than 900 Japanese found that an increased intake of lactic acid foods (yogurt and lactic acid drinks or fermented milk, also known as kefir) was associated significantly with lower periodontal disease [27]. An intake of 55 g or more of lactic acid significantly lower the prevalence of deep probing depth and severe clinical attachment loss compared to those not eating these foods, after adjusting for confounding variables. More recently, a study in older Danish adults showed that intakes of dairy foods, milk, and fermented foods were significantly associated with a reduction of periodontitis risk (p < 0.05), but cheese and other dairy foods intakes were not associated after adjusting for confounding variables [28]. Krall et al. have reported a beneficial effect of calcium and vitamin D supplementation on tooth retention [29] in a small cross-sectional study. Vitamin D status is associated with bone mineral density [30] and a recent meta-analysis showed that vitamin D supplementation is effective in preventing bone loss and fractures, particularly in doses greater than 800 IU/day and combined with calcium supplementation [31]. Because osteoporosis has been proposed as a risk factor for periodontitis, we studied the association between vitamin D status and periodontitis and found lower periodontitis prevalence among subjects older than 50 years of age with higher serum 25(OH)D levels [32]. In a recent publication among 42,730 participants of the Health Professionals Follow-Up Study, the highest quintile of the predicted 25(OH)D score (which is derived from variables known to influence 25(OH)D levels) was associated with a 20 % lower incidence of tooth loss (risk=0.80, 95 % CI: 0.76–0.85) and 20 % lower incidence of periodontitis (risk=0.80; 95 % CI: 0.71–0.90) compared to the lowest quintile [33]. In addition to its established effect on calcium metabolism and bone, vitamin D also has immuno-modulatory functions by which it may reduce periodontitis susceptibility. This hypothesis is consistent with the finding of a strong inverse association between vitamin D status and gingivitis (a precursor of periodontitis not affected by osteoporosis) prevalence [34]. If vitamin D status is truly a risk factor for periodontitis, these findings could have significant public health implications as hypovitaminosis D is highly prevalent in the United States and elsewhere [35], particularly in individuals who are overweight/obese have lower milk intake and/or have less sun protection [36].

Interestingly, VDR polymorphisms are among the genetic factors implicated as putative risk factors for periodontitis. A meta-analysis exploring the association between VDR polymorphisms (Taq-I, Bsm-I, Apa-I, and Fok-I) and periodontitis found that in Asians, chronic periodontitis cases had a significantly lower frequency of bb genotype of BsmI but higher frequency of AA genotype of ApaI with no associations with the other variants or with the aggressive form of the disease [37]. A more recent meta-analysis found that Taq-I variants could be protective of chronic periodontitis, while Fok-I locus could be a risk factor of aggressive periodontitis in Asians, but not in whites [38].
However, intervention studies will be necessary to evaluate if vitamin D supplementation is effective for periodontitis prevention. Up to date, there are no clinical trials examining this. A small cross-sectional study in 51 subjects receiving periodontal maintenance therapy found that those taking calcium ($\geq 1,000 \text{ mg/day}$) and vitamin D ($\geq 400 \text{ IU/day}$) supplements had a better periodontal health compared to those not taking supplements [39]. A 1-year follow-up showed that those still taking the supplements had a modest positive effect on periodontal health compared to those not taking the supplements [40].

Given our increasing understanding of the role of immune function and inflammatory response in periodontitis, it is likely that immune-modulating nutrients, such as some antioxidants and omega 3 fatty acids could alter the inflammatory process in periodontitis. Deficiencies of ascorbate have been associated with severity of gingivitis [41]. Furthermore, NHANES III data demonstrate an inverse association between vitamin C intakes and periodontitis prevalence in the United States [42]. Serum levels of vitamin C have been shown to be strongly associated with periodontitis prevalence, which was also confirmed among never smokers [43]. A retrospective cohort study in community-dwelling older Japanese found that a higher intake of dietary antioxidants (vitamin C, E, and β -carotene) was inversely associated with the number of teeth with periodontal disease progression, controlling for other variables [44]. The carotenoids beta-cryptoxanthin and beta-carotene [45], and omega 3 fatty acids [46], particularly docosahexaenoic acid [47], could also alter the inflammatory process in periodontitis. Interestingly, Merchant et al. reported an inverse association between whole-grain intake and risk of self-reported periodontitis incidence in a large cohort of male U.S. health professionals [48]. Men in the highest quintile of whole-grain intake were 23 % less likely to develop periodontitis.

Associations between obesity and periodontitis have been noted in several cross-sectional studies [49, 50] and in a longitudinal study [51]. More recently, a meta-analysis found a 35 % greater risk of periodontal disease in obese individuals, with some evidence of a stronger association found among younger adults, women, and nonsmokers [52]. The Risk of periodontal disease was found to be significantly greater in men with high BMI (\geq 30 kg/m²) when compared to men with BMI in the normal weight range (18.5–24.9 kg/m²; HR = 1.30; 95 % CI: 1.17–1.45) [53]. Increased WC and WHR were highly associated with a greater risk of periodontal disease (HR for extreme quintiles: WC=1.27, 95 % CI: 1.11–1.46; WHR=1.34, 95 % CI: 1.17–1.54). In addition, a higher mean body mass index (BMI) was found among periodontal patients and a greater mean clinical attachment loss was found among obese individuals. Given that adipose tissue serve as a reservoir for inflammatory cytokines, it is possible that increasing body fat increases the likelihood of an active host inflammatory response in periodontitis. This was recently evidenced in a study in 893 men followed for 40 years, whereas those overweight at baseline with the highest rapid weight gain follow-up (>0.19 kg/year) had significantly more probing pocket depth (PPD) events than those in the lowest tertile of weight gain (\leq -0.05 kg/ year) [54]. In addition, this study found that an increase in waist circumference (WC) in overweight men led to more PPD events than those with little change in WC. Furthermore, BMI and WC-toheight ratio were significantly associated with periodontal disease progression regardless of periodontal disease indicator.

Oral Cancer

Oral cancer is generally preceded by precancerous lesions, which include oral epithelial dysplasia, erythroplakia, leukoplakia, lichen planus, and submucous fibrosis (rare in Western countries). The major risk factors for oral cancer are tobacco and alcohol use. In Asian countries chewing tobacco, beetle nut, and beetle quid are major risk factors. Chewing tobacco use is also increasing in the United States. The relation between nutrition and oral cancer, and the impact of oral cancer on the patient's ability to eat and swallow are discussed below.

Fruits and Vegetables

Fruits and vegetables are rich in many nutrients, including fiber, vitamin C, vitamin E, folate, carotenoids, phenolics, isoflavonoids, isothiocyanates, and indoles, among others. The consumption of these foods has been shown in many studies to be protective of chronic disease, including cancer. A consistent finding across numerous studies is that a diet high in fruits and vegetables is protective against oral pre-cancer [55–57] and cancer [58, 59]. A review of six cohort studies and about 40 case–control studies found a relative risk for high vegetable and fruit consumption of 0.65 (95 % CI: 0.53–0.80) and 0.78 (95 % CI: 0.64–0.95) from three cohort studies on upper aerodigestive tract cancers and 0.52 (95 % CI: 0.45–0.61) and 0.55 (95 % CI: 0.47–0.65) from 18 case–control studies of oral and pharyngeal cancer, respectively [60]. In addition, a generous consumption of fruits is associated with a 20–80 % reduced risk of oral cancer even when smoking and alcohol intake and other factors including total caloric intake are taken into account. Furthermore, a high fruit and vegetable diversity has also been shown to be protective of oral cancer.

However, not all studies show a protective effect of fruits and vegetables in reducing cancer risk [61]. The inconsistencies may be explained by variation in specific vegetables and fruits consumed, as there are differences in the ability between these foods to suppress carcinogenesis. Allium vegetables (such as onions and garlic), carrots, green and leafy vegetables, cruciferous vegetables, and tomatoes, have been shown to be fairly consistently oral cancer protective [60, 62]. In particular, cruciferous vegetables, such as broccoli, cabbage, cauliflower, kale, and mustard, which are rich in carotenoids, vitamin C, folate, soluble fiber, and glucosinolate, also have potent anti-carcinogenic activity. A large Japanese study and the Iowa Women's Health Study also found that a high consumption of green and yellow vegetables decrease oral cavity and pharynx cancer risks [63, 64]. Tomato also shows a strong and consistent inverse association for oral cancer in 12 of 15 studies [65] and in one study on leukoplakia [57]. Citrus fruits have also been shown to have beneficial effect on oral and pharyngeal cancer risk [60].

Also, there is more evidence of an anti-carcinogenic effect of raw vegetables compared to cooked vegetables [66, 67], particularly raw and green/leafy vegetables. Raw tomatoes were more associated with reduce risk of oral cancer than cooked tomatoes [65]. This difference between raw and cooked vegetables could be related to changes that occur during cooking, such as loss of some nutrients, destruction of digestive enzymes, and changes in the structure and digestibility of the food [66]. Studies that have evaluated subgroups have generally found higher beneficial effects of fruits and vegetables and their constituent micronutrients among smokers and drinkers than among abstainers [68].

Anti-oxidants and Other Micronutrients

Several nutrients found in vegetables and fruits show an inverse association with oral cancer. These include vitamin A, vitamin B_{12} , vitamin C, tocopherol (vitamin E), retinoids, carotenoids, lycopene, beta-carotene, folate, glutathione, thiamin, vitamin B_6 , niacin, lutein, and flavonoids have been inversely associated with oral cancer [58] and pre-cancer [57, 69, 70] in one or more studies.

Vitamin C, in particular, has been shown to reduce the risk of oral premalignant lesions, but only from dietary sources, not from supplements in an analysis of the Health Professionals Follow-up Study in 42,340 men [71]. However, a recent pooled analysis of 12 case–control studies of head and neck cancer (7,002 cases and 8,383 controls) did find a decreased risk with the ever use of vitamin C, particularly if used for 10 or more years [72].

Retinoids and beta-carotene in controlled therapeutic doses show protective effects, with fewer new primary tumors in persons with previous oral cancers and reversals or reduction in size of premalignant lesions [73]. High doses of 13-cisretinoic acid (50–100 mg/m² body surface area/day for a year) have been effective in the treatment of oral leukoplakia [74]. This therapy has resulted in complete resolution in 27–57 % of patients and a partial response in 45–90 % of patients. Trials using beta-carotene supplements (60 mg/day for 6 months) have shown reduced risk of oral cancers and remission of pre-cancers with an improvement of at least one grade dysplasia in 39 % and no change in 61 % [75]. Cohort and case–control studies also suggest reduced risk of oral cancer among people who consumed different carotenoids [76], including beta-cryptoxanthin and alpha-carotene [71].

Lycopene, a red-colored carotenoid mainly found in tomatoes but also in other fruits and vegetables, has been shown in epidemiologic studies to have anti-carcinogenic effects for preventing oral cancer and in preliminary human clinical trials to have potential therapeutic effects in precancerous lesions [77]. Folate intake has been found to be also associated with oral cancer risk, but a significant interaction is found with alcohol intake. A cohort of 87,621 women in the Nurses' Health Study found that low alcohol intake reduced the risk of oral cancer compared to nondrinkers and to those with high alcohol intake (RR=0.59; 95 % CI: 0.39–0.87) [78]. However, among those with high alcohol (\geq 30 g/ day) and low folate (<350 µg/day) intakes, cancer risk increased (risk=3.36; 95 % CI: 1.57–7.20) compared with nondrinkers with low folate intake. This was not observed among those with high alcohol intake and high-folate (\geq 350 µg/day). This could be explained to interactions between alcohol and folate intakes with aldehyde dehydrogenase 2 genotype [79]. Glutathione—an antioxidant found in fruits and vegetables—was protective only if it was derived from fruit and raw vegetables [80].

Other Food and Nutrients

A protective effect of fiber and whole grains was observed for both oral submucous fibrosis and leukoplakia [57] and for oral cancer [58], particularly in women [81, 82]. There is a suggestion that meat, desserts, maize, and saturated fats and/or butter may be risk factors while olive oil may be protective [58]. Nitrate, nitrite, and nitrate reductase activity in saliva [83] and high intake of nitrite containing meats [84] have been linked with increased risk. Iron is suggested to be protective against oral cancer [70] and leukoplakia [85]. Ever use of calcium supplement also has been shown to reduce the risk of oral cancer [72].

Dietary Patterns

Studies have also investigated the association of dietary patterns with oral cancer risk, as opposed to individual foods or nutrients. A case-control study in Indonesia found that a dietary pattern characterized by cooked and raw vegetable, fast foods, fermented foods, seafood, canned food, snacks high in fat and sugar increased two times risk of oral cancer (OR = 2.17; 95 % CI: 1.02–4.50) [86]. Although vegetable intake has been shown to reduce the risk in other studies, this case control study did not observe this when combined with other nonhealthy patterns. However, a dietary pattern characterized by dairy products, meat, and fruit reduced the risk (OR = 0.50; 95 % CI: 0.24–1.00), even though total meat consumption is related to an increase risk of oral cancer, particularly red meat [87]. Another case-control study in Brazil found that a pattern characterized by vegetables and raw vegetables, fruit, dairy products, potato, and fish reduced the risk of oral cancer (OR = 0.44; 95% CI: 0.25–0.75) as well as a pattern characterized by rice and pasta and pulses (OR=0.53; 95% CI: 0.30–0.93), while a pattern characterized by bread, butter, cheese, pork, sandwich meat, egg, and sweets and dessert appeared to increase the risk (OR = 1.25; 95% CI: 0.73–2.15) [88]. An Italian case-control study in 804 with oral cancer and 2,080 controls found that a vitamin and fiber-rich dietary pattern was inversely associated with oral cancer (OR = 0.47; 95 % CI: 0.34–0.65) [59]. A recent pooled analysis of five case-control studies also found that a pattern characterized as rich in antioxidant vitamins and fiber decreased oral cancer risk (OR=0.64; 95 % CI 0.45–0.90), whereas the fat-rich pattern or the animal products and cereals pattern did not show a significant association [89]. Another case-control study in Malaysia found that a pattern characterized by intake of beverages and starches or a pattern rich in dairy, fermented/salted, and meat/by-products significantly increased oral cancer risk [90].

In summary, most studies show that a consumption of fruits, vegetables, other fiber-rich foods, and fish are associated with a reduced cancer risk while saturated fat, processed and refined foods, and fast foods are associated with an increased risk.

Impact of Oral Health on Nutrition

This section focuses on the impact of tooth loss and dentition status, oral cancer, and xerostomia on nutrition. Other aspects of oral health such as oral pain, periodontal disease, and altered taste could also have some impact on nutritional status [91] but will not be reviewed here.

Impact of Tooth Loss on Nutritional Status

A number of studies have demonstrated an association between tooth loss and dietary intake. Many studies show that edentulous individuals (people with no teeth) are more likely to eat an unhealthy diet (e.g., ingesting too few nutrient-dense foods and too much calorie-rich, high-fat foods) compared to people with natural teeth. Joshipura et al. [92] observed that edentulous male health professionals consumed fewer vegetables, less fiber and carotene, more cholesterol, saturated fat, and calories than participants with 25 or more teeth after adjusting for age, smoking, exercise, and profession. In a representative sample from a state in Nigeria among 500 randomly selected Nigerians, aged 50 and above, people with 20 or fewer teeth were more likely to indicate that they avoided certain food items because of their teeth/mouth conditions. Multivariable logistic regression analysis, controlling for age, gender, income, and education showed that the number of remaining teeth was significantly associated with the selection of vegetables/fruits and other hard food items [93]. In other studies of healthy older adults, edentulous individuals have been noted to consume fewer fruits and vegetables, lower amounts of fiber, and higher amounts of fat [94, 95]. Edentulous individuals are more likely to have lower intakes of micronutrients, such as calcium, iron, pantothenic acid, vitamins C, and E, than their dentate counterparts [95–97]. In summary, most of the studies relating tooth loss and nutrition suggest that people with fewer teeth are more likely to have compromised nutritional intake. Possible changes in fruit, vegetable, and micronutrient intake after tooth loss may explain part of the associations between tooth loss and cardiovascular and other systemic disease [98]. Therefore patients with tooth loss warrant aggressive counseling regarding methods to maintain dietary quality, such as blending or shredding fresh fruits and vegetables to preserve adequate intake [99].

Although eating with dentures may be preferable to eating with no teeth, most studies suggest that the diet of denture wearers differs from the diet of people who retained their natural teeth. In a study of veterans [100], individuals with full dentures consumed fewer calories, thiamin, iron, folate, vitamin A, and carotene than individuals with a number of natural teeth remaining. Also, those wearing dentures consume more refined carbohydrates, sugar, and dietary cholesterol than their dentate counterparts [101, 102]. The above studies could be interpreted such that the presence of dentures contributes to poorer intake across multiple nutrients compared to dentate subjects. Poor denture fit may contribute to some of these differences. However, all of these studies report cross-sectional associations, and tooth loss resulting from caries and/or periodontitis may well be the outcome of poor nutrition rather than its determinant. Alternatively, both pathways may have a role and explain the cross-sectional association between tooth loss and diet.

Surprisingly, longitudinal studies investigating whether tooth loss leads to dietary changes (secondary to an assumed functional impairment) are scarce. A longitudinal study of 8 years among 31,813 male U.S. health professionals found that men who lost five or more teeth had significant detrimental changes in dietary intakes of dietary fiber, whole fruit, dietary cholesterol, and polyun-saturated fat compared to men who did not lose teeth [103]. Similarly, results from the Nurses' Health Study showed detrimental dietary changes over a 2-year period subsequent to incident tooth loss, with a tendency for women who lost teeth to avoid hard foods such as raw carrot, fresh apple, or pear [104]. However, these differences were relatively small in absolute terms. The significance of these findings with respect to chronic disease risk is uncertain, although small effects on several nutrients could add up to impact disease risk.

Impact of Tooth Loss on Body Weight Status

The relationship between dental status, weight, and BMI varies with the population studied. In nursing home residents, compromised oral functional status was associated with lower BMI (less than 21 kg/ m²) after controlling for functional dependence and age [105]. Another study in community-dwelling older adults found a similar association between masticatory ability and lower body weight, after controlling for age and sex [106]. However, another study found that healthy older edentulous adults actually had higher BMIs, compared to dentate subjects [107]. Similarly, a study in Brazilian older adults found that edentulous persons wearing only upper dentures and those wearing 0-1 prosthesis were more likely to be obese [108]. A recently published longitudinal study of 16,416 Swedish older adults found a significant association between edentulism and obesity, particularly in women [109]. The differences in findings in these studies may be due to the different characteristics of the populations evaluated, with sicker older adults more likely to lose weight in response to altered dentition, and healthier older adults more likely to maintain adequate intake but alter intake to softer foods that are more calorie dense. However, as noted above, cross-sectional studies are insufficient to make causal inferences and obesity may itself be a risk factor for tooth loss secondary to caries or periodontitis. One longitudinal study in community dwelling older adults found that over a 1-year period of follow-up, approximately one-third of the sample had lost 4 % or more of their previous total body weight, 6 % of men and 11 % of women lost 10 % or more of their previous body weight [50]. Edentulism remained an independent risk factor for significant weight loss (OR 1.6 for 4 % weight loss and 2.0 for 10 % weight loss) after controlling for gender, income, advanced age, and baseline weight.

Impact of Dentate Status on Blood Nutrient Status

The largest study to date to evaluate blood nutrient status in relation to dentate status is the British National Diet and Nutrition Survey [96]. In their cross-sectional study of 490 free-living and institutionalized older adults, the authors reported that edentulism subjects had significantly lower mean plasma levels of retinol, ascorbate, and tocopherol than dentate subjects, after controlling for age, sex, social class, and region of residence. Among dentate subjects, mean plasma vitamin C levels were positively associated with increased numbers of occlusal pairs of teeth. Another study of adults in Sweden [107] reported lower serum high-density lipoprotein levels among edentulous individuals compared to those who were dentate. These results are consistent with the studies relating dietary intake to dentition status. Once again, most studies mentioned are cross-sectional and have to be interpreted with caution; it is not clear if nutrition impacts tooth loss through its impact on caries and periodontal disease or if tooth loss impacts nutritional intake or both. Another important issue that needs to be considered when evaluating the evidence relating to tooth loss and diet is the possibility for residual confounding, in particular by socioeconomic status and health-conscious behaviors. Although caries and/or periodontitis are the main causes of tooth loss, a decision to extract a tooth is influenced by many other factors working at the patient, provider (dentist), and community level (access to care), and confounding by socioeconomic factors is a particular concern in this context [110]. In summary, individuals with compromised dentition tend to have poorer dietary quality. Whether or not this association is causal, i.e., whether or not tooth loss leads to important unfavorable changes in a person's diet, is uncertain [91]. Additional longitudinal studies are necessary to answer this important question. Prevention of dental caries and periodontal disease, and preventing tooth loss, is likely important for maintaining chewing ability and facilitating consumption of a healthy diet.

Impact of Tooth Replacement Strategies on the Nutritional Status of Partially-Dentate Elders

Indirect evidence that functional impairment associated with tooth loss may be a determinant of dietary changes also comes from studies comparing different modalities of replacements for missing teeth. In a study of denture wearers in Quebec, those that wore dentures providing poor masticatory performance consumed significantly less fruits and vegetables than those with dentures that provided good masticatory performance [111]. Likewise in Swedish older adults, poorly fitting upper dentures were associated with decreased intake of vitamin C [112]. Among older Australians, women who reported poorly fitting dentures consumed greater amounts of sweets and dessert items [113].

Studies have also examined dietary differences among edentulous subjects with and without dentures. Perhaps not surprisingly, edentulous subjects without dentures usually consume more mashed food. Whether the placement of dentures in an edentulous patient makes a substantial improvement in the patient's intake remains unclear. In the only available randomized controlled trial (RCT) among patients with partial tooth loss, no differences were found in dietary intakes between patients who received either no dentures, fixed partial dentures, or removable partial dentures [114]. Sebring et al. [115] studied the effect of conventional maxillary and implant-supported or conventional mandibular dentures on patients who were edentulous with no prostheses. In both groups, calorie intake decreased; percentage of calories from fat also decreased significantly over the subsequent 3 years. Lindquist [116] evaluated the impact of prosthetic rehabilitation, using optimized complete dentures and then tissue-integrated mandibular fixed prostheses on 64 dissatisfied complete denture wearers. There was no change in diet after optimizing complete dentures, but there was a persistent increase in fresh fruit consumption with the prostheses. Prosthetic relining in addition to dietary counseling in the edentulous found an improvement in chewing ability and fiber intake from fruits and vegetables [117]. However, because there was no group that did not receive dietary counseling, it was not possible to separate the effect of relining from the counseling. More recently, a RCT to evaluate a conventional treatment using removable partial dentures versus functionally orientated treatment based on the shortened dental arch did not find differences in the nutritional assessment tests performed between groups [118]. A RCT comparing dietary intake in subjects with implant-retained overdentures and conventionally relined denture found that among those dissatisfied with current complete mandibular dentures, no significant differences regarding nutrient intake of food choices were found between the two treatments. In summary, dietary quality may improve with the placement of dental prostheses, but the changes are not substantial. Dietary counseling at the same time prostheses are fitted may assist patients in behavioral change and in optimizing the impact of their new chewing capabilities.

Problem	Management
Xerostomia, which could lead to other problems such as caries	Sugar-free mints and gums, artificial saliva, increased intake of water, or induce salivation medically by pilocarpine hydrochloride
Increased caries susceptibility	Instruction on oral hygiene and avoidance of food high in sugar, dental referral, daily fluoride gel
Trismus makes chewing difficult	Recommend appropriate jaw exercises
Dysphagia	Assess swallowing ability and risk of aspiration, monitor feeding capabilities, modify food consistency as indicated, use alternative route of nutritional support if necessary
Risk of aspiration	Use airway protection techniques and use of feeding devices as indicated
Malnutrition	Obtain a dietary consultation. Consider a multivitamin/mineral supplement and/or enteral or parenteral routes for patients who cannot meet their nutritional needs by mouth

Table 5.1 Problem management in oral cancer patients

Impact of Oral Cancer on Nutrition

Oral cancer (OC) has a major impact on eating and swallowing. The location or progression of the tumor itself and the side effects of treatment hamper feeding and swallowing. Side effects of radiation, primarily a result of damage to the salivary glands and reduction in saliva production, include xerostomia, dental caries, oral mucositis, and bacterial and fungal infections. Side effects of chemotherapy include mucositis, fungal infections, xerostomia, throat and mouth pain, taste changes, food aversions, nausea, and diarrhea. Other complications include aspiration, osteoradionecrosis, and trismus [119]. Side effects of surgery vary according to location and extent of surgery. The oral phase of swallowing is affected by surgical resection. Means to counter common problems faced by OC patients are listed in Table 5.1. These interventions can improve nutritional intake and overall quality of life.

Impact of Xerostomia on Nutrition

Xerostomia is associated with increased risk of dental caries, periodontal disease, speaking problems, swallowing problems, reduced chewing ability that contributes to malnutrition, and taste disturbances. With xerostomia, individuals may have inadequate lubrication and moisture in the mouth to chew food and create an adequate food bolus for swallowing. In addition, xerostomia may contribute to altered taste perception and to food sticking to the tongue or hard palate. Three studies of xerostomia have found that diet/nutrition and the quality of saliva were affected by exposure to Sjögren's syndrome (an immunologic disorder in which the body's immune system mistakenly attacks its own moisture producing glands) and xerogenic medications. In patients with Sjögren's syndrome, those with xerostomia have significantly lower caloric and micronutrient intakes [120]. Also, individuals with xerostomia were more likely to avoid crunchy vegetables (e.g., carrots), dry foods (e.g., bread), and sticky foods (e.g., peanut butter) [121] and have significantly lower intakes of energy, protein, fiber, vitamin A, C, and B₆, thiamin, riboflavin, calcium, and iron [122]. These studies suggest that xerostomia impairs optimal nutrient intake; however, these studies are hampered by their small size and cross-sectional design. These studies also noted that BMI was significantly lower in those with xerostomia compared to controls, as well as possibly lower triceps, skin fold thickness, and arm circumference.

Conclusion

Oral conditions that affect and are affected by nutrition, including dental caries, periodontal disease, xerostomia, and oral cancer, are more common in older adults. The causal role of dietary behaviors in the pathogenesis of dental caries throughout the life has been unequivocally demonstrated. Avoidance of in-between meal snacks, especially those high in refined carbohydrates, use of sugar-free candy or gum, and consumption of carbohydrates with meals and water, can reduce caries incidence. On the other hand, a higher meal frequency has metabolic benefits such as improving control of diabetes [123]. Hence, in order to reduce caries risk without increasing metabolic risk, it may be best to avoid refined carbohydrates, have more frequent meals and to brush after meals and snacks. Effects of nutritional factors on periodontitis risk and oral cancer risk on the one hand, and dietary effects of tooth loss on the other hand are highly plausible; however, evidence from well-designed longitudinal or intervention studies is limited.

Nutritional modulation of immune function, as, for example, through the use of antioxidants, may reduce progression of periodontal disease, but intervention studies are limited. Many epidemiologic studies have demonstrated the protective effect of fruits and vegetables and antioxidants on oral cancer risk. Studies suggest tooth loss impacts dietary quality and nutrient intake in a manner that may increase risk for several systemic diseases. Further, impaired dentition may contribute to weight change, depending on age and other population characteristics. Prevention of oral conditions and maintaining a healthy dentition is important for maintaining healthy nutrition status. Attention to dietary quality is particularly important among individuals with chewing disability from tooth loss or edentulism. Patients with oral cancer experience numerous complications that increase their risk for poor dietary intake. Close attention should be given to prevention of caries in patients with xerostomia, modification of food consistency in patients with dysphagia, and alternative feeding routes if nutritional needs cannot be met orally.

Recommendations

- To help in maintaining a healthy diet, oral health and public health professionals should focus on preventing and managing caries and periodontal disease, as well as in retaining teeth, by improving self-care, professional care, and modifying preventable risk factors.
- Clinicians should advise their dentate patients to restrict between-meal snacks that are high in sugar and to prefer low-sugar snacks such as low-sugar dairy products, fresh fruits, and nuts. Also, to eat carbohydrates with meals, and limit sugar and other foods that are cariogenic.
- Consumption of a diet rich in a diversity of fruits and vegetables appears to reduce the risk for the development of oral cancer.
- 4. Patients with tooth loss are at increased risk for poor/inappropriate dietary intake. Clinicians should counsel patients regarding ways to maintain good nutrition and minimize softer calorie-dense foods with low nutritional value. Pureed or shredded fruits and vegetables may serve as a means of ensuring adequate intakes of these food groups. A multivitamin should be considered in this group as well.
- Patients with oral cancer and radiation-induced xerostomia should be counseled to use sugar-free mints and gums and routinely apply fluoride to teeth to prevent dental caries.

References

- Swoboda J, Kiyak HA, Persson RE, Persson GR, Yamaguchi DK, MacEntee MI, et al. Predictors of oral health quality of life in older adults. Spec Care Dentist. 2006;26(4):137–44.
- Tsakos G, Steele JG, Marcenes W, Walls AW, Sheiham A. Clinical correlates of oral health-related quality of life: evidence from a national sample of British older people. Eur J Oral Sci. 2006;114(5):391–5.
- Cunha-Cruz J, Hujoel PP, Kressin NR. Oral health-related quality of life of periodontal patients. J Periodontal Res. 2007;42(2):169–76.
- Beltran-Aguilar ED, Barker LK, Canto MT, Dye BA, Gooch BF, Griffin SO, et al. Surveillance for dental caries, dental sealants, tooth retention, edentulism, and enamel fluorosis: United States, 1988–1994 and 1999–2002. MMWR Surveill Summ. 2005;54(3):1–43.
- 5. Dye BA, Li X, Thorton-Evans G. Oral health disparities as determined by selected healthy people 2020 oral health objectives for the United States, 2009–2010. NCHS Data Brief. 2012;104:1–8.
- 6. Eke EA, Dye BA, Wei L, Thornton-Evans GO, Genco RJ. CDC periodontal disease surveillance workgroup: James Beck (University of North Carolina, Chapel Hill, USA), Gordon Douglass (Past President, American Academy of Periodontology), Roy Page (University of Washington). Prevalence of periodontitis in adults in the United States: 2009 and 2010. J Dent Res. 2012;91(10):914–20.
- 7. Wu B, Liang J, Plassman BL, Remle RC, Bai L. Oral health among white, black, and Mexican-American elders: an examination of edentulism and dental caries. J Public Health Dent. 2011;71(4):308–17.
- Jemal A, Siegel R, Ward E, Murray T, Xu J, Thun MJ. Cancer statistics, 2007. CA Cancer J Clin. 2007; 57(1):43–66.
- 9. Siegel R, Naishadham D, Jemal A. Cancer statistics, 2012. CA Cancer J Clin. 2012;62(1):10-29.
- National Cancer Institute. SEER cancer statistics review, 1975–2009 (Vintage 2009 Populations). 2012; Available at: http://seer.cancer.gov/csr/1975_2009_pops09.
- 11. Zero DT. Sugars: the arch criminal? Caries Res. 2004;38(3):277-85.
- 12. Mandel ID. Calculus update: prevalence, pathogenicity and prevention. J Am Dent Assoc. 1995;126(5):573-80.
- Burt BA, Eklund SA, Morgan KJ, Larkin FE, Guire KE, Brown LO, et al. The effects of sugars intake and frequency of ingestion on dental caries increment in a three-year longitudinal study. J Dent Res. 1988;67(11): 1422–9.
- Lim S, Sohn W, Burt BA, Sandretto AM, Kolker JL, Marshall TA, et al. Cariogenicity of soft drinks, milk and fruit juice in low-income african-american children: a longitudinal study. J Am Dent Assoc. 2008;139(7):959–67. Quiz 995.
- 15. van Loveren C, Duggal MS. The role of diet in caries prevention. Int Dent J. 2001;51(6 Suppl 1):399-406.
- Alfin-Slater RB, Pi-Sunyer FX. Sugar and sugar substitutes. Comparisons and indications. Postgrad Med. 1987;82(2):46–50. 53–6.
- 17. Van Loveren C. Sugar alcohols: what is the evidence for caries-preventive and caries-therapeutic effects? Caries Res. 2004;38(3):286–93.
- Tanaka K, Miyake Y, Sasaki S. Intake of dairy products and the prevalence of dental caries in young children. J Dent. 2010;38(7):579–83.
- 19. Llena C, Forner L. Dietary habits in a child population in relation to caries experience. Caries Res. 2008;42(5):387–93.
- Yoshihara A, Watanabe R, Hanada N, Miyazaki H. A longitudinal study of the relationship between diet intake and dental caries and periodontal disease in elderly Japanese subjects. Gerodontology. 2009;26(2):130–6.
- Jimenez M, Hu FB, Marino M, Li Y, Joshipura KJ. Type 2 diabetes mellitus and 20 year incidence of periodontitis and tooth loss. Diabetes Res Clin Pract. 2012;98:494–500.
- Dietrich T, Maserejian NN, Joshipura KJ, Krall EA, Garcia RI. Tobacco use and incidence of tooth loss among US male health professionals. J Dent Res. 2007;86(4):373–7.
- Kobayashi T, Okano T, Shida S, Okada K, Suginohara T, Nakao H, et al. Variation of 25-hydroxyvitamin D3 and 25-hydroxyvitamin D2 levels in human plasma obtained from 758 Japanese healthy subjects. J Nutr Sci Vitaminol (Tokyo). 1983;29(3):271–81.
- Kobayashi T, Nagata T, Murakami S, Takashiba S, Kurihara H, Izumi Y, et al. Genetic risk factors for periodontitis in a Japanese population. J Dent Res. 2009;88(12):1137–41.
- Nishida M, Grossi SG, Dunford RG, Ho AW, Trevisan M, Genco RJ. Calcium and the risk for periodontal disease. J Periodontol. 2000;71(7):1057–66.
- Al-Zahrani MS. Increased intake of dairy products is related to lower periodontitis prevalence. J Periodontol. 2006;77(2):289–94.
- Shimazaki Y, Shirota T, Uchida K, Yonemoto K, Kiyohara Y, Iida M, et al. Intake of dairy products and periodontal disease: the Hisayama Study. J Periodontol. 2008;79(1):131–7.

- Adegboye AR, Christensen LB, Holm-Pedersen P, Avlund K, Boucher BJ, Heitmann BL. Intake of dairy products in relation to periodontitis in older danish adults. Nutrients. 2012;4(9):1219–29.
- 29. Krall EA. The periodontal-systemic connection: implications for treatment of patients with osteoporosis and periodontal disease. Ann Periodontol. 2001;6(1):209–13.
- Bischoff-Ferrari HA, Dietrich T, Orav EJ, Dawson-Hughes B. Positive association between 25-hydroxy vitamin D levels and bone mineral density: a population-based study of younger and older adults. Am J Med. 2004;116(9): 634–9.
- Bischoff-Ferrari HA, Willett WC, Orav EJ, Lips P, Meunier PJ, Lyons RA, et al. A pooled analysis of vitamin D dose requirements for fracture prevention. N Engl J Med. 2012;367(1):40–9.
- 32. Dietrich T, Joshipura KJ, Dawson-Hughes B, Bischoff-Ferrari HA. Association between serum concentrations of 25-hydroxyvitamin D3 and periodontal disease in the US population. Am J Clin Nutr. 2004;80(1):108–13.
- Jimenez MC, Giovannucci E, Krall EA, Joshipura KJ, Dietrich T. Predicted vitamin d status and incidence of tooth loss and periodontitis. Public Health Nutr. 2013;17:844–52.
- Dietrich T, Nunn M, Dawson-Hughes B, Bischoff-Ferrari HA. Association between serum concentrations of 25-hydroxyvitamin D and gingival inflammation. Am J Clin Nutr. 2005;82(3):575–80.
- 35. van Schoor NM, Lips P. Worldwide vitamin D status. Best Pract Res Clin Endocrinol Metab. 2011;25(4): 671–80.
- Looker AC, Pfeiffer CM, Lacher DA, Schleicher RL, Picciano MF, Yetley EA. Serum 25-hydroxyvitamin D status of the US population: 1988–1994 compared with 2000–2004. Am J Clin Nutr. 2008;88(6):1519–27.
- Deng H, Liu F, Pan Y, Jin X, Wang H, Cao J. BsmI, TaqI, ApaI, and FokI polymorphisms in the vitamin D receptor gene and periodontitis: a meta-analysis of 15 studies including 1338 cases and 1302 controls. J Clin Periodontol. 2011;38(3):199–207.
- Chen LL, Li H, Zhang PP, Wang SM. Association between vitamin D receptor polymorphisms and periodontitis: a meta-analysis. J Periodontol. 2012;83(9):1095–103.
- Miley DD, Garcia MN, Hildebolt CF, Shannon WD, Couture RA, Anderson Spearie CL, et al. Cross-sectional study of vitamin D and calcium supplementation effects on chronic periodontitis. J Periodontol. 2009;80(9): 1433–9.
- Garcia MN, Hildebolt CF, Miley DD, Dixon DA, Couture RA, Spearie CL, et al. One-year effects of vitamin D and calcium supplementation on chronic periodontitis. J Periodontol. 2011;82(1):25–32.
- Leggott PJ, Robertson PB, Jacob RA, Zambon JJ, Walsh M, Armitage GC. Effects of ascorbic acid depletion and supplementation on periodontal health and subgingival microflora in humans. J Dent Res. 1991;70(12):1531–6.
- Nishida M, Grossi SG, Dunford RG, Ho AW, Trevisan M, Genco RJ. Dietary vitamin C and the risk for periodontal disease. J Periodontol. 2000;71(8):1215–23.
- Chapple IL, Milward MR, Dietrich T. The prevalence of inflammatory periodontitis is negatively associated with serum antioxidant concentrations. J Nutr. 2007;137(3):657–64.
- 44. Iwasaki M, Moynihan P, Manz MC, Taylor GW, Yoshihara A, Muramatsu K, et al. Dietary antioxidants and periodontal disease in community-based older Japanese: a 2-year follow-up study. Public Health Nutr. 2013;16(2): 330–8.
- Linden GJ, McClean KM, Woodside JV, Patterson CC, Evans A, Young IS, et al. Antioxidants and periodontitis in 60–70-year-old men. J Clin Periodontol. 2009;36(10):843–9.
- Kesavalu L, Vasudevan B, Raghu B, Browning E, Dawson D, Novak JM, et al. Omega-3 fatty acid effect on alveolar bone loss in rats. J Dent Res. 2006;85(7):648–52.
- Naqvi AZ, Buettner C, Phillips RS, Davis RB, Mukamal KJ. n-3 fatty acids and periodontitis in US adults. J Am Diet Assoc. 2010;110(11):1669–75.
- 48. Merchant AT, Pitiphat W, Franz M, Joshipura KJ. Whole-grain and fiber intakes and periodontitis risk in men. Am J Clin Nutr. 2006;83(6):1395–400.
- Pischon N, Heng N, Bernimoulin JP, Kleber BM, Willich SN, Pischon T. Obesity, inflammation, and periodontal disease. J Dent Res. 2007;86(5):400–9.
- Ritchie CS, Joshipura K, Silliman RA, Miller B, Douglas CW. Oral health problems and significant weight loss among community-dwelling older adults. J Gerontol A Biol Sci Med Sci. 2000;55(7):M366–71.
- Michaud DS, Liu Y, Meyer M, Giovannucci E, Joshipura K. Periodontal disease, tooth loss, and cancer risk in male health professionals: a prospective cohort study. Lancet Oncol. 2008;9(6):550–8.
- 52. Chaffee BW, Weston SJ. Association between chronic periodontal disease and obesity: a systematic review and meta-analysis. J Periodontol. 2010;81(12):1708–24.
- Jimenez M, Hu FB, Marino M, Li Y, Joshipura KJ. Prospective associations between measures of adiposity and periodontal disease. Obesity (Silver Spring). 2012;20(8):1718–25.
- Gorman A, Kaye EK, Apovian C, Fung TT, Nunn M, Garcia RI. Overweight and obesity predict time to periodontal disease progression in men. J Clin Periodontol. 2012;39(2):107–14.
- 55. Morse DE, Pendrys DG, Katz RV, Holford TR, Krutchkoff DJ, Eisenberg E, et al. Food group intake and the risk of oral epithelial dysplasia in a United States population. Cancer Causes Control. 2000;11(8):713–20.

- 56. Maserejian NN, Giovannucci E, Rosner B, Zavras A, Joshipura K. Prospective study of fruits and vegetables and risk of oral premalignant lesions in men. Am J Epidemiol. 2006;164(6):556–66.
- Gupta PC, Hebert JR, Bhonsle RB, Sinor PN, Mehta H, Mehta FS. Dietary factors in oral leukoplakia and submucous fibrosis in a population-based case control study in Gujarat. India Oral Dis. 1998;4(3):200–6.
- 58. Garavello W, Lucenteforte E, Bosetti C, La Vecchia C. The role of foods and nutrients on oral and pharyngeal cancer risk. Minerva Stomatol. 2009;58(1–2):25–34.
- 59. Edefonti V, Bravi F, La Vecchia C, Randi G, Ferraroni M, Garavello W, et al. Nutrient-based dietary patterns and the risk of oral and pharyngeal cancer. Oral Oncol. 2010;46(5):343–8.
- Lucenteforte E, Garavello W, Bosetti C, La Vecchia C. Dietary factors and oral and pharyngeal cancer risk. Oral Oncol. 2009;45(6):461–7.
- McLaughlin JK, Gridley G, Block G, Winn DM, Preston-Martin S, Schoenberg JB, et al. Dietary factors in oral and pharyngeal cancer. J Natl Cancer Inst. 1988;80(15):1237–43.
- Steinmetz KA, Potter JD. Vegetables, fruit, and cancer prevention: a review. J Am Diet Assoc. 1996;96(10): 1027–39.
- Kasum CM, Jacobs Jr DR, Nicodemus K, Folsom AR. Dietary risk factors for upper aerodigestive tract cancers. Int J Cancer. 2002;99(2):267–72.
- 64. Hirayama T. A large scale cohort study on cancer risks by diet: with special reference to the risk reducing effects of green-yellow vegetable consumption. Princess Takamatsu Symp. 1985;16:41–53.
- 65. De Stefani E, Boffetta P, Oreggia F, Brennan P, Ronco A, Deneo-Pellegrini H, et al. Plant foods and risk of laryngeal cancer: a case–control study in Uruguay. Int J Cancer. 2000;87(1):129–32.
- 66. Link LB, Potter JD. Raw versus cooked vegetables and cancer risk. Cancer Epidemiol Biomarkers Prev. 2004;13(9):1422–35.
- Takezaki T, Hirose K, Inoue M, Hamajima N, Kuroishi T, Nakamura S, et al. Tobacco, alcohol and dietary factors associated with the risk of oral cancer among Japanese. Jpn J Cancer Res. 1996;87(6):555–62.
- 68. Tavani A, Gallus S, La Vecchia C, Talamini R, Barbone F, Herrero R, et al. Diet and risk of oral and pharyngeal cancer. An Italian case–control study. Eur J Cancer Prev. 2001;10(2):191–5.
- 69. Nagao T, Ikeda N, Warnakulasuriya S, Fukano H, Yuasa H, Yano M, et al. Serum antioxidant micronutrients and the risk of oral leukoplakia among Japanese. Oral Oncol. 2000;36(5):466–70.
- Negri E, Franceschi S, Bosetti C, Levi F, Conti E, Parpinel M, et al. Selected micronutrients and oral and pharyngeal cancer. Int J Cancer. 2000;86(1):122–7.
- Maserejian NN, Giovannucci E, Rosner B, Joshipura K. Prospective study of vitamins C, E, and A and carotenoids and risk of oral premalignant lesions in men. Int J Cancer. 2007;120(5):970–7.
- 72. Li Q, Chuang SC, Eluf-Neto J, Menezes A, Matos E, Koifman S, et al. Vitamin or mineral supplement intake and the risk of head and neck cancer: pooled analysis in the INHANCE consortium. Int J Cancer. 2012;131(7): 1686–99.
- 73. Zain RB. Cultural and dietary risk factors of oral cancer and precancer: a brief overview. Oral Oncol. 2001;37(3):205–10.
- 74. Gorsky M, Epstein JB. The effect of retinoids on premalignant oral lesions: focus on topical therapy. Cancer. 2002;95(6):1258-64.
- 75. Garewal HS, Katz RV, Meyskens F, Pitcock J, Morse D, Friedman S, et al. Beta-carotene produces sustained remissions in patients with oral leukoplakia: results of a multicenter prospective trial. Arch Otolaryngol Head Neck Surg. 1999;125(12):1305–10.
- World Cancer Research Fund, American Institute for Cancer Research. Food, nutrition, physical activity, and the prevention of cancer: a global perspective. Washington, DC: AICR; 2007.
- 77. Lu R, Dan H, Wu R, Meng W, Liu N, Jin X, et al. Lycopene: features and potential significance in the oral cancer and precancerous lesions. J Oral Pathol Med. 2011;40(5):361–8.
- Shanmugham JR, Zavras AI, Rosner BA, Giovannucci EL. Alcohol-folate interactions in the risk of oral cancer in women: a prospective cohort study. Cancer Epidemiol Biomarkers Prev. 2010;19(10):2516–24.
- 79. Matsuo K, Rossi M, Negri E, Oze I, Hosono S, Ito H, et al. Folate, alcohol, and aldehyde dehydrogenase 2 polymorphism and the risk of oral and pharyngeal cancer in Japanese. Eur J Cancer Prev. 2012;21(2):193–8.
- Flagg EW, Coates RJ, Jones DP, Byers TE, Greenberg RS, Gridley G, et al. Dietary glutathione intake and the risk of oral and pharyngeal cancer. Am J Epidemiol. 1994;139(5):453–65.
- Lam LT, Yang L. Overweight/obesity and attention deficit and hyperactivity disorder tendency among adolescents in China. Int J Obes (Lond). 2007;31(4):584–90.
- 82. Lam TK, Cross AJ, Freedman N, Park Y, Hollenbeck AR, Schatzkin A, et al. Dietary fiber and grain consumption in relation to head and neck cancer in the NIH-AARP diet and health study. Cancer Causes Control. 2011; 22(10):1405–14.
- Badawi AF, Hosny G, el-Hadary M, Mostafa MH. Salivary nitrate, nitrite and nitrate reductase activity in relation to risk of oral cancer in Egypt. Dis Markers. 1998;14(2):91–7.

- Gridley G, McLaughlin JK, Block G, Blot WJ, Winn DM, Greenberg RS, et al. Diet and oral and pharyngeal cancer among blacks. Nutr Cancer. 1990;14(3–4):219–25.
- Gupta PC, Hebert JR, Bhonsle RB, Murti PR, Mehta H, Mehta FS. Influence of dietary factors on oral precancerous lesions in a population-based case–control study in Kerala. India Cancer. 1999;85(9):1885–93.
- Amtha R, Zain R, Razak IA, Basuki B, Roeslan BO, Gautama W, et al. Dietary patterns and risk of oral cancer: a factor analysis study of a population in Jakarta. Indonesia Oral Oncol. 2009;45(8):e49–53.
- Aune D, De Stefani E, Ronco A, Boffetta P, Deneo-Pellegrini H, Acosta G, et al. Meat consumption and cancer risk: a case–control study in Uruguay. Asian Pac J Cancer Prev. 2009;10(3):429–36.
- Marchioni DM, Fisberg RM, Francisco de Gois Filho J, Kowalski LP, Brasilino de Carvalho M, Abrahao M, et al. Dietary patterns and risk of oral cancer: a case–control study in Sao Paulo, Brazil. Rev Saude Publica. 2007;41(1): 19–26.
- Edefonti V, Hashibe M, Ambrogi F, Parpinel M, Bravi F, Talamini R, et al. Nutrient-based dietary patterns and the risk of head and neck cancer: a pooled analysis in the International head and neck cancer epidemiology consortium. Ann Oncol. 2012;23(7):1869–80.
- Helen-Ng LC, Razak IA, Ghani WM, Marhazlinda J, Norain AT, Raja Jallaludin RL, et al. Dietary pattern and oral cancer risk: a factor analysis study. Community Dent Oral Epidemiol. 2012;40(6):560–6.
- Ritchie CS, Joshipura K, Hung HC, Douglass CW. Nutrition as a mediator in the relation between oral and systemic disease: associations between specific measures of adult oral health and nutrition outcomes. Crit Rev Oral Biol Med. 2002;13(3):291–300.
- Joshipura KJ, Willett WC, Douglass CW. The impact of edentulousness on food and nutrient intake. J Am Dent Assoc. 1996;127(4):459–67.
- Akpata E, Otoh E, Enwonwu C, Adeleke O, Joshipura K. Tooth loss, chewing habits, and food choices among older Nigerians in Plateau State: a preliminary study. Community Dent Oral Epidemiol. 2011;39(5):409–15.
- 94. Tsakos G, Herrick K, Sheiham A, Watt RG. Edentulism and fruit and vegetable intake in low-income adults. J Dent Res. 2010;89(5):462–7.
- 95. Ervin RB, Dye BA. The effect of functional dentition on healthy eating index scores and nutrient intakes in a nationally representative sample of older adults. J Public Health Dent. 2009;69(4):207–16.
- 96. Sheiham A, Steele JG, Marcenes W, Lowe C, Finch S, Bates CJ, et al. The relationship among dental status, nutrient intake, and nutritional status in older people. J Dent Res. 2001;80(2):408–13.
- 97. Cousson PY, Bessadet M, Nicolas E, Veyrune JL, Lesourd B, Lassauzay C. Nutritional status, dietary intake and oral quality of life in elderly complete denture wearers. Gerodontology. 2012;29(2):e685–92.
- Joshipura K, Ritchie C, Douglass C. Strength of evidence linking oral conditions and systemic disease. Compend Contin Educ Dent Suppl. 2000;30:12–23. Quiz 65.
- 99. Joshipura KJ. How can tooth loss affect diet and health, and what nutritional advice would you give to a patient scheduled for extractions? J Can Dent Assoc. 2005;71:421–2.
- Krall E, Hayes C, Garcia R. How dentition status and masticatory function affect nutrient intake. J Am Dent Assoc. 1998;129(9):1261–9.
- Papas AS, Palmer CA, Rounds MC, Russell RM. The effects of denture status on nutrition. Spec Care Dentist. 1998;18(1):17–25.
- 102. Yoshida M, Kikutani T, Yoshikawa M, Tsuga K, Kimura M, Akagawa Y. Correlation between dental and nutritional status in community-dwelling elderly Japanese. Geriatr Gerontol Int. 2011;11(3):315–9.
- 103. Hung HC, Willett W, Ascherio A, Rosner BA, Rimm E, Joshipura KJ. Tooth loss and dietary intake. J Am Dent Assoc. 2003;134(9):1185–92.
- 104. Hung HC, Colditz G, Joshipura KJ. The association between tooth loss and the self-reported intake of selected CVD-related nutrients and foods among US women. Community Dent Oral Epidemiol. 2005;33(3):167–73.
- 105. Mojon P, Budtz-Jorgensen E, Rapin CH. Relationship between oral health and nutrition in very old people. Age Ageing. 1999;28(5):463–8.
- 106. Hirano H, Ishiyama N, Watanabe I, Nasu I. Masticatory ability in relation to oral status and general health on aging. J Nutr Health Aging. 1999;3(1):48–52.
- 107. Johansson I, Tidehag P, Lundberg V, Hallmans G. Dental status, diet and cardiovascular risk factors in middle-aged people in northern Sweden. Community Dent Oral Epidemiol. 1994;22(6):431–6.
- 108. Hilgert JB, Hugo FN, de Sousa Mda L, Bozzetti MC. Oral status and its association with obesity in Southern Brazilian older people. Gerodontology. 2009;26(1):46–52.
- Osterberg T, Dey DK, Sundh V, Carlsson GE, Jansson JO, Mellstrom D. Edentulism associated with obesity: a study of four national surveys of 16, 416 Swedes aged 55–84 years. Acta Odontol Scand. 2010;68(6):360–7.
- Joshipura KJ, Ritchie C. Can the relation between tooth loss and chronic disease be explained by socio-economic status? Eur J Epidemiol. 2005;20(3):203–4.
- Laurin D, Brodeur JM, Bourdages J, Vallee R, Lachapelle D. Fibre intake in elderly individuals with poor masticatory performance. J Can Dent Assoc. 1994;60(5):443–6. 449.

- 112. Nordstrom G. The impact of socio-medical factors and oral status on dietary intake in the eighth decade of life. Aging (Milano). 1990;2(4):371–85.
- 113. Horwath CC. Chewing difficulty and dietary intake in the elderly. J Nutr Elder. 1989;9(2):17–24.
- 114. Garrett NR, Kapur KK, Hasse AL, Dent RJ. Veterans administration cooperative dental implant study: comparisons between fixed partial dentures supported by blade-vent implants and removable partial dentures. Part V: comparisons of pretreatment and posttreatment dietary intakes. J Prosthet Dent. 1997;77(2):153–61.
- 115. Sebring NG, Guckes AD, Li SH, McCarthy GR. Nutritional adequacy of reported intake of edentulous subjects treated with new conventional or implant-supported mandibular dentures. J Prosthet Dent. 1995;74(4):358–63.
- 116. Lindquist LW. Prosthetic rehabilitation of the edentulous mandible. A longitudinal study of treatment with tissueintegrated fixed prostheses. Swed Dent J Suppl. 1987;48:1–39.
- 117. Olivier M, Laurin D, Brodeur JM, Boivin M, Leduc N, Levy M, et al. Prosthetic relining and dietary counselling in elderly women. J Can Dent Assoc. 1995;61(10):882–6.
- McKenna G, Allen PF, Flynn A, O'Mahony D, DaMata C, Cronin M, et al. Impact of tooth replacement strategies on the nutritional status of partially-dentate elders. Gerodontology. 2012;29(2):e883–90.
- Minasian A, Dwyer JT. Nutritional implications of dental and swallowing issues in head and neck cancer. Oncology (Williston Park). 1998;12(8):1155–62. Discussion 1162–9.
- 120. Rhodus NL. Qualitative nutritional intake analysis of older adults with Sjogren's syndrome. Gerodontology. 1988;7(2):61–9.
- 121. Loesche WJ, Abrams J, Terpenning MS, Bretz WA, Dominguez BL, Grossman NS, et al. Dental findings in geriatric populations with diverse medical backgrounds. Oral Surg Oral Med Oral Pathol Oral Radiol Endod. 1995;80(1):43–54.
- 122. Rhodus NL, Brown J. The association of xerostomia and inadequate intake in older adults. J Am Diet Assoc. 1990;90(12):1688–92.
- 123. Bertelsen J, Christiansen C, Thomsen C, Poulsen PL, Vestergaard S, Steinov A. Effect of meal frequency on blood glucose, insulin, and free fatty acids in NIDDM subjects. Diabetes Care. 1993;16(1):4–7.

Chapter 6 Loss of Muscle Mass and Muscle Strength in Obese and Nonobese Older Adults

Danielle R. Bouchard and Ian Janssen

Key Points

- *Sarcopenia* refers to the process of age-related skeletal muscle mass loss, and *dynapenia* refers to the process of age-related muscle strength loss. These terms are often used to describe older adults with unhealthy skeletal muscle mass and muscle strength levels, respectively.
- Well-designed longitudinal studies show little or no association between sarcopenia and the risk of impaired physical function, falls, and mortality.
- Conversely, dynapenia is consistently associated with these outcomes.
- The combination of sarcopenia or dynapenia with obesity among older adults (sarcopenic obesity and dynapenic obesity) is highly prevalent. The best treatment for these conditions is unknown, as weight loss is associated with both muscle mass and muscle strength loss.
- Physical activity, particularly resistance exercise, is one of the most promising approaches for preventing and treating both sarcopenia and dynapenia in obese and nonobese older adults.

Keywords Skeletal muscle • Muscle strength • Obesity • Physical function • Resistance exercise • Dietary supplement

Introduction

The terminology used to describe age-related loss of muscle mass and muscle strength has evolved since 1989 (Fig. 6.1). In 1989, Irwin Rosenberg coined the term *sarcopenia* to refer to the process of age-related skeletal muscle loss [1]. Sarcopenia comes from the Greek words *sarx* (flesh) and *penia* (loss).

D.R. Bouchard, Ph.D. (🖂)

Department of Kinesiology and Recreation Management, University of Manitoba, 318 Max Bell Centre, Winnipeg, MB, Canada R3T 2N2 e-mail: Danielle.Bouchard@ad.umanitoba.ca

I. Janssen, Ph.D. Department of Community Health and Epidemiology, Queen's University, Kingston, ON, Canada K7L3N6 e-mail: ian.janssen@queensu.ca



Fig. 6.1 Terms related to loss of muscle mass and muscle strength

The term *sarcopenic obesity* was introduced in 2000 to identify older obese adults that also presented with a low muscle mass [2]. Over time, the definition of sarcopenia as purely a loss in muscle mass evolved into one that included both a loss in muscle mass and muscle function [3, 4]. However, as discussed in more detail later on, the loss in muscle strength is not coupled to the loss in muscle mass. The term *dynapenia* was coined to refer specifically to the process of losing muscle strength with advancing age [5]. The term *dynapenia* is a translation from the Greek—poverty of strength, power, or force. In this chapter, sarcopenia (loss in muscle mass) and dynapenia (loss in muscle strength) will be considered as separate phenomenon. Finally, in 2009, *dynapenic obesity* was coined to identify obese older adults with low muscle strength [6]. This chapter discusses the prevalence, consequences, and possible prevention/treatment strategies of sarcopenia, dynapenia, sarcopenic obesity, and dynapenic obesity.

Sarcopenia

With few exceptions, cross-sectional studies have observed that muscle size remains relatively constant in the third and fourth decades, and then starts to decline noticeably at around the fifth decade [7, 8]. Longitudinal studies, with follow-up lengths of up to 12 years, have confirmed these observations [9, 10]. It has been estimated that skeletal muscle mass decreases by approximately 6 % per decade after mid-life [11].

Definition and Prevalence

Although all older adults have lost some muscle [12], there is a considerable variation in the amount of muscle within the older population. Older adults who had a high peak muscle mass and have lost muscle at a slow rate have reasonably healthy muscle mass values. Others may have had a low peak muscle mass and/or lost muscle quickly as they aged, and subsequently have a very low muscle mass. Many epidemiological studies have tried to identify this later group. Specifically, cut-points have

been used to identify which older adults have skeletal muscle values in the normal and unhealthy (i.e., sarcopenic) ranges. In 1998, Baumgartner and colleagues [13] proposed a process to determine which older persons have unhealthy values. Specifically, they determined the average and standard deviation of the height-adjusted appendicular skeletal mass (muscle mass of legs + arms/height²) of a young and healthy reference population. Sarcopenia was then diagnosed if an older adult had an appendicular skeletal muscle mass value equal to or below two standard deviations of the mean of the young and health reference population. Using this approach, the prevalence of sarcopenia in participants in the New Mexico Elder Health Survey was 14 % in those aged 65–69 years, and over 50 % in those 80 years of age or older [13]. Since Baumgartner and colleagues [13], many other researchers have employed a similar approach for classifying sarcopenia [14, 15] and have reported a prevalence of 8.5 % [14] to 12.3 % [15] with populations aged 60 years and older. Besides this approach, another strategy has been employed to identify sarcopenia, namely using two standard deviations of skeletal muscle mass adjusted for body weight instead of height [16]. Using this approach, the prevalence of sarcopenia was similar for men (30.8 % vs. 29.5 %), but higher for women (30.3 % vs. 10.2 %), when compared with Baumgartner's approach [17].

Consequences of Sarcopenia

Functional Limitation and Physical Disability

The majority of the pre-2005 scientific literature on the health implications of sarcopenia focused on measures of physical function. These early studies were also cross-sectional in design [14] and, for the most part, demonstrated that sarcopenia is associated with functional limitation [13, 14]. These studies have used a variety of methods for measuring muscle size and for classifying sarcopenia. The strength of the association between sarcopenia and physical function in these cross-sectional studies would be considered moderate to strong by epidemiological standards. For instance, cross-sectional findings based on the Cardiovascular Health Study indicated that the likelihood of having physical disability at the start of the study (i.e., baseline exam) was 79 % greater in those with severe sarcopenia than in those with normal muscle mass.

More recently conducted longitudinal analyses have not supported the association observed in the initial cross-sectional studies. For example, in an 8-year follow-up of older adult participants in the Cardiovascular Health Study cohort, the risk of developing physical disability was 27 % greater in those with *severe* sarcopenia than in those with a normal muscle mass [18]. Thus, the effect of sarcopenia on disability risk was considerably smaller in the longitudinal analysis than in the cross-sectional analysis of the same cohort of participants, implying that the effects of sarcopenia on functional impairment and disability inferred from the cross-sectional studies published in the 1990s and early 2000s may have been overestimated. Other prospective studies [19, 20] have also reported no association between loss of muscle mass and functional limitation or physical disability.

Mortality

Studies of community-dwelling [21] and institutionalized [22] older adults have also shown that low muscle area in the upper arm is a good predictor of both short-term and long-term mortality risk. For example, in an 8-year follow-up of ~1,400 older Australians, Miller and colleagues [21] found that mortality risk was increased by twofold in those with an upper arm muscle area below ~21 cm². Studies have also used total fat-free mass (FFM) to predict mortality risk. In a landmark study of 57,053 adults aged 50–64 years, FFM, as measured with bioelectrical impedance, was associated with an ~25 % increased mortality risk over 6 years [23].

Metabolic Health

The metabolic effects of sarcopenia could be based on the fact that loss of muscle mass is associated with a decreased metabolic rate [24]. It has also been postulated that sarcopenia contributes to cardiometabolic diseases such as insulin resistance, type 2 diabetes [25], dyslipidemia, hypertension, and the metabolic syndrome [26]. However, the literature is mixed and in general does not support this postulation. For instance, a study of 22 obese postmenopausal women found that those with sarcopenia had a more favorable lipid and lipoprotein profile than those without sarcopenia [27]. Additional research on over 3,000 members of the Cardiovascular Health Study cohort found that sarcopenia per se was not a risk factor for the development of cardiovascular disease over an 8-year follow-up [28].

Dynapenia

A central belief of early research into the field of sarcopenia was that low muscle mass leads to low muscle strength [29], which in turn increases the risk of functional impairment and physical disability. However, the notion that a loss in muscle strength is tightly coupled to sarcopenia has been disproven. Longitudinal studies of older adults have reported that less than 5 % of the change in muscle strength is explained by the corresponding change in muscle mass [30]. Based on these and other findings, Clark and Manini proposed that the term sarcopenia be used to refer to the loss in muscle and that the term dynapenia be used to refer to the loss in muscle strength [31]. Muscle strength loss, or the process of dyanpenia, beings at around age 50 years, and proceeds at a rate of about 15 % per decade between 50 and 70, and is more accelerated in subsequent years [32].

Prevalence

Even if it is clear that older adults have a lower muscle strength than young people [33], the threshold or cut-point that denotes a critical loss in muscle strength is unclear. At present, researchers are using similar strategies to those that are being used to identify sarcopenia. In a recent study using a small sample (n=46) of healthy 50- to 75-year-old women, dynapenia was defined as muscle strength (measured with a handgrip dynamometer) equivalent to at least one standard deviation below the mean handgrip strength from a reference group made up of young adults [34]. Using this strategy, 50 % were dynapenic.

Consequences of Dynapenia

Functional Limitation and Physical Disability

Cross-sectional [19, 35] and longitudinal [19] studies have consistently reported that a low muscle strength, or dynapenia, is a risk factor for functional limitations and physical disability. For example, Visser and colleagues demonstrated there was an approximately 2.5 times greater risk of developing mobility limitations over 2.5 years in older adults in the lowest quartiles of muscle strength compared with the highest quartile [19]. Furthermore, Moreland and colleagues [36] reported, in a meta-analyses, that low muscle strength was associated with a 76 % increased risk of falling. Muscle strength is associated with functional limitations to a greater extent than muscle mass, implying that muscle function is more important than muscle quantity [37].

Mortality

Rantanen and colleagues reported that the risk of all-cause mortality was increased by 73 % in lowest grip strength tertile. Another study employing other measures of muscle strength (e.g., isokinetic quadriceps strength) made similar observations [38]. In addition, longitudinal studies have shown that rate of muscle strength decline is a strong predictor of mortality risk [39].

Metabolic Health

A comparison of lowest and highest muscle strength groups has shown low strength increases the risk of the metabolic syndrome [40], insulin resistance [41], and type 2 diabetes [42] by almost twofold. Furthermore, it has been reported that dynapenic postmenopausal women had a 24 % reduced maximum oxygen consumption [34].

Sarcopenic Obesity

Definition and Prevalence

The typical aging process is accompanied with an increase of body weight until about age 60 [43]. As a result, an increase in fat mass and a reduction in FFM occurs, but the changes in the two are not completely correlated [44]. As there is a lack of uniformity in the measurement and classification of sarcopenic obesity, the prevalence of this condition is unclear. The reported prevalence ranges from 2% [45] to 48% [46].

Consequences of Sarcopenic Obesity

Most of the literature examining the health implication of sarcopenic-obesity is based on crosssectional study designs. Some studies report that sarcopenic obese people are more likely to have an impaired physical function [13, 47, 48], while others report no differences in function for those with sarcopenic obesity by comparison to those with sarcopenia or obesity alone [49–51].

Messier and colleagues [51] reported that sarcopenic obese older women do not have a more unfavorable metabolic profile compared with non-sarcopenic obese women. Baumgartner and colleagues [47] showed that sarcopenic obesity was a predictor of disability and even a higher mortality rate [52].

In short, it is not clear if being identified as both sarcopenic and obese is more harmful compared with being sarcopenic or obese alone.

Dynapenic Obesity

Definition and Prevalence

In 2009, Bouchard and Janssen defined dynapenic obesity as individuals (in this study aged 55 years or older) who were in the lowest tertile of muscle strength (measured by leg extension strength) and the highest tertile of fat mass (measured by dual-energy X-ray absorptiometry [DEXA]) [6].

Of the 2,039 people studied, 9.2 % of the men and 8.8 % of the women were considered to be dynapenic obese. Recently, Senechal and colleagues (2012) used two standard deviations below the mean strength in young healthy adults to recruit dynapenic-obese women to participate in a randomized controlled study [53]. Obesity was defined as 35 % of fat mass measured by DEXA. While the *dynapenic obesity* term is relatively new, for several years studies have been reporting on the negative consequences of having a poor muscle strength combined with extra body weight/fat.

Consequences of Dynapenic Obesity

Dynapenic obesity is associated with more functional limitations and a greater risk of type 2 diabetes and cardiovascular disease compared with dynapenia alone or obesity alone. For example, Bouchard and Janssen (2009) [6] concluded that dynapenic obese people have 2 % lower physical capacity compared with those who are obese, 7 % lower compared with those who are dynapenic, and 13 % lower compared with those who are nonobese and non-dynapenic. In a 9-year follow-up of almost 5,000 older adults who participated in the Cardiovascular Health Studies, dynapenia alone (poor grip strength) and obesity alone (high BMI) were not sufficient to increase cardiovascular risk; however, dynapenic obesity was associated with a 25 % increased risk of developing cardiovascular disease [28]. Senechal and colleagues [54] recently studied the impact of dynapenic obesity on metabolic health. They reported a higher risk of metabolic syndrome and type 2 diabetes in dynapenic obese adults compared with healthy counterparts.

Prevention and Treatment

The best prevention and treatment strategy for all four of the conditions discussed in this chapter is physical activity. Nutritional strategies, hormonal strategies, and medications have been explored to address muscle mass, but these have not been as successful as resistance training [3, 55].

Prevention and Treatment for Sarcopenia and Dynapenia

Resistance Training

Although most types of physical activity have a positive effect on muscle strength and muscle mass within older adults [56], resistance training is by far the optimal strategy. Based on many intervention studies reported in review papers, interventions have typically ranged from 8 to 16 weeks, and the vast majority have noted a significant increase in skeletal muscle size and muscle strength irrespective of the population group studied [57]. In fact, progressive resistance training has been shown to be effective in increasing muscle strength in frail older adults [58]. Resistance exercise leads to an increase in strength greater than what would be expected for the accompanying increase in muscle mass [59].

Current recommendations of the American College of Sports Medicine are: "To maximize strength development, a resistance should be used that allows 10–15 repetitions for each exercise. The level of effort for muscle-strengthening activities should be moderate to high. On a 10-point scale, where no movement is 0, and maximal effort of a muscle group is 10, moderate-intensity effort is a 5 or 6 and high-intensity effort is a 7 or 8. Muscle strengthening activities include a progressive weight training program, weight bearing calisthenics, and similar resistance exercises that use the major muscle

groups." Older adults should perform at least one set of 10–12 repetitions for 8–10 exercises that train the major muscle groups, and exercises for each of the major muscle groups should occur on 2 or 3 days a week [60].

A growing body of literature reports that resistance exercise that restricts blood flow stimulates muscle protein synthesis to a greater extent than traditional resistance training [61–64]. Blood flow restriction exercise consists of restricting blood flow by applying a wrapping or a blocking device near the muscle being exercised. Studies so far suggest that this strategy would increase muscle mass compared with traditional resistance training in older adults [62].

Resistance Training + Protein Supplements

Controversy exists as to whether protein supplements combined with resistance exercise have a greater impact on muscle hypertrophy than resistance exercise alone. The dose of protein required to increase skeletal muscle hypertrophy is unclear as one of the studies that demonstrated a beneficial effect [65] used a smaller dose of protein (10 g vs. 15 g) than a study showing no effect [66]. Symons and colleagues concluded that ingesting more than 30 g of protein in a single meal did little or nothing to enhance muscle protein synthesis [67]. The timing of protein intake around the exercise may be relevant in determining whether or not additional muscle hypertrophy occurs. Esmarck and colleagues [65] reported that a liquid protein supplement (10 g protein) ingested immediately post-exercise in older men increased skeletal muscle hypertrophy, whereas ingesting the protein supplement 2 h post-exercise did not.

Whole-Body Vibration

The whole-body vibration apparatus is a new tool that proposes to improve muscle mass, muscle strength, and bone mass. To utilize the machine a person stands, sits, or lays on a vibrating platform. As the machine vibrates, it transmits energy to the body and is forcing muscles to contract and relax dozens of times each second to maintain balance. Most intervention studies that have tested this apparatus, with intervention of up to 18 months long, reported increase in muscle strength and cardiorespiratory fitness but not changes in muscle mass [68]. However, when whole-body vibration is embedded in a standard exercise program, it appears to provide only minor effects above those obtained by exercise alone [69].

Supplements

Protein Supplements

Essential amino acids and milk-based proteins, creatine monohydrate, essential fatty acids, and vitamin D may all have beneficial effects on aging muscles [70]. An initial prevention strategy that may reduce the typical progression of sarcopenia in older persons is to ensure that they consume adequate protein in their diet. Data suggests that the average protein intake is lower in older adults than in young adults [71]. Protein supplements can be considered in addition to an individual's ad libitum diet. Protein-based nutritional supplements are attractive given their low cost, ease of administration, and they are well tolerated in older adults [72]. When used appropriately, protein/nutritional supplements may result in an increase in daily energy and protein intake in older adults [72]. The oral ingestion of intact whey protein stimulates muscle protein synthesis in older adults [73]. In a recent study, Pennings and colleagues [74] demonstrated that whey protein stimulates more protein accretion than casein and casein hydrolysate, probably because of a faster digestion and greater leucine content. It is possible that the small increase in energy from a supplement of this nature would not be enough to decrease ad libitum food and protein consumption, as shown in other studies, which used protein/ nutritional supplements with a considerably larger caloric content [75]. Although protein supplement studies have demonstrated promising results on short-term protein synthesis within skeletal muscle, the long-term effectiveness of protein supplements on muscle mass has yet to be demonstrated.

Other Supplements

Experimental studies have shown that 8 weeks of an omega-3 fatty acid supplement increased muscle protein synthesis (from 0.009 ± 0.005 %/h above basal values to 0.031 ± 0.003 %/h above basal values) [76]. Bicarbonate supplements have also shown promising results when taken during a 3-month exercise program, leading to reduced net acid excretion and increased leg press power by 13 % in women; although no positive effects were seen in men [77]. Bicarbonate is well-tolerated and inexpensive.

Hormones

The age-related decrease in anabolic hormones such as growth hormone, estrogen, ghrelin, testosterone, and dehydroepiandosterone (DHEA) have been linked to the development of sarcopenia [3]. A meta-analysis of 11 randomized controlled trials reported that testosterone replacement therapy produced a moderate increase in muscle strength in older men [78]. Although growth hormone administration increases muscle mass and strength in people with hypopituitarism [79], most of the initial studies conducted in older adults found that growth hormone therapy did not increase muscle mass and muscle strength [3, 79]. However, alternative strategies for stimulating the growth hormone and insulin-like growth factor pathway have shown promising results [3, 79]. For example, Nass and colleagues [80] have shown that ghrelin supplementation was well tolerated and resulted in an increase in growth hormone and muscle mass. Meanwhile, replacement studies of estrogen [81] and DHEA [29] have shown poor results.

Prevention and Treatment for Sarcopenic Obesity and Dynapenic Obesity

There is an ongoing debate as to whether weight loss should be recommended in older adults because of a risk of accentuated muscle mass loss, a risk of not meeting the recommended nutrients, or an increase in mortality risk [82]. However, in the last decade most reviews and meta-analyses have concluded that weight reduction in obese older adults has more advantages than disadvantages [83, 84].

Aerobic Exercise

It is possible for obese older adults to reduce their body weight by doing aerobic exercise [85, 86]. However, in older adults and especially obese older adults with low muscle mass and or muscle strength, it might be hard to create a sufficient energy deficit to cause a substantial weight loss. Nonetheless, irrespective of weight loss, aerobic exercise can improve health and functional capacity [87].

Caloric Restriction

Most of the debate in weight loss for older adults concerns caloric restriction as a single strategy to reduce body weight [84, 88]. However, a recent randomized controlled trial concluded that intentional dietary weight loss was *not associated* with increased all-cause mortality over 12 years [89]. However, when weight loss is also accompanied by a low muscle mass and/or strength, it is unclear if weight loss should be recommended considering that weight loss is associated with loss in muscle mass [84].

Specific to Sarcopenic Obese and Dynapenic Obese

The optimal treatment strategy for sarcopenic obesity and dynapenic obesity appears to be a combination of exercise and caloric restriction [82]. More specifically, an energy deficit of about 500 kcal/day with adequate protein and calcium, in addition to multicomponent exercise program, is recommended [83]. With that being said, only one study has specifically looked at the impact of caloric restriction, exercise, and caloric restriction + exercise in a sample of older people classified as being dynapenic obese. Senechal and colleagues [53] studied 38 dynapenic obese postmenopausal women randomized into four different groups (caloric restriction, resistance training, both, or control) for 12 weeks. After a body weight reduction of ~5 % in the caloric restriction group and combined group, the study concluded that caloric restriction with or without resistance training was effective in improving metabolic profile, whereas resistance training was the most effective strategy to improve physical capacity. Furthermore, they did not observe any significant reduction in muscle strength in any group using caloric restriction as part of the intervention.

Clinical Recommendations

- Physical activity, and resistance training in particular, is the most promising approach for preventing and treating sarcopenia and dynapenia. To maximize muscle mass and strength development, 10–15 repetitions per set should be performed to the point of fatigue. Older adults should perform at least one set of repetitions for 8–10 exercises that train the major muscle groups, and exercises for each of the major muscle groups should occur on two or three nonconsecutive days of the week.
- 2. To help reduce the typical progression of sarcopenia in older people, ensure that their protein intake is at or slightly above the recommended dietary allowance. Protein supplements can be considered in addition to a person's ad libitum diet as these supplements are low in cost, are easy to administer, and well tolerated.
- 3. To treat obese older adults who have low muscle mass or low muscle strength, preliminary studies suggest that weight loss in combination with resistance training would be the best approach to improve metabolic health and functional limitations.

Conclusion

Sarcopenia, dynapenia, and obesity are a common part of the aging process. While the negative effects of sarcopenia were thought to be important, recent studies suggest that the effects of sarcopenia on functional impairment, mortality and metabolic health risks are modest. However, there is clear and consistent evidence that dynapenia is a risk factor for development of functional impairment, metabolic diseases, and mortality. While less is known about sarcopenic obesity and dynapenic obesity,

similar risk patterns are observed where dynapenic obese elderly people seem more affected than sarcopenic obese older adults. Resistance exercise is clearly the optimal treatment strategy for sarcopenia and dynapenia. Treatment, particularly for sarcopenia, can also include counseling on nutrition to ensure that dietary protein content is sufficient, and that there is pharmacological evaluation of anabolic hormones that tend to decline with old age. Finally, when sarcopenia or dynapenia is associated with obesity, preliminary data show that weight loss is safe and that greater results are obtained if moderate caloric restriction is prescribed in addition to regular exercise.

References

- 1. Rosenberg IH. Summary comments. Am J Clin Nutr. 1989;50:1231-3.
- 2. Baumgartner RN. Body composition in healthy aging. Ann N Y Acad Sci. 2000;904:437–48.
- 3. Roubenoff R. Origins and clinical relevance of sarcopenia. Can J Appl Physiol. 2001;26(1):78–89.
- 4. Morley JE, Baumgartner RN, Roubenoff R, Mayer J, Nair KS. Sarcopenia. J Lab Clin Med. 2001;137(4):231-43.
- Clark BC, Manini TM. Sarcopenia =/=dynapenia. J Gerontol A Biol Sci Med Sci. 2008;63(8):829–34. PubMed PMID: 18772470. Epub 2008/09/06. eng.
- Bouchard D, Janssen I. Dynapenic-obesity and physical function in older adults. J Gerontol A Biol Sci Med Sci. 2009;65(1):71–7. PubMed PMID: 19887536. Epub 2009/11/06. eng.
- Janssen I, Heymsfield SB, Wang ZM, Ross R. Skeletal muscle mass and distribution in 468 men and women aged 18–88 yr. J Appl Physiol. 2000;89(1):81–8.
- Lynch NA, Metter EJ, Lindle RS, Fozard JL, Tobin JD, Roy TA, et al. Muscle quality. I. Age-associated differences between arm and leg muscle groups. J Appl Physiol. 1999;86(1):188–94. PubMed PMID: 9887130.
- Frontera WR, Hughes VA, Fielding RA, Fiatarone MA, Evans WJ, Roubenoff R. Aging of skeletal muscle: a 12-yr longitudinal study. J Appl Physiol. 2000;88(4):1321–6. PubMed PMID: 10749826.
- Greig CA, Botella J, Young A. The quadriceps strength of healthy elderly people remeasured after eight years. Muscle Nerve. 1993;16(1):6–10. PubMed PMID: 8423833.
- Janssen I, Katzmarzyk PT, Ross R. Body mass index is inversely related to mortality in older people after adjustment for waist circumference. J Am Geriatr Soc. 2005;53(12):2112–8.
- Janssen I, Ross R. Linking age-related changes in skeletal muscle mass and composition with metabolism and disease. J Nutr Health Aging. 2005;9(6):408–19. PubMed PMID: 16395513. Epub 2006/01/06. eng.
- Baumgartner RN, Koehler KM, Gallagher D, Romero L, Heymsfield SB, Ross RR, et al. Epidemiology of sarcopenia among the elderly in New Mexico. Am J Epidemiol. 1998;147(8):755–63.
- 14. Janssen I, Heymsfield SB, Ross R. Low relative skeletal muscle mass (sarcopenia) in older persons is associated with functional impairment and physical disability. J Am Geriatr Soc. 2002;50(5):889–96.
- Tanko LB, Movsesyan L, Mouritzen U, Christiansen C, Svendsen OL. Appendicular lean tissue mass and the prevalence of sarcopenia among healthy women. Metabolism. 2002;51(1):69–74. PubMed PMID: 11782875. Epub 2002/01/10. eng.
- Wen X, Wang M, Jiang CM, Zhang YM. Are current definitions of sarcopenia applicable for older Chinese adults? J Nutr Health Aging. 2011;15(10):847–51. PubMed PMID: 22159771.
- Kim YS, Lee Y, Chung YS, Lee DJ, Joo NS, Hong D, et al. Prevalence of sarcopenia and sarcopenic obesity in the korean population based on the fourth korean national health and nutritional examination surveys. J Gerontol A Biol Sci Med Sci. 2012;67(10):1107–13. PubMed PMID: 22431554.
- Janssen I. Influence of sarcopenia on the development of physical disability: the Cardiovascular Health Study. J Am Geriatr Soc. 2006;54(1):56–62. PubMed PMID: 16420198. Epub 2006/01/20. eng.
- Visser M, Goodpaster BH, Kritchevsky SB, Newman AB, Nevitt M, Rubin SM, et al. Muscle mass, muscle strength, and muscle fat infiltration as predictors of incident mobility limitations in well-functioning older persons. J Gerontol A Biol Sci Med Sci. 2005;60(3):324–33.
- Visser M, Langlois J, Guralnik JM, Cauley JA, Kronmal RA, Robbins J, et al. High body fatness, but not low fat-free mass, predicts disability in older men and women: the Cardiovascular Health Study. Am J Clin Nutr. 1998;68(3):584–90.
- Miller MD, Crotty M, Giles LC, Bannerman E, Whitehead C, Cobiac L, et al. Corrected arm muscle area: an independent predictor of long-term mortality in community-dwelling older adults? J Am Geriatr Soc. 2002;50(7): 1272–7. PubMed PMID: 12133024. Epub 2002/07/23. eng.
- Muhlethaler R, Stuck AE, Minder CE, Frey BM. The prognostic significance of protein-energy malnutrition in geriatric patients. Age Ageing. 1995;24(3):193–7. PubMed PMID: 7645437. Epub 1995/05/01. eng.

- Bigaard J, Frederiksen K, Tjonneland A, Thomsen BL, Overvad K, Heitmann BL, et al. Body fat and fat-free mass and all-cause mortality. Obes Res. 2004;12(7):1042–9. PubMed PMID: 15292467.
- 24. Lammes E, Akner G. Resting metabolic rate in elderly nursing home patients with multiple diagnoses. J Nutr Health Aging. 2006;10(4):263–70. PubMed PMID: 16886096. Epub 2006/08/04. eng.
- Khamseh ME, Malek M, Aghili R, Emami Z. Sarcopenia and diabetes: pathogenesis and consequences. Br J Diabetes Vasc Dis. 2011;11:230–4.
- Karakelides H, Nair KS. Sarcopenia of aging and its metabolic impact. Curr Top Dev Biol. 2005;68:123–48. PubMed PMID: 16124998. Epub 2005/08/30. eng.
- Aubertin-Leheudre M, Lord C, Goulet ED, Khalil A, Dionne IJ. Effect of sarcopenia on cardiovascular disease risk factors in obese postmenopausal women. Obesity (Silver Spring). 2006;14(12):2277–83. PubMed PMID: 17189556. eng.
- Stephen WC, Janssen I. Sarcopenic-obesity and cardiovascular disease risk in the elderly. J Nutr Health Aging. 2009;13(5):460–6. PubMed PMID: 19390754.
- Balagopal P, Proctor D, Nair KS. Sarcopenia and hormonal changes. Endocrine. 1997;7(1):57–60. PubMed PMID: 9449033. Epub 1997/08/01. eng.
- Beliaeff S, Bouchard DR, Hautier C, Brochu M, Dionne IJ. Association between muscle mass and isometric muscle strength in well-functioning older men and women. J Aging Phys Act. 2008;16(4):484–93. PubMed PMID: 19033607. Epub 2008/11/27. eng.
- Clark BC, Manini TM. Functional consequences of sarcopenia and dynapenia in the elderly. Curr Opin Clin Nutr Metab Care. 2010;13(3):271–6. PubMed PMID: 20154609. Pubmed Central PMCID: 2895460.
- 32. Spirduso WW, Francis KL, MacRae PG. Physical dimnensions of aging: second edition. Champaign, IL: Human Kinetics; 2005. p. 384.
- Metter EJ, Conwit R, Tobin J, Fozard JL. Age-associated loss of power and strength in the upper extremities in women and men. J Gerontol A Biol Sci Med Sci. 1997;52(5):B267–76. PubMed PMID: 9310077.
- Barbat-Artigas S, Dupontgand S, Fex A, Karelis AD, Aubertin-Leheudre M. Relationship between dynapenia and cardiorespiratory functions in healthy postmenopausal women: novel clinical criteria. Menopause. 2010;18:400–5. PubMed PMID: 21107297. Epub 2010/11/26. Eng.
- Carmeli E, Reznick AZ, Coleman R, Carmeli V. Muscle strength and mass of lower extremities in relation to functional abilities in elderly adults. Gerontology. 2000;46(5):249–57. PubMed PMID: 10965180.
- Moreland JD, Richardson JA, Goldsmith CH, Clase CM. Muscle weakness and falls in older adults: a systematic review and meta-analysis. J Am Geriatr Soc. 2004;52(7):1121–9. PubMed PMID: 15209650.
- 37. Visser M, Schaap LA. Consequences of sarcopenia. Clin Geriatr Med. 2011;27(3):387–99. PubMed PMID: 21824554.
- Metter EJ, Talbot LA, Schrager M, Conwit RA. Arm-cranking muscle power and arm isometric muscle strength are independent predictors of all-cause mortality in men. J Appl Physiol. 2004;96(2):814–21. PubMed PMID: 14555682.
- Buchman AS, Wilson RS, Boyle PA, Bienias JL, Bennett DA. Change in motor function and risk of mortality in older persons. J Am Geriatr Soc. 2007;55(1):11–9. PubMed PMID: 17233680.
- 40. Jurca R, Jackson AS, LaMonte MJ, Morrow Jr JR, Blair SN, Wareham NJ, et al. Assessing cardiorespiratory fitness without performing exercise testing. Am J Prev Med. 2005;29(3):185–93. PubMed PMID: 16168867. Epub 2005/09/20. eng.
- 41. Karelis AD, Tousignant B, Nantel J, Proteau-Labelle M, Malita FM, St-Pierre DH, et al. Association of insulin sensitivity and muscle strength in overweight and obese sedentary postmenopausal women. Appl Physiol Nutr Metab. 2007;32(2):297–301. PubMed PMID: 17486172.
- 42. Sayer AA, Dennison EM, Syddall HE, Gilbody HJ, Phillips DI, Cooper C. Type 2 diabetes, muscle strength, and impaired physical function: the tip of the iceberg? Diabetes Care. 2005;28(10):2541–2. PubMed PMID: 16186295.
- Mozaffarian D, Hao T, Rimm EB, Willett WC, Hu FB. Changes in diet and lifestyle and long-term weight gain in women and men. N Engl J Med. 2011;364(25):2392–404. PubMed PMID: 21696306. Pubmed Central PMCID: 3151731.
- Forbes GB. Lean body mass-body fat interrelationships in humans. Nutr Rev. 1987;45(8):225–31. PubMed PMID: 3306482.
- 45. Visser M, van Venrooij LM, Vulperhorst L, de Vos R, Wisselink W, van Leeuwen PA, et al. Sarcopenic obesity is associated with adverse clinical outcome after cardiac surgery. Nutr Metab Cardiovasc Dis. 2012;23:511–8. PubMed PMID: 22397879.
- Oliveira RJ, Bottaro M, Junior JT, Farinatti PT, Bezerra LA, Lima RM. Identification of sarcopenic obesity in postmenopausal women: a cutoff proposal. Braz J Med Biol Res. 2011;44(11):1171–6. PubMed PMID: 22002095.
- Baumgartner RN, Wayne SJ, Waters DL, Janssen I, Gallagher D, Morley JE. Sarcopenic obesity predicts instrumental activities of daily living disability in the elderly. Obes Res. 2004;12(12):1995–2004.
- Monteiro MA, Gabriel RC, Sousa MF, Castro MN, Moreira MH. Temporal parameters of the foot roll-over during walking: influence of obesity and sarcopenic obesity on postmenopausal women. Maturitas. 2010;67(2):178–85. PubMed PMID: 20638802.

- Bouchard D, Dionne IJ, Brochu M. Sarcopenic/obesity and physical capacity in older men and women: data from the nutrition as a determinant of successful aging (NuAge)-the Quebec Longitudinal Study. Obesity (Silver Spring). 2009;17(11):2082–8. PubMed PMID: 19373219.
- 50. Davison KK, Ford ES, Cogswell ME, Dietz WH. Percentage of body fat and body mass index are associated with mobility limitations in people aged 70 and older from NHANES III. J Am Geriatr Soc. 2002;50(11):1802–9.
- Messier V, Karelis AD, Lavoie ME, Brochu M, Faraj M, Strychar I, et al. Metabolic profile and quality of life in class I sarcopenic overweight and obese postmenopausal women: a MONET study. Appl Physiol Nutr Metab. 2009;34(1):18–24. PubMed PMID: 19234581.
- Prado CM, Wells JC, Smith SR, Stephan BC, Siervo M. Sarcopenic obesity: a critical appraisal of the current evidence. Clin Nutr. 2012;31:583–601. PubMed PMID: 22809635.
- Senechal M, Bouchard DR, Dionne IJ, Brochu M. The effects of lifestyle interventions in dynapenic-obese postmenopausal women. Menopause. 2012;19:1015–21. PubMed PMID: 22473250. Epub 2012/04/05. Eng.
- Senechal M, Dionne IJ, Brochu M. Dynapenic abdominal obesity and metabolic risk factors in adults 50 years of age and older. J Aging Health. 2012;24(5):812–26. PubMed PMID: 22451528.
- 55. Campbell WW. Synergistic use of higher-protein diets or nutritional supplements with resistance training to counter sarcopenia. Nutr Rev. 2007;65(9):416–22. PubMed PMID: 17958209. Epub 2007/10/26. eng.
- Narici MV, Flueck M, Koesters A, Gimpl M, Reifberger A, Seynnes OR, et al. Skeletal muscle remodeling in response to alpine skiing training in older individuals. Scand J Med Sci Sports. 2011;21 Suppl 1:23–8. PubMed PMID: 21679320.
- Mangione KK, Miller AH, Naughton IV. Cochrane review: improving physical function and performance with progressive resistance strength training in older adults. Phys Ther. 2010;90(12):1711–5. PubMed PMID: 21123213.
- Valenzuela T. Efficacy of progressive resistance training interventions in older adults in nursing homes: a systematic review. J Am Med Dir Assoc. 2012;13(5):418–28. PubMed PMID: 22169509.
- Frontera WR, Meredith CN, O'Reilly KP, Knuttgen HG, Evans WJ. Strength conditioning in older men: skeletal muscle hypertrophy and improved function. J Appl Physiol. 1988;64(3):1038–44.
- 60. Nelson ME, Rejeski WJ, Blair SN, Duncan PW, Judge JO, King AC, et al. Physical activity and public health in older adults: recommendation from the American College of Sports Medicine and the American Heart Association. Circulation. 2007;116(9):1094–105. PubMed PMID: 17671236. Epub 2007/08/03. eng.
- Cook SB, Clark BC, Ploutz-Snyder LL. Effects of exercise load and blood-flow restriction on skeletal muscle function. Med Sci Sports Exerc. 2007;39(10):1708–13. PubMed PMID: 17909396.
- Fry CS, Glynn EL, Drummond MJ, Timmerman KL, Fujita S, Abe T, et al. Blood flow restriction exercise stimulates mTORC1 signaling and muscle protein synthesis in older men. J Appl Physiol. 2010;108(5):1199–209. PubMed PMID: 20150565. Pubmed Central PMCID: 2867530.
- Suga T, Okita K, Morita N, Yokota T, Hirabayashi K, Horiuchi M, et al. Dose effect on intramuscular metabolic stress during low-intensity resistance exercise with blood flow restriction. J Appl Physiol. 2010;108(6):1563–7. PubMed PMID: 20360434.
- 64. Suga T, Okita K, Morita N, Yokota T, Hirabayashi K, Horiuchi M, et al. Intramuscular metabolism during lowintensity resistance exercise with blood flow restriction. J Appl Physiol. 2009;106(4):1119–24. PubMed PMID: 19213931.
- Esmarck B, Andersen JL, Olsen S, Richter EA, Mizuno M, Kjaer M. Timing of postexercise protein intake is important for muscle hypertrophy with resistance training in elderly humans. J Physiol. 2001;535(Pt 1):301–11. PubMed PMID: 11507179. Epub 2001/08/17. eng.
- Fiatarone MA, O'Neill EF, Ryan ND, Clements KM, Solares GR, Nelson ME, et al. Exercise training and nutritional supplementation for physical frailty in very elderly people. N Engl J Med. 1994;330(25):1769–75.
- Symons TB, Sheffield-Moore M, Wolfe RR, Paddon-Jones D. A moderate serving of high-quality protein maximally stimulates skeletal muscle protein synthesis in young and elderly subjects. J Am Diet Assoc. 2009;109(9):1582– 6. PubMed PMID: 19699838. Pubmed Central PMCID: 3197704.
- 68. Verschueren SM, Bogaerts A, Delecluse C, Claessens AL, Haentjens P, Vanderschueren D, et al. The effects of whole-body vibration training and vitamin D supplementation on muscle strength, muscle mass, and bone density in institutionalized elderly women: a 6-month randomized, controlled trial. J Bone Mineral Res. 2011;26(1):42–9. PubMed PMID: 20648661.
- von Stengel S, Kemmler W, Engelke K, Kalender WA. Effect of whole-body vibration on neuromuscular performance and body composition for females 65 years and older: a randomized-controlled trial. Scand J Med Sci Sports. 2012;22(1):119–27. PubMed PMID: 20500555.
- Candow DG, Forbes SC, Little JP, Cornish SM, Pinkoski C, Chilibeck PD. Effect of nutritional interventions and resistance exercise on aging muscle mass and strength. Biogerontology. 2012;13(4):345–58. PubMed PMID: 22684187.
- Campbell WW, Trappe TA, Wolfe RR, Evans WJ. The recommended dietary allowance for protein may not be adequate for older people to maintain skeletal muscle. J Gerontol A Biol Sci Med Sci. 2001;56(6):M373–80. PubMed PMID: 11382798. Epub 2001/05/31. eng.

- Lauque S, Arnaud-Battandier F, Mansourian R, Guigoz Y, Paintin M, Nourhashemi F, et al. Protein-energy oral supplementation in malnourished nursing-home residents. A controlled trial. Age Ageing. 2000;29(1):51–6. PubMed PMID: 10690696. Epub 2000/02/26. eng.
- Paddon-Jones D, Sheffield-Moore M, Katsanos CS, Zhang XJ, Wolfe RR. Differential stimulation of muscle protein synthesis in elderly humans following isocaloric ingestion of amino acids or whey protein. Exp Gerontol. 2006;41(2):215–9. PubMed PMID: 16310330. Epub 2005/11/29. eng.
- 74. Pennings B, Boirie Y, Senden JM, Gijsen AP, Kuipers H, van Loon LJ. Whey protein stimulates postprandial muscle protein accretion more effectively than do casein and casein hydrolysate in older men. Am J Clin Nutr. 2011;93(5):997–1005. PubMed PMID: 21367943.
- 75. Fiatarone Singh MA. Exercise, nutrition, and the older woman: wellness for women over fifty. Boca Raton, FL: CRC; 2000.
- 76. Smith GI, Atherton P, Reeds DN, Mohammed BS, Rankin D, Rennie MJ, et al. Dietary omega-3 fatty acid supplementation increases the rate of muscle protein synthesis in older adults: a randomized controlled trial. Am J Clin Nutr. 2011;93(2):402–12. PubMed PMID: 21159787. Pubmed Central PMCID: 3021432.
- 77. Dawson-Hughes B, Castaneda-Sceppa C, Harris SS, Palermo NJ, Cloutier G, Ceglia L, et al. Impact of supplementation with bicarbonate on lower-extremity muscle performance in older men and women. Osteoporos Int. 2010;21(7):1171–9. PubMed PMID: 19727904. Pubmed Central PMCID: 2888724.
- Ottenbacher KJ, Ottenbacher ME, Ottenbacher AJ, Acha AA, Ostir GV. Androgen treatment and muscle strength in elderly men: a meta-analysis. J Am Geriatr Soc. 2006;54(11):1666–73. PubMed PMID: 17087692. Epub 2006/11/08. eng.
- 79. Borst SE. Interventions for sarcopenia and muscle weakness in older people. Age Ageing. 2004;33(6):548–55. PubMed PMID: 15385272. Epub 2004/09/24. eng.
- Nass RM, Gaylinn BD, Rogol AD, Thorner MO. Ghrelin and growth hormone: story in reverse. Proc Natl Acad Sci U S A. 2010;107(19):8501–2. PubMed PMID: 20435917. Pubmed Central PMCID: 2889295.
- Dionne IJ, Kinaman KA, Poehlman ET. Sarcopenia and muscle function during menopause and hormonereplacement therapy. J Nutr Health Aging. 2000;4(3):156–61. PubMed PMID: 10936902. Epub 2000/08/11. eng.
- Villareal DT, Chode S, Parimi N, Sinacore DR, Hilton T, Armamento-Villareal R, et al. Weight loss, exercise, or both and physical function in obese older adults. N Engl J Med. 2011;364(13):1218–29. PubMed PMID: 21449785. Epub 2011/04/01. eng.
- Mathus-Vliegen EM. Obesity management task force of the European Association for the Study of O. Prevalence, pathophysiology, health consequences and treatment options of obesity in the elderly: a guideline. Obes Facts. 2012;5(3):460–83. PubMed PMID: 22797374.
- 84. Villareal DT, Apovian CM, Kushner RF, Klein S. Obesity in older adults: technical review and position statement of the American Society for Nutrition and NAASO, The Obesity Society. Obes Res. 2005;13(11):1849–63.
- Ross R, Dagnone D, Jones PJ, Smith H, Paddags A, Hudson R, et al. Reduction in obesity and related comorbid conditions after diet-induced weight loss or exercise-induced weight loss in men. A randomized, controlled trial. Ann Intern Med. 2000;133(2):92–103.
- Ross R, Janssen I, Dawson J, Kungl AM, Kuk JL, Wong SL, et al. Exercise-induced reduction in obesity and insulin resistance in women: a randomized controlled trial. Obes Res. 2004;12(5):789–98. PubMed PMID: 15166299. eng.
- Lambert CP, Wright NR, Finck BN, Villareal DT. Exercise but not diet-induced weight loss decreases skeletal muscle inflammatory gene expression in frail obese elderly persons. J Appl Physiol. 2008;105(2):473–8. PubMed PMID: 18535122. Pubmed Central PMCID: 2519937.
- Miller SL, Wolfe RR. The danger of weight loss in the elderly. J Nutr Health Aging. 2008;12(7):487–91. PubMed PMID: 18615231.
- Shea MK, Nicklas BJ, Houston DK, Miller ME, Davis CC, Kitzman DW, et al. The effect of intentional weight loss on all-cause mortality in older adults: results of a randomized controlled weight-loss trial. Am J Clin Nutr. 2011;94(3):839–46. PubMed PMID: 21775558. Epub 2011/07/22. eng.

Chapter 7 Muscle Metabolism, Nutrition, and Functional Status in Older Adults

Douglas Paddon-Jones and Aaron P. Russell

Key Points

- The progression of sarcopenia is often mirrored by a decrease in physical activity, which feeds into a vicious cycle of disuse and negative health outcomes.
- Despite the compelling image of elite endurance athletes with very little muscle mass, aerobic exercise is not an *anti-anabolic* activity. Endurance exercise can increase both mitochondrial and myofibrillar protein synthesis
- Anabolic resistance is characterized by an impaired response to stimuli such as protein ingestion or exercise. Over time, this can be a significant factor contributing to the age-related loss of muscle mass and function.
- Combining human clinical interventions with basic research will provide the greatest opportunity to enhance our understanding of the regulation and maintenance of muscle mass and function during aging.

Keywords Sarcopenia • Protein metabolism • Muscle mass • Nutrition • Strength and function

Introduction

In the last 50 years, the number of individuals over the age of 65 years in the United States has doubled. A further doubling is expected by 2030, dramatically increasing the number of adults at risk of sarcopenia, a condition characterized by an age-related loss of muscle mass with an associated reduction in physical function [1]. A reduction in muscle mass and functional capacity is typically viewed as an undesirable, yet inevitable, consequence of aging, and in its early stages, may be easily

D. Paddon-Jones, Ph.D. (🖂)

Department of Nutrition and Metabolism, The University of Texas Medical Branch, 301 University Blvd, Galveston, TX, USA e-mail: djpaddon@utmb.edu

A.P. Russell, Ph.D. Centre for Physical Activity and Nutrition (C-PAN) Research, School of Exercise and Nutrition Sciences, Deakin University, 221 Burwood Hwy, Melbourne, VIC 3125, Australia e-mail: aaron.russell@deakin.edu.au

masked by subtle lifestyle adaptations. However, advanced sarcopenia is synonymous with physical frailty and is associated with an increased likelihood of falls and impairments in the ability to perform routine activities of daily living [2, 3]. In many instances, the progression of sarcopenia is mirrored by a decrease in physical activity, which feeds into a vicious cycle of disuse and negative outcomes, including impaired insulin action, accelerated loss of muscle and bone mass, fatigue, impaired motor control and functional capacity, and increased morbidity and mortality [4–6].

The inclusion of both muscle mass and function in the definition of sarcopenia has been stressed by several working groups as it provides a framework for understanding clinical consequences as well as potential interventions [1, 7]. While the trajectory of sarcopenia may be greatly accelerated by any number of catabolic events or crises (e.g., illness, injury, physical inactivity), an uncomplicated progression results in a 3-8 % reduction in muscle mass per decade, starting in the fourth or fifth decade of life [8]. Changes in muscle strength and functional capacity, while not directly proportional to changes in muscle mass in all instances, tend to follow a similar pattern with greater losses (approx. 3 % per decade) occurring after the age of 60 years [9].

An increasing number of translational research efforts continue to refine our understanding of sarcopenia and improve our ability to design and implement prevention and treatment-focused interventions. The goal of this chapter is to examine the opportunities and obstacles faced by clinicians and researchers seeking to understand or combat sarcopenia. To this end, we will focus on the regulation of muscle protein metabolism and the nutrition- and exercise-based strategies that provide the foundation for all other concomitant therapies.

Regulation of Muscle Mass with Age

Skeletal muscle represents approximately 40 % of the mass of the human body and plays a key regulatory role in metabolism, the maintenance of posture, and control of movement. Skeletal muscle is continuously being remodeled in response to loading and various metabolic stressors. Any progressive or unchecked reduction in muscle mass is a major health risk with the potential to compromise physical independence, reduce quality of life, and increase the reliance on professional or family care providers. Sarcopenia is the fifth leading cause of death in the United States' aged population and places significant socioeconomic pressure on family members and health care systems [3, 7, 10] (Fig. 7.1).

Skeletal muscle mass is maintained via the regulation of muscle protein synthesis and muscle protein breakdown [11, 12]. Considerable clinical research has focused on the implementation of exercise and nutrition interventions designed to stimulate muscle anabolism and ultimately preserve or increase muscle mass and function in young and aging adults alike [13–16]. While there have been successes, we continue to refine and improve prevention and treatment strategies for sarcopenia. To broaden our understanding of how nutrition and exercise may preserve muscle and function in older adults, recent attention has turned to understanding the regulation of intracellular signals that control muscle growth and breakdown [12, 17, 18]. An understanding of these regulatory signals is important, as very small alterations in the balance between muscle protein synthesis and breakdown (i.e., net balance) could facilitate a significant lean tissue loss over several years.

Signaling hormones including growth hormone (GH) and insulin-like growth factor-1 (IGF-1), as well as signaling pathways for Akt, mammalian target of rapamycin (mTOR) and striated muscle activator of Rho signaling/serum response factor (STARS/SRF), have been linked to sarcopenia (Fig. 7.2). Of particular interest is how these growth-regulating pathways are influenced by nutrition and muscle contraction in both young and older subjects. By understanding these key mechanistic events, we may be able to deliver targeted and effective exercise and nutrition strategies that may ultimately preserve muscle mass and functional capacity.



Fig. 7.1 The cascade of physiological changes leading to sarcopenic muscle loss and decreased physical function



Fig. 7.2 Potential relationship between several molecular signaling pathways involved in skeletal muscle protein synthesis and their attenuation with age. IGF-1, exercise, and BCAA are potent activators of muscle protein synthesis in both younger and older subjects. These factors can activate several upstream targets, including Akt, mTOR, and STARS, that positively impact, directly or indirectly, muscle growth. The *red circles* indicate those known targets that are down-regulated in skeletal muscle from the elderly or that have attenuated activation in response to IGF-1, exercise and BCAA interventions. \rightarrow , known activator; \neg , known inhibitor

IGF-1/Akt-1 Signaling and Muscle Mass

IGF-1 activates a series of intracellular signaling cascades involving phosphatidylinositol 3-kinase (PI3K) and Akt-1 that regulate muscle growth [19–21]. Downstream targets of Akt-1 include glycogen synthase kinase-3 β (GSK-3 β) and mTOR. Phosphorylation of GSK-3 β by Akt-1 [22, 23] releases its inhibition of the translation initiation factor eIF2B [24]. The phosphorylation and activation of mTOR results in further phosphorylation and activation of ribosomal p70^{S6K}, as well as phosphorylation of PHAS-1/4E-BP1 [24]. Phosphorylation of both p70^{S6K} and PHAS-1/4E-BP1 activates pathways promoting protein synthesis and translation initiation, respectively. Hence, the Akt-1/GSK-3 β and Akt-1/mTOR pathways are important for muscle growth and repair. Additionally, IGF-1/Akt can also reduce the activity of pathways involved in muscle protein breakdown [25]. Akt-1 phosphorylates the FoXO family of transcription factors where they remain sequestered to the cytoplasm. This inhibits the FoXO transcription of atrogin-1/MAFbx and MuRF1 [25], both important regulators of skeletal muscle atrophy [26, 27].

In aging human skeletal muscle, a reduction in IGF mRNA has been observed [28, 29]. Lower levels of IGF-1 may compromise that ability to activate Akt-1 in response to meals or exercise. Compared to young males, an elevated level of total Akt-1 protein has been observed in skeletal muscle from older men [28]. However, this was not matched by an elevated increase in phosphory-lated Akt-1. This observation supports those made in older rat skeletal muscle [30] and suggests an age-related reduction in the efficiency to phosphorylate skeletal muscle Akt-1. Inefficient activation of the Akt-1 protein following an anabolic stimuli may contribute to impaired muscle protein synthesis and muscle growth.

The downstream GSK-3 β and mTOR pathways, two axes independently stimulated by Akt-1, have also been measured and compared in muscle biopsies from young and older adults [28, 31]. Increased levels of total and phosphorylated GSK-3 β have been observed in older subjects [28]. An increased pool of GSK-3 β protein may be a result of increased protein translation or protein stability, aimed at providing the cell with a source to maintain protein synthesis. These observations suggest the existence of a mechanism that is able to phosphorylate GSK-3 β , independently of Akt-1, an observation not without precedent [32]. Total and phosphorylated protein levels of mTOR and its downstream targets p70^{S6k} [31], but not 4E-BP1 [28], have also been observed to be reduced in the elderly, when compared with younger muscle.

STARS/SRF Signaling

STARS is a muscle-specific actin-binding protein that binds to the I-band of the sarcomere and to actin filaments [33]. STARS increases actin polymerization by increasing the binding of free G-actin to F-actin filaments [33]. As a consequence the G-actin inhibition of the transcriptional coactivator, myocardin-related transcription factor-A (MRTF-A) is removed, permitting the nuclear translocation of MRTF-A and its enhancement of SRF transcription [34] (see Fig. 7.2). STARS also enhances smooth muscle cell proliferation [35], and in unpublished work, we have observed the same effect in C2C12 myoblasts. Following 8 weeks of hypertrophy-stimulating resistance training, STARS, MRTF-A, and SRF mRNA as well as RhoA and nuclear SRF protein levels are increased [36]. This is associated with increases in several SRF target genes: the structural protein α -actin [37], the motor protein myosin heavy chain type IIa (MHC IIa), and the insulin-like growth factor-1 (IGF-1) [38]. STARS mRNA and protein levels are also increased 3 h following an acute bout of single-leg endurance exercise [39]. STARS has been suggested to play a role in stabilizing the sarcomere, protecting it from mechanically induced damage [33]. This transient increase in STARS following contraction-induced mechanical stress may provide a protective mechanism to reduce the risk of contractile damage to the sarcomere or be required to activate intracellular signals responsible for muscle adaptation to exercise. Recently, STARS, MRTF-A, and SRF have been shown to be reduced in skeletal muscles from aged 24-month-old mice [40]. SRF protein levels are also reduced in mice at 15 months of age with an associated decrease in the SRF target gene, α -actin [41]. Similarly, SRF protein levels are also reduced in muscle biopsies form elderly subjects [41]. Combined, these results suggest that the loss of members of the STARS signaling pathway, in particular SRF, may contribute to the loss of muscle mass and function in older adults. As IGF-1, a transcriptional target of SRF, is also reduced in aged muscle a compromised STARS/SRF signaling pathway may contribute to reduced IGF-1 levels and associated age-related muscle wasting. It is noteworthy that SRF activity is reduced in aged liver [42] and the transgenic disruption of hepatic SRF results in impaired liver function and IGF-1 production [43]. This aging-associated decline in SRF activity may influence the reduction in circulating IGF-1 and therefore perturb the pathways involved in muscle growth and regeneration.

Nutrition and Muscle Metabolism

Many factors that contribute to sarcopenia are precipitated by subtle negative changes in behavior and/or physiology and may require several years to facilitate a significant loss of lean tissue or muscle function [7]. To this end, one area that continues to attract considerable research and lay attention is the potential change in the metabolic response to meals in older adults. In both fasting and fed states, muscle protein synthesis and breakdown are regulated by various molecular switches that in turn control a series of intracellular signaling cascades. Identifying and understanding the activation of these molecular switches complements ongoing efforts to optimize dietary strategies to preserve muscle mass and function.

Anabolic resistance is characterized by an impaired or blunted cellular response to anabolic stimuli, such as protein ingestion [31, 44]. Over time, a chronically diminished capacity to stimulate net muscle protein anabolism is likely to be a significant factor contributing to the age-related loss of muscle mass and function. Until quite recently, most studies exploring changes in muscle protein synthesis in response to nutrition interventions used isolated amino acid cocktails [45-49] or, if they were more translational in design, whey [50, 51] or soy protein [50-53]. Many of these well-controlled, single macronutrient studies clearly demonstrated acute, age-related differences in the magnitude, time-course, and regulation of muscle protein anabolism. However, an increasing number of studies using protein-rich foods (e.g., beef, dairy products) as the anabolic stimulus suggest that ingestion of a moderate-to-large serving of protein in a single meal robustly stimulates muscle protein synthesis and markers of anabolism similarly in young and older adults [54–56]. In practical terms, the threshold for an age-independent increase in muscle protein anabolism appears to be a meal containing approximately 25-30 g of protein. However, older adults, particularly those who are sedentary, experience a blunted anabolic response to lower protein meals, or perhaps even meals containing a mix of protein and carbohydrate [57, 58]. This potential protein "threshold" response is consistent with the concept of anabolic resistance in older adults and is the focus of much ongoing research.

Essential amino acids, or specifically the branched chain amino acids (BCAA) and especially leucine, directly activate the mammalian target of rapamycin (mTOR) signaling pathway [59]. The mTOR complex consists of mTORC1 and mTORC2 that have distinct roles in the regulation muscle protein synthesis. mTORC1 activates several kinases including S6 kinase (S6K) and eIF4E [60, 61]. Activation of this signaling cascade increases cellular protein translation and initiation resulting in increased protein synthesis. As noted, in some older adults there is a blunted anabolic response to insulin and amino acids resulting in reduced mTOR phosphorylation and a decreased ability to initiate protein translation [57, 62, 63]. Similar observations have also been made in animal models (e.g., 20-month- vs. 8-month-old rats) [64]. An attenuated vasodilatory response to insulin in the elderly may also reduce the amount of available amino acids to the muscle [65].

Inflammation and Oxidative Stress

An increasing number of older individuals are at risk of sarcopenic obesity, an increase in fat mass with a concomitant reduction in muscle mass. Lipid accumulation has well-described negative effects on anabolic signaling via its activation of pro-inflammatory factors such as tumor necrosis factor alpha (TNF- α) and nuclear factor kappa-light-chain enhancer of activated B cells (NF κ B) [66]. TNF- α , NF κ B, and interlukin-6 (IL-6) are increased in muscle from older individuals [31, 67]. TNF- α may negatively influence muscle mass and function by reducing muscle protein synthesis and/or increasing protein degradation [68]. The inflammation-induced loss of muscle mass and function is complex and yet to be completely understood. However, TNF- α is a key signaling molecule activating apoptosis in skeletal muscle and it has been shown that apoptosis, inflammation, and oxidative damage play a role in age-related loss in muscle mass and strength [69, 70]. TNF- α also synthesizes ceramide [71], which is elevated in muscle from older individuals [72]. Ceramide reduces amino acid availability and inhibits the activation of Akt, mTOR, and ribosomal S6 protein [73, 74] as well as blunting muscle cell protein synthesis [73] (see Fig. 7.2).

Nutrition is a key modifiable factor with the potential to positively and negatively influence inflammation, apoptosis, and oxidative damage as well as ceramide levels. Consequently, understanding how nutrient ingestion can impact the molecular signals influencing muscle mass and function is important for the development of appropriate nutritional interventions. A TNF- α -dependent mechanism has been shown to reduce muscle protein synthesis during sepsis, an extreme model of systemic inflammation. This was facilitated/accompanied by reductions in mTOR, S6K, and 4E-BP1 [75]. Inflammation and TNF- α may also negatively influence skeletal muscle by decreasing growth hormone (GH) and IGF-1 [76] (see Fig. 7.2). IGF-1 gene transcription is controlled by GH via a Janus kinase-2 (JAK2)/signal transducer and activator of transcription-5b (STAT5b) signaling pathway [77, 78]. Reduced IGF levels in aged muscle may be linked to reduced circulating levels of GH [79], GH-receptor content [28] or GH sensitivity [80]. TNF- α regulates the transcription of suppressor of cytokine signaling-3 (SOCS3) [81], with the latter able to inhibit GH signaling to JAK2 and STAT5b [82, 83]. SOCS3 levels are increased in elderly males, although this is not associated with reduced STATb phosphorylation [28]. This suggests that the age-related reduction in IGF-1 mRNA may be influenced by a GH/SOCS3 pathway but independent of STAT5b transcriptional perturbation.

Exercise

To understand the benefits of exercise, it is useful to first consider the consequences of its absence. A sedentary lifestyle is undoubtedly a key precipitating factor contributing to sarcopenia [84]. Taking a lack of physical activity to an extreme, bed-rest studies clearly show that significant muscle mass and functional capacity can be lost in a very short period of time. Further, the rate of loss appears to be much greater in older adults, compared to a younger study population [48, 85] (e.g., >1 kg muscle loss during 10 days bed rest vs. 0.5 kg during 28 days bed rest). Two of the key negative adaptions are motor unit denervation and the preferential loss of fast twitch/type II muscle fibers. Both of these changes have the potential to reduce the ability to generate muscle power (i.e., the ability to rapidly apply force), which is necessary to successfully perform many activities of daily living and protect against unexpected events such as tripping or falling [86].

Intracellular stress induced by exercise can obviously be a potent anabolic stimulus. While basal (i.e., fasted and rested) muscle protein synthesis rates are similar in healthy young and older adults, like nutrition, some older subjects may also experience a blunted anabolic response to exercise. The age-related factors contributing to this are not precisely known; however, investigations into the molecular regulation of exercise-induced protein synthesis have shed some light.

Resistance Exercise

Resistance exercise training can effectively increase muscle mass and strength in older populations [87, 88]. Looking at the early adaptations that may ultimately contribute to phenotypic change; it has been demonstrated in cell culture models [20], rodents [89], and humans [90] that muscle protein synthesis requires activation of the mTORC1 pathway. Following an acute bout of resistance exercise, protein synthesis increases in both younger and older adults in a dose (i.e., intensity) -dependent manner [91, 92]. However, in many instances the duration or magnitude of the increase in protein synthesis appears to be slightly lower in the older subjects [92]. A similar age-related response has been observed following 4 months of resistance training [93]. The attenuated protein synthesis in the elderly following acute resistance exercise is associated with an impaired activation of mTORC1 (96), p70S6K and 4E-BP1 [92] (see Fig. 7.2). Similarly, a blunted activation of Akt/ mTOR signaling has been observed in aged rat skeletal muscle following contractile stimulation known to induce muscle growth [94].

Endurance Exercise

Despite the compelling image of elite endurance athletes (e.g., marathon runners) with very little muscle mass, aerobic exercise per se is not an *anti-anabolic* activity. Endurance exercise can increase mitochondrial and myofibrillar protein synthesis [95], while chronic endurance training has been shown to activate mTOR and stimulate muscle hypertrophy [96]. These positive adaptations are important as mitochondrial mass and function tend to decrease with age [97]. Peroxisome proliferator-activated receptor-gamma (PPAR- γ) coactivator 1 alpha (PGC-1 α) is a key regulator of mitochondrial biogenesis [98] and is reduced in aging muscle [99]. PGC-1 α suppresses production of reactive oxygen species (ROS) [100], and its overexpression in mice protects against the on-set of sarcopenia [101]. Endurance exercise up regulates PGC-1 α [102], and this may play a role in maintaining aging muscle mass as PGC-1 α also inhibits lysosomal and proteasomal protein degradation via suppression of transcriptional activity of Forkhead (FoXO) transcription factors [103], the latter known to be increased in muscle from aging humans [104] and rats [105].

Low-Intensity Activities

Despite the concern that some types of physical activity may be impractical or medically contraindicated in frail older adults or patient populations, it is nonetheless clear that almost any form of physical activity is preferable to inactivity or bed rest. Numerous studies have reported robust improvements in muscle mass and strength following higher-intensity resistance exercise training programs in older adults [106]. In contrast, in a NASA-supported 21-day bed-rest study, we demonstrated that lowintensity muscular activity can also largely prevent the typical decline in muscle mass and function observed with physical inactivity. Specifically, spending as little as 1 h each day on a specially designed human centrifuge (2.5× gravity at the feet) was sufficient to maintain muscle protein synthesis [107] and partially preserve lower extremity muscle function [86]. In a clinical setting, this supports the growing consensus that even brief periods of very low intensity activity, such as walking or simply weight-bearing, may provide health benefits.

Combining Exercise and Nutrition

In almost every acute muscle metabolism study, the combination of physical activity and protein/ amino acid ingestion has a positive, additive-like effect on skeletal muscle protein synthesis [58]. In a series of recent studies, we reported a 50 % increase in mixed muscle protein synthesis in cohorts of young and older adults following ingestion of a single protein-rich meal (i.e., lean beef). When subjects completed a simple bout of resistance exercise (i.e., leg extension exercise) immediately following the same meal, we observed a further 50 % increase in muscle protein synthesis during the same time frame.

This observation that aging does not necessarily impair the ability to stimulate muscle protein synthesis in response to exercise and an actual protein-rich meal is certainly encouraging news for older adults. However, as noted earlier, it is clear that in some instances the magnitude and/or time course of changes in muscle protein anabolism may differ with age. For example, in young adults, ingestion of 20 g of essential amino acids 1 h following a bout of resistance exercise has been shown to maximally increase muscle protein synthesis during a 1–3-h post-exercise period. In a cohort of older adults, the magnitude of the increase in muscle protein synthesis was similar; however, the peak did not occur until 3–6 h post-exercise [91]. Thus, it appears that while many elements of the signaling pathway (e.g., mTOR, S6K1, 4E-BP1, eEF2) are not directly affected by age, others (e.g., ERK1/2 signaling and AMPK activation) may be less responsive, thereby contributing to the delayed increase in muscle protein synthesis in older adults.

Finally, while carefully controlled, acute research studies can provide a great deal of useful mechanistic data [51, 52, 54], they often fail to reflect a realistic exercise and/or eating experience or predict a longer term outcome (e.g., a change in muscle mass or strength). Further, due to the invasive nature of clinical trials that involve procedures such as stable isotope infusions, muscle biopsies, and rigorous exercise testing, participation is often limited to older adults who are in remarkably good health, and perhaps not representative of the larger aging population. There is clearly a need to adopt a more translational research approach to current acute/mechanistic feeding studies by linking them to longer duration outcome studies.

Conclusion

Sarcopenia has debilitating consequences for the elderly community and considerable impact on health-care systems. As our society ages and continues to demand higher living standards and quality of life, identifying lifestyle and/or therapeutic interventions to maintain skeletal muscle mass and function remains a key focus for biomedical and health research. A multidisciplinary research approach, combining human clinical interventions, with molecular interventions in animal models and cells, will harness the minds and diverse research strengths of scientists interested in healthy aging. This approach will provide the greatest opportunity to enhance our understanding of the regulation and maintenance of muscle mass and function during aging.

References

- 1. Morley JE, Abbatecola AM, Argiles JM, Baracos V, Bauer J, Bhasin S, et al. Sarcopenia with limited mobility: an international consensus. J Am Med Dir Assoc. 2011;12(6):403–9.
- 2. Evans W. Functional and metabolic consequences of sarcopenia. J Nutr. 1997;127(5 Suppl):998S-1003.

^{3.} Mahoney J, Sager M, Dunham NC, Johnson J. Risk of falls after hospital discharge. J Am Geriatr Soc. 1994; 42(3):269–74.

- 4. Urso ML, Clarkson PM, Price TB. Immobilization effects in young and older adults. Eur J Appl Physiol. 2006;96(5):564–71.
- Paddon-Jones D, Wolfe RR, Ferrando AA. Amino acid supplementation for reversing bed rest and steroid myopathies. J Nutr. 2005;135(7):1809S–12.
- Clark BC, Pierce JR, Manini TM, Ploutz-Snyder LL. Effect of prolonged unweighting of human skeletal muscle on neuromotor force control. Eur J Appl Physiol. 2007;100(1):53–62.
- Fielding RA, Vellas B, Evans WJ, Bhasin S, Morley JE, Newman AB, et al. Sarcopenia: an undiagnosed condition in older adults. Current consensus definition: prevalence, etiology, and consequences. International working group on sarcopenia. J Am Med Dir Assoc. 2011;12(4):249–56.
- 8. Evans WJ. Skeletal muscle loss: cachexia, sarcopenia, and inactivity. Am J Clin Nutr. 2010;91(4):1123S-7.
- von Haehling S, Morley JE, Anker SD. An overview of sarcopenia: facts and numbers on prevalence and clinical impact. J Cachexia Sarcopenia Muscle. 2010;1(2):129–33.
- 10. Volpi E, Nazemi R, Fujita S. Muscle tissue changes with aging. Curr Opin Clin Nutr Metab Care. 2004;7(4): 405–10.
- 11. Wolfe RR. Regulation of muscle protein by amino acids. J Nutr. 2002;132(10):3219S-24.
- 12. Glass DJ. Signalling pathways that mediate skeletal muscle hypertrophy and atrophy. Nat Cell Biol. 2003; 5(2):87–90.
- 13. Campbell WW. Synergistic use of higher-protein diets or nutritional supplements with resistance training to counter sarcopenia. Nutr Rev. 2007;65(9):416–22.
- English KL, Paddon-Jones D. Protecting muscle mass and function in older adults during bed rest. Curr Opin Clin Nutr Metab Care. 2010;13(1):34–9.
- 15. Evans WJ. Protein nutrition, exercise and aging. J Am Coll Nutr. 2004;23(6 Suppl):601S-9.
- 16. Koopman R. Dietary protein and exercise training in ageing. Proc Nutr Soc. 2011;70(1):104–13.
- Fry CS, Drummond MJ, Glynn EL, Dickinson JM, Gundermann DM, Timmerman KL, et al. Aging impairs contraction-induced human skeletal muscle mTORC1 signaling and protein synthesis. Skelet Muscle. 2011;1(1):11.
- Drummond MJ, McCarthy JJ, Sinha M, Spratt HM, Volpi E, Esser KA, et al. Aging and microRNA expression in human skeletal muscle: a microarray and bioinformatics analysis. Physiol Genomics. 2011;43(10):595–603.
- 19. Datta SR, Brunet A, Greenberg ME. Cellular survival: a play in three Akts. Genes Dev. 1999;13(22):2905–27.
- Rommel C, Bodine SC, Clarke BA, Rossman R, Nunez L, Stitt TN, et al. Mediation of IGF-1-induced skeletal myotube hypertrophy by PI(3)K/Akt/mTOR and PI(3)K/Akt/GSK3 pathways. Nat Cell Biol. 2001; 3(11):1009–13.
- Vivanco I, Sawyers CL. The phosphatidylinositol 3-Kinase AKT pathway in human cancer. Nat Rev Cancer. 2002; 2(7):489–501.
- Jefferson LS, Fabian JR, Kimball SR. Glycogen synthase kinase-3 is the predominant insulin-regulated eukaryotic initiation factor 2B kinase in skeletal muscle. Int J Biochem Cell Biol. 1999;31(1):191–200.
- Welsh GI, Stokes CM, Wang X, Sakaue H, Ogawa W, Kasuga M, et al. Activation of translation initiation factor eIF2B by insulin requires phosphatidyl inositol 3-kinase. FEBS Lett. 1997;410(2–3):418–22.
- 24. Rhoads RE. Signal transduction pathways that regulate eukaryotic protein synthesis. J Biol Chem. 1999; 274(43):30337–40.
- Stitt TN, Drujan D, Clarke BA, Panaro F, Timofeyva Y, Kline WO, et al. The IGF-1/PI3K/Akt pathway prevents expression of muscle atrophy-induced ubiquitin ligases by inhibiting FOXO transcription factors. Mol Cell. 2004;14(3):395–403.
- Bodine SC, Latres E, Baumhueter S, Lai VK, Nunez L, Clarke BA, et al. Identification of ubiquitin ligases required for skeletal muscle atrophy. Science. 2001;294(5547):1704–8.
- Gomes MD, Lecker SH, Jagoe RT, Navon A, Goldberg AL. Atrogin-1, a muscle-specific F-box protein highly expressed during muscle atrophy. Proc Natl Acad Sci U S A. 2001;98(25):14440–5.
- Leger B, Derave W, De Bock K, Hespel P, Russell AP. Human sarcopenia reveals an increase in SOCS-3 and myostatin and a reduced efficiency of Akt phosphorylation. Rejuvenation Res. 2008;11(1):163–175B.
- 29. Welle S. Cellular and molecular basis of age-related sarcopenia. Can J Appl Physiol. 2002;27(1):19-41.
- Haddad F, Adams GR. Aging-sensitive cellular and molecular mechanisms associated with skeletal muscle hypertrophy. J Appl Physiol. 2006;100(4):1188–203.
- Cuthbertson D, Smith K, Babraj J, Leese G, Waddell T, Atherton P, et al. Anabolic signaling deficits underlie amino acid resistance of wasting, aging muscle. FASEB J. 2005;19(3):422–4.
- Hornberger TA, Stuppard R, Conley KE, Fedele MJ, Fiorotto ML, Chin ER, et al. Mechanical stimuli regulate rapamycin-sensitive signalling by a phosphoinositide 3-kinase-, protein kinase B- and growth factor-independent mechanism. Biochem J. 2004;380(Pt 3):795–804.
- Arai A, Spencer JA, Olson EN. STARS, a striated muscle activator of Rho signaling and serum response factordependent transcription. J Biol Chem. 2002;277(27):24453–9.
- Miralles F, Posern G, Zaromytidou AI, Treisman R. Actin dynamics control SRF activity by regulation of its coactivator MAL. Cell. 2003;113(3):329–42.

- 35. Troidl K, Ruding I, Cai WJ, Mucke Y, Grossekettler L, Piotrowska I, et al. Actin-binding rho activating protein (Abra) is essential for fluid shear stress-induced arteriogenesis. Arterioscler Thromb Vasc Biol. 2009;29(12): 2093–101.
- 36. Lamon S, Wallace MA, Leger B, Russell AP. Regulation of STARS and its downstream targets suggest a novel pathway involved in human skeletal muscle hypertrophy and atrophy. J Physiol. 2009;587(Pt 8):1795–803.
- Carson JA, Schwartz RJ, Booth FW. SRF and TEF-1 control of chicken skeletal alpha-actin gene during slowmuscle hypertrophy. Am J Physiol. 1996;270(6 Pt 1):C1624–33.
- Charvet C, Houbron C, Parlakian A, Giordani J, Lahoute C, Bertrand A, et al. New role for serum response factor in postnatal skeletal muscle growth and regeneration via the interleukin 4 and insulin-like growth factor 1 pathways. Mol Cell Biol. 2006;26(17):6664–74.
- Wallace MA, Hock MB, Hazen BC, Kralli A, Snow RJ, Russell AP. Striated muscle activator of Rho signalling (STARS) is a PGC-1alpha/oestrogen-related receptor-alpha target gene and is upregulated in human skeletal muscle after endurance exercise. J Physiol. 2011;589(Pt 8):2027–39.
- 40. Sakuma K, Akiho M, Nakashima H, Akima H, Yasuhara M. Age-related reductions in expression of serum response factor and myocardin-related transcription factor A in mouse skeletal muscles. Biochim Biophys Acta. 2008;1782(7–8):453–61.
- Lahoute C, Sotiropoulos A, Favier M, Guillet-Deniau I, Charvet C, Ferry A, et al. Premature aging in skeletal muscle lacking serum response factor. PLoS One. 2008;3(12):e3910.
- 42. Supakar PC, Roy AK. Role of transcription factors in the age-dependent regulation of the androgen receptor gene in rat liver. Biol Signals. 1996;5(3):170–9.
- 43. Sun K, Battle MA, Misra RP, Duncan SA. Hepatocyte expression of serum response factor is essential for liver function, hepatocyte proliferation and survival, and postnatal body growth in mice. Hepatology. 2009;49(5): 1645–54.
- 44. Rennie MJ, Wilkes EA. Maintenance of the musculoskeletal mass by control of protein turnover: the concept of anabolic resistance and its relevance to the transplant recipient. Ann Transplant. 2005;10(4):31–4.
- 45. Bohé J, Low A, Wolfe RR, Rennie MJ. Human muscle protein synthesis is modulated by extracellular, not intramuscular amino acid availability: a dose-response study. J Physiol. 2003;552(Pt 1):315–24.
- 46. Ferrando AA, Paddon-Jones D, Hays NP, Kortebein P, Ronsen O, Williams RH, et al. EAA supplementation to increase nitrogen intake improves muscle function during bed rest in the elderly. Clin Nutr. 2010;29(1):18–23.
- 47. Paddon-Jones D, Sheffield-Moore M, Zhang XJ, Volpi E, Wolf SE, Aarsland A, et al. Amino acid ingestion improves muscle protein synthesis in the young and elderly. Am J Physiol Endocrinol Metab. 2004;286(3): E321–8.
- 48. Paddon-Jones D, Sheffield-Moore M, Urban RJ, Sanford AP, Aarsland A, Wolfe RR, et al. Essential amino acid and carbohydrate supplementation ameliorates muscle protein loss in humans during 28 days bedrest. J Clin Endocrinol Metab. 2004;89(9):4351–8.
- Paddon-Jones D, Sheffield-Moore M, Katsanos CS, Zhang XJ, Wolfe RR. Differential stimulation of muscle protein synthesis in elderly humans following isocaloric ingestion of amino acids or whey protein. Exp Gerontol. 2006;41(2):215–9.
- Katsanos CS, Chinkes DL, Paddon-Jones D, Zhang XJ, Aarsland A, Wolfe RR. Whey protein ingestion in elderly persons results in greater muscle protein accrual than ingestion of its constituent essential amino acid content. Nutr Res. 2008;28(10):651–8.
- Pennings B, Boirie Y, Senden JM, Gijsen AP, Kuipers H, van Loon LJ. Whey protein stimulates postprandial muscle protein accretion more effectively than do casein and casein hydrolysate in older men. Am J Clin Nutr. 2011;93(5):997–1005.
- Wilkinson SB, Tarnopolsky MA, Macdonald MJ, Macdonald JR, Armstrong D, Phillips SM. Consumption of fluid skim milk promotes greater muscle protein accretion after resistance exercise than does consumption of an isonitrogenous and isoenergetic soy-protein beverage. Am J Clin Nutr. 2007;85(4):1031–40.
- 53. Yang Y, Churchward-Venne TA, Burd NA, Breen L, Tarnopolsky MA, Phillips SM. Myofibrillar protein synthesis following ingestion of soy protein isolate at rest and after resistance exercise in elderly men. Nutr Metab (Lond). 2012;9(1):57.
- 54. Symons TB, Schutzler SE, Cocke TL, Chinkes DL, Wolfe RR, Paddon-Jones D. Aging does not impair the anabolic response to a protein-rich meal. Am J Clin Nutr. 2007;86(2):451–6.
- 55. Symons TB, Sheffield-Moore M, Wolfe RR, Paddon-Jones D. A moderate serving of high-quality protein maximally stimulates skeletal muscle protein synthesis in young and elderly subjects. J Am Diet Assoc. 2009;109(9): 1582–6.
- 56. Phillips SM. Nutrient-rich meat proteins in offsetting age-related muscle loss. Meat Sci. 2012;92(3):174-8.
- 57. Volpi E, Mittendorfer B, Rasmussen BB, Wolfe RR. The response of muscle protein anabolism to combined hyperaminoacidemia and glucose-induced hyperinsulinemia is impaired in the elderly. J Clin Endocrinol Metab. 2000;85(12):4481–90.
- 58. Yang Y, Breen L, Burd NA, Hector AJ, Churchward-Venne TA, Josse AR, et al. Resistance exercise enhances myofibrillar protein synthesis with graded intakes of whey protein in older men. Br J Nutr. 2012;108:1780.
- Anthony JC, Anthony TG, Kimball SR, Vary TC, Jefferson LS. Orally administered leucine stimulates protein synthesis in skeletal muscle of postabsorptive rats in association with increased eIF4F formation. J Nutr. 2000;130(2):139–45.
- 60. Fenton TR, Gout IT. Functions and regulation of the 70kDa ribosomal S6 kinases. Int J Biochem Cell Biol. 2011;43(1):47–59.
- Meyuhas O, Dreazen A. Ribosomal protein S6 kinase from TOP mRNAs to cell size. Progr Mol Biol Trans Sci. 2009;90:109–53.
- Guillet C, Prod'homme M, Balage M, Gachon P, Giraudet C, Morin L, et al. Impaired anabolic response of muscle protein synthesis is associated with S6K1 dysregulation in elderly humans. FASEB J. 2004;18(13):1586–7.
- 63. Rasmussen BB, Fujita S, Wolfe RR, Mittendorfer B, Roy M, Rowe VL, et al. Insulin resistance of muscle protein metabolism in aging. FASEB J. 2006;20(6):768–9.
- 64. Prod'homme M, Balage M, Debras E, Farges MC, Kimball S, Jefferson L, et al. Differential effects of insulin and dietary amino acids on muscle protein synthesis in adult and old rats. J Physiol. 2005;563(Pt 1):235–48.
- 65. Fujita S, Rasmussen BB, Cadenas JG, Grady JJ, Volpi E. Effect of insulin on human skeletal muscle protein synthesis is modulated by insulin-induced changes in muscle blood flow and amino acid availability. Am J Physiol Endocrinol Metab. 2006;291(4):E745–54.
- 66. Holland WL, Bikman BT, Wang LP, Yuguang G, Sargent KM, Bulchand S, et al. Lipid-induced insulin resistance mediated by the proinflammatory receptor TLR4 requires saturated fatty acid-induced ceramide biosynthesis in mice. J Clin Invest. 2011;121(5):1858–70.
- Chung HY, Cesari M, Anton S, Marzetti E, Giovannini S, Seo AY, et al. Molecular inflammation: underpinnings of aging and age-related diseases. Ageing Res Rev. 2009;8(1):18–30.
- Toth MJ, Matthews DE, Tracy RP, Previs MJ. Age-related differences in skeletal muscle protein synthesis: relation to markers of immune activation. Am J Physiol Endocrinol Metab. 2005;288(5):E883–91.
- Phillips T, Leeuwenburgh C. Muscle fiber specific apoptosis and TNF-alpha signaling in sarcopenia are attenuated by life-long calorie restriction. FASEB J. 2005;19(6):668–70.
- Carter CS, Hofer T, Seo AY, Leeuwenburgh C. Molecular mechanisms of life- and health-span extension: role of calorie restriction and exercise intervention. Appl Physiol Nutr Metab. 2007;32(5):954–66.
- Mathias S, Pena LA, Kolesnick RN. Signal transduction of stress via ceramide. Biochem J. 1998;335(Pt 3):465–80.
- Rivas DA, Morris EP, Haran PH, Pasha EP, Morais Mda S, Dolnikowski GG, et al. Increased ceramide content and NFkappaB signaling may contribute to the attenuation of anabolic signaling after resistance exercise in aged males. J Appl Physiol. 2012;113(11):1727–36.
- Hyde R, Hajduch E, Powell DJ, Taylor PM, Hundal HS. Ceramide down-regulates System A amino acid transport and protein synthesis in rat skeletal muscle cells. FASEB J. 2005;19(3):461–3.
- Peterson RT, Desai BN, Hardwick JS, Schreiber SL. Protein phosphatase 2A interacts with the 70-kDa S6 kinase and is activated by inhibition of FKBP12-rapamycinassociated protein. Proc Natl Acad Sci U S A. 1999;96(8):4438–42.
- Lang CH, Frost RA. Sepsis-induced suppression of skeletal muscle translation initiation mediated by tumor necrosis factor alpha. Metab Clin Exp. 2007;56(1):49–57.
- 76. Roubenoff R. Catabolism of aging: is it an inflammatory process? Curr Opin Clin Nutri Metab Care. 2003;6(3):295–9.
- Lupu F, Terwilliger JD, Lee K, Segre GV, Efstratiadis A. Roles of growth hormone and insulin-like growth factor 1 in mouse postnatal growth. Dev Biol. 2001;229(1):141–62.
- Woelfle J, Rotwein P. In vivo regulation of growth hormone-stimulated gene transcription by STAT5b. Am J Physiol Endocrinol Metab. 2004;286(3):E393–401.
- Zadik Z, Chalew SA, McCarter Jr RJ, Meistas M, Kowarski AA. The influence of age on the 24-hour integrated concentration of growth hormone in normal individuals. J Clin Endocrinol Metab. 1985;60(3):513–6.
- 80. Corpas E, Harman SM, Blackman MR. Human growth hormone and human aging. Endocr Rev. 1993; 14(1):20–39.
- Emanuelli B, Peraldi P, Filloux C, Chavey C, Freidinger K, Hilton DJ, et al. SOCS-3 inhibits insulin signaling and is up-regulated in response to tumor necrosis factor-alpha in the adipose tissue of obese mice. J Biol Chem. 2001;276(51):47944–9.
- Hansen JA, Lindberg K, Hilton DJ, Nielsen JH, Billestrup N. Mechanism of inhibition of growth hormone receptor signaling by suppressor of cytokine signaling proteins. Mol Endocrinol. 1999;13(11):1832–43.
- Ram PA, Waxman DJ. SOCS/CIS protein inhibition of growth hormone-stimulated STAT5 signaling by multiple mechanisms. J Biol Chem. 1999;274(50):35553–61.
- Rennie MJ. Anabolic resistance: the effects of aging, sexual dimorphism, and immobilization on human muscle protein turnover. Appl Physiol Nutr Metab. 2009;34(3):377–81.

- Kortebein P, Symons TB, Ferrando A, Paddon-Jones D, Ronsen O, Protas E, et al. Functional impact of 10 days of bed rest in healthy older adults. J Gerontol A Biol Sci Med Sci. 2008;63(10):1076–81.
- Caiozzo VJ, Haddad F, Lee S, Baker M, Paloski W, Baldwin KM. Artificial gravity as a countermeasure to microgravity: a pilot study examining the effects on knee extensor and plantar flexor muscle groups. J Appl Physiol. 2009;107(1):39–46.
- Fiatarone MA, Marks EC, Ryan ND, Meredith CN, Lipsitz LA, Evans WJ. High-intensity strength training in nonagenarians. Effects on skeletal muscle. JAMA. 1990;263(22):3029–34.
- Fiatarone MA, O'Neill EF, Ryan ND, Clements KM, Solares GR, Nelson ME, et al. Exercise training and nutritional supplementation for physical frailty in very elderly people. N Engl J Med. 1994;330(25):1769–75.
- Bodine SC, Stitt TN, Gonzalez M, Kline WO, Stover GL, Bauerlein R, et al. Akt/mTOR pathway is a crucial regulator of skeletal muscle hypertrophy and can prevent muscle atrophy in vivo. Nat Cell Biol. 2001;3(11):1014–9.
- Drummond MJ, Fry CS, Glynn EL, Dreyer HC, Dhanani S, Timmerman KL, et al. Rapamycin administration in humans blocks the contraction-induced increase in skeletal muscle protein synthesis. J Physiol. 2009;587(Pt 7):1535–46.
- Drummond MJ, Dreyer HC, Pennings B, Fry CS, Dhanani S, Dillon EL, et al. Skeletal muscle protein anabolic response to resistance exercise and essential amino acids is delayed with aging. J Appl Physiol. 2008;104(5): 1452–61.
- 92. Kumar V, Selby A, Rankin D, Patel R, Atherton P, Hildebrandt W, et al. Age-related differences in the dose-response relationship of muscle protein synthesis to resistance exercise in young and old men. J Physiol. 2009;587(Pt 1):211–7.
- Kosek DJ, Kim JS, Petrella JK, Cross JM, Bamman MM. Efficacy of 3 days/wk resistance training on myofiber hypertrophy and myogenic mechanisms in young vs. older adults. J Appl Physiol. 2006;101(2):531–44.
- Funai K, Parkington JD, Carambula S, Fielding RA. Age-associated decrease in contraction-induced activation of downstream targets of Akt/mTor signaling in skeletal muscle. Am J Physiol Regul Integr Comp Physiol. 2006;290(4):R1080–6.
- 95. Wilkinson SB, Phillips SM, Atherton PJ, Patel R, Yarasheski KE, Tarnopolsky MA, et al. Differential effects of resistance and endurance exercise in the fed state on signalling molecule phosphorylation and protein synthesis in human muscle. J Physiol. 2008;586(Pt 15):3701–17.
- Mascher H, Andersson H, Nilsson PA, Ekblom B, Blomstrand E. Changes in signalling pathways regulating protein synthesis in human muscle in the recovery period after endurance exercise. Acta Physiol (Oxf). 2007;191(1):67–75.
- Picard M, Ritchie D, Wright KJ, Romestaing C, Thomas MM, Rowan SL, et al. Mitochondrial functional impairment with aging is exaggerated in isolated mitochondria compared to permeabilized myofibers. Aging Cell. 2010;9(6):1032–46.
- Scarpulla RC, Vega RB, Kelly DP. Transcriptional integration of mitochondrial biogenesis. Trends Endocrinol Metab. 2012;23(9):459–66.
- Short KR, Vittone JL, Bigelow ML, Proctor DN, Rizza RA, Coenen-Schimke JM, et al. Impact of aerobic exercise training on age-related changes in insulin sensitivity and muscle oxidative capacity. Diabetes. 2003;52(8): 1888–96.
- St-Pierre J, Drori S, Uldry M, Silvaggi JM, Rhee J, Jager S, et al. Suppression of reactive oxygen species and neurodegeneration by the PGC-1 transcriptional coactivators. Cell. 2006;127(2):397–408.
- Wenz T, Rossi SG, Rotundo RL, Spiegelman BM, Moraes CT. Increased muscle PGC-1alpha expression protects from sarcopenia and metabolic disease during aging. Proc Natl Acad Sci U S A. 2009;106(48):20405–10.
- 102. Russell AP. PGC-1alpha and exercise: important partners in combating insulin resistance. Curr Diabetes Rev. 2005;1(2):175–81.
- Brault JJ, Jespersen JG, Goldberg AL. Peroxisome proliferator-activated receptor gamma coactivator 1alpha or 1beta overexpression inhibits muscle protein degradation, induction of ubiquitin ligases, and disuse atrophy. J Biol Chem. 2010;285(25):19460–71.
- Welle S, Brooks AI, Delehanty JM, Needler N, Thornton CA. Gene expression profile of aging in human muscle. Physiol Genomics. 2003;14(2):149–59.
- Machida S, Booth FW. Increased nuclear proteins in muscle satellite cells in aged animals as compared to young growing animals. Exp Gerontol. 2004;39(10):1521–5.
- 106. Shackelford LC, LeBlanc AD, Driscoll TB, Evans HJ, Rianon NJ, Smith SM, et al. Resistance exercise as a countermeasure to disuse-induced bone loss. J Appl Physiol. 2004;97(1):119–29.
- 107. Symons TB, Sheffield-Moore M, Chinkes DL, Ferrando AA, Paddon-Jones D. Artificial gravity maintains skeletal muscle protein synthesis during 21 days of simulated microgravity. J Appl Physiol. 2009;107(1):34–8.

Chapter 8 Nutrition in the Prevention and Treatment of Cognitive Decline

Grace E. Giles, Kristen E. D'Anci, and Robin B. Kanarek

Key Points

- Epidemiological studies suggest that nutrients including B vitamins, vitamin D, polyphenol, and omega-3 fatty acid intake are associated with enhanced cognitive function in older adults.
- Randomized controlled trials provide mixed evidence that nutritional manipulations slow the progression of age-related cognitive decline.
- Diets such as the Mediterranean diet may be beneficial in slowing cognitive decline.

Keywords Aging • Cognitive function • Alzheimer's disease • B vitamins • Vitamin D • Polyphenols • Fruit • Tea • Chocolate • Omega-3 polyunsaturated fatty acids • Mediterranean diet

Introduction

Aging is often accompanied by cognitive changes, the direction of which depends on the specific cognitive domain. For instance, knowledge and procedural memory remain relatively unimpaired, whereas selective attention, processing speed, working memory, and episodic memory frequently decline with age [1–3]. Cognitive decline not only goes along with normal aging, but also can serve as an early indicator of more severe cognitive problems, including Alzheimer's disease and other forms of dementia [4]. In elderly individuals, the estimated prevalence of mild cognitive impairment is 16 %, while the estimated prevalence of Alzheimer's disease is 13 %, and all forms of dementia is 46 % [5–7].

G.E. Giles, M.S. • R.B. Kanarek, Ph.D. (⊠) Department of Psychology, Tufts University, 490 Boston Avenue, Medford, MA 02155, USA e-mail: grace.giles@tufts.edu; robin.kanarek@tufts.edu

K.E. D'Anci, Ph.D. Department of Psychology, Salem State University, 352 Lafayette Street, Salem, MA 01970, USA

Department of Psychology, Tufts University, 490 Boston Ave, Medford, MA 02155, USA e-mail: kristen.danci@tufts.edu; kdanci@salemstate.edu

Cognitive decline, however, is not an inevitable consequence of aging. Rather, cognitive functioning may be influenced by a number of lifestyle variables, including nutrient intake [8]. For example, as reviewed in the present chapter, there is evidence that individual nutrients (e.g., vitamin B_6 , vitamin B_{12} , folic acid, and vitamin D) as well as foods containing phytochemicals and fatty acids may moderate age-related cognitive decline.

Three types of study designs have been employed to evaluate the influence of nutrition on cognitive aging: cross-sectional, prospective, and randomized controlled trials. Cross-sectional studies evaluate the association between nutrition and cognition at one time point. Thus, these studies provide a momentary picture of the association of nutrient intake on cognitive behavior; however, they cannot provide information on the important question of the long-term consequences of nutrition on cognitive abilities. In comparison, prospective studies determine the role of initial nutritional status on later cognitive outcomes. Using this methodology, the effect of nutrient intake on changes in mental abilities across time can be evaluated [9, 10]. As an example of these methodologies, Kalmijn and colleagues (1997) employed both a cross-sectional study, which evaluated the association between baseline polyunsaturated fatty acid intake and cognitive function in elderly men, and a prospective study, which assessed whether baseline fatty acid intake predicted cognitive decline 1–3 years later [11].

Findings from epidemiological studies are informative because they offer clues to potential associations between nutrition and cognitive behavior. However, as epidemiological designs are correlational by nature, they cannot be used to draw causal conclusions about the effect of a specific nutrient or food on behavior. For example, a positive association between fish intake and cognitive performance cannot be interpreted as meaning that increased fish consumption will improve mental functioning or prevent cognitive decline in the elderly. It is just as possible that elderly individuals who consume more fish have a healthier diet, a greater level of education, and/or are more physically active than those who consume less fish. Second, epidemiological studies often quantify food intake using food frequency questionnaires (FFQ) or 24-h food recalls. However, it is well known that there are wide variations in day to day food intake, and that memory problems, especially in the elderly, may impair accurate recollections of food intake [12, 13]. For this latter reason, some studies have used blood levels of a nutrient or other biomarkers as a measure of nutrient status.

In contrast to epidemiological studies, randomized double-blind, placebo-controlled trials (RCTs) that directly assess the effect of a dietary supplement on behavior have the potential to identify causal links between diet and cognition. However, these studies also have limitations. For example, determining the appropriate amount of a nutrient to study (e.g., vitamin B_6 or folic acid) is often difficult. Very low levels of a nutrient may be associated with difficulties in performing cognitive tasks. As the level of nutrient intake increases, mental functioning improves until optimal performance is achieved. Further, increases of the nutrient may have no effect, or in some cases, may actually lead to toxicity and impairments in performance of cognitive tasks [14]. The duration of the nutrient manipulation, as well as whether the nutrient is given alone or in combination with other nutrients, can also influence the outcomes of RCTs [14].

The study population must also be considered when comparing RCTs assessing the role of a nutrient on cognitive performance in elderly individuals. Across studies, populations vary with respect to a variety of factors, including overall health, medication use, age, degree of cognitive decline, duration of cognitive problems, and the presence of illnesses (e.g., gastrointestinal disorders) or medications that may interfere with nutrient absorption. Another factor to consider when conducting RCTs is the baseline nutritional status of the participants. Some RCTs have investigated the effects of a particular nutrient on cognitive performance in individuals whose intake of the nutrient is at optimal levels, while others have examined the effect of the same nutrient in individuals with suboptimal intakes.

It is important to emphasize that cognitive function is not a unitary concept, but rather can be divided into a number of domains, including attention, episodic memory, semantic memory, spatial orientation, decision making, and executive functioning. Unfortunately, there is little consistency across studies with respect to either the cognitive domains studied, or the specific cognitive tests used [15].

Indeed, in a recent systematic review of 39 RTCs assessing the effects of flavonoids and micronutrients on cognitive performance, Macready and colleagues (2010) reported that 121 different cognitive tests were used. They also noted that while some cognitive domains (e.g., working memory; executive functioning; and global cognitive impairment) were well represented in RCTs, other domains (e.g., psychomotor processing speed; perception; episodic memory; and procedural memory) received much less attention. Additionally, while some researchers explored the effect of a nutrient on only one cognitive domain, others measured nutrient effects on multiple domains using one or more cognitive task [15].

The most common behavioral measure used to assess global cognitive impairment and quantify agerelated cognitive decline is the Mini-Mental State Examination, which combines a number of cognitive domains, including memory, attention, and consciousness among others, into one composite score [16]. This score ranges from 0 to 30, and averages 27 among adults over 65 years old. It is estimated that scores decrease 3–4 points per year following onset of Alzheimer's disease [17]. Although the MMSE has been used extensively in RCTs examining nutrient intake and cognitive functioning, questions have been raised whether the test is sensitive enough to reveal subtle cognitive changes that may be associated with dietary manipulations [15]. Other frequently used cognitive tests include the forward and backward digit span test; Stroop color test; trails test; and the Ray figure task; see [15, 18] for a summary of cognitive measures utilized in geriatric psychiatry.

Finally, it should be remembered that damages to the areas of the brain mediating cognitive behavior may occur much earlier than outward signs of cognitive decline and that by the time the effects of a nutrient are assessed, the brain may be irreversibly altered.

B Vitamins

Support for the importance of the B vitamins, especially vitamin B_6 , vitamin B_{12} , and folate, for normal brain activity was initially provided by research demonstrating that severe deficiencies of these vitamins were associated with impairments in neurological and psychological functioning. While severe B-vitamin deficiencies are rare in the United States, milder, subclinical B-vitamin deficiencies are not uncommon in the elderly due to poor appetite, interactions with medications, and age-related decrements in the absorption of these vitamins [19–23]. It has been hypothesized that these subclinical B-vitamin deficiencies may contribute to age-related decrements in cognitive abilities and the eventual development of dementia by impairing methylation reactions which are critical for normal brain functioning [22, 24–26].

More specifically, vitamins B_6 and B_{12} , and folate are critical components of the one-carbon metabolism cycle. In brief, the essential amino acid methionine is converted to S-adenosyl-methionine (SAM). SAM is the primary methyl donor in a number of reactions, including creatine synthesis, neurotransmitter synthesis and catabolism, and DNA synthesis, and is the sole methyl donor for reactions in the brain [27–29]. Once SAM yields its methyl group, it is converted to S-adenosylhomocysteine (SAH), which is subsequently hydrolyzed into the sulfur-containing amino acid homocysteine. Homocysteine is then recycled into methionine, a methylation reaction requiring both folate and B₁₂, or catalyzed into the amino acid cysteine via B₆-dependent reactions. In folate or vitamin B_{12} deficiency, remethylation of homocysteine to methionine and its subsequent conversion to SAM are impaired. Vitamin B_{12} and folate are closely linked in the methylation process, such that a B₁₂ deficiency can lead to a secondary folate deficiency. To convert homocysteine to methionine, vitamin B_{12} binds to methionine synthase, and accepts a methyl group from folate (as 5-methyl tetrahydrofolate [5-methyl THF]). Without vitamin B₁₂, 5-methyl THF becomes "trapped" and is not available for reactions (i.e., DNA synthesis) further downstream [30]. With vitamin B₆ deficiency, homocysteine clearance is also impaired. Thus, high levels of homocysteine are seen with deficiency of folate, vitamin B_6 and/or B_{12} .

It has been hypothesized that an elevated level of homocysteine can be neurotoxic and can contribute to the etiology of cerebrovascular disease, including stroke and vascular dementia (e.g., [25, 31, 32]). In support of this hypothesis, a number of studies, but not all (e.g., [33]), have found that individuals with high levels of homocysteine display greater age-related cognitive deficits and a higher risk of dementia than those with lower levels [25, 34–37]. Moreover, several cross-sectional studies have found that patients with Alzheimer's disease have significantly higher total homocysteine than either age-matched hospitalized controls or healthy community-dwelling elderly individuals [38–40]. For example, in one longitudinal study, higher levels of homocysteine in Alzheimer's patients were associated with a greater progression of hippocampal atrophy and a greater decline in Mini Mental State Evaluation scores [41]. Additionally, recent studies using magnetic resonance imaging of healthy participants have found an inverse relationship between plasma homocysteine levels and brain size [42].

Initial epidemiological evidence supporting a relationship between vitamin B status and cognitive function in older individuals was provided in 1983 by Goodwin and colleagues who reported that healthy elderly individuals with low intakes or blood levels of vitamin B_{12} , folate, vitamin E, and riboflavin did more poorly on tests of memory and abstract thinking than elderly individuals with higher intakes or blood levels of these vitamins [43]. A number of subsequent cross-sectional studies have confirmed this initial finding, with individuals with low vitamin B status with or without high homocysteine levels, performing more poorly on cognitive tasks than their more adequately nourished counterparts [34, 44–47]. Additionally, other cross-sectional studies have observed that patients with Alzheimer's disease have lower serum concentrations of B vitamins than those without the disease and further that the severity of the deficiency is associated with the severity of the disease [25, 26, 35, 48–55].

Results of several prospective studies point to an inverse relationship between vitamin B status and cognitive decline with individuals with initially lower blood levels of B vitamins displaying a greater risk of subsequent cognitive impairment and dementia than those with higher blood levels of the vitamins [26, 35]. For example, Tucker and colleagues (2005) reported that over a 3-year period, low baseline levels of B vitamins predicted cognitive decline [47]. However, other studies have found no association between vitamin B status and the risk of subsequent dementia [56–58].

It is important to stress that it is difficult to compare the effects of B vitamin supplementation in mediating age-related cognitive decline across studies as studies have varied widely in a number of variables such as the exact composition of the vitamin supplements; the duration of supplementation; the cognitive tests used; the participant's age, health, nutritional status, and other confounding variables (e.g., socioeconomic status; smoking and physical activity) [58]. Additionally, recent work has suggested that genetic factors may moderate the interaction between cognitive performance and B vitamin status. For example, Vogiatzoglou et al. (2013) found that among well-nourished healthy elderly, the detrimental effects of low levels of vitamin B_{12} on cognitive performance were more pronounced in individuals with the ApoE ε 4 allele, which is the first identified genetic risk factor for Alzheimer's disease [59].

While it has yet to be determined whether homocysteine is causative in cognitive decline, the relationship between B vitamins, homocysteine, and cognition has been intensively studied and reviewed elsewhere [25, 31, 32]. Generally, low B vitamin status with or without high homocysteine is associated with poorer cognitive outcomes [34, 44–47] as well as Alzheimer's disease [52–55] Although a number of studies have shown an inverse relationship between B vitamin status and homocysteine levels, and a positive relationship between B vitamin status and better cognitive performance in older individuals, not all studies show a clear relationship (for review see [57]). However, newer work examining gene–nutrient interactions has cast new light on factors that play an important role in the relationship between B vitamin status and cognitive performance. Discussed below are some of the more widely studied gene–nutrient interactions, but the field continues to evolve.

Interaction Between APOE E4 Genotype and B Vitamins

Carriers of the apolipoprotein E (APOE) ε 4 allele have an increased chance of developing Alzheimer's disease. Presence of the APOE ε 4 allele is also associated with earlier onset of cognitive impairment. However, presence of the APOE ε 4 allele does not guarantee that an individual will inevitably develop Alzheimer's disease. Lifestyle factors such as diet, exercise, and smoking can influence the etiology and progression of the disease.

More recent work has examined the relationship between the APOE ϵ 4 genotype, B vitamin status, and Alzheimer's disease. Some of these studies suggest that there is an interaction between APOE ϵ 4 status and B vitamin status in cognitive performance such that carriers of the APOE ϵ 4 allele may be at greater risk for the negative consequences of high homocysteine or low B vitamin status. It is not clear, however, whether the relationship between APOE ϵ 4 status, B vitamin status, and cognitive performance is directly related to brain function, or if the relationship is more indirectly a result of increased risk for cardiovascular disease [60, 61].

Cross-sectional data show that in carriers of the APOE $\varepsilon 4$ allele, higher vitamin B₁₂ status is more strongly associated with cognitive performance relative to noncarriers [62] and that high homocysteine in APOE $\varepsilon 4$ carriers is more strongly associated with poorer cognitive function relative to noncarriers [63]. In the Hordaland Homocysteine Study [59], vitamin B₁₂ status was positively associated with cognitive performance. In particular, higher levels of vitamin B_{12} were associated with better global cognition in APOE $\varepsilon 4$ carriers relative to noncarriers. However, longitudinal studies show a much weaker outcome dependent upon APOE $\varepsilon 4$ genotype. While Brown et al. (2011) showed a cross-sectional relationship among APOE $\varepsilon 4$ status, B vitamin status, and cognitive function, after 7 years follow-up, there was only a significant relationship between APOE $\varepsilon 4$ carriers, low vitamin B₆ status, and a verbal cognitive task [64]. Other research demonstrated that although presence of the APOE ε 4 allele, elevated homocysteine, and low folate status were more frequently observed in patients with mild cognitive impairment relative to matched controls, these factors were not predictive of subsequent progression to dementia [61]. Presence of the MTHFR C677T polymorphism, discussed in more detail below, may also play a significant role in the relationship between APOE $\varepsilon 4$ genotype and B vitamin status. While the distribution of MTHFR genotype does not appear to be related to APOE $\varepsilon 4$ genotype, some research suggests that presence of both the APOE $\varepsilon 4$ allele and MTHFR TT polymorphism may aggravate the impact of B vitamin deficiency. For example, in comparison with patients with mild cognitive impairment or matched controls, patients with Alzheimer's disease have higher levels of homocysteine, and this observation is stronger in the presence of the MTHFR TT genotype and low folate status [65]. Additionally, some research shows that while total homocysteine may be elevated in patients with Alzheimer's disease, there may be a stronger relationship between high LDL cholesterol levels and cognitive performance in Alzheimer's disease than B vitamin levels [60]. These studies indicate a need for further elucidation of the nature of the relationship between APOE ε 4 genotype, nutritional status, and progression of cognitive decline.

Methylenetetrahydrofolate Reductase (MTHFR) Genotype

Recent research examining the gene–nutrient interaction between folate status and polymorphisms in the methyltetrahydrofolate reductase (MTHFR) gene has shown that individuals with the MTHFR 677T allele (CT or TT) are less efficient in utilizing available folate, specifically in the reduction of 5,10-methylene THF to 5-methyl THF, and thus may have higher folate requirements than those with

the MTHFR 677 CC genotype. In the United States, where folic acid fortification is mandatory, rates of folate deficiency are relatively low. However, carriers of the TT polymorphism have impaired homocysteine clearance that is strongest in the presence of low folate status [66].

A number of epidemiological studies provide data suggesting that poor vitamin B status could contribute to the development of age-related dementias. However, results of randomized-controlled trials comparing the cognitive consequences of vitamin B supplementation to placebo have shown limited utility of the use of B vitamin therapy in combating age-related cognitive decline. Moreover, although the links between high-homocysteine and cognitive impairment have been extensively examined, the evidence is incomplete. At present the data support the idea that high-homocysteine, rather than being causative in the etiology of disease, may serve instead as a biomarker for other pathological processes. Finally, as the science evolves, we are gaining a better understanding about the importance of genetic factors in the larger arena of B vitamins.

Vitamin D

Vitamin D, sometimes known as the "sunshine vitamin," is a fat-soluble steroid hormone naturally synthesized from 7-dehydrocholesterol when ultraviolet-B rays from the sun reach the skin. Small amounts of vitamin D are also obtained from the diet from fatty fish (e.g., salmon, mackerel, and tuna), fish oils, beef liver, and egg yolks, and from dairy foods and cereals specifically fortified with the vitamin. In addition, many individuals obtain vitamin D in the form of dietary supplements [67]. Whether obtained from sunlight or intake of food or supplements, vitamin D is physiologically inert until it is converted to 25-hydroxyvitamin D[25(OH)D] in the liver, and subsequently to the active form of the vitamin 1,25-dihydroxyvitamin D[1,25(OH)₂D], primarily in the kidney. As it represents a reliable and relatively stable measure of bioavailability, serum level of 25(OH)D expressed as either ng/mL or nmole/L is typically used as an indicator of vitamin D status. However, it should be noted that techniques for measuring vitamin D have not been consistent across countries, which has confounded efforts to develop international guidelines for the vitamin [68].

Currently, there is no clear consensus on the optimum serum level of 25(OH)D. However, most experts have concluded that a minimum level of between 50 and 75 nmol/L 25(OH)D is needed for good health [69–71]. Using these values, a significant proportion of the world's population has lower than the recommended levels of vitamin D (hypovitaminosis D). Older individuals are particularly at risk for hypovitaminosis D as a result of (1) consuming a diet with less than adequate vitamin D, (2) having a reduced capacity to metabolize the vitamin, and (3) having limited exposure to sunlight [67, 69, 71, 72].

Vitamin D is best known for its role in building and maintaining strong bones. The vitamin facilitates calcium absorption from the gastrointestinal tract and maintains sufficient serum levels of calcium and phosphate for normal mineralization of bone. In children, vitamin D deficiency is associated with rickets, a disease characterized by soft bones and skeletal abnormalities while in adults, deficiency of the vitamin can result in osteomalacia whose symptoms include bone pain as well as bone and muscle weakness [67, 71]. Recent research has highlighted the fact that vitamin D deficiency also increases the risk for a number of age-related diseases, including cardiovascular disease, metabolic syndrome, type 2 diabetes, stroke, and cancer [70, 73, 74].

Of particular import to the present chapter, is the mounting evidence demonstrating that vitamin D is critical for the normal functioning of the nervous system, and that vitamin D deficiency may contribute to age-related decrements in cognitive behavior. Initial indications of the importance of vitamin D in modulating nervous system activity came from research describing vitamin D receptors (VDR) and the enzymes necessary for the synthesis of the active form of the vitamin throughout the brain and spinal cord. VDR expression is found in both neurons and glial cells particularly in areas of the brain associated with cognitive behavior including the cortex, hippocampus, thalamus, basal ganglia, hypothalamus, and amygdala [75].

Support for the importance of VDR for brain functioning is provided by studies demonstrating that animals born to mothers who were vitamin D deficient during pregnancy displayed morphological changes in the brain including larger brains, cortical layer thinning and larger than normal lateral ventricles relative to control animals [76]. These alterations in brain structure were associated with discrete changes in learning and memory such as impaired latent inhibition and reduced habituation [77]. Similarly, VDR knockout mice display abnormalities in brain function and cognitive behavior including high anxiety, abnormal grooming behavior, impaired nest building, and aberrant maternal behaviors [78, 79].

Vitamin D plays a variety of roles within the nervous system. For example, the vitamin is important for mediating the synthesis of a number of neurotransmitters (e.g., dopamine, serotonin, catecholamines, and acetylcholine) and for up-regulating the synthesis of several neurotrophic factors (e.g., nerve growth factor and glial cell line derived neurotrophic factor) which promote the development and survival of neurons and glial cells [75, 80]. Studies in experimental animals demonstrating that vitamin D administration (1) decreases neuronal injury resulting from neurological trauma, and (2) protects against stress-induced apoptosis in hippocampal neurons have helped to establish a neuroprotective role for the vitamin. Additionally, vitamin D has both antioxidant and anti-inflammatory properties. With respect to aging, it has been proposed that vitamin D protects against age-related inflammatory changes in the hippocampus and reduces hippocampal biomarkers of aging [81, 82]. Moreover, recent work has demonstrated that injections of vitamin D not only mitigate age-related changes in inflammatory state, but also decreases the memory impairments seen in aged animals [83]. Vitamin D also strongly stimulates phagocytosis and augments the removal of amyloid β , one of the characteristic pathological biomarkers found in the brains of individuals with Alzheimer's disease [75, 80, 83, 84].

Results of a number of cross-sectional studies have found that low serum vitamin D levels are associated with impairments in cognitive functioning in older individuals (e.g., [85–89]). For example, Annweiler and coworkers (2010) observed that older community-dwelling French women with lower levels of serum 25(OH)D displayed more cognitive deficits than women with higher serum 25(OH)D levels [90]. In a subsequent study, the same research group, noted that older men and women with mild cognitive impairment (MCI) had lower serum 25(OH)D concentrations than their cognitively healthy counterparts [91]. Further support for the potentially detrimental cognitive effects of hypovitaminosis D is provided by a recent systematic review by van der Schaft and colleagues (2013) who reported that in 18 out of 25 cross-sectional studies, individuals with low vitamin D levels did more poorly on tests of cognitive functioning or displayed a higher frequency of dementia than individuals with higher serum levels of the vitamin [92]. Research demonstrating that individuals with Alzheimer's disease have lower serum levels of 25(OH)D than suitably matched individuals without the disease suggest that hypovitaminosis D could ultimately play a role in the etiology of Alzheimer's disease [93–98].

Prospective studies have examined whether initial vitamin D status predicts later development of cognitive deficits and the risk of suffering from dementia. In the majority of these studies, over time, participants with initially low vitamin D intake or serum levels of 25(OH)D displayed significantly greater declines in performance on cognitive tasks and/or a higher frequency of dementia than participants who began the study with better vitamin D status [92, 97, 99, 100].

While the results of cross-sectional and prospective studies suggest that impaired vitamin D status may contribute to cognitive decline in elderly individuals, as noted above, these types of studies cannot verify causality. Rather than contributing to the development of age-related cognitive decline and Alzheimer's disease, it is possible that vitamin D deficiency represents a consequence, rather than a cause, of cognitive deficits in elderly individuals. Age-related degenerative illnesses, such as Alzheimer's disease are often marked by a reduction in mobility and outdoor activity, and thus a decrease in exposure to sunlight. Moreover, as these diseases progress many patients experience feeding difficulties leading to a decrease in dietary intake of vitamin D, or are taking medications which interfere with absorption of the vitamin [97].

While a number of epidemiological studies suggest a relationship between vitamin D levels and mental functioning, the results of RCTs investigating the effects of vitamin D on cognitive behavior have been more mixed. In an early RCT, Dhesi and colleagues (2004) found that relative to a placebo injection, a single intramuscular injection of 600,000 IU vitamin D increased serum levels of the vitamin and improved reaction time in a choice situation in individuals who had a history of falling and a low baseline serum level of 25(OH)D [101]. More recently, Annweiler et al. (2012) observed that older outpatients given vitamin D_3 supplements for 16 months had significantly higher serum levels of 25(OH)D and better cognitive performance, particularly executive functioning, than controls given a placebo [102]. In contrast, Przybelski and coworkers (2008) reported that while oral administration of 50,000 IU of vitamin D three times a week for 4 weeks increased serum 25(OH)D levels in nursing home residents, vitamin D supplementation did not lead to improvements in either cognitive or motor behavior [103]. As intranasal insulin can acutely improve cognition and vitamin D enhances insulin actions in the brain, Stein et al. (2011) explored the effects of a high dose of vitamin D_2 followed by intranasal insulin in community dwelling participants with mild to moderate Alzheimer's disease [104]. All participants received a low-dose of insulin (1,000 IU) for 16 weeks, followed by either a high-dose of vitamin D or placebo for 8 weeks. While serum levels of 25(OH)D increased following the high dose of vitamin D, the high dose of vitamin D provided no benefits in cognitive behavior relative to the low dose of the vitamin. In the largest study to date, healthy, elderly women from 40 Women's Health Initiatives clinics were randomized to receive either 400 IU vitamin D in combination with 1,000 mg calcium or placebo. After a mean follow-up of 7.8 years, no differences in the incident of mild cognitive impairments or in cognitive functioning on a number of domain-specific cognitive functions were found between the two groups [105].

While RCTs have not consistently supported the concept that vitamin D supplementation has a positive effect on cognitive functioning in the elderly, these types of studies should not be ignored. Rather, more studies systematically assessing variables such as the dose and duration of supplementation, and sex and age should be conducted. Moreover, these studies should begin by investigating the effect of a nutrient in individuals with less than adequate nutrient intake [14].

Polyphenols

Polyphenols are organic compounds found in a variety of foods most notably fruits, vegetables, and beverages. Among others actions, these compounds have antioxidant, antibiotic, and anti-inflammatory properties [106]. The fact that polyphenols reduce oxidation within the nervous system is of particular interest to cognitive aging, as oxidative stress is believed to play a role in the etiology of Alzheimer's disease [107]. Additionally, these compounds stimulate neurogenesis and neuronal regeneration and can protect neurons from injury induced by neurotoxins [2, 108].

There is growing evidence that fruits and vegetables containing polyphenols can counteract agerelated declines in cognitive functioning [109, 110]. Indeed, a recent cross-sectional study found that older adults with "questionable dementia," whose cognitive scores fell between cognitively normal and mild dementia, reported lower appetite and lower intakes of total food, fruits and vegetables, and fluids than cognitively normal older adults [111]. A second cross-sectional study evaluating the relationship between plant food consumption and performance on a number of cognitive tests, including the Mini-Mental State Examination as well as tests measuring memory, executive function, perceptual speed, and visuospatial skills found that older adults who ate more vegetables performed better on tasks measuring executive function, perceptual speed, and semantic memory while those who ate more fruit performed better on tasks measuring visuospatial skills and both episodic and semantic memory. Both fruit and vegetable consumption conferred benefit on Mini-Mental State Examination scores [112]. Unlike cross-sectional studies which are limited to one time point, prospective studies allow us to determine whether higher intake of fruits and vegetables predicts cognitive performance over time. In support of a positive role for foods containing polyphenols in preventing the development of age-related cognitive decline, two studies found that although total fruit intake was unrelated to cognitive decline, higher intake of vegetables, especially green leafy vegetables, predicted reduced cognitive impairment over 6 years [113, 114]. Similar results were found in a 5-year prospective study, in which vegetable intake, particularly root vegetables and cabbage, was associated with a reduced decline in information processing speed. Again, no effect was found for fruit intake [115]. However, other studies have found conflicting results. For example, Peneau and colleagues (2011) reported that while higher combined intake of fruits and vegetables, as well as total intake of fruit at baseline predicted enhanced verbal memory 13 years later, in contrast, however, they found that higher combined intake of fruits and vegetables and total vegetable with lower scores on tests of executive function [116].

Together the previous studies provide preliminary evidence that global polyphenol intake may relate to cognitive performance in older individuals, but the cross-sectional and prospective nature of the studies makes it impossible to determine whether polyphenols play a causal role in cognitive aging. This limitation pertains not only to the aforementioned studies, but also to the majority of those in the subsequent sections, as most research on polyphenols and aging has utilized epidemiological designs. However, a number of RCTs have assessed the efficacy of polyphenol-rich foods including blueberries, grapes, and ginkgo biloba extract on cognitive aging, including Alzheimer's disease.

Berries and Grapes

There is now substantial evidence that consuming foods containing phytochemicals such as those found in berries can help to reduce age-related memory loss. Initial support for this idea came from preclinical studies demonstrating that prolonged intake of blueberries reduced a number of the cognitive impairments typically seen in older laboratory animals [117]. It has been hypothesized that these beneficial effects are related to the polyphenolic compounds found in blueberries and other fruits. These compounds, such as the anthocyanins, which are responsible for the color of these fruits, cross the blood brain barrier to become localized in areas of the brain related to learning and memory (e.g., hippocampus). In the brain, these phytochemicals decrease vulnerability to the oxidative stress that occurs with aging, reduce inflammation, and may increase neuronal signaling [2, 118].

Recent studies indicate that intake of berries may have positive effects on cognitive aging in humans as well as laboratory rodents. For example, Devore et al. (2012) reported that higher consumption of blueberries and strawberries over a 6-year period predicted a slower decline in global cognitive and verbal function in older women [118].

Randomized controlled trials evaluating the efficacy of fruit juice, specifically cranberry, blueberry, and grape juice, on cognitive performance in older adults offer promising evidence for a beneficial effect of fruit juice on cognitive function. For instance, 6 weeks of blueberry juice supplementation in older adults showing signs of memory decline but not dementia resulted in improvement in associative learning and recall relative to placebo [119]. Similarly, 12 weeks of grape juice supplementation in individuals with memory decline but not dementia enhanced verbal learning and retention relative to placebo [120]. The same group found no differences in learning and retention as a result of 16 weeks of grape juice supplementation, but did find reduced semantic interference during recognition memory in individuals with mild cognitive impairment [121]. However not all studies have found beneficial effects of juice consumption on behavior. In particular, 6 weeks of cranberry juice supplementation in healthy older adults had no influence of cognitive measures of memory, intelligence, attention, or executive function [122]. Taken together, epidemiological studies and RCTs provide preliminary support for concluding that fruit, especially berries and grapes, impede cognitive aging (for review, see [2]).

Green Tea

Green tea is a rich source of another class of polyphenols, called flavanols. Catechins, a subclass of flavanols, comprise approximately 30 % dry weight of the tea leaf [123]. Green tea contains a higher concentration of polyphenols than black or oolong tea because unlike black or oolong tea, its preparation does not result in oxidation of leaf polyphenols. In Japanese adults over the age of 70 years, intake of green tea, but not black or oolong tea, was associated with less cognitive impairment, evidenced by higher scores on the Mini Mental State Examination [124]. In comparison, green, black, and oolong tea predicted a slower reduction in scores over a 1- to 2-year follow-up, possibly owing in part, to greater consumption of black and oolong tea than green tea among older individuals [125]. Green tea consumption was also associated with lower psychological distress [126]. Thus, epidemiological evidence suggests that green tea consumption may reduce the risk of cognitive decline, yet RCTs are necessary to verify the causal nature of this relationship.

Chocolate

Chocolate and other cocoa products are becoming recognized as a rich source of flavanols with potent antioxidant and anti-inflammatory actions [127]. While most research has concentrated on the effects of chocolate on the cardiovascular system, there are a growing number of RCTs aimed at appraising the effects of chocolate intake on cognitive functions. The majority of these RCTs have focused on healthy young adults. These trials provide conflicting results, with some showing no impact of chocolate on cognitive behavior [128, 129] and others finding improvement in mental fatigue, sustained attention, and working memory [130, 131]. Research in older adults is similarly inconsistent. Flavanoid- and procyanidin-rich dark chocolate had no effects on a wide range of neuropsychological tests in healthy older adults [132]. However in older adults with mild cognitive impairment, flavanol-rich cocoa improved executive function, verbal memory, and processing speed [133], suggesting that cocoa may have more beneficial effects in individuals who have experienced cognitive impairments than those who are cognitively healthy.

Total Polyphenol Intake

Although most studies have focused on individual classes or sources of polyphenols, recent work has evaluated the association between cognition and polyphenol intake across multiple classes of polyphenols. The first study used a cross-sectional design to assess the correlation between flavanoid intake and cognitive performance among older men and women [134]. Flavanoid intake was determined using a food frequency questionnaire. Cognitive measures, including the Moray House Test, a measure of intelligence and verbal reasoning, the National Adult Reading Test, the Wechsler Adult Intelligence Scale-III, and the Mini-Mental State Examination, were grouped into three factors: memory, intelligence, and processing speed. Intake of a number of flavanoid-rich foods was associated with improved cognitive performance, including total fruit, citrus fruit, total vegetable, red wine, and chocolate. However, these effects were no longer significant after adjusting for childhood IQ and socioeconomic status.

The second study evaluated the relationship between polyphenol consumption at age 66 and cognitive performance 13 years later [135]. Analysis of all cognitive measures factored into two cognitive components: (1) language and verbal abilities, which were positively associated with intake of total polyphenols and total flavanoids, as well as with a number of specific classes of these compounds including catechins and flavanols and (2) executive function, which was not correlated with total polyphenol or flavanoid intake and inversely associated with intake of foods containing catechins, proanthocyanids, and flavanols.

One final study sought to determine the association between chocolate, wine, and tea intake and cognitive function on a number of cognitive tasks including the Mini-Mental State Examination as well as others assessing episodic memory, attention, perceptual speed, and visuospatial skills [136]. Chocolate, wine, and tea intake, alone and together, were associated with enhanced cognitive function across all measures, and wine intake conferred the strongest protection against poor cognitive performance.

Monounsaturated and Polyunsaturated Fatty Acids

Omega-3 (n-3) and omega-6 (n-6) fatty acids comprise the family of polyunsaturated fatty acids. n-6 PUFAs include linoleic acid (LA) and arachidonic acid (ARA) whereas n-3 fatty-acids include alpha linolenic acid (ALA), eicosapentaenoic acid (EPA), and docohexaenoic acid (DHA) [137]. Our current western diet is comprised of a ratio of between 10:1 and 17:1 n-6 to n-3 PUFA. These ratios represent a shift in our diet from consumption of n-3 PUFA rich foods to n-6 PUFA r rich foods. Foods containing n-3 PUFAs include fish, fish oil, wild game, wheat germ, walnuts, and plants. Foods high in n-6 PUFAs include common vegetable oils, including corn, safflower, sunflower, and soybean oil [138, 139].

A host of research has been dedicated to determining whether polyunsaturated fatty acids (PUFA) influence cognitive aging [140–143]. Epidemiological studies assessing the influence of n–3 PUFAs on cognitive decline in older adults generally measure dietary intake of foods containing fatty acids and/or plasma or erythrocyte levels of the fatty acids. Associations with cognitive performance are then determined at one or more time points. The most common behavior measure used in these studies has been the Mini Mental State Exam, which is often used to characterize mild cognitive impairment or cognitive decline on a discrete yes-or-no basis.

The majority, although not all, of epidemiological studies have found positive relationships between intake of n-3 PUFA and/or plasma erythrocyte levels of n-3 fatty acids and cognitive function in older adults [144, 145]. For instance, intake of fish (especially nonfried fish, fatty fish, and tuna fish), fish oil, and n-3 PUFA supplements has been related to reductions in age-related cognitive impairments [11, 146–150]. This relationship is particularly strong in individuals who do not carry the APOE $\varepsilon 4$ allele, a primary risk factor for Alzheimer's disease [147].

Other studies have computed n-3 and n-6 PUFA intake from dietary records and found that higher total n-3 PUFAs, EPA and DHA and lower total n-6 PUFAs and n-6:n-3 ratios are associated with reduced cognitive decline [151–153]. Only one study thus far has found n-3 PUFA intake to be unrelated to cognitive function [145].

Several researchers have reported that lower levels of plasma and/or erythrocyte n-3 fatty acids were related to poorer cognitive performance and increased risk of cognitive decline or dementia [154–157], again with evidence suggesting that the association is particular to individuals without the *APOE* ϵ 4 allele [157]. Further support for a relationship between PUFAs and mental functioning is provided by research demonstrating that erythrocyte EPA was lower and n-6 PUFAs were higher in older adults with mild cognitive impairment than in healthy individuals [158]. Although another study found no associations between plasma n-3 and global cognitive decline, higher plasma n-3 levels were associated with a reduced decline in verbal fluency in individuals with higher hypertensive and dyslipidemic markers and lower depressive symptoms [159]. It should be noted that a few studies have failed to find relationships between PUFAs and cognitive performance [160–162].

RCTs assessing the influence of n-3 PUFA supplementation on cognitive function in healthy older adults have shown that DHA (800–900 mg/day) can improve verbal, visuospatial, and episodic memory, and higher serum and plasma DHA were generally associated with improved cognitive scores on these measures [163, 164]. However a lower dose of DHA (i.e., 252 mg/day) did not influence cognitive function [165]. Two doses of EPA and DHA (226 mg EPA + 176 mg DHA and 2,093 mg EPA + 847 mg DHA) improved attention relative to placebo, but this effect was limited to *APOE* $\varepsilon 4$ carriers [166]. Thus randomized controlled trials in healthy older adults generally suggest that n-3 PUFAs improve cognitive function.

n-6 PUFAs can augment β -amyloid deposition, a primary neurological symptom of Alzheimer's disease, while n-3 PUFAs (or low n-6 to n-3 ratios) may reduce the effects of n-6 PUFAs on β -amyloid deposition [167]. These findings suggest that supplementation with n-3 PUFA may have beneficial cognitive functions in individuals with mild cognitive deficits or Alzheimer's disease. In support of this section, n-3 PUFAs reduced memory and attention decrements in older adults with mild Alzheimer's disease and reduced memory impairments in individuals with more severe Alzheimer's disease [168]. Similarly, n-3 PUFA supplementation improved global clinical status in individuals with mild to moderate Alzheimer's disease. Moreover, the higher proportion of red blood cell membrane EPA, the greater the reduction in cognitive deficits [169]. DHA-rich fish oil enhanced verbal fluency in older adults with mild cognitive impairment, and both EPA- and DHA-rich fish oil improved depressive symptoms [170]. In contrast, other work showed no influence of n-3 PUFA supplementation in individuals with mild to moderate Alzheimer's disease [171].

Whereas the majority of studies focus on polyunsaturated fatty acids, and generally use monounsaturated fatty acids as controls, one study evaluated the relationship between olive oil use and cognitive decline, including tests of global cognitive decline, visual memory, and verbal fluency [172]. At baseline, high or moderate olive oil intake was correlated with better visual memory and verbal fluency but not related to global cognitive function. Across 4 years, "intensive" olive oil intake (i.e., using olive oil both in cooking and dressing foods, rather than one or the other) predicted reduced cognitive decline in visual memory and verbal fluency, but again no effect on global cognitive function.

Epidemiological results showing that higher n-3 PUFA intake and plasma levels are associated with reduced overall cognitive decline are supported by some RCTs showing that n-3 PUFA supplementation, particularly DHA, reverses age-related cognitive decline in otherwise healthy individuals [163, 164, 166], but there is less evidence to suggest such an effect in individuals with mild cognitive impairment and Alzheimer's disease. Alzheimer's disease risk factors included history of depression and carrying the *APOE* $\varepsilon 4$ gene may influence n-3 PUFAs' efficacy in preventing cognitive decline in older adults [147, 152, 157].

Mediterranean Diet

Dietary consumption of polyphenols varies geographically, ranging from approximately 800 mg/day in Finland and Spain [173, 174] to 1,000 mg/day in Poland [175]. Polyphenol intake may be particularly high in Mediterranean countries [174, 176] which rely heavily on polyphenol-rich foods including fruits, vegetables, and legumes. These foods, along with cereals, low-fat dairy products, olive oil, spices, onions, fish, and white meat, as well as wine in moderation, constitute the "Mediterranean Diet," which may confer health benefits over diets with less emphasis on plants and unsaturated oils [177]. Studies that examined the relationship between the Mediterranean Diet and cognitive aging typically utilize a 0–9 point scale, with higher scores indicating greater adherence to the Mediterranean diet [178]. Beneficial food groups, including fruits, vegetables, legumes, cereals, and fish are assigned a 0 for intake below the median and 1 for intake above the median, whereas detrimental food groups, including meat, full-fat dairy, and ratio of monounsaturated to saturated fat, are assigned a 1 for intake below and 0 for intake above the median. Moderate alcohol intake is assigned a 1 while low and high alcohol intake is assigned a 0.

In individuals free of cognitive impairment at baseline, overall adherence to the Mediterranean diet was not associated with developing mild cognitive impairment over a 4 year period. Indeed, counter to expectation, higher intake of monounsaturated fats and fish predicted greater cognitive decline [179]. However, analyses did not take into account fish type or preparation. Other studies have evaluated cognitively healthy individuals at high cardiovascular risk. In a cross-sectional analysis, high consumption of olive oil, coffee, walnuts, and wine as well as total urinary polyphenol excretion was associated with enhanced memory and global cognitive function [180]. In contrast, a prospective study carried over a 5-year period found no evidence that adherence to the Mediterranean diet predicts cognitive decline among individuals with vascular disease or at risk for vascular disease [181].

Results of studies assessing the effects of the Mediterranean diet on cognitive behavior in individuals with Alzheimer's disease have been inconsistent. For example, Scarmeas and colleagues (2007) reported that in older adults with Alzheimer's disease at baseline, adherence to the Mediterranean diet was associated with reduced mortality over a 4-year period [182]. Moreover, higher Mediterranean diet scores were associated with slower global cognitive decline on the Mini-Mental State Examination in older adults with dementia, but not with tests of specific cognitive domains including verbal fluency and episodic memory [183]. It is important to note that in studies which assess adherence to the Mediterranean diet and, thus, results may not be representative of older adults who fully follow the Mediterranean diet. For example, despite similar overall adherence to the Mediterranean diet, individuals in France [183] consumed more fruits and vegetables than those in the United States [184].

In a comparison of older adults cognitively healthy at baseline to those with mild cognitive impairment, greater adherence to the Mediterranean diet was associated with reduced progression from mild cognitive impairment to Alzheimer's disease, as well as marginally significant reduction in procession from healthy cognition to mild cognitive impairment [184]. Also in older adults with and without mild cognitive impairment, higher intake of vegetables, unsaturated to saturated fatty acid ratio and moderate alcohol was associated with reduced risk of mild cognitive impairment, specifically in individuals with impairment in memory, but not individuals with impairments in nonmemory cognitive domains [185]. Finally, the Mediterranean diet but not another measure of healthy diet, i.e., Healthy Eating Index, which is based on recommendation from the Dietary Guideline for Americans, was protective against cognitive decline [186]. Thus the majority of studies found some protective effect of the Mediterranean diet against age-related cognitive decline in healthy older adults [180, 186] as well as those with cognitive impairment or Alzheimer's disease [183–185, 187].

Thus, together with evidence that individual components of the Mediterranean diet, including polyphenol-rich fruits, vegetables, and red wine and n-3 PUFA-rich fish, enhance cognitive function, adherence to the Mediterranean diet may slow the progression of age-related decline. However, the timeline with which nutrition may influence cognitive aging remains relatively unexplored, as the majority of evidence for such a relationship stems from epidemiological studies. Future research should seek to better understand whether adherence to the Mediterranean diet and intake of other macro- and micronutrients discussed in this chapter are necessary in early adulthood to protect against cognitive decline, or whether these nutritional manipulations can slow cognitive decline beginning in old age.

Conclusion

As noted in the introduction, a number of variables should be considered when evaluating the role of vitamins and other nutrients on cognitive behavior in the elderly: age, nutrient status, combination of nutrients, amount of nutrients in RCTs, cognitive tasks, number of participants, and measurements of vitamin status [58]. Additionally, variables in cohort studies, including physical activity, socioeconomic status, comorbid diseases, and genetic factors such as the *APOE* ϵ 4 allele, may confound results.

Nevertheless, the evidence reviewed here that nutrition may influence cognitive function in older adults, and in some cases slow the progression of cognitive aging, as in the case, for example of the Mediterranean diet. Nutrition is of particular interest in older adults, as food intake generally declines with age, often resulting in poor protein, vitamin, and mineral consumption [188].

For instance, high homocysteine and/or low vitamin B status was associated with reduced cognitive scores and increased risk of cognitive decline, particularly in individuals with the *APOE e*4 allele [34–37, 43–47, 59–61]. However, results from RCTs do not present such convincing evidence, suggesting that elevated homocysteine may serve as a biomarker, rather than cause, of age-related cognitive decline. Likewise, serum vitamin D levels and intake have been associated with reduced cognitive decline (for review, see [92]). But, whereas some RCTs found beneficial cognitive effects of vitamin D supplementation [101, 102], others did not [103–105].

Polyphenols, n-3 PUFA, and diets heavy in polyphenol-rich fruits and vegetables, i.e., Mediterranean diet, are also associated with reduced cognitive decline in older adults [183–187]. Polyphenol-rich fruit and vegetable intake, particularly berries and grapes, is correlated with and stimulates improved global cognitive function as evidenced by both epidemiological and randomized controlled trials [118–120]. The same is true for n-3 PUFA plasma and erythrocyte levels [146–150] and intake [163, 164], particularly in healthy older adults rather than those with Alzheimer's disease [166].

Thus, across the vitamins and foods reviewed in this chapter, a consistent finding was that intake was associated with higher cognitive scores across cross-sectional and prospective studies, while the relationship was less clear in randomized controlled trials. Thus future research should focus on determining whether a causal relationship exists between diet and cognitive function, including pinpointing particular populations who may benefit most from improved nutrition (e.g., individuals with versus without heightened risk for Alzheimer's disease) and determining optimal intake values among older adults.

References

- 1. Jagust W. Vulnerable neural systems and the borderland of brain aging and neurodegeneration. Neuron. 2013; 77(2):219–34.
- Miller MG, Shukitt-Hale B. Berry fruit enhances beneficial signaling in the brain. J Agric Food Chem. 2012; 60:5709–15.
- 3. Nyberg L, Lovden M, Riklund K, Lindenberger U, Backman L. Memory aging and brain maintenance. Trends Cogn Sci. 2012;16(5):292–305.
- Britton GB, Rao KS. Cognitive aging and early diagnosis challenges in Alzheimer's disease. J Alzheimers Dis. 2011;24 Suppl 2:153–9.
- Wimo A, Winblad B, Jonsson L. The worldwide societal costs of dementia: estimates for 2009. Alzheimers Dement. 2010;6(2):98–103.
- 6. Alzheimer's Association. 2009 Alzheimer's disease facts and figures. Alzheimers Dement. 2009;5(3):234-70.
- Petersen RC, Roberts RO, Knopman DS, Geda YE, Cha RH, Pankratz VS, et al. Prevalence of mild cognitive impairment is higher in men. The Mayo clinic study of aging. Neurology. 2010;75(10):889–97.
- 8. Ferry M, Roussel AM. Micronutrient status and cognitive decline in aging. Eur Geriatr Med. 2011;2(1):15–21.
- Rafnsson SB, Dilis V, Trichopoulou A. Antioxidant nutrients and age-related cognitive decline: a systematic review of population-based cohort studies. Eur J Nutr. 2013;52(6):1553–67.
- 10. Loef M, Walach H. Fruit, vegetables and prevention of cognitive decline or dementia: a systematic review of cohort studies. J Nutr Health Aging. 2012;16(7):626–30.
- 11. Kalmijn S, Feskens EJ, Launer LJ, Kromhout D. Polyunsaturated fatty acids, antioxidants, and cognitive function in very old men. Am J Epidemiol. 1997;145(1):33–41.
- Schaefer EJ, Augustin JL, Schaefer MM, Rasmussen H, Ordovas JM, Dallal GE, et al. Lack of efficacy of a foodfrequency questionnaire in assessing dietary macronutrient intakes in subjects consuming diets of known composition. Am J Clin Nutr. 2000;71(3):746–51.
- Dunhan A, Johnson EJ. Fruits, vegetables, and their components and mild cognitive impairment and dementia: a review. Food Rev Int. 2013;29(4):407–40.

- Morris MC, Tangney CC. A potential design flaw of randomized trials of vitamin supplements. JAMA. 2011;305(13):1348–9.
- Macready AL, Butler LT, Kennedy OB, Ellis JA, Williams CM, Spencer JP. Cognitive tests used in chronic adult human randomised controlled trial micronutrient and phytochemical intervention studies. Nutr Res Rev. 2010;23(2):200–29.
- Folstein MF, Folstein SE, McHugh PR. "Mini-Mental State": a practical method for grading the cognitive state of patients for the clinician. J Psychiat Res. 1975;12:189–98.
- Cockrell JR, Folstein MF. Mini-mental state examination. In: Copeland JRM, Abou-Saleh MT, Blazer DG, editors. Principles and practice of geriatric psychiatry. Philadelphia, PA: Lippincott Williams & Wilkins; 2002. p. 140–1.
- Logue P. Non-computerized assessment procedures: fundamental assessment issues. In: Copeland JRM, Abou-Saleh MT, Blazer DG, editors. Principles and practice of geriatric psychiatry. Chicester: John Wiley & Sons Ltd.; 2002. p. 137–40.
- Joosten E, van den Berg A, Riezler R, Naurath HJ, Lindenbaum J, Stabler SP, et al. Metabolic evidence that deficiencies of vitamin B-12 (cobalamin), folate, and vitamin B-6 occur commonly in elderly people. Am J Clin Nutr. 1993;58(4):468–76.
- Lindenbaum J, Rosenberg IH, Wilson PW, Stabler SP, Allen RH. Prevalence of cobalamin deficiency in the Framingham elderly population. Am J Clin Nutr. 1994;60(1):2–11.
- Selhub J, Jacques PF, Wilson PW, Rush D, Rosenberg IH. Vitamin status and intake as primary determinants of homocysteinemia in an elderly population. JAMA. 1993;270(22):2693–8.
- 22. Rosenberg IH. Aging, B vitamins and cognitive decline. In: Fernstrom JD, Uauy R, Arroyo P, editors. Nutrition and Brain. 5th ed. Nestle Nutrition Workshop Series Clinical and Performance Program; 2001: 201–17.
- van Asselt DZ, Pasman JW, van Lier HJ, Vingerhoets DM, Poels PJ, Kuin Y, et al. Cobalamin supplementation improves cognitive and cerebral function in older, cobalamin-deficient persons. J Gerontol A Biol Sci Med Sci. 2001;56(12):M775–9.
- 24. Morris MS. The role of B vitamins in preventing and treating cognitive impairment and decline. Adv Nutr. 2012;3(6):801–12.
- 25. Selhub J, Troen A, Rosenberg IH. B vitamins and the aging brain. Nutr Rev. 2010;68 Suppl 2:S112-8.
- Miller JW. Homocysteine, B vitamins, and cognitive function. In: Kanarek RB, Lieberman HR, editors. Diet, brain, behavior practical implications. Boca Raton, FL: CRC; 2012. p. 198–214.
- 27. Crider KS, Yang TP, Berry RJ, Bailey LB. Folate and DNA methylation: a review of molecular mechanisms and the evidence for folate's role. Adv Nutr. 2012;3(1):21–38.
- Paul RT, McDonnell AP, Kelly CB. Folic acid: neurochemistry, metabolism and relationship to depression. Hum Psychopharmacol. 2004;19(7):477–88.
- Park LK, Friso S, Choi SW. Nutritional influences on epigenetics and age-related disease. Proc Nutr Soc. 2012; 71(1):75–83.
- 30. Shane B. Folic acid metabolism in health and disease. New York, NY: Dekker; 1995.
- 31. Balk EM, Raman G, Tatsioni A, Chung M, Lau J, Rosenberg IH. Vitamin B6, B12, and folic acid supplementation and cognitive function: a systematic review of randomized trials. Arch Intern Med. 2007;167(1):21–30.
- 32. D'Anci KE, Rosenberg IH. Folate and brain function in the elderly. Curr Opin Clin Nutr Metab Care. 2004;7(6):659–64.
- Luchsinger JA, Tang MX, Shea S, Miller J, Green R, Mayeux R. Plasma homocysteine levels and risk of Alzheimer disease. Neurology. 2004;62(11):1972–6.
- 34. Haan MN, Miller JW, Aiello AE, Whitmer RA, Jagust WJ, Mungas DM, et al. Homocysteine, B vitamins, and the incidence of dementia and cognitive impairment: results from the Sacramento area latino study on aging. Am J Clin Nutr. 2007;85(2):511–7.
- Reay JL, Smith MA, Riby IM. B vitamins and cognitive performance in older adults: review. ISRN Nutr. 2013;2013(Article ID 650983):7.
- Smith AD. The worldwide challenge of the dementias: a role for B vitamins and homocysteine? Food Nutr Bull. 2008;29(2 Suppl):S143–72.
- Ford AH, Flicker L, Singh U, Hirani V, Almeida OP. Homocysteine, depression and cognitive function in older adults. J Affect Disord. 2013;151(2):646–51.
- Clarke R, Smith AD, Jobst KA, Refsum H, Sutton L, Ueland PM. Folate, vitamin B12, and serum total homocysteine levels in confirmed Alzheimer disease. Arch Neurol. 1998;55(11):1449–55.
- Joosten E, Lesaffre E, Riezler R, Ghekiere V, Dereymaeker L, Pelemans W, et al. Is metabolic evidence for vitamin B-12 and folate deficiency more frequent in elderly patients with Alzheimer's disease? J Gerontol A Biol Sci Med Sci. 1997;52(2):M76–9.
- McCaddon A, Davies G, Hudson P, Tandy S, Cattell H. Total serum homocysteine in senile dementia of Alzheimer type. Int J Geriatr Psychiatry. 1998;13(4):235–9.

- 41. Clarke R, Woodhouse P, Ulvik A, Frost C, Sherliker P, Refsum H, et al. Variability and determinants of total homocysteine concentrations in plasma in an elderly population. Clin Chem. 1998;44(1):102–7.
- 42. Seshadri S, Wolf PA, Beiser AS, Selhub J, Au R, Jacques PF, et al. Association of plasma total homocysteine levels with subclinical brain injury: cerebral volumes, white matter hyperintensity, and silent brain infarcts at volumetric magnetic resonance imaging in the Framingham offspring study. Arch Neurol. 2008;65(5):642–9.
- 43. Goodwin JS, Goodwin JM, Garry PJ. Association between nutritional status and cognitive functioning in a healthy elderly population. JAMA. 1983;249(21):2917–21.
- Clarke R, Birks J, Nexo E, Ueland PM, Schneede J, Scott J, et al. Low vitamin B-12 status and risk of cognitive decline in older adults. Am J Clin Nutr. 2007;86(5):1384–91.
- 45. Kado DM, Karlamangla AS, Huang MH, Troen A, Rowe JW, Selhub J, et al. Homocysteine versus the vitamins folate, B6, and B12 as predictors of cognitive function and decline in older high-functioning adults. MacArthur Studies of Successful Aging. Am J Med. 2005;118(2):161–7.
- 46. Quadri P, Fragiacomo C, Pezzati R, Zanda E, Forloni G, Tettamanti M, et al. Homocysteine, folate, and vitamin B-12 in mild cognitive impairment, Alzheimer disease, and vascular dementia. Am J Clin Nutr. 2004;80(1): 114–22.
- Tucker KL, Qiao N, Scott T, Rosenberg I, Spiro 3rd A. High homocysteine and low B vitamins predict cognitive decline in aging men: the veterans affairs normative aging study. Am J Clin Nutr. 2005;82(3):627–35.
- 48. Snowdon DA, Tully CL, Smith CD, Riley KP, Markesbery WR. Serum folate and the severity of atrophy of the neocortex in Alzheimer disease: findings from the Nun study. Am J Clin Nutr. 2000;71(4):993–8.
- 49. Ikeda T, Furukawa Y, Mashimoto S, Takahashi K, Yamada M. Vitamin B12 levels in serum and cerebrospinal fluid of people with Alzheimer's disease. Acta Psychiatr Scand. 1990;82(4):327–9.
- Karnaze DS, Carmel R. Low serum cobalamin levels in primary degenerative dementia. Do some patients harbor atypical cobalamin deficiency states? Arch Intern Med. 1987;147(3):429–31.
- Selhub J, Bagley LC, Miller J, Rosenberg IH. B vitamins, homocysteine, and neurocognitive function in the elderly. Am J Clin Nutr. 2000;71(2):614S–20.
- 52. Morris MC, Evans DA, Bienias JL, Tangney CC, Bennett DA, Aggarwal N, et al. Dietary intake of antioxidant nutrients and the risk of incident Alzheimer disease in a biracial community study. JAMA. 2002;287(24): 3230–7.
- Seshadri S, Beiser A, Selhub J, Jacques PF, Rosenberg IH, D'Agostino RB, et al. Plasma homocysteine as a risk factor for dementia and Alzheimer's disease. N Engl J Med. 2002;346(7):476–83.
- Nilsson K, Gustafson L, Hultberg B. Plasma homocysteine concentration relates to the severity but not to the duration of Alzheimer's disease. Int J Geriatr Psychiatry. 2004;19(7):666–72.
- 55. Morris MS. Homocysteine and Alzheimer's disease. Lancet Neurol. 2003;2(7):425-8.
- 56. Dangour AD, Whitehouse PJ, Rafferty K, Mitchell SA, Smith L, Hawkesworth S, et al. B-vitamins and fatty acids in the prevention and treatment of Alzheimer's disease and dementia: a systematic review. J Alzheimers Dis. 2010;22(1):205–24.
- 57. Doets EL, van Wijngaarden JP, Szczecinska A, Dullemeijer C, Souverein OW, Dhonukshe-Rutten RA, et al. Vitamin B12 intake and status and cognitive function in elderly people. Epidemiol Rev. 2013;35(1):2–21.
- O'Leary F, Allman-Farinelli M, Samman S. Vitamin B(1)(2) status, cognitive decline and dementia: a systematic review of prospective cohort studies. Br J Nutr. 2012;108(11):1948–61.
- Vogiatzoglou A, Smith AD, Nurk E, Drevon CA, Ueland PM, Vollset SE, et al. Cognitive function in an elderly population: interaction between vitamin B12 status, depression, and apolipoprotein E epsilon4: the Hordaland homocysteine study. Psychosom Med. 2013;75(1):20–9.
- 60. Li L, Cao D, Desmond R, Rahman A, Lah JJ, Levey AI, et al. Cognitive performance and plasma levels of homocysteine, vitamin B12, folate and lipids in patients with Alzheimer disease. Dement Geriatr Cogn Disord. 2008;26(4):384–90.
- 61. Siuda J, Gorzkowska A, Patalong-Ogiewa M, Krzystanek E, Czech E, Wiechula B, et al. From mild cognitive impairment to Alzheimer's disease: influence of homocysteine, vitamin B12 and folate on cognition over time—results from one-year follow-up. Neurol Neurochir Pol. 2009;43(4):321–9.
- Feng L, Li J, Yap KB, Kua EH, Ng TP. Vitamin B-12, apolipoprotein E genotype, and cognitive performance in community-living older adults: evidence of a gene-micronutrient interaction. Am J Clin Nutr. 2009;89(4):1263–8.
- Elias MF, Robbins MA, Budge MM, Elias PK, Dore GA, Brennan SL, et al. Homocysteine and cognitive performance: modification by the ApoE genotype. Neurosci Lett. 2008;430(1):64–9.
- 64. Brown B, Huang MH, Karlamangla A, Seeman T, Kado D. Do the effects of APOE-epsilon4 on cognitive function and decline depend upon vitamin status? MacArthur studies of successful aging. J Nutr Health Aging. 2011; 15(3):196–201.
- 65. Religa D, Styczynska M, Peplonska B, Gabryelewicz T, Pfeffer A, Chodakowska M, et al. Homocysteine, apolipoproteine E and methylenetetrahydrofolate reductase in Alzheimer's disease and mild cognitive impairment. Dement Geriatr Cogn Disord. 2003;16(2):64–70.

8 Nutrition in the Prevention and Treatment of Cognitive Decline

- 66. Moorthy D, Peter I, Scott TM, Parnell LD, Lai CQ, Crott JW, et al. Status of vitamins B-12 and B-6 but not of folate, homocysteine, and the methylenetetrahydrofolate reductase C677T polymorphism are associated with impaired cognition and depression in adults. J Nutr. 2012;142(8):1554–60.
- 67. Holick MF. Vitamin D, deficiency. N Engl J Med. 2007;357(3):266-81.
- 68. Cashman KD, Kiely M, Kinsella M, Durazo-Arvizu RA, Tian L, Zhang Y, et al. Evaluation of Vitamin D standardization program protocols for standardizing serum 25-hydroxyvitamin D data: a case study of the program's potential for national nutrition and health surveys. Am J Clin Nutr. 2013;97(6):1235–42.
- 69. Dawson-Hughes B, Heaney RP, Holick MF, Lips P, Meunier PJ, Vieth R. Estimates of optimal vitamin D status. Osteoporos Int. 2005;16(7):713–6.
- 70. Holick MF. The vitamin D epidemic and its health consequences. J Nutr. 2005;135(11):2739S-48.
- Cherniak P. Vitamin D, energy regulation and mental health. In: Riby L, Smith M, Foster J, editors. Nutrition and mental performance. New York, NY: Palgrave Macmillan; 2012. p. 83–98.
- 72. Boucher BJ. The problems of vitamin d insufficiency in older people. Aging Dis. 2012;3(4):313–29.
- Parker J, Hashmi O, Dutton D, Mavrodaris A, Stranges S, Kandala NB, et al. Levels of vitamin D and cardiometabolic disorders: systematic review and meta-analysis. Maturitas. 2010;65(3):225–36.
- Perez-Lopez FR, Chedraui P, Fernandez-Alonso AM. Vitamin D and aging: beyond calcium and bone metabolism. Maturitas. 2011;69(1):27–36.
- Deluca GC, Kimball SM, Kolasinski J, Ramagopalan SV, Ebers GC. Review: the role of vitamin D in nervous system health and disease. Neuropathol Appl Neurobiol. 2013;39(5):458–84.
- Eyles DW, Feron F, Cui X, Kesby JP, Harms LH, Ko P, et al. Developmental vitamin D deficiency causes abnormal brain development. Psychoneuroendocrinology. 2009;34 Suppl 1:S247–57.
- 77. Becker A, Eyles DW, McGrath JJ, Grecksch G. Transient prenatal vitamin D deficiency is associated with subtle alterations in learning and memory functions in adult rats. Behav Brain Res. 2005;161(2):306–12.
- Kalueff AV, Tuohimaa P. Neurosteroid hormone vitamin D and its utility in clinical nutrition. Curr Opin Clin Nutr Metab Care. 2007;10(1):12–9.
- Eyles DW, Smith S, Kinobe R, Hewison M, McGrath JJ. Distribution of the vitamin D receptor and 1 alphahydroxylase in human brain. J Chem Neuroanat. 2005;29(1):21–30.
- Dickens AP, Lang IA, Langa KM, Kos K, Llewellyn DJ. Vitamin D, cognitive dysfunction and dementia in older adults. CNS Drugs. 2011;25(8):629–39.
- Moore ME, Piazza A, McCartney Y, Lynch MA. Evidence that vitamin D3 reverses age-related inflammatory changes in the rat hippocampus. Biochem Soc Trans. 2005;33(Pt 4):573–7.
- Brewer LD, Porter NM, Kerr DS, Landfield PW, Thibault O. Chronic 1alpha,25-(OH)2 vitamin D3 treatment reduces Ca2+-mediated hippocampal biomarkers of aging. Cell Calcium. 2006;40(3):277–86.
- Briones TL, Darwish H. Vitamin D mitigates age-related cognitive decline through the modulation of proinflammatory state and decrease in amyloid burden. J Neuroinflammation. 2012;9:244. 2094-9-244.
- 84. Grant WB. Does vitamin D reduce the risk of dementia? J Alzheimers Dis. 2009;17(1):151-9.
- 85. Wilkins CH, Sheline YI, Roe CM, Birge SJ, Morris JC. Vitamin D deficiency is associated with low mood and worse cognitive performance in older adults. Am J Geriatr Psychiatry. 2006;14(12):1032–40.
- 86. Przybelski RJ, Binkley NC. Is vitamin D important for preserving cognition? A positive correlation of serum 25-hydroxyvitamin D concentration with cognitive function. Arch Biochem Biophys. 2007;460(2):202–5.
- Peterson A, Mattek N, Clemons A, Bowman GL, Buracchio T, Kaye J, et al. Serum vitamin D concentrations are associated with falling and cognitive function in older adults. J Nutr Health Aging. 2012;16(10):898–901.
- Lee DM, Tajar A, Ulubaev A, Pendleton N, O'Neill TW, O'Connor DB, et al. Association between 25-hydroxyvitamin D levels and cognitive performance in middle-aged and older European men. J Neurol Neurosurg Psychiatry. 2009;80(7):722–9.
- Seamans KM, Hill TR, Scully L, Meunier N, Andrillo-Sanchez M, Polito A, et al. Vitamin D status and measures of cognitive function in healthy older European adults. Eur J Clin Nutr. 2010;64(10):1172–8.
- Annweiler C, Schott AM, Allali G, Bridenbaugh SA, Kressig RW, Allain P, et al. Association of vitamin D deficiency with cognitive impairment in older women: cross-sectional study. Neurology. 2010;74(1):27–32.
- Annweiler C, Fantino B, Schott AM, Krolak-Salmon P, Allali G, Beauchet O. Vitamin D insufficiency and mild cognitive impairment: cross-sectional association. Eur J Neurol. 2012;19(7):1023–9.
- 92. van der Schaft J, Koek HL, Dijkstra E, Verhaar HJ, van der Schouw YT, Emmelot-Vonk MH. The association between vitamin D and cognition: a systematic review. Ageing Res Rev. 2013;12:1013–23.
- Evatt ML, Delong MR, Khazai N, Rosen A, Triche S, Tangpricha V. Prevalence of vitamin d insufficiency in patients with Parkinson disease and Alzheimer disease. Arch Neurol. 2008;65(10):1348–52.
- Ferrier IN, Leake A, Taylor GA, McKeith IG, Fairbairn AF, Robinson CJ, et al. Reduced gastrointestinal absorption of calcium in dementia. Age Ageing. 1990;19(6):368–75.
- 95. Luckhaus C, Mahabadi B, Grass-Kapanke B, Janner M, Willenberg H, Jager M, et al. Blood biomarkers of osteoporosis in mild cognitive impairment and Alzheimer's disease. J Neural Transm. 2009;116(7):905–11.

- 96. Sato Y, Asoh T, Oizumi K. High prevalence of vitamin D deficiency and reduced bone mass in elderly women with Alzheimer's disease. Bone. 1998;23(6):555–7.
- 97. Annweiler C, Llewellyn DJ, Beauchet O. Low serum vitamin D concentrations in Alzheimer's disease: a systematic review and meta-analysis. J Alzheimers Dis. 2013;33(3):659–74.
- Zhao Y, Sun Y, Ji HF, Shen L. Vitamin D levels in Alzheimer's and Parkinson's diseases: a meta-analysis. Nutrition. 2013;29(6):828–32.
- 99. Afzal S, Bojesen SE, Nordestgaard BG. Reduced 25-hydroxyvitamin D and risk of Alzheimer's disease and vascular dementia. Alzheimers Dement. 2013;10:296–302.
- Llewellyn DJ, Lang IA, Langa KM, Muniz-Terrera G, Phillips CL, Cherubini A, et al. Vitamin D and risk of cognitive decline in elderly persons. Arch Intern Med. 2010;170(13):1135–41.
- Dhesi JK, Jackson SH, Bearne LM, Moniz C, Hurley MV, Swift CG, et al. Vitamin D supplementation improves neuromuscular function in older people who fall. Age Ageing. 2004;33(6):589–95.
- 102. Annweiler C, Fantino B, Gautier J, Beaudenon M, Thiery S, Beauchet O. Cognitive effects of vitamin D supplementation in older outpatients visiting a memory clinic: a pre-post study. J Am Geriatr Soc. 2012;60(4):793–5.
- Przybelski R, Agrawal S, Krueger D, Engelke JA, Walbrun F, Binkley N. Rapid correction of low vitamin D status in nursing home residents. Osteoporos Int. 2008;19(11):1621–8.
- 104. Stein MS, Scherer SC, Ladd KS, Harrison LC. A randomized controlled trial of high-dose vitamin D2 followed by intranasal insulin in Alzheimer's disease. J Alzheimers Dis. 2011;26(3):477–84.
- 105. Rossom RC, Espeland MA, Manson JE, Dysken MW, Johnson KC, Lane DS, et al. Calcium and vitamin D supplementation and cognitive impairment in the women's health initiative. J Am Geriatr Soc. 2012;60(12):2197–205.
- 106. Landete JM. Updated knowledge about polyphenols: functions, bioavailability, metabolism, and health. Crit Rev Food Sci Nutr. 2012;52(10):936–48.
- 107. Perry G, Cash AD, Smith MA. Alzheimer disease and oxidative stress. J Biomed Biotechnol. 2002;2(3):120-3.
- 108. Spencer JP. Flavonoids and brain health: multiple effects underpinned by common mechanisms. Genes Nutr. 2009;4(4):243–50.
- 109. Lau FC, Shukitt-Hale B, Joseph JA. Nutritional intervention in brain aging: reducing the effects of inflammation and oxidative stress. Subcell Biochem. 2007;42:299–318.
- 110. El Gharras H. Polyphenols: food sources, properties and applications—a review. Int J Food Sci Tech. 2009; 44:2512–8.
- Lee J, Lam L, Woo J, Kwok T. Lower fluid and fruits/vegetable intake in questionable dementia among older Hong Kong Chinese. J Nutr Health Aging. 2010;14(1):45–9.
- 112. Nurk E, Refsum H, Drevon CA, Tell GS, Nygaard HA, Engedal K, et al. Cognitive performance among the elderly in relation to the intake of plant foods. The Hordaland health study. Br J Nutr. 2010;104(8):1190–201.
- Kang JH, Ascherio A, Grodstein F. Fruit and vegetable consumption and cognitive decline in aging women. Ann Neurol. 2005;57(5):713–20.
- Morris MC, Evans DA, Tangney CC, Bienias JL, Wilson RS. Associations of vegetable and fruit consumption with age-related cognitive change. Neurology. 2006;67(8):1370–6.
- 115. Nooyens AC, Bueno-de-Mesquita HB, van Boxtel MP, van Gelder BM, Verhagen H, Verschuren WM. Fruit and vegetable intake and cognitive decline in middle-aged men and women: the Doetinchem cohort study. Br J Nutr. 2011;106(5):752–61.
- 116. Peneau S, Galan P, Jeandel C, Ferry M, Andreeva V, Hercberg S, et al. Fruit and vegetable intake and cognitive function in the SU.VI.MAX 2 prospective study. Am J Clin Nutr. 2011;94(5):1295–303.
- 117. Joseph JA, Shukitt-Hale B, Willis LM. Grape juice, berries, and walnuts affect brain aging and behavior. J Nutr. 2009;139(9):1813S-7.
- 118. Devore EE, Kang JH, Breteler MM, Grodstein F. Dietary intakes of berries and flavonoids in relation to cognitive decline. Ann Neurol. 2012;72(1):135–43.
- 119. Krikorian R, Shidler MD, Nash TA, Kalt W, Vinqvist-Tymchuk MR, Shukitt-Hale B, et al. Blueberry supplementation improves memory in older adults. J Agric Food Chem. 2010;58(7):3996–4000.
- 120. Krikorian R, Nash TA, Shidler MD, Shukitt-Hale B, Joseph JA. Concord grape juice supplementation improves memory function in older adults with mild cognitive impairment. Br J Nutr. 2010;103(5):730–4.
- 121. Krikorian R, Boespflug EL, Fleck DE, Stein AL, Wightman JD, Shidler MD, et al. Concord grape juice supplementation and neurocognitive function in human aging. J Agric Food Chem. 2012;60(23):5736–42.
- 122. Crews Jr WD, Harrison DW, Griffin ML, Addison K, Yount AM, Giovenco MA, et al. A double-blinded, placebocontrolled, randomized trial of the neuropsychologic efficacy of cranberry juice in a sample of cognitively intact older adults: pilot study findings. J Altern Complement Med. 2005;11(2):305–9.
- 123. Graham HN. Green tea composition, consumption, and polyphenol chemistry. Prev Med. 1992;21(3):334-50.
- 124. Kuriyama S, Hozawa A, Ohmori K, Shimazu T, Matsui T, Ebihara S, et al. Green tea consumption and cognitive function: a cross-sectional study from the Tsurugaya Project 1. Am J Clin Nutr. 2006;83(2):355–61.
- Ng TP, Feng L, Niti M, Kua EH, Yap KB. Tea consumption and cognitive impairment and decline in older Chinese adults. Am J Clin Nutr. 2008;88(1):224–31.

- 126. Hozawa A, Kuriyama S, Nakaya N, Ohmori-Matsuda K, Kakizaki M, Sone T, et al. Green tea consumption is associated with lower psychological distress in a general population: the Ohsaki Cohort 2006 study. Am J Clin Nutr. 2009;90(5):1390–6.
- 127. Sokolov AN, Pavlova MA, Klosterhalfen S, Enck P. Chocolate and the brain: neurobiological impact of cocoa flavanols on cognition and behavior. Neurosci Biobehav Rev. 2013;37:2445–53.
- Francis ST, Head K, Morris PG, Macdonald IA. The effect of flavanol-rich cocoa on the fMRI response to a cognitive task in healthy young people. J Cardiovasc Pharmacol. 2006;47 Suppl 2:S215–20.
- 129. Pase MP, Scholey AB, Pipingas A, Kras M, Nolidin K, Gibbs A, et al. Cocoa polyphenols enhance positive mood states but not cognitive performance: a randomized, placebo-controlled trial. J Psychopharmacol. 2013;27(5): 451–8.
- Field DT, Williams CM, Butler LT. Consumption of cocoa flavanols results in an acute improvement in visual and cognitive functions. Physiol Behav. 2011;103(3–4):255–60.
- 131. Scholey AB, French SJ, Morris PJ, Kennedy DO, Milne AL, Haskell CF. Consumption of cocoa flavanols results in acute improvements in mood and cognitive performance during sustained mental effort. J Psychopharmacol. 2010;24(10):1505–14.
- 132. Crews Jr WD, Harrison DW, Wright JW. A double-blind, placebo-controlled, randomized trial of the effects of dark chocolate and cocoa on variables associated with neuropsychological functioning and cardiovascular health: clinical findings from a sample of healthy, cognitively intact older adults. Am J Clin Nutr. 2008;87(4):872–80.
- 133. Desideri G, Kwik-Uribe C, Grassi D, Necozione S, Ghiadoni L, Mastroiacovo D, et al. Benefits in cognitive function, blood pressure, and insulin resistance through cocoa flavanol consumption in elderly subjects with mild cognitive impairment: the Cocoa, Cognition, and Aging (CoCoA) study. Hypertension. 2012;60(3):794–801.
- 134. Butchart C, Kyle J, McNeill G, Corley J, Gow AJ, Starr JM, et al. Flavonoid intake in relation to cognitive function in later life in the Lothian Birth Cohort 1936. Br J Nutr. 2011;106(1):141–8.
- 135. Kesse-Guyot E, Fezeu L, Andreeva VA, Touvier M, Scalbert A, Hercberg S, et al. Total and specific polyphenol intakes in midlife are associated with cognitive function measured 13 years later. J Nutr. 2012;142(1):76–83.
- 136. Nurk E, Refsum H, Drevon CA, Tell GS, Nygaard HA, Engedal K, et al. Intake of flavonoid-rich wine, tea, and chocolate by elderly men and women is associated with better cognitive test performance. J Nutr. 2009;139(1): 120–7.
- Parker G, Gibson NA, Brotchie H, Heruc G, Rees AM, Hadzi-Pavlovic D. Omega-3 fatty acids and mood disorders. Am J Psychiatry. 2006;163(6):969–78.
- 138. Kiecolt-Glaser JK. Stress, food, and inflammation: psychoneuroimmunology and nutrition at the cutting edge. Psychosom Med. 2010;72(4):365–9.
- Heinrichs SC. Dietary omega-3 fatty acid supplementation for optimizing neuronal structure and function. Mol Nutr Food Res. 2010;54(4):447–56.
- 140. Denis I, Potier B, Vancassel S, Heberden C, Lavialle M. Omega-3 fatty acids and brain resistance to ageing and stress: body of evidence and possible mechanisms. Ageing Res Rev. 2013;12(2):579–94.
- 141. Giles GE, Mahoney CR, Kanarek RB. Omega-3 fatty acids and cognitive behavior. In: Watson RR, De Meester F, editors. Omega 3 fatty acids in brain and neurological health. Philadelphia, PA: Elsevier; 2013.
- 142. Loef M, Walach H. The omega-6/omega-3 ratio and dementia or cognitive decline: a systematic review on human studies and biological evidence. J Nutr Gerontol Geriatr. 2013;32(1):1–23.
- 143. Luchtman DW, Song C. Cognitive enhancement by omega-3 fatty acids from child-hood to old age: findings from animal and clinical studies. Neuropharmacology. 2013;64:550–65.
- 144. Devore EE, Grodstein F, van Rooij FJ, Hofman A, Rosner B, Stampfer MJ, et al. Dietary intake of fish and omega-3 fatty acids in relation to long-term dementia risk. Am J Clin Nutr. 2009;90(1):170–6.
- 145. van de Rest O, Spiro 3rd A, Krall-Kaye E, Geleijnse JM, de Groot LC, Tucker KL. Intakes of (n-3) fatty acids and fatty fish are not associated with cognitive performance and 6-year cognitive change in men participating in the Veterans affairs normative aging study. J Nutr. 2009;139(12):2329–36.
- 146. Gao Q, Niti M, Feng L, Yap KB, Ng TP. Omega-3 polyunsaturated fatty acid supplements and cognitive decline: Singapore longitudinal aging studies. J Nutr Health Aging. 2011;15(1):32–5.
- 147. Huang TL, Zandi PP, Tucker KL, Fitzpatrick AL, Kuller LH, Fried LP, et al. Benefits of fatty fish on dementia risk are stronger for those without APOE epsilon4. Neurology. 2005;65(9):1409–14.
- Morris MC, Evans DA, Tangney CC, Bienias JL, Wilson RS. Fish consumption and cognitive decline with age in a large community study. Arch Neurol. 2005;62(12):1849–53.
- 149. van Gelder BM, Tijhuis M, Kalmijn S, Kromhout D. Fish consumption, n-3 fatty acids, and subsequent 5-y cognitive decline in elderly men: the Zutphen elderly study. Am J Clin Nutr. 2007;85(4):1142–7.
- Whalley LJ, Fox HC, Wahle KW, Starr JM, Deary IJ. Cognitive aging, childhood intelligence, and the use of food supplements: possible involvement of n-3 fatty acids. Am J Clin Nutr. 2004;80(6):1650–7.
- 151. Gonzalez S, Huerta JM, Fernandez S, Patterson AM, Lasheras C. The relationship between dietary lipids and cognitive performance in an elderly population. Int J Food Sci Nutr. 2010;61(2):217–25.

- Heude B, Ducimetiere P, Berr C, EVA Study. Cognitive decline and fatty acid composition of erythrocyte membranes: the EVA study. Am J Clin Nutr. 2003;77(4):803–8.
- 153. Roberts RO, Cerhan JR, Geda YE, Knopman DS, Cha RH, Christianson TJ, et al. Polyunsaturated fatty acids and reduced odds of MCI: the Mayo clinic study of aging. J Alzheimers Dis. 2010;21(3):853–65.
- 154. Chiu CC, Frangou S, Chang CJ, Chiu WC, Liu HC, Sun IW, et al. Associations between n-3 PUFA concentrations and cognitive function after recovery from late-life depression. Am J Clin Nutr. 2012;95(2):420–7.
- 155. Dullemeijer C, Durga J, Brouwer IA, van de Rest O, Kok FJ, Brummer RJ, et al. N 3 Fatty acid proportions in plasma and cognitive performance in older adults. Am J Clin Nutr. 2007;86(5):1479–85.
- 156. Samieri C, Feart C, Letenneur L, Dartigues JF, Peres K, Auriacombe S, et al. Low plasma eicosapentaenoic acid and depressive symptomatology are independent predictors of dementia risk. Am J Clin Nutr. 2008;88(3): 714–21.
- 157. Whalley LJ, Deary IJ, Starr JM, Wahle KW, Rance KA, Bourne VJ, et al. n-3 Fatty acid erythrocyte membrane content, APOE varepsilon4, and cognitive variation: an observational follow-up study in late adulthood. Am J Clin Nutr. 2008;87(2):449–54.
- 158. Milte CM, Sinn N, Street SJ, Buckley JD, Coates AM, Howe PR. Erythrocyte polyunsaturated fatty acid status, memory, cognition and mood in older adults with mild cognitive impairment and healthy controls. Prostaglandins Leukot Essent Fatty Acids. 2011;84(5–6):153–61.
- 159. Beydoun MA, Kaufman JS, Sloane PD, Heiss G, Ibrahim J. n-3 Fatty acids, hypertension and risk of cognitive decline among older adults in the Atherosclerosis Risk in Communities (ARIC) study. Public Health Nutr. 2008;11(1):17–29.
- 160. Kroger E, Verreault R, Carmichael PH, Lindsay J, Julien P, Dewailly E, et al. Omega-3 fatty acids and risk of dementia: the Canadian study of health and aging. Am J Clin Nutr. 2009;90(1):184–92.
- Laurin D, Verreault R, Lindsay J, Dewailly E, Holub BJ. Omega-3 fatty acids and risk of cognitive impairment and dementia. J Alzheimers Dis. 2003;5(4):315–22.
- 162. Ronnemaa E, Zethelius B, Vessby B, Lannfelt L, Byberg L, Kilander L. Serum fatty-acid composition and the risk of Alzheimer's disease: a longitudinal population-based study. Eur J Clin Nutr. 2012;66(8):885–90.
- 163. Johnson EJ, McDonald K, Caldarella SM, Chung HY, Troen AM, Snodderly DM. Cognitive findings of an exploratory trial of docosahexaenoic acid and lutein supplementation in older women. Nutr Neurosci. 2008;11(2): 75–83.
- 164. Yurko-Mauro K, McCarthy D, Rom D, Nelson EB, Ryan AS, Blackwell A, et al. Beneficial effects of docosahexaenoic acid on cognition in age-related cognitive decline. Alzheimers Dement. 2010;6(6):456–64.
- 165. Stough C, Downey L, Silber B, Lloyd J, Kure C, Wesnes K, et al. The effects of 90-day supplementation with the omega-3 essential fatty acid docosahexaenoic acid (DHA) on cognitive function and visual acuity in a healthy aging population. Neurobiol Aging. 2012;33(4):824.e1–3.
- 166. van de Rest O, Geleijnse JM, Kok FJ, van Staveren WA, Dullemeijer C, Olderikkert MG, et al. Effect of fish oil on cognitive performance in older subjects: a randomized, controlled trial. Neurology. 2008;71(6):430–8.
- 167. Corsinovi L, Biasi F, Poli G, Leonarduzzi G, Isaia G. Dietary lipids and their oxidized products in Alzheimer's disease. Mol Nutr Food Res. 2011;55 Suppl 2:S161–72.
- 168. Freund-Levi Y, Eriksdotter-Jonhagen M, Cederholm T, Basun H, Faxen-Irving G, Garlind A, et al. Omega-3 fatty acid treatment in 174 patients with mild to moderate Alzheimer disease: OmegAD study—a randomized doubleblind trial. Arch Neurol. 2006;63(10):1402–8.
- 169. Chiu CC, Su KP, Cheng TC, Liu HC, Chang CJ, Dewey ME, et al. The effects of omega-3 fatty acids monotherapy in Alzheimer's disease and mild cognitive impairment: a preliminary randomized double-blind placebo-controlled study. Prog Neuropsychopharmacol Biol Psychiatry. 2008;32(6):1538–44.
- 170. Sinn N, Milte CM, Street SJ, Buckley JD, Coates AM, Petkov J, et al. Effects of n-3 fatty acids, EPA v. DHA, on depressive symptoms, quality of life, memory and executive function in older adults with mild cognitive impairment: a 6-month randomised controlled trial. Br J Nutr. 2012;107(11):1682–93.
- 171. Quinn JF, Raman R, Thomas RG, Yurko-Mauro K, Nelson EB, Van Dyck C, et al. Docosahexaenoic acid supplementation and cognitive decline in Alzheimer disease: a randomized trial. JAMA. 2010;304(17):1903–11.
- 172. Berr C, Portet F, Carriere I, Akbaraly TN, Feart C, Gourlet V, et al. Olive oil and cognition: results from the threecity study. Dement Geriatr Cogn Disord. 2009;28(4):357–64.
- 173. Ovaskainen ML, Torronen R, Koponen JM, Sinkko H, Hellstrom J, Reinivuo H, et al. Dietary intake and major food sources of polyphenols in Finnish adults. J Nutr. 2008;138(3):562–6.
- 174. Tresserra-Rimbau A, Medina-Remon A, Perez-Jimenez J, Martinez-Gonzalez MA, Covas MI, Corella D, et al. Dietary intake and major food sources of polyphenols in a Spanish population at high cardiovascular risk: the PREDIMED study. Nutr Metab Cardiovasc Dis. 2013;23:953–9.
- 175. Zujko ME, Witkowska AM, Waskiewicz A, Sygnowska E. Estimation of dietary intake and patterns of polyphenol consumption in Polish adult population. Adv Med Sci. 2012;11:1–10.
- Chun OK, Chung SJ, Song WO. Estimated dietary flavonoid intake and major food sources of U.S. adults. J Nutr. 2007;137(5):1244–52.

- 177. Bach-Faig A, Berry EM, Lairon D, Reguant J, Trichopoulou A, Dernini S, et al. Mediterranean diet pyramid today. Science and cultural updates. Public Health Nutr. 2011;14(12A):2274–84.
- 178. deKoning L, Anand SS. Adherence to a Mediterranean diet and survival in a Greek population. Trichopoulou A, Costacou T, Bamia C, Trichopoulos D. N Engl J Med 2003; 348: 2599–608. Vasc Med. 2004; 9(2):145–6.
- 179. Cherbuin N, Anstey KJ. The Mediterranean diet is not related to cognitive change in a large prospective investigation: the PATH through life study. Am J Geriatr Psychiatry. 2012;20(7):635–9.
- Valls-Pedret C, Lamuela-Raventos RM, Medina-Remon A, Quintana M, Corella D, Pinto X, et al. Polyphenol-rich foods in the Mediterranean diet are associated with better cognitive function in elderly subjects at high cardiovascular risk. J Alzheimers Dis. 2012;29(4):773–82.
- Vercambre MN, Grodstein F, Berr C, Kang JH. Mediterranean diet and cognitive decline in women with cardiovascular disease or risk factors. J Acad Nutr Diet. 2012;112(6):816–23.
- Scarmeas N, Luchsinger JA, Mayeux R, Stern Y. Mediterranean diet and Alzheimer disease mortality. Neurology. 2007;69(11):1084–93.
- 183. Feart C, Samieri C, Rondeau V, Amieva H, Portet F, Dartigues JF, et al. Adherence to a Mediterranean diet, cognitive decline, and risk of dementia. JAMA. 2009;302(6):638–48.
- Scarmeas N, Stern Y, Mayeux R, Manly JJ, Schupf N, Luchsinger JA. Mediterranean diet and mild cognitive impairment. Arch Neurol. 2009;66(2):216–25.
- 185. Roberts RO, Geda YE, Cerhan JR, Knopman DS, Cha RH, Christianson TJ, et al. Vegetables, unsaturated fats, moderate alcohol intake, and mild cognitive impairment. Dement Geriatr Cogn Disord. 2010;29(5):413–23.
- 186. Tangney CC, Kwasny MJ, Li H, Wilson RS, Evans DA, Morris MC. Adherence to a Mediterranean-type dietary pattern and cognitive decline in a community population. Am J Clin Nutr. 2011;93(3):601–7.
- 187. Feart C, Samieri C, Barberger-Gateau P. Mediterranean diet and cognitive function in older adults. Curr Opin Clin Nutr Metab Care. 2010;13(1):14–8.
- Drewnowski A, Evans WJ. Nutrition, physical activity, and quality of life in older adults: summary. J Gerontol A Biol Sci Med Sci. 2001;56(Spec No. 2):89–94.

Chapter 9 Food Insecurity and Hunger Among Older Adults

David R. Buys and Julie L. Locher

Key Points

- Food insecurity is a pressing public health issue for older adults.
- Older adults may be less likely than some other segments of the general population to experience food insecurity, but those who are food insecure are likely to remain food insecure and to experience its effects to a greater degree.
- Dealing with food insecurity among older adults requires a multi-pronged approach, including continued support for government programs, such as the Nutrition Services provided for by the Older Americans Act and the Supplemental Nutrition Assistance Program, and private sector programs, such as food banks.

Keywords Food insecurity • Hunger • Undernutrition • Obesity

Introduction: Food (In)Security, Hunger, and Undernutrition

Meeting nutritional needs is necessary for optimal health across the lifespan and especially for older adults [1, 2]. However, it is not always feasible to access the foods needed to obtain desirable health outcomes. Therefore, it is important to understand what factors enable, constrain, or are associated with individuals' ability to access and consume healthful foods. This is especially true for older adults who may be susceptible to irreversible changes in their diet as they encounter illnesses and changes in their eating behaviors with advancing age [3, 4]. Food (in)security is a helpful construct in trying to understand forces that may promote or impede how people obtain and consume nutrition.

The World Health Organization (WHO) defines food security as existing "when all people at all times have access to sufficient, safe, nutritious food to maintain a healthy and active life" [5]. WHO posits that food security has three pillars: availability, access, and use of food. Availability refers to

D.R. Buys, PhD, MSPH (\boxtimes)

Department of Food Science, Nutrition, and Health Promotion State Health Specialist, MSU Extension Service Mississippi State University, PO Box 9805, 107 Herzer, 945 Stone Boulevard MSU, MS 39762-9805, USA e-mail: david.buys@msstate.edu

J.L. Locher, Ph.D.

Translational Nutrition and Aging Program, Departments of Medicine and Health Care Organization, University of Alabama at Birmingham, 3600 Cliff Road S, Birmingham, AL 35205, USA e-mail: jlocher@uab.edu

having an adequate amount of food to feed all people; this is not currently a problem in the United States or most other nations as there is a large enough food supply for most persons around the world. *Access* is more of a problem in the United States, however, and refers to individuals "having sufficient resources to obtain appropriate foods for a nutritious diet." Finally, *Use* refers to "appropriate use based on knowledge of basic nutrition and care, as well as adequate water and sanitation." Proper food use requires adequate access to food and education about what dietary practices are appropriate for a healthy lifestyle [5]. When availability, access and use are not in equilibrium, a state of food insecurity is present. Combating food insecurity and working toward a state of food security is important, so much so that it has been called the "linchpin of healthful living" [6].

Measurement and Definition of Food (In)Security in the United States

The term food (in)security was first operationalized for use in the US context in 1989 by an expert panel of the Life Sciences Research Office (LSRO) as having "access to enough food for an active, healthy life" [7]. Similar to the WHO definition, food security requires availability of and access to nutritionally appropriate foods without food assistance from emergency food supplies, illegal means, or other coping strategies. *Food insecurity*, in contrast, exists "whenever the availability of nutritionally adequate and safe foods or the ability to acquire acceptable foods in socially acceptable ways is limited or uncertain" [8]. Food insecurity specifically refers to a condition stemming from poverty or other factors, not voluntary nutritional modifications such as fasting. While in the general population, poverty is the most likely factor to cause food insecurity, among older adults it may be caused by physical impairments or disabilities that limit their ability to procure, prepare and/or consume the food [9, 10].

After the LSRO conceptualized food (in)security, in 1990, the United States Congress mandated that trends in nutrition and food security in the US be monitored; and these new measures of food (in) security were to be focused on whole households, rather than on individuals. This led to funding for the first administration of the Food Security Supplement (FSS) to assess prevalence rates of food insecurity in 1995. Initially sponsored by the Economic Research Service of the United States Department of Agriculture (USDA), measures of food insecurity have since been collected annually by the Census Bureau's Current Population Survey. Early categories of food security without hunger," and "food security." In 2006, a National Academy of Sciences panel made recommendations for updated categories of food (in)security that would distinguish between food insecurity and hunger [11, 12] and ensure that the categories were conceptually and operationally sound and able to relay pertinent and actionable information to policy makers [13]. Questions from the current Core Food Security Module (CFSM) used to assess food (in)security are presented in Table 9.1.

The categories of food insecurity that emerged from these questions are high, marginal, low, and very low food security. *High food security* is a "state in which there are no reported indications of foodaccess problems or limitations." *Marginal food security* is a state of "one or two reported indications typically of anxiety over food insufficiency or shortage of food in the house." There would be little or no indication of changes in diets or food intake. Households with *low food security* (formerly "food insecurity without hunger") report "reduced quality, variety, or desirability of diet." They also report little or no indication of reduced food intake. Households with *very low food security* (formerly "food insecurity with hunger") report "multiple indications of disrupted eating patterns and reduced food intake" [13]. Households that indicate three or more of any of these conditions of food insecurity are classified as food insecure, including that individuals in households: (1) worried their food would run out before getting more money; (2) bought food that didn't last and didn't have money for more; and (3) couldn't afford to eat balanced meals. Additionally, households with adults who indicated that they

Table 9.1 Survey questions in the FSS that assess household food security

- 1. "We worried whether our food would run out before we got money to buy more." Was that often, sometimes, or never true for you in the last 12 months?
- 2. "The food that we bought just didn't last and we didn't have money to get more." Was that often, sometimes, or never true for you in the last 12 months?
- 3. "We couldn't afford to eat balanced meals." Was that often, sometimes, or never true for you in the last 12 months?
- 4. In the last 12 months, did you or other adults in the household ever cut the size of your meals or skip meals because there wasn't enough money for food? (Yes/No)
- 5. (If yes to question 4) How often did this happen—almost every month, some months but not every month, or in only 1 or 2 months?
- 6. In the last 12 months, did you ever eat less than you felt you should because there wasn't enough money for food? (Yes/No)
- 7. In the last 12 months, were you ever hungry, but didn't eat, because there wasn't enough money for food? (Yes/No)
- 8. In the last 12 months, did you lose weight because there wasn't enough money for food? (Yes/No)
- 9. In the last 12 months did you or other adults in your household ever not eat for a whole day because there wasn't enough money for food? (Yes/No)
- 10. (If yes to question 9) How often did this happen—almost every month, some months but not every month, or in only 1 or 2 months?

(Questions 11–18 were asked only if the household included children age 0–17)

- 11. "We relied on only a few kinds of low-cost food to feed our children because we were running out of money to buy food." Was that often, sometimes, or never true for you in the last 12 months?
- 12. "We couldn't feed our children a balanced meal, because we couldn't afford that." Was that often, sometimes, or never true for you in the last 12 months?
- 13. "The children were not eating enough because we just couldn't afford enough food." Was that often, sometimes, or never true for you in the last 12 months?
- 14. In the last 12 months, did you ever cut the size of any of the children's meals because there wasn't enough money for food? (Yes/No)
- 15. In the last 12 months, were the children ever hungry but you just couldn't afford more food? (Yes/No)
- 16. In the last 12 months, did any of the children ever skip a meal because there wasn't enough money for food? (Yes/No)
- 17. (If yes to question 16) How often did this happen—almost every month, some months but not every month, or in only 1 or 2 months?
- 18. In the last 12 months did any of the children ever not eat for a whole day because there wasn't enough money for food? (Yes/No)

Source: Food Insecurity in the US: Measurement: United States Department of Agriculture: Economic Research Service; 2012 [cited 2012]. Available from: http://www.ers.usda.gov/topics/food-nutrition-assistance/food-security-in-the-us/measurement.aspx#.UV35bKKG3eA

ate less than what they felt they should and that they cut the size of their meals or skipped meals in at least 3 months were classified as having very low food security. Because the U.S. Household Food Security Scale asks about hunger over the previous year, it is sensitive to rare, occasional, or episodic occurrences of food insecurity [14], not just recent episodes or occasions. When a household contains one or more persons responding affirmatively to the questions about food insecurity, the full household is considered food insecure [12].

Communities, likewise, may be considered food (in)secure; that is, they may be in "A situation in which all community residents [are unable to] obtain a safe, culturally acceptable, nutritionally adequate diet through a sustainable food system that maximizes self-reliance and social justice," [15] without resorting to emergency food sources [16]. In contrast, communities that are food secure have conditions in place that enable their residents to access healthy foods that support complete wellbeing [17]. While community food security is relatively easily defined, measuring it is not so easy; Cohen and colleagues proposed an assessment tool for community food insecurity, but uptake has been slow because of its complexity [18]. Food (in)security is a measure at the household, community, or higher-level, but it has implications for more individual-level challenges like hunger and malnutrition, including undernutrition and obesity.



Fig. 9.1 Conceptual model of the association of food insecurity with individual nutrition-related outcomes

Hunger

Hunger is "a potential consequence of food insecurity that, because of prolonged, involuntary lack of food, results in discomfort, illness, weakness, or pain that goes beyond the usual uneasy sensation" [13]; furthermore, it "may produce malnutrition over time" [7]. Hunger is distinct from food insecurity, though sometimes, it is an indicator or consequence of it. Furthermore, hunger is an individual-level condition that should be assessed at that level, in contrast to food insecurity, which is usually assessed at the household or community level [12, 19]. Hunger, as a social problem, has been recognized by society for much longer than food insecurity [12]. However, even given the longstanding recognition that hunger is a problem, older adults may experience unrecognized hunger and consequently be less likely to benefit from efforts to combat hunger through government services and private organizations like SNAP or food banks [20]. This is likely because older adults may have subclinical nutrient deficiencies such as iron, folic acid, and vitamin A, but not demonstrable clinical indications of undernutrition such as wasting or other signs that would indicate the presence of hunger to the naked eye [21].

Malnutrition

Malnutrition is "a condition brought about by insufficient intake of nutrients to meet biological requirements" [12] and may manifest itself as either undernutrition, leading to wasting or as overnutrition, leading to obesity [22]. Both of these will be described further in the next sections (Fig. 9.1).

Undernutrition

Undernutrition is a specific nutritional state which has been characterized by unintentional weight loss, low anthropometric measurements, abnormal biochemical markers, poor nutritional intake as indicated by nutrition screening instruments [23]. Undernourishment has also been defined by the

FAO as a state "when caloric intake is below the minimum dietary energy requirement," and is considered to be "an extreme form of food insecurity" [24]. The specific causes of weight loss in older adults include anorexia, cachexia, malabsorption, hypermetabolism, dehydration, and sarcopenia. The prevalence rates of these conditions are estimated to be 5-10 % of persons in nursing homes and up to 50 % of persons discharged from hospitals [25]. More information on definitions and measurement of undernutrition, malnutritional risk is presented in Chap. 3.

Overnutrition and Obesity

Food insecurity has historically been associated with undernutrition, and older adults have been stereotyped as underweight and frail. However, in the past decade, there has been a rise in the proportion of obese older adults, and food insecurity has become increasingly recognized as a factor contributing to overnutrition and obesity [19, 25–27]. "Overnutrition" [or consumption of excess calories] may stem from access to and use of foods of high calorie/low nutrient density that exceed the caloric needs of those needed for the individual consuming them [21]. Habits of excess kcal consumption that may have existed over a lifetime become exacerbated in older age with the loss of lean muscle mass, with the net effect of sarcopenic obesity [28, 29]. Thus, ironically, it is possible to be both overnourished according to one measure (i.e., calories consumed), yet undernourished according to another (i.e., macro- and micro-nutrients consumed).

It is also the case that persons who are food insecure may be more likely to live in poorer neighborhoods with less access to stores that sell nutritious foods or those for special dietary needs [30], limiting their ability to buy foods necessary for a healthy lifestyle. They may resort to purchasing low nutrient-dense foods with higher energy that are more convenient and affordable for them [35]. Ultimately, the overabundance of inexpensive and nutritionally-depleted foods contributes to obesity among persons in food insecure communities or households [19]. This may cyclically exacerbate food insecurity in older adults who become unable because of mobility restrictions to travel beyond their own neighborhoods to shop for more nutritionally dense foods.

The Distribution of Food (In)Security in the Population

Individual and Household Characteristics of Food (In)Security

The characteristics of individuals and households living with food insecurity vary across the population. In 2011, 14.9 % of all US households were food insecure (living with low or very low food security), an increase from 10.5 % in 2000 [31]. Figure 9.2 demonstrates current rates of food (in) security in the United States.

Among the total population, 5.7 % were living with very low food security in 2011, up from 5.4 % the previous year, which was a statistically significant increase. Those numbers were as low as 3 % in 2000, increasing to 4 % in 2004, and remaining unchanged through 2007 when they began to increase to current levels. Factors associated with food (in)security among the general population, were low social capital, rural residence, lower income, lower education, and minority status [32].



Fig. 9.2 U.S. households by food security status, 2011. Reprinted from [33]



Fig. 9.3 U.S. households with older adults by food security status, 2011. Reprinted from [33]

Food (In)Security Characteristics for Middle and Older Adults

In 2011, among households with older adults (65+), 8.4 % were food insecure (Fig. 9.3).

Among this older adult population, 5.3 % lived with low food security, and 3.1 % lived with very low food security [33]. Also, about 19 % of households with adults ages 60 and over who were living below 185 % of the poverty line were food insecure [22].

Prior to 2011, between 2007 and 2009, when the US economy was under duress, food insecurity among middle and older adults saw some notable changes. Persons 40–49 years old experienced a 68 % increase between 2007 and 2009, compared to 38 % increases for 50–59 year olds, and a 25 % increase among those 60 and older. Overall however, older individuals were less likely to be food insecure than younger ones. Yet, food insecurity was more prevalent in larger households of older adults (with two or more persons) and among individuals that had less than a high school diploma, had disabilities related to work, or were in rented homes [34, 35].

Between 2006 and 2008, the number of poor and near poor older adults who did not know where their next meal would come from doubled from 4.7 to 10.1 % [36]. This rate is projected to increase to nearly 13 % by 2025, when the youngest Baby Boomers turn 65. Factors predictive of food

insecurity among older adults, include living with a grandchild, living alone, being African American or Hispanic, living in or near poverty, having less than a high school education and being a renter. Specifically, food insecurity among older adults with a grandchild in the home is at least twice that of older adults without a grandchild living in their home [37]. Furthermore, food insecurity among the poor and near poor are two to three times greater for persons over age 50 than for the general population. Notably, though, after 2007, food insecurity did not increase at the same rate among the poor and near poor as it did for persons in higher income brackets. This is indicative that the economic recession had the greatest impact for food insecurity among those with higher incomes [31].

In general, food insecurity is strongly linked with poverty and may be either transitory [i.e., varying throughout the year or through the lifecourse] or chronic. Unlike children and/or young and middle-aged adults, older adults with limited resources are more likely to experience chronic poverty and food insecurity because they are less likely to acquire new, additional resources and move out of poverty. Whereas younger persons experience seasonal or temporary food insecurity [e.g., that which may be associated with an economic downturn or job loss], older adults who are dependent upon Social Security or Supplemental Security Income for the rest of their lives are not able to break out of poverty, and, thus are likely to remain food insecure without assistance.

Geographic Variation in Food (In)Security

Regional differences in food insecurity among older adults were significant in 2011, ranging from 16 % in the South to 15.8 % in the West, to 13.5 % in the Midwest, and the Northeast [33]. The disparity in food insecurity between states is also substantial, with the lowest rate of food insecurity at 1.5 % in North Dakota and climbing to over eight times greater in Mississippi, with a rate of 12.3 %. Furthermore, persons living with very low food security ranged from 2.6 % in North Dakota to 6.8 % in Alabama [35]; with regards to older adults, eight of the states with the highest rates of food insecurity are in the South.

Large cities and rural areas have more food insecure older adults than suburban areas and other outlying areas around large cities [33]. Food insecurity differs, as well, for those older adults in metropolitan versus nonmetropolitan areas. For example, married persons are less likely to be food insecure than unmarried persons in metro areas, but not nonmetro areas. Also, older adults living alone in metropolitan areas are twice as likely to be food insecure as those not living alone, but there is no relationship in nonmetro areas [36].

Health Implications of Food (In)Security

Food insecurity, hunger, and undernutrition represent different health-related challenges for all individuals and specifically for older adults. Risk factors associated with inadequate nutritional intake include poverty, race, living alone, chronic diseases, cognitive status, dentition, location of residence, polypharmacy, and household food sufficiency [2]. Among the general population, an association has been observed between food insecurity, hypertension, hyperlipidemia, and diabetes. Among persons living in households with very low food security, there was a twofold increase in diabetes risk compared to those in food-secure households. Furthermore, adults with diabetes in food-insecure households had higher hemoglobin A1c values relative to persons in food-secure households, indicating that people in food insecurity has been associated with poor physical and mental health status, including depression in women [6]. Specifically, anxiety, feelings of alienation and deprivation, distress, and adverse changes in family and social interactions sometimes occur as a result of food insecurity [12].

Food (In)Security and Health Among Older Adults

Food insecurity poses specific health-related challenges for older adults. Among persons 60–69 years old who live at less than 200 % of the poverty level and who are food insecure, 28 % of them have diabetes, compared to 19 % of persons who are food secure. Among all persons 60–69 who are food insecure, 10 % report depressive symptomology compared to 4 % of those who are food secure. Of those who are food secure, 87 % indicated at least one activities of daily living limitation, compared to 72 % of those who are food secure. Finally, among persons food insecure, 42 % reported having excellent, very good, or good health, compared to 61 % of persons who are food secure [35, 36]. Older adults 65+ who were food insecure reported greater health services utilization [38], and those who presented at emergency departments from food insecure contexts had greater comorbidities than those who came from food-secure ones [39]. Also, Lee et al. [9] found that food insecure older adults were more likely to have poorer dietary intake, nutritional status, and health status than food-secure persons. Finally, food insecure persons ages 60 years and older have statistically significant lower nutrient intakes; and food insecure persons ages 50 years and older are less likely to be in excellent or very good health [40].

Among slightly younger persons, those 50–59, 19 % of food insecure people had diabetes compared with 10 % food-secure people. Among persons who were food insecure, 16 % reported depressive symptomology compared to 3 % of those who were food secure. Of those who were food secure, 52 % indicated at least one activities of daily living limitation, compared to 21 % of those who were food secure. After adjusting for income, race, and education, food insecurity affected activities of daily living, such that, a food insecure person had a rating like that of a food-secure adult 14 years older. These food-insecure adults were also more likely than food-secure persons to have diabetes. Furthermore, prevalence of food insecurity was nearly twice as high for persons with multiple physical impairments (29 %) than those with none (15 %). Finally, among these food insecure persons, only 48 % reported having excellent, very good, or good health, compared to 78 % of persons who were food secure [31].

Malnutrition and Health Among Older Adults

Malnutrition may manifest itself differently among people of different ages. Undernutrition, one aspect of malnutrition, varies for older adults according to the setting in which it is measured. Undernutrition is prevalent among 5-12% of the community-dwelling older adult population; 5-10%of the nursing home population; 32–50 % of hospitalized patients, and 70–92 % of homebound persons [25, 41]. Undernutrition has been shown to be a cause of pressure ulcers, hip fractures, falls, weakness, fatigue, anemia, edema, cognitive abnormalities, infections, immune dysfunction, thymic atrophy, decreased delayed hypersensitivity, decreased mitogen lymphocytes, and decreased response to immunizations, and death [25]. Furthermore, undernutrition has been associated with functional decline and increased health care utilization, including nursing home placement, and costs of care [42–44]. Weight loss stemming from undernutrition may accelerate functional decline, loss of muscle mass, balance impairment, and decreased strength and power and walking speed [2]. Also among older adults, weight loss of 5 % or more doubles the risk of death [25, 45]. Other possible causes of undernutrition are polypharmacy, which may alter desire for food; poor dentition, which may alter a person's ability to chew food; and functional status, which may impede people from shopping for and preparing food. Additionally, psychosocial issues, including mental and cognitive impairment may lead to lower nutritional status. Specifically, depression may lead to a loss of motivation to consume

enough or appropriate foods [30]. In a study of urban homebound older adults, 29 % of the women and 37 % of the men were underweight; and 38 % consumed inadequate amounts of protein and energy. Age, educational status, and oral symptoms were all predictors of lower BMI. Wearing of dentures was positively associated with BMI [46].

Overnutrition may also stem from food insecurity, as described above. Overnutrition leads to obesity, a condition affecting approximately 30 % of all older adults. Obesity-related conditions include hypertension, hyperlipidemia, and diabetes. Costs for obese versus nonobese older adults among Medicare beneficiaries are projected to be 34 % greater. In one study of older adults receiving Medicare home health services, nutritional risk among overweight and obese patients was associated with nursing home placement within 1 year of follow-up [44]. Nutritional risk, including undernutrition, is described in greater detail in Chap. 3, and obesity is discussed in Chaps. 6 and 10.

Interventions for Food Insecurity

The public health response to the food and nutritional needs of food insecure older adults may have an influence on health outcomes for many years to come [47]. By 2030, it is projected that there will be over 72 million persons over 65 years of age in the United States, nearly double that of 1996. With this shifting population, it is imperative that society address the need for appropriate nutritional health. On average, older adults reported spending 12.4 % of their income on food [48]. One report on older adults observed that while income for making food purchases was not a problem, obtaining and preparing food was challenging due to limited transportation, functional limitations, or other health impairments [10]. Because of the increasing numbers and proportion of older adults in the United States, it is especially important to consider what interventions on food insecurity may be effective.

Food Assistance for Older Adults

Because adequate nutrition is necessary for health, functionality, and the ability of older adults to remain at home in the community, it is imperative that researchers, practitioners, and advocates consider what programs exist and what needs remain for ensuring food security for older adults. Food and nutrition programs for community-residing older adults have been well described in a position paper issued jointly in 2010 by the American Dietetic Association, American Society for Nutrition, and Society for Nutrition Education [30]. In the U.S., nutrition assistance programs designed for older adults include the Older Americans Act (OAA) Nutrition Program, Senior Farmers' Market Nutrition Program, Commodity Supplemental Food Program (CSFP) for adults age 60 years and older who meet income eligibility requirements, and the Child and Adult Care Food Program (CACFP) for chronically impaired disabled adults and adults age 60 years and older in adult day care centers. Programs designed for all citizens, including older adults, include Supplemental Nutrition Assistance Program (SNAP), the Emergency Food Assistance Program, and the Emergency Food and Shelter National Board Program. Among food-insecure households, only 57 % participated in one or more of the three largest Federal food and nutrition assistance programs during the month prior to the 2011 survey [33].

Among Older Americans Act programs, there are three primary nutrition services for nutrition assistance: Home-Delivered Nutrition Services (HDN), Congregate Nutrition Services (CN), and a Nutrition Services Incentive Program. In 2010 the HDN program provided approximately 145 million meals to more than 880,000 older adults; and the CN program provided over 96.4 million meals to more than 1.7 million older adults in different community settings. These services are crucial for their

recipients as nearly 90 % of AOA clients have multiple chronic conditions, which can be combated through proper nutrition. Furthermore, nearly 35 % of older adults receiving HDN meals were unable to perform three or more activities of daily living, while 69 % were unable to perform three or more instrumental activities of daily living; these impairments are also associated with nutritional needs which were being met through the Administration on Aging (AOA) programs, and possibly preventing or delaying their need for institutional care.

In 2010, however, approximately 89 % of food-insecure older adults did not receive homedelivered or congregate meals. It was estimated that eligible persons may not participate in the OAA programs such as HDN or CN offerings for a number of potential reasons—they may have limited awareness of available services; live in areas with limited available services; receive informal services through friends, family, or other organizations; choose not to obtain government assistance; or receive nutrition assistance through other federal programs, such as SNAP. However, a recent report found that only 7 % of CN participants and 16 % of home-delivered meal recipients also received SNAP benefits [49].

While participation rates in HDN and CN program seems to be low, there is an increasing demand for OAA Elderly Nutrition Program Services. The Government Accountability Office reported that in 2010 requests for HDN and CN services have increased 79 % and 47 %, respectively, since 2007. This may reflect the increasing rates of older adults who stay in their homes rather than moving to institutional settings; it is important because the growing number of individuals eligible for HDN services are those most vulnerable for poor dietary intake, poor nutritional status, and physical disability [2].

Because, the chronic diseases that are the major causes of death among older adults include heart disease, cancer, stroke, influenza, and pneumonia, all of which are impacted by diet and nutrition, it is important to consider how to maintain food security among this population. Similarly, preventing malnutrition in older adults—including undernutrition and overnutrition—through healthy nutrition may delay or prevent the disablement process, reducing fall risk and hip fractures. Reducing these conditions may also increase functional capacity and reduce medical costs to individuals and the Medicare system [19].

One of the most frequently used programs among the general population is SNAP; not only is the SNAP program helpful for relieving individual food insecurity and hunger, the USDA found that for every \$5 in new food stamp benefits, individuals spent \$9 in their communities, indicating that there is an economic ripple effect [19]. Older adults have the lowest participation rates in SNAP; therefore, food access may be improved among older adults through greater participation in the program. However, the USDA's research in 2004 on why individuals who may qualify, including older adults, do not apply for benefits found that "one-third [of non-participants] did not think they were eligible, 18 % were unsure whether or not they would qualify for benefits, and a small % had never heard of the [program]." Some older adults who were aware of the process reported that they believed SNAP benefits are too low. Furthermore, the application process for SNAP may be too long and complicated for older adults, especially those with cognitive or physical impairments, who may be even less likely to apply for SNAP.

Private and community initiatives to promote food security among individuals in communities should focus on improving access to healthy and nutritious food items, increasing the self-reliance of communities in providing their own foods, and developing cross-disciplinary, inter-institutional responses to food and nutrition-related needs. These goals may be accomplished through farmers markets, community-supported agriculture programs and economic development activities [50]. Innovative programs partnering farmers markets with low income older adults to provide seasonal vegetables may be useful toward this end [51]. Further research on the effectiveness of the existing programs would also bolster the evidence-base for the continuation of these programs.

Conclusion

The evidence presented here indicates that food insecurity is hazardous to the health of the general population and older adults, alike, because of its association with malnutrition, including under- and overnutrition; physical and mental health problems; health services utilization; chronic disease management; medication management; and poor food and nutrition intake leading to malnutrition [52]. Continuing to combat food insecurity with programs like the OAA Nutrition Services and finding new ways to connect food insecure persons with resources through the OAA and SNAP is important. These programs that promote food security need to be emphasized because they have the potential to promote health, reduce disease risk, and cultivate effective disease management [30].

References

- 1. Chernoff R. Nutrition and health promotion in older adults. J Gerontol Ser A Biol Sci Med Sci. 2001;56 suppl 2:47–53.
- 2. Sharkey JR. Diet and health outcomes in vulnerable populations. Ann N Y Acad Sci. 2008;1136(1):210-7.
- 3. Roberts Sb FPHMB, et al. COntrol of food intake in older men. JAMA. 1994;272(20):1601-6.
- 4. MacIntosh C, Morley JE, Chapman IM. The anorexia of aging. Nutrition. 2000;16(10):983-95.
- 5. Food security: World Health Organization; 2012 [updated October 12, 2012; cited 2012 October 13]. Available from: http://www.who.int/trade/glossary/story028/en/
- Holben D. Position of the American Dietetic Association: food insecurity in the United States. J Am Diet Assoc. 2010;110(9):1368–77.
- 7. Anderson S. Core indicators of nutritional state for difficult-to-sample populations. J Nutr. 1990;120 Suppl 11: 1559–600.
- Food insecurity in the US: measurement: United States Department of Agriculture: Economic Research Service; 2012 [cited 2012]. Available from: http://www.ers.usda.gov/topics/food-nutrition-assistance/food-security-in-theus/measurement.aspx#.UV35bKKG3eA
- Lee JS, Frongillo Jr EA. Factors associated with food insecurity among U.S. elderly persons: importance of functional impairments. J Gerontol Ser B Psychol Sci Soc Sci. 2001;56(2):S94–9.
- Lee JS, Frongillo EA. Nutritional and health consequences are associated with food insecurity among U.S. elderly persons. J Nutr. 2001;131(5):1503–9.
- Food insecurity in the US: history and background: United States Department of Agriculture: Economic Research Service; 2012 [cited 2012]. Available from: http://www.ers.usda.gov/topics/food-nutrition-assistance/foodsecurity-in-the-us/history-background.aspx#.UV35naKG3eA
- 12. National Research Council. Food insecurity and hunger in the United States. An assessment of the measure. Washington, DC: The National Academies Press; 2006.
- Food insecurity in the US: definitions of food insecurity: United States Department of Agriculture: Economic Research Service; 2012 [cited 2012]. Available from: http://www.ers.usda.gov/topics/food-nutrition-assistance/ food-security-in-the-us/definitions-of-food-security.aspx#.UV33YKKG3eA
- Food insecurity in the US: frequency of food insecurity: United States Department of Agriculture: Economic Research Service; 2012 [cited 2012]. Available from: http://www.ers.usda.gov/topics/food-nutrition-assistance/ food-security-in-the-us/frequency-of-food-insecurity.aspx#.UV341aKG3eA
- 15. Hamm MW, Bellows AC. Community food security: background and future directions. J Nutr Educ Behav. 2003;35(1):37–43.
- 16. Gottlieb R, Fisher A. Community food security and environmental justice: Searching for a common discourse. Agric Hum Values. 1996;13(3):23–32.
- 17. Haering SA, Syed SB. Community food security in United States cities: a survey of the relevant scientific literature. Baltimore, MD: Johns Hopkins Center for a Livable Future; 2009. Available from: http://www.jhsph.edu/research/ centers-and-institutes/johns-hopkins-center-for-a-livable-future/research/clf_publications/pub_rep_desc/CFS_ USA.html
- Cohen B, Andrews M, Kantor L. Community food security assessment toolkit. USDA Economic Research Service, 2002 Contract No.: E-FAN-02-013.
- 19. The Food Assistance National Input-Output Multiplier (FANIOM) Model and Stimulus Effects of SNAP. United States Department of Agriculture, 2010 Contract No. 103.

- 20. Cunnyngham K. State trends in supplemental nutrition assistance program eligibility and participation among elderly individuals. Washington, DC: Mathematical Policy Research; 2010.
- 21. Tanumihardjo SA, Anderson C, Kaufer-Horwitz M, Bode L, Emenaker NJ, Haqq AM, et al. Poverty, obesity, and malnutrition: an international perspective recognizing the paradox. J Am Diet Assoc. 2007;107(11):1966–72.
- 22. Brown KE. Nutritional assistance: additional efficiencies could improve services to older adults. Washington, DC: Government Accountability Office; 2011.
- 23. Committee on Nutrition Services for Medicare B, Food and Nutrition B. The role of nutrition in maintaining health in the nation's elderly: evaluating coverage of nutrition services for the medicare population. Washington, DC: The National Academies Press; 2000.
- 24. FAO, WFP, IFAD. The state of food insecurity in the world 2012. Economic growth is necessary but not sufficient to accelerate reduction of hunger and malnutrition. Rome, Italy: Food and Agriculture Organization; 2012.
- 25. Morley JE. Undernutrition in older adults. Fam Pract. 2012;29 Suppl 1:i89–93.
- 26. Chubinsky J, Carrozza MA. Obesity and food insecurity. In: Ludke RL, Obermiller PJ, Couto RA, editors. Appalachian health and well-being. Lexington, KY: University Press of Kentucky; 2012. p. 149–66.
- 27. Institute of Medicine. Hunger and obesity: understanding a food insecurity paradigm: workshop summary. Washington, DC: The National Academies Press; 2011.
- Jensen GL, Mirtallo J, Compher C, Dhaliwal R, Forbes A, Grijalba RF, et al. Adult starvation and disease-related malnutrition: a proposal for etiology-based diagnosis in the clinical practice setting from the International Consensus Guideline Committee. J Parenteral Enteral Nutr. 2010;34(2):156–9.
- 29. Waters DL, Baumgartner RN. Sarcopenia and obesity. Clin Geriatr Med. 2011;27(3):401-21.
- Kamp BJ, Wellman NS, Russell C. Position of the American Dietetic Association, American Society for Nutrition, and Society for Nutrition Education: food and nutrition programs for community-residing older adults. J Nutr Educ Behav. 2010;42(2):72–82.
- 31. Ziliak JP, Gunderson C. Food insecurity among older adults. Washington, DC: AARP; 2011.
- 32. Dean WR, Sharkey JR. Food insecurity, social capital and perceived personal disparity in a predominantly rural region of Texas: an individual-level analysis. Social Sci Med. 2011;72(9):1454–62.
- Coleman-Jensen A, Nord M, Andrews M, Carlson S. Household food security in the United States in 2011. Washington, DC: USDA, Economic Research Service; 2012.
- Brown KE. Older Americans act: more should be done to measure the extent of unmet need for services. Washington, DC: Government Accountability Office; 2011.
- 35. Ziliak J, Gundersen C, Haist M. The consequences causes future of senior hunger in America. Special report by the University of Kentucky Center for Poverty Research for the Meals on Wheels Association of America. Lexington, KY: University of Kentucky Center for Poverty Research; 2008. p. 2008.
- 36. Ziliak J, Gundersen C. Senior hunger in the United States: differences across states and rural and urban areas. Special report. Lexington, KY: University of Kentucky Center for Poverty Research; 2009.
- 37. Ziliak JP, Gundersen C. Multigenerational families and food insecurity. Lexington, KY: University of Kentucky Center for Poverty Research; 2012.
- Nelson K, Cunningham W, Andersen R, Harrison G, Gelberg L. Is food insufficiency associated with health status and health care utilization among adults with diabetes? J Gen Intern Med. 2001;16(6):404–11.
- Sullivan AF, Clark S, Pallin DJ, Camargo Jr CA. Food security, health, and medication expenditures of emergency department patients. J Emerg Med. 2010;38(4):524–8.
- 40. Food Nutrition Board IoM. Nutrition and healthy aging in the community: workshop summary. Washington, DC: The National Academies Press; 2012.
- Locher JL, Ritchie CS, Robinson CO, Roth DL, Smith West D, Burgio KL. A multidimensional approach to understanding under-eating in homebound older adults: the importance of social factors. Gerontologist. 2008;48(2): 223–34.
- 42. Payette H, Coulombe C, Boutier V, Gray-Donald K. Weight loss and mortality among free-living frail elders: a prospective study. J Gerontol A Biol Sci Med Sci. 1999;54(9):M440–5.
- Janssen I. Influence of sarcopenia on the development of physical disability: the Cardiovascular Health Study. J Am Geriatr Soc. 2006;54(1):56–62.
- 44. Yang Y, Brown CJ, Burgio KL, Kilgore ML, Ritchie CS, Roth DL, et al. Undernutrition at baseline and health services utilization and mortality over a 1-year period in older adults receiving medicare home health services. J Am Med Direct Assoc. 2011;12(4):287–94.
- 45. Newman AB, Yanez D, Harris T, Duxbury A, Enright PL, Fried LP, et al. Weight change in old age and its association with mortality. J Am Geriatr Soc. 2001;49(10):1309–18.
- 46. Ritchie CS, Burgio KL, Locher JL, Cornwell A, Thomas D, Hardin M, et al. Nutritional status of urban homebound older adults. Am J Clin Nutr. 1997;66(4):815–8.
- Seligman HK, Schillinger D. Hunger and socioeconomic disparities in chronic disease. N Engl J Med. 2010; 363(1):6–9.

- 9 Food Insecurity and Hunger Among Older Adults
- 48. Statistics FIFoA-R. Older Americans 2012: key indicators of well-being. In: statistics. Washington, DC: GPO; 2012.
- Barrett A, Schimmel J. Multiple service use among OAA Title III Program participants. Washington, DC: Mathematical Policy Research; 2010.
- Food insecurity in the US: community food security: United States Department of Agriculture: Economic Research Service; 2012 [cited 2012]. Available from: http://www.ers.usda.gov/topics/food-nutrition-assistance/foodsecurity-in-the-us/community-food-security.aspx#.UV35_aKG3eA
- 51. Dover SE, Buys DR, Allocca S, Locher JL. Farmers' market produce delivery program for mitigating nutritional risk in older adults. J Hunger Environ Nutr. 2013;8(1):106–8.
- Johnson MA, Dwyer JT, Jensen GL, Miller JW, Speakman JR, Starke-Reed P, et al. Challenges and new opportunities for clinical nutrition interventions in the aged. J Nutr. 2011;141(3):535–41.
Part III Common Clinical Conditions

Chapter 10 Obesity in Older Adults and Strategies for Weight Management

Dennis T. Villareal and Krupa Shah

Key Points

- The increasing prevalence of obese older adults is a major public health issue.
- Obesity causes frailty in older adults by exacerbating the age-related decline in physical function.
- Treatment plans for obese older adults should include lifestyle interventions such as weight loss, behavior modification, and exercise therapy to improve physical function and quality of life and reduce medical complications associated with obesity.
- While treatment of obesity may improve a number of health parameters, the potential for adverse effects of weight loss on bone and muscle mass must also be taken into consideration.

Keywords Obesity • Older adults • Frailty • Weight loss • Exercise • Behavior modification • Sarcopenia

Introduction

Obesity is defined as an unhealthy excess of body fat, which increases the risk of morbidity and premature mortality. Obesity is a growing concern among adults. It has not only increased in prevalence, but it has also been associated with significant morbidity and mortality. Some of its medical risks include hypertension, diabetes, hyperlipidemia, coronary artery disease, and osteoarthritis. More so in older adults, obesity exacerbates the age-related decline in physical function, impairs quality of life, and leads to frailty. The current therapeutic and management tools designed for weight management in older persons include lifestyle intervention (diet, physical activity, and behavior modification), pharmacotherapy, and surgery. Current evidence suggests that weight-management strategies in obese

D.T. Villareal, M.D. (🖂)

Department of Medicine, Baylor College of Medicine and Michael E. De Bakey VA Medical Center, 1 Baylor Plaza, Houston, TX 77030, USA e-mail: dennis.villareal@bcm.edu

K. Shah, M.D., M.P.H. Department of Medicine and Orthopedics, Highland Hospital, University of Rochester School of Medicine and Dentistry, Rochester, NY 14692, USA e-mail: Krupa_Shah@URMC.Rochester.edu older adults improves physical function, quality of life, and reduces medical complications. The treatment must consider the potential adverse effects of weight loss on bone and muscle mass.

This chapter will review the clinical issues related to obesity in older adults and provide health professionals with the appropriate weight-management guidelines on the basis of current evidence.

Obesity: An Epidemic

Obesity among older adults continues to grow in prevalence in the United States. Data from the National Health and Nutrition Examination Survey (NHANES) indicate that approximately one-third of United States adults are obese [1]. Likewise, in developed countries, the prevalence of obesity is increasing among older adults. The underlying reasons for the increased prevalence are an increase in the older person population, and an increase in the percentage of obesity in that population. Past studies have compared point-in-time statistics of the American older adult population and highlighted the increase in prevalence of obesity. For example, in a 10-year period between 1991 and 2000, obesity was found to grow from 14.7 to 22.9 % in the 60–69-year age group, while obesity grew from 11.4 to 15.5 % in the >70-year age group. This represents an increase of 56 and 36 % in the respective age groups [2]. More recent estimates suggest that 37 % of adults 65 years of age or older are obese (Body Mass Index, $BMI \ge 30 \text{ kg/m}^2$), and such prevalence is expected to become even more evident with the aging of "baby boomers" [3]. The prevalence of obesity in older adults will continue to challenge our health care systems [4]. Furthermore, obesity poses an increasing problem for long-term care facilities [5]. On a positive note, obesity is less likely to develop in the very old population (>80-year-olds). In this age group, the prevalence rate of obesity is considerably lower than for younger cohorts. The relatively low prevalence of obesity after 80 years of age could be due to the survival advantage of being lean [6]. Nonetheless, overall, more than 15 % of the older American population is obese, and obesity is more common in older women than in men [2]. Moreover, concern about the prevalence of geriatric obesity is not contained to the United States. It is an increasing problem of older populations throughout the world [7].

Pathophysiology of Obesity

Aging is associated with marked changes in body composition. After 30 years of age, fat-free mass (FFM), which is comprised predominantly of muscle, progressively decreases, whereas fat mass increases. FFM reaches its peak during the third decade of life, while fat mass (FM) reaches its peak during the seventh decade [8]. Subsequently, after about age 70, both indices (FFM and fat mass) decrease. Aside from quantitative changes of FFM and fat mass, aging is also associated with a redistribution of body fat and FFM. The intra-abdominal fat increases, while the subcutaneous fat and total body fat decrease [9].

Body fat is accumulated when energy input exceeds energy expenditure. The intake of energy does not change or may even decline with aging. Energy output comprises the resting metabolic rate (accounts for ~70 %), the thermal effect of food (~10 %), and physical activity (20 %). Aging is associated with a decrease in all major components of energy output. Resting metabolic rate decreases by 3 % every decade after 20 years of age. About three-fourths of this decline can be accounted for by a loss in FFM [10]. The thermal effect of food is 20 % lower in older men than in younger men [11]. Physical activity decreases with increasing age, and it accounts for about one-half of the decrease in energy output that occurs with aging [12].

As one ages, the growth hormone and testosterone production decreases, which results in a reduction in FFM and increased accumulation of fat mass [13]. Thyroid hormone-induced oxidative bursts are decreased with aging. Changes in neurohumoral modulators of appetite and body composition such as leptin and ghrelin have also been implicated as causes of obesity in later life. Resistance to leptin could result in a diminished ability to downregulate appetite [14]. These changes in hormone levels with aging could play an important role in the pathogenesis of obesity.

Measuring Overweight and Obesity

It is difficult to accurately measure body fat mass in most clinical settings because such assessments require the use of sophisticated technologies that are not readily available. Two measures for assessing overweight and total body fat content are widely used and accepted as simple methods to classify medical risk with regards to body fat; they are the body mass index (BMI) and waist circumference.

Body Mass Index

BMI is calculated as weight (kg)/height squared (m²). The BMI is used to assess overweight and obesity and to monitor changes in body weight status. It allows meaningful comparisons of weight status within and between populations. However, in older adults, age-related changes in body composition and loss of height caused by compression of vertebral bodies and kyphosis alter the relationship between BMI and percentage body fat. Therefore, at any given BMI value, changes in body composition tend to underestimate fatness, whereas the loss of height would tend to overestimate fatness.

Waist Circumference

The presence of excess fat in the abdomen out of proportion to total body fat is an independent predictor of comorbidities such as cardiovascular disease, diabetes, and hypertension [15]. Men with a waist circumference of >40 in. and women with a waist circumference of >35 in. are considered to have increased disease risk.

Table 10.1 incorporates both BMI and waist circumference in the classification of overweight and obesity and provides an indication of relative disease risk [16].

Health Implications of Obesity

Adverse Effects of Obesity

Obesity is associated with a number of health hazards. Some adverse effects include increased mortality, health complications, poor quality of life, and disability. These hazards are discussed in detail below.

	BMI (kg/m ²)	Obesity class	Disease risk ^a (Relative to normal weight and waist circumference)	
			Men <40 in. Women <35 in.	Men >40 in. Women >35 in.
Underweight	<18.5		_	-
Normal ^b	18.5-24.9		-	-
Overweight	25.0-29.9	Ι	Increased	High
Obesity	30.0-34.9	II	High	Very high
	35.0-39.9		Very high	Very high
Extreme obesity	≫40	III	Extremely high	Extremely high

Table 10.1 Classification of overweight and obesity by BMI, waist circumference, and associated disease risk

^aDisease risk for type 2 diabetes, hypertension, and CVD

^bIncreased waist circumference can also be a marker for increased risk even in persons of normal weight

Mortality

Obesity was associated with significantly higher all-cause mortality relative to the normal weight BMI in both younger and older adults [17]. Although the relative risk of death associated with obesity is greater for younger adults than for older ones [18, 19], a high BMI increases absolute mortality and health risks linearly up to 75 years of age [20]. That is, from a clinical standpoint, the health complications associated with obesity increase linearly with increasing BMI until the age of 75. The relationship of obesity in >75 years of age with total mortality is unclear. Some previous epidemiological studies do not show that excess body weight is detrimental to mortality in advancing age [21, 22]. However, underlying diseases that can themselves increase the risk of early mortality may cause the underestimation of the relation between obesity and mortality in older adults. Since those who are susceptible to the effects of obesity die at a younger age, the surviving group of obese older adults are said to be the "resistant" survivors. Furthermore, waist circumference, an indicator of central adiposity, is identified as a potentially good risk factor for mortality in obese elderly adults [23].

Comorbid Disease

Obesity and increased visceral fat are associated with increased morbidity and poor quality of life. Most studies evaluating obesity-related complications focus on middle-aged and younger adults. The prevalence of the medical complications associated with obesity, such as hypertension, diabetes, cardiovascular disease, and osteoarthritis, increases with age. Therefore, obesity and weight gain during middle age may contribute to medical complications and development of obesity-associated chronic disease, and subsequent increased health care expenditures that occur during old age [24].

Metabolic Abnormalities

There is an age-related increase in the prevalence of all of the components of the metabolic syndrome. Relative to those who are 20–34 years of age, the odds of having metabolic syndrome for those who are over 65 years of age were reported to be 5.8 in men and 4.9 in women [25]. Additionally, increased abdominal fat is independently associated with metabolic syndrome in adults aged 70–79 years [26]. Fasting plasma glucose increases by 1–2 mg/dL and postprandial glucose by 10–20 mg/dL for each decade of age after 30 years. Accordingly, the prevalence of type 2 diabetes mellitus based on standard criteria is highest in older persons [27]. The age-related increase in fat

and, more importantly, visceral fat, could be the main causative factor for the increased prevalence of diabetes mellitus and insulin resistance in the elderly. At the age of 65, there is a strong influence of BMI on the remaining lifetime risk of diabetes, with the increased risk ranging from ~3 % with BMI < 18.5 kg/m², 10 % with BMI 18.5–25 kg/m², 14 % with BMI 25–30 kg/m², 29 % with BMI 30–35 kg/m² and 35 % with BMI \geq 35 kg/m² [28].

Hypertension is extremely prevalent in the older population, affecting 65 % of all persons aged >60 years [29]. Obesity and high blood pressure continue to be correlated, even in old age [30]. Obesity-related dyslipidemia (i.e., low HDL cholesterol and high serum triglyceride concentrations) is seen in both younger and older adults. In the United States, 35–42 % of white men and women who are over 65 years of age with metabolic syndrome have low HDL cholesterol (<40 mg/dL in men and <50 mg/dL in women) and high triglyceride (>150 mg/dL) concentrations [25]. Data from longitudinal studies suggest that obesity increases the risk of cardiovascular disease in older men. Elevated BMI in older 70-year-old men and women was associated with an increase in new cases of coronary artery disease, fatal and nonfatal myocardial infarction, and cardiovascular disease mortality during 12–15 years of observation [31] in older men but not in older women.

Arthritis

Osteoarthritis (OA) is the most common type of arthritis and its prevalence increases progressively with age in both sexes in parallel with the increase in body weight and fat observed with aging. The age-related increase in prevalence of OA presumably reflects bodily changes as a result of a lifetime of being overweight, which results in chronic mechanical strain on weight-bearing joints [32, 33]. In a population-based study of older adults, with a mean age of 73, the relative risk of developing knee OA increased from 0.1 for a BMI lower than 20 kg/m² to 13.6 for a BMI of 36 kg/m² or higher [34].

Pulmonary Abnormalities

Obesity is associated with obstructive sleep apnea (OSA), obesity-hypoventilation syndrome, and pulmonary function abnormalities [35]. Increased fat on the chest wall decreases lung compliance, increases the work of breathing, and reduces ventilation [36]. The prevalence of OSA increases with age. Both waist circumference and waist changes were the most powerful predictors of OSA in older obese and normal-weight men in a 30-year follow-up study [37].

Urinary Incontinence

The prevalence of urinary incontinence increases after the age of 65 and affects 15–30 % of that population. Obesity further exacerbates the high prevalence of urinary incontinence in older adults, which has been shown to be directly associated with elevated BMI [38, 39].

Cancer

Obesity is a risk factor for several types of cancer, including breast, colon, gallbladder, pancreas, and bladder amongst both men and women, more so in older than younger adults [40]. A study in older women has shown that breast cancer occurs more frequently in obese older women than all older women [40]. Obese postmenopausal women are also at higher risk of developing malignant melanoma and endometrial cancer than their nonobese counterparts [41].



Fig. 10.1 Muscle quality (strength per muscle mass) in nonobese nonfrail, nonobese frail, and obese elderly subjects. Reprinted by permission from Macmillan Publishers Ltd: Obesity. Villareal DT, Banks M, Siener C, Sinacore DR, Klein S. Physical frailty and body composition in obese elderly men and women, 12(6), copyright 2004

Functional Impairment and Quality of Life

Aging causes a progressive decline in physical function because of a continued decline in muscle mass, strength, and power, and an increase in joint instability and arthritis [42]. These functional impairments affect activities of daily living, decrease quality of life, and lead to an increased utilization of health care services. Obesity has important functional implications in older adults because it worsens their age-related decline in physical function. Data from both cross-sectional [43–45] and longitudinal studies [46–48] have consistently demonstrated a strong link between increasing BMI and worsening physical function in older persons. High BMI is associated with self-reported impairment in ADLs, limitations in mobility, decreased physical performance, and increased risk for functional decline [45–50]. Moreover, obesity is associated with increasing nursing home admissions [51, 52].

Although obesity is associated with increases in FM, aging is associated with a decline in FFM (primarily skeletal muscle) and function, referred to as sarcopenia [53] and obesity does not appear to protect against sarcopenia. In one study [54], the prevalence of sarcopenia in obese persons increased with age, suggesting that many obese persons maintain a constant fat mass while losing muscle mass. In another study [55], obese older adults were found to have sarcopenia based on lower relative muscle mass and low muscle strength per muscle area (low muscle quality, Fig. 10.1), despite having more than adequate body weight, which is opposite of the stereotypical frail older adult. Their functional performance, aerobic capacity, strength, balance, and walking speed were as severely reduced as the frail nonobese adults [55]. Thus, obesity in older adults acts synergistically with sarcopenia (sarcopenic obesity; see also Chap. 6) to augment disability. Accordingly, the "sarcopenic-obese" individual has two problems that lead to frailty: (1) decreased muscle mass and strength, which occur with aging, and (2) a need to carry greater weight due to excess body fat [55, 56]. Figure 10.2 is a cross-sectional MRI image from the mid-thigh in a frail obese older adult. This figure demonstrates the excessive adipose tissue infiltration of skeletal muscle mass with obesity.



Fig. 10.2 Cross-sectional MRI image from the mid-thigh in a frail obese female participant (76 years of age). *IMAT* intermuscular adipose tissue, *SAT* subcutaneous adipose tissue, *SM* skeletal muscle

In one study [55], 96 % of community-living older adults with BMIs >30 were frail, as determined by physical performance test scores [57], peak oxygen consumption [58], and self-reported ability to perform activities of daily living [59]. Data from another study [60] also demonstrated that obesity was associated with a marked increased risk of frailty (odds ratio=3.5), determined by weakness, slowness, weight loss, low physical activity, and exhaustion. In another study [61] obesity was identified as one of the five modifiable risk factors that predict functional decline in both vigorous and basic activities among older women. Recently in a study of community dwelling older adults, frailty was related to BMI in a U-shaped manner (i.e., increased frailty in people with extremes of low or high BMI). However, in people with large WC (\geq 35 in. in women and \geq 40 in. in men), frailty was shown to exist in all BMI categories [62].

Beneficial Effects of Obesity

Increased body weight is associated with increased bone mineral density (BMD) and decreased osteoporosis and hip fracture in older men and women, whereas the converse is true for being of low body weight. Both body fat mass and FFM are directly correlated with BMD. Although the increase in BMD has been attributed to mechanical stress on the weight-bearing skeleton, the protective effects have also been observed in non-weight-bearing bones [63]. Therefore, hormonal factors that are increased in obese persons, such as circulating estrogens, insulin, and leptin, might contribute to the osteoprotective effects of obesity, by stimulating bone formation and inhibiting bone resorption. The increase in both BMD and the extra cushioning effect of the fat surrounding crucial areas such as the hip might provide protection against hip fracture during a fall in obese older adults [64]. On the other hand, recent studies also suggest that excess adiposity could be detrimental to bone and also increases fall risk [65, 66].

Effects of Intentional Weight Loss in Older Adults

Body Composition

Loss of body weight is accompanied by a decrease in both FM and FFM. Therefore, it is possible that weight loss in obese older persons could increase sarcopenia by worsening the age-related loss of muscle mass. In younger adults, ~75 % of diet-induced weight loss is composed of fat tissue and ~25 % is composed of FFM [67]. The relative amount of diet-induced weight loss as FFM and fat mass in older men and women is similar to that observed in younger adults [68]. Therefore, diet-induced weight loss does not produce a disproportionate loss of lean tissue in old persons. Despite much evidence linking high body fat to functional disability [45, 47, 48], weight loss has not been typically instituted in obese older persons because of the fear that it will exacerbate sarcopenia. Additionally, it is a general belief among many geriatricians that some "reserve" of body fat is advantageous in the older people particularly if they are hospitalized [69].

In a randomized controlled trial conducted in obese older subjects, there was no significant difference in loss of FFM after a diet-induced weight loss plus regular exercise compared with the control group who did not lose weight. These encouraging findings suggest that regular exercise can attenuate a diet-induced loss of FFM in older persons [70].

Medical Complications

Data from young and middle-aged adults show that weight loss improves or normalizes metabolic abnormalities associated with obesity. Similarly, it has been shown that moderate weight loss in obese older adults simultaneously decreases an array of metabolic coronary heart disease risk factors [71].

Physical Function and Quality of Life

Moderate weight loss in conjunction with physical activity improves physical function and healthrelated quality of life in obese older persons. Data from studies conducted in overweight and obese older persons with or without joint disease have shown that the combination of moderate diet-induced weight loss and exercise therapy improved both subjective and objective measures of physical function and health-related quality of life and had a greater beneficial effect than did either diet or exercise interventions alone [70, 72–74]. These findings suggest that obesity is a reversible cause of frailty and impaired quality of life in older adults.

Recently, in a 1-year randomized controlled clinical trial, 107 obese adults aged 65 years and older were recruited to investigate the independent and combined effects of weight loss and exercise on physical function, body composition, and quality of life [75]. Participants were randomized into a weight management program, exercise training program, weight management plus exercise training program, and a control group. The weight management program consisted of a balanced diet producing an energy deficit of 500–750 kcal/day in participants' daily requirements. The exercise intervention included both aerobic and resistance training components. Results demonstrated that while physical function improved in all of the intervention groups compared to the control group, the group simultaneously adopting the weight management and exercise training programs demonstrated a significantly higher physical function compared to the other interventions. Moreover, the group that received the combined intervention also tended to demonstrate more benefits in terms of reduced loss of lean body mass and bone mineral density, as well as improved aerobic capacity, strength, balance,

and gait speed. These findings suggest that the combination of weight loss and exercise programs is more effective at preventing frailty and preserving quality of life for obese older adults compared to interventions focused on one single modality.

In this context, it is important to underline that the combination of weight loss and exercise programs is safe, and the exercise benefits may effectively compensate the reduction of lean mass due to the weight loss program [76] (see also Chap. 22). Further studies are needed to determine whether weight loss can be maintained beyond 1 year and may be able to prevent major health-related outcomes (including mortality and nursing home admissions) in obese older adults. These studies should focus on evaluating whether long-term weight maintenance is likely to produce the most meaningful changes in health outcomes in obese older adults.

Mortality

It has been observed in several population-based studies that community-dwelling older adults who lost weight, or who experienced weight variability, had an increased relative mortality risk compared with those who were weight stable [37].

However, most studies have inherent weaknesses, such as not reporting whether the observed weight changes were intentional or unintentional, relying upon self-reported weight change, or not distinguishing between weight loss in obese and lean subjects. Moreover, a recent randomized controlled clinical trial demonstrated that intentional dietary weight loss was not significantly associated with increased all-cause mortality over 12-year of follow-up in older overweight or obese adults [77].

Bone Mineral Density

Weight loss can have adverse effects on bone mass. Previous interventional studies conducted in young and middle-aged adults reported that weight loss causes bone loss that may be proportional to the amount of weight loss [78–80]. However, it is not known whether the bone loss associated with intentional weight loss increases the risk of osteoporotic fractures in obese persons. One study showed that diet-induced weight loss, but not exercise-induced weight loss, is associated with reductions in BMD at weight-bearing sites, suggesting that exercise should be an important component of a weight-loss program to offset adverse effects of diet-induced weight loss on bone [81]. In a more recent study, regular exercise was able to attenuate weight-loss-induced bone loss [76]. It is thought that this beneficial effect may be specific for sites involved in weight-bearing exercise [82], mediated by prevention of weight-loss-induced increase in sclerostin (Sclerostin is a secreted Wnt antagonist that inhibits osteoblastic proliferation and differentiation, thus reducing bone formation, an inhibitor of bone formation secreted by osteocytes) [83]. Therefore, including exercise as part of a weight-loss program is particularly important in older persons to reduce bone loss.

Interventions and Treatment

Weight loss in obese persons of any age can improve obesity-related medical complications, physical function, and quality of life. In older adults, improving physical function and quality of life may be the most important goals of therapy. The current therapeutic tools and recommendations available for weight management in older persons are (1) lifestyle intervention involving diet, physical activity, and behavior modification; (2) pharmacotherapy; and (3) surgery.

Weight Management Intervention

Weight management intervention is just as effective in older as in younger subjects [70, 72–74]. The combination of an energy-deficit producing diet, increased physical activity, and behavior therapy causes moderate weight loss and is associated with a lower risk of treatment-induced complications. Weight-management therapy that minimizes muscle and bone losses is recommended for older adults who are obese and who have functional impairments or metabolic complications that can benefit from weight loss.

Diet Therapy

In order for weight loss to be successful, an energy deficit must be achieved. A low-calorie diet that reduces energy intake by 500–750 kcal/day results in a weight loss of 0.4–0.9 kg (about 1–2 lb)/week and a weight loss of 8–10 % by 6 months. The diet should contain 1.0 g/kg high-quality protein/day [70], and multivitamin, and mineral supplements to ensure that all daily recommended requirements are met, including daily intakes of 1,500 mg calcium and 1,000 IU vitamin D to prevent bone loss. Very low-calorie diets (<800 kcal/day) should be avoided because of an increased risk of medical complications. Also, depending on the patient's cardiovascular risk status, the diet therapy should be consistent with the National Cholesterol Education Program Expert Panel (Adult Treatment Panel III)'s Therapeutic Lifestyle Changes Diet and or possibly the Dietary Approaches to Stop Hypertension (DASH diet) [84].

Referral to a registered dietitian, who has weight-management experience, is often necessary to ensure that appropriate nutritional counseling is provided. Patients should be educated on food composition, preparation, and portion control and their food preferences should be supported to improve compliance.

Successful weight loss and maintenance program should be based on a sound scientific rationale. The program should be safe and nutritionally adequate, as well as practical and applicable to patient's social and ethnic background.

Physical Activity

Introducing an exercise component early in the treatment course can improve physical function and can ameliorate frailty in the older adult [70]. The exercise program should be individualized according to a person's medical conditions and disability (again, see Chap. 22). The program should start at a low-to-moderate intensity, duration, and frequency to promote adherence and avoid musculoskeletal injuries. If possible, the program should be gradually progressed over a period of several weeks or months to a longer, more frequent, and more vigorous effort. The goals of regular exercise in obese older persons are to increase flexibility, endurance, and strength; therefore, a multicomponent exercise program that includes stretching, aerobic activity, and strength exercises is recommended. Even very old or frail persons can participate in these types of physical activities.

Behavior Modification

Clinicians should help obese older adults set personal goals, monitor progress, and use motivational strategies to improve adherence to the weight-loss program (see also Chap. 1). The cognitive behavioral therapy strategies that should be considered include goal setting, self-monitoring, social support, stimulus control techniques, and problem-solving skills. Lifestyle and behavior modification can be

facilitated by counseling from a behavioral therapist, exercise specialist, or dietitian who has weightmanagement experience.

Implementation of changes in the diet and activity habits of older adults may be challenging. An increased burden of disease, adverse quality of life, depression, hearing and visual difficulties, and cognitive dysfunction may make it difficult to change one's lifestyle. This increase in chronic disabilities with aging reduces physical activity and exercise capacity. Common geriatric situations, such as depression, cognitive impairment, dependency on others, institutionalization, widowhood, loneliness, and isolation should be addressed, because these factors can make it more difficult to adopt or adhere to lifestyle changes designed to lose weight. Lifestyle-change programs should also encourage participation by family members and care providers for better compliance.

Pharmacotherapy

Since most clinical trials that evaluate the use of pharmacotherapy excluded older adults or included only a small number of older adults, the available data are insufficient to determine the efficacy and safety of pharmacotherapy in this population.

The use of pharmacological agents to treat obesity can add an additional burden on older persons. Many obese older patients are already taking multiple medications for other diseases, which can increase the chances of nonadherence, drug–drug interactions, and errors with obesity pharmacotherapy.

Moreover, potential side effects can have more serious consequences in older adults. Some examples of weight loss drugs that are FDA approved include phenteramine, orlistat, and a recently approved drug Qysmia (combination of phentermine and topiramate extended-release). These drugs are seldom covered by health insurance or Medicare, which can add an additional financial burden in older patients who have a fixed income. A thorough review of all the medications should be conducted for older adults who are obese, because some medications may cause weight gain (e.g., antipsychotic, antidepressants, anticonvulsants, or steroids). Furthermore, weight- loss-induced clinical improvements might require changes in medications to avoid iatrogenic complications.

Weight-Loss Surgery

A few studies have provided information on the effectiveness and safety of bariatric surgery among older adults. Data from case series that evaluate the effect of bariatric surgery in patients who are over 60 years of age suggest that the relative weight loss and improvement in obesity-related medical complications are lower, whereas the perioperative morbidity and mortality are greater, in older compared to younger patients [85]. However, bariatric surgery can result in considerable weight loss and marked improvements in obesity-related physical impairment and medical complications in the older patients such as reversal of diabetes. The laparoscopic-adjustable gastric band is associated with fewer serious complications and a lower mortality rate; therefore the gastric band may be a better choice than the Roux-en-Y gastric bypass for older patients. However, it should be emphasized that the efficacy and safety of these procedures have not been compared in randomized trials in older adults. Therefore, careful patient selection, intensive preoperative education, and expert operative and perioperative management are advised. Surgery should be considered in selected older adults who have disabling obesity that can be ameliorated with weight loss and who meet the criteria for surgery. The preoperative evaluation should include an assessment for depression, which is common amongst older adults and could influence outcome. Postoperative management should include monitoring for nutritional and metabolic problems; particularly, vitamin B_{12} deficiency, iron deficiency, and osteoporosis.

Conclusion

The increasing prevalence of obese older adults is a major public health issue. Decreased muscle mass with aging and the burden of extra body mass due to obesity make it particularly difficult for obese older adults to function independently and may lead to the secondary complication of frailty. Treatment plans for obese older adults should include comprehensive lifestyle interventions such as dietary changes that promote weight loss, behavior modification, and exercise therapy to improve physical function, quality of life, and the medical complications associated with obesity. Finally, the treatment must consider the potential adverse effects of weight loss on bone and muscle mass, as well as physical function and quality of life. Randomized controlled trials to determine health benefits and risks from long-term weight management in obese older adults are necessary.

Clinical Recommendations for Older Adult Weight-loss Therapy

- 1. Initial assessment
 - A thorough medical history, physical examination, appropriate laboratory tests, and review of medications should be conducted to assess the patient's current health and comorbidity risks.
 - Additional information such as the patient's readiness to lose weight, previous attempts at weight loss, and current lifestyle habits should be collected before initiating weight-loss therapy.
 - Clinicians should help obese older adults set their personal goals and welcome participation by family members and care providers.
 - Clinicians should individualize the weight-loss plan after taking into account the special needs of this population.
- 2. Diet therapy
 - Advocate a modest reduction in energy intake (500–750 kcal/day) containing 1.0 g/kg highquality protein/day, multivitamin, and mineral supplements (including 1,500 mg calcium and 1,000 IU vitamin D/day).
 - Consider referrals to a registered dietitian for appropriate nutritional counseling and education.
- 3. Behavior therapy
 - Behavior therapy should highlight both diet and exercise—the integral parts of weight loss therapy and weight maintenance. The self-monitoring of nutrient intake and better understanding of physical activity accomplishes this task.
 - Consider referring to a behavioral therapist for counseling.
 - Stress management, stimulus control, problem solving, contingency management, and social support should be addressed.
- 4. Exercise therapy
 - Clinicians should assess the need for stress test before any physical activity.
 - Advocate an exercise program that is gradual, individualized, and monitored.
 - A multicomponent exercise program including stretching, aerobic activity, and strength exercises is recommended.

5. Additional recommendations

- Advocate a combination of energy deficit-producing diet, increased physical activity, and behavior therapy. Such combinations are associated with the low risk of treatment-induced complications.
- Bariatric surgery may be an option for patients who have failed multiple weight-loss attempts.
- Weight maintenance efforts should be implemented once weight-loss goals have been achieved.

References

- Flegal KM, Carroll MD, Ogden CL, Johnson CL. Prevalence and trends in obesity among US adults, 1999-2000. JAMA. 2002;288(14):1723–7.
- Mokdad AH, Bowman BA, Ford ES, Vinicor F, Marks JS, Koplan JP. The continuing epidemics of obesity and diabetes in the United States. JAMA. 2001;286(10):1195–200.
- Flegal KM, Carroll MD, Ogden CL, Curtin LR. Prevalence and trends in obesity among US adults, 1999-2008. JAMA. 2010;303(3):235–41.
- Arterburn DE, Crane PK, Sullivan SD. The coming epidemic of obesity in elderly Americans. J Am Geriatr Soc. 2004;52(11):1907–12.
- 5. Lapane KL, Resnik L. Obesity in nursing homes: an escalating problem. J Am Geriatr Soc. 2005;53(8):1386-91.
- Wallace JI, Schwartz RS. Involuntary weight loss in elderly outpatients: recognition, etiologies, and treatment. Clin Geriatr Med. 1997;13(4):717–35.
- 7. Kopelman PG. Obesity as a medical problem. Nature. 2000;404(6778):635-43.
- Gallagher D, Visser M, De Meersman RE, Sepulveda D, Baumgartner RN, Pierson RN, et al. Appendicular skeletal muscle mass: effects of age, gender, and ethnicity. J Appl Physiol. 1997;83(1):229–39.
- Beaufrere B, Morio B. Fat and protein redistribution with aging: metabolic considerations. Eur J Clin Nutr. 2000;54 Suppl 3:S48–53.
- 10. Tzankoff SP, Norris AH. Effect of muscle mass decrease on age-related BMR changes. J Appl Physiol. 1977; 43(6):1001–6.
- 11. Schwartz RS, Jaeger LF, Veith RC. The thermic effect of feeding in older men: the importance of the sympathetic nervous system. Metabolism. 1990;39(7):733–7.
- 12. Elia M, Ritz P, Stubbs RJ. Total energy expenditure in the elderly. Eur J Clin Nutr. 2000;54 Suppl 3:S92-103.
- 13. Schwartz RS. Trophic factor supplementation: effect on the age-associated changes in body composition. J Gerontol A Biol Sci Med Sci 1995;50 Spec No: 151–6.
- Enriori PJ, Evans AE, Sinnayah P, Cowley MA. Leptin resistance and obesity. Obesity (Silver Spring). 2006;14 Suppl 5:254S–8.
- 15. Kissebah AH, Krakower GR. Regional adiposity and morbidity. Physiol Rev. 1994;74(4):761-811.
- 16. Obesity: preventing and managing the global epidemic. Report of a WHO consultation. World Health Organ Tech Rep Ser 2000; 894: i-253.
- Flegal KM, Kit BK, Orpana H, Graubard BI. Association of all-cause mortality with overweight and obesity using standard body mass index categories: a systematic review and meta-analysis. JAMA. 2013;309(1):71–82.
- Flegal KM, Graubard BI, Williamson DF, Gail MH. Cause-specific excess deaths associated with underweight, overweight, and obesity. JAMA. 2007;298(17):2028–37.
- Calle EE, Thun MJ, Petrelli JM, Rodriguez C, Heath Jr CW. Body-mass index and mortality in a prospective cohort of U.S. adults. N Engl J Med. 1999;341(15):1097–105.
- Villareal DT, Apovian CM, Kushner RF, Klein S. Obesity in older adults: technical review and position statement of the American Society for Nutrition and NAASO, The Obesity Society. Am J Clin Nutr. 2005;82(5):923–34.
- 21. Troiano RP, Frongillo Jr EA, Sobal J, Levitsky DA. The relationship between body weight and mortality: a quantitative analysis of combined information from existing studies. Int J Obes Relat Metab Disord. 1996;20(1):63–75.
- 22. Allison DB, Gallagher D, Heo M, Pi-Sunyer FX, Heymsfield SB. Body mass index and all-cause mortality among people age 70 and over: the Longitudinal Study of Aging. Int J Obes Relat Metab Disord. 1997;21(6):424–31.
- Donini LM, Savina C, Gennaro E, De Felice MR, Rosano A, Pandolfo MM, et al. A systematic review of the literature concerning the relationship between obesity and mortality in the elderly. J Nutr Health Aging. 2012;16(1):89–98.
- Daviglus ML, Liu K, Yan LL, Pirzada A, Manheim L, Manning W, et al. Relation of body mass index in young adulthood and middle age to Medicare expenditures in older age. JAMA. 2004;292(22):2743–9.

- Park YW, Zhu S, Palaniappan L, Heshka S, Carnethon MR, Heymsfield SB. The metabolic syndrome: prevalence and associated risk factor findings in the US population from the Third National Health and Nutrition Examination Survey, 1988-1994. Arch Intern Med. 2003;163(4):427–36.
- 26. Goodpaster BH, Krishnaswami S, Harris TB, Katsiaras A, Kritchevsky SB, Simonsick EM, et al. Obesity, regional body fat distribution, and the metabolic syndrome in older men and women. Arch Intern Med. 2005;165(7): 777–83.
- Kahn SE, Schwartz RS, Porte Jr D, Abrass IB. The glucose intolerance of aging. Implications for intervention. Hosp Pract. 1991;26(4A):29–38.
- Narayan KM, Boyle JP, Thompson TJ, Gregg EW, Williamson DF. Effect of BMI on lifetime risk for diabetes in the U.S. Diabetes Care. 2007;30(6):1562–6.
- Hajjar I, Kotchen TA. Trends in prevalence, awareness, treatment, and control of hypertension in the United States, 1988-2000. JAMA. 2003;290(2):199–206.
- Masaki KH, Curb JD, Chiu D, Petrovitch H, Rodriguez BL. Association of body mass index with blood pressure in elderly Japanese American men. The Honolulu Heart Program. Hypertension. 1997;29(2):673–7.
- Dey DK, Lissner L. Obesity in 70-year-old subjects as a risk factor for 15-year coronary heart disease incidence. Obes Res. 2003;11(7):817–27.
- 32. Badley EM, Ansari H. Arthritis and arthritis-attributable activity limitations in the United States and Canada: a cross-border comparison. Arthritis Care Res (Hoboken). 2010;62(3):308–15.
- 33. Al SS, Graham JE, Kuo YF, Goodwin JS, Markides KS, Ottenbacher KJ. Obesity and disability: relation among older adults living in Latin America and the Caribbean. Am J Epidemiol. 2010;171(12):1282–8.
- Coggon D, Reading I, Croft P, McLaren M, Barrett D, Cooper C. Knee osteoarthritis and obesity. Int J Obes Relat Metab Disord. 2001;25(5):622–7.
- Lazarus R, Sparrow D, Weiss ST. Effects of obesity and fat distribution on ventilatory function: the normative aging study. Chest. 1997;111(4):891–8.
- 36. Franssen FM, O'Donnell DE, Goossens GH, Blaak EE, Schols AM. Obesity and the lung: 5. Obesity and COPD. Thorax. 2008;63(12):1110–7.
- Carmelli D, Swan GE, Bliwise DL. Relationship of 30-year changes in obesity to sleep-disordered breathing in the Western Collaborative Group Study. Obes Res. 2000;8(9):632–7.
- Brown JS, Seeley DG, Fong J, Black DM, Ensrud KE, Grady D. Urinary incontinence in older women: who is at risk? Study of Osteoporotic Fractures Research Group. Obstet Gynecol. 1996;87(5 Pt 1):715–21.
- Hunskaar S. A systematic review of overweight and obesity as risk factors and targets for clinical intervention for urinary incontinence in women. Neurourol Urodyn. 2008;27(8):749–57.
- 40. Wolk A, Gridley G, Svensson M, Nyren O, McLaughlin JK, Fraumeni JF, et al. A prospective study of obesity and cancer risk (Sweden). Cancer Causes Control. 2001;12(1):13–21.
- Reeves GK, Pirie K, Beral V, Green J, Spencer E, Bull D. Cancer incidence and mortality in relation to body mass index in the Million Women Study: cohort study. BMJ. 2007;335(7630):1134.
- 42. Jordan JM, Luta G, Renner JB, Linder GF, Dragomir A, Hochberg MC, et al. Self-reported functional status in osteoarthritis of the knee in a rural southern community: the role of sociodemographic factors, obesity, and knee pain. Arthritis Care Res. 1996;9(4):273–8.
- Apovian CM, Frey CM, Rogers JZ, McDermott EA, Jensen GL. Body mass index and physical function in obese older women. J Am Geriatr Soc. 1996;44(12):1487–8.
- 44. Himes CL. Obesity, disease, and functional limitation in later life. Demography. 2000;37(1):73-82.
- 45. Davison KK, Ford ES, Cogswell ME, Dietz WH. Percentage of body fat and body mass index are associated with mobility limitations in people aged 70 and older from NHANES III. J Am Geriatr Soc. 2002;50(11):1802–9.
- Jensen GL, Friedmann JM. Obesity is associated with functional decline in community-dwelling rural older persons. J Am Geriatr Soc. 2002;50(5):918–23.
- Launer LJ, Harris T, Rumpel C, Madans J. Body mass index, weight change, and risk of mobility disability in middle-aged and older women. The epidemiologic follow-up study of NHANES I. JAMA. 1994;271(14):1093–8.
- 48. Galanos AN, Pieper CF, Cornoni-Huntley JC, Bales CW, Fillenbaum GG. Nutrition and function: is there a relationship between body mass index and the functional capabilities of community-dwelling elderly? J Am Geriatr Soc. 1994;42(4):368–73.
- 49. Jenkins KR. Obesity's effects on the onset of functional impairment among older adults. Gerontologist. 2004; 44(2):206–16.
- 50. Rolland Y, Lauwers-Cances V, Cristini C, van Abellan KG, Janssen I, Morley JE, et al. Difficulties with physical function associated with obesity, sarcopenia, and sarcopenic-obesity in community-dwelling elderly women: the EPIDOS (EPIDemiologie de l'OSteoporose) Study. Am J Clin Nutr. 2009;89(6):1895–900.
- Zizza CA, Herring A, Stevens J, Popkin BM. Obesity affects nursing-care facility admission among whites but not blacks. Obes Res. 2002;10(8):816–23.

- 52. Yang Y, Brown CJ, Burgio KL, Kilgore ML, Ritchie CS, Roth DL, et al. Undernutrition at baseline and health services utilization and mortality over a 1-year period in older adults receiving Medicare home health services. J Am Med Dir Assoc. 2011;12(4):287–94.
- Roubenoff R. Sarcopenia: effects on body composition and function. J Gerontol A Biol Sci Med Sci. 2003;58(11): 1012–7.
- 54. Baumgartner RN. Body composition in healthy aging. Ann N Y Acad Sci. 2000;904:437-48.
- Villareal DT, Banks M, Siener C, Sinacore DR, Klein S. Physical frailty and body composition in obese elderly men and women. Obes Res. 2004;12(6):913–20.
- 56. Roubenoff R. Sarcopenic obesity: the confluence of two epidemics. Obes Res. 2004;12(6):887-8.
- Brown M, Sinacore DR, Binder EF, Kohrt WM. Physical and performance measures for the identification of mild to moderate frailty. J Gerontol A Biol Sci Med Sci. 2000;55(6):M350–5.
- 58. Holloszy JO, Kohrt WM. Handbook of physiology aging. London: Oxford University Press; 1995.
- 59. Jette AM, Cleary PD. Functional disability assessment. Phys Ther. 1987;67(12):1854–9.
- 60. Blaum CS, Xue QL, Michelon E, Semba RD, Fried LP. The association between obesity and the frailty syndrome in older women: the women's health and aging studies. J Am Geriatr Soc. 2005;53(6):927–34.
- Sarkisian CA, Liu H, Gutierrez PR, Seeley DG, Cummings SR, Mangione CM. Modifiable risk factors predict functional decline among older women: a prospectively validated clinical prediction tool. The Study of Osteoporotic Fractures Research Group. J Am Geriatr Soc. 2000;48(2):170–8.
- Hubbard RE, Lang IA, Llewellyn DJ, Rockwood K. Frailty, body mass index, and abdominal obesity in older people. J Gerontol A Biol Sci Med Sci. 2010;65(4):377–81.
- Reid IR, Cornish J, Baldock PA. Nutrition-related peptides and bone homeostasis. J Bone Miner Res. 2006;21(4): 495–500.
- 64. Schott AM, Cormier C, Hans D, Favier F, Hausherr E, Dargent-Molina P, et al. How hip and whole-body bone mineral density predict hip fracture in elderly women: the EPIDOS Prospective Study. Osteoporos Int. 1998;8(3): 247–54.
- Nielson CM, Marshall LM, Adams AL, LeBlanc ES, Cawthon PM, Ensrud K, et al. BMI and fracture risk in older men: the osteoporotic fractures in men study (MrOS). J Bone Miner Res. 2011;26(3):496–502.
- 66. Himes CL, Reynolds SL. Effect of obesity on falls, injury, and disability. J Am Geriatr Soc. 2012;60(1):124-9.
- Garrow JS, Summerbell CD. Meta-analysis: effect of exercise, with or without dieting, on the body composition of overweight subjects. Eur J Clin Nutr. 1995;49(1):1–10.
- 68. Gallagher D, Kovera AJ, Clay-Williams G, Agin D, Leone P, Albu J, et al. Weight loss in postmenopausal obesity: no adverse alterations in body composition and protein metabolism. Am J Physiol Endocrinol Metab. 2000;279(1):E124–31.
- 69. Inelmen EM, Sergi G, Coin A, Miotto F, Peruzza S, Enzi G. Can obesity be a risk factor in elderly people? Obes Rev. 2003;4(3):147–55.
- Villareal DT, Banks M, Sinacore DR, Siener C, Klein S. Effect of weight loss and exercise on frailty in obese older adults. Arch Intern Med. 2006;166(8):860–6.
- Villareal DT, Miller III BV, Banks M, Fontana L, Sinacore DR, Klein S. Effect of lifestyle intervention on metabolic coronary heart disease risk factors in obese older adults. Am J Clin Nutr. 2006;84(6):1317–23.
- Jensen GL, Roy MA, Buchanan AE, Berg MB. Weight loss intervention for obese older women: improvements in performance and function. Obes Res. 2004;12(11):1814–20.
- 73. Messier SP, Loeser RF, Miller GD, Morgan TM, Rejeski WJ, Sevick MA, et al. Exercise and dietary weight loss in overweight and obese older adults with knee osteoarthritis: the Arthritis, Diet, and Activity Promotion Trial. Arthritis Rheum. 2004;50(5):1501–10.
- Miller GD, Nicklas BJ, Davis C, Loeser RF, Lenchik L, Messier SP. Intensive weight loss program improves physical function in older obese adults with knee osteoarthritis. Obesity (Silver Spring). 2006;14(7):1219–30.
- Villareal DT, Chode S, Parimi N, Sinacore DR, Hilton T, Armamento-Villareal R, et al. Weight loss, exercise, or both and physical function in obese older adults. N Engl J Med. 2011;364(13):1218–29.
- 76. Shah K, Armamento-Villareal R, Parimi N, Chode S, Sinacore DR, Hilton TN, et al. Exercise training in obese older adults prevents increase in bone turnover and attenuates decrease in hip bone mineral density induced by weight loss despite decline in bone-active hormones. J Bone Miner Res. 2011;26(12):2851–9.
- 77. Shea MK, Nicklas BJ, Houston DK, Miller ME, Davis CC, Kitzman DW, et al. The effect of intentional weight loss on all-cause mortality in older adults: results of a randomized controlled weight-loss trial. Am J Clin Nutr. 2011;94(3):839–46.
- Avenell A, Richmond PR, Lean ME, Reid DM. Bone loss associated with a high fibre weight reduction diet in postmenopausal women. Eur J Clin Nutr. 1994;48(8):561–6.
- 79. Riedt CS, Cifuentes M, Stahl T, Chowdhury HA, Schlussel Y, Shapses SA. Overweight postmenopausal women lose bone with moderate weight reduction and 1 g/day calcium intake. J Bone Miner Res. 2005;20(3):455–63.

- 80. Jensen LB, Kollerup G, Quaade F, Sorensen OH. Bone minerals changes in obese women during a moderate weight loss with and without calcium supplementation. J Bone Miner Res. 2001;16(1):141–7.
- Villareal DT, Fontana L, Weiss EP, Racette SB, Steger-May K, Schechtman KB, et al. Bone mineral density response to caloric restriction-induced weight loss or exercise-induced weight loss: a randomized controlled trial. Arch Intern Med. 2006;166(22):2502–10.
- Ryan AS, Nicklas BJ, Dennis KE. Aerobic exercise maintains regional bone mineral density during weight loss in postmenopausal women. J Appl Physiol. 1998;84(4):1305–10.
- Armamento-Villareal R, Sadler C, Napoli N, Shah K, Chode S, Sinacore DR, et al. Weight loss in obese older adults increases serum sclerostin and impairs hip geometry but both are prevented by exercise training. J Bone Miner Res. 2012;27(5):1215–21.
- 84. Expert panel on detection, evaluation, and treatment of high blood cholesterol in adults. Executive summary of the third report of The National Cholesterol Education Program (NCEP) (Adult Treatment Panel III). JAMA. 2001;285(19):2486–97.
- Sugerman HJ, DeMaria EJ, Kellum JM, Sugerman EL, Meador JG, Wolfe LG. Effects of bariatric surgery in older patients. Ann Surg. 2004;240(2):243–7.

Chapter 11 Nutrition and Lifestyle Change in Older Adults with Diabetes Mellitus and Metabolic Syndrome

Barbara Stetson, Holly M. Knight, and Sri Prakash L. Mokshagundam

Key Points

- In the United States, 22–33 % of adults >65 years of age have diabetes and constitute 44 % of persons with self-reported diagnosed diabetes [1].
- The aim of diabetes intervention is to prevent or delay the development of long-term complications of high blood glucose and related metabolic abnormalities and improve the quality of life.
- "Metabolic syndrome" refers to a cluster of abnormalities that includes hypertension, dyslipidemia, abnormal blood glucose, and abdominal obesity. Over 40 % of adults over the age of 70 have metabolic syndrome.
- Hypoglycemia is a major limiting factor in the management of diabetes. Factors that may play a role in the increased risk of hypoglycemia in older adults include poor nutritional status, cognitive dysfunction, poly-pharmacy, and comorbid illnesses.
- Diabetes-related comorbidities such as diabetic retinopathy, cardiovascular disease, peripheral vascular disease, and congestive heart failure may result in decreased usual activity and limit activities of daily living, including transportation, shopping for food, and ability to read food labels and restaurant menus.
- Given the high rates of depression in the diabetes population, careful assessment of depressive symptomology and its impact on dietary intake, diabetes self-care, and health outcomes and provision of support is critical.

Keywords Blood glucose • Metabolic syndrome • Hypoglycemia • Depression • Quality of life

B. Stetson, Ph.D. (🖂) • H.M. Knight, B.A.

Department of Psychological and Brain Sciences, University of Louisville, 317 Life Sciences Building, Louisville, KY 40292, USA e-mail: barbara.stetson@louisville.edu; hmknig01@louisville.edu

S.P.L. Mokshagundam, M.D.

Division of Endocrinology and Metabolism, University of Louisville, 530 S Jackson Street, Louisville, KY 40292, USA e-mail: slmoks01@louisville.edu

Introduction

Diabetes mellitus (DM) (diabetes) is a major health problem in the United States. The estimated number of individuals with diabetes is approximately 25.8 million, of whom 7 million are undiagnosed [2]. In older adults type 2 diabetes disproportionately affects minority populations, with higher prevalence in African Americans and Hispanics, who also have poor blood glucose control and more comorbidities and complications relative to non-Hispanic whites [3]. The prevalence of obesity is rising so rapidly in so many countries that the World Health Organization has declared that there is now a global epidemic of obesity. Internationally, emergence of new cases of diabetes parallels the increases seen in Western countries and incidence is are rapidly increasing in low and middle income countries and global prevalence is at an all-time high [4].

Diabetes in Older U.S. Adults

The graying of America is contributing to the increasing number of U.S. cases of diabetes, as prevalence increases with age. In developing countries, the majority of people with diabetes are between 45 and 64 years of age. In developed countries, the majority of people with diabetes are age 65 or older. The aging population has and will continue to contribute to the now evident diabetes epidemic and the resultant burdens of increased functional impairment, intensive care, and mortality [3].

Health Consequences of Diabetes

Diabetes is a chronic disease that leads to a variety of micro (e.g., nephropathy, retinopathy, and neuropathy) and macrovascular (e.g., coronary artery disease, peripheral vascular disease, and stroke) complications that affect almost all bodily systems. While the primary abnormality, elevated blood glucose level, remains largely asymptomatic, the consequences of sustained elevation are potentially devastating. Diabetes is the leading cause of blindness, chronic renal insufficiency, peripheral neuropathy, and nontraumatic limb amputations in the United States.

Type 2 diabetes exerts a tremendous economic burden and substantial impact on comorbidities such as visual impairment, end stage renal disease, cardiovascular disease, and events such as myocardial infarction and amputations, particularly in older adults [3]. Older adults have a particularly high prevalence of diabetes-related conditions, approximately three-quarters living with diabetes or prediabetes. It is estimated that 30 % of older adults have diabetes, with almost half of these individuals as yet undiagnosed [5]. A recent review of diabetes and diabetes complications in older adults indicates that older adults with diabetes have higher rates of falling, incontinence, and psychiatric comorbidities such as depression and dementia relative to younger adults with diabetes [5].

Cardiovascular disease (CVD) is the most frequent and costly complication of type 2 diabetes. When cardiovascular events are stratified by diabetes status, relative risk for men is twice that of gender-matched non-diabetics and for women is threefold. Among all CVD events, diabetes accounts for 56 % of events in men and 78 % of events in women. A number of diabetes-related risk factors, as well as post-glucose challenge hyperglycemia in nondiabeteic individuals, have been associated with CVD [6].

General Aims of Diabetes Treatment

The management of diabetes requires a combination of lifestyle interventions and medications, with dietary approaches to maintaining optimal glycemic control as a key component. The aims of diabetes treatment are to: (1) decrease/prevent the development of long-term complications of high blood glucose and related metabolic abnormalities, (2) improve quality of life, and (3) treat or prevent the development of symptoms of high or low blood glucose.

Diagnosis and Classification

Diagnosis

The American Diabetes Association (ADA) recommends the following criteria for the diagnosis of DM [7]:

- Hemoblobin A1c \geq 6.5 %.
- The Hemoblobin A1c test should be performed in a laboratory using a method that is National Glycohemoglobin Standardization Program (NGSP)-certified and standardized to the Diabetes Control and Complications Trial (DCCT) assay; or
- Fasting plasma glucose (FPG) ≥126 mg/dL (7.0 mmol/L). Fasting is defined as no caloric intake for at least 8 h; or
- 2-h plasma glucose \geq 200 mg/dL (11.1 mmol/L) during an oral glucose tolerance test (OGTT); or
- In a patient with classic symptoms of hyperglycemia or hyperglycemic crisis, a random plasma glucose ≥200 mg/dL (11.1 mmol/L).

In the absence of unequivocal hyperglycemia, the result should be confirmed by repeat testing. The ADA diagnostic criteria were developed for general use, and apply broadly to all age groups. No specific ADA guidelines exist for older adults.

Typologies of Diabetes in Older Adults

The proper classification of diabetes is important in setting goals for nutritional management of individuals with diabetes, with broad classification of type 1 and type 2 diabetes.

Type 1 Diabetes

Type 1 diabetes is an autoimmune disorder resulting from cell mediated and antibody mediated destruction of beta-cells of the islets [8]. Although type 1 diabetes most commonly occurs in the first three decades of life, it can develop at any age, even in older adults. An additional contributor to the population of older individuals is that the number of individuals with type 1 diabetes living into old age has increased [9]. Research investigating potential nutritional influences on the development and progression of diabetes has led to a widespread interest in the role of vitamin D. Several studies indicate that in type 1 diabetes the role of vitamin D is in prenatal or early life vitamin D deficiency. Vitamin D deficiency has been gaining increasing attention as a potential factor in the development of both type 1 and type 2 diabetes. Several epidemiological studies have indicated that vitamin D deficiency could be a factor in risk of development of type 1 and type 2 DM [10]. Vitamin D has

been shown in animal models to be associated with both insulin resistance and insulin secretion, abnormalities of which play a key role in the development of type 2 DM [11]. The NIH has recently funded a multicenter trial to assess the role of vitamin D supplementation in prevention of type 2 DM. Vitamin D has also been shown to play an important role in the regulation of both innate and reactive immune system [12]. In animal models of type 1 DM, 1–25 dihydroxyvitamin D has been demonstrated to delay development of type 1 DM [13]. Definitive human trials are awaited. The complex interaction between vitamin D and DM has been recently reviewed by Van Belle et al. [14].

Type 2 Diabetes

The majority of older adults with diabetes have type 2 diabetes, which is characterized by two defects: insulin resistance and defective insulin secretion [15]. The majority of individuals with type 2 diabetes are obese. However, in the older population the proportion of subjects with type 2 diabetes who are underweight increases and could be as high as 20 %. This is particularly true in the nursing home population due to sarcopenia and frailty [16].

The exact mechanism of insulin resistance in type 2 diabetes is unclear. A variety of genetic and environmental factors lead to decreased insulin sensitivity. Obesity and decreased physical activity are known to decrease insulin sensitivity [17]. Aging is associated with body composition changes of increased fat and decreased muscle mass [18, 19]. This could be partly responsible for the increase in insulin resistance with aging. Aging is also associated with a decline in insulin secretion, particularly a blunting of the first-phase insulin secretion [20]. Age-related increases in sedentary lifestyle may also further compound these changes. Vitamin D deficiency has been linked to increased insulin resistance and higher prevalence of type 2 diabetes.

Establishing Medication and Nutritional Management Goals

Once the type of diabetes is established, medication and nutritional management goals should be developed. In addition to tailoring the nutritional recommendation to assist glycemic control, consideration of other important CVD risk factors is critical. Obesity, dyslipidemia, hypertension, and insulin resistance are important and often overlapping factors warranting consideration when planning dietary interventions for older adults with type 2 diabetes. Avoidance of hypoglycemia, particularly recurrent and/or severe hypoglycemia, is a major consideration in type 1 diabetes. Lifestyle interventions that have been recommended for the management of diabetes have positive effects on both insulin secretion and insulin resistance. Aggressive lifestyle intervention can prevent the progression of impaired glucose tolerance to diabetes and could decrease the dose and number of medications for the management of type 2 diabetes.

Medication Use and Glycemic Control in Older Adults with Diabetes

Goals of Diabetes Treatment in Older Adults

The major aim of treating diabetes is to decrease the rate of micro and macrovascular disease associated with elevated blood glucose. Two landmark trials have served as the basis for the recommendations for the management of blood glucose levels in DM until 2012. Neither of these trials enrolled individuals over age 65 years. The DCCT was conducted in adults with type 1 diabetes and compared intensive insulin treatment using multiple insulin injections or an insulin pump to conventional treatment using twice daily injections of intermediate- and short-acting insulin [21, 22]. The results showed significant reduction in risk of all microvascular disease endpoints in the intensively treated group. The United Kingdom Prospective Diabetes Study (UKPDS) was a long-term study of a variety of treatment options in adults with type 2 diabetes [22]. Major UKPDS findings were that: (1) a reduction of 1 % in hemoglobin A1c resulted in ~22 % reduction in microvascular complications, (2) glycemic control in type 2 DM worsened over time and necessitated changes in medication, and (3) in a subgroup of subjects treated with metformin there was a significant reduction in macrovascular disease. Ten-year follow-up noted persistent microvascular benefits [23]. While the DCCT and UKPDS informed earlier guidelines, three new studies that included older participants [24–26] have raised questions about the role of intensive glycemic control in reducing macrovascular events in older adult subjects. This has been reviewed extensively in a combined ADA—AHA review [27].

Current ADA goals are to aim for HbA1c levels \leq 7.0 %. In addition the ADA recommends individualization of glycemic control goals with an HbA1c of <6 % when feasible. The new guidelines by the ADA recognize the uncertainty arising from the newer trials and the need to consider multiple factors in determining glycemic goals for each individual such as duration of diabetes, risk of hypo-glycemia, presence of comorbidities, and age [28]. This is particularly true for older adults. The 2003 California Healthcare Foundation/American Geriatric Society (AGS) guidelines [29] have helped clarify some of these concerns, and several concepts, including cognitive function, social support, and presence of comorbidities, have been subsequently incorporated in the ADA guidelines. The AGS guidelines also recognize the importance of comorbidities that are common in the older adult with diabetes that influence ability to maintain strict glycemic control [29]. Approaches to the management of hyperglycemia may be seen in Fig. 11.1.

In addition to glycemic goals, several metabolic and cardiovascular risk factors must also be considered due to the higher rates of cardiovascular disease and its substantial impact on morbidity and mortality. The high risk of cardiovascular morbidity and mortality in diabetes is likely due to a variety of factors, including overall blood glucose control, glycemic fluctuation, postprandial blood glucose levels, high LDL cholesterol, low HDL cholesterol, elevated serum triglycerides, blood pressure, and altered coagulation profile. In addition, inflammation, pro-oxidant state, and endothelial dysfunction play a significant role. Despite multiple trials that have shown benefits of statins in reducing CVD events, few have specifically addressed benefits of statins in individuals with diabetes aged 65 and over [30]. Based on these trials, it is reasonable to use statins in older adults <80 years age and exercise caution with older individuals. The benefits of fibrates in the older adult are not clear. Similarly, the benefits of blood pressure reduction have not been specifically addressed in older adults with diabetes; however, some subpopulation analyses indicate benefits of blood pressure reduction consistent with ADA goals [30]. Optimal implementation of the current guidelines may require changes in the way that risk-factor management is viewed by caregivers and patients [31]. Many providers emphasize the importance of patient awareness of their diabetes-related labs that reflect ADA goals, particularly their A1c, blood pressure, and lipids. A nationally representative sample of adults with type 2 diabetes from the NHANES survey found that knowledge was limited (48 % were able to report their last A1c level, 63 % their last blood pressure level, and 22 % their last LDL level) and was greatest among non-Hispanic whites and those with higher education and income [32]. In general, knowledge about personal diabetes-related labs was not significantly associated with meeting related clinical recommendations. This lack of a significant association between knowledge and meeting clinical recommendations suggests that an emphasis on awareness of these test values in and of themselves may not be a necessary priority, particularly in older adults with other competing demands [32].

Any nutritional approach to the management of DM must specifically address the issues related to cardiovascular risk. Cardiovascular risk reduction in DM is achieved through a combination of lifestyle changes and pharmacological interventions that address the multiple risk factors. Taken together, these lifestyle and pharmacological interventions can play a major role in cardiovascular risk reduction in persons with diabetes.

Approach to management of hyperglycemia: less More stringent stringent Highly motivated, adherent, Patient attitude and Less motivated, non-adherent, excellent self-care capacities poor self-care capacities expected treatment efforts Risks potentially associated High I ow with hypoglycemia, other adverse events Long-standing Disease duration Newly diagnosed Life expectancy Long Short Important comorbidities Severe Absent Few / mild Absent Severe Established vascular Few / mild complications Resources, support system Readily available Limited

Fig. 11.1 Approach to management of hyperglycemia. Reproduced with permission from the American Diabetes Association, from Inzucchi SE, Bergenstal RM, Buse JB, Diamant M, Ferrannini E, Nauck M, et al. Management of hyperglycemia in type 2 diabetes: a patient-centered approach: position statement of the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetes Care. 2012 Jun;35(6):1364–79; permission conveyed through Copyright Clearance Center, Inc.

General Diabetes Dietary Goals

The general goals of nutritional recommendations for the management of diabetes include:

- 1. Achieve and maintain blood glucose levels as outlined above.
- 2. Achieve and maintain optimum lipid levels.
- 3. Achieve and maintain reasonable body weight. This would include weight loss, if overweight, and weight gain, if undernourished.
- 4. Prevent acute complications.
- 5. Maintain overall health.

Macronutrient Recommendations

The carbohydrate composition (amount and type) of the diet has been a focus of many recommendations and subject of recent controversy. There are no well-designed studies that have compared different dietary approaches. In a study of a high carbohydrate (60 %), low fat (25 %) diet, compared to a low carbohydrate (35 %), high monounsaturated fat (50 %) diet [33], plasma glucose, triglyceride, and VLDL cholesterol were lower in the subjects in the low carbohydrate/high monounsaturated fat diet group [33]. However, use of high-fat/low-carbohydrate diets could lead to more hypoglycemic episodes and ketosis, particularly in type 1 diabetes [34].

Higher protein diets have been recommended and have been popular for weight loss. However, the efficacies of these diets in persons with diabetes have not been well tested. Higher protein content has been shown to increase risk of development and progression of diabetic nephropathy (as discussed in Chap. 16 of this book), contraindicating this type of diet for persons with poorly controlled diabetes or complications.

Low-carbohydrate diets are not recommended in the management of diabetes. Dietary carbohydrate is the major contributor to postprandial glucose concentration, and foods containing carbohydrates are an important source of energy, water-soluble vitamins and minerals, and fiber. Thus, in agreement with the National Academy of Sciences-Food and Nutrition Board, a recommended range of carbohydrate intake is 45–65 % of total calories [35]. In addition, because the brain and central nervous system have an absolute requirement for glucose as an energy source, restricting total carbohydrate to <130 g/day is not recommended [36]. Monitoring carbohydrate intake, whether by carbohydrate counting, using the exchange system, or experience based estimation remains a key component of managing diabetes. Use of complex carbohydrates is preferred. The use of lower fat content in the diet is based on the need to restrict caloric intake, improve lipid levels, and assist weight loss. While use of vitamin supplements is not generally recommended for patients with type 2 diabetes, the ADA recommends using a multivitamin for older adults. There has been no convincing evidence to recommend the routine use of antioxidants, vitamin E, or C in individuals with diabetes.

Balancing Diet and Medication

The interaction of diet and medication is of particular importance in diabetes management. Insulin and drugs that increase insulin secretion are likely to induce hypoglycemia if meals are not taken at appropriate times. Erratic eating patterns might require readjustment of medications, dose, timing, or both. This may be of particular concern in the hospitalized older patient with diabetes. Poor eating habits might also necessitate change to medications that are less likely to cause hypoglycemia when used alone. Metformin (Glucophage), thiazolidenediones (Pioglitazone—Actos), and incretin-based therapies (Exenatide-Byetta; Liraglutide—Victoza; Sitagliptin—Januvia; Saxagliptin—Onglyza; and Linagliptin—Tradjenta) are least likely to cause hypoglycemia when used alone. The problem of unwanted weight gain is another issue for consideration for many individuals. Insulin and the thiazolidenediones, particularly when used in combination, are most likely to result in weight gain. Loss of appetite may occur with Metformin, Exenatide, Liraglutide (glucagon-like peptide 1 receptor agonists), and pramlinitide (Symlin, an amylin analog) and would be of concern in undernourished individuals.

Body Weight and Functional Status in Older Adults with Diabetes

Overweight and Obesity in Older Adults

Overweight and obesity are both important risk factors for the development of type 2 diabetes and are prevalent in older adults (as is discussed in Chap. 10 of this book). Obesity appears to be common in older adults until the eighth decade of life and then declines in the oldest old. Muscle mass and body strength decreases observed in aging may result in sarcopenia in normal, over-, and underweight older

adults. This may be further enhanced by inactivity, which may result from diabetes complications and comorbidities. Combining physical activity interventions with nutritional approaches may help to combat these difficulties while improving fitness and functional status [3].

Underweight and Malnutrition

While obesity is clearly a problem that greatly impacts diabetes in older adults, for many individuals, malnutrition may be the more pressing nutritional concern. Even with the problem of increasingly prevalent overweight, obesity, and diabetes, the prevalence of obesity in older persons with diabetes is still less than that in younger person with type 2 diabetes. This may be particularly true for the oldest old, and those who have impaired functional status [37]. Factors influencing undernourishment in older adults include changed sense of taste and/or smell, oral issues such as poor dentition or swallowing difficulties and functional difficulties resulting in problems in preparing food or feeding oneself [3]. Brief malnutrition screening with validated instruments such as the Mini-Nutritional Assessment [38] may help to streamline evaluation for determining the need for further specific nutrition therapy [3].

Special Nutrition Intervention Situations for Persons with Diabetes

Acute Illness, Hospitalization, Enteral and Parenteral Nutrition

Management of hyperglycemia in the hospital setting has gained increasing attention over the last few years. Hyperglycemia is common among hospitalized subjects and has been shown to be associated with higher mortality and morbidity in a variety of studies [39–44]. This is particularly relevant to the older population, since they are more likely to be admitted to the hospital and have higher rates of diabetes. The American Association of Clinical Endocrinologists (AACE) and ADA have recommended glycemic targets for patients with hyperglycemia [45]. Initial studies in the ICU setting suggested that tight blood glucose control to a level of <110 mg/dL would result in significant improvement in mortality and morbidity [46]. However, this was not confirmed by subsequent studies, which also pointed to a higher mortality with efforts to achieve goal blood sugar of <110 mg/dL, perhaps related to higher rate of hypoglycemia [47, 48]. Current recommendations are to achieve a glucose level of 140–180 mg/dL [49, 50].

Hospitalization

A variety of systemic problems have been identified that affect glycemic control in the hospital setting. Barriers that may impact an individual's nutrition status and subsequently affect glycemic control include poor appetite, inability to eat, increased nutrient and calorie needs due to catabolic stress, variation in diabetes medications, and the possible need for enteral or parenteral nutrition support. Proper timing of meals and the relation to medications is important. Insulin should be administered immediately before or after a meal. Due to the wide heterogeneity in the hospital population, individualization of nutrition recommendations is key to improving outcomes. The common practice of ordering an "ADA Diet" is strongly discouraged, as the ADA does not endorse any specific diet. The consistent carbohydrate meal planning system is encouraged. For this system to be effective, it is important that nursing and nutrition services coordinate their services. The key areas of focus to improve inpatient glycemic control are:

- 1. Establishing screening criteria for appropriate referral to a registered dietitian.
- 2. Identifying nutrition-related issues in clinical pathways and patient care plans.
- 3. Implementing and maintaining standardized diet orders such as consistent carbohydrate menus.
- 4. Integrating blood glucose monitoring results with nutrition care plans.
- 5. Using standing orders for diabetes education and diabetes medical nutrition therapy (MNT) as appropriate.
- Standardizing discharge follow-up orders for MNT and diabetes education post-discharge when necessary [51, 52].

Patients requiring clear or full liquid diets should receive 200 g carbohydrate/day in equally divided amounts at meal and snack times [53]. Liquids should not be sugar-free. Patients require carbohydrate and calories, and sugar-free liquids do not meet these nutritional needs. For tube feedings, either a standard enteral formula (50 % carbohydrate) or a lower carbohydrate content formula (33–40 % carbohydrate) may be used. Calorie needs for most patients are in the range of 25–35 kcal/kg every 24 h. Care must be taken not to overfeed patients because this can exacerbate hyperglycemia. After surgery, food intake should be initiated as quickly as possible. Progression from clear liquids to full liquids to solid foods should be completed as rapidly as tolerated [54].

Enteral and Parenteral Nutrition

Enteral and parenteral nutrition might pose additional challenges in the management of patients with diabetes. While the glycemic goals for individuals receiving enteral and parenteral nutrition are the same as glycemic goals for the general population with diabetes, achievement of normoglycemia may be more difficult in patients who are acutely ill. There is evidence that poor glycemic control in individuals on parenteral or enteral nutrition is related to poor outcomes. It is estimated that up to 30 % of patients who receive parenteral nutrition have diabetes, many of whom have no previous history of diagnosed diabetes and develop hyperglycemia due to stress-induced increases in counter-regulatory hormones and cytokines.

The relative value of high-carbohydrate versus high-fat enteral feeds for persons with diabetes has been debated [55]. The most widely used commercial enteral preparations for individuals with diabetes provide, 1 cal/mL, 40 % (CHOICEdmTF; Novartis Medical Nutrition) to 34 % (Glucerna; Abott Laboratories, Inc.) carbohydrate, and 43 % (CHOICEdmTF) to 49 % (Glucerna) fat. They also have high monounsaturated fatty acids (MUFA; 35 % of kcal in Glucerna). MUFA has been shown to be beneficial in improving lipid profile, glycemic control, and lowering insulin levels in both acutely ill and ambulatory individuals [56]. CHOICEdmTF has a higher content of medium-chain triglycerides and no fructose. The use of insulin or oral agents in persons receiving enteral nutrition should be tailored to match the timing of feeds. Parenteral nutrition fluids are high in carbohydrate and derive fewer calories from fat. In persons with diabetes, particularly in less severely stressed individuals, the proportion of carbohydrate may be decreased but is still very high. The usual rate of glucose infusion is 4-5 g/kg body weight and lipid infusion of 1-1.5 g/kg body weight. This requires adequate use of insulin to maintain normoglycemia [57]. Insulin infusion not only maintains glycemic control, but prevents protein breakdown and promotes protein synthesis.

Long-Term Care

Residents of long-term care facilities may face additional or unique problems. They tend to be underweight and do not necessitate any caloric restrictions. Low body weight has been associated with higher mortality and morbidity in these populations. Restricting food choices may lead to poor overall nutritional status and has not been shown to improve glycemic control [58]. Hence, the use of "no concentrated sugar," "no sugar added," or "liberal diabetic diet" is discouraged [59, 60].

Metabolic Syndrome

The term "metabolic syndrome" refers to a cluster of abnormalities that was initially described as including hypertension, dyslipidemia, abnormal blood glucose, and abdominal obesity. Insulin resistance was considered to be a central pathogenetic abnormality in this syndrome. In 2001, the National Cholesterol Education Program-Adult Treatment Panel III (NCEP-ATP III) [61] recommended the use of the metabolic syndrome in cardiovascular risk assessment. The NCEP-ATP III defined metabolic syndrome [62] is the most widely used definition. Prevalence of metabolic syndrome as defined by the NCEP-ATP III in the NHANES III cohort increases with age, with over 40 % of adults over the age of 70 having the metabolic syndrome [63]. NCEP-ATP III Criteria for Diagnosis of Metabolic Syndrome reflect presence of three or more of the following risk factors: abdominal obesity (waist circumference): men: >102 cm (>40 in.), women: >88 cm (>35 in.); triglyceride: >150 mg/dL; HDL cholesterol: men: <40 mg/dL, women: <50 mg/dL; blood pressure: <130/85 mmHg; fasting blood glucose: >110 mg/dL.

The exact significance of the metabolic syndrome has been controversial. A variety of clinical studies have demonstrated an association between metabolic syndrome and risk of heart disease, stroke, and development of diabetes [64, 65]. However, several other studies have shown a lack association between the "syndrome" and CVD risk that cannot be attributed to its individual components [66, 67]. Identification of individual risk factors related to the metabolic syndrome construct should prompt health care providers to look for other risk factors and resultantly, encourage the use of behavioral approaches such as diet and increased physical activity rather than always prescribing a different drug for each medical condition. The management of the cluster of risk factors represented in metabolic syndrome is primarily aimed at reducing cardiovascular risk by addressing the individual risk factors. In individuals with pre-diabetes, the aim is to prevent progression to type 2 diabetes.

The metabolic syndrome predicts a higher rate of progression to type 2 diabetes. The landmark Diabetes Prevention Program (DPP) evaluated the impact of intensive lifestyle intervention and pharmacological treatment on the delay or prevention of progression of individuals with high risk of diabetes (most of whom had metabolic syndrome) to type 2 diabetes. The DPP demonstrated the efficacy of both lifestyle and pharmacological interventions in preventing the progression of impaired fasting glucose or impaired glucose tolerance. In older DPP participants, lifestyle intervention had an even greater impact than metformin. Among participants age 60 and older, lifestyle intervention reduced the risk of development of diabetes by 71 %. Participants in the 60- to 85-year-old age group were the most likely to achieve weight loss (5–7 % loss in body weight was achieved with dietary change and activity) and physical activity goals [68].

Hypoglycemia in Older Adults

Hypoglycemia is a major limiting factor in the management of diabetes. The incidence of hypoglycemia is relatively high in older compared to younger adults, with a variety of factors playing a role in this increased risk. These include poor nutritional status, cognitive dysfunction, polypharmacy, and comorbid illnesses. Except in the severely malnourished, poor dietary intake by itself does not lead to hypoglycemia. The most common cause of hypoglycemia remains the use of blood glucose-lowering agents. Drugs that increase insulin secretion and insulin itself can cause hypoglycemia. A major finding of the DCCT, which was conducted with young, healthy adults, was that the major deleterious health consequence of tight blood glucose control in persons with type 1 diabetes is hypoglycemia [69]. Drugs that enhance insulin sensitivity (thiazolidenediones), decrease hepatic glucose production (metformin), or decrease carbohydrate absorption (alpha-glucosidase inhibitors) have very low risk of hypoglycemia, except when used in combination with insulin or an insulin secretagogue. When medication use creates problems of consistent hypoglycemia, patients must learn how to avoid and manage hypoglycemic episodes. Older adults taking insulin who have high variability in blood glucose levels, exhibit very low average blood glucose concentrations, have had diabetes for a long duration, have a low body mass index, or who have high levels of vigorous physical activity may be at particular risk of severe hypoglycemia [70].

Self-Monitoring and Dietary Treatment of Hypoglycemia

Frequent self-monitoring of blood glucose levels (SMBG) provides specific information to guide decisions about moment-to-moment treatment needs, thus helping individuals to anticipate or prevent severely low glucose levels. By increasing the frequency of blood glucose testing and making informed decisions about when to eat additional carbohydrate (e.g., eat 15 g of carbohydrate to raise blood glucose levels about 45 mg/dL) or to identify personal sources of vigorous physical activity contributing to low blood glucose levels, patients may learn to prevent severe hypoglycemia. Educating patients about the importance of always carrying glucose tabs, gel, or fast-acting carbohydrate snacks or placing them in various locations such as the car, or relative's homes may also aid in the treatment of mild to moderate hypoglycemic episodes. Recommendations for dietary management of hypoglycemia are below:

- 1. Check blood glucose level by glucose monitor.
- 2. If blood glucose less than 60 mg/dL or symptomatic—treat with 15 g of carbohydrate (e.g., four ounces of or regular soft drink [not diet], glucose tablets or gel tube, two tablespoons of raisins, one tablespoon of sugar, honey, or corn syrup, eight ounces of nonfat, or 1 % milk).
- 3. Repeat blood sugar reading in 15 min after treatment and again after 60 min.
- 4. Repeat step 2 until blood glucose is >60 mg/dL.
- 5. If meals are due within 60 min, eat meal now.
- 6. If meals are not due within 60 min, follow the glucose treatment with a snack containing carbohydrate and one protein (cheese and crackers, peanut butter and crackers, skim milk and crackers, or a small sandwich).
- If blood glucose <40 mg/dL and/or subject is stuporous, confused, or unresponsive, give 1 amp of D50W as IV push and start D10W at 60 cc/h. Check blood glucose every 5 min and repeat till blood glucose >60 mg/dL or till awake. Give oral carbohydrate once awake.

Hypoglycemia Unawareness and Treatment of Hypoglycemia

As previously described, older adults with type 1 diabetes and those with type 2 diabetes who are on exogenous insulin regimens are at risk for hypoglycemia. Many individuals develop the syndrome of hypoglycemia unawareness, in which the warning symptoms that indicate that hypoglycemia is developing (e.g., tremulousness, tachycardia) are decreased or not detected. Without these warning symptoms, individuals are not able to take actions such as eating to prevent continued reductions in blood glucose levels and severe hypoglycemic episodes may result. Following episodes of hypoglycemia,

counterregulatory hormone stores may not be available, and thresholds for symptoms of hypoglycemia may shift to lower glucose concentrations. Thus, patients with recurrent hypoglycemia may be particularly at risk for unawareness and for severely low hypoglycemic episodes. Failure to test blood glucose levels regularly can contribute to the problem of hypoglycemia unawareness. This cycle is particularly problematic for older adults who are highly physically active or who skip meals, do not eat sufficient quantities of food to match their insulin doses, or consume a high-fat diet, which delays carbohydrate absorption and is not accounted for in the timing of insulin administration.

Alcohol consumption, while not typically problematic when consumed in moderation, can pose risks for hypoglycemia in older adults taking insulin. The disinhibiting effect of alcohol intoxication poses the risk of hypoglycemia unawareness due to inattention to internal cues, making blood glucose monitoring essential. The potential for a delayed risk of hypoglycemia the morning after evening alcohol intake should also be emphasized [71]. The problem of patient hypoglycemia unawareness should be considered if the patient's HbA1c is low (e.g., <6.0), and if s/he describes inability to detect counterregulatory autonomic symptoms (e.g., tremulousness, pounding heart, anxiety, queasy stomach, sweating, flushed face) when blood glucose levels are low [72].

Structured psychoeducational intervention and print materials to promote reduce hypoglycemia unawareness have been developed and systematically evaluated and have been shown to improve accuracy of recognition of current blood glucose levels, improved judgment regarding when to treat, reduced occurrence of severe hypoglycemia and improved judgment about not driving while hypoglycemic [73].

Diabetes Lifestyle Change and Physical Limitations

Diabetes-related comorbidities such as diabetic retinopathy, cardiovascular disease, peripheral vascular disease, and congestive heart failure may result in decreased usual activity and limit activities of daily living (ADLs) and instrumental activities in daily living (IADLs), including limited transportation options, limited ability to shop for food or ability to read restaurant menus. SMBG may also be influenced by reduced visual acuity. Hearing impairment may influence understanding of provider recommendations. Analyses from NHANES participants found twice the prevalence of hearing impairment in people with diabetes relative to those without, after controlling for age [74]. Comorbid health conditions may heighten the risk of these sensory difficulties, as the presence of heart disease, peripheral neuropathy, and having poor health were associated with hearing impairment in participants with diabetes [74]. Sensory limitations should be considered when providing education and instructions for older adults with diabetes. Diminished fine motor skills may also impact ability to functionally conduct a finger stick, conduct the steps necessary for using the glucometer and reading the results. Patients may need assistance adapting the choice of meter or approach to SMBG [75]. Comorbid health conditions increase polypharmacy, so pill boxes and mediplanners may also be helpful in simplifying medication management. Comorbid medical conditions and complications such as peripheral neuropathy may also contribute to pain, gait, and balance difficulties, increasing risks for falls. Hypoglycemia may also pose fall risks [3]. Management of severe hyperglycemia and hypoglycemia, avoiding polypharmacy and incorporating regular strength/stability exercises or participation in physical therapy into lifestyle approaches may also help to reduce the risks of falling in older adults with diabetes [3].

Self-Management Behaviors

Few studies of diabetes self-management education and training (DSME/T) have focused on older adults. Intervention guidelines have been predominantly based on expert consensus. As such, it is unclear if strategies for DSME that are effective with younger patients are optimal for older adults [76].

Older age may be associated with reduced readiness to make lifestyle changes such as increasing physical activity and diet modifications [77]. Many of the oldest of older adults may have limited knowledge or understanding of diabetes care. As previously described, sensory limitations may impact instruction and patient education delivery. Older individual's abilities and skills to effectively plan meals and make adjustments to food and insulin intake to accommodate blood glucose levels may be influenced by their cognitive functioning, health and numeracy literacy, functional status, and personal resources. Spreading information delivery over multiple contacts and use of memory cues such as large-print handouts and cues to use at home may facilitate learning and retention. Simplified self-care regimens that consider day-to-day quality of life are optimal [76].

Dietary Habits of Older Adults with Diabetes

Both the type of food consumed and the pattern of eating behaviors are important influences on nutritional status in older adults [78]. Adherence to a healthy diet has been found to be greater in older adults that viewed diabetes as a stable, long-term health condition relative to those who viewed diabetes symptoms as an intermittent problem, particularly for men [79].

Unhealthful snacking may threaten optimal diabetes diet adherence in older adults with diabetes, influencing the balance and timing of meals, medication use, and physical activity. A random telephone survey of older U.S. adults found that the 98 % reported snacking at least once each day, with evening the most common time and nearly all snacking occurring at home [80]. Taste outranked nutrition as selection criteria. Concrete suggestions for replacing highly processed, high-fat snack foods with fruits and vegetables and other nutritious snacks may assist older adults in selection of healthier snacks. Planning snack supplements as part of a meal plan may also be helpful, as demonstrated by a study of mid-life and older adults with diabetes found that low-glycemic, moderately high-protein snacks throughout the day promoted body weight and fat-mass loss, while not changing glycemic control, lipids, or inflammatory markers [81]. Patients may be encouraged to incorporate specific items into their shopping lists and keep them in the home as replacements for preferred but unhealthy ones. Evening activities that are alternatives to snacking (e.g., walking, crafts) may also be encouraged.

An additional influence on eating behavior in older adults with diabetes may be the social structure of mealtimes. Older adults who eat alone have been found to be more likely to be depressed and have less diversity in foods eaten relative to those who share mealtimes with others [82]. Homebound older adults may have particular difficulties with undernutrition. A study of homebound older adults found that 70 % were under-eating, with the highest risks in men, those receiving either infrequent care or receiving intensive caregiver aid, and those with hospitalization preceding home care. Under-eating was not limited to underweight individuals, but was also problematic for those with a higher BMI [83].

Psychosocial and Behavioral Issues Related to Self-Care and Dietary Intake in Older Adults with Diabetes

Depression

Estimates of the prevalence of depression in the general adult population with diabetes indicate a rate of 2–3 times that of the general population [84]. Adults with diabetes who are depressed have average HbA1c levels 0.5–1.0 % higher relative to those who are nondepressed. Depression in diabetes is also associated with additional functional impairment, lower levels of diabetes self-care, increased risk of macrovascular and microvascular complications, higher medical costs, and greater mortality [85].

In older adults with type 2 diabetes, depression was associated with increased risk of dementia [86]. In the Hispanic Established Population for the Epidemiologic Study of the Elderly (EPESE), death rates were substantially higher when elevated depressive symptomology was comorbid with diabetes, cardiovascular disease, hypertension, stroke, and cancer. Odds of death among individuals with diabetes with high levels of depressive symptoms are three times that of those without elevated depressive symptomology [87]. For both those newly diagnosed and those with ongoing diabetes, the demands of the regimen, particularly the often complicated dietary issues, may be perceived as "one more thing" to do in what may be seen as an already stressful existence [88]. Given the high rates of depression in this population, assessment of depressive symptomology and its impact on dietary intake, related aspects of diabetes self-care and health outcomes are critical.

Social Support

Level of social support, reflected by connections with individuals as well as community support, including social events, church and senior centers, has been linked to mortality rates in older adults with diabetes, with highest levels of social support having the highest survival rate. Having higher levels of social support is associated with lower rates of depressed mood, less difficulty with stress, fewer limitations in activities of daily living, better health ratings, lower likelihood of heart disease and shorter diabetes duration [89].

Social integration and network size also predict involvement in HbA1c provider monitoring and foot examinations, highlighting the potential role of social support in adherence to self-care recommendations [90]. Of note, substantial gender differences have been observed in diabetes health care behaviors, with higher levels of social support associated with improved glycemic control in women with type 2 diabetes and less control in men, perhaps reflecting differing social influences on patterns of eating, drinking, and physical activity level [91]. DSME/T programs that include the older adult's spouse may help to promote optimal diabetes knowledge, psychosocial functioning, and metabolic control. Spousal support can also promote adherence to diabetes self-care behaviors and may ultimately reduce diabetesrelated distress [92]. In older adults who live with a spouse, the husband's food preference is often the best predictor of family meal nutrition [93]. Furthermore, negative spousal perceptions of diabetes patient's diet adherence and setbacks can increase their own diabetes-related distress and depressive symptomology in the long run [94]. This indicates that social isolation is an important influence on dietary intake, but also that day-to-day social support from family and friends may both positively and negatively impact dietary intake in older persons with diabetes and their spouses. As such, it is imperative to evaluate the social context of patients' dietary behaviors. Additionally, instrumental assistance with issues such as transportation, finances, and food preparation should be considered. Social support resources could also include community based options such as religious groups [95]. Adult day centers, health department wellness programs, and home-based nursing can also help to provide support. Examples of useful resource options have been compiled in a resource guide [96].

Cognitive Dysfunction

Systematic and meta-analytic review of population-based, prospective cohort studies of older adults and diabetes-related outcomes indicate that diabetes is associated with a faster decline in cognitive function among older adults. Prevalence rates of both Alzheimer's type and vascular dementia were twice that in persons with diabetes relative to those without [97]. Blood glucose control is associated with cognitive impairment. Hyperglycemia in nondiabetic individuals and higher average glucose levels in individuals with diabetes are related to increased risk of dementia [98]. Multiple crosssectional studies linking hypoglycemia to cognitive dysfunction [99, 100] and research indicating a bidirectional relationship of hypoglycemia and cognitive dysfunction [101]. A greater occurrence of hypoglycemic incidents is found in individuals with cognitive impairment [3], and a higher incidence of dementia is observed in individuals with a history of severe hypoglycemic episodes [102].

If frequent self-monitoring and adherence to specific dietary guidelines is within the cognitive abilities of the older patient with diabetes, then attainment of tight blood glucose control may be a reasonable goal. However, intensive self-management may not always be realistic for many cognitively impaired older adults. Unfortunately, despite the potential physical benefits, intensive management and tight control may require so many day-to-day demands, that this may be difficult to practically achieve. Cognitive dysfunction can make adherence to dietary recommendations particularly difficult. A large study of adults over age 75 with type 2 diabetes found that over one fifth had impaired cognitive status and that understanding of diabetes management (knowing how to adjust medication when ill) and selfcare (blood glucose testing) was worse in those with impairment [103]. Older adults who are cognitively impaired may have difficulty remembering structured mealtimes that are coordinated with their insulin regimen or with following a complicated meal plan, such as carbohydrate counting or using a sliding scale to match insulin units to intake, making such treatment regimens too overwhelming to be practical. For some older individuals, a concrete, structured meal plan can help minimize ambiguity regarding their diabetes diet. Helpful strategies include using cues such as regular meals, setting alarms, providing written information with large print and pictures, and assessing comprehension and skill by asking for demonstrations. In addition, provision of home-based caretakers or meal services may also assist the older person with cognitive impairment or significant physical disability or other barriers to obtain access to optimal nutrition that is consistent with diabetes goals.

Attitudes and Dietary Intake in Older Adults with Diabetes

Older patients' personal views of diabetes may impact their diabetes self-care. A study of adults over age 60 with type 2 diabetes found that perceptions regarding the cause of diabetes, treatment effectiveness, and seriousness of one's diabetes were all significantly associated with quality of life and negative affect. Beliefs regarding treatment effectiveness were particularly predictive of dietary intake and physical activity [104]. An interview study of older adults with diabetes found that friends, family, and media were influential and critical in shaping expectations and diabetes care goals. Personal diabetes care goals predominantly centered on activities of daily living, maintaining independence, and avoiding becoming a burden to family. Diet and exercise were underemphasized relative to medication management [105]. Effective interactions with older patients should consider their experiences and influences, and focus on quality of life and patient definitions of their health care goals. Clearly communicating priorities in self-care tasks may facilitate adherence to medical goals and optimal care. Collaborative provider-patient discussions regarding self-care, and personal preferences and goals, and awareness of personal risk factors, may enhance patient motivation and treatment engagement. Integrating patient values and beliefs about health and quality of life, considering available resources, may best promote long-term regimen adherence [106].

Ethnic/Cultural Issues Influencing Self-Care and Dietary Intake

Research also suggests that ethnic minority, older adults, and traditionally "hard to reach" persons may have culturally unique health-related perspectives that are not effectively targeted by traditionally delivered health promotion interventions [107]. For example, focus group studies with

underserved ethnic minorities found that a prevailing belief was that better health behaviors could build resistance to acute illnesses and keep them healthy, but that chronic diseases such as diabetes were due to fate and heredity and beyond their individual control [105]. Most participants expressed an interest in "doing better" but were not able to specify how such healthful changes might be made. Qualitative evaluations of cultural influences on diabetes reveal the complexity of psychosocial influences on diabetes lifestyle change and why traditional health provider perspective based dietary interventions may fail. An interview-based study of Mexican American women with type 2 diabetes found that personal understanding and interpretation of their diabetes was most heavily based on their family's experiences and on community influences [108]. From the participants' perspectives, the severity of their diabetes was indicated by being treated with insulin injections and the provider being vigilant, while treatment with oral medications and the perception that providers had a lax attitude was taken to mean that the diabetes was not severe. Participant comments revealed that many found that provider comments were predominantly focused on negative aspects of their behavior, were confrontational, and at times demeaning. Provider focus on positive gains from behavior change, reinforcement for accomplishments and avoiding pejorative terms (e.g., obese) may go a long way in engaging many patients and enhancing a more collaborative relationship. Health provider cultural sensitivity, empathy, responsiveness, and continuity in care are also related to patient satisfaction with care in samples of older, underserved adults with diabetes [109]. Cultural beliefs and traditions may also directly impact use of nontraditional approaches to diabetes self-care. In-depth, structured interviews with older adults with type 2 diabetes divided into multiethnic groups (African Americans, Mexican Americans, Native Americans, and rural white Americans) found that one-fourth had used complementary approaches for their diabetes care (50 % of Mexican American participants, 20 % of African Americans, 15 % of Native Americans, and 15 % of rural whites). Many used herbal (15 %) or dietary (9 %) remedies, and teas, spiritual interventions, and other approaches (5 % each) [109].

The strong influence of family and culture on adherence to a diabetes diet and lifestyle change is also evident in focus group based qualitative research with predominantly older Southern, ruraldwelling African American women with type 2 diabetes [95]. The psychological impact of diabetes was characterized as stronger than the physical impact. Participants reported considerable life stress other than diabetes, particularly having a multi-caregiver role. Family members complaining about and being resistant towards healthy food preparation methods was common. Spirituality and religiosity were prominent and viewed as a primary source of emotional support, a positive influence on diabetes and a contributor to quality of life, with church described as an important source of social and emotional support. Findings demonstrate why consideration of the social and cultural context of older adult's lives is critical for the development of interventions to promote diet and lifestyle change.

Quality of Life and Diabetes Diet in Older Adults

For individuals with limited cognitive function or multiple comorbidities, the primary goal of care may be achieving a satisfactory quality of life rather than aggressive treatment regimens. In other cases, for those who are robust, goals and care may be similar to those for younger persons with diabetes. ADA and American Geriatrics Society Consensus Report guidelines for improving care of older persons with diabetes addresses quality of life issues that are more common in this population, including depression, pain, falls with injury, and declining functional status. These guidelines suggest that goals be developed based on functional status and personal desires [3]. See Table 11.1 for additional Consensus Report guidelines.

Table 11.1 Additional consensus recommendations for care of older adults with diabetes^a

Screening for and prevention of diabetes

- Screen older adults for prediabetes and diabetes according to ADA recommendations, if the patient will be likely to benefit from identification of the condition/disease and subsequent intervention
- Implement lifestyle intervention for older adults with prediabetes who are able to participate and are likely to benefit from the prevention of type 2 diabetes

Management of diabetes

- Encourage physical activity, even if not to optimal levels, and implement MNT using simple teaching strategies and community resources while considering patient safety and preferences
- DSME/T in older adults should take into account sensory deficits, cognitive impairment, and different learning styles and teaching strategies and should include caregivers
- In order to develop and update an individualized treatment plan, screen older adults periodically for cognitive dysfunction, functional status, and fall risk, using simple tools such as those at http://www.hospitalmedicine.org/geriresource/toolbox/determine.htm

Pharmacotherapy

- Carefully choose antihyperglycemic therapies, considering polypharmacy. Avoid glyburide in older adult patients. Metformin can be used safely and is the preferred initial therapy in many older adults with type 2 diabetes, but at reduced dose in those with stage III chronic kidney disease, and avoid in those with stage IV or worse. Assess renal function using eGFR, not serum creatinine alone
- Assess patients for hypoglycemia regularly by asking the patient and caregiver about symptoms or signs and reviewing blood glucose logs. In type 2 diabetic patients, hypoglycemia risk is linked more to treatment strategies than to achieved lower A1c (e.g., a patient with a low A1c on metformin alone may be at considerably lower risk of hypoglycemia than a patient with a high A1c on insulin)
- If recurrent or severe hypoglycemia occurs, strongly consider changing therapy and/or targets
- Assess the burden of treatment on older adult patients [caregivers], consider patient/caregiver preferences, and attempt to reduce treatment complexity

Management of older adults with diabetes in settings outside the home

- The glycemic goals for hospitalized older adults with diabetes are usually similar to those for the general population. The use of SSI alone for chronic glycemic management is discouraged in inpatient settings as well as in LTC facilities
- Transitions of older adults with diabetes (e.g., from home or LTC facility to hospital to post-discharge setting) are periods of high risk. Careful medication reconciliation and written information regarding medication dosing and timing help to minimize risk for hyper- and hypoglycemia. Early transition of diabetes care to an outpatient provider is important to modify drug therapy according to changes in clinical status

^aReproduced from Kirkman SM, Briscoe VJ, Clark N, Florez H, Haas LB, Halter JB, et al. Diabetes in older adults: a consensus report. J Am Geriatr Soc. 2012 Oct 25: 2342–56, with permission from John Wiley & Sons

Weight Loss and Quality of Life in Older Adults with Diabetes

Weight loss issues that must be considered for older adults with diabetes include the impact of restrictions on quality of life and potential loss of lean muscle mass from decreased protein intake. In research undertaken with younger adults with diabetes, weight loss programs that combine diet, physical activity, and theoretically guided behavior change techniques have been shown to be the most effective over the short term and can decrease insulin resistance to improve glycemic control [110, 111]. Hypocaloric diets, independent of weight loss, can also improve glucose tolerance and lipid levels and may also be appropriate for older, obese persons with diabetes. Physical activity was successfully incorporated into intensive weight loss support and education intervention with older adults with type 2 diabetes in the Look AHEAD (Action for Health in Diabetes) study. Mean participant age was 58 years, restriction of caloric intake was the primary approach with a goal of limiting total fat calories to 30 %, with a maximum of 10 % saturated fat and minimum of 15 % from protein.

At 1 year follow-up, intervention participants had lost an average of 8.6 % body weight, with greater weight loss and improved cardiovascular risk factors relative to participants in the control condition [112]. Four-year follow-up showed that at all annual assessments, the study's oldest participants (age 65–76 years at baseline) lost significantly more weight than the next oldest group (age 55–64), who, in turn, lost significantly more weight than participants in a younger age group (age 45–54 years) [113].

Recommendations for Imparting Dietary Information

When imparting diabetes diet information for behavior changes, it is critical to consider the psychosocial and cultural influences for each individual patient. Simple and concrete statements such as "eat less fat" or "get more walking in each day" may promote learning and minimize failure. Nutrition information is best presented in sequenced manageable steps that can then be individualized to the patient's setting. The National Diabetes Education Program has prepared materials adapted from those used in the DPP for use by primary care providers for middle age and older adults. Materials address motivational approaches that consider readiness for change, behavioral relapse, and the development of walking programs. This toolkit, the NDEP GAMEPLAN [114] is copyright-free and contains health care provider information with background information and patient handouts that may be copied. It is also important to be mindful of the range of functioning in older adults. Older adults of the World War II generation have tended to be characterized as somewhat reverential toward physicians and the health care system. However, baby boomers tend to have high expectations of their health providers, seeking a convenient yet collaborative relationship with access to additional resources such as self-help publications and internet-based supplements [115]. In order to meet the needs of the range of older adults with diabetes, it is clear that a "one size fits all" approach will not be effective. Rather issues related to culture and ethnicity and generational cohort must be considered.

Technological Advances in Diet Education Distribution

Advances in the delivery of diet and lifestyle change education to older adults in their homes have been made with the use of Internet, telephone, and remote telehealth monitoring approaches. A systematic review indicated that Internet-mediated interventions aimed at promoting a healthy lifestyle in adults over age 50 have yielded improvements in physical activity and weight loss. It is noteworthy that the majority of studies reviewed had developed social network forums as part of their website; however, all reported that they were rarely used by participants [116]. A health department-led individual counseling program, delivered predominantly over the telephone to low-income, ethnic minority adults with diabetes risk factors and an average age over 50, resulted in consumption of more fruits and vegetables and less fat, weight loss and triglyceride improvement relative to a wait list control group at a 6-month follow-up. Phone sessions had substantially higher completion rates relative to in-person group meetings and were rated as the most useful aspect of the program [117]. Systematic review of controlled studies also indicates benefits of telemedicine in home-based care interventions for individuals with chronic health conditions who are over age 60, including older adults with diabetes [118]. The most frequently studied telemedicine intervention involves monitoring of vital signs combined with personal interaction between health care provider and patient, followed by only personal interaction via telephone or videoconferencing. Randomized studies indicate predominantly positive results, particularly for behavioral endpoints such as adherence to medication or diet, and self-efficacy relative to medical endpoints such as blood pressure. In the review, the majority of studies with older adults with diabetes reported positive outcomes for endpoints of glycemic control, blood pressure, lipids, and quality of life and behavioral outcomes. Across chronic conditions, studies including personal contact between the patients and the health care provider show superior outcomes compared to those without. The generalizability of studies of telemedicine in older adults, however, is limited by the exclusion of participants with cognitive, hearing visual impairments, or barriers to communication in many trials.

Conclusion: Clinical Recommendations

- 1. Establish the type of diabetes and medication regimen in order to appropriately integrate dietary goals.
- 2. Consider the importance of cardiovascular risk, including obesity and lipids, in developing diabetes dietary goals and routine assessment.
- 3. Work with the individual to set a goal of achieving and maintaining reasonable body weight. For obese older adults, moderate weight loss may achieve dramatic results and exercise may greatly enhance dietary intervention. Maintenance of behavior change and weight loss is critical. For underweight adults, focus on promotion of optimal nutritional intake and functional status.
- 4. Educate older adults with diabetes about the rationale for diet and lifestyle change and link to health outcomes; promote self-efficacy for change.
- 5. Consider the risk of hypoglycemia for older adults taking insulin—particularly those with poor nutritional status, cognitive dysfunction, poly-pharmacy, and comorbid illness. Encourage frequent self-monitoring and dietary self-treatment and preventive strategies.
- 6. Assess older adults' specific dietary patterns such as food choices, quantity eaten, and unplanned snacking and the lifestyle contexts in which they occur.
- 7. Address psychosocial issues that may influence dietary intake, including depression, social support, cognitive status, attitudes and perceptions, and the impact of the diabetes regimen on quality of life.
- 8. Address and intervene within individuals' cultural context, including family influences and church, when appropriate.
- 9. Provide a collaborative relationship with each patient, offer resources and provide concrete, behavioral strategies to promote behavior change.
- 10. When appropriate, provide self-help materials, including Internet resources.

References

- 1. Centers for Disease Control and Prevention. National diabetes fact sheet: general information and national estimates on diabetes in the United States. Atlanta, GA: U.S. Department of Health and Human Services; 2011.
- 2. Prevention CfDCa. 2011 National diabetes fact sheet Atlanta, GA2013 (October 30 2013). Available from: cdc. gov/diabetes/pubs/factsheet.
- 3. Sue Kirkman M, Briscoe VJ, Clark N, Florez H, Haas LB, Halter JB, et al. Diabetes in older adults: a consensus report. J Am Geriatr Soc. 2012;60:2342–56. PubMed PMID: 23106132. Epub 2012/10/31. Eng.
- 4. Whiting DR, Guariguata L, Weil C, Shaw J. IDF diabetes atlas: global estimates of the prevalence of diabetes for 2011 and 2030. Diabetes Res Clin Pract. 2011;94(3):311–21. PubMed PMID: 22079683. Epub 2011/11/15. eng.
- 5. Corriere M, Rooparinesingh N, Kalyani RR. Epidemiology of diabetes and diabetes complications in the elderly: an emerging public health burden. Curr Diab Rep. 2013;13(6):805–13. PubMed PMID: 24018732.
- 6. Howard BV, Magee MF. Diabetes and cardiovascular disease. Curr Atheroscler Rep. 2000;2(6):476-81.
- 7. American Diabetes Association. Diagnosis and classification of diabetes mellitus. Diabetes Care. 2010;33(S1): S62–9.
- Falorni A, Kockum I, Sanjeevi CB, Lernmark A. Pathogenesis of insulin-dependent diabetes mellitus. Baillieres Clin Endocrinol Metab. 1995;9(1):25–46.
- Schutt M, Fach EM, Seufert J, Kerner W, Lang W, Zeyfang A, et al. Multiple complications and frequent severe hypoglycaemia in 'elderly' and 'old' patients with Type 1 diabetes. Diabet Med. 2012;29(8):e176–9. PubMed PMID: 22506989. Epub 2012/04/18. eng.
- Group TESS. Vitamin D supplement in early childhood and risk for Type 1 (insulin-dependent) diabetes mellitus. Diabetologia. 1999;42:51–4.
- Chiu K, Chu A, Go VL, Saad MF. Hypovitaminosis D is associated with insulin resistance and beta cell dysfunction. Am J Clin Nutr. 2004;79(5):820–5.
- Borges M, Martini LA, Rogero MM. Current perspectives on vitamin D, immune system, and chronic diseases. Nutrition. 2011;27(4):399–404.
- Van Etten E, Mathieu C. Immunoregulation by 1,25-dihydroxyvitamin D3: basic concepts. J Steroid Biochem Mol Biol. 2005;97(1–2):93–101.
- 14. Van Belle TL, Gysemans C, Mathieu C. Vitamin D and diabetes: the odd couple. Trends Endocrinol Metab. 2013;24(11):561–8.
- 15. Morley J. An overview of diabetes mellitus in older persons. Clin Geriatr Med. 1999;15(2):211–24.
- 16. Morley JE. Diabetes, sarcopenia, and frailty. Clin Geriatr Med. 2008;24(3):455-69.
- Amati F, Dube JJ, Coen PM, Stefanovic-Racic M, Toledo FG, Goodpaster BH. Physical inactivity and obesity underlie the insulin resistance of aging. Diabetes Care. 2009;32(8):1547–9. PubMed PMID: 19401446. Pubmed Central PMCID: PMC2713647. Epub 2009/04/30. eng.
- 18. Elahi D, Muller DC. Carbohydrate metabolism in the elderly. Eur J Clin Nutr. 2000;54 Suppl 3:S112-20.
- Beaufrere B, Morio B. Fat and protein redistribution with aging: metabolic considerations. Eur J Clin Nutr. 2000;54 Suppl 3:S48–53.
- Chiu KC, Lee NP, Cohan P, Chuang LM. Beta cell function declines with age in glucose tolerant Caucasians. Clin Endocrinol (Oxf). 2000;53(5):569–75.
- The diabetes control and complications trial research group. The effect of intensive treatment of diabetes on the development and progression of long-term complications in insulin-dependent diabetes mellitus. N Engl J Med. 1993;329(14):977–86.
- 22. Turner RC, Holman RR. Lessons from UK prospective diabetes study. Diabetes Res Clin Pract. 1995;28:S151-7.
- Holman RR, Paul SK, Bethel MA, Matthews DR, Neil HA. 10-year follow-up of intensive glucose control in type 2 diabetes. N Engl J Med. 2008;359(15):1577–89. PubMed PMID: 18784090. Epub 2008/09/12. eng.
- ADVANCE Collaborative Group, Patel A, MacMahon S, Chalmers J, Neal B, Billot L, Woodward M, et al. Intensive blood glucose control and vascular aoutcomes in patients with type 2 diabetes. N Engl J Med. 2008; 358:256–72.
- Duckworth W, Abraira C, Moritz T, Reda D, Emanuele N, Reaven PD. VADT Intestigators. Glucose control and vascular complications in veterans with type 2 diabetes. N Engl J Med. 2009;360(2):129–39.
- Gerstein HC, Miller ME, Byington RP, Goff Jr DC, Bigger JT, Buse JB, et al. Effects of intensive glucose lowering in type 2 diabetes. N Engl J Med. 2008;358:2545–59. PubMed PMID: 18539917. Epub 2008/06/10. eng.
- 27. Skyler JS, Bergenstal R, Bonow RO, Buse J, Deedwania P, Gale EA, et al. Intensive glycemic control and the prevention of cardiovascular events: implications of the ACCORD, ADVANCE, and VA diabetes trials: a position statement of the American Diabetes Association and a Scientific Statement of the American College of Cardiology Foundation and the American Heart Association. J Am Coll Cardiol. 2009;53(3):298–304. PubMed PMID: 19147051. Epub 2009/01/17. eng.
- 28. Inzucchi SE, Bergenstal RM, Buse JB, Diamant M, Ferrannini E, Nauck M, et al. Management of hyperglycemia in type 2 diabetes: a patient-centered approach: position statement of the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetes Care. 2012;35(6):1364–79. PubMed PMID: 22517736. Pubmed Central PMCID: PMC3357214. Epub 2012/04/21. eng.
- Brown AF, Mangione CM, Saliba D, Sarkisian CA. Guidelines for improving the care of the older person with diabetes mellitus. J Am Geriatr Soc. 2003;51(5 suppl):S265–80.
- Cigolle CT, Blaum CS, Halter JB. Diabetes and cardiovascular disease prevention in older adults. Clin Geriatr Med. 2009;25(4):607–41. vii–viii. PubMed PMID: 19944264. Epub 2009/12/01. eng.
- Helmy T, Patel AD, Alameddine F, Wenger NK. Management strategies of dyslipidemia in the elderly: review. Med Gen Med. 2005;7(4):8.
- 32. Casagrande SS, Ríos Burrows N, Geiss LS, Bainbridge KE, Fradkin JE, Cowie CC. Diabetes knowledge and its relationship with achieving treatment recommendations in a national sample of people with type 2 diabetes. Diabetes Care. 2012;35(7):1556–65.
- Garg A, Bonanome A, Grundy SM, Zhang ZJ, Unger RH. Comparison of a high-carbohydrate diet with a highmonosaturated fat diet in patients with non-insulin-dependent diabetes mellitus. N Engl J Med. 1988;319(13):829–34.
- Burge M, Castillo KR, Schade DS. Meal composition is a determinant of lispro-induced hypoglycemia in IDDM. Diabetes Care. 1997;20(2):152–5.

- 35. Chiu CJ, Wray LA. Factors predicting glycemic control in middle-aged and older adults with type 2 diabetes. Prev Chronic Dis. 2010;7(1):A08. PubMed PMID: 20040223. Pubmed Central PMCID: PMC2811503. Epub 2009/12/31. eng.
- 36. Sheard NF, Clark NG, Brand-Miller JC, Franz MJ, Pi-Sunyer FX, Mayer-Davis E, Kulkarni K, Geil P. Dietary carbohydrate (amount and type) in the prevention and management of diabetes: a statement by the American Diabetes Association. Diabetes Care. 2004;27(9):2266–71.
- Mooradian AD, Osterweil D, Petrawek D, Morley JE. Diabetes mellitus in elderly nursing home patients. J Am Geriatric Soc. 1988;36:391–6.
- Kaiser MJ, Bauer JM, Ramsch C, Uter W, Guigoz Y, Cederholm T, et al. Frequency of malnutrition in older adults: a multinational perspective using the mini nutritional assessment. J Am Geriatr Soc. 2010;58(9):1734–8. PubMed PMID: 20863332. Epub 2010/09/25. eng.
- Bochicchio GV, Salzano L, Joshi M, Bochicchio K, Scalea TM. Admission preoperative glucose is predictive of morbidity and mortality in trauma patients who require immediate operative intervention. Am Surg. 2005; 71(2):171–4. PubMed PMID: 16022019.
- 40. Capes SE, Hunt D, Malmberg K, Gerstein HC. Stress hyperglycaemia and increased risk of death after myocardial infarction in patients with and without diabetes: a systematic overview. Lancet. 2000;355(9206):773–8. PubMed PMID: 10711923.
- Capes SE, Hunt D, Malmberg K, Pathak P, Gerstein HC. Stress hyperglycemia and prognosis of stroke in nondiabetic and diabetic patients: a systematic overview. Stroke. 2001;32(10):2426–32. PubMed PMID: 11588337.
- 42. Krinsley JS. Association between hyperglycemia and increased hospital mortality in a heterogeneous population of critically ill patients. Mayo Clin Proc. 2003;78(12):1471–8.
- Pomposelli JJ, Baxter 3rd JK, Babineau TJ, Pomfret EA, Driscoll DF, Forse RA, et al. Early postoperative glucose control predicts nosocomial infection rate in diabetic patients. JPEN J Parenter Enteral Nutr. 1998;22(2):77–81. PubMed PMID: 9527963.
- 44. Umpierrez GE, Isaacs SD, Bazargan N, You X, Thaler LM, Kitabchi AE. Hyperglycemia: an independent marker of in-hospital mortality in patients with undiagnosed diabetes. J Clin Endocrinol Metab. 2002;87(3):978–82. PubMed PMID: 11889147.
- 45. Rodbard HW, Jellinger PS, Davidson JA, Einhorn D, Garber AJ, Grunberger G, et al. Statement by an American Association of Clinical Endocrinologists/American College of Endocrinology consensus panel on type 2 diabetes mellitus: an algorithm for glycemic control. Endocr Pract. 2009;15(6):540–59. PubMed PMID: 19858063. Epub 2009/10/28. eng.
- 46. van den Berghe G, Wouters P, Weekers F, Verwaest C, Bruyninckx F, Schetz M, et al. Intensive insulin therapy in critically ill patients. N Engl J Med. 2001;345(19):1359–67. PubMed PMID: 11794168.
- Investigators N-SS, Finfer S, Chittock DR, Su SY, Blair D, Foster D, et al. Intensive versus conventional glucose control in critically ill patients. N Engl J Med. 2009;360(13):1283–97. PubMed PMID: 19318384.
- Kansagara D, Fu R, Freeman M, Wolf F, Helfand M. Intensive insulin therapy in hospitalized patients: a systematic review. Ann Intern Med. 2011;154(4):268–82. PubMed PMID: 21320942.
- 49. Qaseem A, Chou R, Humphrey LL, Shekelle P, Clinical Guidelines Committee of the American College of Physicians. inpatient glycemic control: best practice advice from the Clinical Guidelines Committee of the American College of Physicians. Am J Med Quality. 2013;29:95–8. PubMed PMID: 23709472.
- 50. American Diabetes Association. Standards of medical care in diabetes: 2013. Diabetes Care. 2013;36 Suppl 1:S11–66. PubMed PMID: 23264422. Pubmed Central PMCID: 3537269.
- 51. Swift CS, Boucher JL. Nutrition therapy for the hospitalized patient with diabetes. Endocr Pract. 2006;12 Suppl 3:61–7.
- 52. Boucher JL, Swift CS, Franz MJ, Kulkarni K, Schafer RG, Pritchett E, Clark NG. Inpatient management of diabetes and hyperglycemia: implications for nutrition practice and the food and nutrition professional. J Am Diet Assoc. 2007;107(1):105–11.
- 53. Gosmanov AR, Umpierrez GE. Medical nutrition therapy in hospitalized patients with diabetes. Curr Diab Rep. 2012;12(1):93–100. PubMed PMID: 21997598. Pubmed Central PMCID: 3746498.
- American Diabetes Association. Diabetes nutrition recommendations for health care institutions (position statement). Diabetes Care. 2004;27 Suppl 1:S55–7.
- 55. Wright J. Total parenteral nutrition and enteral nutrition in diabetes. Curr Opin Clin Nutr Metab Care. 2000; 3:5–10.
- 56. Garg A. High-MUFA diets for patients with DM: a meta-analysis. Am J Clin Nutr. 1998;67 Suppl 3:577S-82.
- Hongsermeier T, Bistrian BR. Evaluation of a practical technique of determining insulin requirements in diabetic patients receiving total parenteral nutrition. J Parenter Enter Nutr. 1993;17:16–9.
- 58. Morley JE. Nutritional status of the elderly. Am J Med. 1986;81:679-95.
- Coulston AM, Mandelbaum D, Reaven GM. Dietary management of nursing home residents with non-insulindependent diabetes mellitus. Am J Clin Nutr. 1990;51:67–71.
- 60. Tariq SH, Karcic E, Thomas DR, Thomson K, Philpot C, Chapel DL, Morley JE. The use of a no-concentratedsweets diet in the management of type 2 diabetes in nursing homes. J Am Diet Assoc. 2001;101:1463–6.

- Grundy SM, Cleeman JI, Merz CN, Brewer Jr HB, Clark LT, Hunninghake DB, et al. Implications of recent clinical trials for the National Cholesterol Education Program Adult Treatment Panel III guidelines. Circulation. 2004;110(2):227–39. PubMed PMID: 15249516. Epub 2004/07/14. eng.
- 62. Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults. Executive summary of the third report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). JAMA. 2001;285:2486–97.
- 63. Ford ES, Giles WH, Dietz WH. Prevalence of the metabolic syndrome among US adults: findings from the third National Health and Nutrition Examination Survey. JAMA. 2002;287:356–9.
- 64. Stern MP, Williams K, Gonzalez-Villalpando C, Hunt KJ, Haffner SM. Does the metabolic syndrome improve identification of individuals at risk of type 2 diabetes and/or cardiovascular disease? Diabetes Care. 2004;27: 2676–81.
- Lorenzo C, Okoloise M, Williams K, Stern MP, Haffner SM. The metabolic syndrome as predictor of type 2 diabetes: the San Antonio Heart Study. Diabetes Care. 2003;26:3153–9.
- 66. Wang J, Ruotsalainen S, Moilanen L, Lepisto P, Laakso M, Kuusisto J. The metabolic syndrome predicts cardiovascular mortality: a 13-year follow-up study in elderly non-diabetic Finns. Eur Heart J. 2007;28(7):857–64. PubMed PMID: 17303589. Epub 2007/02/17. eng.
- Mozaffarian D, Kamineni A, Prineas RJ, Siscovick DS. Metabolic syndrome and mortality in older adults: the Cardiovascular Health Study. Arch Intern Med. 2008;168(9):969–78. PubMed PMID: 18474761. Epub 2008/05/14. eng.
- 68. Crandall J, Schade D, Ma Y, Fujimoto WY, Barrett-Conner E, Fowlder S, Dagogo-Jack S, Andres R, Diabetes Prevention Program Research Group. The influence of age on the effects of lifestyle modification and metformin in prevention of diabetes. J Gerontol. 2006;61A(10):1075–81.
- 69. Diabetes Control and Complications Trial Research Group. Effect of intensive diabetes treatment on the development and progression of long-term complications in adolescents with insulin-dependent diabetes mellitus: diabetes control and complications trial. J Pediatr. 1994;125(2):177–88. PubMed PMID: 8040759.
- 70. Janssen MM, Snoek FJ, de Jongh RT, Casteleijn S, Deville W, Heine RJ. Biological and behavioural determinants of the frequency of mild, biochemical hypoglycaemia in patients with type 1 diabetes on multiple injection therapy. Diabetes Metab Res Rev. 2000;16(3):157–63.
- Meeking DR, Cavan DA. Alcohol ingestion and glycemic control in patients with insulin-dependent diabetes mellitus. Diabet Med. 1997;14(4):279–83.
- 72. Bolli GB. How to ameliorate the problem of hypoglycemia in intensive as well as nonintensive treatment of type 1 diabetes. Diabetes Care. 1999;22 Suppl 2:B43–52.
- 73. Cox DJ, Kovatchev B, Koev D, Koeva L, Dachev S, Tcharaktchiev D, et al. Hypoglycemia anticipation, awareness and treatment training (HAATT) reduces occurrence of severe hypoglycemia among adults with type 1 diabetes mellitus. Int J Behav Med. 2004;11(4):212–8. PubMed PMID: 15657021. Epub 2005/01/20. eng.
- 74. Bainbridge KE, Hoffman HJ, Cowie CC. Risk factors for hearing impairment among U.S. adults with diabetes: National Health and Nutrition Examination Survey 1999–2004. Diabetes Care. 2011;34(7):1540–5.
- Educators AAoD. AADE guidelines for the practice of diabetes self-management education 2011 November 28 2012. Available from: http://www.diabeteseducator.org/export/sites/aade/_resources/pdf/general/PracticeGuide lines2011.pdf.
- Suhl E, Bonsignore P. Diabetes self-management education for older adults: general principles and practical application. Diab Spectr. 2006;19(4):234–40.
- Bouchard DR, Langlois MF, Domingue ME, Brown C, LeBrun V, Baillargeon JP. Age differences in expectations and readiness regarding lifestyle modifications in individuals at high risk of diabetes. Arch Phys Med Rehabil. 2012;93(6):1059–64. PubMed PMID: 22475056. Epub 2012/04/06. eng.
- 78. Horwath CC, Worsley A. Dietary habits of elderly persons with diabetes. J Am Diet Assoc. 1991;91(5):553-7.
- Hemphill RC, Stephens MA, Rook KS, Franks MM, Salem JK. Older adults' beliefs about the timeline of type 2 diabetes and adherence to dietary regimens. Psychol Health. 2012;17. PubMed PMID: 22594631. Pubmed Central PMCID: PMC3447991. Epub 2012/05/19. Eng.
- 80. Cross AT, Babicz D, Cushman LF. Snacking habits of senior Americans. J Nutr Elder. 1995;14(2-3):27-38.
- Navas-Carretero S, Abete I, Zulet MA, Martinez JA. Chronologically scheduled snacking with high-protein products within the habitual diet in type-2 diabetes patients leads to a fat mass loss: a longitudinal study. Nutr J. 2011;10:74. PubMed PMID: 21756320. Pubmed Central PMCID: PMC3155966. Epub 2011/07/16. eng.
- Kimura Y, Wada T, Okumiya K, Ishimoto Y, Fukutomi E, Kasahara Y, et al. Eating alone among community-Dwelling Japanese elderly: association with depression and food diversity. J Nutr Health Aging. 2012;16(8):728– 31. PubMed PMID: 23076516. Epub 2012/10/19. eng.
- Locher JL, Ritchie CS, Robinson CO, Roth DL, Smith West D, Burgio KL. A multidimensional approach to understanding under-eating in homebound older adults: the importance of social factors. Gerontologist. 2008;48(2):223–34. PubMed PMID: 18483434. Pubmed Central PMCID: PMC2756416. Epub 2008/05/17. eng.
- Lustman PJ, Penckofer SM, Clouse RE. Recent advances in understanding depression in adults with diabetes. Curr Diab Rep. 2007;4:114–22.

- Katon W, Russo J, Lin EH, Heckbert SR, Ciechanowski P, Ludman EJ, et al. Depression and diabetes: factors associated with major depression at five-year follow-up. Psychosomatics. 2009;50(6):570–9. PubMed PMID: 19996227. Pubmed Central PMCID: PMC3087499. Epub 2009/12/10. eng.
- Katon WJ, Lin EH, Williams LH, Ciechanowski P, Heckbert SR, Ludman E, et al. Comorbid depression is associated with an increased risk of dementia diagnosis in patients with diabetes: a prospective cohort study. J Gen Intern Med. 2010;25(5):423–9. PubMed PMID: 20108126. Pubmed Central PMCID: PMC2855007. Epub 2010/01/29. eng.
- Black SA, Markides KS. Depressive symptoms and mortality in older Mexican Americans. Ann Epidemiol. 1999;9:45–52.
- Polonsky WH, Dudl RJ, Peterson M, Steffian G, Lees J, Hokai H. Depression in type 2 diabetes: Links to health care utilization, self-care and medical markers. Diabetes. 2000;49 Suppl 1:A64.
- Zhang X, Norris SL, Gregg EW, Beckles G. Social support and mortality among older persons with diabetes. Diabetes Educ. 2007;33(2):273–81.
- Arcury TA, Grzywacz JG, Ip EH, Saldana S, Nguyen HT, Bell RA, et al. Social integration and diabetes management among rural older adults. J Aging Health. 2012;24(6):899–922. PubMed PMID: 22764154. Epub 2012/07/06. eng.
- Kaplan RM, Hartwell SL. Differential effects of social support and social network on physiological and social outcomes in men and women with type II diabetes mellitus. Health Psychol. 1987;6:387–98.
- Stephens MA, Franks MM, Rook KS, Iida M, Hemphill RC, Salem JK. Spouses' attempts to regulate day-to-day dietary adherence among patients with type 2 diabetes. Health Psychol. 2012;32:1029–37. PubMed PMID: 23025302. Epub 2012/10/03. Eng.
- Weidner G, Healy AB, Matarazzo JD. Family consumption of low fat foods: stated preference versus actual consumption. J Appl Social Psychol. 1985;15:773–9.
- Franks MM, Hemphill RC, Seidel AJ, Stephens MA, Rook KS, Salem JK. Setbacks in diet adherence and emotional distress: a study of older patients with type 2 diabetes and their spouses. Aging Ment Health. 2012;16(7):902– 10. PubMed PMID: 22533446. Pubmed Central PMCID: PMC3416920. Epub 2012/04/27. eng.
- 95. Samuel-Hodge CD, Headen SW, Skelly AH, Ingram AF, Keyserling TC, Jackson EJ, Ammerman AS, Elasy TA. Influences on day to day self-management of type 2 diabetes among African American women: spirituality, the multi-caregiver role, and other social context factors. Diabetes Care. 2000;23(7):928–33.
- 96. Brownson CA, Lovegreen SL, Fisher EB. Community and society support for diabetes self-management. In: Mensing C, editor. The art and science of diabetes self-management education: a desk reference for healthcare professionals. Chicago: American Association of Diabetes Educators; 2006.
- Lu FP, Lin KP, Kuo HK. Diabetes and the risk of multi-system aging phenotypes: a systematic review and metaanalysis. PLoS One. 2009;4(1):e4144. PubMed PMID: 19127292. Pubmed Central PMCID: PMC2607544. Epub 2009/01/08. eng.
- Crane P, Walker R, Hubbard RA, Li G, Nathan DM, Zheng H, et al. Glucose levels and risk of dementia. N Engl J Med. 2013;369(6):540–8.
- 99. Launer L, Miller ME, Williamson JD, Lazar RM, Gerstein HC, Murray AM, ACCORD-MIND Investigators, et al. Effects of intensive glucose lowering on brain structure and function in people with type 2 diabetes (ACCORD MIND): a randomized open-label substudy. Lancet Neurol. 2011;10:969–77.
- 100. Punthakee Z, Miller ME, Launer LK, Williamson JD, Lazar RM, Cukierman-Yaffee T, ACCORD Group of Investigators, ACCORD-MIND Investigators, et al. Poor cognitive function and risk of severe hypoglycemia in type 2 diabetes: post hoc epidemiologic analysis of the ACCORD trial. Diabetes Care. 2012;35:787–93.
- 101. Yaffe K, Falvey CM, Hamilton N, Harris TB, Simonsick EM, Strotmeyer ES, Health ABC Study, et al. Association between hypglycemia and dementia in a biracial cohort of older adults with diabetes mellitus. JAMA Intern Med. 2013;173(14):1300–6.
- 102. Whitmer RA, Karter AJ, Yaffe K, Quesenberry Jr CP, Selby JV. Hypoglycemic episodes and risk of dementia in older patients with type 2 diabetes mellitus. JAMA. 2009;301(15):1565–72. PubMed PMID: 19366776. Pubmed Central PMCID: PMC2782622. Epub 2009/04/16. eng.
- 103. Hewitt J, Smeeth L, Chaturvedi N, Bulpitt CJ, Fletcher AE. Self management and patient understanding of diabetes in the older person. Diabet Med. 2011;28(1):117–22. PubMed PMID: 21166853. Epub 2010/12/21. eng.
- Hampson SE, Glasgow RE, Foster LS. Personal models of diabetes among older adults: Relationship to selfmanagement and other variables. Diabetes Educ. 1995;21(4):300–7.
- Huang ES, Gorawara-Bhat R, Chin MH. Self-reported goals of older patients with type 2 diabetes. J Am Geriatr Soc. 2005;53:306–11.
- Jack Jr L, Airhihenbuwa CO, Namageyo-Funa A, Owens MD, Vinicor F. The psychosocial aspects of diabetes care. Geriatrics. 2004;59(5):26–32.
- 107. White SL, Maloney SK. Promoting healthy diets and active lives to hard-to-reach groups: market research study. Public Health Rep. 1990;105(3):224–31.
- Alcozer F. Secondary analysis of perceptions and meanings of type 2 diabetes among Mexican American women. Diabetes Educ. 2000;26(5):785–95.

- 109. Schoenberg NE, Traywick LS, Jacobs-Lawson J, Kart CS. Diabetes self-care among a multiethnic sample of older adults. J Cross Cult Gerontol. 2008;23(4):361–76. PubMed PMID: 18369715. Pubmed Central PMCID: PMC3079270. Epub 2008/03/29. eng.
- 110. Wing RR, Goldstein MG, Acton KJ, Birch LL, Jakicic JM, Sallis Jr JF, Smith-West D, et al. Behavioral science research in diabetes: lifestyle changes related to obesity, eating behavior, and physical activity. Diabetes Care. 2001;24(1):1–2.
- 111. Mason C, Foster-Schubert KE, Imayama I, Kong A, Xiao L, Bain C, et al. Dietary weight loss and exercise effects on insulin resistance in postmenopausal women. Am J Prev Med. 2011;41(4):366–75. PubMed PMID: 21961463. Pubmed Central PMCID: PMC3185302. Epub 2011/10/04. eng.
- 112. Group TLAR. Reduction in weight and cardiovascular disease risk factors in individuals with type 2 diabetes. Diabetes Care. 2007;30(6):1374–83.
- 113. Wadden TA, Neiberg RH, Wing RR, Clark JM, Delahanty LM, Hill JO, et al. Four-year weight losses in the Look AHEAD study: factors associated with long-term success. Obesity (Silver Spring). 2011;19(10):1987–98. PubMed PMID: 21779086. Pubmed Central PMCID: PMC3183129. Epub 2011/07/23. eng.
- 114. Program NDEP. Small Steps. Big Rewards. Your GAME PLAN to prevent type 2 diabetes: information for patients. Bethesda, MD: National Diabetes Education Program; 2006.
- 115. Clark B. Older, sicker, smarter, and redefining quality: the older consumer's quest for service. In: Dychtwald K, editor. Healthy aging challenges and solutions. Gaithersburg, MD: Aspen Publishers, Inc.; 1999.
- 116. Aalbers T, Baars MA, Rikkert MG. Characteristics of effective Internet-mediated interventions to change lifestyle in people aged 50 and older: a systematic review. Ageing Res Rev. 2011;10(4):487–97. PubMed PMID: 21628005. Epub 2011/06/02. eng.
- 117. Kanaya AM, Santoyo-Olsson J, Gregorich S, Grossman M, Moore T, Stewart AL. The live well, be well study: a community-based, translational lifestyle program to lower diabetes risk factors in ethnic minority and lower-socioeconomic status adults. Am J Public Health. 2012;102(8):1551–8. PubMed PMID: 22698027. Pubmed Central PMCID: PMC3395772. Epub 2012/06/16. eng.
- 118. van den Berg N, Schumann M, Kraft K, Hoffmann W. Telemedicine and telecare for older patients: a systematic review. Maturitas. 2012;73(2):94–114. PubMed PMID: 22809497. Epub 2012/07/20. eng.

Chapter 12 Preventive Cardiology: Counseling Older At-Risk Adults on Nutrition

William E. Kraus and Julie D. Pruitt

Key Points

- Preventive cardiology is a discipline that focuses on the prevention of future cardiovascular disease in all populations, by reducing the burden of known cardiovascular risk factors such as hypertension, dyslipidemia, glucose intolerance, and obesity [1]. Preventive cardiology focuses on heart health by preventing the development of risk factors (health promotion or primordial prevention), the first clinical coronary heart disease, or stroke event (primary prevention) and recurrent events (secondary prevention) [2].
- Special concerns that can affect implementation of cardiac prevention strategies in older at-risk adults include complicated comorbidities; functional limitations, alterations in taste, smell, and appetite; difficulties with medication use/effectiveness; and limited financial, social, and/or care-giver resources.
- Preventive cardiology nutrition recommendations stress increasing the intake of fruits, vegetables, whole grains, and omega-3 and omega-6 fatty acids while substituting nonhydrogenated unsaturated fats for saturated and *trans*-fats. It further recommends minimizing the intake of beverages and foods with added sugar.
- While the basic tenets of these recommendations work well for older at-risk adults, clinical supervision becomes critical in successfully implementing healthy eating in this age group.

Keywords Cardiac event • Hypertension • Dyslipidemia • Lipid-modified diet • Sodium • Sugars

Core Components of Preventive Cardiology

Preventive cardiology brings a detailed understanding of the interplay between known and emerging risk factors to the long-term treatment of cardiac patients particularly targeting cardiometabolic risk factors. In essence, it translates a growing scientific knowledge base into clinical practice. The discipline of preventive cardiology applies research in vascular biology, clinical genetics, cardiovascular

Division of Cardiology, Department of Medicine,

W.E. Kraus, M.D. (🖂) • J.D. Pruitt, M.S., R.D., L.D.N.

Duke University School of Medicine, Duke Molecular Physiology Institute,

³⁰⁰ N. Duke Street, Durham, NC 27701, USA

e-mail: william.kraus@duke.edu; Julie.pruitt@duke.edu

epidemiology, clinical pharmacology, and clinical trials to practical prevention strategies for patients. A key concept is the integration of multidisciplinary components such as smoking cessation, exercise therapy, nutritional counseling, and the best in pharmacologic medical therapy [1]. In this chapter, we will present a practical description of the nutrition component of preventive cardiology with an emphasis on the dietary recommendations available to older at-risk adults.

Nutrition and Cardiovascular Risk

Even with the advances in drug therapy for preventing coronary heart disease, diet, and lifestyle therapies remain the foundation of preventive clinical intervention. This is especially so for older adults. The American Heart Association AHA Diet and Lifestyle Recommendations note that with advancing age, the risk of developing coronary heart disease increases dramatically [2]. A number of longitudinal trials have shown that, as a group, older adults stand to benefit significantly from nutritional and lifestyle interventions that lower their risk or slow the advance of heart disease and cardiovascular events [3].

Research shows that older adults are open and interested in making and sustaining lifestyle changes. It also demonstrates that even relatively small improvements in risk factors (such as small reductions in BP and LDL cholesterol through diet and lifestyle changes) may be of substantial benefit. Improved dietary behaviors can also have a major impact on hypertension, type 2 diabetes mellitus, and dyslip-idemias of many varieties [2].

In general, the goals and recommendations that apply to all populations covered by the AHA Diet and Lifestyle Recommendations are appropriate for older at-risk adults. However, with decreased energy needs and potentially higher vitamin and mineral requirements, older adults are well served by individual counseling that emphasizes nutrient-dense food choices [2].

Bette Davis, the iconic actor is famously attributed with the expression, "Old age is no place for sissies" [4]. She was, of course, commenting on the inevitable toll that aging has on the body and mind. With aging, every system, physiological and mental, experiences a decline and often a break-down in function. This decline, wherever it occurs, cannot be treated in isolation. An approach that integrates all of the older at-risk adult's comorbidities in a pragmatic whole is the one that works the best for each individual.

Special Issues of Concern for Older at-Risk Patients

As persons age, multiple changes occur throughout their bodies and in their lifestyle. Physical changes that affect metabolism, digestion, and caloric intake requirements have a direct and significant impact on cardiovascular risk. Socio-economic change in financial status, mobility, and ability to live independently also impact the person's ability to manage cardiovascular risk. For each person the interplay of physical and socio-economic status is unique and diverse. It is therefore essential that a nutritional approach for preventing or managing coronary heart disease be tailored to the reality of each person [5].

For a health care provider, practicing preventive cardiology in older populations means making a thorough assessment of the person's physical and environmental factors and using them to set up a personalized health plan. In addition to nutritional recommendations, the clinician should identify ongoing nutritional counseling that is both convenient and appropriate to the individual.

With this in mind, there are several issues that present special concerns for cardiac event prevention in older at-risk adults. In fact comorbidities, such as arthritis, chronic obstructive coronary artery disease, and dementia can prevent patients from being referred to nutritional counseling and interfere with heart-healthy eating plans for persons who are referred. In the likely presence of comorbid disease states and associated treatments and altered physiologic response to food, according to Podrabsky and Remig, an average 75-year-old person has three chronic disease conditions and takes five prescription medications [6]. It is highly likely that each disease state and its treatments have their own dietary concerns and restrictions.

One hallmark of the aging process is a progressive decline in physical activity, which is itself a significant cardiovascular risk factor [7]. Although some may view frailty as an indication that lifestyle behavioral counseling, including nutrition, is far beyond the point of having an impact on cardiovascular risk, persons with limited functional capacity may actually derive the greatest benefit in the form of proportional improvements with minimal risk and improved quality of life [8]. In fact, the AHA Diet and Lifestyle Recommendations finds that older patients can successfully follow and sustain healthy behavior change [2]. However, at present, it must be noted that the benefits of nutritional counseling for older at-risk adults remain more conceptual than proven.

Some cardiovascular risk factors present particular challenges in older at-risk adults that can interact with nutritional factors. For hypertension, salt sensitivity increases in the already stiff vasculature of older persons and makes the use of diuretics as a first-line medication more justifiable. However, fragile volume status might make overtreatment and positional hypotensive episodes more likely. At the same time, a decrease in taste sensation may increase a person's desire to use more salt. Dietary sodium content may interact and complicate the process of finding a stable medical regimen for treatment of blood pressure, especially if the dietary sodium content is not stable. Counseling might include alternatives for salt and salt-free food enhancers to ensure the inclusion of foods that are nutritionally dense but uninspiring in taste. For dyslipidemia, accumulating evidence suggests that treating elevated low-density lipoprotein cholesterol levels is beneficial even in older individuals, and that physicians should not shy away from aggressive treatment of lipids, irrespective of the age of the patient [9]. For example, in the PROSPER study, designed to study the efficacy of cholesterol modification with pravastatin in older adults aged 70–82 years, at 3.2 years the active therapy reduced the risk of a primary cardiovascular endpoint by 15 % [9].

Given the well-known difficulty in compliance with statin therapy even in younger cardiovascular patients, nutritional therapy can be particularly valuable for the older at-risk adult in encouraging compliance and reinforcing dietary approaches to cardiovascular risk factors such as hypertension and dyslipidemias. For example, in the course of nutritional assessment the clinician might uncover a symptom that on the surface may not seem nutrition-related, such as lack of concentration or fatigue. This symptom might contribute to patient noncompliance with cardiac medication. By addressing the nutrition issue, the person's level of concentration or fatigue may improve enough for him or her to remember to take the prescribed cardiac medication.

Providing nutrition counseling to an older at-risk adult population elicits the need to discuss population-specific challenges and issues that must be considered in a comprehensive strategy. Implementation of the clinical recommendations that follow may be particularly difficult for this group. Many older at-risk adults have a limited or restricted ability to meet their nutritional needs. As a whole older adults often have limited financial resources for healthy food choices, be they inexpensive canned or frozen vegetables or a loaf of whole-grain bread, much less omega-3 fatty acid-rich fish or fresh fruit. They may also be faced with restricted access to markets and grocery stores or may be physically less able to take trips out of the house.

Dietary Intake and the Management of Heart Disease

Extensive research has been conducted on many different individual nutritional components and the impact they have on coronary heart disease and associated risk factors. Mente and de Koning completed an extensive review of prospective cohort studies or randomized trials investigating dietary

exposures in relation to coronary heart disease. They identified increased intake of fruits, vegetables, and whole grains, substitution of nonhydrogenated unsaturated fats for saturated and trans-fats, and increased intake of omega-3 fatty acids as effective dietary strategies for preventing coronary heart disease [10]. Additionally, incorporation of low-fat dairy products on a daily basis helps to control blood pressure and may, possibly, assist in weight management, if needed [11, 12]. Structuring heart-healthy nutrition education around these principles will support the goals of preventive cardiology.

Fruits and Vegetables

Numerous studies have demonstrated that diets rich in fruits and vegetables are correlated with prevention of coronary heart disease and associated risk factors [10]. In a large epidemiologic study, a significant inverse correlation between risk of coronary heart disease and intake of fruits and vegetables was noted. Every single serving/day increase in intake was correlated with a 4 % decrease in risk. Persons in the highest quintile of intake (9.15–10.15 servings/day) had a 20 % lower relative risk of developing coronary heart disease than persons in the lowest quintile of intake (2.5–2.9 servings/day). Especially beneficial were leafy green vegetables and vitamin C-rich fruits and vegetables [13]. In the landmark DASH trials, increased consumption of fruits and vegetables in the setting of additional healthy food behaviors, led to markedly decreased blood pressure levels [11]. The beneficial effects of including fruits and vegetables in the diet are thought to come, in part, from the provision of potassium, fiber, phytochemicals, and the displacement of unhealthier food choices.

Counseling persons to include additional fruits and vegetables in their diets should take into account several key elements. First, while respecting the person's dignity, discussion of ability to procure fruits and vegetables is important. Exploration of viable options for increasing fruit and vegetable intake, such as selecting inexpensive preparations (canned or frozen), selection of inseason varieties, home-delivered farm shares, and the like is important. Second, emphasis on the importance of serving the fruits and vegetables in a healthy and appealing manner is essential. People often offset the benefit gained from the inclusion of fruits and vegetables in their diet by adding copious amounts of fats, sugars, and salt during food preparation. Third, starchy vegetables such as corn, potatoes, peas, and beans should be considered part of the starches food group and not the vegetable food group. Nonstarchy vegetables are very low in calories and can be consumed liberally, even in the setting of weight management. Last, at-risk people should be encouraged to eat whole fruits and vegetables whether they are fresh, frozen, or canned in their own juice or light syrup over drinking fruit and vegetable juices is important. Whole fruits and vegetables provide the extra benefit of fiber and increased satiety. In general, over consumption of calories from fruit rarely occurs and only when a person consumes more than the recommended amount of juice or dried fruit-less satisfying foods. Increasing the consumption of healthy fruits and vegetables is often a new experience for many at-risk older adults. Increasing their self-efficacy through education and counseling to establish these new habits is critical in their eventual success.

Dietary Fats

There is undeniable evidence that diets high in saturated and *trans*-fats are significant contributors to the risk of coronary heart disease, while replacing saturated fats and *trans*-fats with healthy monoun-saturated and polyunsaturated fats reduce risk. Differences in total fat intake do not significantly

Table 12.1 Benefits vs. risks of eating fish^a

Some types of fish may contain high levels of mercury, PCBs (polychlorinated biphenyls), dioxins, and other environmental contaminants. Levels of these substances are generally highest in older, larger predatory fish and marine mammals.

The benefits and risks of eating fish vary depending on a person's stage of life.

- Children and pregnant women are advised by the U.S. Food and Drug Administration (FDA) to avoid eating those
 fish with the potential for the highest level of mercury contamination (e.g., shark, swordfish, king mackerel, or
 tilefish); to eat up to 12 oz (two average meals) per week of a variety of fish and shellfish that are lower in mercury
 (e.g., canned light tuna, salmon, pollock, catfish); and to check local advisories about the safety of fish caught by
 family and friends in local lakes, rivers, and coastal areas.
- For middle-aged and older men and postmenopausal women, the benefits of eating fish far outweigh the potential risks when the amount of fish eaten is within the recommendations established by the FDA and Environmental Protection Agency.
- Eating a variety of fish will help minimize any potentially adverse effects due to environmental pollutants.
- Potential exposure to some contaminants can be reduced by removing the skin and surface fat from these fish before cooking. Consumers should also check with local and state authorities about types of fish and watersheds that may be contaminated and visit the FDA Web site for the most up-to-date information on recommendations for specific subgroups of the U.S. population (e.g., children, pregnant women).

^aAmerican Heart Association. (2013). Fish 101. Retrieved from http://www.heart.org/HEARTORG/GettingHealthy/ NutritionCenter/Fish-101_UCM_305986_Article.jsp

influence disease risk; only substitution of saturated and *trans*-fats with monounsaturated and polyunsaturated fats significantly impact risk [14, 15]. This is an extremely important concept to grasp. In the Women's Health Initiative Study, an 8.2 % reduction in total fat intake (2.9 % from saturated fat, 0.6 % trans-fats, 3.3 % monounsaturated fats, 1.5 % polyunsaturated fats) did not result in significant impact on coronary heart disease risk [14]. However, replacement of 5 % of energy from saturated fat and 2 % of energy from trans-fat with monounsaturated and polyunsaturated fats was associated with a 42 % and 53 % lower risk of coronary heart disease, respectively, in the Nurses' Health Study [15]. Targeted reductions and substitutions of fat subtypes are critical to improving a person's cardiac risk profile.

In addition to replacing unhealthy fats in the diet with healthy fats, inclusion of omega-3 and omega-6 fats can provide substantial benefits. The 2010 Dietary Guidelines for Americans' recommends including 8 oz of fatty fish in the diet each week as moderate evidence has shown that this level of consumption is associated with reduced cardiac deaths among individuals with and without preexisting cardiovascular disease [16]. Plant-based sources of omega-3 fatty acids have also been indicated in the secondary prevention of coronary heart disease [17]. It is important to encourage older at-risk adults to increase omega-3 fatty acids in their diet by including eight ounces of low-contaminate (Table 12.1) fatty fish each week and plant-based sources such as flaxseed, English walnuts, and canola oil, within the context of recent safety advisories. Higher amounts of omega-6 may improve insulin resistance, reduce diabetes risk, and lower blood pressure. The AHA recommends a daily intake of at least five to ten percent of daily energy as omega-6 fatty acids. Replacing saturated fats with polyunsaturated fats is a good solution as they are a natural source of omega-6 fatty acids [18].

Managing the balance of fat in the diet through the replacement of unhealthy saturated and *trans-fats* with healthy omega-3, omega-6, monounsaturated, and polyunsaturated fats can represent a significant lifestyle change for many older at-risk adults. This will likely be a shift away from many of the comfort foods that are ingrained in their home environment. Acknowledging the difficulty of these changes can be helpful as the person begins to set goals. Smaller, more attainable goals lead to early success and increased self-efficacy.

Dairy Products

Dairy products may play a role in managing blood pressure and weight. Adding low-fat dairy products such as milk and yogurt to a diet rich in fruits and vegetables strengthens the diet's ability to reduce cardiac risk. In the setting of weight maintenance, the DASH feeding studies noted that addition of two to three servings of low-fat dairy products to the basic diet more than doubled the reduction in blood pressure achieved by subjects using this dietary approach [11]. While these positive effects on blood pressure have been well documented, research also suggests that dairy products may be a useful addition to weight loss programs. A meta-analysis of 29 randomized controlled trials concluded that dairy products may have a modest impact on facilitating weight loss in short-term or energy-restricted programs [19].

Studies also support the role of calcium and low-fat dairy products in weight and body fat mass management. Recent meta-analyses found that including dairy in the setting of energy restriction significantly improves loss of weight, fat mass, and waist circumference [19, 20]. A randomized controlled trial compared increased calcium through supplementation and inclusion of dairy products. While individuals taking calcium supplements experienced improvement over the control group in weight, total body fat, and trunk fat loss, individuals in the dairy group had substantially greater losses [12].

These smaller trials indicate that adding dairy products to the diet is beneficial in the management of weight and body fat. However, more extensive research is needed. Virtually all older at-risk adults can benefit from the inclusion of two to three servings of dairy products in their daily diet. Primarily, these servings should come from low-fat milk and yogurt. Owing to their higher fat content and lower calcium content, cheese and butter, should be considered as daily fat servings, rather than as dairy servings.

Whole Grains and Starches

In the past several years, starches have received a lot of negative press. Some fad diet creators even suggest virtually eliminating starches from the diet. However, starches are essential components of a healthy diet, as they provide certain essential nutrients and serve as a vital energy source. Starches also serve an important function in satiation. A distinction to be made during nutritional counseling is in the *type* of starch to include. While refined grains are stripped of their nutrients, whole grains provide a bounty of nutrients, phytochemicals, and fiber.

Inclusion of whole grains and cereal fiber in the diet decreases risk of disease. In the Iowa Women's Health study, Jacobs et al. found a clear inverse relation between intake of whole grains and risk of heart disease [21]. Persons in the highest quintile of intake (3.2 servings/day) had a 30 % lower relative risk of heart disease than persons in the lowest quintile of intake (0.2 servings/day) [21]. Similarly, the Nurse's Health Study also demonstrated a 34 % lower risk of heart disease in women in the highest quintile of intake of fiber. The decreased risk was only significant for dietary fiber from cereal grains, and not for fiber from fruits and vegetables [22].

Introducing or increasing whole grain products in the diet can be structured in a stepwise manner with the goal of at least half of all starch servings in the diet provided by whole grain products. It is productive to have the person begin to examine the starch products he or she currently consumes and identify areas for change. Initial increases may come from mixing whole grain breakfast cereals in with favorite refined grain cereals or purchasing whole grain pastas. Eventually, the person can continue to transition to additional whole grain products as the palate and gastrointestinal tract adjusts.

Added Sugars

As the American diet has become increasingly reliant on processed food and beverages, the general intake of added sugars has risen dramatically. It is estimated that the mean intake of sugars by all persons lies between 22.2 teaspoons (355 cal) and 34.3 teaspoons (549 cal) per day [23]. Dietary sugars can be divided into two types. The first type includes sugars that occur naturally in fruits and vegetables and are consumed as an integral part of the food. The second type includes sugars that are added to processed food or added manually just before they are consumed. For the purposes of cardiac prevention, nutrition counseling should emphasize natural sugar intake and set strict limits for any foods with added sugars.

Added sugars come in many varieties including sucrose, maltose, glucose, cane sugar, fructose, and high-fructose corn syrup, to name only a few. Research indicates that a heavy intake of added sugars, especially fructose consumption in the form of sugared soft drinks is driving the epidemics of insulin resistance, obesity, hypertension, dyslipidemia, and type 2 diabetes, all major risk factors for coronary heart disease [23]. For this reason the American Heart Association recommends limiting consumption of products with added sugar to no more than 6 teaspoons (100 cal) for women and 9 teaspoons (150 cal) for men per day [23]. Beverages sweetened with sugars may not be readily identified by at-risk people as problematic. In teaching about sources of hidden sugars, clinicians should remind their clients that sugary beverages count and encourage older at-risk adults to reach for plain water when they are thirsty rather than a sugared drink containing empty calories.

Teaching older adults about controlling their sugar consumption can be a very difficult issue and might take a two-pronged approach. In the first instance, the clinician can teach the difference between the natural sugars found in fruits and vegetables and the added sugars used in processed foods and beverages or by the spoonful to sweeten coffee, tea, or breakfast cereal. In the second instance the clinician can provide motivation to the person to satisfy the urge for something sweet with a piece of fruit rather than a piece of candy or a sweet drink. To help older at-risk adults manage their added sugar intake, clinicians might teach how to read and interpret the nutrition labeling on the food items. Not only will this skill come in handy for making better choices at the grocery store, it may help in exerting portion control at the table.

Additional Relevant Dietary Components

Stanols and Sterols

Plant stanols and sterols are organic compounds found naturally in vegetable oils, cereals, fruits, and vegetables. Additionally, they are now added to products such as margarines and orange juice. Due to their cholesterol-like structure, these compounds interfere with the absorption of cholesterol and cholesterol building blocks in the digestive tract. The net effect of this lowered absorption is a 6–15 % reduction in LDL cholesterol [24]. The Third Report of the National Cholesterol Education Program Expert Panel of Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III) recommends consuming two grams of plant stanol and sterol esters per day as a therapeutic option for managing LDL cholesterol [24]. To achieve an optimal intake level, inclusion of fortified products is necessary as the amount of stanols and sterols occurring naturally in foods is minimal.

Alcohol

While alcohol intake and a reduction in cardiac events has been widely publicized in the popular media and supported by several research studies, most guidelines do not recommend encouraging the addition of alcohol in the heart-healthy diet. The addictive nature and adverse consequences of overconsuming alcohol may outweigh the potential benefits. In fact, as noted, excessive alcohol intake can be especially problematic in the older at-risk adult population. Also, it is becoming clear that many of the studies that found a potential benefit of moderate alcohol consumption perhaps overestimated the benefits by incorrectly including reformed high-volume alcoholic consumers in the abstinence group. While the proper role of alcohol in the diet and for risk reduction are still being investigated, if a participant currently consumes alcohol, the American Heart Association recommends limiting intake to not more than one drink per day for women and two drinks per day for men [2].

Sodium

Numerous studies have shown that reducing intake of dietary sodium can have a significant effect on lowering blood pressure, a key risk factor for heart disease. In the DASH-Sodium trial, subjects were maintained on sodium intakes of 3,300, 2,400, and 1,500 mg/day in a cross-over design. Blood pressure was significantly reduced with each incremental drop in sodium intake [25]. While sodium intake at the lowest level showed the greatest overall lowering effect, maintaining this level of intake is difficult at best. Therefore, the American Heart Association recommends consuming no more than 1,500 mg of sodium per day [2].

To remain under the recommended level of sodium intake, counsel older at-risk adults to steer clear of high sodium seasonings, canned meats, soups and vegetables, and salty snacks and to avoid adding salt when cooking or at the table. Encouraging persons to experiment with low-sodium flavor agents to maintain or add flavor to their foods is a viable strategy [25].

Alternative Dietary Patterns

One of the most significant ways that cardiac risk can be reduced is through achieving and maintaining a healthy body weight. As a result, older at-risk adults who are overweight or obese often solicit advice about initiating one of the numerous popular diets for weight loss. For this reason health professionals need to be aware of two key points emerging from research in this field. Several studies have noted that weight loss is not significantly different after 12 months in subjects following any one of several popular diets [26–28]. In a randomized clinical trial comparing a low-carbohydrate, highprotein, high-fat diet to a low-calorie, high-carbohydrate, low-fat diet, greater improvement of HDL cholesterol was experienced on the low-carbohydrate diet [27]. Changes in other risk factors (LDL, blood pressure, insulin sensitivity), however, were not significantly different between the groups or, as demonstrated by meta-analysis, were found to be unfavorable on a low-carbohydrate diet [27, 28]. When discussing alternative popular diets with persons, examine the restrictions set forth by the ones that limit the consumption of proven beneficial foods (fruits, vegetables, low-fat dairy products, and whole grains) discussed previously. The long-term health ramifications and safety of excluding or limiting these foods is, as yet, unknown.

Recommendations for Appropriate Caloric Intakes and Dietary Patterns

Practical implementation of a healthy diet pattern begins with laying a foundation of appropriate caloric intake. For the older at-risk adult population achieving an appropriate caloric intake may mean either a decrease or an increase in current caloric intake. In this population less strict adherence to specific calorie levels and more emphasis on healthy eating patterns should be the emphasis. The tool presented in Table 12.2 was developed by dietitians at the Duke Center for Living and Duke Cardiac Rehabilitation program. Based on the Harris Benedict equation, it is a user-friendly tool requiring minimal calculations to approximate caloric intake (see Table 12.2). Once an appropriate calorie level is identified, one can determine the recommended contributions from each dietary component. A healthy cardiac diet allows for an estimated 25–30 % of total calories from fat—mainly from healthy monounsaturated and polyunsaturated fats by limiting unhealthy saturated fat to 7 % of total calories. A quick reference for total fat and saturated fat gram allowance for varying calorie levels is provided (Table 12.3). Selection of the calorie level closest to the older at-risk adult's calculated needs is facilitated.

Table 12.3 shows two suggested meal plans that take into account the calories from fat and the recommendations for following a diet pattern rich in fruits, vegetables, low-fat dairy products, and whole grains. The first plan (Table 12.4, Chart 1) includes provisions for persons who consume beef, poultry, seafood, eggs, and cheese, while the second (see Table 12.4, Chart 2) is specifically designed for lacto-ovo vegetarians (individuals who do not consume animal products with the exception of eggs, cheese, and dairy products). To advise a specific person, one begins by finding the caloric level that most closely matches the patient's recommended caloric intake on the appropriate chart.

Table 12.2 Determining your daily calorie allowance^a

Step 1:	Write current weight (lbs.) then multiply by 10=	
Step 2:	Choose one from Steps a–e below. (<i>It is important not to have a calorie level <1,200 without evaluation by nutritionist.</i>)	
	a. If you want to tone up body, <i>maintain weight or lose less than 10 lb</i> , then add 500	
	b. If you want to <i>lose 10–25 lb</i> , then add 0.	
	c. If you want to lose greater than 25 lb, subtract 500	
	d. If you weigh 350 lb or more, subtract 1,000.	
	e. If you want to gain weight, add 1,000.	
Step 3:	Add calories in right hand column from Steps 1 and 2. This is your estimated Calorie needs per day	Calories per day

^aCalorie levels are based on a person engaging in ~30 min of exercise 3-5 days/week

Calorie needs	Maximum daily <i>total</i> fat gram budget	Maximum daily saturated fat gram budget
1,300	40	10
1,400	43	11
1,600	48	12
1,800	51	14
2,000	58	16
2,200	66	17
2,400	73	19
2,600	79	20
2,800	87	22
3,000	95	23

Table 12.3 Daily fat gram budget chart

Table 12.4	Recommended n	umber of serving	s by food group	for varying level	s of kilocalorie i	ntake for vegeta	ians and nonveg	etarians		
Chart 1 ^a	1,300	1,400	1,600	1,800	2,000	2,200	2,400	2,600	2,800	3,000
Fat Grams	≤43	≤45	≤52	≤58	≤64	≤70	≤78	≤84	≤90	≤98
Starch ^c	4	4	5	5	6	7	7	8	6	10
Fruit	2	3	4	4	5	5	5	6	7	7
Veg	3+	4+	4+	5+	5+	6+	6+	6+	7+	7+
Dairy	2	2	2	2	2	2	e G	ю	e G	n
M&Pd	3	33	4	6	6	8	8	8	8	6
Chart 2 ^b	1,300 VEG	1,400 VEG	1,600 VEG	1,800 VEG	2,000 VEG	2,200 VEG	2,400 VEG	2,600 VEG	2,800 VEG	3,000 VEG
Fat Grams	≤43	≤45	≤52	≤58	≤64	≤70	≤78	≤84	≤90	≤98
Starch ^e	5	5	6	6	7	8	6	10	11	12
Fruit	2	2	e G	e G	4	4	4	5	6	6
Veg	3+	3+	4+	5+	5+	5+	6+	6+	7+	7+
Dairy	3	6	3	3	ю	3	3	ю	3	e,
$M\&P^d$	2	2	2	3	ю	3	3	4	4	5
^a Chart 1 is a 1 ^b Chart 2 is a 1	olan for people v	vho eat beef, pou vho are lacto-ovo	ltry, seafood, egg vegetarian (eat o	gs, and cheese only eggs, cheese	e, and dairy anin	al products)				

-Chart 2 is a plain for people who are lacto-over vegetarian (can only eggs, cheese, and dairy an "Include at least half of all starch servings from whole grain products ^dM&P: Meat (beef, poultry, seafood, pork, etc.) and other proteins such as eggs, cheese, nuts

Conclusion

The goal of nutrition therapy for older adults at-risk of coronary heart disease is the adaptation and maintenance of healthy behaviors in order to improve cardiovascular risk profile and prevent all cardiac events over his or her lifetime. Nutrition therapy begins with an assessment of current dietary practices and a review of potential barriers, followed by identification of targeted areas for change and creation of a practical and effective plan. Including the services of a registered dietitian in an overall heart-healthy strategy may be beneficial in managing the complexities of changing dietary behaviors and providing nutrition education. Involving individuals who are part of the at-risk adult's support network and providing support and encouragement for each step will foster continued positive changes and cardiac risk reduction.

Clinical Recommendations for a Heart-Healthy Diet in Older Individuals

- Complete a thorough assessment of the nutrition status and home socioeconomic context of the person.
- 2. Provide education on a diet rich in fruits, vegetables, low-fat dairy products, and whole grains and low in saturated and trans-fats.
- 3. Assist the at-risk adult in identifying changes they can make to substitute nonhydrogenated unsaturated fats for saturated and trans-fats in their diets.
- Encourage the at-risk adult to increase their intake of natural food sources of omega-3 and omega-6 fatty acids.
- 5. Approximate the at-risk adult's optimal caloric intake (see Table 12.2) for supporting healthy weight maintenance goals.
- Develop a person-specific implementation plan that acknowledges areas needing improvement and solutions to potential barriers.

References

- Fuster V, Hirshfeld Jr JW, Brown AS, Brundage BH, Fye WB, Lewis RP, et al. Working group 8: defining the different types of cardiovascular specialists and developing a new model for training general clinical cardiologists. J Am Coll Cardiol. 2004;44(2):267–71.
- Lichtenstein AH, Appel LJ, Brands M, Carnethon M, Daniels S, Franch HA, et al. Diet and lifestyle recommendations revision 2006. A Scientific statement from the American Heart Association Nutrition Committee. Circulation. 2006;2006(114):82–96.
- 3. Tourlouki E, Matalas AL, Panagiotakos DB. Dietary habits and cardiovascular disease risk in middle-aged and elderly populations: a review of evidence. Clin Interv Aging. 2009;4:319–30.
- Search quotes. [Internet] Bette Davis quotes. [updated 2013 cited 2013 May 6] Retrieved from http://www.searchquotes.com/quotation/Old_age_is_no_place_for_sissies./218285/
- 5. Culross B. Nutrition: meeting the needs of the elderly. ARN Network 2008; Aug./Sept. 7
- 6. Podrabsky M, Remig V. Public policy initiative III: meeting the nation's new aging reality. ADA Times 2005; 3(2).
- Fleg JL, Morrell CH, Bos AG, Brant LJ, Talbot LA, Wright JG, et al. Accelerated longitudinal decline of aerobic capacity in healthy older adults. Circulation. 2005;112(5):674–82.
- Ades PA, Savage PD, Cress ME, Brochu M, Lee NM, Poehlman ET. Resistance training on physical performance in disabled older female cardiac patients. Med Sci Sports Exerc. 2003;35(8):1265–70.
- Shepherd J, Blauw GJ, Murphy MB, Bollen EL, Buckley BM, Cobbe SM, et al. Pravastatin in elderly individuals at risk of vascular disease (PROSPER): a randomised controlled trial. Lancet. 2002;360(9346):1623–30.
- Mente A, de Koning L, Shannon HS, Anand SS. A systematic review of the evidence supporting a causal link between dietary factors and coronary heart disease. Arch Intern Med. 2009;169(7):659–69.

- Appel LJ, Moore TJ, Obarzanek E, Vollmer WM, Svetkey LP, Sacks FM, et al. A clinical trial of the effects of dietary patterns on blood pressure. DASH Collaborative Research Group. N Engl J Med. 1997;336(16):1117–24.
- 12. Zemel MB. The role of dairy foods in weight management. J Am Coll Nutr. 2005;24(6 Suppl):537S-46.
- Joshipura KJ, Hu FB, Manson JE, Stampfer MJ, Rimm EB, Speizer FE, et al. The effect of fruit and vegetable intake on risk for coronary heart disease. Ann Intern Med. 2001;134(12):1106–14.
- Howard BV, Van Horn L, Hsia J, Manson JE, Stefanick ML, Wassertheil-Smoller S, et al. Low-fat dietary pattern and risk of cardiovascular disease: the Women's Health Initiative Randomized Controlled Dietary Modification Trial. JAMA. 2006;295(6):655–66.
- 15. Kyungwon O, Hu FB, Manson JE, Stampfer MJ, Willertt WC. Dietary fat intake and risk of coronary heart disease in women: 20 years of follow-up of the Nurses' Health Study. Am J Epidemiol. 2005;161(7):672–9.
- 16. US Department of Agriculture. Dietary guidelines for Americans. 7th ed. Washington, DC: U.S. Government Printing Office; 2010.
- 17. Kris-Etherton P, Daniels SR, Eckel RH, Engler M, Howard BV, Krauss RM, et al. Summary of the scientific conference on dietary fatty acids and cardiovascular health: conference summary from the nutrition committee of the American Heart Association. Circulation. 2001;103(7):1034–9.
- American Heart Association. (2010). Omega-6 fatty acids Science advisory. Retrieved from http://www.heart.org/ HEARTORG/GettingHealthy/FatsAndOils/Fats101/Omega-6-Fatty-Acids---Science-Advisory_UCM_306808_ Article.jsp
- Chen M, Pan A, Malik VS, Hu FB. Effects of dairy intake on body weight and fat: a meta-analysis of randomized controlled trials. Am J Clin Nutr. 2012;96(4):735–47.
- Abargouei AS, Janghorbani M, Salehi-Marzijarani M, Esmaillzadeh A. Effect of dairy consumption on weight and body composition in adults: a systematic review and meta-analysis of randomized controlled clinical trials. Int J Obes. 2012;36(12):1485–93.
- Jacobs Jr DR, Meyer KA, Kushi LH, Folsom AR. Whole-grain intake may reduce the risk of ischemic heart disease death in postmenopausal women: the Iowa Women's Health Study. Am J Clin Nutr. 1998;68(2):248–57.
- 22. Wolk A, Manson JE, Stampfer MJ, Colditz GA, Hu FB, Speizer FE, et al. Long-term intake of dietary fiber and decreased risk of coronary heart disease among women. JAMA. 1999;281(21):1998–2004.
- Johnson RK, Appel LJ, Brands M, Howard BV, Lefevre M, Lustig RH, et al. Dietary sugars and cardiovascular health: a scientific statement from the American Heart Association. Circulation. 2009;120:1011–20.
- Expert Panel on Detection, Evaluation, And Treatment of High Blood Cholesterol In Adults. Executive summary of the third report of the National Cholesterol Education Program (NCEP) (Adult Treatment Panel III). JAMA. 2001;285(19):2486–97.
- Sacks FM, Svetkey LP, Vollmer WM, Appel LJ, Bray GA, Harsha D. Effects on blood pressure of reduced dietary sodium and the Dietary Approaches to Stop Hypertension (DASH) diet. DASH-Sodium Collaborative Research Group. N Engl J Med. 2001;344(1):3–10.
- 26. Freedman MR, King J, Kennedy E. Popular diets: a scientific review. Obes Res. 2001;9 Suppl 1:1S-40.
- 27. Foster GD, Wyatt HR, Hill JO, Makris AP, Rosembaum DL, Brill C, et al. Weight and metabolic outcomes after 2 years on a low-carbohydrate versus low-fat diet: a randomized trial. Ann Intern Med. 2010;153(3):147–57.
- Nordmann AJ, Nordmann A, Briel M, Keller U, Yancy Jr WS, Brehm BJ, et al. Effects of low-carbohydrate vs lowfat diets on weight loss and cardiovascular risk factors: a meta-analysis of randomized controlled trials. Arch Intern Med. 2006;166(3):285–93.

Chapter 13 Chronic Heart Failure

Justin M. Vader, Christopher L. Holley, and Michael W. Rich

Key Points

- Heart failure is the leading cause of hospitalization in the Medicare age group. The prognosis for established heart failure in persons over age 65 is poor, with 5-year survival rates of less than 50 % in both men and women.
- The pharmacotherapy of heart failure with reduced ejection fraction (HFREF) is well established, with angiotensin-converting enzyme inhibitors, beta-blockers, and aldosterone antagonists having the most proven benefit.
- Treatment of heart failure with preserved ejection fraction (HFPEF) is an area of active research, but no therapies have been definitively shown to reduce mortality.
- Unintentional weight loss in heart failure is likely due to both increased energy utilization and decreased availability of fat, protein (amino acids), and carbohydrates despite "normal" caloric intake.
- Moderate dietary sodium restriction, such as a 2-g sodium diet, is appropriate for most patients with heart failure, and excess fluid intake should be avoided.
- Some patients will require supplementation with potassium, calcium, and/or magnesium if adequate amounts cannot be obtained from the diet. However, the importance of most vitamins and other micronutrients in the pathogenesis and treatment of chronic heart failure has not been well characterized.

J.M. Vader, M.D.

Cardiovascular Division, Department of Internal Medicine, Washington University School of Medicine, 660 South Euclid Ave., Campus Box 8086, St. Louis, MO 63110, USA e-mail: jvader@dom.wustl.edu

C.L. Holley, M.D., Ph.D.

M.W. Rich, M.D. (🖂) Washington University School of Medicine, 660 S. Euclid Avenue, Campus Box 8086, St. Louis, MO 63110, USA e-mail: mrich@dom.wustl.edu

Cardiovascular Division, Department of Internal Medicine, Barnes-Jewish Hospital, 660 S. Euclid Avenue, Box 8086, St. Louis, MO 63110, USA e-mail: cholley@dom.wustl.edu

Keywords Cardiac cachexia • Cardiac function • Electrolytes • Heart failure • Heart failure with preserved ejection fraction • Heart failure with reduced ejection fraction • Nutrition • Sodium restriction

Dropsy [heart failure] is usually produced when a patient remains for a long time with impurities of the body following a long illness. The flesh is consumed and becomes water. The abdomen fills with fluid; the feet and legs swell; the shoulders, clavicles, chest, and thighs melt away.

Hippocrates [1]

Introduction

Age-related changes in the cardiovascular system coupled with the increasing prevalence of cardiovascular disorders with advancing age, especially hypertension, coronary artery disease, and valvular heart disease, lead to a progressive rise in the prevalence of heart failure with increasing age. Nutritional factors may contribute to the development of heart failure and, conversely, heart failure may lead to nutritional deficiencies. Dietary factors also play a key role in heart failure management. This chapter provides an overview of the clinical features of heart failure and reviews the interactions between heart failure, nutrition, and aging. The chapter concludes with a summary of nutritional recommendations for management of older patients with heart failure.

Overview of Heart Failure

Background

Heart failure is a condition in which one or more abnormalities in cardiac function lead to an inability of the heart to pump sufficient blood to meet the body's metabolic needs while maintaining normal or near-normal intra-cardiac pressures and blood volumes. As of 2010, 6.6 million American adults are estimated to have heart failure, and by 2030 prevalence is expected to increase by 25 % [2]. Both the incidence and prevalence of heart failure increase with advancing age [3], with over 70 % of hospital admissions for heart failure occurring in persons 65 years of age or older [4] and more than 50 % occurring in persons over the age of 75 [5]. As a result, heart failure is the leading cause of hospitalization in the Medicare age group, and it is currently the most costly cardiovascular illness in the United States [6]. Moreover, it is anticipated that the rapid growth in the older adult population will result in a marked increase in the number of older persons with heart failure during the next two to three decades [7].

Etiology

In the United States, chronic hypertension and coronary heart disease account for 70–80 % of heart failure cases [8, 9]. In older women, hypertension is the most common etiology of heart failure, accounting for almost 60 % of cases, while in older men coronary heart disease and hypertension each account for 30–40 % of heart failure cases [9]. Valvular heart disease (especially aortic stenosis and mitral regurgitation) and non-ischemic cardiomyopathies are also common causes of heart failure in

older adults. Less frequent causes include infective endocarditis, pericardial disease, thyroid disorders, and drug toxicity (e.g., alcohol or anthracyclines).

Pathophysiology

The cardiac cycle is divided into a filling phase (diastole) and a pumping phase (systole). Impaired cardiac filling due to increased "stiffness" of the heart (e.g., from left ventricular hypertrophy as a consequence of hypertension) results in increased intra-cardiac pressures and reduced cardiac output despite preserved contractile function, leading to the syndrome of heart failure with preserved ejection fraction (HFPEF, sometimes called "diastolic heart failure"). Conversely, damage to the heart muscle (e.g., from a myocardial infarction) that results in impaired pumping action is referred to as heart failure with reduced ejection fraction (HFREF) or "systolic heart failure." Of note, both systole and diastole require energy in the form of adenosine triphosphate (ATP), with diastole being more sensitive to ATP depletion [10]. This explains why diastolic dysfunction precedes systolic dysfunction in the setting of ischemia. Although most patients with heart failure have evidence for both systolic and diastolic dysfunction, those with a left ventricular ejection fraction of less than 40–45 % are usually classified as having predominantly HFREF, whereas those with an ejection fraction above this range are classified as having HFPEF.

Recent studies indicate that about half of heart failure cases are associated with impaired systolic function, while the remainder have normal or near-normal systolic function at rest (i.e., HFPEF) [8, 11]. HFPEF is more common in women than in men, and the proportion of patients with HFPEF increases markedly with age. In the Cardiovascular Health Study, two-thirds of women over age 65 with heart failure had preserved systolic function, as compared with only 41 % of men in this age group [12]. Although treatment of systolic and diastolic heart failure is similar in some respects, it is important to evaluate ventricular function by echocardiography, radionuclide angiography, magnetic resonance imaging, or cardiac catheterization in all patients with newly diagnosed heart failure because, as discussed below, there are important differences in pharmacotherapy depending on the degree of impairment in contractile function.

Clinical Features

The cardinal symptoms of heart failure include exertional shortness of breath and fatigue, reduced exercise tolerance, orthopnea, and lower extremity edema. Palpitations and orthostatic light-headedness are also common, but chest discomfort in the absence of ischemia is not usually present. Physical findings may include tachycardia, tachypnea, elevated jugular venous pressure, moist pulmonary rales, an S_3 or S_4 gallop, hepatomegaly, and dependent pitting edema. In patients with advanced or long-standing heart failure, there is loss of lean body mass, particularly muscle mass, which in severe cases may progress to the syndrome of cardiac cachexia.

Prognosis

The prognosis for established heart failure in persons over age 65 is poor, with 5-year survival rates of less than 50 % in both men and women [13]. Mortality models, particularly the Seattle Heart Failure Model, can individualize this estimate based upon clinical and laboratory data [14], although

these estimates may be inaccurate in patients over age 80 [15]. In addition to poor survival, chronic heart failure is characterized by recurrent hospitalizations for acute exacerbations [16, 17], a marked increase in the risk of sudden death due to arrhythmia [18], and substantially impaired quality of life due to diminished activity tolerance. Although the short-term prognosis (i.e., 3–6 months) is somewhat more favorable in patients with HFPEF compared to HFREF, the long-term prognosis is similar [19, 20], as are hospitalization rates, symptom severity, and functional capacity [21]. Noncardiovascular death is more common in HFPEF than in HFREF, but arrhythmias and progressive heart failure are the most common causes of death in both groups [22].

Treatment

Optimal treatment of chronic heart failure combines both nonpharmacological, pharmacological, and device-based approaches [23]. Essential nonpharmacological measures include patient education, dietary counseling, sodium and in some cases fluid restriction, attention to psychosocial and financial concerns, and close follow-up. Older patients with multiple comorbid conditions or complex environmental issues often benefit from a multidisciplinary approach to care delivery, involving nurses, social workers, dietitians, therapists, pharmacists, and physicians [24, 25].

The pharmacotherapy of systolic heart failure has been studied extensively over the last 25 years. Angiotensin-converting enzyme (ACE) inhibitors are the cornerstone of treatment, and available evidence indicates that these agents are as effective in older as in younger heart failure patients [26]. Angiotensin II receptor blockers (ARBs) and the combination of hydralazine and isosorbide dinitrate are suitable alternatives in patients who are unable to tolerate ACE inhibitors [27–30]. The beta blockers carvediolol, metoprolol, and bisoprolol have also been shown to reduce mortality and improve left ventricular function in stable heart failure patients at least up to the age of 80 [31, 32]. A fourth agent, nebivolol, showed a nonsignificant reduction in mortality in a trial conducted exclusively in patients >70 years of age [33]. Metoprolol and carvedilol are the only beta blockers approved for systolic heart failure in the U.S. Digoxin improves symptoms and reduces hospitalizations for heart failure but has no effect on survival, with similar effects in older and younger patients [34, 35]. Diuretics are important for maintaining normal volume status and for managing acute heart failure exacerbations, but with the exception of the aldosterone antagonists spironolactone and eplerenone, diuretics have no discernible effect on the natural history of heart failure. Spironolactone reduces mortality in patients with advanced systolic heart failure and is indicated in patients who remain symptomatic despite the above therapeutic measures [36]. The recent findings of the EMPHASIS trial, in which eplerenone provided a mortality benefit in New York Heart Association (NYHA) Class II patients, have led to expanded indications for spironolactone and eplerenone in the most recent European Heart Failure guidelines (Class IA recommendation for patients with NYHA Class II–IV symptoms and EF \leq 35 %) [37, 38]. Current pharmacotherapy of systolic heart failure is summarized in Fig. 13.1.

In contrast to systolic heart failure, no agents are specifically indicated for the treatment of HFPEF. Trials with ACE inhibitors and ARBs have shown no mortality benefit in this population [39–41]. Pharmacologic agents that have been shown to improve *symptoms* in selected patients with diastolic heart failure include nitrates, ACE-inhibitors, and beta blockers [21, 39]. Guidelines recommend that hypertension be treated aggressively and coronary artery disease be managed with medications and/or revascularization as indicated [42]. Diuretics are indicated for controlling volume overload, but over-diuresis should be avoided.

While pharmaceutical management of heart failure has proven successful, surgically implanted left ventricular assist devices (LVADs) have emerged as an option for patients with advanced heart failure, either as a bridge to heart transplantation (BTT) or as stand-alone "destination" therapy (DT). Pulsatile LVADs improve longevity and quality of life compared to medical therapy in NYHA class IV heart



Fig. 13.1 Pharmacotherapy of left ventricular systolic dysfunction. Hatched regions denote conditions for which improved outcomes have been documented in prospective randomized clinical trials. *ACE* angiotensin-converting enzyme. *Angiotensin-receptor blockers (ARBs) and the combination of hydralazine/isosorbide dinitrate are acceptable alternatives or adjuncts to ACE inhibitors in selected patients

failure patients ineligible for heart transplant [43]. Second-generation continuous-flow devices now offer benefit beyond older pulsatile devices [44], and have supplanted pulsatile devices in clinical practice. Limited data suggest that carefully selected patients up to age 80 experience improved symptoms and quality of life with DT-LVAD, although the risk of gastrointestinal bleeding is higher than in younger patients [45].

General Nutritional Aspects of Heart Failure

Heart Failure as a Metabolic Syndrome

Heart failure is a chronic progressive disorder characterized by a host of neurohormonal, immunologic, and metabolic derangements (Table 13.1) [46]. In acute heart failure, activation of the sympathetic nervous system and renin-angiotensin-aldosterone axis serve to maintain cardiac output and preserve tissue perfusion. However, chronic activation of these systems is deleterious and perpetuates progression of the heart failure syndrome. Indeed, current therapy for heart failure focuses on antagonizing the harmful effects of these two neurohormonal pathways through the use of beta blockers, ACE inhibitors, angiotensin receptor blockers, and aldosterone antagonists.

In addition to activation of neurohormonal systems, chronic heart failure is associated with immunological dysregulation, as evidenced by increased levels of circulating tumor necrosis factor-alpha (TNF- α), interleukins 1 and 6, soluble adhesion molecules, and certain leucocyte chemokines [47, 48]. Activation of these cytokines likely plays a pivotal role in the apoptosis (programmed cell death) and anorexia that are features of chronic heart failure. In addition, some cytokines may exert direct cardiotoxic effects (e.g., through increased oxygen free radical activity), thereby contributing to heart failure progression.

Activation/up-regulation/high levels	Reduction/resistance/down-regulation
Sympathetic nervous system	Vagal tone
• Increased circulating norepinephrine and epinephrine	-
Activation of renin-angiotensin-aldosterone system	
Increased angiotensin II levels	
 Increased aldosterone levels 	
Atrial and B-type natriuretic peptide levels	
TNF-α and IL-1/IL-6	
Endothelin-1 and vasopressin (anti-diuretic hormone)	Peripheral blood flow (decreased nutrient delivery)
Cortisol levels	Dehydroepiandrosterone (DHEA) ^a
Insulin levels (in noncachectic patients)	Insulin sensitivity (manifest as insulin resistance ^a)
Growth hormone	Normal or reduced insulin-like growth factor-1 (IGF-1)
Leptin	Ghrelin
Basal metabolic rate (BMR)	Thyroid function
^a Denotes anti-anabolic effect	

Table 13.1 Neurohormonal and metabolic abnormalities in chronic heart failure

Many of the neurohormonal and immunologic abnormalities in chronic heart failure are also associated with important effects on metabolism. While the mechanisms underlying these effects are complex and not fully understood, the net effect is an imbalance between catabolic (tissue-wasting) and anabolic (tissue-building) factors [49]. Cardinal features of advanced heart failure include an increase in basal metabolic rate (BMR) [50], altered protein and fat metabolism, and impaired peripheral blood flow with reduced nutrient delivery to bodily tissues. In chronic advanced heart failure, these effects lead to tissue wasting and loss of lean body mass [51].

Cardiac Cachexia

Hippocrates's early description of "dropsy" (i.e., heart failure) cited at the beginning of this chapter provides a remarkably apt characterization of cardiac cachexia. While tissue wasting and loss of muscle mass occur early in chronic heart failure, marked tissue wasting and cachexia are hallmarks of advanced or end-stage heart failure.

Cardiac cachexia has been variously defined over the years, but a 2008 consensus statement defines adult cachexia as weight loss of 5 % or more within 12 months (or BMI <20) plus three of five clinical or laboratory criteria: decreased muscle strength, fatigue, anorexia, low fat-free mass index, and abnormal laboratory values (anemia, low serum albumin, or elevated markers of inflammation) [52]. Estimates of the prevalence of cachexia in patients with heart failure vary widely, but typically range from 10 to 20 %. Importantly, cachexia differs markedly from starvation (e.g., due to anorexia nervosa). Although patients with heart failure may exhibit signs of malnutrition, in cachexia there tends to be a greater loss of lean body mass, principally muscle mass but also bone mass, whereas in starvation there is preferential loss of adipose tissue in the early stages, with subsequent loss of muscle mass as malnutrition progresses [51]. In addition, prolonged starvation is almost invariably associated with a very low body mass index (BMI), whereas patients with cardiac cachexia may experience only modest reductions or even increased body weight, in part due to increased extracellular fluid accumulation (edema) and replacement of muscle mass by fat mass.

As noted above, the complex cascade of metabolic disturbances leading to cardiac cachexia is incompletely understood. However, circulating levels of TNF- α are invariably elevated in patients with cardiac cachexia; indeed, the TNF- α level is the strongest predictor of weight loss in heart failure patients [53]. It is currently unknown whether TNF- α plays a direct pathophysiological role in the

development of cardiac cachexia or if it merely serves as a marker for the cachectic state (and the severity of heart failure), but circulating TNF- α levels are a strong independent predictor of mortality in heart failure patients. Unfortunately, clinical trials of TNF- α blocking agents have been disappointing (see below).

Other circulating factors that may play a role in cardiac cachexia include endotoxin, leptin, and ghrelin. Circulating endotoxin may arise when extensive fluid overload causes gut edema, allowing bacterial toxins to move from the bowel to the blood stream. This leads to increased inflammatory cytokines with systemic effects as noted above. Plasma leptin levels have recently been shown to correlate with prognosis in cases of non-ischemic heart failure, with high plasma leptin predicting disease progression [54]. In contrast to leptin, ghrelin is a centrally acting peptide produced in the stomach that *stimulates* food intake and can increase circulating levels of growth hormone. Heart failure patients appear to have some degree of ghrelin-resistance, which normalizes after heart transplantation [55]. Administration of ghrelin has shown promising results in both animal models of heart failure as well as in a small clinical trial, in which heart failure patients treated with ghrelin for 3 weeks had increased left ventricular function, exercise capacity, and lean body mass [56, 57]. B-type natriuretic peptide (BNP), which is expressed at high levels in patients with decompensated heart failure, has been shown to exert anorectic effects and to reduce ghrelin concentrations when administered to healthy men, underscoring the interrelatedness of heart failure and energy balance [58].

Caloric Intake, Fat, and Protein

The occurrence of weight loss in patients with advanced heart failure is somewhat paradoxical, since heart failure is often associated with reduced physical activity. Furthermore, excepting those heart failure patients who develop anorexia (due either to heart failure itself or as a result of medications), caloric intake is similar in heart failure patients (including those with cachexia) and persons without heart failure [59]. This combination of preserved caloric intake and apparently reduced activity would be expected to result in weight gain rather than weight loss. What, then, accounts for the net loss in non-edematous body mass encountered in end-stage heart failure patients?

First, as noted above, most studies have shown that despite reduced muscle mass, there is an increase in BMR in most patients with heart failure, most likely due to increased energy requirements for respiration and the generalized catabolic state arising from neurohormonal dysregulation (esp. increased circulating catecholamines) [60, 61]. Second, although caloric intake is maintained, there is evidence that fat absorption is impaired in patients with heart failure, perhaps due to bowel edema [62]. Although intestinal handling of protein is usually preserved [63], alterations in protein and carbohydrate metabolism result in impaired delivery of these nutrients to the body's tissues [51]. Excessive adrenergic tone resulting in vasoconstriction is an important and potentially reversible mechanism that limits insulin and glucose delivery to skeletal muscle [64]. Further, compared to matched controls with similar dietary protein and carbohydrate intake, nonobese heart failure patients have increased nitrogen excretion and consequently negative nitrogen balance [65]. In sum, weight loss in heart failure is likely due to increased energy utilization, decreased availability of fat, protein (amino acids), and carbohydrates, and increased nitrogen excretion—all despite "normal" caloric intake.

Management of cardiac cachexia is problematic, and the best approach is to focus on prevention. In this regard, preliminary data suggest that both ACE inhibitors and beta-blockers may exert favorable effects on weight loss in heart failure patients. In a post-hoc analysis of the Studies of Left Ventricular Dysfunction (SOLVD) trial, patients receiving the ACE inhibitor enalapril were 19 % less likely to experience weight loss of 6 % or more during a mean follow-up period of 35 months compared to patients receiving placebo (p=0.05) [66]. Although cachexia is an ominous sign associated with significant mortality, there is some evidence that pharmacologic treatment of heart failure can

lead to partial reversal of the cachectic state. In a small observational study involving 13 heart failure patients with cachexia, 6 months of treatment with either carvedilol or metoprolol was associated with significant weight gain accompanied by favorable effects on plasma norepinephrine and leptin levels [67]. In another study of eight patients with advanced heart failure and cardiac cachexia, treatment with the combination of digoxin, the ACE-inhibitor enalapril, and the loop diuretic furosemide was associated with significant clinical improvement, as well as increased muscle bulk, subcutaneous fat, and serum albumin and hematocrit levels [68]. In contrast to these studies, two large trials of etanercept, a tumor necrosis factor (TNF) blocking agent, failed to demonstrate significant benefit and were discontinued [69]. These findings were particularly disappointing since TNF- α may play a role in the development of cardiac cachexia.

Beyond pharmacologic approaches to prevent or treat cachexia, a few studies have examined the role of nutritional support in advanced heart failure patients. In one small randomized trial, a high caloric diet failed to result in significant changes in nutritional status or clinical outcomes of patients with advanced heart failure [70]. In another study of patients with moderate to severe heart failure and malnutrition receiving high-energy nasogastric tube feedings for 2 weeks, total body weight and extracellular fluid weight declined but lean body mass increased [71]. There was, however, no change in oxygen consumption or cardiac function [72]. A third study involving only six patients undergoing mitral valve surgery showed that perioperative nutritional support was associated with improved clinical status and stable cardiac function. Regarding amino acid supplementation, in a study of eight patients with cardiac cachexia, infusion of branched-chain amino acids had no discernible effect on protein metabolism [73]. Conversely, a specially formulated oral amino acid supplement, in conjunction with standard pharmacologic therapy, appeared to increase exercise capacity via improved circulatory function, muscle oxygen consumption, and aerobic energy production in a small randomized trial of outpatients aged 65–74 with heart failure [74]. Finally, the GISSI-HF trial randomized patients with chronic systolic heart failure to placebo or 1 g/day of n-3 polyunsaturated fatty acids (n-3 PUFA). There was a modest, but statistically significant decrease in cardiovascular hospitalization and overall mortality [75]. The applicability of this observation to dietary intake of n-3 PUFA is not known. In summary, there are limited data at present to assess the impact of various modes of nutritional support on metabolic parameters or clinical outcomes in patients with advanced heart failure. While the observations in amino acid and PUFA trials are intriguing, additional study is needed before these interventions can be recommended for routine heart failure care.

Heart Failure and Obesity

The epidemic of obesity in the United States and elsewhere has led to an increasing proportion of heart failure patients who are overweight or obese. Obesity is independently associated with the development of heart failure, as well as with hypertension, diabetes, obstructive sleep apnea, pulmonary hypertension, and other comorbidities that adversely affect the heart failure patient. Obesity may also result in misdiagnosis of heart failure, since exertional shortness of breath may be due to pulmonary hypertension or physical deconditioning, while lower extremity edema may be caused by venous insufficiency. Conversely, sedentary obese patients may not experience significant shortness of breath until heart failure is far advanced, and early diagnosis of heart failure, at a stage when treatment is more likely to be effective, can easily be missed. Finally, the excess weight carried by overweight heart failure patients results in increased cardiac work, straining the already weakened heart. Indeed, obesity is often associated with increased heart rate and vascular resistance, effects that not only increase cardiac work, but which are diametrically opposed to the actions of beta-blockers and ACE inhibitors. The effect of intentional weight loss in patients with heart failure is an area of active investigation. An emerging body of literature involving obese subjects describes favorable cardiac morphologic [76, 77], metabolic [78], and functional changes after marked weight loss [79]. However, there is currently no evidence that weight loss is associated with improved clinical outcomes in older heart failure patients. Furthermore, several studies have shown that older patients with heart failure and increased body mass index (BMI) have better outcomes than those with normal body mass index (BMI 20–25 kg/m²), and that those with lower BMI (<20 kg/m²) or cardiac cachexia have the worst prognosis [80]. This has been termed the "obesity paradox" and may reflect the fact that those with a higher BMI have better nutritional status and lower NYHA functional class than those with a lower BMI [81]. Thus, the value of weight loss in obese older patients with heart failure is unknown, and additional study is needed before specific recommendations can be made. In the meantime, it seems reasonable to advise older patients who are morbidly obese (BMI \geq 40 kg/m²) to lose a modest amount of weight through a combination of increased physical activity (including strengthening exercises to maintain or increase muscle mass) and reduced caloric intake.

Specific Nutrients in Heart Failure

Water and Sodium

Activation of the renin-angiotensin-aldosterone system in patients with heart failure results in sodium and water retention. As a result, untreated heart failure is usually associated with increased total body water and total body sodium. Of note, total body sodium is generally increased even when serum sodium levels are reduced (i.e., hyponatremia). This situation occurs in patients with advanced heart failure because fluid retention is more pronounced than sodium retention, in part due to the action of vasopressin (anti-diuretic hormone). Indeed, hyponatremia in patients with heart failure is associated with more severe hemodynamic and neurohormonal disturbances, and is a marker for poor prognosis [82]. Antagonists of the vasopressin type-2 receptor promote free water excretion and are approved for treatment of hyponatremia. Though clinical trials in acute heart failure failed to show a mortality benefit [83], these agents may be used for refractory symptomatic hyponatremia.

Diuretics are the mainstay of therapy for fluid overload in heart failure patients. Ideally, diuretic dosages are adjusted to maintain a normal state of hydration (euvolemia). However, over- and underdiuresis are both common, so that at any given time, a patient may be volume-overloaded, euvolemic, or relatively dehydrated, and careful assessment of volume status is thus essential in managing heart failure patients. From a practical standpoint, the simplest way to do this is by monitoring daily weights. Patients should be instructed to weigh themselves every morning without clothing, after voiding, and before eating, and weights should be recorded on a daily weight chart. An optimal or "dry" weight should be established, and variances of more than 2–3 lb in either direction should lead to adjustments in diuretic dosage. The rationale behind this approach is that short-term variability in body weight primarily reflects changes in total body water. Note, however, that non-edematous weight may change over longer periods of time, usually decreasing but occasionally increasing if the overall nutritional status improves. Therefore, periodic reassessment of the patient's desirable weight is appropriate.

In addition to monitoring daily weights and adjusting diuretic dosages, dietary sodium restriction plays a pivotal role in maintaining normal volume status and avoiding acute heart failure exacerbations, as evidenced by the fact that several studies have shown that dietary sodium excess is a common precipitant of repetitive heart failure hospitalizations [84, 85]. Dietary sodium excess contributes to fluid retention, and an acute dietary sodium load (e.g., potato chips, canned soup, "fast food") may

result in a sudden increase in intravascular blood volume, triggering a rise in intra-cardiac pressures and precipitating acute heart failure. Older patients with diastolic heart failure are particularly sensitive to salt intake and changes in blood volume, and are therefore less tolerant of a salt load. While there are no clinical trial data demonstrating improved outcomes with sodium restriction, it is standard of care that heart failure patients, family members, and other caregivers should be educated about the importance of avoiding high sodium foods [23]. Although some patients may find it difficult to adhere to a sodium-restricted diet, careful instruction and guidance from a dietitian is often effective in overcoming this barrier. In contrast to sodium restriction, fluid restriction is not usually required for most patients with mild to moderate heart failure unless significant renal impairment is also present. However, patients should be advised to avoid *excess* fluid intake—the oft-quoted dictum to "drink 8–10 glasses of water every day" does not apply to patients with heart failure. In addition, patients with advanced heart failure accompanied by hyponatremia may benefit from more stringent fluid restriction, e.g., 1.5 L/day total fluid intake.

Other Electrolytes

Apart from their effect on body water, diuretics have important effects on key electrolytes, including sodium, potassium, chloride, magnesium, and calcium. Thiazide and "loop" diuretics (furosemide, bumetanide, torsemide), as well as metolazone, promote urinary loss of sodium, potassium, chloride, and magnesium. As a result, these diuretics may be associated with hyponatremia, hypokalemia, hypochloremia, and hypomagnesemia. In addition, loop diuretics increase calcium excretion and may contribute to a negative calcium balance, although hypocalcemia due to loop diuretics is uncommon. Conversely, the potassium-sparing diuretics spironolactone, eplerenone, triamterine, and amiloride, as well as ACE-inhibitors and angiotensin receptor blockers, are all associated with potassium retention and may occasionally induce significant hyperkalemia. For these reasons, serum electrolytes should be monitored periodically in patients receiving long-term diuretic therapy, especially during periods of dosage adjustment, whenever thiazide and loop diuretics are used together, and in patients with impaired renal function.

Diet and nutrition play an important role in managing electrolytes in heart failure patients. Unfortunately, many trials evaluating micronutrient supplementation in cardiac disease states have excluded patients with heart failure [86]. Despite this limitation, some general principles apply. Patients with heart failure and preserved renal function should consume a diet rich in potassium, magnesium, and calcium, but low in sodium. Most patients on chronic loop diuretic therapy will require potassium replacement, either through high-potassium foods (e.g., fresh fruits) or as potassium supplements (usually administered as potassium chloride, which also aids in chloride replacement). Diuretic-induced hyponatremia is potentially life-threatening and may require hospitalization (e.g., if the serum sodium concentration falls to <120-125 meq/L). Treatment includes fluid restriction, reduction in diuretic dosage, temporary liberalization of sodium intake, and occasionally the use of vasopressin receptor antagonists. Hypomagnesemia is relatively common during long-term diuretic therapy, but may be overlooked unless serum magnesium levels are assessed. Importantly, magnesium deficiency may contribute to muscle fatigue and risk for arrhythmia via prolongation of the QT interval. Treatment consists of dietary therapy and magnesium supplements. The combined potassium and magnesium depleting effects of loop and thiazide diuretics may be moderated by the addition of an aldosterone receptor antagonist, thereby reducing the risk of arrhythmias [87, 88]. Patients with chronic heart failure often suffer bone loss (osteopenia) due to low levels of vitamin D and secondary hyperparathyroidism [46]. However, the value of calcium supplements in heart failure patients, with or without vitamin D, is currently unknown.

Other Minerals

Zinc, manganese, copper, and selenium all have anti-oxidant effects, and deficiencies of these minerals may be associated with increased lipid peroxidation and oxidative stress [89]. In addition, severe copper and selenium deficiency have been associated with cardiomyopathies in humans [90, 91], while zinc and manganese deficiency have been associated with myocardial contractile dysfunction in laboratory animals [92, 93]. Diuretics appear to increase urinary zinc excretion, and clinically significant zinc deficiency is common in older heart failure patients on chronic diuretic therapy [94]. Conversely, serious deficiencies of manganese, copper, and selenium occur infrequently in older adults consuming a normal diet. Based on currently available data, daily intake of each of these minerals should be sufficient to meet Dietary Reference Intakes (DRIs). Although some patients with diuretic-associated zinc deficiency may benefit from zinc supplements, there are currently no data to support routine use of such supplements in older heart failure patients.

Iron is essential for the production of hemoglobin and iron deficiency is common in older adults. Chronic anemia leads to an increase in cardiac work in order to preserve tissue oxygen delivery, and in severe cases may lead to high-output cardiac failure. In a randomized controlled trial, intravenous iron was beneficial in the setting of heart failure and iron depletion, even in the absence of anemia, with respect to symptoms, functional capacity, and quality of life [95]. Conversely, iron overload due to multiple blood transfusions or hemochromatosis has been associated with restrictive cardiomyopathy [96]. Guidelines do not stipulate what constitutes adequate or inadequate stores of iron in the setting of heart failure and it should be noted that oral repletion of iron has not been investigated for benefit in this setting. Iron intake should therefore be sufficient to maintain tissue stores and prevent chronic iron-deficiency anemia, but excess iron intake should be avoided.

Vitamins

Vitamin B₁ (thiamine) deficiency impairs oxidative metabolism and has been unequivocally linked to high-output cardiac failure [89, 97]. In addition, thiamine deficiency may contribute to "diuretic-resistance" in patients receiving moderate to high doses of loop diuretics over a prolonged period of time [98, 99]. In the United States, clinically important thiamine deficiency is most commonly encountered in alcoholics and in older heart failure patients treated with loop diuretics. Of note, both digoxin and furosemide diminish uptake of thiamine by cardiac myocytes, and the effects of these drugs are additive [100]. Thiamine deficiency responds promptly to either oral or parenteral thiamine administration, and is usually associated with substantial improvement in cardiac function and symptoms. Although chronic thiamine supplementation may be considered in selected high risk populations (e.g., alcoholics and poorly nourished older adults treated with high-dose loop diuretics), in most cases maintaining a well-balanced diet will ensure adequate thiamine intake.

Vitamin C supplementation has been associated with improved endothelial function [101–103], and some epidemiologic studies have suggested that increased intake of vitamin C correlates with reduced risk for cardiovascular disease [104–106]. However, there is no convincing evidence that vitamin C deficiency contributes to the development of heart failure, or that vitamin C supplements are beneficial in heart failure patients [89]. Vitamin E has anti-oxidant properties and reduces platelet adhesion [107], and several epidemiologic studies have reported that diets high in vitamin E, alone or in combination with vitamin C, are associated with a lower incidence of coronary heart disease [29, 108–110]. Despite this, several large randomized trials of vitamin E therapy failed to show significant benefit, and one meta-analysis suggested that high-dose vitamin E intake may be associated with increased mortality [111, 112]. Moreover, follow-up analysis of the Heart Outcomes Prevention

Evaluation (HOPE) trial and its extension (HOPE-TOO), showed that vitamin E therapy was associated with increases in the incidence of heart failure and in the risk of hospitalization for heart failure [113]. Therefore, vitamin E is not recommended for prevention or treatment of HF.

Deficiencies of folic acid, vitamin B_6 , and vitamin B_{12} are common in older adults and contribute to age-associated increases in homocysteine levels [114]. Elevated homocysteine is an established risk marker for coronary and cerebrovascular disease in both older and younger adults [115–117], and elevated homocysteine levels have been associated with more severe heart failure and worse prognosis [118]. Despite plausible mechanisms for an adverse effect of homocysteine on myocardial function, there is currently no convincing evidence that lowering homocysteine levels through the use of folic acid and B-vitamin supplements reduces the risk of coronary or cerebrovascular events, or improves myocardial function or outcomes in heart failure patients [118].

Vitamin D is essential for maintaining normal calcium homeostasis, and marked vitamin D deficiency has been associated with decreased myocardial contractility in laboratory animals [119]. Vitamin D deficiency is common in older adults with or without heart failure [120, 121], and is associated with diminished functional capacity [122]. Although vitamin D supplementation would seem appropriate in these individuals, there is no evidence that such treatment alters the clinical course of patients with heart failure. In fact, vitamin D supplementation did not improve 6 min walk distance or quality of life in a randomized trial conducted in older heart failure patients with low vitamin D levels [123]. There are several ongoing trials investigating higher dose vitamin D supplementation. Currently, the DRI for vitamin D is 800 IU in persons older than age 70 and 600 IU daily for those younger than age 70.

Although high-dose niacin is FDA approved for the treatment of dyslipidemia, there is no evidence that niacin deficiency contributes to the development of cardiovascular disease [89], and a niacin preparation failed to reduce cardiovascular events compared with placebo in patients already on statin therapy [124]. Low beta-carotene intake has been associated with increased risk for myocardial infarction [125], but there is no evidence that vitamin A levels correlate with heart failure risk or that vitamin A supplements are useful in the prevention of cardiovascular disease [89]. In the Physician's Health Study, a trial notable for demonstrating the benefit of aspirin in preventing cardiovascular disease, beta-carotene was ineffective [126]. Similarly, there are no established links between vitamins B_2 (riboflavin) and B_{17} (pantothenic acid) and either the development or treatment of cardiac disorders [89].

Other Nutritional Supplements

Despite continued interest in the antioxidant coenzyme Q_{10} (ubiquinone), its role in the pathophysiology and treatment of heart failure remains controversial. Myocardial coenzyme Q_{10} levels are reduced in patients with heart failure, and low plasma coenzyme Q_{10} levels are associated with increased mortality [89], although an analysis of the CORONA trial did not identify coenzyme Q_{10} levels as *independently* prognostic [127]. As the commonly-prescribed lipid-lowering HMG-CoA reductase inhibitors (statins) are associated with depletion of coenzyme Q_{10} [128], the issue of coenzyme Q_{10} repletion arises frequently. Observational studies and some (but not all) small randomized trials indicate that coenzyme Q_{10} supplementation may improve LV function, symptoms, and exercise tolerance [129– 133], but there are no large outcomes trials of coenzyme Q_{10} in heart failure, and routine administration of coenzyme Q_{10} is not recommended by guidelines—even in patients on statin therapy [134].

Carnitine, propionyl L-carnitine, and creatine phosphate are nutritional supplements which may enhance skeletal muscle performance in some patients with heart failure [135–137], but there is little evidence that oral administration improves cardiac function. In addition, there is no evidence that these agents improve long-term clinical outcomes in heart failure patients, and there are also concerns about the safety of these agents during chronic use [89].

Multinutrient Therapy

Older heart failure patients often have multiple nutritional deficiencies, suggesting that therapeutic interventions may need to be broad-based, rather than focusing on a single or even a relatively small number of micronutrients. This is supported by the favorable results in a study reported by Witte et al., in which 30 heart failure patients over age 70 were randomized to receive high-dose micronutrient capsules or placebo in double-blind fashion [138]. The capsules contained calcium ($0.3 \times \text{RDI}$), magnesium ($0.5 \times \text{RDI}$), zinc ($1 \times \text{RDI}$), copper ($1 \times \text{RDI}$), selenium ($0.75 \times \text{RDI}$), vitamin A ($1 \times \text{RDI}$), thiamine ($140 \times \text{RDI}$), riboflavin ($1 \times \text{RDI}$), vitamin B₆ ($100 \times \text{RDI}$), folate ($25 \times \text{RDI}$), vitamin B₁₂ ($200 \times \text{RDI}$), vitamin C ($8 \times \text{RDI}$), vitamin D ($2 \times \text{RDI}$), vitamin E ($40 \times \text{RDI}$), and coenzyme Q₁₀ ($10 \times \text{RDI}$). During a 9-month follow-up period, patients receiving the micronutrient capsules demonstrated decreased left ventricular volumes, an increase in left ventricular ejection fraction (mean 5.3 %), and improved quality of life scores, whereas no changes occurred in the placebo group. These findings require large-scale replication and enthusiasm is tempered by the fact that to date no trials have demonstrated a benefit with regard to death or rehospitalization when one or more micronutrients were added to standard therapy for HF [139].

Impact of Heart Failure Interventions and Age on Nutritional Parameters

Medication Effects

As noted previously, many of the agents used in the treatment of chronic heart failure may have beneficial effects on nutritional status. Conversely, there is a risk for drug-related side-effects and a potentially negative impact on the nutritional status of these patients. Diuretics directly impact fluid and electrolyte homeostasis, and diuretic-induced electrolyte abnormalities are very common. In addition, loop diuretics have been associated with thiamine deficiency, and thiazide diuretics in particular may adversely affect carbohydrate and lipid metabolism. Digoxin may be associated with nausea and anorexia, and these symptoms may occur in older patients even at therapeutic dosages. The ACEinhibitor captopril occasionally causes dysgeusia (altered taste), nausea, and anorexia, and other ACE-inhibitors may be associated with similar side effects, although less frequently. Beta blockers may also influence carbohydrate and lipid metabolism, and depressive symptoms, including reduced appetite, may occur in older patients treated with these agents. Finally, the calcium channel blockers diltiazem and especially verapamil are commonly associated with constipation in older individuals.

Age-Specific Nutritional Issues

Older age is associated with increased risk for a broad range of nutritional deficiencies, and this risk is potentiated by the presence of cardiovascular disease in general and by heart failure in particular (see Chap. 21). In addition, older adults are more susceptible to the adverse effects of pharmacological agents and dietary interventions on nutritional parameters, in part due to preexisting nutritional deficiencies coupled with an increased prevalence of comorbid conditions. The latter issue may be particularly problematic, since the presence of several common comorbidities, e.g., coronary artery disease, diabetes mellitus, and renal insufficiency, may lead to serial dietary restrictions (low fat, low carbohydrate, low protein, low salt) culminating in a diet that is unpalatable and severely deficient in both calories and essential nutrients. It is therefore critically important that an appropriately detailed

nutritional evaluation, including dietary history, body weight, selected laboratory tests (hemoglobin, albumin, cholesterol, electrolytes, creatinine, blood urea nitrogen), and in some cases anthropometric assessments, be incorporated into the routine management of older patients with chronic illnesses, including heart failure.

Device-Specific Nutritional Therapies

As described above, the emergence of LVADs in the care of older patients with heart failure has introduced a new level of complexity. In addition to standard preoperative and postoperative nutritional considerations such as ileus, LVAD therapy may pose unique side-effects including chronic intravascular hemolysis, gastrointestinal bleeding, and chronic soft tissue infection. Ensuring adequate iron stores is important in patients with gastrointestinal blood loss, while chronic intravascular hemolysis calls for supplementation with vitamin B_{12} and folate to ensure adequate erythropoiesis to match erythrocyte destruction. Chronic soft tissue infection is rarely severe enough to result in net catabolism, but one important consideration in patients on long term antibiotics is the effect of mineral supplements on gut absorption of antibiotics. The most notable of these are the interactions between calcium and both tetracyclines and ciprofloxacin, in which absorption of both calcium and the antibiotic is reduced. Guidelines for nutrition in the setting of LVADs are emerging, but contemporary practice is largely shaped by analogy with similar disease states [140].

Recommendations

Nutritional guidelines for managing chronic heart failure in older adults are summarized in Table 13.2. As noted previously, nutritional management of older patients begins with a nutritional assessment, ideally with the assistance of an experienced dietitian or nurse. As with other chronic illnesses, the guiding principle in making nutritional recommendations to older heart failure patients is, first and foremost, maintenance of a well-balanced diet with sufficient calories, nutrients, and fluids to meet daily requirements. In addition, the diet should be both palatable and within the patient's financial means and physical capabilities. Few older patients with heart failure require a therapeutic weight reduction diet, since body weight correlates inversely with mortality in heart failure patients [80]. Indeed, in most cases it is appropriate to prescribe a diet that will either maintain current non-edematous weight or promote a modest increase in lean body mass. Although it has been suggested that the proportion of calories derived from protein and fat should perhaps be increased in older heart failure patients [51], there is little evidence to support this contention, and current recommendations are that 15–20 % of total calories be derived from protein, 25–30 % from fat, and the remaining 50–60 % from complex carbohydrates [141].

Moderate dietary sodium restriction, such as a 2-g sodium diet, is appropriate for most patients with heart failure [23]. More stringent sodium restriction is of unproven benefit; indeed, a recent meta-analysis of randomized trial data demonstrated that sodium restricted patients (average 1.8 g/ day) had a nearly twofold higher risk of death compared to patients with normal sodium intake (average 2.8 g/day). Thus, overzealous salt restriction may be harmful [142]. Patients should be instructed to avoid very high-sodium foods, such as canned soups and sauces, tomato juice, most prepared lunch meats and prepackaged frozen entrees, pickles, fast foods, and certain ethnic foods which are high in sodium (e.g., Asian cuisine). Dining out is potentially problematic, and patients should be advised to call ahead to see if low-sodium options are available. Patients should also be instructed about the widespread availability of alternative seasonings which contain little or no salt.

Component	Recommendation
Nutritional assessment	
Basic (all patients)	Obtain detailed dietary history
	Assess body weight and habitus
	Laboratory: hemoglobin, serum albumin, cholesterol, serum electrolytes (sodium, potassium, calcium, phosphorus, magnesium), creatinine, blood urea nitrogen
Supplemental (selected	Anthropometric measures (e.g. skinfold thickness)
patients)	Determination of lean body mass
	Folate, B ₁₂ levels
	Bone mineral density
	Iron stores (ferritin, iron, iron-binding capacity)
General diet	Well-balanced, rich in fruits and vegetables, whole grains, dairy products, lean meats
Caloric intake	Sufficient to maintain lean body mass; 1,600–2,000 cal/day in most cases
Frotein	15-20 % of total calories
Fat	23-50% of total calories
Eluide	-2 I /day
Tiulus	~2 L/day
	1.5 L/day in setting of hyponatremia, severe renal failure, diuretic-resistance
	Avoid excess fluid intake
Electrolytes	
Sodium	2 g Na ⁺ /day. Avoid overzealous restriction
Potassium, calcium, magnesium	Sufficient to maintain body stores and serum levels; supplement as indicated
Minerals	
Zinc, copper, manganese, selenium	Sufficient intake to meet DRIs; zinc supplements in selected patients
Iron	Sufficient to maintain body stores; avoid iron overload
Vitamins	
Thiamine (Vit B ₁)	Supplement in alcoholics, possibly patients on chronic high-dose loop diuretics
Folate, B_6 , and B_{12}	Supplement if deficient (common)
Cholecalciferol (Vit D)	Supplement if deficient (esp. in osteoporosis) DRI 800 IU/day if >70 year old, otherwise 600 IU
Beta carotene (Vit A)	No known relation to heart failure; maintain DRI
Riboflavin (Vit B ₂)	
Niacin (Vit B ₃)	
Ascorbic acid (Vit C)	
Alpha-tocopherol (Vit E)	
Dietary supplements	
Ubiquinone (coenzyme Q ₁₀)	Unproven benefit, not recommended
Carnitine	Unproven benefit, not recommended
Creatine phosphate	Unproven benefit, not recommended

Table 13.2 Nutritional guidelines for older adults with chronic heart failure

Source is Food and Nutrition Board, Institute of Medicine. Dietary Reference Intakes: The Essential Guide to Nutrient Requirements. Washington, DC: National Academies Press; 2006. Updated Dietary Reference Intakes for Calcium and Vitamin D (2010)

DRI dietary reference intake

Fluid intake should be adequate to maintain hydration while avoiding volume overload. In patients with preserved renal function, about 2 L of fluid per day is appropriate. Excess fluid intake ("8–10 glasses of water a day") should be avoided, but fluid restriction is unnecessary in the absence of hyponatremia, severe renal failure, or advanced heart failure with diuretic resistance. In such cases, fluid intake should be limited to about 1.5 L/day. Alcohol should be avoided in cases of alcohol-induced cardiomy-opathy, but in the systolic heart failure population light-to-moderate drinkers (one to ten drinks/week) do not appear to be at increased risk for adverse outcomes compared to nondrinkers [143, 144].

Caffeinated beverage consumption does not appear to be associated with increased incidence of heart failure [145] and may have favorable effects with respect to exercise in heart failure patients [146]. However, there are few data on the safety of large amounts of caffeine in patients with heart failure, especially with respect to the risk of inducing supraventricular or ventricular arrhythmias—moderation is therefore advised.

Dietary potassium, calcium, and magnesium requirements vary considerably depending on medications, renal function, and comorbid conditions. As a general principle, a well-balanced diet rich in fresh fruits and vegetables, whole grain breads and cereals, and dairy products will provide sufficient amounts of potassium, calcium, and magnesium to meet normal needs. However, many older heart failure patients will require supplemental administration of one or more of these electrolytes to overcome losses through urinary excretion or as a result of other metabolic abnormalities. Since individual requirements cannot easily be predicted, periodic assessment of serum electrolyte levels is appropriate.

As discussed above, the importance of most vitamins and other micronutrients in the pathogenesis and treatment of chronic heart failure has not been well characterized, and it is difficult to make specific nutritional recommendations in most cases. Since older patients are at increased risk for multiple nutritional deficiencies, it is appropriate to maintain a high index of suspicion, particularly in frail, socially isolated, or institutionalized elders, as well as those with multiple comorbidities and those receiving multiple medications. In particular, deficiencies of folate, B_{12} , vitamin D, and zinc are common, and dietary or pharmacological supplementation is indicated when specific deficiencies are identified or suspected. Since long-term administration of loop diuretics may deplete thiamine stores, thiamine replacement should be considered in such cases, particularly in the setting of increasing diuretic-resistance. Finally, although evidence supporting high-dose multivitamin and mineral supplements in older heart failure patients is sparse, daily use of a nonprescription multivitamin and mineral supplement may ease concerns about multinutrient deficiencies and is unlikely to be harmful. Conversely, the use of other dietary supplements, such as coenzyme Q_{10} , carnitine, or creatine phosphate, is not currently recommended, and patients should be screened for the use of these and other neutraceuticals of uncertain safety.

Conclusion

Optimal management of older adults with heart failure requires careful attention to their overall diet and nutritional status, as well as to specific nutrients. Additional research is needed to better define the role of nutritional factors in the pathogenesis and treatment of heart failure in patients of all ages.

Recommendations for Clinicians

- Nutritional management begins with a nutritional assessment. Obtain baseline body weight and monitor at regular intervals. Recognize that caloric needs may change as body weight changes, and provide increased nutritional support if rapid unintentional weight loss occurs.
- Patients should be encouraged to choose a well-balanced diet high in fruits and vegetables as excellent sources of vitamins and electrolytes.
- 3. Sodium restriction to 2 g/day is sufficient. Dietary counseling may be required to assist patients in achieving this goal. Overzealous sodium restriction (<1.5 g/day) is not advised.
- 4. Fluid restriction is not usually necessary, except when hyponatremia, severe renal failure, or advanced heart failure is present. In these cases, fluid should be restricted to ~1.5 L/day. Excess fluid intake ("8–10 glasses of water per day") should be avoided.

(continued)

- 5. Magnesium, potassium, and calcium levels should be monitored and supplemented as needed. Other nutrients of concern in high-risk patients include thiamine, folate, vitamin B₁₂, vitamin D, and zinc. Routine daily use of an oral vitamin/mineral supplement may be helpful in alleviating any deficits.
- 6. Caution should be exercised in translating the results of small trials or trials of proprietary pharmaceutical agents to dietary modifications.

References

- 1. Katz AM, Katz PB. Diseases of the heart in the work of Hippocrates. Br Heart J. 1961;24:257-64.
- Roger VL, Go AS, Lloyd-Jones DM, Benjamin EJ, Berry JD, Borden WB, et al. Heart disease and stroke statistics—2012 update: a report from the American Heart Association. Circulation. 2012;125(1):e2–220.
- 3. Kannel WB. Incidence and epidemiology of heart failure. Heart Fail Rev. 2000;5(2):167-73.
- 4. DeFrances CJ, Podgornik MN. 2004 National hospital discharge survey. Adv Data. 2006;371:1–19.
- Kozak LJ, DeFrances CJ, Hall MJ. National hospital discharge survey: 2004 annual summary with detailed diagnosis and procedure data. Vital Health Stat 13. 2006;(162):1–209.
- Lee WC, Chavez YE, Baker T, Luce BR. Economic burden of heart failure: a summary of recent literature. Heart Lung. 2004;33(6):362–71.
- 7. Heidenreich PA, Trogdon JG, Khavjou OA, Butler J, Dracup K, Ezekowitz MD, et al. Forecasting the future of cardiovascular disease in the United States: a policy statement from the American Heart Association. Circulation. 2011;123(8):933–44.
- Gottdiener JS, Arnold AM, Aurigemma GP, Polak JF, Tracy RP, Kitzman DW, et al. Predictors of congestive heart failure in the elderly: the Cardiovascular Health Study. J Am Coll Cardiol. 2000;35(6):1628–37.
- Levy D, Larson MG, Vasan RS, Kannel WB, Ho kalon KL. [The progression from hypertension to congestive heart failure]. JAMA. 1996;275(20):1557–62.
- 10. Kapila R, Mahajan RP. Diastolic dysfunction. Continuing education in anaesthesia. Crit Care Pain. 2009;9(1): 29–33.
- Vasan RS, Larson MG, Benjamin EJ, Evans JC, Reiss CK, Levy D. Congestive heart failure in subjects with normal versus reduced left ventricular ejection fraction: prevalence and mortality in a population-based cohort. J Am Coll Cardiol. 1999;33(7):1948–55.
- 12. Kitzman DW, Gardin JM, Gottdiener JS, Arnold A, Boineau R, Aurigemma G, et al. Importance of heart failure with preserved systolic function in patients ≥65 years of age. Am J Cardiol. 2001;87(4):413–9.
- Croft JB, Giles WH, Pollard RA, Keenan NL, Casper ML, Anda RF. Heart failure survival among older adults in the United States. Arch Intern Med. 1999;159(5):505–10.
- 14. Mozaffarian D, Anker SD, Anand I, Linker DT, Sullivan MD, Cleland JG, et al. Prediction of mode of death in heart failure: the Seattle Heart Failure Model. Circulation. 2007;116(4):392–8.
- Benbarkat H, Addetia K, Eisenberg MJ, Sheppard R, Filion KB, Michel C. Application of the Seattle Heart Failure Model in Patients >80 years of age enrolled in a tertiary care heart failure clinic. Am J Cardiol. 2012;110(11): 1663–6.
- Vinson JM, Rich MW, Sperry JC, Shah AS, McNamara T. Early readmission of elderly patients with congestive heart failure. J Am Geriatr Soc. 1991;39(10):1290–5.
- Krumholz HM, Parent EM, Tu N, Vaccarino V, Wang Y, Radford MJ, et al. Readmission after hospitalization for congestive heart failure among medicare benificiaries. Arch Intern Med. 1997;157:99–104.
- Ho KK, Anderson KM, Kannel WB, Grossman W, Levy D. Survival after the onset of congestive heart failure in Framingham Heart Study subjects. Circulation. 1993;88(1):107–15.
- Pernenkil R, Vinson JM, Shaha S, Beckham V, Wittenberg C, Rich MW. Course and prognosis in patients > or =70 years of age with congestive heart failure and normal versus abnormal left ventricular ejection fraction. Am J Cardiol. 1997;79(2):216–9.
- Senni M, Tribouilloy CM, Rodeheffer RJ, Jacobsen SJ, Evans JM, Bailey KR, et al. Congestive heart failure in the community: a study of all incident cases in Olmsted county, Minnesota, in 1991. Circulation. 1998;98(21): 2282–9.
- Chinnaiyan KM, Alexander D, Maddens M, McCullough PA. Curriculum in cardiology: integrated diagnosis and management of diastolic heart failure. Am Heart J. 2007;153(2):189–200.

- 22. Zile MR, Gaasch WH, Anand IS, Haass M, Little WC, Miller AB, et al. Mode of death in patients with heart failure and a preserved ejection fraction: results from the Irbesartan in Heart Failure with Preserved Ejection Fraction Study (I-Preserve) trial. Circulation. 2010;121(12):1393–405.
- Heart Failure Society of America. HFSA 2010 comprehensive heart failure practice guideline. J Card Fail. 2010;16(6):e1–192.
- Rich MW, Beckham V, Wittenberg C, Leven CL, Freedland KE, Carney RM. A multidisciplinary intervention to prevent the readmission of elderly patients with congestive heart failure. N Engl J Med. 1995;333(18):1190–5.
- McAlister FA, Lawson FM, Teo KK, Armstrong PW. A systematic review of randomized trials of disease management programs in heart failure. Am J Med. 2001;110(5):378–84.
- Flather MD, Yusuf S, Køber L, Pfeffer M, Hall A, Murray G, et al. Long-term ACE-inhibitor therapy in patients with heart failure or left-ventricular dysfunction: a systematic overview of data from individual patients. Lancet. 2000;355(9215):1575–81.
- Pitt B, Poole-Wilson PA, Segal R, Martinez FA, Dickstein K, Camm AJ, et al. Effect of losartan compared with captopril on mortality in patients with symptomatic heart failure: randomised trial—the Losartan Heart Failure Survival Study ELITE II. Lancet. 2000;355(9215):1582–7.
- Cohn JN, Archibald DG, Ziesche S, Franciosa JA, Harston WE, Tristani FE, et al. Effect of vasodilator therapy on mortality in chronic congestive heart failure. N Engl J Med. 1986;314(24):1547–52.
- Cohn J, Johnson G, Ziesche S. A comparison of enalapril with hydralazine–isosorbide dinitrate in the treatment of chronic congestive heart failure. N Engl J Med. 1991;325(5):303–10.
- Cohn J, Tognoni G. A randomized trial of the angiotensin-receptor blocker valsartan in chronic heart failure. N Engl J Med. 2001;345(23):1667–75.
- CIBIS II Investigators and Committees. The Cardiac Insufficiency Bisoprolol Study II (CIBIS-II): a randomised trial. Lancet. 1999;353(9146):9–13.
- MERIT-HF study group. Effect of metoprolol CR/XL in chronic heart failure: metoprolol CR/Xl randomised intervention trial in congestive heart failure (MERIT-HF). Lancet. 1999;353(9169):2001–7.
- 33. Flather MD, Shibata MC, Coats AJS, Van Veldhuisen DJ, Parkhomenko A, Borbola J, et al. Randomized trial to determine the effect of nebivolol on mortality and cardiovascular hospital admission in elderly patients with heart failure (SENIORS). Eur Heart J. 2005;26(3):215–25.
- The Digitalis Investigation Group. The effect of digoxin on mortality and morbidity in patients with heart failure. N Engl J Med. 1997;336(8):525–33.
- 35. Rich MW, McSherry F, Williford WO, Yusuf S. Effect of age on mortality, hospitalizations and response to digoxin in patients with heart failure: the DIG study. J Am Coll Cardiol. 2001;38(3):806–13.
- Pitt B, Zannad F, Remme W. The effect of spironolactone on morbidity and mortality in patients with severe heart failure. N Engl J Med. 1999;341(10):709–17.
- 37. Zannad F, McMurray J. Eplerenone in patients with systolic heart failure and mild symptoms. N Engl J Med. 2011;364(1):11–21.
- 38. McMurray JJV, Adamopoulos S, Anker SD, Auricchio A, Böhm M, Dickstein K, et al. ESC Guidelines for the diagnosis and treatment of acute and chronic heart failure 2012: the task force for the diagnosis and treatment of acute and chronic heart failure 2012 of the European Society of Cardiology. Developed in collaboration with the Heart. Eur Heart J. 2012;33(14):1787–847.
- Cleland JGF, Tendera M, Adamus J, Freemantle N, Polonski L, Taylor J. The perindopril in elderly people with chronic heart failure (PEP-CHF) study. Eur Heart J. 2006;27(19):2338–45.
- Yusuf S, Pfeffer MA, Swedberg K, Granger CB, Held P, McMurray JJ, et al. Effects of candesartan in patients with chronic heart failure and preserved left-ventricular ejection fraction: the CHARM-Preserved Trial. Lancet. 2003;362(9386):777–81.
- 41. Massie BM, Carson PE, McMurray JJ, Komajda M, McKelvie R, Zile MR, et al. Irbesartan in patients with heart failure and preserved ejection fraction. N Engl J Med. 2008;359(23):2456–67.
- 42. Yancy CW, Jessup M, Bozkurt B, Butler J, Casey DE Jr, Drazner MH, et al. 2013 ACCF/AHA guideline for the management of heart failure: a report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines. J Am Coll Cardiol [Internet]. 2013 [cited 2013 Jul 12]; J Am Coll. Available from: http://jvsmedicscorner.com/Medicine_files/ACCF%3AAHAGuideline for the Management of Heart Failure 2013.pdf
- Rose E, Gelijns A. Long-term use of a left ventricular assist device for end-stage heart failure. N Engl J Med. 2001;345(20):1435–43.
- 44. Slaughter MS, Rogers JG, Milano CA, Russell SD, Conte JV, Feldman D, et al. Advanced heart failure treated with continuous-flow left ventricular assist device. N Engl J Med. 2009;361(23):2241–51.
- 45. Adamson RM, Stahovich M, Chillcott S, Baradarian S, Chammas J, Jaski B, et al. Clinical strategies and outcomes in advanced heart failure patients older than 70 years of age receiving the HeartMate II left ventricular assist device: a community hospital experience. J Am Coll Cardiol. 2011;57(25):2487–95.

- 46. Anker SD, Rauchhaus M. Heart failure as a metabolic problem. Eur J Heart Fail. 1999;1(2):127-31.
- 47. Berry C. Catabolism in chronic heart failure. Eur Heart J. 2000;21(7):521-32.
- Anker S, Rauchhaus M. Insights into the pathogenesis of chronic heart failure: immune activation and cachexia. Curr Opin Cardiol. 1999;14(3):211.
- 49. Anker S, Chua T, Ponikowski P. Hormonal changes and catabolic/anabolic imbalance in chronic heart failure and their importance for cardiac cachexia. Circulation. 1997;96(2):526–34.
- Poehlman ET, Scheffers J, Gottlieb SS, Fisher ML, Vaitekevicius P. Brief communications increased resting metabolic rate in patients with congestive heart failure. Ann Intern Med. 2012;121(11):860–2.
- 51. Freeman LM, Roubenoff R. The nutrition implications of cardiac cachexia. Nutr Rev. 2009;52(10):340-7.
- Anker SD, Ponikowski P, Varney S, Chua TP, Clark AL, Webb-Peploe KM, et al. Wasting as independent risk factor for mortality in chronic heart failure. Lancet. 1997;349(9058):1050–3.
- 53. von Haehling S, Lainscak M. Cardiac cachexia: a systematic overview. Pharmacol Ther. 2009;121(3):227-52.
- Murdoch DR, Rooney E, Dargie HJ, Shapiro D, Morton JJ, McMurray JJ. Inappropriately low plasma leptin concentration in the cachexia associated with chronic heart failure. Heart. 1999;82(3):352–6.
- Lund LH, Williams JJ, Freda P, LaManca JJ, LeJemtel TH, Mancini DM. Ghrelin resistance occurs in severe heart failure and resolves after heart transplantation. Eur J Heart Fail. 2009;11(8):789–94.
- 56. Strassburg S, Anker SD. Metabolic and immunologic derangements in cardiac cachexia: where to from here? Heart Fail Rev. 2006;11(1):57–64.
- Nagaya N, Moriya J, Yasumura Y, Uematsu M, Ono F, Shimizu W, et al. Effects of ghrelin administration on left ventricular function, exercise capacity, and muscle wasting in patients with chronic heart failure. Circulation. 2004;110(24):3674–9.
- Vila G, Grimm G, Resl M, Heinisch B, Einwallner E, Esterbauer H, et al. B-type natriuretic peptide modulates ghrelin, hunger, and satiety in healthy men. Diabetes. 2012;61(10):1–5.
- Zhao SP, Zeng LH. Elevated plasma levels of tumor necrosis factor in chronic heart failure with cachexia. Int J Cardiol. 1997;58(3):257–61.
- 60. Bobbert P, Jenke A, Bobbert T, Kühl U, Rauch U, Lassner D, et al. High leptin and resistin expression in chronic heart failure: adverse outcome in patients with dilated and inflammatory cardiomyopathy. Eur J Heart Fail. 2012;14(11):1265–75.
- Fukagawa N. Sarcopenia in aging humans: the impact of menopause and disease. J Gerontol A Biol Sci Med Sci. 1995;50A:73–7.
- 62. King D, Smith ML, Chapman TJ, Stockdale HR, Lye M. Fat malabsorption in elderly patients with cardiac cachexia. Age Ageing. 1996;25(2):144–9.
- King D, Smith ML, Lye M. Gastro-intestinal protein loss in elderly patients with cardiac cachexia. Age Ageing. 1996;25(3):221–3.
- Gomes MER, Mulder A, Bellersen L, Verheugt FWA, Smits P, Tack CJ. Alpha-receptor blockade improves muscle perfusion and glucose uptake in heart failure. Eur J Heart Failure [Internet]. 2010 [cited 2013 Sep 15];12(10):1061– 6. Available from: http://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=2980999&tool=pmcentrez&render type=abstract
- 65. Aquilani R, Opasich C, Verri M, Boschi F, Febo O, Pasini E, et al. Is nutritional intake adequate in chronic heart failure patients? J Am Coll Cardiol [Internet]. 2003 [cited 2013 Sep 15];42(7):1218–23. Available from: http:// linkinghub.elsevier.com/retrieve/pii/S073510970300946X
- 66. Anker SD, Negassa A, Coats AJS, Afzal R, Poole-Wilson PA, Cohn JN. Prognostic importance of weight loss in chronic heart failure and the effect of treatment with angiotensin-converting-enzyme inhibitors: an observational study. Lancet. 2003;361:1077–83.
- 67. Hryniewicz K, Androne ANAS, Hudaihed A, Katz SD. Partial reversal of cachexia by b-adrenergic receptor blocker therapy in patients with chronic heart failure. J Card Fail. 2003;9(6):464–8.
- Adigun A, Ajayi A. The effects of enalapril-digoxin-diuretic combination therapy on nutritional and anthropometric indices in chronic congestive heart failure: preliminary findings in cardiac cachexia. Eur J Heart Fail. 2001;3:359–63.
- 69. Mann DL, McMurray JJV, Packer M, Swedberg K, Borer JS, Colucci WS, et al. Targeted anticytokine therapy in patients with chronic heart failure: results of the Ranodimized Etanercept Worldwide Evaluation (RENEWAL). Circulation. 2004;109:1594–602.
- Broqvist M, Arnqvist H. Nutritional assessment and muscle energy metabolism in severe chronic congestive heart failure—effects of long-term dietary supplementation. Eur Heart J. 1994;15:1641–50.
- Heymsfield B, Casper K. Congestive continuous heart failure: clinical management by use of continuous nasoenteric feeding. Am J Clin Nutr. 1989;50(3):539–44.
- Paccagnella A, Calò MA, Caenaro G, Salandin V, Jus P, Simini G, et al. Cardiac cachexia: preoperative and postoperative nutrition management. JPEN J Parenter Enteral Nutr. 1994;18(5):409–16.
- Morrison W. Skeletal muscle and whole body protein turnover in cardiac cachexia: influence of branched-chain amino acid administration. Eur J Clin Invest. 1988;18(6):648–54.
- 74. Aquilani R, Viglio S, Iadarola P, Opasich C, Testa A, Dioguardi FS, et al. Oral amino acid supplements improve exercise capacities in elderly patients with chronic heart failure. Am J Cardiol. 2008;101(11A):104E–10.
- Tavazzi L, Maggioni A, Marchioli R. Effect of n-3 polyunsaturated fatty acids in patients with chronic heart failure (the GISSI-HF trial): a randomised, double-blind, placebo-controlled trial. Lancet. 2008;372(9645):1223–30.
- Garza CA, Pellikka P, Somers VK, et al. Structural and functional changes in left and right ventricles after major weight loss following bariatric surgery for morbid obesity. Am J Cardiol. 2010;105(4):550–6.
- Garza CA, Pellikka PA, Somers VK, Sarr MG, Seward JB, Collazo-Clavell ML, et al. Major weight loss prevents long-term left atrial enlargement in patients with morbid and extreme obesity. Eur J Echocardiogr. 2008;9(5): 587–93.
- Lin CH, Kurup S, Herrero P, Schechtman KB, Eagon JC, Klein S, et al. Myocardial oxygen consumption change predicts left ventricular relaxation improvement in obese humans after weight loss. Obesity (Silver Spring, MD). 2011;19(9):1804–12.
- Varli M, Turhan S, Aras S, Atli T, Erdogan G. Effects of weight loss on ventricular systolic and diastolic functions and left ventricular mass assessed by tissue doppler imaging in obese geriatric women: preliminary report. Aging Clin Exp Res. 2010;22(3):206–11.
- Horwich TB, Fonarow GC, Hamilton MA, MacLellan WR, Woo MA, Tillisch JH. The relationship between obesity and mortality in patients with heart failure. J Am Coll Cardiol. 2001;38(3):789–95.
- Casas-Vara A, Santolaria F, Fernández-Bereciartúa A, González-Reimers E, García-Ochoa A, Martínez-Riera A. The obesity paradox in elderly patients with heart failure: analysis of nutritional status. Nutrition. 2012;28(6): 616–22.
- Huynh B, Rovner A, Rich M. Long-term survival in elderly patients hospitalized for heart failure: 14-year followup from a prospective randomized trial. Arch Intern Med. 2006;166:1892–8.
- Konstam MA, Maggioni AP, Swedberg K, Udelson JE. Effects of oral tolvaptan in patients hospitalized for worsening heart failure The EVEREST outcome trial. JAMA. 2012;297(12):1319–31.
- 84. Ghali J, Kadakia S, Cooper R, Ferlinz J. Precipitating factors leading to decompensation of heart failure. Arch Intern Med. 1988;148:20–3.
- Vinson JM, Rich MW, Sperry JC, Shah A, McNamara T. Early readmission of elderly patients with congestive heart failure. J Am Geriatr Soc. 1991;39(10):1045–6.
- Witte KK, Clark AL. Micronutrients and their supplementation in chronic cardiac failure. An update beyond theoretical perspectives. Heart Fail Rev. 2006;11(1):65–74.
- Gao X, Peng L, Adhikari C, Lin J, Zuo Z. Spironolactone reduced arrhythmia and maintained magnesium homeostasis in patients with congestive heart failure. J Card Fail. 2007;13(3):170–7.
- Ramires FJ, Mansur A, Coelho O, Maranhão M, Gruppi CJ, Mady C, et al. Effect of spironolactone on ventricular arrhythmias in congestive heart failure secondary to idiopathic dilated or to ischemic cardiomyopathy. Am J Cardiol. 2000;85(10):1207–11.
- Witte KK, Clark AL, Cleland JG. Chronic heart failure and micronutrients. J Am Coll Cardiol. 2001;37(7): 1765–74.
- Kopp SJ, Klevay LM, Feliksik JM, Feliksik JM. Physiological and metabolic characterization of a cardiomyopathy induced by chronic copper deficiency. Am J Physiol. 1983;245:h855–6.
- 91. Lockitch G. Cardiomyopathy associated deficiency in a Caucasian with nonendemic. Am J Clin Nutr. 1990;52:572–7.
- 92. Coudray C, Boucher F, Richard MJ, Arnaud J, De Leiris J, Favier A. Zinc deficiency, ethanol, and myocardial ischemia affect lipoperoxidation in rats. Biol Trace Elem Res. 1991;30(2):103–18.
- Li Y, Huang T, Carlson E, Melov S. Dilated cardiomyopathy and neonatal lethality in mutant mice lacking manganese superoxide dismutase. Nat Genet. 1995;11(4):376–81.
- 94. Golik A, Cohen N, Ramot Y, Maor J, Moses R, Weissgarten J, et al. Type II diabetes mellitus, congestive heart failure, and zinc metabolism. Biol Trace Elem Res. 1993;39(2–3):171–5.
- 95. Anker SD, Comin Colet J, Filippatos G, Willenheimer R, Dickstein K, Drexler H, et al. Ferric carboxymaltose in patients with heart failure and iron deficiency. N Engl J Med. 2009;361(25):2436–48.
- 96. Liu P, Olivieri N. Iron overload cardiomyopathies: new insights into an old disease. Cardiovasc Drugs Ther. 1994;8(1):101–10.
- 97. Djoenaidi W, Notermans S, Dunda G. Beriberi cardiomyopathy. Eur J Clin Nutr. 1992;46(3):227-34.
- Shimon I, Almog S, Vered Z, Seligmann H, Shefi M, Peleg E, et al. Improved left ventricular function after thiamine supplementation in patients with congestive heart failure receiving long-term furosemide therapy. Am J Med. 1995;98(5):485–90.
- 99. Seligmann H, Halkin H. Thiamine deficiency in patients with congestive heart failure receiving long-term furosemide therapy: a pilot study. Am J Med. 1991;91(2):151–5.
- Zangen A, Botzer D, Zangen R, Shainberg A. Furosemide and digoxin inhibit thiamine uptake in cardiac cells. Eur J Pharmacol. 1998;361(1):151–5.

- 101. Ting HH, Timimi FK, Boles KS, Creager SJ, Ganz P, Creager MA. Vitamin C improves endothelium-dependent vasodilation in patients with non-insulin-dependent diabetes mellitus. J Clin Invest. 1996;97(1):22–8.
- 102. Gokce N, Keaney JF, Frei B, Holbrook M, Olesiak M, Zachariah BJ, et al. Long-term ascorbic acid administration reverses endothelial vasomotor dysfunction in patients with coronary artery disease. Circulation. 1999;99(25): 3234–40.
- 103. Hornig B, Arakawa N, Kohler C, Drexler H. Vitamin C improves endothelial function of conduit arteries in patients with chronic heart failure. Circulation. 1998;97(4):363–8.
- 104. Khaw K-T, Bingham S, Welch A, Luben R, Wareham N, Oakes S, et al. Relation between plasma ascorbic acid and mortality in men and women in EPIC-Norfolk prospective study: a prospective population study. Lancet. 2001;357(9257):657–63.
- Gale C, Martyn C, Winter P, Cooper C. Vitamin C and risk of death from stroke and coronary heart disease in cohort of elderly people. BMJ. 1995;310(6994):1563–6.
- Enstrom JE, Kanim LE, Klein MA. Vitamin C intake and mortality among a sample of the United States population. Epidemiology. 1992;3(3):194–202.
- Calzada C, Bruckdorfer KR, Rice-Evans CA. The influence of antioxidant nutrients on platelet function in healthy volunteers. Atherosclerosis. 1997;128(1):97–105.
- Stampfer M, Hennekens CH, Manson JE, Colditz G, Rosner B, Willett W. Vitamin E consumption and the risk of coronary disease in women. N Engl J Med. 1993;328(20):1444–9.
- 109. Losonczy K, Harris B, Havlik R. Vitamin E and vitamin C supplement use and risk of all-cause and coronary heart disease mortality in older persons: the established populations for epidemiologic studies of the elderly. Am J Clin Nutr. 1996;64:190–6.
- Kushi L, Folsom A. Dietary antioxidant vitamins and death from coronary heart disease in postmenopausal women. N Engl J Med. 1996;334(18):1156–62.
- 111. Eidelman RS, Hollar D, Hebert PR, Lamas GA, Hennekens CH. Randomized trials of vitamin E in the treatment and prevention of cardiovascular disease. Arch Intern Med. 2004;164(14):1552–6.
- 112. Miller ER, Pastor-Barriuso R, Dalal D, Riemersma RA. Meta-analysis: high-dosage vitamin E supplementation may increase all-cause mortality. Ann Intern Med. 2005;142(1):37–47.
- 113. Lonn E, Bosch J, Yusuf S, Sheridan P, Pogue J, Arnold JM, et al. Effects of long-term vitamin E supplementation on cardiovascular events and cancer. JAMA. 2005;293(11):1338–47.
- 114. Selhub J, Jacques PF, Wilson PW, Rush D, Rosenberg IH. Vitamin status and intake as primary determinants of homocysteinemia in an elderly population. JAMA. 1993;270(22):2693–8.
- 115. Bostom AG, Rosenberg IH, Silbershatz H, Jacques PF, Selhub J, D'Agostino RB, et al. Nonfasting plasma total homocysteine levels and stroke incidence in elderly persons: the Framingham Study. Ann Intern Med. 1999;131(5):352–5.
- Bots M, Launer L, Lindemans J. Homocysteine and short-term risk of myocardial infarction and stroke in the elderly: the Rotterdam Study. Arch Intern Med. 1999;159:38–44.
- 117. Eikelboom JW, Lonn E, Genest J, Hankey G, Yusuf S. Homocyst(e)ine and cardiovascular disease: a critical review of the epidemiologic evidence. Ann Intern Med. 1999;131(5):363–75.
- Herrmann M, Taban-Shomal O, Hübner U, Böhm M, Herrmann W. A review of homocysteine and heart failure. Eur J Heart Fail. 2006;8(6):571–6.
- Weishaar R, Simpson R. Involvement of vitamin D3 with cardiovascular function. II. Direct and indirect effects. Am J Physiol Endocrinol Metab. 1987;253:e675–86.
- 120. MacLaughlin J, Holick MF. Aging decreases the capacity of human skin to produce vitamin D3. J Clin Invest. 1985;76(4):1536–8.
- 121. Field MH. Hyperparathyroidism in congestive heart failure. Am J Med. 1999;107(1):102–3.
- Boxer R, Kenny A, Cheruvu V. Serum 25-hydroxyvitamin D concentration is associated with functional capacity in older adults with heart failure. Am Heart J. 2010;160(5):893–9.
- 123. Witham MD, Crighton LJ, Gillespie ND, Struthers AD, McMurdo MET. The effects of vitamin D supplementation on physical function and quality of life in older patients with heart failure: a randomized controlled trial. Circ Heart Fail. 2010;3(2):195–201.
- Boden W, Probstfield J, Anderson T. Niacin in patients with low HDL cholesterol levels receiving intensive statin therapy. N Engl J Med. 2011;365(24):2255–67.
- 125. Tavani A, Negri E, D'Avanzo B, La Vecchia C. Beta-carotene intake and risk of nonfatal acute myocardial infarction in women. Eur J Epidemiol. 1997;13(6):631–7.
- 126. Hennekens CH, Buring JE, Manson JE, Stampfer M, Rosner B, Cook NR, et al. Lack of effect of long-term supplementation with beta carotene on the incidence of malignant neoplasms and cardiovascular disease. N Engl J Med. 1996;334(18):1145–9.
- 127. McMurray JJV, Dunselman P, Wedel H, Cleland JG, Lindberg M, Hjalmarson A, et al. Coenzyme Q10, rosuvastatin, and clinical outcomes in heart failure: a pre-specified substudy of CORONA (controlled rosuvastatin multinational study in heart failure). J Am Coll Cardiol. 2010;56(15):1196–204.

- 128. De Pinieux G, Chariot P, Ammi-Saïd M, Louarn F, Lejonc JL, Astier A, et al. Lipid-lowering drugs and mitochondrial function: effects of HMG-CoA reductase inhibitors on serum ubiquinone and blood lactate/pyruvate ratio. Br J Clin Pharmacol. 1996;42(3):333–7.
- 129. Hofman-Bang C, Rehnqvist N, Swedberg K, Wiklund I, Aström H. Coenzyme Q10 as an adjunctive in the treatment of chronic congestive heart failure. The Q10 Study Group. J Card Fail. 1995;1(2):101–7.
- 130. Watson PS, Scalia GM, Galbraith A, Burstow DJ, Bett N, Aroney CN. Lack of effect of coenzyme Q on left ventricular function in patients with congestive heart failure. J Am Coll Cardiol. 1999;33(6):1549–52.
- 131. Weant KA, Smith KM. The role of coenzyme Q10 in heart failure. Ann Pharmacother. 2005;39(9):1522-6.
- 132. Khatta M, Alexander BS, Krichten CM, Fisher ML, Freudenberger R, Robinson SW, et al. The effect of coenzyme Q10 in patients with congestive heart failure. Ann Intern Med. 2000;132(8):636–40.
- 133. Belardinelli R, Muçaj A, Lacalaprice F, Solenghi M, Seddaiu G, Principi F, et al. Coenzyme Q10 and exercise training in chronic heart failure. Eur Heart J. 2006;27(22):2675–81.
- 134. Nawarskas JJ. HMG-CoA reductase inhibitors and coenzyme Q10. Cardiol Rev. 2005;13(2):76-9.
- 135. Anand I, Chandrashekhan Y, De Giuli F, Pasini E, Mazzoletti A, Confortini R, et al. Acute and chronic effects of propionyl-L-carnitine on the hemodynamics, exercise capacity, and hormones in patients with congestive heart failure. Cardiovasc Drug Ther. 1998;12(3):291–9.
- 136. Study on propionyl-L-carnitine in chronic heart failure. Eur Heart J. 1999;20(1):70-6.
- 137. Gordon A, Hultman E, Kaijser L, Kristjansson S, Rolf CJ, Nyquist O, et al. Creatine supplementation in chronic heart failure increases skeletal muscle creatine phosphate and muscle performance. Cardiovasc Res. 1995;30(3): 413–8.
- 138. Witte KKA, Nikitin NP, Parker AC, von Haehling S, Volk HD, Anker SD, et al. The effect of micronutrient supplementation on quality-of-life and left ventricular function in elderly patients with chronic heart failure. Eur Heart J. 2005;26(21):2238–44.
- 139. Soukoulis V, Dihu JB, Sole M, Anker SD, Cleland J, Fonarow GC, et al. Micronutrient deficiencies an unmet need in heart failure. J Am Coll Cardiol. 2009;54(18):1660–73.
- 140. Holdy K, Dembitsky W, Eaton LL, Chillcott S, Stahovich M, Rasmusson B, et al. Nutrition assessment and management of left ventricular assist device patients. J Heart Lung Transplant. 2005;24(10):1690–6.
- Otten JJ, Pitzi Hellwig J, Meyers LD. Dietary reference intakes essential guide nutrient requirements. Washington, DC: The National Academies Press; 2006.
- 142. Dinicolantonio JJ, DiPasquale P, Taylor RS, Hackam DG. Low sodium versus normal sodium diets in systolic heart failure: systematic review and meta-analysis. Heart. 2013;99(11):820.
- 143. Aguilar D, Skali H, Moyé LA, Lewis EF, Gaziano JM, Rutherford JD, et al. Alcohol consumption and prognosis in patients with left ventricular systolic dysfunction after a myocardial infarction. J Am Coll Cardiol. 2004;43(11): 2015–21.
- 144. Cooper HA, Exner DV, Domanski MJ. Light-to-moderate alcohol consumption and prognosis in patients with left ventricular systolic dysfunction. J Am Coll Cardiol. 2000;35(7):1753–9.
- Mostofsky E, Rice MS, Levitan EB, Mittleman MA. Habitual coffee consumption and risk of heart failure: a doseresponse meta-analysis. Circ Heart Fail. 2012;5(4):401–5.
- 146. Notarius CF, Morris B, Floras JS. Caffeine prolongs exercise duration in heart failure. J Card Fail. 2006;12(3): 220–6.

Chapter 14 The Relationship of Nutrition and Pressure Ulcers

David R. Thomas

Key Points

- A strong epidemiological association exists between nutritional status and the incidence, progression, and severity of pressure sores.
- The results of trials of prevention and treatment of pressure ulcers with nutritional interventions to date have been disappointing. While nutrient deficiencies are linked with poor wound healing, providing supplements to patients who are not deficient has not been shown to be of benefit for pressure ulcers.
- This paradoxical finding could be explained by a mechanism of weight loss occurring in a cycle of anorexia and cachexia. Cytokine-induced cachexia is remarkably resistant to hypercaloric feeding.
- Acknowledging these ambivalent findings, it is still important that general nutritional support be provided to persons with pressure ulcers, consistent with medical goals and patient wishes.

Keywords Wound healing • Under-nutrition • Nutritional supplements • Cytokines • Pressure ulcers

Introduction and Background

Wound healing is intricately linked to nutrition. Severe protein-calorie undernutrition in humans alters tissue regeneration, the inflammatory reaction, and immune function [1]. After vascular surgery, hypoalbuminemia and low serum transferrin levels predict wound healing complications [2]. Undernourished patients are more likely to have post-operative complications than well-nourished patients [3]. Although these markers do predict outcome, they do not correlate well with nutritional status [4].

Experimental studies in animal models suggest a biologically plausible relationship between undernutrition and development of pressure ulcers. When pressure was applied for 4 h to the skin of well-nourished animals and malnourished animals, pressure ulcers occurred equally in both groups. However, the degree of ischemic skin destruction was more severe in the malnourished animals. Epithelialization of the pressure lesions occurred in normal animals at 3 days post-injury, while

D. R. Thomas M.D., F.A.C.P., A.G.S.F.,

Division of Geriatric Medicine, Department of Internal Medicine, Saint Louis University Hospital, 1402 South Grand Blvd, M238, Saint Louis, MO 63104, USA e-mail: thomasdrstl@gmail.com

necrosis of the epidermis was still present in the malnourished animals [5]. These data suggest that while pressure damage may occur independently of nutritional status, malnourished animals may have impaired healing after a pressure injury.

Epidemiological Associations of Nutrition and Pressure Ulcers

A strong epidemiological association exists between nutritional status and the incidence, progression, and severity of pressure sores. In a prospective study of high-risk patients, undernutrition (defined by an index of biochemical and anthropometric variables) was present in 29 % of patients at hospital admission. At 4 weeks, 17 % of the undernourished patients had developed a pressure ulcer, compared to 9 % of the non-undernourished patients. Thus, patients who were undernourished at hospital admission were twice as likely to develop pressure ulcers as non-undernourished patients (RR 2.1, 95 % confidence intervals [CI] 1.1, 4.2) [6].

In a 2007 study of 4,067 patients from 22 hospitals in Germany, a positive relationship was shown between the presence of a pressure ulcer and each of the following variables: unintentional weight loss of 5–10 %, a body mass index less than 18.5 kg/m², poor nutritional intake (from the Braden Scale), and being bedfast. Being bedfast for hospitalized subjects was the most critical risk factor with an odds ratio of 23 (95 % CI 10, 52). In 2,393 patients from 29 German nursing homes, a positive association for the presence of a pressure ulcer was observed for each of the following variables: an unintentional weight loss of 5 to greater than 10 %, a body mass index less than 20 kg/m², poor nutritional intake (from the Braden Scale), and probable inadequate nutritional intake (from the Braden Scale) [7].

In a long-term-care setting, pressure ulcers developed in 65 % of residents who were diagnosed as severely undernourished upon admission [8]. In another long-term care setting, the estimated percent intake of dietary protein, but not total caloric intake, predicted development of pressure ulcers [9]. Impaired nutritional intake, defined as a persistently poor appetite, meals held due to gastrointestinal disease, or a prescribed diet less than 1,100 kcal or 50 g protein per day, predicted pressure ulcer development in another long-term care setting [10]. Table 14.1 demonstrates the association of serum albumin and other nutritional variables with the development of a pressure ulcer. Pressure ulcers appear to be associated with traditional markers of nutritional status in some, but not all, studies.

Nutrition in the Prevention of Pressure Ulcers

This strong association of undernutrition with the development of pressure ulcers led to the hypothesis that providing hypercaloric feeding to persons at risk for undernutrition might lead to the prevention of pressure ulcers. Several trials have examined this hypothesis (Table 14.2). However, the combined results of trials of nutritional intervention for the prevention of pressure ulcers have been disappointing [11, 12].

The effect of oral nutrition supplements was observed in a nonrandomized group of hospitalized, severely ill patients. Nutritional supplements were given to 33 % of subjects on one ward and to 87 % of subjects on another hospital ward. There was no difference between groups in pressure ulcer incidence (26.4 % vs. 20.2 %), pressure ulcer prevalence at discharge (14.7 % vs. 10.3 %), mortality (15.6 % vs. 14.2 %), length of stay (17.3 days vs. 17.4 days), or nosocomial infections (26.4 % vs. 19.0 %) [13].

A multi-center, cluster randomized trial in persons older than 65 years who were in the acute phase of a critical illness examined the effect of two oral supplements per day in addition to the normal hospital diet. At the end of 15 days, the supplemented group had a reduced incidence of all stages of pressure ulcers (40 % in the nutritional intervention group versus 48 % in the control group) (relative risk 0.83, 95 % confidence intervals 0.70–0.99) [14].

First author	Setting	Associated with presence of PU	Not associated with presence of PU
Allman [89]	AC	Albumin	Weight, hemoglobin, TLC, nutritional assessment
Gorse [1]	AC	Albumin	Nutritional assessment score
Inman [2]	AC, ICU	Albumin (measured at 3 days)	Serum protein, hemoglobin, weight
Allman [3]	AC	BMI, TLC	Albumin, TSF, arm circumference, weight loss, hemoglobin, nitrogen balance
Hartgrink [18]	AC, orthopedic		Nocturnal enteral feeding
Anthony [4]	AC	Albumin <32 g/L	
Moolten [5]	LTC	Albumin <35 g/L	
Pinchcofsky- Devin [8]	LTC	Severe malnutrition	Mild-to-moderate malnutrition or normal nutrition
Berlowitz [10]	LTC	Impaired nutritional intake	Albumin, serum protein, hemoglobin, TLC, BMI/weight
Bennett [6]	LTC		Weight, BMI, weight gain
Brandeis [7]	LTC	Dependency in feeding	BMI/weight, TSF
Trumbore [8]	LTC	Albumin, cholesterol	
Breslow [9]	LTC	Albumin, hemoglobin	Serum protein, cholesterol, zinc, copper, transferrin, body weight, BMI, TLC
Bergstrom [9]	LTC	Dietary protein intake 93 % of RDA vs. 119 %, dietary iron	Serum protein, cholesterol, zinc, copper, transferrin, weight, BMI, TLC
Ferrell [10]	LTC	-	Albumin, serum protein, BMI, hematocrit,
Bourdel- Marchasson [14]	LTC		Oral nutritional supplement (26 % vs. 20 % incidence)
Guralnik [11]	Community		Albumin, BMI, impaired nutrition, hemoglobin

Table 14.1 Epidemiological association of nutritional markers with development of a pressure ulcer

AC acute care, *LTC* long-term care, *BMI* body mass index, *TLC* total lymphocyte count, *TSF* triceps skinfold thickness, *ICU* intensive care unit, *RDA* Recommended Daily Allowance

Intervention First author Setting Outcome Delmi Hospitalized with One oral nutrition supplement per All-stage pressure ulcers 9 % in the 1990 [15] femoral neck day in addition to hospital diet nutritional intervention group vs. fracture vs. standard hospital diet alone 7 % in the control group RR 0.79 (0.14 - 4.39, p = 0.8)Stage 2 or greater pressure ulcers 52 % Hartgrink Hospitalized with hip Overnight nasogastric tube feeding 1998 [18] fracture and vs. standard hospital diet in the nutritional intervention group vs. 56 % in the control group increased pressure ulcer risk RR = 0.92 (0.64 - 1.32, p = 0.6)Bourdel-M Acute phase of a Two oral supplements per day in All-stage pressure ulcers 40 % in the 2000 [13] critical illness addition to normal diet vs. nutritional intervention group vs. standard hospital diet alone 48 % in the control group RR 0.83 (0.70 - 0.99)Houwing Hip-fracture patients One supplement daily in addition to Stage 1 and 2 pressure ulcers 55 % in the nutritional intervention group 2003 [16] the standard hospital diet vs. noncaloric water-based placebo vs. 59 % in the placebo group RR and standard hospital diet 0.92 (0.65-1.3)

 Table 14.2
 Nutritional interventions in the prevention of pressure ulcers

RR relative risk (95 % confidence intervals)

In hospitalized persons with femoral neck fracture, one group was randomized to receive one oral nutrition supplement in addition to the hospital diet or the standard hospital diet alone. There was no difference in the incidence of pressure ulcers between the two groups in this study [15].

In another population of persons with hip fracture, subjects were randomized to receive either 400 ml daily of a nutritional supplement enriched with protein, arginine, zinc, and antioxidants in addition to the standard hospital diet or the standard hospital diet and a noncaloric water-based placebo. There was no difference in the incidence of stage 2 or greater pressure ulcers between groups after 28 days (55 % in the supplement group vs. 59 % in the placebo group) [16].

Of these four trials of oral nutritional supplements for the prevention of pressure ulcers, only one suggests that a nutritional intervention may reduce the incidence of pressure ulcers. Similar results were found when another trial that used a group randomization was included in a meta-analysis [17]. The calculated number needed to treat in this analysis suggested that 20 patients would need to receive oral nutritional supplements to prevent one pressure ulcer.

In a trial of overnight enteral feeding in patients with hip fracture, no difference in pressure ulcer incidence, total serum protein, serum albumin, or the severity of pressure sores after 1 and 2 weeks was observed. After 2 weeks, 52 % of subjects in the enterally fed group and 56 % of the control group developed stage 2 or greater pressure ulcers (p=0.06). Of the 62 patients randomized for enteral feeding, only 25 tolerated their tube for more than 1 week, and only 16 tolerated their tube for 2 weeks. Comparison of the actually tube-fed group (n=25 at 1 week, n=16 at 2 weeks) and the control group showed two to three times higher protein and energy intake (p<0.0001), and a significantly higher total serum protein and serum albumin after 1 and 2 weeks in the actually tube-fed group (all p<0.001). In an intention to treat analysis there was also no difference in the incidence of sores of grade 2 or above [18]. It is possible that the lack of effect on supplemental enteral feeding was due to poor tolerance of the feedings.

A study of enteral tube feedings in patients with a pressure ulcer in a long-term care setting, observed 49 patients for 3 months [19]. Patients received 1.6 times basal energy expenditure daily, 1.4 g of protein per kilogram per day, and 85 % or more of their total recommended daily allowance. At the end of 3 months, there was no difference in number or in healing of pressure ulcers.

All of these clinical trials suffer from methodological problems, including weaknesses in study design and statistical power. Large, prospective, randomized controlled trials will be required to define the effect of nutritional interventions in prevention of pressure ulcers.

Nutrition in the Healing of Pressure Ulcers

Although correction of poor nutrition is part of total patient care and should be addressed in each patient, controversy exists about the ability of nutritional support to reduce wound complications or improve wound healing [20, 21].

Randomized, controlled trials have evaluated the effect of increased protein, vitamin C, zinc, and oral supplements in the treatment of pressure ulcers. One trial randomized by group found no difference in pressure ulcer healing between the nutritionally supplemented and control groups [17]. With the exception of a single trial of higher protein intake, no trial has demonstrated improved healing with the intervention. Table 14.3 summarizes the interventional trials for pressure ulcers.

First author	Setting	Intervention	Outcome
Breslow [27]	Long-term care	24 % protein vs. 14 % protein enteral feeding	-4.2 cm ² vs2.1 cm ² decrease in surface area
Chernoff [27]	Long-term care	1.8 g/kg protein vs. 1.2 g/kg protein enteral feeding	73 % vs. 43 % improvement in surface area
Henderson [19]	Long-term care	1.6 times basal energy expenditure,1.4 g of protein per kilogram per day	65 % PU at onset; 61 % prevalence at 3 months

Table 14.3 Nutritional interventions in the treatment of pressure ulcers

(continued)

First author	Setting	Intervention	Outcome
ter Riet [47] Taylor [48]	Long-term care Acute surgical patients	Vitamin C 10 mg vs. 1,000 mg Vitamin C large dose vs. none	No difference in healing rate 84 % vs. 43 % (control) reduction surface area at 30 days
Norris [56]	Acute hip fracture	Zinc	No difference
Lee [29]	Long-term care	Concentrated, fortified, collagen protein hydrolysate supplement, 15 g vs. placebo	Pressure Ulcer Scale for Healing Score (mean 3.55 intervention vs. 3.22 control)
Cerrada [1]	Long-term care	Oral = Diet plus 500 kcal, 34 g protein, 6 g arginine, vit C 500 mg, 18 mg zinc; Tube = 1,000 kcal, 55 g protein (20 %), 8 g arginine, vit C 380 mg, 20 mg zinc vs. standard 16 % protein	Reaching target of 30 kcal/kg/day did not affect wound healing. Controls had a greater increase in caloric intake (4.6 vs. 3.1 kcal/kg/ day). No benefit from increased protein intake No independent effect of arginine or zinc
Desneves [38]	Long-term care	Standard hospital diet vs. Standard diet plus two high-protein/energy supplements vs. Standard diet plus two high-protein/energy supplements containing additional arginine (9 g), vitamin C (500 mg) and zinc (30 mg)	Improved Pressure Ulcer Score Healing group 3 (9.4±1.2 vs. 2.6±0.6 at week 3). No significant changes in biochemical markers, oral dietary intake or weight in any group
Leigh [39]	Spinal cord injury	Standard hospital diet plus 4.5 vs. Standard diet plus 9 g arginine	No evidence of a difference in healing rate between the two arginine dosages
van Anholt [40]	Multisite, non-malnourished	200 mL three times daily of a high-energy supplement enriched with arginine, antioxidants, and other micronutrients (not specified) vs. a noncaloric placebo	Complete healing of 6 ulcers in the supplemented group vs. 5 ulcers in placebo group

Table 14.3 (continued)

RR relative risk (95 % confidence intervals)

General Nutritional Support for Persons with Pressure Ulcers

It is clear that nutritional deficiency in the form of starvation is associated with increased mortality and morbidity. Therefore, nutritional requirements must be addressed in every patient and corrected when possible, consistent with the wishes and plan of care for each individual (Table 14.4).

Energy

An important question relates to whether the measured resting energy expenditure (mREE) is higher in persons with pressure ulcers. A meta-analysis of five trials, the mREE was 23.7 ± 2.2 kcal/kg/day in persons with a pressure ulcer (n=92) versus 20.7 ± 0.8 kcal/kg/day in the controls without a pressure ulcer (n=101), suggesting a small but significant difference (p=0.0001) [22]. However, 43 % of the subjects with a pressure ulcer had a spinal cord injury, perhaps accounting for the difference.

In another study of 29 older hospitalized persons with a pressure ulcer, the measured resting metabolic rate did not differ from controls, and did not vary by ulcer size or severity [23]. The data suggest that a pressure ulcer may be associated with an increase in energy requirement, but the magnitude is small. Both of these two trials confirmed an estimated daily caloric requirement of 25–30 kcal/kg using the Harris-Benedict equation.

Hubic I iii Huandonar alerapy for pressure aleers		
30-35 kcal/kg/day		
1.2–1.5 g/kg/day		
Slight, if any, benefit		
No demonstrated benefit		
No demonstrated benefit		

Table 14.4 Nutritional therapy for pressure ulcers

The Harris-Benedict equation can be used to predict caloric requirements, but controversy exists over accuracy in obese or severely undernourished individuals [24]. Other formulas have been adjusted for severely stressed hospitalized subjects [25]. The use of prediction formulas achieve almost the same results for caloric requirements as bedside clinical estimates [26].

General estimates for daily caloric requirements range from 25 kcal/kg/day for sedentary adults to 40 kcal/kg/day for stressed adults. Stress generally includes persons with burns, pressure ulcers, cancer, infections, and other similar conditions. Therefore, caloric requirements can be met at 25–30 kcal/kg/day for elderly patients under moderate stress.

Protein

Greater healing of pressure ulcers has been reported with a higher protein intake irrespective of whether or not a positive nitrogen balance was observed. In 48 patients with stage 3 through 4 pressure ulcers who were being fed enterally, undernutrition was defined as a serum albumin below 35 g/L *or* a body weight more than 10 % below the midpoint of the age-specific weight range. Total truncal pressure ulcer surface area showed more decrease $(-4.2 \text{ cm}^2 \text{ vs}. -2.1 \text{ cm}^2)$ in surface area in patients fed the enteral formula containing 24 % protein compared to a formula containing 14 % protein. However, changes in body weight or in biochemical parameters of nutritional status did not differ between the two groups. The study was limited by a small sample size (only 28 patients completed the study), nonrandom assignment to treatment groups, confounding effects of air-fluidized beds, and the use of two different feeding routes [27].

In a small study of 12 enterally fed patients with pressure ulcers, the group who received 1.8 g/kg of protein had a 73 % improvement in pressure ulcer surface area compared to a 42 % improvement in surface area in the group receiving 1.2 g/kg of protein despite the fact that the group that received the higher protein level began the study with larger surface area pressure ulcers (22.6 cm² vs. 9.1 cm²) [28].

A concentrated, fortified, collagen protein hydrolysate supplement was evaluated in 44 subjects with 75 pressure ulcers, who were compared with 27 subjects with 33 pressure ulcers who received a noncaloric control. At 8 weeks there was a small but statistically significant improvement in the Pressure Ulcer Scale for Healing score for the protein-treated group (3.55 intervention vs. 3.22 control) [29]. The randomization process did not produce equal group size, and the number and severity of the pressure ulcers at baseline was not balanced between groups, potentially biasing the study.

In a long-term care setting, 93 residents with a wound were compared to 57 residents without a wound. Persons with a wound were started on an enteral tube feeding formula containing 1.25 g/ protein/day, and persons without a wound were started on an enteral tube feeding formula containing 1.0 g/protein/day [30]. At admission, only 12 % of persons with a wound, and 21 % of persons without a wound had normal prealbumin levels. Over 14 months, the amount of protein in the feeding formula was increased or maintained based on the serum prealbumin levels. The prealbumin normalized or increased by 8 points in 42 % of persons with a wound and in 46 % of persons without a wound. However, protein intake was not predictive for persons who improved their serum albumin versus those who did not. There was no correlation observed between the Pressure Ulcer Scale for

Healing (PUSH) score and improvement in serum prealbumin. These data suggest that increasing protein intake, in some cases to greater than 2 g/kg/day, is not associated with greater healing of pressure ulcers. It also indicates that serum prealbumin is a poor marker of nutritional status.

The optimum dietary protein intake in patients with pressure ulcers is unknown, but may be much higher than current adult recommendation of 0.8 g/kg/day. Current recommendations for dietary intake of protein in stressed elderly patients lies between 1.2 and 1.5 g/kg/day. Yet half of chronically ill elderly persons cannot maintain nitrogen balance at this level [31]. On the other hand, increasing protein intake beyond 1.5 g/kg/day may not increase protein synthesis and may cause dehydration [32]. Studies in critically ill adults have shown that a protein intake of 1.2 g/kg/day of normally hydrated weight or 1.0 g/kg/day of resuscitated weight was more effective than a lower protein intake in achieving complete protein sparing in persons with severe systemic inflammatory response. However, further protein sparing cannot be demonstrated at protein intakes greater than 1.5 g/kg/day [33, 34]. The optimum protein intake for these patients has not been defined, but may lie between 1.2 and 1.5 g/kg per day.

Amino Acids

The association of dietary protein intake with wound healing has led to investigation of the use of specific amino acids. Leucine seems to be important in severely ill patients. Glutamine is essential for the immune system function, but supplemental glutamine has not been shown to have noticeable effects on wound healing [35].

Arginine enhances wound collagen deposition in healthy volunteers [36]. However, no effect on healing of pressure ulcers has been observed with arginine supplementation [37]. Sixteen inpatients with a stage 2–4 pressure ulcer were randomized to a standard hospital diet compared to a standard diet plus two high-protein/energy supplements or a standard diet plus two high-protein/energy supplements containing 9 g of arginine, 500 mg of vitamin C, and 30 mg of zinc for 3 weeks. PUSH tool scores improved in both the control and arginine/vitamin C/zinc groups, but not in the protein/energy supplemented group. The major improvement in PUSH score was in the arginine/vitamin C/zinc group. The groups were not similar at baseline, and no changes in biochemical markers, oral dietary intake, or body weight was observed in any group [38].

In a study of 23 participants with 31 nonhealing wounds, 4.5 g supplemental arginine was compared to 9 g supplemental arginine and healing evaluated over 3 weeks. Most of the pressure ulcers were stage 2 (74 %), or stage 3 (19 %), with 7 % stage 4. No difference in healing rates between the two treatment groups was observed, suggesting no dose–response effect for arginine [39].

Forty-three non-malnourished subjects with stage 3 or 4 pressure ulcers were evaluated in a multisite, randomized, controlled, double-blind trial. Subjects were screened to exclude malnutrition, defined as a body mass index less than 18.5 kg/m² for those younger than 70 years or 21 kg/m² for those older than 70 years. Other exclusion criteria were severe medical conditions, life expectancy shorter than 6 months, receiving palliative care, use of corticosteroids, and/or dietary restrictions, such as a protein-restricted diet. Subjects were offered 200 mL of a high energy supplement enriched with arginine, antioxidants, and other micronutrients (not specified) three times daily for a maximum of 8 weeks, or a noncaloric placebo, similar in taste and appearance, over the same timeframe. In the supplemented group, the mean healing rate measured by wound size was 0.26 cm²/day compared to 0.14 cm²/day in the control group over the first 3 weeks. By 8 weeks, the mean healing rate in the supplemented group was similar to the control group (0.16 cm²/day vs. 0.15 cm²/day, respectively). In this population screened to exclude undernutrition, complete healing was observed in six ulcers in the supplemented group, compared to five ulcers in the control group by 8 weeks. No change was observed in body mass index [40].

Twenty-eight subjects older than 65 years with a stage 3-4 pressure ulcer that was present for less than 1 month were randomized to receive a standard hospital diet with no supplement (n=15), or a standard hospital diet plus a supplement containing 500 kcal with 34 g protein, 6 g arginine, 500 mg vitamin C, and 18 mg zinc (n=13). Nine subjects who were tube fed in the treatment group (69 %) received a formula containing 1,000 kcal, 55 g protein (20 %), 8 g arginine, 380 mg vitamin C, and 20 mg zinc. Nine subjects who were tube-fed in the control group (60 %) received an enteral formula containing 16 % protein. All subjects were adjusted to a target of 30 kcal/kg/day. After 12 weeks, complete pressure ulcer healing was documented for only one person in the treatment group. The PUSH score between groups was different only at week 12. There was no difference in ulcer area measured in square millimeters at any time point. The percentage of decrease in pressure ulcer area in the treatment group was greater compared to the control group at week 8 and week 12. No nutritional parameter was different between groups except for the treatment group having a higher zinc level. Surprisingly, reaching a target of 30 kcal/kg/day did not affect wound healing, and no benefit from increased protein intake or independent effect of arginine or zinc was observed. These findings call into question empirically derived recommendations. In addition, no effect on wound healing measured by change in PUSH score or change in ulcer area was observed despite an increase in energy intake and protein intake over a 12-week period (controls 4.6 kcal/kg/day vs. treatment 3.1 kcal/kg/ day). These data suggest that a specific nutritional formula may produce small benefits in improving pressure ulcer size. However, no specific supplemental nutritional component could account for the variation [41]. Additionally, based on a single trial to date, no improvement in wound healing has been demonstrated in response to high supplements of branched-chain amino acid formulations [42].

Vitamins and Minerals

The deficiency of several vitamins has significant effects on wound healing. However, supplementation of vitamins to accelerate wound healing in the absence of a deficiency state is controversial. Vitamin C is essential for wound healing and impaired wound healing has been observed in clinical scurvy. However, in studies of clinically impaired wound healing, 6 months of an ascorbate-free diet is required to produce a deficient state [43]. In animals that are vitamin C–deficient, wound healing is abnormal at 7 days but completely normal at 14 days [44].

There is no evidence of acceleration of wound healing by vitamin C supplementation in patients who are not vitamin C-deficient [45]. Supertherapeutic doses of vitamin C have not been shown to accelerate wound healing [46]. Two clinical trials have evaluated the effect of supplemental vitamin C in the treatment of pressure ulcers. In a multicenter, blinded trial, 88 patients with pressure ulcers were randomized to either 10 or 500 mg twice daily of vitamin C. The wound closure rate, relative healing rate, and wound improvement score were not different between groups [47]. An earlier trial in acute surgical patients with pressure ulcers found a mean reduction in surface area at 1 month of 84 % in patients treated with large doses of vitamin C compared to a reduction in surface area of 43 % in the control group (p<0.005) [48]. The recommended daily allowance (RDA) of vitamin C is 60 mg. This RDA is easily achieved from dietary sources that include citrus fruits, green vegetables, peppers, tomatoes, and potatoes.

Vitamin A deficiency results in delayed wound healing and increased susceptibility to infection [49]. Vitamin A has been shown to be effective in counteracting delayed healing in patients on corticosteroids [50]. Vitamin E deficiency does not appear to play an active role in wound healing [51].

Zinc was first implicated in delayed wound healing in 1967 [52]. No study to date has shown improved wound healing in patients supplemented with zinc who were not zinc-deficient [53, 54]. Zinc levels have not been associated with development of pressure ulcers in patients with femoral neck fractures [55]. In a small study of patients with pressure ulcers, no effect on ulcer healing was

seen at 12 weeks in zinc-supplemented versus non-zinc-supplemented patients [56]. Indiscriminate or long-term zinc supplementation should be avoided since high serum zinc levels may inhibit healing, impair phagocytosis, and interfere with copper metabolism [11, 57]. The RDA for zinc is 12–15 mg. but most elderly persons intake 7–11 mg of zinc per day [58], chiefly from meats and cereal.

Mode of Delivery

Enteral tube feeding is often considered for persons with poor oral intake. An observational study of persons (N=1,124) referred to a hospital for insertion of a percutaneous endoscopic gastrostomy (PEG) tube, but who had no pressure ulcer, found that these nursing home residents were twice as likely to develop a new pressure ulcer (OR 2.27, 95 % CI, 1.95–2.65) compared to matched controls who had no PEG tube. Similarly, those persons (N=461) with a prevalent pressure ulcer who had a PEG feeding tube inserted were less likely to have the ulcer heal (OR 0.70, 95 % CI, 0.55–0.89) compared to matched controls [59].

Factors Contributing to the Nutritional Paradox

Traditionally, weight loss and undernutrition have been classified as due to either a relative lack of dietary protein (kwashiokor) or lack of both dietary protein and calories (marasmus) [60]. This classification system focused clinical attention on a lack of adequate food. Simply put, weight loss in older persons was attributed to starvation. Measures of serum proteins or anthropormorphological parameters were thought to detect early starvation. Remedial nutritional strategies were aimed at increasing voluntary food intake, prescribing hypercaloric supplements, or by instituting parenteral or enteral feeding [61]. In general, these interventions have had only a modest effect in reversing weight loss [62, 63].

Recent data suggest that the major cause of weight loss in older adults is the anorexia/cachexia syndrome [64]. This has resulted in a new paradigm for defining the mechanism of weight loss in older persons. Current understanding of undernutrition defines three categories: (1) pure chronic starvation without inflammation, (2) acute disease or major injury with inflammation, and (3) chronic diseases or conditions that impose sustained inflammation of a mild to moderate degree [65]. One of the chief distinguishing factors is that starvation without inflammation is amenable to hypercaloric feeding in all but the terminally undernourished patients, while acute and chronic inflammatory conditions are remarkably resistant to hypercaloric feeding [66, 67].

Cachexia is directly related to inflammatory states, such as cancer or acquired immunodeficiency syndrome, and also occurs in other common conditions such as rheumatoid arthritis, chronic renal insufficiency, chronic obstructive pulmonary disease, ischemic cardiomyopathy, and infectious diseases [68]. Interleukin (IL)-1 concentrations are elevated in elderly patients with severe undernutrition of unknown etiology, [69] and levels of IL-1 β and IL-6 can be increased in elderly persons without evidence of infection or cancer [70]. Tumor necrosis factor- α (TNF) levels are elevated in patients with severe undernutrition and congestive heart failure, but not in patients with congestive heart failure who do not have severe undernutrition [71]. Severe undernutrition occurs in both chronic infections and neoplastic disorders, suggesting that severe undernutrition develops along a common pathway and is not dependent on a specific infection or a particular neoplasm.

Measures of serum proteins are increasingly seen as acute phase reactants reflecting underlying inflammation rather than measures of starvation. Physiological stress (such as surgical operations), cortisol excess, and hypermetabolic states reduce serum albumin even in the presence of adequate

protein intake. Decreases in serum albumin may reflect the presence of inflammatory cytokine production or comorbidity rather than nutritional status [72]. Soluble IL-2 receptors are negatively associated with albumin, prealbumin, cholesterol, transferrin, and hemoglobin. The use of albumin and cholesterol in these patients as nutritional markers could potentially lead to over diagnosis of malnutrition [73]. This may explain why serum albumin has not consistently been an independent predictor of pressure ulcers.

Several cytokines, particularly IL-1 α , IL-1 β , and IL-6, have been suggested to be elevated in subjects with pressure ulcers. Whether these levels change with healing, or are predictive of healing is not known. These cytokines are known to also increase in severe undernutrition.

Serum IL-1 β is elevated in patients with pressure ulcers [74]. Levels of IL-1 α are elevated in pressure ulcers but low in acute wound fluid [75]. In hospitalized elderly patients suffering from bacterial pneumonia, cerebrovascular disease, or femoral bone fracture, serum IL-1 beta (but not IL-6) was higher in subjects with pressure ulcers. Albumin, hemoglobin, C-reactive protein, fibrinogen, and white cell count were also lower in subjects with pressure ulcers, despite no significant differences in age, gender, Braden scale, or underlying diseases between the two groups [75].

Circulating serum levels of IL-6, IL-2, and IL-2R are higher in spinal cord injured patients compared to normal controls, and highest in subjects with pressure ulcers. The highest concentration of cytokines were in subjects with the slowest healing pressure ulcers [76]. In other studies, IL-6 serum levels were increased in patients with pressure ulcers but IL-1 and TNF were not elevated [77].

Existing studies are not clear as to whether the elevation of inflammatory cytokines is due to the presence of a pressure ulcer or due to underlying severe undernutrition. Alternatively, the elevation of cytokine levels may be a common pathway for both conditions. The hypotheses are demonstrated in Fig. 14.1. Cytokine-mediated anorexia and weight loss are common in the population that develops pressure ulcers. The interrelationship is outlined in Table 14.5.



Fig. 14.1 Inflammatory disease may initiate the proinflammatory cytokine cascade, leading directly to development of a pressure ulcer. Alternatively, the development of a pressure ulcer may initiate the cytokine cascade, leading to undernutrition through suppression of appetite and cachexia. Undernutrition may lead to development of a pressure ulcer through either cachexia and loss of body mass or through the cytokine cascade

Undernutrition		
Poor wound healing		
Increased risk of infection		
Increased incidence of pressure ulcers		
Proinflammatory cytokines		
Suppress appetite		
Promote/interfere with wound healing		
Chronic wounds		
Source of cytokines		
Increased association with undernutrition		
Increased serum levels of cytokines		

Table 14.5 Documented relations among cytokines, undernutrition, and chronic wounds

Interventions for Inflammatory-Mediated Cachexia

This new paradigm suggests that interventions targeting weight loss in older persons must address the anorexia/cachexia continuum. Assessment of appetite can suggest anorexia as a cause of decreased food intake [78]. A common cause may be loss of appetite, due to dysregulation of a variety of psychological, gastrointestinal, metabolic, and nutritional factors [79]. Loss of appetite may initiate a vicious cycle of weight loss and increasing undernutrition.

Cytokines may regulate appetite directly through the central feeding drive. Significant interaction between the central feeding drive, neuropeptide Y, and IL-1 β has been demonstrated in rats [80, 81]. IL-1, IL-6, TNF, interferon- γ , leukemia inhibitory factor (D-factor), and prostaglandin E2 have all been implicated in cancer-induced severe undernutrition [82, 83]. Leptin, a central regulator of food intake and body fat mass, increases under the stress of hip operations [84] but is low in undernour-ished men [85].

The lack of effect of hypercaloric feeding in pressure ulcers may reflect that the underlying pathophysiology is cytokine-induced cachexia rather than simple starvation. Starvation is amenable to hypercaloric feeding in all but the terminally undernourished patients. Cytokine-induced cachexia is remarkably resistant to hypercaloric feeding [66, 67]. This may explain the modest results of clinical nutrition intervention trials in older persons [86].

Where possible, the underlying inflammatory condition should be sought. The importance of defining the distinction lies in developing a targeted therapeutic approach to weight loss in older persons [87]. Failure to distinguish among these causes of weight loss often results in frustration over the clinical response to therapeutic interventions.

Interventions to modulate cytokine activity are possible. Cytokine modulation has been postulated as a potential treatment for cachexia [88–94]. If a significant positive relationship exists between circulating cytokines and pressure ulcers, an opportunity for potential intervention to promote healing exists.

Conclusion

Wound nutrition is whole body nutrition. Unquestionably, providing nutritional support can prevent the effects of starvation. Death is an inevitable consequence of starvation. Whether or not nutrition can improve the outcome of pressure ulcers remains in dispute. Improvements in nutritional markers, such as serum protein concentrations, nitrogen balance, and weight gain, have not usually been accompanied by clinical benefits. There is no doubt that undernutrition is not good for wound healing. However, there is no magic nutritional bullet that will accelerate wound healing. General nutritional support should be provided to persons with pressure ulcers, consistent with medical goals and patient wishes.

Clinical Recommendations

- 1. Provide optimum nutrition consistent with goals of care for each patient.
- 2. Optimize protein intake with a goal of 1.2–1.5 g/kg/day of protein.
- 3. A simple multivitamin supplement may be indicated for nutritionally compromised patients, but there is no data to support the routine use of vitamin C and zinc in patients with pressure ulcers.
- 4. Consider vitamin A supplements in patients on corticosteriods.
- 5. Consider whether a cytokine-associated inflammatory condition may be present and potentially treatable.

References

- 1. Young ME. Malnutrition and wound healing. Heart Lung. 1988;17:60-7.
- Casey J, Flinn WR, Yao JST, Fahey V, Pawlowski J, Bergan JJ. Correlation of immune and nutritional status with wound complications in patients undergoing vascular operations. Surgery. 1983;93:822–7.
- Detsky AS, Baker JP, O'Rourke K, Johnston N, Whitwell J, Mendelson RA, et al. Predicting nutrition-associated complications for patients undergoing gastrointestinal surgery. JPEN. 1987;11:440–6.
- Covinsky KE, Covinsky MH, Palmer RM, Sehgal AR. Serum albumin concentration and clinical assessments of nutritional status in hospitalized older people: different sides of different coins? J Am Geriatr Soc. 2002;50:631–7.
- Takeda T, Koyama T, Izawa Y, Makita T, Nakamura N. Effects of malnutrition on development of experimental pressure sores. J Dermatol. 1992;19:602–9.
- Thomas DR, Goode PS, Tarquine PH, Allman R. Hospital acquired pressure ulcers and risk of Death. J Am Geriatr Soc. 1996;44:1435–40.
- Shahin ES, Meijers JM, Schols JM, Tannen A, Halfens RJ, Dassen T. The relationship between malnutrition parameters and pressure ulcers in hospitals and nursing homes. Nutrition. 2010;26(9):886–9.
- Pinchcofsky-Devin GD, Kaminski Jr MV. Correlation of pressure sores and nutritional status. J Am Geriatr Soc. 1986;34:435–40.
- Bergstrom N, Braden B. A prospective study of pressure sore risk among institutionalized elderly. J Am Geriatr Soc. 1992;40:747–58.
- Berlowitz DR, Wilking SV. Risk factors for pressure sores: a comparison of cross-sectional and cohort-derived data. J Am Geriatr Soc. 1989;37:1043–50.
- 11. Thomas DR. The role of nutrition in prevention and healing of pressure ulcers. Clin Geriatr Med. 1997;13:497–511.
- 12. Thomas DR. Improving the outcome of pressure ulcers with nutritional intervention: a review of the evidence. Nutrition. 2001;17:121–5.
- Bourdel-Marchasson I, Barateau M, Sourgen C, Pinganaud G, Salle-Montaudon N, Richard-Harston S, et al. Prospective audits of quality of PEM recognition and nutritional support in critically ill elderly patients. Clin Nutr. 1999;18:233–40.
- 14. Bourdel-Marchasson I, Barateau M, Rondeau V, Dequae-Merchadou L, Salles-Montaudon N, Emeriau JP, et al. A multi-center trial of the effects of oral nutritional supplementation in critically ill older inpatients. GAGE Group. Groupe Aquitain Geriatrique d'Evaluation. Nutrition. 2000;16:1–5.
- Delmi M, Rapin CH, Bengoa JM, Delmas PD, Vasey H, Bonjour JP. Dietary supplementation in elderly patients with fractured neck of the femur. Lancet. 1990;335(8696):1013–6.
- Houwing RH, Rozendaal M, Wouters-Wesseling W, Beulens JW, Buskens E, Haalboom JR. A randomised, doubleblind assessment of the effect of nutritional supplementation on the prevention of pressure ulcers in hip-fracture patients. Nutrition. 2003;22(4):401–5.

- 17. Stratton RJ, Ek AC, Engfer M, Moore Z, Rigby P, Wolfe R, Elia M. Enteral nutritional support in prevention and treatment of pressure ulcers: a systematic review and meta-analysis. Ageing Res Rev. 2005;4:422–50.
- Hartgrink HH, Wille J, Konig P, Breslau PJ. Pressure sores and tube feeding in patients with a fracture of the hip: a randomized clinical trial. Clin Nutr. 1998;6:287–92.
- Henderson CT, Trumbore LS, Mobarhan S, Benya R, Miles TP. Prolonged tube feeding in long-term care: nutritional status and clinical outcomes. J Am Coll Clin Nutr. 1992;11:309.
- 20. Albina JE. Nutrition and wound healing. JPEN. 1994;18:367-76.
- 21. Thomas DR. Issues and dilemmas in managing pressure ulcers. J Gerontol Med Sci. 2001;56:M238-340.
- 22. Cereda E, Klersy C, Rondanelli M, Caccialanza R. Energy balance in patients with pressure ulcers: a systematic review and meta-analysis of observational studies. J Am Diet Assoc. 2011;111(12):1868–76.
- Dambach B, Salle A, Marteau C, Mouzet JB, Ghali A, Favreau AM, et al. Energy requirements are not greater in elderly patients suffering from pressure ulcers. J Am Geriatr Soc. 2005;53(3):478–82.
- 24. Choban PS, Burge JC, Flanobaum L. Nutrition support of obese hospitalized patients. Nutr Clin Pract. 1997;12:149–54.
- 25. Ireton-Jones CS. Evaluation of energy expenditures in obese patients. Nutr Clin Pract. 1989;4:127-9.
- 26. Food and Nutrition Board, Institute of Medicine. Dietary reference intakes for energy, carbohydrates, fiber, fat, protein and amino acids (macronutrients). In: Energy. Washington, DC: The National Academies Press, 2002, pp. 93–206.
- Breslow RA, Hallfrisch J, Guy DG, Crawley B, Goldberg AP. The importance of dietary protein in healing pressure ulcers. J Am Geriatr Soc. 1993;41:357–62.
- Chernoff RS, Milton KY, Lipschitz DA. The effect of very high-protein liquid formula (Replete) on decubitus ulcer healing in long-term tube-fed institutionalized patients. Investigators Final Report 1990. J Am Diet Assoc. 1990;90(9):A–130.
- Lee SK, Posthauer ME, Dorner B, Redovian V, Maloney MJ. Pressure ulcer healing with a concentrated, fortified, collagen protein hydrolysate supplement: a randomized controlled trial. Adv Skin Wound Care. 2006;19(2):92–6.
- Pompeo M. Misconceptions about protein requirements for wound healing: results of a prospective study. Ostom Wound Manag. 2007;53(8):30–2. 34, 36–38.
- Gersovitz M, Motil K, Munro HN, Scrimshaw A. Human protein requirements: assessment of the adequacy of the current Recommended Dietary Allowance for dietary protein in elderly men and women. Am J Clin Nutr. 1982;35:6–14.
- Long CL, Nelson KM, Akin Jr JM, Geiger JW, Merrick HW, Blakemore WZ. A physiologic bases for the provision of fuel mixtures in normal and stressed patients. J Trauma. 1990;30:1077–86.
- Wolfe RR, Goodenough RD, Burke JF, Wolfe MH. Response of protein and urea kinetics in burn patients to different levels of protein intake. Ann Surg. 1983;197:163–71.
- Ishibashi N, Plank LD, Sando K, Hill GL. Optimal protein requirements during the first 2 weeks after the onset of critical illness. Crit Care Med. 1998;26:1529–35.
- 35. McCauley R, Platell C, Hall J, McCulloch R. Effects of glutamine on colonic strength anastomosis in the rat. J Parenter Enter Nutr. 1991;116:821.
- Barbul A, Lazarous S, Efron DT, Wasserkrug HL, Efron G. Arginine enhances wound healing in humans. Surgery. 1990;108:331–7.
- Langkamp-Henken B, Herrlinger-Garcia KA, Stechmiller JK, Nickerson-Troy JA, Lewis B, Moffatt L. Arginine supplementation is well tolerated but does not enhance mitogen-induced lymphocyte proliferation in elderly nursing home residents with pressure ulcers. J Parenter Enteral Nutr. 2000;24(5):280–7.
- Desneves KJ, Todorovic BE, Cassar A, Crowe TC. Treatment with supplementary arginine, vitamin C and zinc in patients with pressure ulcers: a randomised controlled trial. Clin Nutr. 2005;24:979–87.
- Leigh B, Desneves K, Rafferty J, Pearce L, King S, Woodward MC, et al. The effect of different doses of an arginine-containing supplement on the healing of pressure ulcers. J Wound Care. 2012;21(3):150–6.
- 40. van Anholt RD, Sobotka L, Meijer EP, Heyman H, Groen HW, Topinkova E, et al. Specific nutritional support accelerates pressure ulcer healing and reduces wound care intensity in non-malnourished patients. Nutrition. 2010;26(9):867–72.
- Cereda E, Gini A, Pedrolli C, Vanotti A. Disease-specific, versus standard, nutritional support for the treatment of pressure ulcers in institutionalized older adults: a randomized controlled trial. J Am Geriatr Soc. 2009;57(8):1395–402.
- McCauley C, Platell C, Hall J, McCullock R. Influence of branched chain amino acid solutions on wound healing. Aust NZ J Surg. 1990;60:471.
- 43. Crandon JH, Lind CC, Dill DB. Experimental human scurvy. N Engl J Med. 1940;223:353.
- 44. Levenson SM, Upjohn HL, Preston JA. Effect of thermal burns on wound healing. Ann Surg. 1957;146:357-68.
- Rackett SC, Rothe MJ, Grant-Kels JM. Diet and dermatology. The role of dietary manipulation in the prevention and treatment of cutaneous disorders. J Am Acad Dermatol. 1993;29:447–61.
- 46. Vilter RW. Nutritional aspects of ascorbic acid: uses and abuses. West J Med. 1980;133:485.

- ter Riet G, Kessels AG, Knipschild PG. Randomized clinical trial of ascorbic acid in the treatment of pressure ulcers. J Clin Epidemiol. 1995;48:1453–60.
- Taylor TV, Rimmer S, Day B, Butcher J, Dymock IW. Ascorbic acid supplementation in the treatment of pressure sores. Lancet. 1974;2:544–6.
- 49. Hunt TK. Vitamin A, and wound healing. J Am Acad Dermatol. 1986;15:817-21.
- 50. Ehrlich HP, Hunt TK. Effects of cortisone and vitamin A on wound healing. Ann Surg. 1968;167:324.
- 51. Waldorf H, Fewkes J. Wound healing. Adv Dermatol. 1995;10:77-96.
- 52. Pories WJ, Henzel WH, Rob CG, Strain WH. Acceleration of healing with zinc sulfate. Ann Surg. 1967;165:423.
- 53. Hallbrook T, Lanner E. Serum zinc and healing of leg ulcers. Lancet. 1972;2:780.
- Sandstead SH, Henrikson LK, Greger JL. Zinc nutriture in the elderly in relation to taste acuity, immune response, and wound healing. Am J Clin Nutr. 1982;36(Supp):1046.
- Goode HF, Burns E, Walker BE. Vitamin C depletion and pressure ulcers in elderly patients with femoral neck fracture. Br Med J. 1992;305:925–7.
- 56. Norris JR, Reynolds RE. The effect of oral zinc sulfate therapy on decubitus ulcers. JAGS. 1971;19:793.
- Reed BR, Clark RAF. Cutaneous tissue repair: practical implications of current knowledge: II. J Am Acad Dermatol. 1985;13:919–41.
- 58. Gregger JL. Potential for trace mineral deficiencies and toxicities in the elderly. In: Bales CW, editor. Mineral homeostasis in the elderly. New York, NY: Marcel Dekker; 1989. p. 171–200.
- Teno JM, Gozalo P, Mitchell SL, Kuo S, Fulton AT, Mor V. Feeding tubes and the prevention or healing of pressure ulcers. Arch Intern Med. 2012;172(9):697–701.
- 60. Thomas DR. Distinguishing starvation from cachexia. Geriatr Clin North Am. 2002;18:883–92.
- 61. Thomas DR. Weight loss in older adults. Rev Endocrinol Metab. 2005;6:129-36.
- 62. Haddad RY, Thomas DR. Enteral nutrition and tube feeding: a review of the evidence. Geriatr Clin North Am. 2002;18:867–82.
- 63. Thomas DR. A prospective, randomized clinical study of adjunctive peripheral parenteral nutrition in adult subacute care patients. J Nutr Health Aging. 2005;9:321–5.
- 64. Morley JE, Thomas DR, Wilson MM. Cachexia: pathophysiology and clinical relevance. Am J Clin Nutr. 2006;83(4):735–43.
- 65. Jensen G. Adult starvation and disease-related malnutrition: a proposal for etiology-based diagnosis in the clinical practice setting from the International Consensus Guideline Committee. Parenter Enteral Nutr. 2010;34:156–9.
- 66. Souba WW. Drug therapy: nutritional support. N Engl J Med. 1997;336:41-8.
- Atkinson S, Sieffert E, Bihari D. A prospective, randomized, double-blind, controlled clinical trial of enteral immunonutrition in the critically ill. Crit Care Med. 1998;26:1164–72.
- 68. Thomas DR. Loss of skeletal muscle mass in aging: examining the relationship of starvation, sarcopenia and cachexia. Clin Nutr. 2007;26(4):389–99.
- Liso Z, Tu JH, Small CB, Schnipper SM, Rosenstreich DL. Increased urine IL-1 levels in aging. Gerontology. 1993;39:19–27.
- Cederholm T, Whetline B, Hollstrom K, Andersson B, Engstrom L, Brismar K, et al. Enhanced generation of Interleukin 1β and 6 may contribute to the cachexia of chronic disease. Am J Clin Nutr. 1997;65:876–82.
- Ikeda U, Yamamoto K, Akazawa H, Hojo Y, Ohkawa F, Fujikawa H, et al. Plasma cytokine levels in cardiac chambers of patients with mitral stenosis with congestive heart failure. Cardiology. 1996;87:476–8.
- Friedman FJ, Campbell AJ, Caradoc-Davies TH. Hypoalbuminemia in the elderly is due to disease not malnutrition. Clin Exp Gerontol. 1985;7:191–203.
- Rosenthal AJ, Sanders KM, McMurtry CT, Jacobs MA, Thompson DD, Gheorghiu D, et al. Is malnutrition overdiagnosed in older hospitalized patients? Association between the soluble interleukin-2 receptor and serum markers of malnutrition. J Gerontol A Biol Sci Med Sci. 1998;53:M81–6.
- Matsuyama N, Takano K, Mashiko T, Jimbo S, Shimetani N, Ohtani H. The possibility of acute inflammatory reaction affects the development of pressure ulcers in bedridden elderly patients. Rinsho Byori Jpn J Clin Pathol. 1999;47(11):1039–45.
- 75. Barone EJ, Yager DR, Pozez AL, Olutoye OO, Crossland MC, Diegelmann RF, et al. Interleukin-1α and collagenase activity are elevated in chronic wounds. Plast Reconstr Surg. 1998;102:1023–7.
- 76. Segal JL, Gonzales E, Yousefi S, Jamshidipour L, Brunnemann SR. Circulating levels of IL-2R, ICAM-1, and IL-6 in spinal cord injuries. Arch Physical Med Rehab. 1997;78:44–7.
- Bonnefoy M, Coulon L, Bienvenu J, Boisson RC, Rys L. Implication of Cytokines in the aggravation of malnutrition and hypercatabolism in elderly patients with severe pressure sores. Age Ageing. 1995;24:37–42.
- Wilson MM, Thomas DR, Rubenstein LZ, Chibnall JT, Anderson S, Baxi A, et al. Appetite assessment: simple appetite questionnaire predicts weight loss in community-dwelling adults and nursing home residents. Am J Clin Nutr. 2005;82(5):1074–81.
- 79. Morley JE, Thomas DR. Anorexia and aging: pathophysiology. Nutrition. 1999;15:499–503.

- Chance WT, Balasubramahiam A, Dayal R, Brown J, Fischer JE. Hypothalamic concentration and release of neuropeptide Y into microdialysis is reduced in anorectic tumor bearing rates. Life Sci. 1994;54:1869–74.
- Leibowitz SF. Neurochemical-neuroendocrine systems in the brain controlling macronutirent intake and metabolism. Trends Neurosci. 1992;12:491–7.
- Noguchi Y, Yoshikawa T, Marsumoto A, Svaninger G, Gelin J. Are cytokines possible mediators of cancer cachexia? Jpn J Surg. 1996;26:467–75.
- 83. Keiler U. Pathophysiology of cancer cachexia. Support Care Cancer. 1993;1:290-4.
- Straton RJ, Dewit O, Crowe R, Jennings G, Viller RN, Elia M. Plasm leptin, energy intake and hunger following total hip replacement surgery. Clin Sci. 1997;93:113–7.
- 85. Cederholm T, Arter P, Palmviad J. Low circulation leptin level in protein-energy malnourished chronically ill elderly patients. J Intern Med. 1997;242:377–82.
- Milne AC, Potter J, Avenel A. Protein and energy supplementation in elderly people at risk from malnutrition. Cochrane Database Syst Rev 2006;(1).
- 87. Thomas DR. Guidelines for the use of orexigenic drugs in long-term care. Nutr Clin Pract. 2006;21(1):82-7.
- Bruera E, Macmillan K, Hanson J, MacDonald RN. A controlled trial of megestrol acetate on appetite, caloric intake, nutritional status, and other symptoms in patients with advance cancer. Cancer. 1990;66:1279–82.
- Allman RM, Goode PS, Patrick MM. Pressure ulcer risk factors among hospitalized patients with severe limitation. JAMA. 1995;273:865–70.
- 90. Schmoll E, Wilke H, Thole R. Megestrol acetate in cancer cachexia. Semin Oncol. 1991;1 Suppl 2:32-4.
- Heckmayr M, Gatzenneier U. Treatment of cancer weight loss in patients with advance lung cancer. Oncology. 1992;49 suppl 2:32–4.
- Feliu J, Gonzalez-Baron M, Berrocal A. Usefulness of megestrol acetate in cancer cachexia and anorexia. Am J Clin Oncol. 1992;15:436–40.
- Azcona C, Castro L, Crespo E, Jimenez M, Sierrasesumaga L. Megestrol acetate therapy for anorexia and weight loss in children with malignant solid tumours. Aliment Pharmacol Ther. 1996;10:577–86.
- 94. Mantovani G, Maccio A, Bianchi A, Curreli L, Ghiani M, Santona MC, et al. Megestrol acetate in neoplastic anorexia/cachexia: clinical evaluation and comparison with cytokine levels in patients with head and neck carcinoma treated with neoadjuvant chemotherapy. Int J Clin Lab Res. 1995;25(3):135–41.

Chapter 15 Nutrition Support in Solid Tumor Cancer Patients

Krishna A. Patel and Aminah Jatoi

Key Points

- Nutrition support in the form of enteral or parenteral nutrition confers minimal benefit to patients with advanced, incurable cancer and sometimes leads to complications.
- Severely malnourished cancer patients who are being considered for curative surgery or highly effective cancer treatment gain modest benefits from nutrition support.
- A multidisciplinary approach is key for deciding how and when to start nutrition support for a cancer patient.

Keywords Nutrition support • Enteral • Parenteral • Preoperative • Immunonutrition • Dietary counseling

Introduction

Few situations in medicine evoke such controversy as nutrition support in patients with cancer. For purposes of this chapter, nutrition support is defined as the prescription and administration of enteral or parenteral nutrition, either of which are given in a manner that can potentially override volitional factors on the part of the patient. Despite the fact that many organizations, such as the American College of Physicians, the Working Group for Developing Guidelines for Parenteral Nutrition of the German Association for Nutritional Medicine, and the American Society of Parenteral Nutrition, have cautioned against the indiscriminate use of nutritional support and specifically against the use of parenteral nutritional support in patients with advanced cancer, a malignant diagnosis represents the most common primary diagnosis among patients receiving parenteral nutrition [1-3].

K.A. Patel M.D.

A. Jatoi M.D.

Department of Internal Medicine, Mayo Clinic Rochester, 200 1st Street SW, Rochester, MN 55902, USA e-mail: patel.krishna@mayo.edu

Department of Oncology, Mayo Clinic, 200 First Street SW, Rochester, MN 55905, USA e-mail: Jatoi.Aminah@mayo.edu

Several factors might explain this controversy. First, although weight-losing, incurable cancer patients manifest a pathophysiology, which can be characterized by loss of appetite, a disproportionate loss of lean tissue, and possible hypermetabolism, these patients present with a clinical phenotype that is remarkably similar to that of a starving but otherwise disease-free individual [4]. In effect, these weight-losing cancer patients appear as if they are starving, thus providing a scenario that sometimes convinces health care providers to aggressively feed, when in fact many sources in the published medical literature, as alluded to below, indicate that feeding is perhaps not the most appropriate intervention. Second, for some health care providers, it seems almost to make intuitive sense to feed a weight-losing cancer patient. Weight loss suggests an energy deficit, and the easiest way to compensate for this deficit is to feed the patient aggressively, even in the setting of anorexia and even in the setting of an incurable malignancy. Although clinical trials do not support this approach, some health care providers develop a mindset that drives the mission of aggressively prescribing nutrition support to patients with advanced, incurable cancer. Finally, the emotional impact of watching an incurable cancer patient wither and die is difficult for the patient, for family members, and for health care providers [5]. The desire to do something, really anything—such as provision of nutrition support sometimes becomes an emotionally driven, although less well scientifically justified, approach.

Nutrition Support in Patients with Advanced, Incurable Cancer

Some examples of patients with advanced, incurable cancer include the lung cancer patient who has liver metastases, has no chance of cure, and has no appetite and is losing weight. Another example is the patient with recurrent ovarian cancer, progressive weight loss, and a malignancy that is no longer responding to chemotherapy. Is nutrition support with either parenteral or enteral nutrition always the best choice for the advanced cancer patient with incurable disease? The answer is "no." For example, in 1981, Nixon and others described the findings of a controlled trial in 50 colorectal patients who were randomly assigned to receive either chemotherapy plus parenteral nutrition or chemotherapy alone [6]. Hypertonic glucose, amino acids, vitamin E, folic acid, and vitamin B_{12} comprised the parenteral nutrition. Of note, among the patients who had adverse events on the parenteral nutrition arm, one died of unclear reasons shortly after entering the trial, another developed sepsis with a staphylococcal infection, and yet another suffered from a cerebral vascular accident. More importantly, overall, parenteral nutrition-treated cancer patients died earlier than those who did not receive this intervention: 79 days versus 305 days (P=0.03). This prospective study represents one of many that indicate the detriments of parenteral nutrition in patients with incurable solid malignancies.

As further evidence of the detriments of parenteral nutrition in patients with advanced cancer, a series of prospective studies culminated in a series of meta-analyses, which substantiate this same conclusion of detrimental effects. In one of these meta-analyses, McGeer and others reported that parenteral nutrition is associated with inferior survival when administered to cancer patients receiving chemotherapy (relative risk 0.81; 95 % confidence interval 0.62, 1.0), and it was also associated with lower cancer response rates, (odds ratio 0.68; 95 % confidence interval 0.40, 1.1) [7]. Although these individual studies are often criticized for their smaller sample sizes and although these meta-analyses are often criticized for the heterogenous nature of their data, these observations continue to bear out the conclusion that parenteral nutrition should be prescribed to patients on chemotherapy and with advanced, incurable cancer at best with great caution.

Similarly, Baldwin and others reported a meta-analysis that consisted of studies that specifically focused on enteral nutrition [8]. This meta-analysis was spurred by the fact that some international guidelines fall short of recommending parenteral nutrition to cancer patients but allude to the use of enteral nutritional supplements. Thus, this meta-analysis sought randomized controlled trials that focused on oral nutrition supplements versus routine care among cancer patients with weight loss/

malnutrition or among cancer patients at risk for weight loss/malnutrition. Among the 13 identified studies that included 1,414 patients, the oral nutritional supplements were associated with greater improvements in weight and energy intake (mean difference in weight = 1.86 kg, 95 % CI = 0.25 - 3.47, P = 0.02; and mean difference in energy intake = 432 kcal/day, 95 % CI = 172 - 693, P = 0.001) but with no major improvements on mortality (relative risk = 1.06, 95 % CI = 0.92 - 1.22, P = 0.43). Of note, however, this meta-analysis also demonstrated notable heterogeneity among studies, and re-analysis after removal of these sources of heterogeneity showed that these statistically significant differences in clinical outcomes disappeared. Thus, although cancer patients appear to consume more calories with oral nutritional supplementation, this approach does not appear to lead to major improvements in survival, as hoped.

Alternative Palliative Strategies in Patients with Advanced Incurable Cancer

Admittedly, the aforementioned studies suggest that nutrition support does not, as a rule, prolong life in patients with advanced, incurable cancer, but might there be other goals—such as palliation of symptoms-to be achieved with nutrition support in patients with advanced, incurable cancer? To attempt to answer this question, McCann and others undertook a 32-patient study that monitored patient-reported outcomes over a 12-month period [9]. The study was small, but given the intensity of assessment, duration of assessment, and sensitive nature of studying and monitoring dying patients, this limited sample size seems most appropriate. Of note, not all these patients had cancer, and these patients were not receiving aggressive nutritional support. Importantly, 20 (63 %) had no symptoms of hunger and described that a sense of thirst was only fleeting. Following these patients over time, these investigators observed that symptoms such as loss of appetite and increased thirst were attenuated by allowing patients to eat small meals or snacks at will, by allowing them to drink fluids ad lib, by sucking on ice chips, and by applying lubrication to the lips. This study helps resolve the argument that aggressive nutrition support provides an important palliative strategy for patients at the end of life. In fact, many patients are not hungry or thirsty at the end of life, aggressive nutrition support does not appear necessary to palliate symptoms, and other relatively simple strategies, as alluded to above, can play a fundamental palliative role under these circumstances.

Along similar lines, Bruera and others randomly assigned 129 hospice-enrolled cancer patients to normal saline 1 L intravenously per day versus normal saline 100 mL intravenously per day (placebo) over 4 h [10]. The primary outcome was change in the sum of four dehydration symptoms (fatigue, myoclonus, sedation, and hallucinations, 0=best and 40=worst possible) between day 4 and baseline. Saline administration did not improve patient-reported dehydration symptoms, thus emphasizing further that the so-called "palliative effects" of parenteral hydration/nutritional interventions may be far more limited than we would hope and thus providing the impetus to further study or utilize simpler, less-invasive interventions.

Acknowledging Exceptional Circumstances Among Advanced, Incurable Cancer Patients

It should be noted that there are occasionally exceptions and that some patients with metastatic cancer appear to benefit from nutrition support, specifically from parenteral nutrition support. A single-institution, 20-year experience included 52 patients with a variety of diagnoses were included: carcinoid/islet cell tumor (n=10), ovarian carcinoma (n=6), amyloidosis/multiple

myeloma (n=6), colorectal carcinoma (n=5), sarcoma (n=5), pancreatic carcinoma (n=4), gastric carcinoma (n=3), lymphoma (n=2), pseudomyxoma peritonei (n=2), and other (n=9) [11]. The median interval from the initiation of parenteral nutrition to death was 5 months (range: 1–154 months), but 16 patients survived for 1 year or longer. Parenteral nutrition-related complications included catheter infections, thromboses, pneumothoraces, and liver disease, but none of these complications occurred at an especially high rate to raise alarm. Overall, it appears that parenteral nutrition may have contributed to the prolonged survival observed in some of these patients, but importantly it should be emphasized that this cohort represents a highly select group of patients and that the findings of this study should not be extrapolated to the majority of metastatic cancer who are losing weight.

Meeting the Nutritional Needs of Severely Malnourished Cancer Patients Who Have Highly Treatable Cancers

Are there any cancer patients—for example those without metastatic disease—who might truly benefit from nutrition support? It is important to emphasize that the sparing use of nutrition support applies primarily to patients with advanced, incurable cancer. Previous studies indicate that solid cancer patients with potentially curable or highly treatable cancers can, at times, benefit from nutrition support. Hence, the importance of making decisions on nutrition support in a highly collaborative, multidisciplinary manner cannot be overemphasized. In further support of expending the time to collaborate, a recent multicenter trial compared early versus late initiation of parenteral nutrition among 2,312 intensive care unit-treated adults, slightly under 20 % of whom had a cancer diagnosis [12]. If parenteral nutrition was not initiated before day 8, patients in fact manifested a faster recovery and fewer complications. Hence, taking the time to assess the status of the cancer, the goals of antineoplastic therapy, and the likelihood of cure, long-term survival, or durable tumor response should, can, and must go hand-in-hand with decisions on nutrition support, and the data above suggest that under some circumstances, there may be time to undertake such discussions. Three studies, as summarized below, are particularly instructive and provide guidance, as relevant to this point and as relevant to parenteral nutrition.

First, although the Veterans Affairs Cooperative Study did not exclusively focus on cancer patients, 65 % of the 395-patient cohort included cancer patients [13]. This study otherwise focused on preoperative malnourished patients, who were enrolled within a randomized trial that included either total parenteral nutrition for 7-15 days before surgery and 3 days postoperatively versus no intravenous nutritional supplementation. Major, 30-day postoperative complication rates were similar in both groups (25.5 % with total parenteral nutrition versus 24.6 %). Similarly, there were no major differences between groups with respect to 90-day mortality rates (13.4 % versus 10.5 %). Of note, in other analyses, patients who were assigned to the parenteral nutrition arm suffered a higher infection rate (14 % versus 6 %), but a planned subgroup analysis, which included only 24 patients who seemed to have severe malnutrition, also observed that parenteral-nutrition-treated patients also had fewer noninfectious complications (5 % versus 43 %). The conclusions from such subset analyses should be viewed with caution because they do not comprise the primary study analysis and because they provide conclusions that were drawn from a highly contracted subset of patients. However, such findings might suggest that parenteral nutrition in a small group of cancer patients, specifically those who are severely malnourished in a preoperative setting, might confer modest benefit.

Second, Fan and others studied 124 hepatocellular cancer patients; these patients, too, were studied in a preoperative setting where they were about to undergo a partial hepatectomy for their cancer [14].

Less than 20 % of this cohort reported losing greater than 10 % premorbid weight. Prior to this surgery, patients received parenteral nutrition versus placebo fluids. The latter had minimal nutritional or caloric content. Although mortality rates were not markedly different between treatment arms, those parenteral nutrition-treated patients manifested lower rates of morbidity after surgery; specifically they manifested less-problematic ascites and better liver function, as assessed biochemically. Assessing malnutrition in hepatocellular cancer patients can be challenging, particularly because ascites can mask weight loss, an often-used marker of malnutrition. This study is important because it suggests that parenteral nutrition can provide modest benefit to hepatocellular carcinoma patients, who are often malnourished, prior to surgery.

Finally, in a third preoperative study that examined the role of preoperative nutrition, Bozzetti and others focused on 90 gastric or colorectal cancer patients, all of whom had suffered loss of at least 10 % of their premorbid weight [15]. As was the case with the other two studies described above, this study examined these patients in a preoperative setting, providing half the cohort preoperative parenteral nutrition and the other half no nutrition support. Patients in the former group manifested lower postoperative complication rates (37 % versus 57 %) of both infectious and noninfectious etiology. A smaller trial, this study nonetheless adds further support to the premise that parenteral nutrition can reduce complication rates in preoperative cancer patients.

What is the role of nutrition support among head and neck cancer patients? This group of cancer patients offers somewhat of a unique situation that calls for the consideration of enteral nutrition support. The latter is perhaps the preferred consideration based on the old adage, "If the gut works, use it." Although the data are sparse to provide absolute justification for this enteral nutrition support, a confluence of factors, which includes eating impairment as a result of the tumor and that include mucositis from cancer treatment such as radiation, suggests that some type of enteral support makes sense under these circumstances. The data that support this claim include retrospective studies as well as some prospectively derived data.

For example, Daly and others conducted a randomized trial of 40 head and neck cancer patients who were receiving radiation and who were assigned to either oral nutrition versus nasogastric tube feedings, greater benefit appeared manifest in the latter group of patients [16]. These patients were observed to have higher caloric intake, less weight loss, and a more rapid resumption of the activities of daily living. Of note, tumor response rates with cancer treatment and overall survival were not statistically different between groups. However, this issue of reduced weight loss may not be minor. A recent prospective study among 533 head and neck cancer patients who were being treated with curative intent with radiation found that 30 % lost 0.1-5.0 % of their weight during cancer treatment, 26 % lost 5.1–10.0 %, and 24 % lost >10 % [17]. Importantly, these investigators observed a statistically significant association between weight loss and decline in global quality of life as well as physical functioning, social functioning, social eating, and social contact. Hence, weight loss under these circumstances represents more than a drop in numbers on a scale but rather a clinical change that is associated with major decrements in quality of life. Thus, the randomized trial alluded to earlier in this paragraph may have been too small to capture subtle improvements between treatment arms, but it nonetheless provides a hint of evidence that aggressive enteral feeding may provide an advantage over dietary counseling in head and neck cancer patients who are receiving radiation and that this advantage might also include improvements in quality of life. Other studies have suggested that the need to hold cancer treatment is reduced with the use of enteral nutritional support. To our knowledge, no studies have demonstrated that nutritional support clearly improves survival or treatment effect outcomes from radiation or combined modality therapy for head and neck cancer, but the fact that these patients suffer eating impairments and weight loss before and during cancer treatment leads many health care providers to readily adopt enteral feeding strategies as part of the cancer treatment plan.

Nasogastric Versus Percutaneous Endoscopic Gastrostomy Feeding in Head and Neck Cancer Patients?

The above raises the unresolved issue of how best to provide enteral nutrition support. Should these patients receive nasogastric or percutaneous endoscopic gastrostomy feeding? A recent Cochrane review attempted to provide further direction to answer this question and compiled the randomized trials that had focused on this issue [18]. This review raised important points, including the fact that the duration of percutaneous endoscopic feeding tended to be much longer and more expensive than enteral feeding, at least in one of the trials reviewed in this meta-analysis. Although the authors concluded that they were unable to provide clear direction on which method to use based on quality of life and cancer outcomes, the issues of duration of use and cost associated with percutaneous endoscopic feeding provide further details that might enter into the discussions as health care providers discuss options with cancer patients. Moreover, in our own experience, concerns over seeding tumor cells with insertion of a nasogastric tube, continued contact and erosion of tissue around the nares and further down from the tube itself, and the cosmetic challenges of interacting socially with a nasogastric tube in place appear to shift patients and health care providers towards considering percutaneous endoscopic feeding.

Should all cancer patients about to receive surgery or about to receive highly effective cancer therapy receive nutrition support in some manner? We do not think so. Although this question is controversial, justification for enteral or parenteral nutritional support can be found under the circumstances where a cancer patient faces the prospect of cure of their cancer or another major favorable outcome to be derived from cancer treatment. Importantly, it appears that cancer patients must also be showing clear signs of severe malnutrition and/or risk of malnutrition in order to derive benefit from nutrition support. Even under such circumstances, it should be noted that nutritional support does not appear to provide an improvement in survival but rather it provides a more modest set of benefits that center around reduced morbidity. Thus, it stands to reason that nutrition support can and should be prescribed under some circumstances to cancer patients but that a balanced discussion with the patient should precede this intervention.

A Word on Nutritional Counseling

As an aside, this chapter focuses on prescribing parenteral or enteral nutrition to cancer patients, the question often arises: What is the value of nutritional counseling in the absence of nutritional support to cancer patients? Although no final answer exists to this question, some studies, including those from Ravasco and others suggest benefit [19, 20]. A meta-analysis from Halfdanarson attempted to put these studies in context. These investigators accepted the fact that nutritional interventions have not yielded a survival advantage and instead focused on quality of life [21]. Quality of life had to have been measured by a validated questionnaire, and all randomized trials were sought that evaluated dietary counseling. These investigators identified five such trials that met this study's eligibility criteria. Quality of life scores among patients who received dietary counseling were only marginally improved compared to patients who did not receive this intervention (P=0.06). However, this trend towards statistical significance suggests that human interactions and discussions about how to improve dietary intake definitely are exerting some positive influences on cancer patients. Thus, regardless of cancer status and regardless of the mixed findings reported in previous studies, it appears that dietary counseling can also be of value to weight-losing cancer patients.

Immunonutrition

Several formulas that are used for enteral nutrition include supplemental agents such as l-arginine and omega-3 fatty acids, and the term immunonutrition has been used to describe such feeding sources. Although it is outside the scope of this chapter to comment on all the available enteral formulations and methods for administering parenteral nutrition, some comments on immunonutrition seem appropriate. Several clinicians have started incorporating such formulations into their clinical practices, as a result of a growing number of clinical trials and meta-analyses. Cresci has offered guidance on the type of patient who may be best suited [22]: "Ideal candidates for these formulations... include the following: malnourished and non-malnourished elective surgery patients that may also have comorbid conditions such as diabetes, heart disease, lung disease, and obesity; critically ill ICU patients that have several comorbid conditions placing them at higher risk of infections; and critically ill trauma and surgical patients that are anticipated to remain in the ICU for >3 to 5 days."

In our opinion, however, the benefits of immunonutrition are compelling with a reduction in postoperative complications, but such results are not definitive. For example, one meta-analysis included 21 randomized controlled trials that in total included a robust sampling of 2,730 patients [23]. Although not all these patients were cancer patients, some were; and some were about to have surgery for either a gastrointestinal malignancy. Immunonutrition significantly reduced perioperative complication rates and even resulted in a reduced interval of hospitalization. Importantly, this meta-analysis represented clinical trial data that had been generated by only a limited number of authors, and notable heterogeneity existed among studies. Thus, although one can clearly justify the use of immunonutrition, in our opinion, the data in their totality do not seem compelling enough to enable us to consider this approach the clear standard of care. Of parenthetical note, the use of immunonutrition in patients with advanced cancer has yielded mostly negative neutral outcomes.

Conclusion

As described above, multiple negative, randomized controlled trials and subsequent meta-analyses underscore a limited role of nutrition support—particularly parenteral nutrition support—in patients with advanced cancer. These studies had prompted the American College of Physicians to issue a 1989 position statement that clearly advises against parental nutrition in cancer patients who are being prescribed chemotherapy with noncurative intent, and, over the span of more than two decades, this position statement has not been revised [1]. Moreover, even in patients with potentially curable malignancies, it appears that the benefits of nutrition support are restricted to only subgroups of patients with severe malnutrition. However, in reality, nutrition support-and, in particular, parenteral nutrition support—appears to be prescribed quite readily to cancer patients. In past years, 40 % of patients who are receiving home parenteral nutrition in the United States have a cancer diagnosis [24]. Similarly, in Sweden, the Netherlands, and France, these percentages are 80 %, 60 %, 27 %, respectively. Only in the United Kingdom is this percentage low at 5 %. It remains unclear whether all these patients have metastatic cancer or what sorts of circumstances resulted in these patients receiving such interventions. In view of these high rates of usage, factors other than evidence derived from clinical trials are likely driving these decisions. Understanding what motivates health care providers, patients, and their families to reach the point where they choose nutritional support will be key to making sure cancer patients are in fact receiving the most appropriate intervention at the right time in their disease course. Such an understanding will also provide assurance that patients' educational needs on this topic are being adequately met.

References

- 1. Parenteral nutrition in patients receiving cancer chemotherapy. American College of Physicians. Ann Intern Med 1989;110:734–6.
- Arends J, Zuercher G, Dossett A, Fitkau R, Huq M, Schmid I, et al. Non-surgical oncology—guidelines on parenteral nutrition. Ger Med Sci 2009;7:Doc 09.
- August DA, Huhmann MB, American Society for Parenteral and Enteral Nutrition Board of Directors. ASPEN clinical guidelines: nutrition support therapy during adult anticancer treatment and in hematopoietic cell transplantation. JPEN J Parenter Enteral Nutr. 2009;33:472–500.
- 4. Dodson S, Baracos VE, Jatoi A, Evans WJ, Cella D, Dalton JT, Steiner MS. Muscle wasting in cancer cachexia: clinical implications, diagnosis, and emerging treatment strategies. Annu Rev Med. 2011;62:265–79.
- 5. Hopkinson JB. The emotional aspects of cancer anorexia. Curr Opin Support Palliat Care. 2010;4:254-8.
- Nixon DW, Lawson DH, Kutner MH, Moffitt SD, Ansley J, Heymsfield SB, et al. Effect of total parenteral on survival in advanced colon cancer. Cancer Detect Prev. 1981;4:421–7.
- McGeer AJ, Detsky AS, O'Rourke K. Parenteral nutrition in cancer patients undergoing chemotherapy: a metaanalysis. Nutrition. 1990;6:233–40.
- Baldwin C, Spiro A, Ahern R, Emery PW. Oral nutritional interventions in malnourished patients with cancer: a systematic review and meta-analysis. J Natl Cancer Inst. 2012;104:371–85.
- 9. McCann RM, Hall WJ, Groth-Juncker A. Comfort care for terminally ill patients. The appropriate use of nutrition and hydration. JAMA. 1994;272:1263–6.
- 10. Bruera E, Hui D, Dalal S, Torres-Vigil I, Trumble J, Roosth J, et al. Parenteral hydration in patients with advanced cancer: a multicenter, double-blind, placebo controlled randomized trial. J Clin Oncol. 2013;31:111–8.
- 11. Hoda D, Jatoi A, Burnes J, Loprinzi C, Kelly D. Should patients with advanced, incurable cancers ever be sent home with total parenteral nutrition? A single institution's 20-year experience. Cancer. 2005;103:863–8.
- 12. Casaer MP, Mesotten D, Hemans G, Wouters PJ, Schetz M, Meyfroidt G, et al. Early versus late parenteral nutrition is critically ill adults. N Engl J Med. 2011;365:506–17.
- 13. Perioperative total parenteral nutrition in surgical patients. The Veterans Affairs Total Parenteral Nutrition Cooperative Study Group. N Engl J Med 1991;325:525–32.
- Fan ST, Lo CM, Lai EC, Chu KM, Liu CL, Wong J. Perioperative nutritional support in patients undergoing hepatectomy for hepatocellular carcinoma. N Engl J Med. 1994;331:1547–52.
- Bozzetti F, Gavazzi C, Miceli R, Rossi N, Mariani L, Cozzaglio L, et al. Perioperative total parenteral nutrition in malnourished, gastrointestinal cancer patients: a randomized, clinical trial. JPEN J Parenter Enteral Nutr. 2000;24:7–14.
- 16. Daly JM, Hearne B, Dunaj J, LePorte B, Vikram B, Strong E, et al. Nutritional rehabilitation in patients with advanced head and neck cancer receiving radiation therapy. Am J Surg. 1984;148:514.
- Langius JA, van Dijk AM, Doornaert P, Kruizenga HM, Langendijk JA, Leemans CR, et al. More than 10 % weight loss in head and neck cancer patients during radiotherapy is independently associated with deterioration in quality of life. Nutr Cancer. 2013;65:76–83.
- Gomes CA, Lustosa SA, Matos D, Andriolo RB, Waisberg DR, Waisberg J. Percutaneous endoscopic gastrostomy versus nasogastric tube feeding for adults with swallowing disturbances. Cochrane Database Syst Rev. 2012;14:3.
- Ravasco P, Monteiro-Grillo I, Camilo M. Individualized nutrition intervention is major benefit to colorectal cancer patients: long-term follow-up of a randomized controlled trial of nutritional therapy. Am J Clin Nutr. 2012;96:1346–53.
- Ravasco P, Monteiro-Grillo I, Marques Vidal P, Camilo ME. Impact of nutrition on outcome: a prospective randomized controlled trial in patients with head and neck cancer undergoing radiotherapy. Head Neck. 2005;27:659–68.
- Halfdanarson TR, Thordardottir E, West CP, Jatoi A. Does dietary counseling improve quality of life in cancer patients? A systematic review and meta-analysis. J Support Oncol. 2008;6:234–7.
- 22. Cresci G. http://intl-pen.sagepub.com/content/29/1_suppl/S92.full. Last accessed 8 Jun 2013.
- Cerantola Y, Hubner M, Grass F, Demartines N, Schafer M. Immunonutrition in gastrointestinal surgery. Br J Surg. 2011;98:37–48.
- 24. Howard L. Home parenteral nutrition: a translatlantic view. Clin Nutr. 1999;18:131-3.

Chapter 16 Nutrition and Chronic Kidney Disease

Xiaorui Chen and Srinivasan Beddhu

Key Points

- The development and progression of chronic kidney disease (CKD) is influenced by a number of dietary factors, including salt and protein intake and energy balance (obesity).
- While the benefits of a low protein intake in preventing the development of CKD are not firmly established, it is likely that a high-protein intake is detrimental to individuals with even mild impairment of renal function.
- Specific dietary recommendations regarding protein level and intakes of sodium, phosphorus, potassium, and fluids must be carefully individualized as appropriate for the level of renal function.
- Obesity is a risk factor for the development of CKD via a variety of mechanisms, including clinical or subclinical insulin resistance, hypertension, and possibly other metabolic derangements.
- However, obesity is associated with better survival in dialysis patients. This is likely because the deleterious metabolic effects of obesity are outweighed by its protective nutritional effects in Stage Five (V) CKD patients on dialysis.

Keywords Renal disease • Hemodialysis • Uremia • Metabolic acidosis

X. Chen M.S. (🖂)

S. Beddhu M.D.

Medical Service, Salt Lake Veterans Affairs Healthcare System, Salt Lake City, UT, USA

Department of Nephrology, University of Utah, Salt Lake City, UT, USA e-mail: Xiaorui.Chen@hsc.utah.edu

Division of Nephrology & Hypertension, University of Utah School of Medicine, Salt Lake City, UT, USA e-mail: srinivasan.beddhu@hsc.utah.edu

Introduction

Based on the National Health and Nutrition Examination Survey (NHANES), it was estimated that 14 % of U.S. adults have chronic kidney disease (CKD) [1], and the prevalence of CKD in U.S. adults increased by 30 % between 1988–1994 and 1999–2004 [2]. Furthermore, CKD is a strong independent predictor of atherosclerotic events [3, 4]. The annual mortality on dialysis is approximately 20 %, and the total Medicare cost of end-stage renal disease in 2009 was estimated to be \$29 billion [5]. Thus, CKD is extremely common, carrying a significant health and economic burden.

Nutritional status plays a major role in development and progression of CKD as well as strongly influences survival in established CKD. Thus, dietary interventions could potentially have a significant impact on decreasing the progression of CKD and improving the outcomes of those with established CKD. The data on (1) the impact of diet and obesity on the development and progression of CKD and (2) nutritional issues unique to advanced CKD (uremia) are reviewed in this chapter.

Role of Diet and Obesity on Development and Progression of CKD

Salt Intake, Hypertension, and the Risk of CKD

Hypertension is a very strong risk factor for CKD. In those with preexisting hypertension, lowering the dietary sodium intake to approximately 1 g/day is associated with better control of blood pressure [6, 7]. In those without preexisting hypertension, the beneficial effects of lowering salt intake have been controversial [8]. However, in the Dietary Approaches to Stop Hypertension (DASH) Sodium Study, 412 participants were randomly assigned to eat either a control diet (typical of intake in the United States) or the DASH diet (which is rich in vegetables, fruits, and low-fat dairy products). Within the assigned diet, participants ate foods with high, intermediate, and low levels of sodium for 30 consecutive days each, in random order. The effects of sodium were observed in participants with and in those without hypertension. The DASH diet was associated with a significantly lower systolic blood pressure at each sodium level. As compared to the control diet with a high sodium level, the DASH diet with a low sodium level led to a mean systolic blood pressure that was 7.1 mmHg lower in participants without hypertension, and 11.5 mmHg lower in participants with hypertension. Furthermore, high sodium intake [9] causes hyperfiltration patterns in healthy men with a BMI \geq 25 kg/m². These studies suggest that lower sodium intake is not only beneficial for preventing the development of CKD, but also plays a role in slowing progression of kidney disease. Nonetheless, there are no randomized controlled trials that have examined whether reduction of dietary sodium in normotensive individuals will reduce the development of CKD.

Protein Intake and Kidney Function

The impact of protein intake on renal function likely differs based on the level of baseline renal function. Hence, these are discussed separately.

In Those with Normal Renal Function

A high-protein, low-carbohydrate diet has been shown to produce greater weight loss in obese subjects compared to a low-fat, high-carbohydrate diet [10]. However, there are concerns on whether a high-protein diet will cause nephrotoxicity. In a short-term study of 24 healthy young men, high-protein diet has been found to significantly increase the glomerular filtration rate, as well as blood urea nitrogen, serum uric acid, glucagon, natriuresis, urinary albumin, and urea excretion [11]. In a 6-month randomized controlled trial of 65 healthy, overweight, and obese individuals, dietary protein intake changed from 91.1 g/day to a 6 months intervention average of 70.4 g/day (P<0.05) in the low-protein group and from 91.4 to 107.8 g/day (P<0.05) in the high-protein group [12]. These resulted in changes in glomerular filtration rate (GFR) of -7.1 mL/min in the low-protein group and +5.2 mL/min in the high-protein group (group effect: P<0.05). Kidney volume decreased by -6.2 cm³ in the low-protein group and increased by +9.1 cm³ in the high-protein group (P<0.05), whereas albuminuria remained unchanged in all groups. Thus, a high-protein diet led to increased renal mass and hyperfiltration. As hyperfiltration is a recognized risk factor for progressive kidney damage, theoretically a high-protein diet over the long term might result in kidney failure. On the other hand, as discussed below, obesity per se is a risk factor for kidney damage. Therefore, the risk-benefit ratio of long-term use of high-protein diet on kidney function remains unclear. Hence, based on the currently available data [10], short-term (6–12 months) use of high-protein diet in obese individuals for weight loss might be reasonable.

In Those with Chronic Kidney Disease

In animal studies, restricting protein intake delays the progression of renal disease. In smaller clinical trials of insulin-dependent diabetic nephropathy, low-protein intake resulted in slower decline of GFR [13, 14]. However, in the Modification of Diet in Renal Diseases (MDRD) Study [9], in 585 individuals with moderate CKD (GFR of 25–55 mL/min/1.73 m²) compared to the usual-protein diet (1.3 g/ kg/day), a low-protein diet (0.58 g/kg/day) did not result in slower decline of kidney function. A longer term follow-up using national registries of these patients after the study ended suggested that there might be an early benefit with low protein intake regarding the development of ESRD or the composite of ESRD or all-cause mortality [15].

In the same study, in 255 patients with more advanced CKD (GFR 13–24 mL/min/1.73 m²), compared to a low-protein diet (0.58 g/kg/day), a very-low-protein diet (0.28 g/kg/day) with a keto acidamino acid supplement did not result in significantly slower decline in GFR [10]. A recent longer term follow-up of these patients using the national registries after the study was completed, suggests that assignment to a very low-protein diet did not delay progression to kidney failure, but appeared to increase the risk of death [16]. More recently, 423 patients with stage 4–5 CKD were randomly assigned to low-protein diet (0.73 ± 0.04 g/kg/day) or a moderate protein diet (0.9 ± 0.06 g/kg/day). Over 48 months of follow-up, the mean monthly decline in the GFR did not significantly differ between the low protein diet and moderate protein diet (0.19 ± 0.48 mL/min/1.73 m² and 0.18 ± 0.46 mL/ min/1.73 m²) [17]. Thus, the role of protein restriction in slowing the progression of CKD remains controversial.

In summary, high-protein diet might be beneficial for weight loss in obese individuals with normal kidney function but should be avoided in those with even mild impairment of renal function. Therefore, kidney function should be estimated with a GFR estimating equation before high-protein diet could be prescribed particularly because mild kidney dysfunction is very common and is almost always asymptomatic.

Obesity, Insulin Resistance, and the Risk of CKD

While the impact of obesity on cardiovascular disease, hypertension, and diabetes are well established, only recently the data on the impact of obesity on kidney disease are emerging [18–22]. In an analysis of 320,252 adults, compared to those with normal BMI (18.5–24.9 kg/m²), the risk of



Fig. 16.1 Mechanisms of kidney damage in obesity

developing end-stage renal disease (ESRD) needing dialysis therapy was 1.87 (95 % CI, 1.64–2.14) for those who were overweight (BMI, 25.0–29.9 kg/m²), 3.57 (CI, 3.05–4.18) for those with class I obesity (BMI, 30.0–34.9 kg/m²), 6.12 (CI, 4.97–7.54) for those with class II obesity (BMI, 35.0–39.9 kg/m²), and 7.07 (CI, 5.37–9.31) for those with extreme obesity (BMI \geq 40 kg/m²) [21]. Furthermore, in morbidly obese individuals who underwent bariatric surgery, loss of weight is accompanied by better control of blood pressure and reduction of glomerular hyperfiltration and proteinuria [22–24]. These data suggest that obesity is a risk factor for kidney disease.

In the above studies, higher baseline BMI remained an independent predictor for CKD/ESRD after additional adjustments for baseline blood pressure level and the presence or absence of diabetes mellitus [21]. However, diabetes is only one extreme of insulin resistance. It is possible that subclinical insulin resistance in nondiabetic individuals might reflect a poor metabolic milieu and might still lead to kidney injury. Indeed, in nondiabetic adults, insulin resistance is associated with development of CKD [25].

The potential molecular mechanisms for this phenomenon are depicted in Fig. 16.1. Adipose tissue is not a mere storage depot of fat. It is metabolically active and produces adipokines such as tumor necrosis factor- α (TNF- α), interleukin-6 (IL-6), plasminogen activator inhibitor, leptin, angiotensinogen, and adiponectin [26–40]. Alterations in production of these adipokines in obesity result in metabolic derangements that cause insulin resistance, dyslipidemia, hypertension, and inflammation [26–40]. In morbidly obese individuals who underwent bariatric surgery, loss of weight is accompanied by better control of blood pressure [41, 42] and reduction of glomerular hyperfiltration and proteinuria [22–24]. These data suggest that weight loss reduces kidney damage in obesity. A part of this effect might be mediated through hypertension and a part of this effect might be independent of hypertension. Insulin resistance, inflammation, and oxidative stress as consequences of altered production of adipokines in obesity might also result in kidney damage.

Nutritional Issues in Uremia

The previous section discussed the role of diet and obesity in the development and progression of CKD. In this section, nutritional issues that might be unique to uremia, i.e., the obesity paradox, the pathophysiology of wasting syndrome, and serum phosphorus in advanced CKD are discussed.

Obesity Paradox in Uremia

The annual mortality in dialysis patients is approximately 20 % a year. Better nutritional status as evidenced by higher body size, muscle mass, and serum albumin is associated with better survival in dialysis patients. In particular, in contrast to the associations of high body mass index (BMI) with increased mortality in the general population, high BMI is associated with better survival in dialysis patients [4, 43–51]. This phenomenon has been described as a risk factor paradox or reverse epidemiology of wasting disease [47, 52, 53]. Hence, it has been suggested that obesity is protective rather than harmful in dialysis patients [47].

Although high BMI is associated with better survival in dialysis patients, the associations of adiposity with traditional cardiovascular risk factors such as diabetes and nontraditional risk factors such as inflammation are not confined to the nondialysis population. Indeed, previous studies have shown that in dialysis patients, adiposity and/or high BMI is associated with insulin resistance [54], diabetes [55], inflammation [56], anemia [57], coronary calcification [58, 59], and carotid atherosclerosis [54]. It is perplexing: if adiposity is associated with these apparent cardiovascular risk factors, why is adiposity associated with better survival in dialysis patients?

These apparently perplexing associations might be explained if (1) adiposity has dual competing effects on survival; a protective nutritional effect and a deleterious metabolic effect resulting in insulin resistance, dyslipidemia, hypertension, and inflammation and (2) the level of kidney function modifies the relative importance of these effects [60]. In this paradigm, the deleterious metabolic effects of obesity outweigh its protective nutritional effects in the non-CKD population, the deleterious metabolic effects of obesity are neutralized by its protective nutritional effects in the moderate CKD population and the deleterious metabolic effects of obesity are outweighed by its protective nutritional effects in stage V CKD on dialysis. In other words, the overall effects of obesity on survival vary according to the level of kidney function, and there is an interaction of body size and presence or absence of CKD on survival, even though the metabolic effects of adiposity are not modified by the level of kidney function.

Pathophysiology of Malnutrition in Uremia

Figure 16.2 summarizes the likely causal pathways for sarcopenia (low muscle mass) in uremia. Muscle mass is the net result of muscle protein synthesis and breakdown. Decreased protein intake results in decreased muscle protein synthesis, whereas inflammation, oxidative stress, and metabolic acidosis promote muscle protein breakdown. Uremic toxins decrease muscle protein synthesis by decreasing protein intake as well as by directly inhibiting muscle protein synthetic machinery. Uremic toxins also increase muscle protein breakdown by inflammation, oxidative stress, and metabolic acidosis as well as by directly stimulating muscle protein catabolic pathways.

The relative importance of these pathways (decreased anabolism vs. increased catabolism) in causing uremic malnutrition remains controversial and the following discusses these issues.

Protein Intake and Nutritional Status

The current clinical guidelines recommend a dietary protein intake of 1.2 g/kg/day for hemodialysis patients [61]. These guidelines are based upon opinion and observational data. High-protein intake might result in increased serum phosphorus (discussed later) and metabolic acidosis which might be



Fig. 16.2 Likely causal pathways for sarcopenia in uremia

detrimental in dialysis patients. Furthermore, as discussed below, there are controversies whether dietary interventions could have an impact on the nutritional status in dialysis patients.

It has been suggested that malnutrition in dialysis patients is a misnomer because uremic wasting (the state of low muscle mass, body weight, and serum proteins in dialysis patients) is caused by hypercatabolism and not decreased dietary intake [62]. In support of this theory, there are experimental data that suggest that acidosis [1, 5, 63] and TNF- α [64, 65] activate the ubiquitin-proteasome proteolytic system, the major pathway for protein catabolism [66]. Thus, it has been suggested that metabolic acidosis and inflammation may be the major determinants of the "state of loss of protein" in dialysis patients [67]. Further, in a small study of eight peritoneal dialysis patients, increase in dialysate lactate (which could buffer acids) was associated with decreased expression of skeletal muscle ubiquitin and improvement of nutritional indices in 1 month [68]. The other factors that are thought to contribute to this state of increased catabolism include resistance to anabolic activities of insulin and muscle breakdown induced by the hemodialysis procedure [62, 69, 70].

Nonetheless, while increased catabolism might play a causative role in the state of low muscle mass, body mass index, and serum proteins in dialysis patients, it remains unclear whether inflammation, oxidative stress, and metabolic acidosis are the key determinants of muscle mass in uremia. Dietary intake might still be the most important determinant of muscle mass and body size in dialysis patients. For instance, after starvation for a prolonged period, malnutrition is as likely in dialysis patients as in the general population. However, once malnutrition is established, it might be more difficult to reverse it in dialysis patients as compared to the general population. Thus, difficulties in reversing malnourished state with increased nutrition do not preclude that malnourished state is result of deficient dietary intake.

Metabolic Acidosis and Nutritional Status

As discussed above, metabolic acidosis [1, 5, 63] activates ubiquitin-proteasome proteolytic system, the major pathway for protein catabolism [66], and, hence, metabolic acidosis is proposed as the major cause of low muscle mass in dialysis patients. Thus, low serum creatinine levels, which reflect lower muscle mass in prevalent hemodialysis patients is expected in those with lower serum bicarbonate concentrations. However, in an analysis of the well-dialyzed participants of the HEMO Study, the

opposite association was present, i.e., those with higher serum creatinine concentrations (higher muscle mass) had lower serum bicarbonate concentrations [71], suggesting that acidosis was not a dominant determinant of low muscle mass.

Inflammation and Nutritional Status

TNF- α also activates the ubiquitin-proteasome proteolytic system [64, 65]. Thus, inflammation is thought to be a major determinant of muscle wasting in dialysis patients [62]. In an earlier analysis of the HEMO Study data, very high levels of CRP were associated with decreased serum creatinine levels [71]. In a separate study, mid-thigh muscle mass estimated by computed tomography was also negatively associated with CRP [72]. In addition to increased catabolism, inflammation may also suppress appetite resulting in malnutrition [52, 73–75]. Thus, it is biologically plausible that inflammation is associated with malnutrition in dialysis patients.

However, longitudinal associations between inflammation and low BMI have never been documented in dialysis patients. Furthermore, in the above longitudinal analysis of HEMO Study, each unit increase in natural logarithm of CRP was associated with a very modest 0.15 mg/dL decrease in serum creatinine [71]. Hence, a tenfold increase in natural logarithm of CRP will be associated with a drop in serum creatinine of 1.5 mg% in prevalent hemodialysis patients. Thus, the association of severe inflammation with loss of muscle mass in dialysis patients is plausible but might not be sufficient to explain the high prevalence of low muscle mass in dialysis patients.

In summary, as suggested in Fig. 16.2, the effects of deficient diet and increased catabolism need not be mutually exclusive and on the other hand, they might have synergistic effects. Interventional studies are warranted to determine the relative importance of anabolic vs. catabolic processes in uremic malnutrition.

Calcium-Phosphorus, Parathyroid Hormone, and Vascular Calcification

With decline in GFR, phosphorus excretion in the urine is reduced and phosphorus accumulates in the blood. This results in hypocalcemia as well as hyperparathyroidism.

Higher phosphorus, calcium phosphorus product, and parathyroid hormone levels are associated with increased vascular calcification [76] and mortality in dialysis patients [77]. Serum phosphorus could be reduced by decreasing dietary phosphorus intake. However, this is problematic because foods that are high in phosphorus are also high in protein. Therefore, in order to let the patient consume reasonable amounts of protein, phosphorus binders are used along with meals. These act by binding the phosphorus in the gut resulting in nonabsorbable compounds that are eliminated in stools.

Aluminum-containing phosphorus binders resulted in elevated serum aluminum levels with consequent osteomalacia [78, 79] and decreased mentation [80] (dialysis dementia). Therefore, aluminumcontaining phosphorus binders have been largely abandoned. Currently, the two main classes of phosphorus binders are calcium-based or non-calcium-based. Randomized controlled trials showed that therapy with sevelamer (a non-calcium-containing phosphorus binder) as opposed to calciumcontaining phosphorus binders were associated with decreased arterial calcification, particularly in those who are older and have preexisting calcification [81, 82]. In a follow-up analysis of one of these trials of 129 incident hemodialysis patients, in sevelamer-treated subjects, mortality, a prespecified secondary endpoint was lower. However, in a multicenter, randomized, open-label, parallel design trial (n=2,103) of sevelamer and calcium-based binders, all-cause mortality rates and cause-specific mortality rates were not significantly different [83]. However, there was a significant age interaction on the treatment effect. Only in patients over 65 years of age was there a significant effect of sevelamer

Condition	Appropriate diet
Obesity with normal kidney function (estimated MDRD GFR>80 mL/min/1.73 m ²)	High-protein, low-carbohydrate diet (or other well-balanced, reduced-calorie diet)
	No fluid restriction
Impaired kidney function (MDRD GFR 30 to ≤80 mL/	Avoid high-protein diet
$min/1.73 m^2$)	Potential benefit of protein restriction (0.6 g/kg/day)
	Salt restriction (3–4 g/day)
	No fluid restriction if urine output is normal
Stage IV CKD (MDRD GFR 15-29 mL/min/1.73 m ²)	Avoid high-protein diet
	Potential benefit of protein restriction (0.6 g/kg/day)
	Salt restriction (3–4 g/day)
	Restrict potassium and phosphorus as needed
	No fluid restriction if urine output is normal
On dialysis	Potential benefit of high-protein diet (1.2 g/kg/day)
	Salt restriction (3–4 g/day)
	Restrict potassium and phosphorus
	Restrict fluid intake to 1–1.5 L/day

Table 16.1 Summary of dietary recommendations based on the level of renal function

in lowering the mortality rate. There was a suggestion that sevelamer was associated with lower overall, but not cardiovascular-linked, mortality in older patients. Taken together with the above observational data, sevelamer might be beneficial in elderly who are likely to have arterial calcification but not in younger dialysis patients.

Conclusion: Dietary Recommendations Based on Level of Kidney Function

Table 16.1 provides a summary of diet recommendations individualized appropriately for the level of renal function. More extensive information about appropriate dieting modifications during various levels of CKD can be found at the National Kidney Foundation website (www.kidney.org).

In summary, diet and nutritional status exert a strong effect on both the development and the progression of CKD, and is a strong determinant of the outcomes in those with established CKD. Dietary and lifestyle interventions could reduce the burden of CKD in the United States.

References

- 1. May RC, Kelly RA, Mitch WE. Mechanisms for defects in muscle protein metabolism in rats with chronic uremia. Influence of metabolic acidosis. J Clin Invest. 1987;79:1099–103.
- Coresh J, Selvin E, Stevens LA, Manzi J, Kusek JW, Eggers P, et al. Prevalence of chronic kidney disease in the United States. JAMA. 2007;298:2038–47.
- Go AS, Chertow GM, Fan D, McCulloch CE, Hsu CY. Chronic kidney disease and the risks of death, cardiovascular events, and hospitalization. N Engl J Med. 2004;351:1296–305.
- Beddhu S, Allen-Brady K, Cheung AK, Horne BD, Bair T, Muhlestein JB, et al. Impact of renal failure on the risk of myocardial infarction and death. Kidney Int. 2002;62:1776–83.
- May RC, Bailey JL, Mitch WE, Masud T, England BK. Glucocorticoids and acidosis stimulate protein and amino acid catabolism in vivo. Kidney Int. 1996;49:679–83.
- Melander O, von Wowern F, Frandsen E, Burri P, Willsteen G, Aurell M, et al. Moderate salt restriction effectively lowers blood pressure and degree of salt sensitivity is related to baseline concentration of renin and N-terminal atrial natriuretic peptide in plasma. J Hypertens. 2007;25:619–27.

- Swift PA, Markandu ND, Sagnella GA, He FJ, MacGregor GA. Modest salt reduction reduces blood pressure and urine protein excretion in black hypertensives: a randomized control trial. Hypertension. 2005;46:308–12.
- Fodor JG, Whitmore B, Leenen F, Larochelle P. Lifestyle modifications to prevent and control hypertension. 5. Recommendations on dietary salt. Canadian Hypertension Society, Canadian Coalition for High Blood Pressure Prevention and Control, Laboratory Centre for Disease Control at Health Canada, Heart and Stroke Foundation of Canada. CMAJ. 1999;160:S29–34.
- Krikken JA, Lely AT, Bakker SJ, Navis G. The effect of a shift in sodium intake on renal hemodynamics is determined by body mass index in healthy young men. Kidney Int. 2007;71:260–5.
- Gardner CD, Kiazand A, Alhassan S, Kim S, Stafford RS, Balise RR, et al. Comparison of the Atkins, Zone, Ornish, and LEARN diets for change in weight and related risk factors among overweight premenopausal women: the A TO Z Weight Loss Study: a randomized trial. JAMA. 2007;297:969–77.
- Frank H, Graf J, Amann-Gassner U, Bratke R, Daniel H, Heemann U, et al. Effect of short-term high-protein compared with normal-protein diets on renal hemodynamics and associated variables in healthy young men. Am J Clin Nutr. 2009;90:1509–16.
- 12. Skov AR, Toubro S, Bulow J, Krabbe K, Parving HH, Astrup A. Changes in renal function during weight loss induced by high vs low-protein low-fat diets in overweight subjects. Int J Obes Relat Metab Disord. 1999;23:1170–7.
- Walker JD, Bending JJ, Dodds RA, Mattock MB, Murrells TJ, Keen H, et al. Restriction of dietary protein and progression of renal failure in diabetic nephropathy. Lancet. 1989;2:1411–5.
- Zeller K, Whittaker E, Sullivan L, Raskin P, Jacobson HR. Effect of restricting dietary protein on the progression of renal failure in patients with insulin-dependent diabetes mellitus. N Engl J Med. 1991;324:78–84.
- Levey AS, Greene T, Sarnak MJ, Wang X, Beck GJ, Kusek JW, et al. Effect of dietary protein restriction on the progression of kidney disease: long-term follow-up of the Modification of Diet in Renal Disease (MDRD) Study. Am J Kidney Dis. 2006;48:879–88.
- Menon V, Kopple JD, Wang X, Beck GJ, Collins AJ, Kusek JW, et al. Effect of a very low-protein diet on outcomes: long-term follow-up of the Modification of Diet in Renal Disease (MDRD) Study. Am J Kidney Dis. 2009; 53:208–17.
- Cianciaruso B, Pota A, Bellizzi V, Di Giuseppe D, Di Micco L, Minutolo R, et al. Effect of a low- versus moderateprotein diet on progression of CKD: follow-up of a randomized controlled trial. Am J Kidney Dis. 2009; 54:1052–61.
- Hallan S, de Mutsert R, Carlsen S, Dekker FW, Aasarod K, Holmen J. Obesity, smoking, and physical inactivity as risk factors for CKD: are men more vulnerable? Am J Kidney Dis. 2006;47:396–405.
- Kramer H, Luke A, Bidani A, Cao G, Cooper R, McGee D. Obesity and prevalent and incident CKD: the Hypertension Detection and Follow-Up Program. Am J Kidney Dis. 2005;46:587–94.
- Gelber RP, Kurth T, Kausz AT, Manson JE, Buring JE, Levey AS, et al. Association between body mass index and CKD in apparently healthy men. Am J Kidney Dis. 2005;46:871–80.
- Hsu CY, McCulloch CE, Iribarren C, Darbinian J, Go AS. Body mass index and risk for end-stage renal disease. Ann Intern Med. 2006;144:21–8.
- 22. Chagnac A, Weinstein T, Herman M, Hirsh J, Gafter U, Ori Y. The effects of weight loss on renal function in patients with severe obesity. J Am Soc Nephrol. 2003;14:1480–6.
- Vogel JA, Franklin BA, Zalesin KC, Trivax JE, Krause KR, Chengelis DL, et al. Reduction in predicted coronary heart disease risk after substantial weight reduction after bariatric surgery. Am J Cardiol. 2007;99:222–6.
- Navarro-Diaz M, Serra A, Romero R, Bonet J, Bayés B, Homs M, et al. Effect of drastic weight loss after bariatric surgery on renal parameters in extremely obese patients: long-term follow-up. J Am Soc Nephrol. 2006; 17:S213–7.
- Kurella M, Lo JC, Chertow GM. Metabolic syndrome and the risk for chronic kidney disease among nondiabetic adults. J Am Soc Nephrol. 2005;16:2134–40.
- Bastard JP, Jardel C, Delattre J, Hainque B, Bruckert E, Oberlin F. Evidence for a link between adipose tissue interleukin-6 content and serum C-reactive protein concentrations in obese subjects. Circulation. 1999;99:2221–2.
- Bhagat K, Vallance P. Inflammatory cytokines impair endothelium-dependent dilatation in human veins in vivo. Circulation. 1997;96:3042–7.
- Bouloumie A, Marumo T, Lafontan M, Busse R. Leptin induces oxidative stress in human endothelial cells. FASEB J. 1999;13:1231–8.
- Hotamisligil GS, Peraldi P, Budavari A, Ellis R, White MF, Spiegelman BM. IRS-1-mediated inhibition of insulin receptor tyrosine kinase activity in TNF-alpha- and obesity-induced insulin resistance. Science. 1996;271:665–8.
- Nakajima J, Mogi M, Kage T, Chino T, Harada M. Hypertriglyceridemia associated with tumor necrosis factoralpha in hamster cheek-pouch carcinogenesis. J Dent Res. 1995;74:1558–63.
- Ouchi N, Kihara S, Funahashi T, Nakamura T, Nishida M, Kumada M, et al. Reciprocal association of C-reactive protein with adiponectin in blood stream and adipose tissue. Circulation. 2003;107:671–4.
- Quehenberger P, Exner M, Sunder-Plassmann R, Ruzicka K, Bieglmayer C, Endler G, et al. Leptin induces endothelin-1 in endothelial cells in vitro. Circ Res. 2002;90:711–8.

- Schneider JG, von Eynatten M, Schiekofer S, Nawroth PP, Dugi KA. Low plasma adiponectin levels are associated with increased hepatic lipase activity in vivo. Diabetes Care. 2005;28:2181–6.
- Stephens JM, Pekala PH. Transcriptional repression of the GLUT4 and C/EBP genes in 3T3-L1 adipocytes by tumor necrosis factor-alpha. J Biol Chem. 1991;266:21839–45.
- Tham DM, Martin-McNulty B, Wang YX, Wilson DW, Vergona R, Sullivan ME, et al. Angiotensin II is associated with activation of NF-kappaB-mediated genes and downregulation of PPARs. Physiol Genomics. 2002;11:21–30.
- 36. Verma S, Li SH, Wang CH, Fedak PW, Li RK, Weisel RD, et al. Resistin promotes endothelial cell activation: further evidence of adipokine-endothelial interaction. Circulation. 2003;108:736–40.
- von Eynatten M, Schneider JG, Humpert PM, Rudofsky G, Schmidt N, Barosch P, et al. Decreased plasma lipoprotein lipase in hypoadiponectinemia: an association independent of systemic inflammation and insulin resistance. Diabetes Care. 2004;27:2925–9.
- 38. Wellen KE, Hotamisligil GS. Inflammation, stress, and diabetes. J Clin Invest. 2005;115:1111-9.
- Whitehead JP, Richards AA, Hickman IJ, Macdonald GA, Prins JB. Adiponectin—a key adipokine in the metabolic syndrome. Diabetes Obes Metab. 2006;8:264–80.
- 40. Ziccardi P, Nappo F, Giugliano G, Esposito K, Marfella R, Cioffi M, et al. Reduction of inflammatory cytokine concentrations and improvement of endothelial functions in obese women after weight loss over one year. Circulation. 2002;105:804–9.
- MacLaughlin HL, Cook SA, Kariyawasam D, Roseke M, van Niekerk M, Macdougall IC. Nonrandomized trial of weight loss with orlistat, nutrition education, diet, and exercise in obese patients with CKD: 2-year follow-up. Am J Kidney Dis. 2010;55:69–76.
- MacLaughlin HL, Sarafidis PA, Greenwood SA, Campbell KL, Hall WL, Macdougall IC. Compliance with a structured weight loss program is associated with reduced systolic blood pressure in obese patients with chronic kidney disease. Am J Hypertens. 2012;25:1024–9.
- Johansen KL, Young B, Kaysen GA, Chertow GM. Association of body size with outcomes among patients beginning dialysis. Am J Clin Nutr. 2004;80:324–32.
- 44. Abbott KC, Glanton CW, Trespalacios FC, Oliver DK, Ortiz MI, Agodoa LY, et al. Body mass index, dialysis modality, and survival: analysis of the United States Renal Data System Dialysis Morbidity and Mortality Wave II Study. Kidney Int. 2004;65:597–605.
- 45. Aoyagi T, Naka H, Miyaji K, Hayakawa K, Ishikawa H, Hata M. Body mass index for chronic hemodialysis patients: stable hemodialysis and mortality. Int J Urol. 2001;8:S71–5.
- Beddhu S, Pappas LM, Ramkumar N, Samore M. Effects of body size and body composition on survival in hemodialysis patients. J Am Soc Nephrol. 2003;14:2366–72.
- Kalantar-Zadeh K, Abbott KC, Salahudeen AK, Kilpatrick RD, Horwich TB. Survival advantages of obesity in dialysis patients. Am J Clin Nutr. 2005;81:543–54.
- Kopple JD. Nutritional status as a predictor of morbidity and mortality in maintenance dialysis patients. ASAIO J. 1997;43:246–50.
- 49. Kutner NG, Zhang R. Body mass index as a predictor of continued survival in older chronic dialysis patients. Int Urol Nephrol. 2001;32:441–8.
- 50. Leavey SF, McCullough K, Hecking E, Goodkin D, Port FK, Young EW. Body mass index and mortality in 'healthier' as compared with 'sicker' haemodialysis patients: results from the Dialysis Outcomes and Practice Patterns Study (DOPPS). Nephrol Dial Transplant. 2001;16:2386–94.
- Leavey SF, Strawderman RL, Jones CA, Port FK, Held PJ. Simple nutritional indicators as independent predictors of mortality in hemodialysis patients. Am J Kidney Dis. 1998;31:997–1006.
- 52. Kalantar-Zadeh K, Ikizler TA, Block G, Avram MM, Kopple JD. Malnutrition-inflammation complex syndrome in dialysis patients: causes and consequences. Am J Kidney Dis. 2003;42:864–81.
- 53. Nishizawa Y, Shoji T, Ishimura E, Inaba M, Morii H. Paradox of risk factors for cardiovascular mortality in uremia: is a higher cholesterol level better for atherosclerosis in uremia? Am J Kidney Dis. 2001;38:S4–7.
- 54. Yamauchi T, Kuno T, Takada H, Nagura Y, Kanmatsuse K, Takahashi S. The impact of visceral fat on multiple risk factors and carotid atherosclerosis in chronic haemodialysis patients. Nephrol Dialysis Transplant. 2003;18:1842–7.
- Beddhu S, Pappas LM, Ramkumar N, Samore MH. Malnutrition and atherosclerosis in dialysis patients. J Am Soc Nephrol. 2004;15:733–42.
- 56. Axelsson J, Rashid Qureshi A, Suliman ME, Honda H, Pecoits-Filho R, Heimbürger O, et al. Truncal fat mass as a contributor to inflammation in end-stage renal disease. Am J Clin Nutr. 2004;80:1222–9.
- 57. Axelsson J, Qureshi AR, Heimburger O, Lindholm B, Stenvinkel P, Barany P. Body fat mass and serum leptin levels influence epoetin sensitivity in patients with ESRD. Am J Kidney Dis. 2005;46:628–34.
- 58. Goodman WG, Goldin J, Kuizon BD, Yoon C, Gales B, Sider D, et al. Coronary-artery calcification in young adults with end-stage renal disease who are undergoing dialysis. N Engl J Med. 2000;342:1478–83.
- 59. Stompor T, Pasowicz M, Sullowicz W, Dembińska-Kieć A, Janda K, Wójcik K, et al. An association between coronary artery calcification score, lipid profile, and selected markers of chronic inflammation in ESRD patients treated with peritoneal dialysis. Am J Kidney Dis. 2003;41:203–11.
- Kwan BC, Beddhu S. A story half untold: adiposity, adipokines and outcomes in dialysis population. Semin Dial. 2007;20:493–7.
- Clinical practice guidelines for nutrition in chronic renal failure. K/DOQI, National Kidney Foundation. Am J Kidney Dis 2000;35:S1–140.
- 62. Mitch WE. Malnutrition: a frequent misdiagnosis for hemodialysis patients. J Clin Invest. 2002;110:437-9.
- Hara Y, May RC, Kelly RA, Mitch WE. Acidosis, not azotemia, stimulates branched-chain, amino acid catabolism in uremic rats. Kidney Int. 1987;32:808–14.
- 64. Goodman MN. Tumor necrosis factor induces skeletal muscle protein breakdown in rats. Am J Physiol. 1991;260:E727–30.
- 65. Goldberg AL, Kettelhut IC, Furuno K, Fagan JM, Baracos V. Activation of protein breakdown and prostaglandin E2 production in rat skeletal muscle in fever is signaled by a macrophage product distinct from interleukin 1 or other known monokines. J Clin Invest. 1988;81:1378–83.
- Mitch WE, Goldberg AL. Mechanisms of muscle wasting. The role of the ubiquitin-proteasome pathway. N Engl J Med. 1996;335:1897–905.
- 67. Ahuja TS, Mitch WE. The evidence against malnutrition as a prominent problem for chronic dialysis patients. Semin Dial. 2004;17:427–31.
- Pickering WP, Price SR, Bircher G, Marinovic AC, Mitch WE, Walls J. Nutrition in CAPD: serum bicarbonate and the ubiquitin-proteasome system in muscle. Kidney Int. 2002;61:1286–92.
- Pupim LB, Flakoll PJ, Brouillette JR, Levenhagen DK, Hakim RM, Ikizler TA. Intradialytic parenteral nutrition improves protein and energy homeostasis in chronic hemodialysis patients. J Clin Invest. 2002;110:483–92.
- Ikizler TA, Pupim LB, Brouillette JR, Levenhagen DK, Farmer K, Hakim RM, et al. Hemodialysis stimulates muscle and whole body protein loss and alters substrate oxidation. Am J Physiol Endocrinol Metab. 2002;282: E107–16.
- 71. Kaysen GA, Greene T, Daugirdas JT, Kimmel PL, Schulman GW, Toto RD, et al. Longitudinal and cross-sectional effects of C-reactive protein, equilibrated normalized protein catabolic rate, and serum bicarbonate on creatinine and albumin levels in dialysis patients. Am J Kidney Dis. 2003;42:1200–11.
- Kaizu Y, Ohkawa S, Odamaki M, Ikegaya N, Hibi I, Miyaji K, et al. Association between inflammatory mediators and muscle mass in long-term hemodialysis patients. Am J Kidney Dis. 2003;42:295–302.
- Kalantar-Zadeh K, Kopple JD, Block G, Humphreys MH. A malnutrition-inflammation score is correlated with morbidity and mortality in maintenance hemodialysis patients. Am J Kidney Dis. 2001;38:1251–63.
- 74. Stenvinkel P, Heimburger O, Paultre F, Diczfalusy U, Wang T, Berglund L, et al. Strong association between malnutrition, inflammation, and atherosclerosis in chronic renal failure. Kidney Int. 1999;55:1899–911.
- Stenvinkel P, Heimburger O, Lindholm B, Kaysen GA, Bergstrom J. Are there two types of malnutrition in chronic renal failure? Evidence for relationships between malnutrition, inflammation and atherosclerosis (MIA syndrome). Nephrol Dial Transplant. 2000;15:953–60.
- Chertow GM, Raggi P, Chasan-Taber S, Bommer J, Holzer H, Burke SK. Determinants of progressive vascular calcification in haemodialysis patients. Nephrol Dial Transplant. 2004;19:1489–96.
- 77. Ganesh SK, Stack AG, Levin NW, Hulbert-Shearon T, Port FK. Association of elevated serum PO(4), Ca x PO(4) product, and parathyroid hormone with cardiac mortality risk in chronic hemodialysis patients. J Am Soc Nephrol. 2001;12:2131–8.
- Savory J, Bertholf RL, Wills MR. Aluminium toxicity in chronic renal insufficiency. Clin Endocrinol Metab. 1985;14:681–702.
- 79. Malluche HH. Aluminium and bone disease in chronic renal failure. Nephrol Dial Transplant. 2002;17 Suppl 2:21–4.
- Delmez JA, Slatopolsky E. Hyperphosphatemia: its consequences and treatment in patients with chronic renal disease. Am J Kidney Dis. 1992;19:303–17.
- Chertow GM, Burke SK, Raggi P, Treat to Goal Working Group. Sevelamer attenuates the progression of coronary and aortic calcification in hemodialysis patients. Kidney Int. 2002;62:245–52.
- Block GA, Spiegel DM, Ehrlich J, Mehta R, Lindbergh J, Dreisbach A, et al. Effects of sevelamer and calcium on coronary artery calcification in patients new to hemodialysis. Kidney Int. 2005;68:1815–24.
- Suki WN, Zabaneh R, Cangiano JL, Reed J, Fischer D, Garrett L, et al. Effects of sevelamer and calcium-based phosphate binders on mortality in hemodialysis patients. Kidney Int. 2007;72:1130–7.

Chapter 17 Nutritional Concerns in Osteoporosis

Bess Dawson-Hughes

Key Points

- Adequate calcium intake, 1,000–1,200 mg per day, supports the preservation of bone mass in older adults.
- Higher calcium intake should be avoided because it adds no value and may increase risk of kidney stones and cardiovascular disease, although the evidence is inconsistent.
- Vitamin D lowers fracture risk by improving bone mass and lowering risk of falling. An intake of 800–1,000 IU per day is sufficient for most older adults.
- The magnitude of the risk reduction for falls and fractures with vitamin D supplementation is approximately 20 %.
- High, infrequent dosing with vitamin D should be avoided because it increases risk of falls and fractures.
- Lowering the acid load of the diet with alkaline salts of potassium has been shown to lower rates of bone resorption and, in some trials, to lower rates of bone loss, but the effect on fracture risk is unknown. There is some evidence that these salts improve muscle performance over the short term; long-term studies are needed to determine whether these favorable effects persist.

Keywords Calcium • Vitamin D • Acid–base balance • Potassium bicarbonate • Bone mass • Falls • Fractures • Nutrition

Introduction

Osteoporosis is defined as the state of reduced mass and architectural deterioration of bone that contribute to risk of fracture. The clinical outcome of ongoing bone loss is low-trauma fractures. The lifetime risk of a fracture is about 40 % [1] and one in four women will sustain a low-trauma fracture after age 50 [2]. There have been encouraging downward trends in risk of hip fracture over the last

B. Dawson-Hughes, M.D.

Bone Metabolism Laboratory, Jean Mayer USDA Human Nutrition Research Center on Aging, Tufts University School of Medicine, 711 Washington St., Boston, MA 02111, USA e-mail: bess.dawson-hughes@tufts.edu

15–20 years in the United States [3, 4] and elsewhere [5], but not apparently at other skeletal sites [4]. Any small declines in fracture rates, however, would not result in fewer total fractures because of changing demographics. In the United States, the population at highest risk for fracture, men and women aged 65 years and older, is expected to increase from 40 to 89 million between 2010 and 2050 [6]. As a result of this demographic shift, the health care cost of fractures in the United States, estimated at \$17 billion for the year 2005, is projected to be over \$25 billion in 2025 [7].

Many factors contribute to age-related bone loss and increased fracture risk. Of central importance is genetics. In a recent study in 2,716 female twins who had at least two bone mineral density (BMD) assessments more than 4 years apart (mean follow-up 9.7 years), Moayyeri et al. [8] estimated heritability of bone loss at different ages from age 40 to 80 years. At age 40–45 years, heritability of the changes in BMD averaged 39.9 % for the total hip, 46.4 % for the femoral neck, and 69.5 % for the lumbar spine. Heritability of bone loss declined with increasing age and by age 65 years, there was no heritability of the rates of bone loss at any of these sites. The lack of heritability of bone loss in older subjects implies a greater role for life style in preserving bone mass in this vulnerable population. Exercise is an important determinant of bone health. Intense exercise programs in the elderly can improve bone mass modestly, by 1–2 % over the course of a year. The most important impact of exercise, particularly exercise that improves strength and balance, is to lower the risk of falling [9–11], a major risk factor for fracture.

Many nutrients affect bone development and preservation. The most extensively studied of these are calcium and vitamin D. Vitamin D also has an important role in muscle function. An emerging area of interest to the research and clinical communities is the role that the acid–base balance of the diet plays in preserving bone and muscle mass and function in older adults. This chapter is focused on the relationship of calcium intake, vitamin D status, and the acid–base balance of the diet to bone and muscle mass and strength and to the important clinical outcomes of falls and fractures in older men and women.

Calcium

Physiology

Calcium has many functions, including an important role in intracellular signaling and vascular and neuromuscular function. It is also critical for the development and preservation of the skeleton. About 99 % of the calcium in the human body is in bone. Calcium is part of hydroxyapatite, the crystalline component of bone that lends bone stiffness and strength. In addition to its structural role, calcium intake affects bone mass in adults through its impact on the remodeling rate. An inadequate intake of calcium results in less calcium absorbed, a lower circulating ionized calcium concentration, and an increased secretion of parathyroid hormone (PTH), a potent bone-resorbing agent. Typically about 5 nmol (200 mg) of calcium is removed from the adult skeleton and replaced each day. Dietary or supplemental calcium at sufficiently high levels, such as 1,000 mg per day or more, lowers the bone remodeling rate by 10–15 % in older men and women and the degree of suppression appears to be dose-related [12]. The reduction in remodeling rate accounts for the increase in BMD that occurs in the first 12–18 months of treatment with calcium. A high remodeling rate is generally associated with a greater rate of bone loss. The bone remodeling rate increases with aging; however, treatment with calcium will lower the rate in older women to rates commonly observed in premenopausal women [13]. It will also significantly lower circulating levels of PTH.

Calcium is absorbed by active transport across the mucosal cells, a process promoted by the active metabolite of vitamin D, 1,25-dihydroxyvitamin D $[1,25(OH)_2D]$. It is also absorbed by passive transport, a process that occurs in proportion to the luminal:serosal concentration gradient. Some calcium may be absorbed by solvent drag in the setting of large intakes of sugar. Active transport is the dominant mode of absorption at low calcium intakes, and passive transport dominates at high intakes. Calcium absorption has an hereditary component, linked to alleles of the vitamin D receptor (VDR); this component is apparent at low but not high calcium intakes [14]. Calcium is absorbed at the greatest rate in the duodenum; however, because of its greater length and absorptive surface, most calcium absorption efficiency declines with aging [15]. This is thought to be due in part to the age-related decline in intestinal VDRs [16]. Additionally, vitamin D levels tend to decline with aging, making vitamin D deficiency a contributing cause.

Basis for Setting Calcium Intake Requirements

The 2011 Institute of Medicine (IOM) recommendations for calcium intake are based primarily on short-term calcium balance studies. Randomized calcium intervention trials with changes in rates of bone loss and fractures were used as supporting rather than primary evidence because of limited quantities of trial data.

Hunt and Johnson analyzed calcium balance data in 155 young adults with calcium intakes ranging from 415 to 1,740 mg per day [17]. In this analysis, neutral balance was predicted at an intake of 741 mg per day. Several meta-analyses of randomized, placebo-controlled trials have addressed whether supplemental calcium lowers fracture risk. The Shea meta-analysis [18] of trials in post-menopausal women found that calcium supplementation increased BMD at every measured site by 1–2 %. It did not, however, significantly lower risk of vertebral (RR 0.77 [95 % CI 0.54–1.09]) or nonvertebral fractures (RR 0.86 [CI 0.43–1.72]). The 15 trials included in this analysis administered doses in the range of 500–2,000 mg per day over periods ranging from 18 months to 4 years. A subsequent meta-analysis found no significant effect of calcium on risk of nonvertebral (RR 0.92 [95 % CI: 0.81, 1.05]) and a significant *increase* in risk of hip fracture on calcium (RR 1.64 [95 % CI: 1.02, 2.64]) [19]. A subsequent and third meta-analysis confirmed the significant *increase* in hip fracture risk (RR 1.50 [95 % CI: 1.06, 2.12]) [20]. Thus, the weight of the evidence is that supplemental calcium alone will improve BMD modestly, but it is not effective in preventing fractures.

Based largely on the Hunt and Johnson data described above [17], the IOM identified 800 mg as the estimated average requirement (EAR) for adults. From the EAR, the IOM derived the Recommended Dietary Allowances (RDAs) shown in Table 17.1 [21]. The Tolerable Upper Limit for calcium was set at 2,000 mg per day.

Table 17.1 Recommended calcium and vitamin D intak	s for older adults
Tuble 17.1 Recommended calculation and vitamini D maak	is for order dualts

	Men	Women
Calcium, mg/day		
Age 51–70	1,000	1,200
Age 71+	1,200	1,200
Vitamin D, IU/day		
Age 51–70	600	600
Age 71+	800	800

^aTable created with data from [21]

Safety

Calcium in amounts recommended has long been considered safe, but questions have been raised about the safety of high doses of calcium supplements. High intake from supplements has been associated with a 20 % increase in risk of kidney stones in women [22] but not in men [23]. In the Women's Health Initiative (WHI), treatment with 1,000 mg of calcium and 400 IU of vitamin D daily for 7 years was associated with an increased risk of kidney stones [24]. In contrast, high intakes of calcium from food sources may lower risk of kidney stones in men [23] and women [22].

Of more serious concern is the possibility that calcium supplementation may increase risk of cardiovascular disease. Surprisingly, in view of evidence that calcium lowers lipids levels [25], Bolland et al. reported in 2008 that healthy postmenopausal women treated with 1,000 mg of supplemental calcium, when compared with placebo, had significantly higher risk of myocardial infarction (RR 2.12 [95 % CI: 1.01, 4.47]) [26]. A subsequent meta-analysis by the same authors also identified an increased risk (OR 1.31 [95 % CI: 1.02, 1.67]) [27]. A recent large observational study found that calcium supplement use, particularly at higher doses, was associated with increased risk of cardiovascular disease mortality in men but not in women [28]. They found no association of dietary calcium intake with mortality in men or women. In the WHI, there was no increase in risk of cardiovascular disease with calcium and vitamin D [29]. Similarly, in 1,460 women, mean age 75 years, treatment with 1,200 mg per day of calcium, compared with placebo, for 4.5 years didn't increase risk of cardiovascular disease [30]. Thus, the question remains open.

Since there is no evidence that intake in excess of the RDA adds any benefit either in the general population or in patients being treated with pharmacotherapy for osteoporosis, and since the possibility for harm remains open, supplement use that takes total calcium intake substantially above the RDA should generally be avoided.

Calcium Recommendations

Calcium intake in older men and women in the U.S. averages about 750 mg per day [21], so the average gap between actual and recommended calcium intake is 250–450 mg per day. Increasing intake from foods sources is the first strategy. This can be achieved by adding 1–1.5 servings of a calcium-rich food such as dairy foods or fortified orange juice to the diet.

For those unable or unwilling to increase their calcium intake from food sources, a calcium supplement can be used to fill the gap. Few elders will need more than about 500 mg of supplemental calcium per day, and this can be taken in a single dose. For individuals who have unusually low calcium intakes from food, such as individuals who avoid dairy foods, higher supplement doses may be needed. When that dose exceeds 500 or 600 mg per day, the dose should be split for better absorption efficiency [31]. Several forms of calcium are available. The most commonly used supplement is calcium carbonate (40 % calcium by weight) and the second most common is calcium citrate (21 % by weight). Calcium from citrate is slightly better absorbed than calcium from carbonate; however, differences in absorbability are generally offset by differences in their calcium contents, so the amounts of calcium absorbed per gram of supplement are similar. With regard to timing of calcium supplements, calcium carbonate should be taken with a meal for better absorption [32], whereas citrate may be taken at any time.

Vitamin D

Physiology and Interactions with the Aging Process

Vitamin D is acquired mainly through synthesis in the skin after sun exposure. UVB rays stimulate the conversion of the precursor, 7-dehydrocholesterol, to vitamin D_3 . Vitamin D_2 and vitamin D_3 are absorbed from vegetable and plant foods, respectively. Their absorption requires the presence of bile acids and occurs in the proximal jejunum and the distal ileum by passive diffusion [33]. Few foods contain significant amounts of vitamin D and in periods when sun exposure is lacking or ineffective, many people will need supplements. The effectiveness of sun exposure is related to latitude, altitude, time of day, degree of skin pigmentation, sunscreen use, and other factors. At higher latitudes such as 42° North (Boston), sun exposure in the winter does not promote skin synthesis because the needed UVB rays don't reach earth's surface. Vitamin D status, assessed clinically by measuring the serum 25-hydroxyvitamin D (250HD) level, is commonly poor in the elderly for several reasons. Sun exposure produces less vitamin D in older adults than in young adults because the concentration of the precursor in the skin is lower [34]. Additionally, many elders avoid sun exposure because of concern about skin cancers. Absorption efficiency of ingested vitamin D does not appear to decline with aging [35].

Vitamin D affects fracture risk through its impact on muscle, balance, and risk of falling and through its effects on calcium homeostasis, bone mass, and strength. The means by which vitamin D insufficiency affects bone and muscle and influences fracture risk are illustrated in Fig. 17.1.

Muscle Performance, Balance, and Falls

The active form of vitamin D, $1,25(OH)_2D$, acts on muscle by binding to classical nuclear VDRs. Mice lacking the VDR have smaller muscle fibers throughout adult life [36]. VDRs in human muscle have been identified and been shown to decline as a function of aging [37]. However, it is not universally agreed that VDRs exist in human muscle [38]. In humans, severe vitamin D deficiency is characterized clinically by profound muscle weakness, particularly in proximal muscles, and by muscle pain and impaired gait [39, 40]. Even mild vitamin D insufficiency appears to influence muscle



Fig. 17.1 The pathway by which vitamin D influences fracture risk

performance in the general older population. In 4,100 ambulatory adults aged 60 years and older participating in The Third National Health and Nutrition Examination Survey (NHANES III), lower extremity muscle performance, measured as the 8-foot walk test and the repeated sit-to-stand test, was poorest in subjects with the lowest 25OHD levels and was progressively better at increasing 25OHD levels throughout and even beyond the upper end of the reference range [41]. A similar association was observed in a prospective cohort of older Dutch men and women [42]. In this study, however, performance reached its maximum at a 25OHD level of about 50 nmol/L. Randomized, controlled vitamin D intervention trials with muscle performance outcomes have presented a mixed picture. In a meta-analysis of these studies, Stockton and colleagues concluded that vitamin D had no significant effect on lower extremity muscle strength except in individuals with low starting serum 25OHD levels (<25 nmol/L) [43].

Poor balance is a known contributor to risk of falling. Balance in clinical trials has been measured by quantifying the degree of sway in the anterior-posterior and medial-lateral directions in subjects standing on a force platform. Two trials have evaluated the effect of vitamin D on sway. Both trials compared the effect of 800 IU of vitamin D₃ plus 1,000 mg of calcium per day with calcium alone in elderly adults. The vitamin D groups had an up to 28 % improvement (reduction) in body sway [44, 45] over periods of 2 and 12 months, when compared with the calcium alone groups. These studies implicate a role for vitamin D supplementation in improving balance in elders. This may be an important means by which vitamin D affects risk of falling.

The impact of vitamin D on risk of falling has been examined in several randomized controlled trials and in as many meta-analyses. In one meta-analysis, risk of falling was significantly reduced, by 20 %, in the trials administering 700–1,000 IU per day of vitamin D, whereas doses of 400 IU per day were ineffective [46]. In a different meta-analysis, the risk reduction with vitamin D was 17 % when compared with placebo [47]. In acute hip fracture patients, 2,000 IU of vitamin D₃ per day was no more or less effective than 800 IU per day in affecting fall rates [10]. A recent evaluation of a 500,000 IU dose given orally once per year to elders actually increased their risk of falling when compared with placebo [48]. Thus while further evidence is needed, it appears that intake in the range of 700–1,000 IU per day is the amount needed for maximal protection against falling and that large single doses should be avoided.

Vitamin D: Bone Mass and Fractures

In several cross-sectional studies, serum 25OHD levels are inversely associated with serum PTH levels and positively associated with BMD [49–51]. Moreover, in randomized, controlled clinical trials, supplementation with vitamin D has lowered rates of bone loss in older adults [52, 53], with most of the benefit occurring in the wintertime [53]. The magnitude of the reduction in bone loss is on the order of 1-2 % over a 1- to 2-year period, or approximately one-half of the usual loss. These BMD changes are too small to account for the significant impact that vitamin D has on fracture risk (to be described below); most of the anti-fracture benefit from vitamin D is thought to result from its improving strength and balance and lowering risk of falling.

Many randomized controlled trials have examined the effect of vitamin D with and without calcium on fracture rates in older men and women. Several investigators have performed meta-analyses of the trials, each focusing on slightly different questions and therefore applying different trial inclusion criteria. It's not surprising that they have reached somewhat different conclusions. An individual subject-level meta-analysis, while not free of shortcomings, allows much more analytical flexibility and refinement than a study-level meta-analysis. A subject-level meta-analysis utilizing age, gender, living arrangement (institutionalized or free-living), treatment assignment, dose of vitamin D, calcium intake, and compliance with administered supplements (placebo or vitamin D with or without calcium) of each participant was performed in 31,022 persons, age 65 years and older [54].



Fig. 17.2 Adapted from the individual subject-level meta-analysis of 11 randomized controlled trials to determine the impact of vitamin D on hip fracture risk [54]. The bars represent the relative risk of hip fracture by quartile of vitamin D intake in 31,022 participants, after adjustment for study, age group, sex, and type of dwelling

In the intent-to-treat analysis, participants assigned to treatment with vitamin D had a nonsignificant 10 % reduction in risk of hip fracture when compared with placebo (hazard ratio, 0.90; 95 % confidence interval [CI] 0.80–1.01) and a significant 7 % reduction in risk of nonvertebral fracture (hazard ratio 0.93 [CI] 0.76–0.96). The risk of hip fracture was also examined by quartile of total vitamin D intake, including intake from study supplements and outside supplements, as was allowed in several of the trials, including the WHI. There was a 30 % lower risk of hip fracture in the highest quartile of intake, 792–2,000 IU per day, median 800 IU per day, than in each of the other quartiles, after adjustment for study, age group, sex, and type of dwelling (Fig. 17.2). Hip fracture risk in the third quartile did not differ from the risk in the lower two quartiles. The highest quartile also had significantly lower risk of nonvertebral fracture (hazard ratio 0.86, [CI] 0.76-0.96). These results suggest that an intake of at least 800 IU per day is needed to lower fracture risk in men and women age 65 and older. This intake is in accord with the IOM recommended intake of 800 IU per day for men and women age 71 years and older, but higher than the intake of 600 IU recommended for persons age 50-70 years (see Table 17.1) [21]. Strengths of this subject-level meta-analysis are that it contains data from a large number of trials including the WHI; it takes account of compliance with study pills, which was as low as 50 % in one of the trials [55], and it accounts for vitamin D intake not only from study pills but also from personal supplement use during the trial, a practice that was allowed and widespread in the WHI and several other trials. Meta-analyses can be used to estimate the dose or 25OHD level needed to minimize fracture risk, but they can never be as precise as a single multiple-dose study with fracture as the primary outcome. A multiple-dose anti-fracture efficacy trial is not likely to be conducted, and we must therefore make the best possible interpretation of smaller, single-dose trials when seeking to define the optimal dose of vitamin D for musculoskeletal health. Another limitation of current trial data for defining the vitamin D requirement is that most of the higher dose vitamin D trials also administered calcium to the vitamin D group, so one cannot differentiate the effects of vitamin D alone from the combined effects of vitamin D and calcium.

Recommendations for Vitamin D Replacement

Based on available evidence, an intake of about 800 IU per day of vitamin D is sufficient to lower risk of falls and fractures in the general population of older men and women. Daily and weekly dosing give similar increases in serum 25OHD levels [56]. Monthly dosing, favored by some patients, is also effective, although the variability in increment in 25OHD is somewhat greater than it is with more frequent dosing [56]. Most now agree that vitamin D_3 is preferred over vitamin D_2 , because it gives a larger 25OHD increment. This, of course, can be offset by giving a higher dose of vitamin D_2 . With monthly administration of oral vitamin D_2 , the circulating 25OHD level starts to decline about 2 weeks after each dose, a phenomenon not seen with monthly administration of vitamin D_3 .

Acid–Base Balance of the Diet

The effects of the acid–base balance of the diet on bone and muscle, while less well studied than those of calcium and vitamin D, is of growing interest, and lowering the acid load of the diet may be a promising adjunct to more established nutritional preventive strategies. Aging is associated with a low-grade, progressive metabolic acidosis [57]. In young adults, the kidney responds to metabolic acidosis by increasing renal net acid excretion (NAE) to minimize perturbation in blood pH [58]. The gradual decline in renal function in older adults limits their capacity to excrete hydrogen ions [59]. The diet consumed by many people in the United States and elsewhere contributes to the acidosis. An acid-producing diet is one in which intake of alkali-producing fruits and vegetables is inadequate to neutralize the intake of acid-producing cereal grains and animal protein. Fruits and vegetables are metabolized to the alkaline bicarbonate and protein and cereal grains are metabolized to sulfuric, phytic, and other acids [60–62]. Figure 17.3 illustrates the distribution of NAE of values of 171 healthy older men and women enrolled in our recent bicarbonate trial (described below); 96 % of the subjects had positive NAE values, indicating that their usual diets were acid-producing.

There is extensive evidence that an acidic environment has negative effects on bone. It impairs osteoblast function [63–65] and increases bone resorption [66] through activation of a proton receptor on osteoclasts [67]. Acid also has a direct physicochemical effect on bone [68]. In several small metabolic studies, a single daily dose of alkali over a period of 1–3 weeks lowered biochemical markers of bone turnover [69, 70]. A meta-analysis of calcium balance studies found no evidence that the dietary acid load affected calcium balance [71]. In contrast, a recent randomized controlled trial found that 90 mmol per day of the alkaline salt, potassium citrate, significantly improved calcium balance over a 6-month period [72]. Several studies have described associations of diets rich in fruits and vegetables with higher BMD and/or lower rates of bone loss [73–77]. Three relatively small randomized controlled trials in older men and women have also been reported. In one, supplementation of postmenopausal women with 35 meq per day of an alkaline salt of potassium reduced rates of bone loss over 1 year [78]. In a 2-year trial with a larger dose of 60 meq per day in 201 older men and women, potassium citrate significantly lowered rates of bone loss, increased a biochemical marker of bone formation, and decreased a biochemical marker of bone resorption, when compared with placebo [79]. In a third trial, neither an alkaline salt of potassium nor fruit and vegetable intake significantly affected changes in BMD [80]. The reason why the latter trial had discordant results is not clear.

An acidic environment also has negative effects on muscle. Clinical states of chronic metabolic acidosis including starvation [81–83], trauma, sepsis and burns [84–87], and chronic renal failure are states of muscle wasting in humans [88]. In rats, acid loading promoted nitrogen excretion, an indicator of muscle wasting [89]. In a small metabolic study, daily treatment with an alkaline salt of potassium reduced nitrogen excretion in postmenopausal women on acid-producing high-protein diets, and discontinuation of the alkali was followed by an immediate increase in the urinary nitrogen excretion toward



Fig. 17.3 The distribution of net acid excretion (NAE) values of 171 healthy men and women age 50 years and older with normal renal function. Ninety-six percent of the subjects had positive NAE values, indicating that they were consuming acid-producing diets. Vegetarians were excluded, but there were no other dietary restrictions in these subjects

its baseline level [62]. The authors calculated that the bicarbonate-induced decline in nitrogen wasting, if sustained, would be sufficient to offset the decline in muscle mass that occurs with aging (64 g of nitrogen or 1.0 kg of lean body mass per decade). In our recent 3-month trial, administration of 60 meq per day of potassium bicarbonate to healthy older men and women significantly lowered nitrogen excretion and, in the women, improved lower extremity muscle power. Longer-term studies are needed to determine whether these favorable effects persist and result in improved muscle mass and performance. Treatment with alkaline salts of potassium (or preferably increasing the intake of fruits and vegetables) warrants further evaluation as a potential risk modifier of age-related losses in both bone and muscle. Even small gains in prevention would have large consequences for the elderly population.

Conclusion

A calcium intake of 1,000–1,200 mg per day is needed to support the preservation of bone mass in older adults. Higher intakes offer no added benefit, may increase risk of kidney stones, and should be avoided. A vitamin D intake of 800–1,000 IU per day lowers risk of falls and fractures in older adults who are vitamin D insufficient. High, infrequent dosing with vitamin D should be avoided because it increases risk of falls and fractures. There is evidence that supplementation with alkaline salts of potassium (potassium citrate or bicarbonate) improves muscle performance and calcium balance over the short term. Long-term intervention studies are needed to determine whether these favorable effects persist.

Disclosure This material is based on work supported by the U.S. Department of Agriculture, under agreement number 58-1950-7-707. Any opinions, findings, conclusion, or recommendations expressed in this publication are those of the authors and do not necessarily reflect the views of the U.S. Department of Agriculture.

References

- 1. Kanis JA. Diagnosis of osteoporosis and assessment of fracture risk. Lancet. 2002;359(9321):1929-36.
- National Osteoporosis Foundation. America's bone health: the state of osteoporosis and low bone mass in our nation. Washington, DC: National Osteoporosis Foundation; 2002.
- Brauer CA, Coca-Perraillon M, Cutler DM, Rosen AB. Incidence and mortality of hip fractures in the United States. JAMA. 2009;302(14):1573–9.
- Wright NC, Saag KG, Curtis JR, Smith WK, Kilgore ML, Morrisey MA, et al. Recent trends in hip fracture rates by race/ethnicity among older US adults. J Bone Miner Res. 2012;27(11):2325–32.
- Leslie WD, O'Donnell S, Jean S, Lagace C, Walsh P, Bancej C, et al. Trends in hip fracture rates in Canada. JAMA. 2009;302(8):883–9.
- Projections of the population by selected age groups and sex for the United States, 2020 to 2050. Washington, DC: US Census Bureau; 2008.
- Burge R, Dawson-Hughes B, Solomon DH, Wong JB, King A, Tosteson A. Incidence and economic burden of osteoporosis-related fractures in the United States, 2005–2025. J Bone Miner Res. 2007;22(3):465–75.
- Moayyeri A, Hammond CJ, Hart DJ, Spector TD. Effects of age on genetic influence on bone loss over 17 years in women: the Healthy Ageing Twin Study (HATS). J Bone Miner Res. 2012;27(10):2170–8.
- Campbell AJ, Robertson MC, Gardner MM, Norton RN, Tilyard MW, Buchner DM. Randomised controlled trial of a general practice programme of home based exercise to prevent falls in elderly women. [see comment]. BMJ. 1997;315(7115):1065–9.
- Bischoff-Ferrari HA, Dawson-Hughes B, Platz A, Orav EJ, Stahelin HB, Willett WC, et al. Effect of high-dosage cholecalciferol and extended physiotherapy on complications after hip fracture: a randomized controlled trial. Arch Intern Med. 2010;170(9):813–20.
- Wolf SL, Barnhart HX, Kutner NG, McNeely E, Coogler C, Xu T. Reducing frailty and falls in older persons: an investigation of Tai Chi and computerized balance training. Atlanta FICSIT Group. Frailty and injuries: cooperative studies of intervention techniques. J Am Geriatr Soc. 1996;44(5):489–97.
- Elders PJ, Lips P, Netelenbos JC, van Ginkel FC, Khoe E, van der Vijgh WJ, et al. Long-term effect of calcium supplementation on bone loss in perimenopausal women. J Bone Miner Res. 1994;9(7):963–70.
- McKane WR, Khosla S, Egan KS, Robins SP, Burritt MF, Riggs BL. Role of calcium intake in modulating agerelated increases in parathyroid function and bone resorption. J Clin Endocrinol Metab. 1996;81(5):1699–703.
- 14. Dawson-Hughes B, Harris SS, Finneran S. Calcium absorption on high and low calcium intakes in relation to vitamin D receptor genotype. J Clin Endocrinol Metab. 1995;80(12):3657–61.
- Bullamore JR, Wilkinson R, Gallagher JC, Nordin BE, Marshall DH. Effect of age on calcium absorption. Lancet. 1970;2(7672):535–7.
- Ebeling PR, Sandgren ME, DiMagno EP, Lane AW, DeLuca HF, Riggs BL. Evidence of an age-related decrease in intestinal responsiveness to vitamin D: relationship between serum 1,25-dihydroxyvitamin D3 and intestinal vitamin D receptor concentrations in normal women. J Clin Endocrinol Metab. 1992;75(1):176–82.
- Hunt CD, Johnson LK. Calcium requirements: new estimations for men and women by cross-sectional statistical analyses of calcium balance data from metabolic studies. Am J Clin Nutr. 2007;86(4):1054–63.
- Shea B, Wells G, Cranney A, Zytaruk N, Robinson V, Griffith L, et al. VII. Meta-analysis of calcium supplementation for the prevention of postmenopausal osteoporosis. Endocr Rev. 2002;23(4):552–9.
- Bischoff-Ferrari HA, Dawson-Hughes B, Baron JA, Burckhardt P, Li R, Spiegelman D, et al. Calcium intake and hip fracture risk in men and women: a meta-analysis of prospective cohort studies and randomized controlled trials. Am J Clin Nutr. 2007;86(6):1780–90.
- 20. Reid IR, Bolland MJ, Grey A. Effect of calcium supplementation on hip fractures. Osteoporos Int. 2008; 19(8):1119–23.
- 21. IOM. Dietary reference intakes for calcium and vitamin D. Washington, DC: IOM; 2011.
- Curhan GC, Willett WC, Speizer FE, Spiegelman D, Stampfer MJ. Comparison of dietary calcium with supplemental calcium and other nutrients as factors affecting the risk for kidney stones in women. [see comments]. Ann Intern Med. 1997;126(7):497–504.
- Curhan GC, Willett WC, Rimm EB, Stampfer MJ. A prospective study of dietary calcium and other nutrients and the risk of symptomatic kidney stones. [see comments]. N Engl J Med. 1993;328(12):833–8.

- Wallace RB, Wactawski-Wende J, O'Sullivan MJ, Larson JC, Cochrane B, Gass M, et al. Urinary tract stone occurrence in the Women's Health Initiative (WHI) randomized clinical trial of calcium and vitamin D supplements. Am J Clin Nutr. 2011;94(1):270–7.
- Reid IR, Ames R, Mason B, Bolland MJ, Bacon CJ, Reid HE, et al. Effects of calcium supplementation on lipids, blood pressure, and body composition in healthy older men: a randomized controlled trial. Am J Clin Nutr. 2010;91(1):131–9.
- Bolland MJ, Barber PA, Doughty RN, Mason B, Horne A, Ames R, et al. Vascular events in healthy older women receiving calcium supplementation: randomised controlled trial. BMJ. 2008;336(7638):262–6.
- Bolland MJ, Avenell A, Baron JA, Grey A, MacLennan GS, Gamble GD, et al. Effect of calcium supplements on risk of myocardial infarction and cardiovascular events: meta-analysis. BMJ. 2010;341:c3691.
- Xiao Q, Murphy RA, Houston DK, Harris TB, Chow WH, Park Y. Dietary and supplemental calcium intake and cardiovascular disease mortality: the National Institutes of Health-AARP Diet and Health Study. JAMA Intern Med. 2013;4:1–8.
- Hsia J, Heiss G, Ren H, Allison M, Dolan NC, Greenland P, et al. Calcium/vitamin D supplementation and cardiovascular events. Circulation. 2007;115(7):846–54.
- Lewis JR, Calver J, Zhu K, Flicker L, Prince RL. Calcium supplementation and the risks of atherosclerotic vascular disease in older women: results of a 5-year RCT and a 4.5-year follow-up. J Bone Miner Res. 2011;26(1):35–41.
- Harvey JA, Zobitz MM, Pak CY. Dose dependency of calcium absorption: a comparison of calcium carbonate and calcium citrate. J Bone Miner Res. 1988;3(3):253–8.
- 32. Recker RR. Calcium absorption and achlorhydria. N Engl J Med. 1985;313(2):70-3.
- Hollander D, Muralidhara KS, Zimmerman A. Vitamin D-3 intestinal absorption in vivo: influence of fatty acids, bile salts, and perfusate pH on absorption. Gut. 1978;19(4):267–72.
- MacLaughlin J, Holick MF. Aging decreases the capacity of human skin to produce vitamin D3. J Clin Invest. 1985;76(4):1536–8.
- Harris SS, Dawson-Hughes B. Plasma vitamin D and 25OHD responses of young and old men to supplementation with vitamin D3. J Am Coll Nutr. 2002;21(4):357–62.
- 36. Endo I, Inoue D, Mitsui T, Umaki Y, Akaike M, Yoshizawa T, et al. Deletion of vitamin D receptor gene in mice results in abnormal skeletal muscle development with deregulated expression of myoregulatory transcription factors. Endocrinology. 2003;144(12):5138–44.
- 37. Bischoff-Ferrari HA, Borchers M, Gudat F, Durmuller U, Stahelin HB, Dick W. Vitamin D receptor expression in human muscle tissue decreases with age. J Bone Miner Res. 2004;19(2):265–9.
- 38. Wang Y, DeLuca HF. Is the vitamin D receptor found in muscle? Endocrinology. 2011;152(2):354-63.
- Glerup H, Mikkelsen K, Poulsen L, Hass E, Overbeck S, Andersen H, et al. Hypovitaminosis D myopathy without biochemical signs of osteomalacic bone involvement. Calcif Tissue Int. 2000;66(6):419–24.
- 40. Schott GD, Wills MR. Muscle weakness in osteomalacia. Lancet. 1976;1(7960):626-9.
- Bischoff-Ferrari HA, Dietrich T, Orav EJ, Zhang Y, Karlson EW, Dawson-Hughes B. Higher 25-hydroxyvitamin D levels are associated with better lower extremity function in both active and inactive adults 60+ years of age. Am J Clin Nutr. 2004;80:752–8.
- 42. Wicherts IS, van Schoor NM, Boeke AJ, Visser M, Deeg DJ, Smit J, et al. Vitamin D status predicts physical performance and its decline in older persons. J Clin Endocrinol Metab. 2007;92:2058–65.
- Stockton KA, Mengersen K, Paratz JD, Kandiah D, Bennell KL. Effect of vitamin D supplementation on muscle strength: a systematic review and meta-analysis. Osteoporos Int. 2011;22(3):859–71.
- 44. Pfeifer M, Begerow B, Minne HW, Abrams C, Nachtigall D, Hansen C. Effects of a short-term vitamin D and calcium supplementation on body sway and secondary hyperparathyroidism in elderly women. J Bone Miner Res. 2000;15:1113–8.
- 45. Pfeifer M, Begerow B, Minne HW, Suppan K, Fahrleitner-Pammer A, Dobnig H. Effects of a long-term vitamin D and calcium supplementation on falls and parameters of muscle function in community-dwelling older individuals. Osteoporos Int. 2009;20(2):315–22.
- 46. Bischoff-Ferrari HA, Dawson-Hughes B, Staehelin HB, Orav EJ, Stuck AE, Theiler R, et al. Fall prevention with supplemental and alpha-hydroxylated vitamin D: a meta-analysis of randomized controlled trials. BMJ. 2009; 339:b3692.
- Murad MH, Elamin KB, Abu Elnour NO, Elamin MB, Alkatib AA, Fatourechi MM, et al. The effect of vitamin D on falls: a systematic review and meta-analysis. J Clin Endocrinol Metab. 2011;96(10):2997–3006.
- 48. Sanders KM, Stuart AL, Williamson EJ, Simpson JA, Kotowicz MA, Young D, et al. Annual high-dose oral vitamin D and falls and fractures in older women: a randomized controlled trial. JAMA. 2010;303(18):1815–22.
- Steingrimsdottir L, Gunnarsson O, Indridason OS, Franzson L, Sigurdsson G. Relationship between serum parathyroid hormone levels, vitamin D sufficiency, and calcium intake. [see comment]. JAMA. 2005;294(18):2336–41.
- 50. Bischoff-Ferrari HA, Kiel DP, Dawson-Hughes B, Orav JE, Li R, Spiegelman D, et al. Dietary calcium and serum 25-hydroxyvitamin D status in relation to BMD among U.S. adults. J Bone Miner Res. 2009;24(5):935–42.

- 51. Joo NS, Dawson-Hughes B, Kim YS, Oh K, Yeum KJ. Impact of calcium and vitamin D insufficiencies on serum parathyroid hormone and bone mineral density: analysis of the fourth and fifth Korea National Health and Nutrition Examination Survey (KNHANES IV-3, 2009 and V-1, 2010). J Bone Miner Res. 2012;28(4):764–70.
- 52. Ooms ME, Lips P, Roos JC, van der Vijgh WJ, Popp-Snijders C, Bezemer PD, et al. Vitamin D status and sex hormone binding globulin: determinants of bone turnover and bone mineral density in elderly women. J Bone Miner Res. 1995;10(8):1177–84.
- 53. Dawson-Hughes B, Dallal GE, Krall EA, Harris S, Sokoll LJ, Falconer G. Effect of vitamin D supplementation on wintertime and overall bone loss in healthy postmenopausal women. Ann Intern Med. 1991;115(7):505–12.
- Bischoff-Ferrari HA, Willett WC, Orav EJ, Lips P, Meunier PJ, Lyons RA, et al. A pooled analysis of vitamin D dose requirements for fracture prevention. N Engl J Med. 2012;367(1):40–9.
- 55. Grant AM, Avenell A, Campbell MK, McDonald AM, MacLennan GS, McPherson GC, et al. Oral vitamin D3 and calcium for secondary prevention of low-trauma fractures in elderly people (Randomised Evaluation of Calcium Or vitamin D, RECORD): a randomised placebo-controlled trial. Lancet. 2005;365(9471):1621–8.
- 56. Ish-Shalom S, Segal E, Salganik T, Raz B, Bromberg IL, Vieth R. Comparison of daily, weekly, and monthly vitamin D3 in ethanol dosing protocols for two months in elderly hip fracture patients. J Clin Endocrinol Metab. 2008;93(9):3430–5.
- Frassetto LA, Morris Jr RC, Sebastian A. Effect of age on blood acid-base composition in adult humans: role of age-related renal functional decline. Am J Physiol. 1996;271(6 Pt 2):F1114–22.
- Kurtz I, Maher T, Hulter HN, Schambelan M, Sebastian A. Effect of diet on plasma acid-base composition in normal humans. Kidney Int. 1983;24(5):670–80.
- 59. Lindeman RD, Tobin J, Shock NW. Longitudinal studies on the rate of decline in renal function with age. J Am Geriatr Soc. 1985;33(4):278–85.
- 60. Remer T, Manz F. Potential renal acid load of foods and its influence on urine pH. J Am Diet Assoc. 1995; 95(7):791–7.
- Frassetto LA, Todd KM, Morris Jr RC, Sebastian A. Estimation of net endogenous noncarbonic acid production in humans from diet potassium and protein contents. Am J Clin Nutr. 1998;68(3):576–83.
- Frassetto L, Morris Jr RC, Sebastian A. Potassium bicarbonate reduces urinary nitrogen excretion in postmenopausal women. J Clin Endocrinol Metab. 1997;82(1):254–9.
- Ludwig MG, Vanek M, Guerini D, Gasser JA, Jones CE, Junker U, et al. Proton-sensing G-protein-coupled receptors. Nature. 2003;425(6953):93–8.
- Tomura H, Mogi C, Sato K, Okajima F. Proton-sensing and lysolipid-sensitive G-protein-coupled receptors: a novel type of multi-functional receptors. Cell Signal. 2005;17(12):1466–76.
- Frick KK, Krieger NS, Nehrke K, Bushinsky DA. Metabolic acidosis increases intracellular calcium in bone cells through activation of the proton receptor OGR1. J Bone Miner Res. 2009;24:305–13.
- 66. Arnett TR, Dempster DW. Effect of pH on bone resorption by rat osteoclasts in vitro. Endocrinology. 1986; 119(1):119-24.
- Komarova SV, Pereverzev A, Shum JW, Sims SM, Dixon SJ. Convergent signaling by acidosis and receptor activator of NF-kappaB ligand (RANKL) on the calcium/calcineurin/NFAT pathway in osteoclasts. Proc Natl Acad Sci U S A. 2005;102(7):2643–8.
- Bushinsky DA. Metabolic alkalosis decreases bone calcium efflux by suppressing osteoclasts and stimulating osteoblasts. Am J Physiol. 1996;271(1 Pt 2):F216–22.
- 69. Sebastian A, Morris Jr RC. Improved mineral balance and skeletal metabolism in postmenopausal women treated with potassium bicarbonate. N Engl J Med. 1994;331(4):279.
- Maurer M, Riesen W, Muser J, Hulter HN, Krapf R. Neutralization of Western diet inhibits bone resorption independently of K intake and reduces cortisol secretion in humans. Am J Physiol Renal Physiol. 2003;284(1): F32–40.
- Fenton TR, Lyon AW, Eliasziw M, Tough SC, Hanley DA. Meta-analysis of the effect of the acid-ash hypothesis of osteoporosis on calcium balance. J Bone Miner Res. 2009;24(11):1835–40.
- 72. Moseley K, Weaver C, Appel L, Sebastian A, Sellmeyer DE. Potassium citrate supplementation results in sustained improvement in calcium balance in older men and women. J Bone Miner Res. 2012. doi:10.1002/jbmr.1764.
- Tucker KL, Chen H, Hannan MT, Cupples LA, Wilson PW, Felson D, et al. Bone mineral density and dietary patterns in older adults: the Framingham Osteoporosis Study. Am J Clin Nutr. 2002;76(1):245–52.
- Jones G, Riley MD, Whiting S. Association between urinary potassium, urinary sodium, current diet, and bone density in prepubertal children. Am J Clin Nutr. 2001;73(4):839–44.
- 75. New SA, Bolton-Smith C, Grubb DA, Reid DM. Nutritional influences on bone mineral density: a cross-sectional study in premenopausal women. Am J Clin Nutr. 1997;65(6):1831–9.
- Chen Y, Ho SC, Lee R, Lam S, Woo J. Fruit intake is associated with better bone mass among Hong Kong Chinese early postmenopausal women. J Bone Miner Res. 2001;16 Suppl 1:S386.

- 77. MacDonald HM, New SA, Golden MHN, Campbell MK, Reid DM. Nutritional associations with bone loss during the menopausal transition: evidence of a beneficial effect of calcium, alcohol, and fruit and vegetable nutrients and of a detrimental effect of fatty acids. Am J Clin Nutr. 2004;79(1):155–65.
- Jehle S, Zanetti A, Muser J, Hulter HN, Krapf R. Partial neutralization of the acidogenic Western diet with potassium citrate increases bone mass in postmenopausal women with osteopenia. J Am Soc Nephrol. 2006;17(11): 3213–22.
- 79. Jehle S, Hulter HN, Krapf R. Effect of potassium citrate on bone density, microarchitecture, and fracture risk in healthy older adults without osteoporosis: a randomized controlled trial. J Clin Endocrinol Metab. 2013;98(1): 207–17.
- Macdonald HM, Black AJ, Sandison R, Aucott L, Hardcastle AJ, Lanham-New SA, Fraser WD, Reid DM. Two year double blind randomized controlled trial in postmenopausal women shows no gain in BMD with potassium citrate treatment. J Bone Miner Res. 2006;21 Suppl 1:S15.
- Ruderman NB, Berger M. The formation of glutamine and alanine in skeletal muscle. J Biol Chem. 1974; 249(17):5500–6.
- 82. Cahill Jr GF. Starvation in man. N Engl J Med. 1970;282(12):668-75.
- Owen EE, Robinson RR. Amino acid extraction and ammonia metabolism by the human kidney during the prolonged administration of ammonium chloride. J Clin Invest. 1963;42:263–76.
- Aulick LH, Wilmore DW. Increased peripheral amino acid release following burn injury. Surgery. 1979;85(5):560–5.
- Askanazi J, Carpentier YA, Michelsen CB, Elwyn DH, Furst P, Kantrowitz LR, et al. Muscle and plasma amino acids following injury. Influence of intercurrent infection. Ann Surg. 1980;192(1):78–85.
- Souba WW, Smith RJ, Wilmore DW. Glutamine metabolism by the intestinal tract. JPEN J Parenter Enteral Nutr. 1985;9(5):608–17.
- Williamson DH. Muscle protein degradation and amino acid metabolism in human injury. Biochem Soc Trans. 1980;8(5):497.
- Garibotto G, Russo R, Sofia A, Sala MR, Sabatino C, Moscatelli P, et al. Muscle protein turnover in chronic renal failure patients with metabolic acidosis or normal acid-base balance. Miner Electrolyte Metab. 1996;22(1–3): 58–61.
- Williams B, Layward E, Walls J. Skeletal muscle degradation and nitrogen wasting in rats with chronic metabolic acidosis. Clin Sci. 1991;80(5):457–62.

Chapter 18 Dementia-Related Mealtime Difficulties: Assessment and Management in the Long-Term Care Setting

Melissa Batchelor-Aselage, Elaine J. Amella, Sarah Broome Rose, and Connie Watkins Bales

Key Points

- Mealtime difficulties in persons with dementia can lead to chronic under-nutrition, with profound effects on both physical health and quality of life.
- Mealtime difficulties related to dementia are best dealt with in the clinical setting by an interdisciplinary team approach, where clear communication among team members can promote effective care management.
- The Need-Driven Dementia Compromised Model provides a problem-solving framework that prompts caregivers to examine behaviors through the lens of background factors (e.g., lifetime habits and routines, physical health, physiological factors) and adaptable proximal factors (e.g., caregiver interactions, environmental factors) in order to look for modifiable factors.
- Careful hand feeding is a viable alternative to percutaneous gastrostomy feeding (PEG) and is often the preferable intervention to support oral intake of food and fluids in advanced dementia.

Keywords Mealtime difficulties • Dementia • Long-term care • Feeding

E.J. Amella, Ph.D., R.N., F.A.A.N. College of Nursing, Medical University of South Carolina, 99 Jonathan Lucas Street, MSC-160, Charleston, SC 29425, USA e-mail: amellaej@musc.edu

S.B. Rose, M.S., R.D. Department of Molecular and Medical Genetics, Oregon Health and Science University, 3181 SW Sam Jackson Park Road, L103A, Portland, OR 97239, USA e-mail: rosesarahb@gmail.com

C.W. Bales, Ph.D., R.D. Durham VA Medical Center and Duke University Medical Center, Box 3003, Durham, NC 27710, USA e-mail: bales001@mc.duke.edu

M. Batchelor-Aselage, Ph.D., R.N.-B.C., F.N.P.-B.C. (⊠) School of Nursing, Duke University, DUMC 3322 307 Trent Drive, Durham, NC 27710, USA e-mail: melissa.aselage@duke.edu

Significance and Impact of the Problem

Background

By the year 2050, the number of persons with dementia aged 65 and older in the United States is expected to increase from 5.4 to 16 million persons [1]. While most persons with dementia (PWD) prefer to be cared for in their home environment, as the disease causes progressive functional and behavioral decline many will require institutional care. For these PWD, this means placement in assisted-living facilities or skilled nursing homes (NH). The number of PWD in the NH setting is expected to double to three million by the year 2050 [1]. While NHs attempt to address the multiple needs of their residents, the very basic need for adequate nutrition to promote health may be compromised by frailty, chronic illness, and cognitive decline. In fact, the ability to feed oneself is the last activity of daily living (ADL) to be lost; this is especially true in persons with Alzheimer's disease and related dementias. The adverse impact of chronic under-nutrition due to ongoing mealtime difficulties is profoundly felt on both physical health and quality of life for the PWD [2]. This chapter focuses on this frequently neglected topic, exploring both the assessment and management of mealtime difficulties in PWD and offering some innovative and practical strategies for managing such difficulties.

Impact of Mealtime Difficulties

In the institutional setting, complaints about mealtimes are common, ranging from noisy and cluttered meal environments and over-burdened staff, to restricted diets that appear and smell unappetizing and that may not be accepted by residents. Within this context, dining may be transformed into more of a task to be accomplished by the staff who assist, rather than a process to be facilitated and enjoyed. As a result, staff and resident may lose an opportunity for social interaction, which is often missing within institutional settings [3].

Adding to the concern regarding this high-risk mealtime scenario is the extreme vulnerability of PWD. They are often unable to express their needs due to a loss of language ability and have only their behaviors as a means to express their needs to caregivers. The behaviors, such as turning away of the head, clamping the mouth shut, and/or physically pushing away feeding assistance, are frequently thought of as "resistant" or "aversive" when, in fact, they may be the only means of communication available to the PWD. When these behaviors are interpreted as "resistive" or "aversive," feeding assistance attempts may stop, leading to malnutrition and dehydration, and potentially contribute to an earlier death if not acknowledged and appropriate interventions taken.

In NHs across the country, malnutrition, dehydration, and weight loss due to chronic poor oral intake remains one of the largest, silent epidemics [4]. A major risk factor for experiencing nutritional problems is cognitive impairment (dementia), and approximately 65 % of NH residents have documented moderate to severe dementia [5]. When a PWD loses the ability to self-feed adequately, meal-time support from NH staff becomes critical for their nutritional health. In 2013, the national average of the "percentage of long-stay residents who lose too much weight" was 7.5 %, a rate higher than falls (3.3 %) and pressure ulcers (6.4 %) [2]. Yet, under-nutrition due to mealtime difficulties is a largely preventable, reversible problem.

Seeking a Solution

While several members of the health care team may play important roles in dealing with feeding issues in PWD in other settings (see subsequent section), in the long-term care (LTC) setting it is the nursing staff that interacts most directly with patients during mealtimes and it is they who have the best opportunity to assess and address mealtime challenges. However, current training programs for direct care workers, certified nursing assistants and licensed nursing (cumulatively referred to as nursing staff), do not include content on dementia feeding skills [6, 7]. Certified nursing assistants (CNAs) typically provide about 90 % of feeding assistance to PWD, and interventions used are learned "on the job" and are typically non-evidence-based [7, 8]. Recent work provided evidence that when a group of nursing staff observe the identical mealtime experience, they frequently develop quite different interpretations of the meal and interaction: for example, believing a resident was refusing food halted one staff member's attempts, while another staff member who saw the same resident's repetitive behaviors as a misunderstanding of eating was able to alter the routine and achieve a successful interaction during which food was consumed [9]. In 2003, the first nursing clinical practice guidelines for dealing with mealtime difficulties were published, and they were updated in 2008 and 2012 [10, 11]. In 2010, the American Medical Directors Association (AMDA) published clinical practice guidelines for altered nutritional status in the LTC setting [12]. In subsequent sections, we will provide recommendations to foster clinical integration of various disciplinary approaches and offer mealtime interventions that are personalized to individual PWD and thus most likely to be effective.

Causes of Mealtime Difficulty in Persons with Dementia

Underlying causes of mealtime difficulties can be framed through the lens of problems with the place (environment), with the people (caregiver interactions), and with the person (with dementia) [13]. We will use this framework that is more easily taught to caregivers, illustrated in Table 18.1, to address

Person	People	Place
Physiologic: Appetite/food intake regulation Oral health problems Dysphagia or swallowing issues Sensory impairments Altered absorption or digestion Acute and chronic diseases, associated therapies Constipation Pain Difficulty with self-feeding (physical or neurological disability) Dependence on enteral or parenteral feeding	Poor continuity of care during transitions Outside staff unaware of usual needs/patterns Inadequate oral care provided Sensory equipment not provided or not functioning Nonfunctional positioning; left in bed for meals	Poor quality of diet offered Unappetizing dining environment (odors, sounds) Disruption of meal routines Mismanagement of snack/ med-pass selections Medication mismanagement Adaptive equipment not provided Inadequate lighting
<i>Psychological:</i> History of poor nutrition habits Depression Dementia Other cognitive impairments Long-standing emotional or mental illness	Lack of family involvement Burn-out and/or compassion fatigue of families and staff who are tired and burdened with lack of resources, support	Social interaction at meals de-emphasized Inadequate staffing Lack of involvement of families or significant others at meals (could be not welcomed by agency) Distractions—noise, television

 Table 18.1
 Precipitators of mealtime difficulties for persons with dementia

complex, yet common, mealtime challenges encountered, including broader environmental considerations and caregiver behaviors and how they may be interpreted by PWD. This multilayered approach has been supported in a model introduced in the mid-1990s that shifted the thinking of care for PWD from caregiver focused to providing person-centered care for this vulnerable population, the Need-Driven Dementia Compromised (NDB) Model [14]. When a dementia-related behavior is exhibited, the underlying premise is that the PWD is attempting to communicate an "unmet" need. The NDB Model provides a problem-solving framework by prompting caregivers to examine the behavior through the lens of more stable, background factors (e.g., lifetime habits and routines, physical health, physiological factors) and adaptable proximal factors (e.g., caregiver interactions, environmental factors). By carefully examining background (change the person) and proximal factors (change the place and/or people) that may contribute to or be antecedents to "problematic" behaviors, caregivers can look for a reversible or modifiable factor to alleviate dementia-related mealtime behaviors.

The Interdependent Roles of the Caregiving Team

The study of mealtime difficulties related to dementia has increased in prevalence over the past 30 years [15]. As this clinical condition is observed by many members of the interdisciplinary team, each discipline is able to focus on alleviating mealtime difficulties from different perspectives. When there is clear communication among team members, this creates the best possible condition for effective care management.

Physician/Nurse Practitioner/Physician Assistant

The key role of these health care providers in minimizing the impact of mealtime difficulties is to address modifiable issues that may be interfering with food and fluid intake. Examples include treatment of acute illness, addressing oral and/or swallowing problems, and adjustment of medications that might interfere with appetite or ability to optimally participate in a meal. These team members may also prescribe/provide oral high-density protein supplements or appetite stimulant agents; the former has been found effective in a systematic review while the latter have questionable efficacy in this population [16].

Dietitians and Dietary Managers

Patients with cognitive decline frequently have poor nutrient intake and weight loss associated with increased morbidity and mortality [17]. The Registered Dietitian (RD) plays an integral role in the nutritional care of elderly PWD in a variety of areas, including oversight of the overall nutritional adequacy of the meals and supplements being provided, whether orally or enterally [18]. The RD is responsible for menu planning that balances micro- and macronutrients, provides the appropriate amount of calories, and allows for appropriate modifications for therapeutically modified diet and texture prescriptions [19]. Often the RD acts as an advocate for the resident so as to tailor severe dietary restrictions to better match individual preferences, especially when life expectancy is not long [20]. Additionally, RDs are responsible for the evaluation of the nutritional status of all residents, including PWD, using anthropometric and clinical measures. Quick nutrition intervention when there are changes in these status indicators can lessen negative trajectories and improve quality

of life for these residents. The RD is thus a key contributor to the management of comorbidities, being responsible for the development of revised nutrition care plans related to new medical issues that may arise (e.g., pressure ulcers, recent weight loss) and for discussing end-of-life issues surrounding enteral nutrition with the family caregivers. It is common for PWD to be receiving enteral nutrition by tube feeding; the RD is responsible not only for the composition of these feedings but also may provide a careful discussion of these decisions with family members and the interdisciplinary care team [18].

Of utmost importance is the need for collaboration and coordination of care among all members of the care team. In some LTC settings, especially smaller institutions, RDs are not available on a full time basis but work as consultants, visiting the facilities on a part time basis. So it is especially important for RDs to work closely with all the other members of the team, especially those who have direct daily contact with the resident. The RD works with Certified Dietary Managers to plan and implement dietary interventions for residents that are tailored to their needs, as prescribed by the physician, and to plan/implement prescribed meals that are visually appealing and with appetizing aromas. Keller et al. [21] found that allocating additional RD time to residents with dementia, while further individualizing their meal plans, resulted in positive outcomes in relation to weight maintenance.

Nursing

Nursing's role on the team includes coordinating each team member's contribution for problemsolving, inclusive of Registered Nurses (RNs), and through other levels of professional nursing. Licensed Practical Nurses (LPNs) and CNAs monitor meal intake and changes in condition on a daily basis, monitor bowel and bladder contributing factors, provide the direct feeding assistance required for meals, and monitor environmental considerations that may impact meal intake. Additionally, they may have the greatest contact with family or friends who could be welcomed at mealtimes and their assistance of the PWD could be tailored through teaching them to use the Change the Person, People and Place model [12].

Consultant Therapists

Social Worker

This specialist is consulted when depression or other affective impediments to adequate dietary intake are a concern. As a liaison to friends and family, they can obtain critical information about past preferences, practices, and cultural issues that could be enacted at meals. In many NHs, they update information about end of life wishes, including use of artificial nutrition and hydration. The social worker may also assist when PWDs need assistance in getting access to food programs and related services when discharged.

Speech Language Pathologist

The speech language pathologist (SLP) conducts instrumental swallow assessments when dysphagia is suspected. Based on these evaluations the SLP may recommend changes in food/fluid viscosities and textures to improve oral intake while avoiding the risk of aspiration, and swallow exercises or technique changes for people with mild cognitive impairment or early-stage dementia.

Occupational Therapy

This specialty may be consulted for assistance with the external limitations to being able to self-feed. These specialists often approach the problem from the perspective of recommending adaptive equipment related to meals (e.g., built up plates and silverware), positioning devices and environmental adaptations.

Physical Therapy

Another determinant of meal intake is the physical ability to take in the meal. The physical therapists may offer strategies to improve upper extremity strength and reduce the development of hand/arm contractures through exercises that build strength and/or promote better range of motion.

Family Caregivers

Family caregivers play an important role by being able to provide a social and cultural history for PWD that may impact nutrition. Examples of information provided include food preferences, normal body weight, and daily eating patterns. Reviewing specific mealtime difficulties and associated factors with the family caregivers may provide contextual insight and assist with development of problem-solving strategies. The family caregiver may also be particularly effective in assisting with mealtimes by providing social support as well as feeding assistance.

Interdisciplinary Team Approach

All members of the interdisciplinary team should document results of care plan meetings and goals of care set in conjunction with legally authorized representatives in the medical record. Documentation should include the extent of the nutritional problems, a description of all identified or probable contributing etiologies, and the prognosis. Updates to the care plan should be ongoing, and indicative of appropriate palliative care measures, with progress notes maintained indicating effectiveness of the plan. In addition, Advance Directives should be documented (e.g., desires for Artificial Nutrition and Hydration, Do Not Resuscitate [DNR] and/or Do Not Hospitalize, and/or Medical Orders for Scope of Treatment [MOST] forms).

Assessment of Mealtime Difficulties in Persons with Dementia

In LTC settings, often times the first sign that a PWD is experiencing a mealtime difficulty is weight loss. Weight loss is defined by the Minimum Data Set (MDS) 3.0 as unplanned weight loss of 5 % in 1 month or 10 % in 3 months [22]. This is true even when the current body mass index (BMI) is normal or elevated since excess body fat can mask low levels of lean muscle mass. When weight loss occurs, careful investigation into potential reversible causes is the cornerstone of quality care.

Evaluation of Compromised Meal Behaviors Using the Need-Driven Dementia Compromised Model

Aversive feeding behaviors have been identified in the current literature as: clamping mouth shut, turning away of the head, pushing away assistance, holding food in the mouth, refusing to swallow, and/or allowing food to drop out of the mouth [23]. These behaviors were identified in the mid-1990s and subsequently tested by a tool known as the Edinburgh Feeding Evaluation in Dementia Scale (EdFED) [23–30]. More recent work has grouped these feeding difficulties according to their occurrence during the phases of the feeding cycle, which are initiating self-feeding, maintaining attention, getting food to the mouth, chewing food, and swallowing food [31].

When the EdFED was developed, much of clinical care provided to PWD was viewed from the perspective of the caregivers. Behaviors were interpreted as interfering with clinical care and the tasks needing to be completed. When viewed through the NDB Model, the behaviors are seen as meaningful and can be useful when providing and planning care [14].

When these behaviors are seen as meaningful forms of communication, it becomes more appropriate to refer to them as dementia-related mealtime behaviors, which carries less of a negative connotation. When caregivers interpret dementia-related mealtime behaviors as an attempt by the PWD to communicate preferences such as food or fluids desired at that moment in the meal, needing more time to chew/swallow, and/or an attempt to exert some control over the meal assistance process, caregivers tend to spend more time providing feeding assistance and the PWD consumes more food [32–34]. Conducting a meal observation to determine which dementia-related mealtime behaviors are being exhibited and at which point in the feeding cycle they occur provides the basis for beginning to solve mealtime difficulties.

Background Factors: Change the Person

Neurologic/Cognitive

The loss in ability to self-feed has been linked to disease progression, weight loss, and functional decline [35, 36]. As a person moves through the disease trajectory, caregivers should anticipate changes in functional, motor, and sensory abilities. As these changes occur, PWD will need more caregiver support at meal time. Assessment of cognitive ability can be achieved by routine Mini Mental State Examinations (MMSE) and comparing results over time, in addition to use of the MDS 3.0 Brief Inventory for Mental State (BIMS). Both instruments assess/screen attention, memory, visuospatial ability, and language skills.

As cognition declines, it impairs the ability of the person with dementia to initiate and maintain self-feeding. As a person becomes more dependent, behaviors may emerge that may often be interpreted as aversive or resistive. Use of the EdFED can provide insight into the types of behaviors observed and whether or not they are active (e.g., clamping mouth shut) or passive (e.g., allowing food to fall out of the mouth) [11].

Health Status

Changes in instrumental activities of daily living (IADLs) and ADLs may signal diminished functional ability and a requirement for more caregiver support. In early dementia, the loss of one or more IADLs, such as the ability to use the telephone or pay bills correctly is where family caregivers may notice deterioration. In long-term care settings, changes in ADLs will likely be the indicator that PWD need more caregiver support.

In addition to cognitive assessment, assessment of affective state should be conducted. In situations where the PWD still has the ability to communicate feelings with words; use of the Geriatric Depression Scale (GDS) would be appropriate. In the later stages of dementia, observation of behavior may provide insight into whether or not a PWD is expressing sadness, anxiety, boredom, or loneliness. Caregivers may see an emergence of repetitive vocalizations, or increased agitation as attempts to communicate affective difficulties.

Review of Vital Signs/Lab Work and Weight Patterns

The examination of vital signs and weight patterns over the past 6–12 months is a critical factor when beginning to assess a PWD exhibiting weight loss. This information is critical for providing insight into a slow and insidious weight loss, versus a more acute change in condition. In the NH, body weights for residents have emerged as a principle monitoring and screening indicator [12]. Weights are reasonably accurate, noninvasive, acceptable to patients, and fairly inexpensive indicators to maintain [12]. Weights should be obtained at admission, weekly thereafter for 4 weeks to establish baseline, and then monthly thereafter if weight is stable [12].

Any change in vital signs must be investigated for an underlying cause, including differential diagnoses of an infectious process. Assessment of blood chemistries can provide insight into correct dosing of certain medications, elevation in white blood cell counts, and or other hematologic changes that could contribute to or be an indicator of an acute change in the physical condition. Serum albumin is the most frequently used biomarker for establishing protein malnutrition and can provide insight into nutritional status in the previous 3 weeks; however, this is a stress-related marker and the level may be indicator for acute responses to nutritional interventions [12]. Assessment for hypermetabolic states is critical for adequately treating conditions such as hyperthyroidism or a malabsorption syndrome [12]. Baseline laboratory assessments should include a complete blood count, basic metabolic panel, drug levels as indicated, hemoccult, thyroid-stimulating hormone, urinalysis with culture and sensitivity as indicated, and testing for malabsorption syndromes [12].

Meal Intake and Bowel Movement Review

When a PWD begins to lose weight, caregivers must begin the process of assessing how and why the weight loss is occurring. In skilled nursing homes, all residents are monitored every shift for percentages of meal intake and the occurrence of bowel movements. Meal intake must be examined for patterns that can help identify the causes of weight loss. For example, if the PWD is indicated to be eating 100 % of every meal served and yet still losing weight, perhaps not enough food is being offered at each meal time. Another example would be a pattern of every breakfast intake being 0 %, and lunch and dinner patterns being 100 %; this would prompt investigation into reasons why the person isn't eating breakfast. Possible reasons might include being too sleepy at breakfast or not liking the types of foods offered at the breakfast meal. Meal patterns should be investigated with direct caregivers to determine the underlying problem, some of which might be easily modified (e.g., serve breakfast at a later time in the morning). Constipation is a common problem in older adults that can contribute to reduced food intake by causing bloating and abdominal discomfort. This is an easily modified, reversible condition that caregivers can identify for appropriate intervention. Absence of bowel movement documentation of 3 days or more is the "gold standard" for identifying and intervening for constipation.

Other assessments for determining potentially reversible causes for low meal intake include determination of stage of dementia, screening for depression, identifying any medications that might contribute to low intake, and dealing with underlying nausea, vomiting, or diarrhea, fluid retention and/or edema, or infection [12].

Psychosocial Factors

Knowing the personal life history of the PWD may help put mealtime difficulties into context. A PWD's age, gender, personality type, previous occupation, history of psychosocial stress, and general behavioral responses to stress are important factors in identifying potential triggers for behavioral problems [14, 37]. Adaptive strategies that were used as coping strategies by the PWD prior to cognitive impairment may not work when the PWD no longer has control over their environment. An example follows:

A PWD is exhibiting anxious behaviors that frequently interrupt the evening meal (e.g., wringing of the hands, repetitive smoothing of their hair). In a discussion with the family caregiver, the staff determines that this behavior is associated with fatigue and that the PWD was accustomed to an afternoon nap in her past routine. The staff institutes an afternoon rest period that helps to alleviate the anxious behavior, dinner intake is improved, and the addition of an anti-anxiety medication is avoided.

Proximal Factors: Change the People, Change the Place

Personal Factors

Emotional Fact-ors. Depression is a common culprit in weight loss in LTC settings, and is a treatable etiology [12]. Mealtimes are the greatest opportunity for social interaction, yet many direct care workers treat meals as simply a task to be completed [38]. Determination of the amount of support needed, and the amount of time needed to provide social interaction while increasing meal intake is vital. Family caregivers, when available, may be able to help either directly or indirectly when emotional factors interfere with meal intake.

Physiological Determinants of Intake. Potential physiologic etiologies for weight loss include frank anorexia. However, a milder but long-term disparity between energy expenditure and intake can also lead to weight loss [12]. Assessment of meal intake less than 75 % is often associated with weight loss, but if a person with dementia is eating 100 % of meals and weight loss is still occurring, assessment of energy expenditure is critical. Registered Dietitians can assist the team by estimating caloric needs, while nursing can provide information on repetitive movements, fidgeting, rocking, and/or wandering that can contribute to weight loss by elevating energy expenditure [12].

Other physiological factors that may contribute to poor food intake include dry mouth, oral, or dental pain. PWD should be evaluated by a dentist, preferably with a specialization in geriatrics, for poor fitting dentures, broken teeth or decay, or gingivitis [12]. CNAs should be instructed in proper methods to promote oral hygiene in dependent persons. Swallowing ability should be carefully observed and it may be necessary to discuss a swallowing evaluation with speech therapy [12]. The continuation of previously prescribed therapeutic diets (e.g., low sodium, low fat diets) needs to be carefully weighed in risk/benefit discussions with the interdisciplinary team and legally authorized representatives.

Table 18.2 Recommendations for improving the physical and social environment^a

Physical

- Choose the dining environment best suited for the individual PWD (resident room, dining room, or other location).
 Some PWD may eat more food when less activity is going on around them, while others may benefit from increased social interaction
- If traffic of other staff or residents is distracting to the PWD, limit or eliminate these interruptions during the mealtime
- Adequate staffing to assist in all aspects of mealtime care including oral care after meals, provision of adaptive equipment and functional positioning for seating
- Food preparation close to eating environment may provide smells associated with mealtimes, another cue that it is nearing time to eat
- · Music from the PWDs cohort that promotes a calm environment
- Adequate lighting, and contrasting table cloths/placemats. Be sure cups, plate, and silverware are on contrasting background. Reduce glare
- · Adaptive equipment as needed and suited to PWD ability

Social

- · Consistent seating if dining environment promotes meal intake
- Caregivers should talk with the PWD about the meal, explain what items are, and provide verbal and visual cueing as appropriate
- Position PWD so that eye contact can be made; may sit in front of PWD or to the side if providing hand-feeding assistance
- · Family members should be welcomed into dining experience and share meals with PWD, if appropriate

^aAdapted from [10]

Functional Performance. The ability of a person with dementia to feed themselves independently must be assessed with consideration given to the person's ability to initiate a meal and difficulties in using proper utensils, scooping the right amount of food, recognizing the amount of food provided, maintaining attention while eating, and/or maintaining alertness during a meal [36, 39, 40]. Upper extremity ability, positioning, fine-motor ability for holding utensils and/or finger foods, and any difficulty chewing or swallowing should be assessed and taken into account [12].

Physical and Social Environment

A list of recommendations for improving the physical and social environment is provided in Table 18.2 and more specific details are discussed in the following paragraphs. When possible, low-intake meals should be observed for potential problems. The nature of caregiver interactions (positive or negative) can strongly impact a PWD's meal intake [33]. For some, it may improve if a family member or friend is present; it has been shown that eating with a family member or caregiver increases calorie intake in homebound older adults [41].

Meal observations can provide insight as to whether or not a change has occurred in the level of support needed with feeding assistance, and/or if an environmental change might be necessary. Noise levels, dining area traffic, inability to maintain attention on the meal, and position in dining room can all contribute to distractibility at mealtime.

Light Level, Noise Level, and/or Temperature. The physical environment is touted as a vital component for dining. Where one eats can be as important as what one eats, and providers can ask themselves if they, too, would be willing to eat in the conditions under which LTC residents consume their meals [11, 12, 37]. Every effort should be made to make the dining setting comfortable and familiar (*not* in bed if at all possible). Cleanliness and lack of clutter, adequate lighting and contrasting colors to enhance visualization are basic considerations for supporting adequate meal intake [11, 12, 37, 42].

Dining Ambience. It has been suggested that the smells of freshly cooked foods can be used to stimulate appetite. The difficulty with this in LTC settings is that the kitchen is often in a location far from resident and dining rooms. If possible, some food items can still be prepared and cooked on the unit to provide the aroma, such as baked bread or vegetable soup. Some facilities have achieved the same effect by infusing essential oils of vanilla, cinnamon, or apple pie spice into the dining area.

Staff Stability and Staff Mix. The critical importance of staff training and stability factors for ensuring a well-trained and skilled workforce to care for persons with dementia cannot be over-emphasized. Consistent assignment of staff to work with PWD allows for staff to get to know the individual resident and respond to mealtime behaviors in ways that permit the meal to continue. By appropriately responding to the cues given by the PWD, meal intake is maximized. For example, a staff member working with the same PWD every day would know that the PWD holding food in their mouth will do so until a drink is offered. Rather than interpreting the behavior as negative and ceasing feeding attempts, a drink is offered and the meal continues.

Choosing the Best Course of Nutritional Care for the PWD

When determining which interventions are most appropriate for any given individual PWD, discussions with the interdisciplinary team and family members/legally authorized representatives are a vital part of goal-setting as the prognosis is discussed [12]. The most important and obvious goal is to promote self-feeding as much as possible, offering support only when functionally challenged or fatigued but still hungry to finish meal. Many recommendations for promoting this goal have been discussed in the preceding sections; they are also summarized at the end of the chapter. Depending upon where the PWD is in the disease process, goals of care should be established for maintenance or end-of-life management (end-stage dementia) [12]. Advance Directives should be reviewed for insight into wishes related to artificial nutrition and hydration through Living Wills. Plans of care should include interventions for each identified risk factor and/or potential etiology identified in the assessment. However, all caregivers, both professional and family/friends, need to realize that even the most individualized care strategies may not be successful in altering mealtime issues; the overall goal then becomes maintaining dignity and quality of life. Because of its critical importance in determining the best nutritional quality of life for the PWD, the closing sections of the chapter are devoted to a discussion of the benefits and risks of tube feeding versus careful hand feeding for those with advanced dementia.

Weighing the Risks/Benefits of Percutaneous Endoscopic Gastrostomy Tube (PEG) Feeding

As dementia progresses, loss of the ability to swallow and/or consume adequate food/fluids to sustain life are features of the disease and indicative of entering end-stage dementia. This stage in the disease process can create tremendous moral and ethical dilemmas for family members and/or legally authorized representatives regarding whether or not to maintain continued careful hand feeding or opt for tube feeding placement. The decision to place a feeding tube (or not) must be a well-informed decision based on current evidence and accurate information. At the point the decision needs to be made, families need support answering questions including but not limited to:

- Are there advance directives indicating a preference for life-sustaining artificial food/fluids?
- If the person with dementia could speak for themselves, would they want the current quality of life prolonged until death?

- Will the nutritional intervention have a significant positive impact on the person's overall condition or prognosis?
- What does the patient or family hope to achieve by placing the tube feeding—what is the desired outcome?

Families and residents need to be aware that placing a feeding tube does not decrease risk of aspiration or risk of pressure ulcer development, nor does it decrease a person's risk of suffering. In fact, quite often feeding tubes placed in persons with dementia cause agitation, discomfort, diarrhea, abdominal pain, and increased risk of local complications [12, 43, 44].

Careful Hand Feeding as an Alternative to PEG Feeding

Thoughtful consideration for engaging a person with dementia with actual food intake is an ethical and moral consideration that is often overlooked by harried staff who simply want to complete the task of providing food to residents. Careful hand feeding is a viable alternative to PEG feeding and we believe that in most cases it is the preferable intervention to support oral intake of food and fluids for PWD. This recommendation, that tube feeding should not be the approach for persons with late-stage dementia, was recently supported by two major professional medical groups – the American Geriatrics Society and the American Academy of Hospice and Palliative Medicine [45]. Our recommendation reflects the shift in thinking that moves the plan of care from curative to care focused. Careful hand feeding requires time and a highly skilled workforce to adequately and appropriately support a PWD during mealtimes [46, 47]. Unfortunately, careful hand feeding has not yet been recognized as a reimbursable nursing skill and therefore long-term care facilities only receive compensation for feeding tube management [48].

Three techniques for hand feeding are illustrated in Fig. 18.1. Current clinical practice guidelines for hand feeding present the first two techniques as evidence-based options for providing feeding assistance, namely direct hand and hand over hand feeding [37, 47, 49, 50]. *Direct hand feeding* is the most widespread hand feeding technique and is accepted in common practice for providing meal assistance to those who require it. *Hand over hand* is seen as a second option if a person still has fine motor ability to hold a utensil and simply need guidance in getting the utensil from the plate after acquiring a food item, to the mouth. The caregiver places their hand *over* the person's hand (holding the utensil), moving it towards the mouth. From the perspective of the PWD, this may inherently feel like the caregiver is forcing their hand towards their mouth; as such, it may elicit an aversive behavior of pushing the caregiver's hand away.

Evidence is emerging that a third hand feeding technique, *hand under hand*, exists and should be considered a third, viable careful hand-feeding technique [51, 52]. Hand under hand feeding assistance is beneficial for persons with dementia who have lost fine motor skills to hold the utensils, but still have gross motor skills. Hand-under-hand feeding assistance requires the caregiver to hold the utensil, and place their hand *under* the PWD's hand. From the perspective of the PWD, this feeding assistance elicits feelings of control over the movement, and feeling as though they initiated the movement. Evidence supports the benefit of providing persons with dementia verbal and auditory cues with meals, in agreement with the hand-under-hand technique, which provides a motor cue for a lifelong movement associated with intake [51, 52]. Further study is needed to determine under what conditions each hand-feeding technique would be most appropriate, and what intra-individual differences predict the best technique to use.



Fig. 18.1 Hand feeding (a) is the most commonly used hand-feeding technique, but should be reserved only for fully dependent persons. The hand-over-hand technique (b) is an option if a person still has sufficient fine motor ability to hold a utensil; it is used to guide a person's hand towards his/her face. A third technique is the hand-under-hand feeding approach (c). For persons with dementia, this approach may elicit more of a sense of control over the movement and allow them to feel as though they initiated the movement. Figures courtesy of Melissa Batchelor-Aselage

Clinical Recommendations

- 1. Individualized interventions are critically important to prevent chronic under-nutrition due to on-going mealtime difficulties in PWD.
- 2. An interdisciplinary approach is the best way to focus on alleviating mealtime issues; clear communication among team members creates the best possible condition for effective care management.
- 3. Based on the Need-Driven Dementia Compromised Model, caregivers are prompted to examine the background (change the person) and proximal factors (change the place and/or people) that may contribute to "problematic" behaviors and look for reversible or modifiable factors.
- 4. Self-feeding should be promoted for PWD whenever possible, offering support only when functionally challenged or fatigued but still hungry to finish the meal.
- 5. Careful hand feeding is a preferable alternative to PEG feeding to support oral intake of food and fluids for PWD with end-stage dementia.

References

- 1. Alzheimer's Association. 2013 Alzheimer's disease facts and figures. Alzheimers Dement. 2013;9(2):208-45.
- Center for Medicare and Medicaid Services. Nursing Home Compare. 2013; http://www.medicare.gov/NHCompare. Accessed 22 Sept 2013.
- 3. Vesnaver E, Keller HH. Social influences and eating behavior in later life: a review. J Nutr Gerontol Geriatr. 2011;30(1):2–23.
- Burger S, Kayser-Jones J, Bell J. Malnutrition and dehydration in nursing homes: key issues in prevention and treatment. 2000
- 5. Compendium MD. 2012.
- 6. Institute of Medicine. Retooling for for an Aging America: building the healthcare workforce. Washington, DC: National Academies Press; 2008.
- Pelletier CA. What do certified nurse assistants actually know about dysphagia and feeding nursing home residents? Am J Speech Lang Pathol. 2004;13(2):99–113.
- Pelletier CA. Innovations in long-term care. Feeding beliefs of certified nurse assistants in the nursing home: a factor influencing practice. J Gerontol Nurs. 2005;31(7):5–10.
- 9. Pasman H, The B, Onwuteaka-Philipsen B, van der Wal G, Ribbe M. Feeding nursing home patients with severe dementia: a qualitative study. J Adv Nurs. 2003;42(3):304–11.
- Amella EJ. Mealtime difficulties. In: Capezuti E, Zwicker D, Mezey M, Fulmer T, editors. Evidence-based geriatric nursing protocols for best practice. 3rd ed. New York, NY: Springer Publishing Company; 2008. p. 347–52.
- Amella EJ, Aselage M. Mealtime difficulties. In: Boltz M, Capezuti E, Fulmer T, Zwicker D, editors. Evidencebased geriatric nursing protocols for best practice. 4th ed. New York, NY: Springer Publishing Company; 2012. p. 453–68.
- 12. American Medical Directors Association. Altered nutritional status in the long-term care setting clinical practice guideline 2010. Columbia, MD: American Medical Directors Association; 2010.
- Amella EJ, Batchelor-Aselage MB. Facilitating ADLs by caregivers of persons with dementia: the C3P model. Occupational Ther Health Care. 2014;28(1):51–61.
- 14. Algase DL, Beck C, Kolanowski A, et al. Need-driven dementia-compromised behavior: an alternative view of disruptive behavior. Am J Alzheimer's Dis. 1996;11(6):10–9.
- 15. Aselage M, Amella E, Watson R. State of the science: alleviating mealtime difficulties for persons with dementia in the nursing home. Nurs Outlook. 2011;59(4):210–4.
- Hanson LC, Ersek M, Gilliam R, Carey TS. Oral feeding options for people with dementia: a systematic review. J Am Geriatr Soc. 2011;59(3):463–72.
- Salva A, Coll-Planas L, Bruce S, De Groot L, Andrieu S, Abellan G, et al. Nutritional assessment of residents in long-term care facilities (LTCFs): recommendations of the task force on nutrition and ageing of the IAGG European region and the IANA. J Nutr Health Aging. 2009;13(6):475–83.
- Dorner B, Posthauer ME, Friedrich EK, Robinson GE. Enteral nutrition for older adults in nursing facilities. Nutr Clin Pract. 2011;26(3):261–72.
- Academy Quality Management C, Scope of Practice Subcommittee of Quality Management C. Academy of nutrition and dietetics: scope of practice for the registered dietitian. J Acad Nutr Dietetics. 2013;113(6 Suppl):S17–28.
- Dorner B, Friedrich EK, Posthauer ME. Position of the American Dietetic Association: individualized nutrition approaches for older adults in health care communities [corrected] [published erratum appears in. J. Am. Diet. Assoc. 2010 Dec;110(12):1941]. J Am Diet Assoc. 2010;110(10):1549–53.
- Keller HH, Gibbs AJ, Boudreau LD, Goy RE, Pattillo MS, Brown HM. Prevention of weight loss in dementia with comprehensive nutritional treatment. J Am Geriatr Soc. 2003;51(7):945–52.
- 22. Center for Medicare and Medicaid Services. LongTerm Care Minimum Data Set (MDS). 2012; http://www.cms. gov/Research-Statistics-Data-and-Systems/Files-for-Order/IdentifiableDataFiles/ LongTermCareMinimumDataSetMDS.html. Accessed 1 May 2013
- 23. Watson R. Measuring feeding difficulty in patients with dementia: perspectives and problems. J Adv Nurs. 1993;18(1):25-31.
- 24. Watson R. Measuring feeding difficulty in patients with dementia: replication and validation of the EdFED Scale #1. J Adv Nurs. 1994;19(5):850–5.
- 25. Watson R. Measuring feeding difficulty in patients with dementia: developing a scale. J Adv Nurs. 1994;19(2):257-63.
- 26. Watson R. Research in brief. Measurement of feeding difficulty in patients with dementia. J Psychiatr Ment Health Nurs. 1994;1(1):45–6.
- 27. Watson R. The Mokken scaling procedure (MSP) applied to the measurement of feeding difficulty in elderly people with dementia. Int J Nurs Stud. 1996;33(4):385–93.
- Watson R. Construct validity of a scale to measure feeding difficulty in elderly patients with dementia. Clin Eff Nurs. 1997;1(2):114–5.

- 29. Watson R, Deary IJ. Measuring feeding difficulty in patients with dementia: multivariate analysis of feeding problems, nursing intervention and indicators of feeding difficulty. J Adv Nurs. 1994;20(2):283–7.
- Watson R, Deary IJ. Feeding difficulty in elderly patients with dementia: confirmatory factor analysis. Int J Nurs Stud. 1997;34(6):405–14.
- 31. Chia-Chi C, Roberts BL. Strategies for feeding patients with dementia. Am J Nurs. 2011;111(4):36–46.
- 32. Aselage M. Feasibility of implementing a web-based dementia feeding skills training module for nursing home staff [Doctoral Dissertation]. Charleston, SC: Doctoral dissertation College of Nursing, Medical University of South Carolina; 2011.
- 33. Amella EJ. Resistance at mealtimes for persons with dementia. J Nutr Health Aging. 2002;6(2):117-22.
- 34. Amella EJ. Feeding and hydration issues for older adults with dementia. Nurs Clin North Am. 2004;39(3): 607–23.
- Berkhout AM, Cools HJ, van Houwelingen HC. The relationship between difficulties in feeding oneself and loss of weight in nursing-home patients with dementia. Age Ageing. 1998;27(5):637–41.
- Edahiro A, Hirano H, Yamada R, Chiba Y, Watanabe Y, Tonogi M, et al. Factors affecting independence in eating among elderly with Alzheimer's disease. Geriatr Gerontol Int. 2012;12(3):481–90.
- Hellen C. Eating-mealtimes: challenges and interventions. In: Kaplan M, Hoffman S, editors. Behaviors in dementia: best practices for successful management. Baltimore, MD: Health Professions Press; 1998. p. 193–226.
- Amella EJ. Factors influencing the proportion of food consumed by nursing home residents with dementia. J Am Geriatr Soc. 1999;47(7):879–85.
- 39. Van Ort S, Phillips L. Feeding nursing home residents with Alzheimer's disease. Geriatr Nurs. 1992;13(5): 249–53.
- 40. Van Ort S, Phillips LR. Nursing intervention to promote functional feeding. J Gerontol Nurs. 1995;21(10):6–14.
- Locher JL, Robinson CO, Roth DL, Ritchie CS, Burgio KL. The effect of the presence of others on caloric intake in homebound older adults. J Gerontol A Biol Sci Med Sci. 2005;60(11):1475–8.
- 42. Liu W, Cheon J, Thomas S. Interventions on mealtime difficulties in older adults with dementia: a systematic review. Int J Nurs Stud. 2013;51:14–27. http://dx.doi.org/10.16/j.ijnurstu.2012.12.021.
- Finucane TE, Christmas C, Travis K. Tube feeding in patients with advanced dementia: a review of the evidence. JAMA. 1999;282(14):1365–70.
- Parker M, Power D. Management of swallowing difficulties in people with advanced dementia. Nurs Older People. 2013;25(2):26–31.
- 45. Fischberg D, Bull J, Casarett D, Hanson LC, Klein SM, Rotella J, et al. Five things physicians and patients should question in hospice and palliative medicine. J Pain Symptom Manage. 2013;45(3):595–605.
- 46. DiBartolo MC. Careful hand feeding: a reasonable alternative to PEG tube placement in individuals with dementia. J Gerontol Nurs. 2006;32(5):25–33. quiz 34–25.
- Simmons S, Schnelle J. Feeding assistant needs of long-stay nursing home residents and staff time to provide care. J Am Geriatr Soc. 2006;54:919–24.
- Zheng L, Qi T, Yang D, et al. MicroRNA-9 suppresses the proliferation, invasion and metastasis of gastric cancer cells through targeting cyclin D1 and Ets1. PLoS One. 2013;8:e55719.
- 49. Amella EJ. Amount of food consumed by elderly nursing home residents with dementia as a function of the quality of the reciprocal relationship between nurse aide caregivers and residents, caregivers' power and empathy. New York, NY: New York University; 1997.
- 50. Chang CC, Roberts BL. Strategies for feeding patients with dementia. Am J Nurs. 2011;111(4):36–44. author reply 45–36.
- Aselage M. Comparison of careful hand feeding techniques for persons with dementia in the nursing home. Durham, NC: Duke University School of Nursing: National Centers for Gerontological Nursing Excellence, Claire M. Fagin Fellowship; 2012.
- 52. Aselage M, Anderson R. A comparison of hand feeding techniques for persons with dementia in the nursing home. Durham, NC: Duke University School of Nursing Office of Research Affairs; 2011.

Chapter 19 Nutrition at the End of Life

Michi Yukawa and Christine Seel Ritchie

Key Points

- The majority of patients, patients' families, and friends consider nutrition and fluid intake as an essential part of life and not an aspect of medical treatment. As such, health care providers need to be sensitive to the complex psychosocial issues surrounding decisions regarding food and fluid intake in terminally ill patients.
- Case law in the United States considers artificial nutrition (enteral and parenteral) to be medical treatment and thus recognizes a patient's right to refuse artificial nutritional support.
- During discussion of feeding options, health care providers should keep in mind the overall goals of care for the patient and frame the conversation with patients, patient's families, and friends focused on their goals, whether it is for curative treatment, rehabilitative treatment, or comfort-focused treatment.

Keywords Nutrition support • Terminal illness • Palliative treatment

Introduction

Nutritional support during the final phase of life is an important and sensitive issue that must be addressed through careful discussion between patients, the patient's loved ones, and the health care team. Because food is an integral part of day-to-day life, the loss of interest in or the inability to partake in this activity often causes great distress for caregivers. Furthermore, caregivers view provision of food and hydration as basic care and not medical treatment, and as such they frequently struggle to determine which feeding options to choose for their loved ones. In the final weeks of an individual's

M. Yukawa M.D., M.P.H.

- Division of Geriatrics, Department of Medicine, San Francisco VA Medical Center, University of California San Francisco, 4180 Clement St. 181 NH, San Francisco, CA 94121, USA e-mail: Michi.Yukawa@va.gov
- C.S. Ritchie M.D., M.S.P.H. (⊠) Department of Medicine, University of California San Francisco, 3333 California Street, Suite 380, San Francisco CA 94118, USA e-mail: christine.ritchie@ucsf.edu

life, the issues central to decisions about nutritional support shift to a cautious weighing of its burdens and benefits. Therefore, health care providers need to be well versed in the issues surrounding nutrition and hydration in terminal illness in order to assist patients and their families in treatment decisions. This chapter will focus on the benefits and limitations of nutritional support in the final *weeks to months* of life. The principles presented in this chapter may not be generalizable to populations with a longer life expectancy, even if the underlying disease process is similar.

Definitions

Comfort feeding only/recreational feeding is assisted oral feeding provided by a caregiver to provide food or fluid as requested or desired by the patient [1]. Feeding of this kind may not provide adequate nutrition or hydration for patients; the goal is enjoyment for the patients. Non-oral feeding means giving nutritional support through a nasogastric tube, gastrostomy tube, or gastrojejunostomy tube. Providing water or electrolyte solution via non-oral route is *artificial hydration. Artificial nutrition* includes enteral nutrition by nasogastric tube, percutaneous endoscopic gastrostomy tube, percutaneous jejunostomy tube, gastrostomy tube or gastrojejunostomy tube, and parenteral nutrition.

Legal, Religious, and Ethical Precedents for Decision-Making

Multiple legal decisions are available to provide guidance to physicians concerning the nutritional treatment of terminal patients. In the Karen Ann Quinlan Case of 1976, the court upheld the right of individuals to forgo life-sustaining care and enabled her parents as surrogate to make treatment plans on behalf of Ms. Quinlan [2]. The Barber Case in 1983 involved a patient in a deep coma after cardio-pulmonary arrest following surgery to close an ileostomy. A California lower court initially indicted two physicians who stopped intravenous fluids and feeding tubes for murder [3]. However, the spouse testified that the patient had stated "no Karen Quinlan" and thus the appellate court dismissed the murder charges because the charge of murder requires proof of an act of commission. The distinction between acts of omission and acts of commission is central to discussions of the ethics of terminal nutrition and hydration. Acts of omission may be unlawful if there is a known duty to act. There was no duty to act in the Barber Case because of the patient's known prior wishes for no treatment [3].

In 1986, the appellate court in the Bouvia Case ruled that refusal of medical support by a competent patient is a fundamental right. The patient was a 29-year-old female with severe cerebral palsy who was bedridden, immobile, in constant pain, and competent to make her own medical decisions. A feeding tube was placed against the patient's wishes. The patient then petitioned to remove the feeding tube. The lower court refused the petition, stating it was a form of suicide. An appeals court later ruled in favor of the patient [4]. In the 1990 Nancy Cruzan case, the parent of Nancy Cruzan, a woman in a persistent vegetative state, requested cessation of nutritional support. The US Supreme Court stated that patients have a right to die and that competent patients can refuse therapy. They also made it clear that artificial nutrition/hydration was no different than any other medical treatment. However, they also concluded that the state (in this case, Missouri) can set its own evidentiary standards regarding what is considered clear and convincing evidence of the patient's wishes to withdraw life support [2].

Although the courts see artificial nutrition/hydration (ANH) as medical treatment, many patients and caregivers do not share this opinion; they see ANH as basic treatment that cannot be withdrawn. This tension around ANH is demonstrated by the fact that many states have separate additional statutory requirements for ANH, beyond that required for other forms of medical treatment [5, 6]. This tension was also made manifest in the Terry Schiavo case, a case where there was substantial

public disagreement regarding the ethics of discontinuing tube feeding in a woman in a persistent vegetative state [7].

Religious, cultural, and personal views often influence and take precedence over prevailing legal/ ethical positions on ANH. Prior to Terry Schiavo's case, the National Conference of Catholic Bishops stated that "Catholics are not obligated to use extraordinary or disproportionate means where there is no hope"; nevertheless, they also recommended a "presumption in favor of providing nutrition and hydration to all patients, including patients who require medically assisted nutrition and hydration, as long as this is of sufficient benefit to outweigh burdens involved to the patients" [8, 9]. However, since the Schiavo case, the Catholic position on ANH has become less clear [10]. According to the most recent two Popes, providing nutrition and hydration is considered "ordinary care" and not an extraordinary life preserving measure. The US Conference of Catholic Bishops in November of 2009 stated that Catholic health care facilities should offer food and water and even ANH to all patients regardless of their irreversible condition [10]. However, the Bishops' list of exceptions include use of ANH in cases where it will fail to prolong life or is "excessively burdensome or causes marked physical discomfort" [10].

Other religions such as Orthodox Judaism promote the continuation of ANH once it has been started [11]. Buddhists believe that a person who dies of hunger will become a hungry and restless soul and thus some Buddhists may also prefer ANH at the end of life [12]. However, in Hindu culture, decreased oral intake is often considered an indication of imminent death and not a cause of death. They view voluntary reduction of nutrition and fluid as a terminally ill person's preparation for an expected death in a self-controlled and dignified manner [12]. Therefore, they may be less likely to choose ANH at the end of life.

The lack of public consensus surrounding ANH requires that health care providers pay close attention to the burden and benefit of ANH and make sure patients and their proxies understand as fully as possible the benefits, burdens, and uncertainties associated with these therapies. Only then can patients or their caregivers truly determine what is consistent with their values and priorities.

In summary, case law in the United States confirms that enteral nutrition is medical treatment and not basic care. Patients have the right to refuse this form of medical treatment. Withdrawal of artificial nutritional support to allow a patient to die is not equivalent to euthanasia. In the former instance, the goal of discontinuing therapy is to remove burdensome interventions in the setting of a life-threatening or terminal condition; in the latter, the intended result is the death of the patient.

Nutrition and Hydration in Advanced Illness

The scientific literature provides sparse guidance regarding the benefits and burdens of nutrition and hydration at the end of life, with most studies conducted in cancer or advanced dementia patients. A few studies have begun to address patients' perspective on ANH at end of life and quality of life for those terminally ill patients.

Artificial Nutrition and Hydration in Terminal Cancer

Malnutrition and unintentional weight loss are cardinal symptoms in advanced cancer, and are poor prognostic indicators. Prevalence of malnutrition and weight loss can range from 46 to 85 % depending on types of cancer [13]. In addition, cancer cachexia, which is associated with metabolic abnormalities, anorexia, early satiety, decreased food intake, and loss of lean body mass, occurs in 80 % of patients in the advanced stages of cancer [14–16]. The underlying pathophysiology of cancer cachexia

is unknown, but it probably is due to combination of tumor burden, effects of cytokines, and adverse effects of cancer treatments [13, 14, 17]. The benefit of nutritional support in cancer cachexia remains questionable. Two systematic reviews of literatures failed to show improvement in survival tumor response, decreased toxicity, or decreased surgical complications in cancer patients receiving enteral nutrition [13, 14, 18]. The only exception may be in GI, head and neck, and esophageal cancer patients who are malnourished [19, 20]. A recent study in esophageal cancer patients demonstrated decreased frequency of hematological toxicities (neutropenia, leukopenia, and thrombocytopenia) in patients receiving enteral nutrition during neoadjuvant chemotherapy [21]. These findings are somewhat equivocal, however, and balanced by studies that question benefit. In a secondary analysis of the Radiation Therapy Oncology Group 90-03 Study, a trial primarily aimed at evaluating four different radiation fractionation schedules in head and neck cancer, findings did not suggest benefit of prophylactic nutrition support (PNS). Whereas patients receiving PNS had less weight loss by the end of treatment and less grade 3–4 mucositis, by 5 years, patients receiving PNS had poorer locoregional control and poorer survival [20].

Artificial Nutrition and Hydration in Terminal Dementia

Dementia is a progressive terminal condition with an average life expectancy of 4–9 years after an initial diagnosis of dementia [22, 23]. The Functional Assessment Staging System (FAST) is one system used to follow the course of Alzheimer's disease thereby helping to decide how far the disease has progressed [24]. There are seven stages in the FAST system, with the first stage comprising essentially no symptoms and the last stage describing advanced, end-stage dementia. At stage six, the patient needs supervision in dressing, bathing, toileting, and eating and becomes dependent on the caregiver. Patients typically either die or are institutionalized after 3 years in this stage. This is the stage when Alzheimer's disease patients may stop eating spontaneously, but can be encouraged to eat. At stage seven, Alzheimer's patients lose the ability to speak, ambulate, eat, and control their muscles and smile. When patients reach this stage it is very difficult to maintain nutrition because encouragement to eat becomes less successful. Patients at stage 7 typically die within a year, and difficulty eating is a marker for the terminal phase of Alzheimer's dementia.

Addressing the Costs and Benefits of Artificial Nutrition and Hydration in Advanced Dementia

Physicians and families have difficulty discussing nutrition support for patients with advanced dementia. Decisions around nutritional support may elicit more emotion from families than ventilator support or cardiac resuscitation. Families may articulate a concern that they are "starving" their loved one. They fear the ill person will not survive as a direct result of not eating. These issues are so difficult that sometimes physicians and families do not initiate discussions around nutrition.

Finucane attempted to address these difficult questions based on the limited clinical data available [25]. He reviewed primary concerns commonly cited as rationale for enteral nutrition, and included aspiration pneumonia, skin breakdown, quality of life, and survival as outcomes. His review of the literature did not indicate that enteral nutrition improved any of these outcomes. However, many of the studies reviewed were not solely of patients with dementia and many had methodological limitations. Multiple other controlled observational studies showed that tube feeding in dementia patients do not improve their survival rate, decrease aspiration risk, or improve wound healing [26–28]. A recent Cochrane review of tube feeding in advanced dementia patients also failed to provide sufficient evidence to suggest benefit in these patients [29]. Enteral tube feeding did not increase survival or prevent pressure ulcers in these patients.

Gillick addressed quality of life issues in nutritional support and found that advanced dementia patients who were tube fed were often deprived of taste, touch, and social interaction [30]. Sanders also evaluated survival with tube feeding [31]. He found that in a nursing home, the patients who were fed by a gastrostomy tube and those fed by hand had the same survival rates. He found that among gastrostomy patients, patients with dementia had a much worse prognosis (54 % 1-month mortality) compared to those without dementia (28 % 1-month mortality).

Despite the results of these studies, nursing home residents with advanced cognitive impairment often receive tube feeding. According to one study, the incidence of PEG tube placement among 97,241 nursing home residents with dementia in the United States was 53.9/1,000 patients with 48.3 % of them dying within a year after PEG tube placement [32]. Furthermore, one survey of primary care physicians in Hawaii showed that decisions to insert tube feedings were associated with internal medicine specialty, family preference, and fear of liability [33]. Another study explored the expected benefit of tube feeding held by physicians [34]. Physicians endorsed more benefit of tube feeding for patients after stroke or for those without ability to eat orally. Even for patients with neurodegenerative diseases, physicians felt that tube feeding would improve nutrition (93 %), hydration (69 %), and prolong life (49 %) [34]. One group of investigators performed an interdisciplinary team education study to teach attending physicians about the evidence suggesting lack of benefit of PEG tubes in advanced dementia [35]. Further education of medicine and surgical subspecialists about lack of benefit of PEG tubes in advanced dementia patients may facilitate more informed decisions about feeding options by patient's family and surrogate decision maker.

Alternative Approaches to Artificial Nutrition and Hydration in Advanced Dementia

In Alzheimer's dementia patients, skillful feeding approaches may provide an effective alternative to artificial nutrition. These include the appropriate selection of food consistencies, and minimization of distractions. Additionally, adequate time for feeding and verbal cuing to chew the food and swallow must occur. The mid-day meal is when food intake is often greatest, and presents an opportunity to effectively focus feeding efforts [36]. "Comfort feeding only" is an alternative method to provide nutrition and fluids to advanced dementia patients. Patients are offered sustenance if they would like to eat or drink and they are not forced. "Comfort feeding only" focuses on patient's individual comfort. This approach allows family members and surrogate decision maker a way to continue nutrition and hydration for patients without the added burden of tube feeding [1].

Artificial Nutrition at Terminal Stages of Other Neurological Diseases

Amyotrophic lateral sclerosis (ALS) is a progressive motor neuron disease which is characterized by progressive paralysis and respiratory failure. The prevalence of malnutrition varies from 16 to 53 % in ALS patients and the mortality rate is 7.7-fold higher for ALS patients who are malnourished [37, 38]. A recent Cochrane review stated that there was insufficient evidence to show survival advantage or improvement in nutrition after PEG tube placement [39]. Since there were no randomized controlled trials, they reviewed prospective nonrandomized and retrospective studies. Some studies showed increased survival rates and improved nutritional status with tube feeding while others did not [39]. Two studies revealed no enrichment in quality of life after instituting enteral nutrition [39]. Another study compared survival rates of bulbar and spinal onset ALS patients after PEG placement, which showed no difference in survival rates [40]. Having low forced vital capacity (FVC < 50 %) and old age were associated with poor prognosis after PEG placement [38, 40]. Therefore, enteral nutrition may be more beneficial for younger patients with bulbar-onset ALS at early stage of the disease.

Parkinson's disease is another neurodegenerative disease, which, at its terminal stage is associated with dysphagia and malnutrition. One small study showed lack of weight gain in patients after PEG placement [41]. In fact, male patients continued to lose weight after starting enteral nutrition. No information was collected on quality of life [41].

Patient's Perspective on Food and Hydration at End of Life

In an often-quoted study, the authors completed a prospective study of patients in a comfort care unit, the majority of whom had cancer or stroke as their terminal diagnosis. The patients were alert and competent. Food was offered and feeding was assisted but not forced. The patients were followed for thirst, hunger, and dry mouth and to see if food or fluid relieved the symptoms [42]. Sixty-three percent of the patients studied did not experience hunger; and additional thirty-four percent reported hunger initially that subsequently resolved. Similarly, 62 % had no thirst or thirst only initially. Most of the patient's symptoms were easily controlled with a small amount of food or water. The authors concluded by stating that hunger and thirst were uncommon in the terminal phase despite food and water intake inadequate to sustain basal energy requirements. Furthermore, a recent qualitative study of patients in Palliative Care Units revealed that patients did not want to be forced to eat and they did not want their weights monitored [43]. Some patients ate to please their caregiver rather than to satisfy themselves [44]. Caregivers who push food on the patient with anorexia may inadvertently contribute to the patient's distress instead of comforting the patient [42]. Less is known about how dementia patients perceive symptoms related to feeding. It is hard to know if patients with dementia experience discomfort from not eating or burdens related to assisted eating because they are noncommunicative at this stage.

Addressing Treatment Goals and Decision-Making for Terminal Patients

When treatment goals are discussed early following the diagnosis of a terminal condition, the patient often can decide for himself or herself what their wishes are regarding ANH at the end-of-life. In this way, the goals will reflect the preferences and values of the patient. Unfortunately, patient's wishes are often not known regarding tube feeding.

When a patient becomes terminal and is nondecisional, there are established guidelines as to how decisions are to be made. The advanced directive, if one exists, is the document that should be consulted first to know the patient's prior expressed wishes. If the advanced directive does not answer the specific question that needs to be addressed, then the legal guardian or the agent of the advanced directive makes the decision based on what they think the patient would have wanted or the wishes previously expressed by the patient. If there is no document and no designated decision-maker, then the first-order relative makes decisions (usually spouse, then adult children, then siblings but this may vary by state). Finally the opinion of other relatives can be considered. If none of the above exists, then the physician's judgment can be used to determine the best treatment for the patient.

Practical Considerations Before Providing Enteral/Parenteral Nutrition and Hydration

Decision making regarding the initiation or discontinuation of tube feeding is never easy. Practical matters may be more likely to influence the decision regarding feeding than ethical principles. In an imminently terminal patient the usual goal of care should be aggressive efforts at optimizing comfort.

This means managing the symptoms of the disease or side effects of the treatment while maintaining optimal quality of life. Thus, the goal of nutrition support in this phase of illness should be to optimize energy and strength while being attentive to potentially negative quality of life effects of coercive feeding or artificial nutrition and hydration. During this time, the clinician will be most effective if they seek to understand the caregiver's feelings and offer input. The caregiver may feel frustrated over the inability to find and prepare foods that are tolerated by the patient. They may also sense that the food they are offering is not providing the comfort that they were hoping for. The caregiver should be educated to understand that the loss of appetite and the inability to eat are common experiences in the terminally ill. Also, physical and emotional changes influence the ability to eat. For example, the disease itself, medications, fear, or depression may make it difficult to eat. Changes in the sense of smell, diarrhea, constipation, and nausea or vomiting also decrease the patient's appetite. Thirst and hunger often are diminished in the dying process. Practical options in caring for the terminally ill include eliminating most dietary restrictions and giving only the amounts of foods and liquids tolerated or accepted. Finally, the patient should be assisted with meals, but not forced to eat. It can be helpful to share with caregivers that (1) withholding nutrition and hydration at the end of life can be beneficial with regard to patient comfort, and (2) the injudicious use of ANH may aggravate symptoms. Without hydration there are less oral and airway secretions, less congestion, coughing and fewer symptoms associated with ascites and edema. In addition, terminal patients cannot always cough secondary to weakness; therefore, aspiration risk increases. Peripheral edema may increase pain and predispose the patient to pressure ulcers. Finally, increased gastrointestinal fluids can cause nausea and vomiting, especially for patients with intestinal strictures or obstruction from neoplasms. During the dying process, dehydration occurs from inadequate intake and losses from GI, renal, skin, and pulmonary secretion. Fluid deprivation at this stage rarely causes headache, nausea, cramps, or vomiting. Dehydration leads to mental changes, which may decrease the patient's awareness of their suffering. Families are sometimes concerned about the dry mouth that occurs as a person dies. Ice chips, sips of liquid, lip moisteners, salivary substitutes, mouth swabs, hard candy, and routine mouth care may all help to relieve the xerostoma (dry mouth) that occurs.

Potential Ways to Assist Families and Surrogate Decision-Makers Regarding Feeding Options for Advanced Dementia Patients

In a previous study of family members whose relatives died of dementia, 71.6 % participants stated that there was no decision about feeding tubes [45]. Furthermore, 13.7 % reported that there was no discussion about feeding tube insertion and 41.6 % stated that discussion about feeding tube was less than 15 min [45]. Family members of patients who died with a feeding tube felt that end-of-life care was suboptimal. These findings led to two studies conducted in 24 skilled nursing home facilities in North Carolina that showed that use of audio, audiovisual, or printed decision aids on feeding options for patients with advanced dementia significantly improved surrogate knowledge of feeding options for their loved ones and reduced decisional conflict regarding assisted oral feeding or tube feeding [46, 47]. Topics discussed in the audio, audiovisual, or printed aid included surrogate decision making, and clinical course of dementia, including development of feeding problems. It also discussed advantages and disadvantages of feeding tubes or assisted oral feeding. This intervention led to higher mean knowledge score about dementia and feeding options as well as increase the frequency of surrogates discussing feeding treatment with a physician, nurse practitioner, or physician assistant. After 3 months, the interventional group used assisted oral feeding more than the control group [47]. Investigators felt that a combination of audiovisual, audio, or printed decision aids and provider discussion may further improve the quality of decision making for surrogates about feeding options.
Conclusion

Caring for patients with a terminal illness such as cancer or advanced Alzheimer's disease is difficult for the family and the physician. The issues surrounding feeding are some of the hardest to resolve. This chapter defines the problems and offers broad guidelines for addressing these challenges.

Decisions regarding nutritional support in end-of-life care should be consistent with treatment goals and patient preference. Case law regards enteral nutrition as medical treatment rather than basic care. With the exception of head and neck cancer and esophageal cancer, no studies have demonstrated improved survival in cancer or advanced dementia with enteral support. In advanced cancer patients, nausea and pain should be addressed and corticosteroids and progestational agents considered. In advanced dementia, emphasis should be placed on oral food intake, allowing adequate time for feeding, avoiding distractions, and using verbal cueing. Every person, family, and physician must decide for himself/herself to what extent to nourish a person with a terminal illness based on available information about the risks and benefits.

Recommendations for Clinicians

- 1. Before deciding on a specific form of nutritional support, establish treatment goals.
- 2. In most cases, artificial nutritional support in cancer patients does not improve survival or tumor response, decrease toxicity, or decrease surgical complications.
- 3. Artificial nutritional support may be appropriate in head and neck cancer patients and esophageal cancer patients who are unable to swallow properly and still have an appetite.
- Current data do not demonstrate that artificial nutritional support improves survival or quality of life in advanced dementia patients.

References

- Palecek EJ, Teno JM, Casarett DJ, Hanson LC, Rhodes RL, Mitchell SL. Comfort feeding only: a proposal to bring clarity to decision-making regarding difficulty with eating for persons with advanced dementia. J Am Geriatr Soc. 2010;58(3):580–4. PubMed PMID: 20398123, Pubmed Central PMCID: 2872797.
- Annas GJ, Arnold B, Aroskar M, Battin P, Bartels D, Beauchamp T, et al. Bioethicists' statement on the U.S. Supreme Court's Cruzan decision. N Engl J Med. 1990;323(10):686–7.
- 3. Burck R. Feeding, withdrawing, and withholding: ethical perspectives. Nutr Clin Pract. 1996;11(6):243–53. PubMed PMID: 9016142.
- 4. Bouvia versus Superior Court (Glenchur). West California Report. 1986; 225:297-308.
- 5. Mayo TW. Living and dying in a post-Schiavo world. J Health Law. 2005;38(4):587–608. PubMed PMID: 16673631.
- Sieger CE, Arnold JF, Ahronheim JC. Refusing artificial nutrition and hydration: does statutory law send the wrong message? J Am Geriatr Soc. 2002;50(3):544–50. PubMed PMID: 11943054.
- Perry JE, Churchill LR, Kirshner HS. The Terri Schiavo case: legal, ethical, and medical perspectives. Ann Intern Med. 2005;143(10):744–8. PubMed PMID: 16287796.
- Nutrition and hydration: moral and pastoral reflections. National Conference of Catholic Bishops Committee for Pro-life Activities. J Contemp Health Law Policy. 1999;15(2):455–77.
- 9. Ethical and Religious Directives for Catholic Health Care Services in United States Catholic Conference. Washington, DC: National Conference of Catholic Bishops; 1995.
- Brody H, Hermer LD, Scott LD, Grumbles LL, Kutac JE, McCammon SD. Artificial nutrition and hydration: the evolution of ethics, evidence, and policy. J Gen Intern Med. 2011;26(9):1053–8. PubMed PMID: 21380599, Pubmed Central PMCID: 3157529.
- 11. Dorff EN. A Jewish approach to end-stage medical care. Conserv Jud. 1991;43(3):3-51. PubMed PMID: 11651008.

19 Nutrition at the End of Life

- Del Rio MI, Shand B, Bonati P, Palma A, Maldonado A, Taboada P, et al. Hydration and nutrition at the end of life: a systematic review of emotional impact, perceptions, and decision-making among patients, family, and health care staff. Psychooncology. 2012;21(9):913–21. PubMed PMID: 22162174.
- Elia M, Van Bokhorst-de van der Schueren MA, Garvey J, Goedhart A, Lundholm K, Nitenberg G, et al. Enteral (oral or tube administration) nutritional support and eicosapentaenoic acid in patients with cancer: a systematic review. Int J Oncol. 2006;28(1):5–23. PubMed PMID: 16327975.
- Gullett NP, Mazurak VC, Hebbar G, Ziegler TR. Nutritional interventions for cancer-induced cachexia. Curr Probl Cancer. 2011;35(2):58–90. PubMed PMID: 21420558, Pubmed Central PMCID: 3106221.
- 15. Lasheen W, Walsh D. The cancer anorexia-cachexia syndrome: myth or reality? Suppor Care Cancer. 2010;18(2):265–72. PubMed PMID: 19937260.
- Yavuzsen T, Walsh D, Davis MP, Kirkova J, Jin T, LeGrand S, et al. Components of the anorexia-cachexia syndrome: gastrointestinal symptom correlates of cancer anorexia. Suppor Care Cancer. 2009;17(12):1531–41. PubMed PMID: 19350287.
- Evans WJ. Skeletal muscle loss: cachexia, sarcopenia, and inactivity. Am J Clin Nutr. 2010;91(4):1123S-7. PubMed PMID: 20164314.
- Gullett N, Rossi P, Kucuk O, Johnstone PA. Cancer-induced cachexia: a guide for the oncologist. J Soc Integr Oncol. 2009;7(4):155–69. PubMed PMID: 19883531.
- Senesse P, Assenat E, Schneider S, Chargari C, Magne N, Azria D, et al. Nutritional support during oncologic treatment of patients with gastrointestinal cancer: who could benefit? Cancer Treat Rev. 2008;34(6):568–75. PubMed PMID: 18455316.
- 20. Rabinovitch R, Grant B, Berkey BA, Raben D, Ang KK, Fu KK, et al. Impact of nutrition support on treatment outcome in patients with locally advanced head and neck squamous cell cancer treated with definitive radiotherapy: a secondary analysis of RTOG trial 90-03. Head Neck. 2006;28(4):287–96. PubMed PMID: 16287132.
- Miyata H, Yano M, Yasuda T, Hamano R, Yamasaki M, Hou E, et al. Randomized study of clinical effect of enteral nutrition support during neoadjuvant chemotherapy on chemotherapy-related toxicity in patients with esophageal cancer. Clinical Nutr. 2012;31(3):330–6. PubMed PMID: 22169459.
- Hebert LE, Scherr PA, Bienias JL, Bennett DA, Evans DA. Alzheimer disease in the US population: prevalence estimates using the 2000 census. Arch Neurol. 2003;60(8):1119–22. PubMed PMID: 12925369.
- Larson EB, Shadlen MF, Wang L, McCormick WC, Bowen JD, Teri L, et al. Survival after initial diagnosis of Alzheimer disease. Ann Intern Med. 2004;140(7):501–9. PubMed PMID: 15068977.
- 24. Reisberg B. Functional assessment staging (FAST). Psychopharmacol Bull. 1988;24(4):653–9. PubMed PMID: 3249767.
- Finucane TE, Christmas C, Travis K. Tube feeding in patients with advanced dementia: a review of the evidence. JAMA. 1999;282(14):1365–70. PubMed PMID: 10527184.
- Teno JM, Gozalo P, Mitchell SL, Kuo S, Fulton AT, Mor V. Feeding tubes and the prevention or healing of pressure ulcers. Arch Intern Med. 2012;172(9):697–701. PubMed PMID: 22782196, Pubmed Central PMCID: 3555136.
- Arinzon Z, Peisakh A, Berner YN. Evaluation of the benefits of enteral nutrition in long-term care elderly patients. J Am Med Direct Assoc. 2008;9(9):657–62. PubMed PMID: 18992698.
- Murphy LM, Lipman TO. Percutaneous endoscopic gastrostomy does not prolong survival in patients with dementia. Arch Intern Med. 2003;163(11):1351–3. PubMed PMID: 12796072.
- Sampson EL, Candy B, Jones L. Enteral tube feeding for older people with advanced dementia. Cochrane Database Syst Rev. 2009; (2):CD007209, PubMed PMID: 19370678.
- Gillick MR. Rethinking the role of tube feeding in patients with advanced dementia. N Engl J Med. 2000;342(3):206– 10. PubMed PMID: 10639550.
- Sanders DS, Carter MJ, D'Silva J, James G, Bolton RP, Bardhan KD. Survival analysis in percutaneous endoscopic gastrostomy feeding: a worse outcome in patients with dementia. Am J Gastroenterol. 2000;95(6):1472–5. PubMed PMID: 10894581.
- 32. Teno JM, Mitchell SL, Skinner J, Kuo S, Fisher E, Intrator O, et al. Churning: the association between health care transitions and feeding tube insertion for nursing home residents with advanced cognitive impairment. J Palliat Med. 2009;12(4):359–62. PubMed PMID: 19327073, Pubmed Central PMCID: 2700356.
- Bell C, Somogyi-Zalud E, Masaki K, Fortaleza-Dawson T, Blanchette PL. Factors associated with physician decision-making in starting tube feeding. J Palliat Med. 2008;11(6):915–24. PubMed PMID: 18715184, Pubmed Central PMCID: 2988454.
- Hanson LC, Garrett JM, Lewis C, Phifer N, Jackman A, Carey TS. Physicians' expectations of benefit from tube feeding. J Palliat Med. 2008;11(8):1130–4. PubMed PMID: 18980454, Pubmed Central PMCID: 2982717.
- Campbell ML, Dove-Medows E, Walch J, Sanna-Gouin K, Colomba S. The impact of a multidisciplinary educational intervention to reduce PEG tube placement in patients with terminal-stage dementia: a translation of research into practice. J Palliat Med. 2011;14(9):1017–21. PubMed PMID: 21790469.

- Suski NS, Nielsen CC. Factors affecting food intake of women with Alzheimer's type dementia in long-term care. J Am Diet Assoc. 1989;89(12):1770–3. PubMed PMID: 2512336.
- 37. Muscaritoli M, Kushta I, Molfino A, Inghilleri M, Sabatelli M, Rossi Fanelli F. Nutritional and metabolic support in patients with amyotrophic lateral sclerosis. Nutrition. 2012;28(10):959–66. PubMed PMID: 22677356.
- Verschueren A, Monnier A, Attarian S, Lardillier D, Pouget J. Enteral and parenteral nutrition in the later stages of ALS: an observational study. Amyotroph Lateral Scler. 2009;10(1):42–6. PubMed PMID: 18615338.
- 39. Katzberg HD, Benatar M. Enteral tube feeding for amyotrophic lateral sclerosis/motor neuron disease. Cochrane Database Syst Rev. 2011; (1):CD004030, PubMed PMID: 21249659.
- 40. Pena MJ, Ravasco P, Machado M, Pinto A, Pinto S, Rocha L, et al. What is the relevance of percutaneous endoscopic gastrostomy on the survival of patients with amyotrophic lateral sclerosis? Amyotroph Lateral Scler. 2012;13(6):550–4. PubMed PMID: 22708872.
- Yamazaki Y, Kobatake K, Hara M, Katagiri M, Matsumoto M. Nutritional support by "conventional" percutaneous endoscopic gastrostomy feeding may not result in weight gain in Parkinson's disease. J Neurol. 2011;258(8): 1561–3. PubMed PMID: 21547382.
- 42. McCann RM, Hall WJ, Groth-Juncker A. Comfort care for terminally ill patients. The appropriate use of nutrition and hydration. JAMA. 1994;272(16):1263–6. PubMed PMID: 7523740.
- 43. Muir CI, Linklater GT. A qualitative analysis of the nutritional requirements of palliative care patients. J Human Nutr Diet. 2011;24(5):470–8. PubMed PMID: 21733142.
- 44. Hopkinson J, Corner J. Helping patients with advanced cancer live with concerns about eating: a challenge for palliative care professionals. J Pain Symp Manag. 2006;31(4):293–305. PubMed PMID: 16632077.
- 45. Teno JM, Mitchell SL, Kuo SK, Gozalo PL, Rhodes RL, Lima JC, et al. Decision-making and outcomes of feeding tube insertion: a five-state study. J Am Geriatr Soc. 2011;59(5):881–6. PubMed PMID: 21539524, Pubmed Central PMCID: 3254052.
- 46. Snyder EA, Caprio AJ, Wessell K, Lin FC, Hanson LC. Impact of a decision aid on surrogate decision-makers' perceptions of feeding options for patients with dementia. J Am Med Direct Assoc. 2013;14(2):114–8. PubMed PMID: 23273855, Pubmed Central PMCID: 3563876.
- 47. Hanson LC, Carey TS, Caprio AJ, Lee TJ, Ersek M, Garrett J, et al. Improving decision-making for feeding options in advanced dementia: a randomized, controlled trial. J Am Geriatr Soc. 2011;59(11):2009–16. PubMed PMID: 22091750, Pubmed Central PMCID: 3227016.

Part IV Contemporary Diet-Focused Concerns

Chapter 20 Anti-aging Effects of Nutritional Modification: The State of the Science on Calorie Restriction

L. Anne Gilmore, Eric Ravussin, and Leanne M. Redman

Key Points

- Calorie restriction (CR) is a dietary intervention hypothesized to improve quality of life and extend lifespan.
- Prolonged CR has been shown to extend both the median and maximal lifespan in a variety of animal species.
- Mechanisms of this CR-mediated lifespan extension possibly involve alterations in energy metabolism, oxidative damage, insulin sensitivity, and functional changes in both the neuroendocrine and sympathetic nervous systems.
- CALERIE is a randomized controlled trial that is testing the effects of prolonged CR in humans on biomarkers of aging and the rate of living theory hypothesis.

Keywords Calorie restriction • Caloric restriction • Humans • Longevity • Metabolism • Quality of life • Physical activity • Aging • Hormones • Weight loss

Introduction: Calorie Restriction and Lifespan

Aging is considered to be either "primary," that is, the inevitable deterioration of cells and tissues structure and function that occurs independent of disease, lifestyle, and environmental causes, or "secondary," where the decline in tissue structure and function occurs as a result of external influences, including diseases [52]. Attenuation of primary aging results in an increase in maximal lifespan, whereas delays in the progression of age-related disease or secondary aging increase mostly average lifespan. Calorie restriction (CR) has been shown since the 1930s by McCay et al. to retard the aging process [1], extending the median and maximal lifespan in various models and species [2].

L.A. Gilmore, Ph.D. • L.M. Redman, M.S., Ph.D. (🖂)

Department of Reproductive Endocrinology and Women's Health, Pennington Biomedical Research Center, 6400 Perkins Road, Baton Rouge, LA 70808, USA e-mail: anne.gilmore@pbrc.edu; leanne.redman@pbrc.edu

E. Ravussin, Ph.D.

Department of Health and Human Performance, Pennington Biomedical Research Center, 6400 Perkins Road, Baton Rouge, LA 70808, USA e-mail: eric.ravussin@pbrc.edu

While the exact mechanisms through which CR is able to extend the lifespan have yet to be fully elucidated, CR reduces metabolic rate and oxidative damage, improves markers of age-related diseases such as insulin resistance in diabetes, and has been shown to alter neuroendocrine activities in animals [3]. Results from studies in rhesus monkeys suggest that prolonged CR can also oppose many age-associated pathophysiological changes including learning and behavior changes, plasma insulin concentrations, and resting energy expenditure [4–6]. Since many changes associated with prolonged CR are important to the health and survival of humans, and excessive caloric intake is associated with morbidity and development of chronic diseases, it has become an important research objective to assess the feasibility, the safety, and the effects of prolonged CR in well-controlled human trials.

Calorie Restriction May Alter the "Rate of Living" and "Oxidative Stress"

The aging process may be influenced by CR through a reduction in the "rate of living," [7] leading ultimately to reduced oxidative damage. The rate of living theory suggests that increased metabolism and thus increased production of ROS leads to a shorter life span. An ongoing controversy among investigators appears to be whether chronic CR leads to "metabolic adaptation," a reduction in the metabolic rate that is greater than what is expected for the diminished metabolic mass of the organism [8]. Results from rats and monkeys suggest that most of the collected data should be reevaluated using appropriate methods of normalizing the metabolic rate for changes in metabolic size to truly test this theory [9]. For example, Blanc et al. recently calculated a 13 % reduction in resting energy expenditure after adjusting for fat-free mass in an 11-year-long study of energy restricted monkeys [4]. Recently, however, Selman et al., using doubly labeled water to measure total energy expenditure, reported that calorie-restricted rats expended 30–50 % more energy than expected [10]. Yamada et al. explain that while there is a decrease in resting energy expenditure, calorie-restricted monkeys had an increased physical activity level and intensity, leading to a greater than expected energy expenditure [11].

The "free radical theory of aging" or "oxidative stress" hypothesis is one of the well-supported theories of aging. It is widely accepted that the metabolic rate of an organism is a major factor in the rate of aging and is inversely related to its lifespan [12]. Additionally, since 1–3 % of consumed oxygen is associated with the production of reactive oxygen species (ROS), namely superoxide (O_2^{-}), hydrogen peroxide ($H_2O_2^{-}$), and the hydroxyl ion (OH⁻) [13], the production of these highly reactive molecules from normal aerobic metabolism is also in direct proportion to an organism's metabolic rate. Many investigators have shown that modulation of the oxidative stress of an organism through prolonged CR is able to retard the aging process in various species, including mammals [14, 15]. As a result of increased oxygen consumption, aerobic exercise is associated with increased production of ROS in muscle tissues [16]. However, exercise training boosts the antioxidant capacity of skeletal muscle, probably resulting in decreased overall oxidative stress [17]. Mitochondria consume the majority of cellular oxygen resulting in the production of ROS [18]. In humans and monkeys, longterm energy restriction has been shown to induce robust increases PGC-1 and mitochondrial biogenesis, which in turn is hypothesized to delay the onsets of sarcopenia, as well as loss of muscle function [19, 20]. Despite an increase in mitochondrial biogenesis, CR results in improved mitochondrial function, decreased total body oxygen consumption, and therefore decreased production of ROS [21].

Calorie Restriction, Cardiovascular Disease, Insulin Resistance, Type 2 Diabetes Mellitus

Elevated levels of LDL, excessive ROS generation, hypertension, and diabetes are all potential causes for the development of endothelial dysfunction, a precipitating event in the progression of atherosclerosis. These factors are believed to initiate an inflammatory response in the injured endothelial tissue. Long-term CR is associated with sustained reductions in factors related to endothelial dysfunction in humans, such as decreased blood pressure [22], reduced levels of total plasma cholesterol and triglycerides [23], and reduced markers of inflammation such as C-reactive protein, NFkB, and plasminogen activator inhibitor type-1 [24–27]. A long-term CR study in humans supports the feasibility of using CR to protect against atherosclerosis by showing a 40 % reduction in carotid artery intima-media thickness in CR participants relative to a control group [28]. Additionally, long-term CR counters expected age-related cardiac autonomic changes resulting in function equal to that of an average human 20 years younger [19].

Strong evidence shows that long-term CR in lean and obese subjects improves insulin sensitivity, a mechanism by which CR may act to extend lifespan [23, 29]. Improved insulin sensitivity is due in part to the down-regulation of IGF-1/insulin pathway that results in decreased P13K and AKT transcripts in skeletal muscle [27]. Additionally, prolonged CR reduces fasting glucose and insulin concentration, two factors believed to contribute to the aging process due to protein glycation [30] and mitogenic action [31], respectively. This compelling evidence suggests that weight loss due to CR may be the most effective means of improving insulin sensitivity, thereby decreasing the risk for the development of diabetes mellitus.

Why Calorie Restriction?

Since the first report of prolonged lifespan in rodents more than 70 years ago [1], CR has been gaining momentum as an intervention with the potential to ward off age-associated diseases and delay death. While the first observations were reported in rodents, similar observations have been reported across a wide range of species including yeast, worms, spiders, flies, fish, mice, and rats [8]. While the effects of CR in longer-lived species remains unknown, results reported thus far from three nonhuman primate colonies suggest that CR might have a similar effect in longer-lived species. Two of the largest longitudinal studies in Rhesus monkeys agreed that CR was beneficial on several markers of metabolic health, including weight, body composition, blood lipids, and cancer incidence but disagreed that young-onset CR reduced all-cause mortality and age-related death [5, 32]. Differences in diet composition and supplementation are two possible reasons for the conflicting results as well as a slight CR in the control animals, genetic diversity, and age at time of CR have been conducted with survival being the primary end-point [3]. There is, however, a lot that can be learned from controlled trials of CR where biomarkers of aging are measured and a handful of epidemiological and cross-sectional observations in longer-lived humans, centenarians, and individuals who self-impose CR.

Centenarians from Okinawa

Probably the most intriguing epidemiological evidence supporting the role of CR in lifespan extension in humans comes from the Okinawans [33]. Compared to most industrialized countries, Okinawa, Japan has four to five times the average number of centenarians, with an estimated 50 in every 100,000 people [34]. Reports from the Japanese Ministry of Health, Labor and Welfare show that both the average (50th percentile) and maximum (99th percentile) lifespan are increased in Okinawans. From age 65, the expected lifespan in Okinawa is 24.1 years for women and 18.5 years for men compared to 19.3 years for women and 16.2 years for men in the United States [35]. What is interesting about this population is that a low calorie intake was reported in school children on the island more than 40 years ago and later studies confirmed a 20 % CR in adults residing on Okinawa compared to mainland Japan [36]. A recent estimate of the energy balance of a cohort of Okinawa septuagenarians during youth to middle age suggested a 10–15 % energy deficit [37]. This energy deficit can be attributed to laborious occupations and daily

activities as farmers and a diet that was rich in nutrients yet low in energy density [37]. The nutrient-dense diet of the Okinawans allowed for a negative energy balance while providing an abundance of vitamins, minerals, antioxidants, and flavonoids [38]. Unfortunately, with the increase in American presence on Okinawa in recent years (location of US military base and increased availability of fast food chains), unpublished reports suggest that the longevity-promoting lifestyle (i.e., CR and high levels of physical activity) is threatened and may be reversed.

The Vallejo Study

To our knowledge there is only one study that was designed to test the effects of CR without malnutrition in nonobese humans [39]. This was a study of alternate day feeding in 120 men whereby the 60 participants in the CR group received an average of 1,500 kcal per day for 3 years, whereas the 60 others were ad libitum. This amounted to approximately 35 % CR compared to the control group. While the initial report was brief, analyses conducted several years later [40] indicated that the death rate tended to be lowered in the CR group and hospital admissions were reduced in these individuals by approximately 50 % (123 days for CR vs. 219 days for Control).

Unexpected Calorie Restriction in Biosphere 2

The unexpected low availability of food during the 2-year Biosphere 2 experiment provided a unique opportunity to observe the effects of CR in a group of nonobese humans.

Biosphere 2 was an enclosed 3.15-acre ecological laboratory that housed seven ecosystems or biomes resembling the earth: rainforest, savannah, ocean, marsh, desert, and agriculture and human/ animal habitats [41]. For 2 years, eight individuals, including Dr. Roy Walford, were completely iso-lated within this "mini-world," where 100 % of the air and water was recycled and all the food grown inside. Due to unforeseen problems with agriculture early on, food supply became quickly insufficient. Food intake for the eight individuals was projected at ~2,500 kcal/day and estimates from food records maintained by one of the biospherians suggested diets were restricted by ~750 kcal/day in each person during the first 6 months. The resulting ~15 % weight loss in the Biospherians was associated with many physiological, hematological, biochemical, and metabolic alterations [23, 42] consistent with calorie-restricted rodents and primates, including reductions in insulin, core temperature, and metabolic rate (Fig. 20.1) [32].

Randomized Controlled Trials of Calorie Restriction in Nonobese Humans

As for randomized controlled trials, results from a 2-year study of CR in humans will be known shortly. The National Institute on Aging (NIA) is sponsoring a trial, CALERIE (Comprehensive Assessment of the Long-term Effect of Reducing Intake of Energy), which is, for the first time, scientifically testing the effects of 25 % CR in ~150 nonobese healthy men and women aged 25–45 years. Three clinical sites are involved in the trial: Washington University in St. Louis, MO; Tufts University in Boston, MA; and the Pennington Biomedical Research Center in Baton Rouge, LA. The protocol and endpoints for this multicenter trial were developed from experience acquired in three independent Phase 1 trials conducted at each clinical site [3, 43, 44]. The 2-year CALERIE trial (CALERIE 2) was recently completed and data analyses are underway.



Fig. 20.1 Biosphere 2 (**a**), a 3.15-acre ecological enclosure provided an unexpectedly low availability of food for eight individuals who were housed inside for 2 years in the early 1990s. This study of nature of CR resulted in ~15 % weight loss (**b**), changes in energy expenditure (24 EE) and physical activity measured in a metabolic chamber (spontaneous physical activity) or as the ratio of total daily energy expenditure to resting metabolic rate (TEE/RMR) (**c**) and many hematological, biochemical, and metabolic alterations (**d**) consistent with calorie-restricted rodents and primates including reductions in insulin, core temperature, and metabolic rate [23, 42]

The remainder of this review will concentrate on the results from CALERIE Phase 1 conducted at the Pennington Center. For 6 months, 48 men and women were randomized to one of four treatment groups [21, 45–52]. For the CR group, overweight individuals were restricted to 75 % (a 25 % CR) of their weight maintenance energy requirements assessed by doubly labeled water [53]. The other groups were: (1) CR plus exercise group, for which the calorie deficit was also 25 % from weight maintenance but half (12.5 %) was achieved by CR and half (12.5 %) by increasing energy expenditure with structured aerobic exercise; (2) a low calorie-diet group in which participants consumed 890 kcal/day to achieve a 15 % weight loss and thereafter followed a weight maintenance diet; and (3) a healthy diet control group that followed a weight-maintaining diet based on the American Heart Association Step 1 diet. A sample menu can be found in Table 20.1. The effects of the CR interventions were determined from changes in various physiological and psychological endpoints after 3 and 6 months.

Table 2	0.1 Sam	ple menus	of food provided in the (CALERIE I	Phase I stu	ıdy at 1,500,	1,800, and 2,100 k	cal/day			
Day I			·								
Breakfi	ast			Lunch				Dinner			
1,500	1,800	2,100		1,500	1,800	2,100		1,500	1,800	2,100	
1/2 c	2/3 c	3/4 c	Low-fat granola with raisins				Greek wrap	1 c	1 1/4 c	1 1/3 c	Lentils with olives and feta
1 c	1 c	1 c	Skim milk	1	1	1 1/2	10 in. tortilla	1/2 c	2/3 c	3/4 c	Couscous
1	1	1	Banana	2 T	2 1/2 T	3 T	Hummus	1/2 c	3/4 c	3/4 c	Zucchini
				1/2 c	1/2 c	1/2 c	Cucumber	3/4 c	1 c	1 c	Strawberries
				1/3 c	1/3 c	1/3 c	Tomato	,			
				1 1/2 T	1 1/2 T	1 1/2 T	Onion	· · · · · ·			
				1 1/4 T	1 1/4 T	1 3/4 T	Olives	,			
				1 1/2 T	2 T	3 T	Feta cheese				
				1/3 c	1/2 c	1/2 c	Cheddar cheese	[]			
				1 1/3 c	1 2/3 c	1 2/3 c	Red grapes				
Day 2											
Breakfi	ast			Lunch				Dinner			
1,500	1,800	2,100		1,500	1,800	2,100		1,500	1,800	2,100	
1/4 c	1/2 c	1/2 c	Oatmeal	1 2/3 c	2 c	2 1/4 c	Pesto pasta	1 c	1 c	1 1/4 c	Greek-style potatoes
2/3 c	2/3 c	3/4 c	Peaches	3/4 oz	3/4 oz	1 1/4 oz	Chicken breast	3 oz	3 oz	3 oz	Salmon steak
1 T	1 T	1 1/2 T	Almonds	1	1	1	Apple	3/4 c	3/4 c	3/4 c	Green beans
1/2 c	1 c	1 c	Skim milk	-		2	Dinner roll	3/4 c	1 c	1 c	Mandarin oran <i>s</i> e

1/4 0 8 C č Ē E f τ 4 ŭ 2

Physiological Effects of Calorie Restriction

Six months of CR produced favorable alterations in physiological and behavioral outcomes.

Body Composition

Throughout the 6-month intervention there was a progressive decline in body weight that reached ~10 % for the CR group at the completion of the study (Fig. 20.2) [45]. Body composition analysis by dual X-ray absorptiometry and multi-slice computed tomography showed that the loss of tissue mass was attributable to significant reductions in both fat mass (CR: -24 ± 3 %) and fat-free mass (CR: -4 ± 1 %). There was a 27 % decrease in both visceral and subcutaneous fat depots, but it was interesting to note that the fat distribution within the abdomen was not altered by CR [45]. We also observed a reduction in subcutaneous abdominal mean fat cell size by ~20 %, a lowering of hepatic lipid by ~37 % but no change in skeletal muscle lipid content [50].

Biomarkers of Longevity

A "biomarker of aging or longevity" is considered to be any parameter that reflects physiological or functional age; it must undergo significant age-related changes, be slowed or reversed by treatments that increase longevity (e.g., calorie restriction), and must be reliably measured. Numerous biomarkers have been identified in rodents and primates, including body temperature and hormones such as DHEA-S and insulin (Fig. 20.3) [54]. In the CALERIE study, two out of the three biomarkers of longevity [54] were improved with 6 months of 25 % CR [3]. Significant reductions were observed in both fasting insulin concentrations (-29 ± 6 %) and core body temperature (-0.20 ± 0.05 °C),



Fig. 20.2 Our 6-month study of 25 % CR resulted in a progressive decline in body weight that reached ~10 % at the completion of the study [45]. Body composition analysis by dual X-ray absorptiometry showed that the loss of tissue mass was attributable to significant reductions in both fat mass (CR: -24 ± 3 %) and fat-free mass (CR: -4 ± 1 %)



Fig. 20.3 Can CR improve biological age and extend chronological age? This figure summarizes some of the potential biomarkers of aging. It is hypothesized that CR will change the biological trajectory of these biomarkers and therefore improve biological age and extend chronological age. For example, the *left panel* shows an individual aged 75 years. With prolonged CR it is hypothesized that fasting insulin and oxidative damage will be reduced in this individual. The *dotted line* represents the theoretical effects of CR. Therefore, an individual although 75 will have a biological age 17 years younger. Similarly the individual on the *right* at 90 years with prolonged CR will be biologically similar to an individual aged 66 years

whereas DHEA-S was unchanged by the intervention. These findings echo results previously reported in nonhuman primates and rodents on CR and long-lived men in The Baltimore Longitudinal Study of Aging [54].

Cardiovascular and Diabetes Risk Factors

With heart disease and stroke ranked numbers one and three in the causes of death in the United States [55], delaying the progression of atherosclerotic cardiovascular disease may be one potential mechanism by which CR promotes longevity. The risk factors for CVD, including abnormalities in blood lipids, blood pressure, hemostatic factors, inflammatory markers, and endothelial function, are worsened with aging [56, 57]. At least a portion of these age-related changes appear to be secondary to increases in adiposity and/or reductions in physical activity [58, 59] and, therefore, may be amenable to improvements through prolonged CR. Six months CR significantly reduced triacylglycerol (TG) and factor VIIc by 18 % and 11 %, respectively [60]. HDL-cholesterol was increased and fibrinogen, homocysteine, and endothelial function were not changed. According to total and HDL cholesterol (expressed as their ratio), systolic blood pressure, age, and gender, estimated 10-year CVD risk was 28 % lower after only 6 months of CR.

Insulin resistance is an early metabolic abnormality that precedes the development of hyperglycemia, hyperlipidemia, and overt type 2 diabetes. Both insulin resistance and β -cell dysfunction are associated with obesity [61–63]. Calorie restriction reduces fat mass and delays the development of age-associated diseases such as type 2 diabetes. While in obese humans it is well established that CR and weight loss improve insulin sensitivity [64, 65], the effects of CR on insulin sensitivity and, therefore, diabetes risk are not well understood in overweight and lean individuals. In our study of 6 months CR we observed a 40 % improvement in insulin sensitivity in the CR group, although this did not reach significance (p=0.08; p-values assess level of statistical significance. In most cases to be statistically significant a p-value of ≤ 0.05 must be reached) [50]. The acute insulin response to glucose (AIRg), however, was significantly decreased from baseline (CR: 29 ± 7 %, p < 0.01), indicating an improvement in β -cell responsiveness to glucose.

Metabolic Adaptation and Oxidative Stress

One of the most popular proposed theories by which CR promotes lifespan extension is the "rate of living theory" [66]. It is hypothesized that a lowering of the metabolic rate reduces the flux of energy with a consequential lowering of ROS and rate of oxidative damage to vital tissues [14]. Indeed, CR is associated with a robust decrease in energy metabolism, including an absolute lowering of resting metabolic rate (or sleeping metabolic rate), and thermic effect of meals and a decrease in the energy cost of physical activity. However, as mentioned earlier, whether total energy expenditure is reduced beyond the expected level (i.e., metabolic adaptation) for the reduction in the metabolizing mass (fatfree and fat mass: FFM and FM) following CR is a matter of debate.

As expected, absolute 24-h energy expenditure and sleeping metabolic rate (both measured in a respiratory chamber) were significantly reduced from baseline with CR (p < 0.001). Importantly, however, both 24-h sedentary and sleeping energy expenditures were reduced ~6 % beyond what was expected for the loss of metabolic mass i.e. FFM and FM [3]. This metabolic adaptation was also observed for RMR measured by a ventilated hood indirect calorimeter [46]. These physiological responses were associated with a reduced amount of oxidative stress as measured by DNA damage. DNA damage was reduced from baseline after 6 months in CR (p=0.0005), but not in controls [3]. In addition, 8-0x07,8-dihidro-2'deoxyguanosine (80x0dG) was also significantly reduced from baseline in CR subjects (p<0.0001). These data confirm findings in animals that CR reduces energy metabolism, oxidative stress to DNA, both potentially attenuating the aging process.

Endocrine Adaptations

Thyroid Function

Short-term studies of CR in humans have reported alterations in thyroid function. Four weeks of complete fasting resulted in a decrease in triiodothyronine (T3) and an increase in reverse triiodothyronine (rT3), which was associated with a reduction in metabolic rate [67]. The CRONIES (a self-selected group engaging on long-term CR) have significantly lower T3, but not thyroxine (T4) or thyroidstimulating hormone (TSH), concentrations compared to age-, sex-, and weight-matched controls [68].

In the CALERIE study, plasma T3 concentrations were reduced from baseline in the CR group after 3 (p < 0.01) and 6 months (p < 0.02) of intervention [3]. Similar results were found for the change in plasma T4 in response to the treatment. When the data of the subjects in the three CR groups were combined into one intervention sample, we observed significant linear relationships between the change in plasma thyroid hormones and the degree of 24-h metabolic adaptation after 3 months of intervention (T3; r=0.40, p=0.006 and T4; r=0.29, p=0.05) [3].

The Somatotropic Axis

Aging is marked by a reduction in both growth hormone (GH) and insulin like growth factor-1 (IGF-1) concentrations in healthy adults, resulting from a reduced amount of GH secreted at each burst without alterations of burst frequency or GH half-life [69]. Unlike for rodents, weight loss via CR in humans increases GH [70]. After 6 months of CR, 11-h mean GH concentrations were not changed with CR, nor was the secretory dynamics in terms of the number of secretion events, secretion amplitude, and secretion mass (unpublished data). The fasting plasma concentration of ghrelin, a GH secretagogue was significantly increased from baseline, but IGF-1 was unaffected. Despite a significant reduction in weight and visceral fat and an improvement in insulin sensitivity, mean GH concentrations were not altered by the 6 month intervention. In agreement with this observation was the finding that both GH and IGF-1 were not affected by the chronic food shortage experienced by the individuals in Biosphere 2 [23].

DHEA-S

Given the evidence from cross-sectional [71] and longitudinal studies [72] that DHEA-S declines with age, DHEA-S, which is a metabolite of DHEA, an abundant steroid hormone in the body, is considered to be a reliable endocrine marker of human aging and longevity [6]. It was hypothesized that CR will delay or attenuate the age-associated decline in DHEA-S. In our 6-month study in young individuals (37±2 years), we observed no alteration in DHEA-S [3]. Similarly, DHEA-S was not changed with 2 years of energy restriction in the individuals within Biosphere 2 [23]. To our knowledge, there has been no report of DHEA-S levels in those individuals from the Calorie Restriction Society (CRONIES) who are self-imposing CR. The lack of agreement between the human and nonhuman primate data is believed to be due to first, the chronological age of the subjects at the onset of CR, and second, to the duration of CR. Young adult monkeys undergoing CR for 3–6 years had an age-related decline in DHEA-S of 3 % compared to 30 % in monkeys fed ad libitum [73]. In contrast, CR initiated in older animals (~22 years) did not attenuate the age-associated decline in DHEA-S [74]. These explanations remain to be tested in longer term studies of CR in humans.

Leptin

Leptin, which influences body composition and energy balance by regulating energy intake and expenditure, also decreases with CR [75–77]. Metabolic adaptation or a drop in energy expenditure as a result of CR that is greater than what is expected on the basis of changes in weight and energy stores. The CR-mediated change in leptin is an independent determinant of metabolic adaptation. This suggests that leptin response to CR diets could be a possible biomarker for aging [75].

Physical Activity

Daily energy expenditure has three major components: resting metabolic rate (RMR), the thermic effect of food, and the energy cost of physical activity. Investigation of changes in physical activity are important in studies of CR not only because the contribution of physical activity to daily energy expenditure is variable, but also because it is not known if individuals volitionally or nonvolitionally decrease their level of physical activity in an attempt to conserve energy [78]. In our study, we observed no change in spontaneous physical activity in a respiratory chamber [46] consistent with earlier reports of no alterations in spontaneous physical activity [79] or posture allocation in obese individuals following weight loss [80]. These findings are not surprising if the current hypothesis that



Fig. 20.4 The effect of CR on all components of daily energy expenditure (*top panel*). The components of energy expenditure were determined by combining a measure of sedentary energy expenditure in the metabolic chamber (sleeping metabolic rate, SMR; spontaneous physical activity, SPA; TEF, thermic effect of food) and free-living energy expenditure by doubly labeled water (physical activity); The changes in total daily energy expenditure after 3 and 6 months of CR (*bottom panel*) are shown and those representing a metabolic adaptation (larger than due to weight loss) are *highlighted in grey* [53]. Combining two state-of-the-art methods (indirect calorimetry in the metabolic chamber and doubly labeled water) for quantifying precisely the complete energy expenditure response to CR in nonobese individuals, we identified a reduction in sedentary energy expenditure that was 6 % larger than what could be accounted for by the loss in metabolic size, i.e., a "metabolic adaptation" [3] and a metabolic adaptation in the free-living situation as well. This adaptation comprised not only a reduction in cellular respiration (energy cost of maintaining cells, organs and tissue alive) but also a decrease in free-living activity thermogenesis, *highlighted in blue* (behavioral adaptation)

spontaneous physical activity is biologically determined is true [80, 81]. However with a measure of energy metabolism in free-living conditions (doubly labeled water) we found that a metabolic adaptation exists after 3 months (-386 ± 69 kcal/day) but not after 6 months of CR (Fig. 20.4) [53]. This adaptation was evident even after total daily energy expenditure (TDEE) was adjusted for sedentary energy metabolism (24-h or sleeping energy expenditure) indicating that changes in other components

Table 20.2 Summary of the psychological and behavioral responses to 6 months of CR in humans Psychological/behavioral responses Development of eating disorder symptoms ↓ Disinhibition ↓ Binge eating ↓ Concern about body size and shape ↔ Fear of fatness ↔ Purgative behavior Depressed mood ↓ MAEDS Depression scale ↔ Beck Depression Inventory II Subjective feelings of hunger ↓ Eating Inventory, Perceived Hunger Scale Quality of life ↑ Physical functioning \leftrightarrow Vitality Cognitive performance \leftrightarrow Verbal memory ↔ Short-term memory and retention ↔ Visual perception and memory ↔ Attention/concentration

of daily energy expenditure, mostly as physical activity and less via diet-induced thermogenesis, are also involved. In support of this, physical activity level calculated by either the ratios of TDEE to RMR or sleeping metabolic rate [46], or TDEE adjusted for sleeping metabolic rate, was significantly reduced at month 3 by 12 % and returned toward baseline values after 6 months of intervention. Interestingly, despite lower physical activity levels, participants reported an improvement in physical functioning, a primary component of quality of life. All the effects of CR on physiological outcomes are summarized in Table 20.2.

Psychological and Behavioral Effects of Calorie Restriction

Calorie restriction in humans might prove to have positive effects on physical health and longevity, resulting in the practice of CR or the identification of CR mimetics. Very little is known about the effect of CR on the quality of life and, furthermore, if people attempt to follow CR for health promotion, important questions must be answered about possible negative effects of CR on psychological well-being, cognitive functioning, mood, and subjective feelings of appetite. Determining the effect of CR on these parameters is critical to learn if adhering to a CR regimen is feasible and if CR has unintended negative consequences that would offset the potential of its health benefits. Phase I of CALERIE provided a unique opportunity to examine the effect of 6 months of CR on psychological and behavioral endpoints. We here summarize the effects of CR on the development of eating disorder symptoms, quality of life, mood (symptoms of depression), subjective ratings of appetite, and cognitive function.

Development of Eating Disorder Symptoms

One of the most pressing concerns about CR is that the adherence to a sustained reduction in food intake will potentiate the development of symptoms of eating disorders. This concern is based in part on the Keys [78] study, which found that 50 % CR for 6 months among healthy men was associated with the development of eating disorder symptoms, e.g., binge eating [82]. Additionally, CR or the intent to restrict intake has been associated with the onset of eating disorders, including anorexia [83], bulimia nervosa [84], and binge-eating disorder [85]. Hence, there is a need to examine both the benefit and potential harm of CR in humans, particularly for people who are not obese, and to answer important safety questions before CR is recommended [86, 87].

In our study, participants completed an assessment battery that included: (1) the Multifactorial Assessment of Eating Disorder Symptoms (MAEDS), which measures six symptom domains associated with eating disorders (binge eating, purgative behavior, depression, fear of fatness, avoidance of forbidden foods, restrictive eating) [88], (2) the Eating Inventory, which measures dietary restraint, disinhibition, and perceived hunger [89], and (3) the Body Shape Questionnaire (BSQ) [90], which measures concern about body size and shape.

As reported by Williamson et al. [51], the three "dieting" groups in CALERIE, including the CR group, reported higher dietary restraint scores in comparison to the Control group at months 3 and 6, but measures of eating disorder symptoms did not increase and some decreased. All groups, except the control group, reported a significant reduction in disinhibition at month 6 whereas binge eating, decreased in all groups at months 3 and 6. Concern about body size/shape decreased at 3 and 6 months among the three dieting groups but did not change in the control group. The Fear of Fatness and Purgative Behavior subscales of the MAEDS did not change during CR.

Subjective Feelings of Hunger

The ability of people to follow CR could be limited by feelings of increased hunger. We evaluated change in appetite ratings during CR using the perceived hunger scale of the Eating Inventory [89] and the Visual Analogue Scales (VAS), which have been found to be reliable and valid measures of appetite: hunger, fullness, desire to eat, satisfaction, and prospective food consumption [91]. During the 6-month study, appetite ratings changed, but the changes among the dieting groups were not different from those in the control group. Moreover, based on the perceived hunger scale of the Eating Inventory, hunger was reduced in the CR group at month 6 [51].

Quality of Life and Mood

The Minnesota Semi Starvation study [78] indicated that CR can negatively affect mood; therefore, the effect of CR on mood and quality of life (QOL) becomes an important factor when considering the feasibility of CR in humans. During CALERIE Phase I, the Medical Outcomes Study Short-Form 36 Health Survey (SF-36) [92, 93] was used to measure QOL, and the Beck Depression Inventory II [94] and depression scale of the MAEDS were used to measure mood. Our results indicate that depressed mood, measured by the BDI-II, did not change during the trial. Additionally, in the CR group, scores on the MAEDS depression subscale decreased at 3 and 6 months in comparison to baseline [51]. Together, the results indicate that CR had no negative effect on mood during this trial and, in fact, symptoms of depressed mood, measured with the MAEDS, decreased in the CR group.

The SF-36 was used to test the effects of CR on two components of QOL—physical functioning and vitality. All dieting groups, but not the control group, had improved physical functioning during the trial. For the CR group, physical functioning was significantly improved at both months 3 and 6 but CR had no significant effect on vitality.

Cognitive Function and Performance

Self-reported dieting or CR has been associated with deficits in cognitive performance (e.g., memory and concentration deficits) [95, 96]. Nevertheless, cognitive impairment is frequently mediated by preoccupation with food and body weight [97], suggesting that obsessive thoughts about food and weight, rather than CR, negatively affect cognitive performance. If CR has negative effects on cognitive performance, the feasibility of CR in humans would be in doubt.

In our trial, cognitive performance was evaluated empirically at baseline and months 3 and 6 with a comprehensive neuropsychological battery [47]. Verbal memory was measured with the Rey Auditory and Verbal Learning Test (RAVLT) [98], short-term memory and retention with the Auditory Consonant Trigram (ACT) [99, 100], visual perception and memory with the Benton Visual Retention Test (BVRT) [101], and attention/concentration with the Conners' Continuous Performance Test-II (CPT-II) [102]. During CR, no pattern of memory or attention/concentration deficits emerged. The degree of daily energy deficit also was not correlated with change in cognitive performance; hence, these data indicate that CR did not have a negative effect on cognitive performance [47]. All the effects of CR on psychological and behavioral outcomes are summarized in Table 20.2.

The psychological and behavioral findings from CALERIE provide important information about the feasibility and safety of CR in humans. Calorie restriction was not associated with the development of eating disorder symptoms, decreased quality of life, depressed mood, or cognitive impairment. In fact, many of these endpoints improved, and changes in subjective ratings of appetite were similar in the CR group to those of the control group. These results suggest that CR might be feasible and have few unintended consequences, at least among overweight individuals. Additional research is needed to determine the feasibility and safety of CR in other samples of the population and over a longer duration.

Could Calorie Restriction Increase Longevity in Humans?

The wealth of CR literature in rodents allows us to address some important questions relating to the practicality and feasibility of CR in humans. Relevant and practical questions are (1) How much CR do we need to improve age-related health and possibly longevity? (2) How long do we need to sustain CR in order to obtain these benefits? Analysis of 24 published studies of CR in rodents (CR up to 55 %) indicated a strong negative relationship between survival and energy intake [103] and a positive relationship between the duration of CR and longevity. Using the prediction equations derived from the rodent data above [103], we and others estimated that a 5-year life extension could be induced by 20 % CR starting at age 25 and sustained for 52 years, i.e., the life expectancy of a male in the United States. However, if a 30 % CR was initiated at age 55 for the next 22 years, the gain would only be 2 months (Fig. 20.5).

Certainly there are individuals who self-impose CR with the CRON (Calorie Restriction with Optimal Nutrition) diet for health and longevity. A group of 18 CRONIES (only 3 women) have recently been studied after 3–15 years of CR [28, 104]. Dietary analysis indicated an energy intake ~50 % less than age-matched controls. In terms of body composition, the mean BMI of the males was



Fig. 20.5 How can CR impact lifespan in humans? By extrapolating the data from rodents to humans [103], one can predict the potential effect of CR in humans [105]. As an example, if Albert Einstein started a 20 % CR diet at 25 years of age, he could have increased his life by approximately 5 years. On the other hand, undertaking a 30 % CR diet 45 years later (age 60) would have extended his life by only 2 months. Therefore, CR needs to be initiated early in adult life to significantly increase life expectancy

 $19.6 \pm 1.9 \text{ kg/m}^2$ with an extremely low percent body fat of ~7 %. Atherosclerosis risk factors including total cholesterol, LDL-c, HDL-c, and triglycerides fell within the 50th percentile of values for people in their age group. This report provides further evidence that longer term CR is highly effective in lowering the risk of developing coronary heart disease and other age-related comorbidities. It remains to be seen if the CRONIES live longer than their age and sex matched counterparts.

Conclusion

While the rodent and primate data indicate that lifespan extension is possible with CR, collective analysis of the rodent data suggest that intensity and onset of CR required to induce these effects is probably not suitable for many individuals [105]. Epidemiological studies certainly support the notion that a reduced energy intake that is nutritionally sound improves age-associated health. While results of the first randomized trials of CR, albeit short in duration suggest a reduction in risk of age-related disease and improvements in some biomarkers of longevity, the ultimate effect of this intervention on lifespan in humans will probably never been determined in the scientific setting. In our short-term study, CR was not associated with the development of eating disorder symptoms, decreased quality of life, depressed mood, or cognitive impairment all probably indicating the feasibility and safety of CR in humans. However, it is a challenge for most individuals to practice CR in an "obesogenic" environment so conducive to overfeeding. Only a very few will be able to practice a lifestyle of CR and probably benefit from it. There is therefore a need for the search for organic or inorganic compounds that mimic the biological effects of CR. If such compounds often called "CR mimetics" (such as resveratrol [106, 107]) prove viable in humans, individuals for the most part will opt to enjoy the effects of anti-aging via a "pill" rather than CR.

Acknowledgements Studies of calorie restriction in nonobese individuals were supported by funding from the National Institutes of Health (grant number U01AG20478).

References

- McCay CM, Crowel MF, Maynard LA. The effect of retarded growth upon the length of the life span and upon the ultimate body size. J Nutr. 1935;10:63–79.
- Walford RL, Harris SB, Weindruch R. Dietary restriction and aging: historical phases, mechanisms and current directions. J Nutr. 1987;117(10):1650–4.
- 3. Heilbronn LK, de Jonge L, Frisard MI, DeLany JP, Larson-Meyer DE, Rood J, et al. Effect of 6-month calorie restriction on biomarkers of longevity, metabolic adaptation, and oxidative stress in overweight individuals: a randomized controlled trial. JAMA. 2006;295(13):1539–48.
- Blanc S, Schoeller D, Kemnitz J, Weindruch R, Colman R, Newton W, et al. Energy expenditure of rhesus monkeys subjected to 11 years of dietary restriction. J Clin Endocrinol Metab. 2003;88(1):16–23.
- Bodkin NL, Alexander TM, Ortmeyer HK, Johnson E, Hansen BC. Mortality and morbidity in laboratorymaintained Rhesus monkeys and effects of long-term dietary restriction. J Gerontol A Biol Sci Med Sci. 2003; 58(3):212–9.
- Roth GS, Mattison JA, Ottinger MA, Chachich ME, Lane MA, Ingram DK. Aging in rhesus monkeys: relevance to human health interventions. Science. 2004;305(5689):1423–6.
- 7. Sacher GA, Duffy PH. Genetic relation of life span to metabolic rate for inbred mouse strains and their hybrids. Fed Proc. 1979;38(2):184–8.
- Heilbronn LK, Ravussin E. Calorie restriction and aging: review of the literature and implications for studies in humans. Am J Clin Nutr. 2003;78(3):361–9.
- 9. Ravussin E, Bogardus C. Relationship of genetics, age, and physical-fitness to daily energy-expenditure and fuel utilization. Am J Clin Nutr. 1989;49(5):968–75.
- Selman C, Phillips T, Staib JL, Duncan JS, Leeuwenburgh C, Speakman JR. Energy expenditure of calorically restricted rats is higher than predicted from their altered body composition. Mech Ageing Dev. 2005;126(6–7): 783–93.
- Yamada Y, Colman RJ, Kemnitz JW, Baum ST, Anderson RM, Weundruch R, et al. Long-term calorie restriction decreases metabolic cost of movement and prevents decrease of physical activity during aging in rhesus monkeys. Exp Gerontol. 2013;48(11):1226–35.
- Sohal RS, Allen RG. Relationship between metabolic rate, free radicals, differentiation and aging: a unified theory. Basic Life Sci. 1985;35:75–104.
- 13. Alexeyev MF, Ledoux SP, Wilson GL. Mitochondrial DNA and aging. Clin Sci (Lond). 2004;107(4):355-64.
- 14. Sohal RS, Weindruch R. Oxidative stress, caloric restriction, and aging. Science. 1996;273(5271):59-63.
- Weindruch R, Walford RL, Fligiel S, Guthrie D. The retardation of aging in mice by dietary restriction: longevity, cancer, immunity and lifetime energy intake. J Nutr. 1986;116(4):641–54.
- Fulle S, Protasi F, Di Tano G, Pietrangelo T, Beltramin A, Boncompagni S, et al. The contribution of reactive oxygen species to sarcopenia and muscle ageing. Exp Gerontol. 2004;39(1):17–24.
- Sachdev S, Davies KJ. Production, detection, and adaptive responses to free radicals in exercise. Free Radic Biol Med. 2008;44(2):215–23.
- Ames BN, Shigenaga MK, Hagen TM. Mitochondrial decay in aging. Biochim Biophys Acta. 1995;1271(1): 165–70 [Research Support, U.S. Gov't, P.H.S. Review].
- Stein PK, Soare A, Meyer TE, Cangemi R, Holloszy JO, Fontana L. Caloric restriction may reverse age-related autonomic decline in humans. Aging Cell. 2012;11(4):644–50 [Research Support, N.I.H., Extramural Research Support, Non-U.S. Gov't].
- McKiernan SH, Colman RJ, Aiken E, Evans TD, Beasley TM, Aiken JM, et al. Cellular adaptation contributes to calorie restriction-induced preservation of skeletal muscle in aged rhesus monkeys. Exp Gerontol. 2012;47(3):229– 36 [Research Support, N.I.H., Extramural Research Support, Non-U.S. Gov't].
- Civitarese AE, Carling S, Heilbronn LK, Hulver MH, Ukropcova B, Deutsch WA, et al. Calorie restriction increases muscle mitochondrial biogenesis in healthy humans. PLoS Med. 2007;4(3):e76.
- Velthuis-te Wierik EJ, van den Berg H, Schaafsma G, Hendriks HF, Brouwer A. Energy restriction, a useful intervention to retard human ageing? Results of a feasibility study. Eur J Clin Nutr. 1994;248:138–48.
- Walford RL, Mock D, Verdery R, MacCallum T. Calorie restriction in biosphere 2: alterations in physiologic, hematologic, hormonal, and biochemical parameters in humans restricted for a 2-year period. J Gerontol A Biol Sci Med Sci. 2002;57(6):B211–24.

- Heilbronn LK, Noakes M, Clifton PM. Energy restriction and weight loss on very-low-fat diets reduce C-reactive protein concentrations in obese, healthy women. Arterioscler Thromb Vasc Biol. 2001;21(6):968–70.
- 25. Mavri A, Alessi MC, Bastelica D, Geel-Georgelin O, Fina F, Sentocnik JT, et al. Subcutaneous abdominal, but not femoral fat expression of plasminogen activator inhibitor-1 (PAI-1) is related to plasma PAI-1 levels and insulin resistance and decreases after weight loss. Diabetologia. 2001;44(11):2025–31.
- Bastard JP, Jardel C, Bruckert E, Blondy P, Capeau J, Laville M, et al. Elevated levels of interleukin 6 are reduced in serum and subcutaneous adipose tissue of obese women after weight loss. J Clin Endocrinol Metab. 2000; 85(9):3338–42.
- Mercken EM, Crosby SD, Lamming DW, Jebailey L, Krzysik-Walker S, Villareal DT, et al. Calorie restriction in humans inhibits the PI3K/AKT pathway and induces a younger transcription profile. Aging Cell. 2013;12(4): 645–51.
- Fontana L, Meyer TE, Klein S, Holloszy JO. Long-term calorie restriction is highly effective in reducing the risk for atherosclerosis in humans. Proc Natl Acad Sci U S A. 2004;101(17):6659–63.
- Walford RL, Mock D, MacCallum T, Laseter JL. Physiologic changes in humans subjected to severe, selective calorie restriction for two years in biosphere 2: health, aging, and toxicological perspectives. Toxicol Sci. 1999;52(2 Suppl):61–5.
- Robertson RP. Chronic oxidative stress as a central mechanism for glucose toxicity in pancreatic islet beta cells in diabetes. J Biol Chem. 2004;279(41):42351–4.
- Stenkula KG, Said L, Karlsson M, Thorn H, Kjolhede P, Gustavsson J, et al. Expression of a mutant IRS inhibits metabolic and mitogenic signalling of insulin in human adipocytes. Mol Cell Endocrinol. 2004;221(1–2):1–8.
- 32. Lane MA, Baer DJ, Tilmont EM, Rumpler WV, Ingram DK, Roth GS, et al. Energy balance in rhesus monkeys (Macaca mulatta) subjected to long-term dietary restriction. J Gerontol A Biol Sci Med Sci. 1995;50(5): B295–302.
- Kagawa Y. Impact of Westernization on the nutrition of Japanese: changes in physique, cancer, longevity and centenarians. Prev Med. 1978;7(2):205–17.
- 34. Japan Ministry of Health Law. Tokyo: Health and Welfare Statistics Association; 2005.
- Willcox DC, Willcox BJ, Todoriki H, Curb JD, Suzuki M. Caloric restriction and human longevity: what can we learn from the Okinawans? Biogerontology. 2006;7(3):173–7.
- Suzuki M, Wilcox BJ, Wilcox CD. Implications from and for food cultures for cardiovascular disease: longevity. Asia Pac J Clin Nutr. 2001;10(2):165–71.
- Willcox BJ, Willcox DC, Todoriki H, Yano K, Curb D, Suzuki M. Caloric restriction, energy balance and healthy aging in Okinawans and Americans: biomarker differences in Septuagenarians. Okinawan J Am Stud. 2007;4: 62–74.
- Willcox DC, Willcox BJ, Todoriki H, Suzuki M. The Okinawan diet: health implications of a low-calorie, nutrientdense, antioxidant-rich dietary pattern low in glycemic load. J Am Coll Nutr. 2009;28(Suppl):500S–16 [Review].
- 39. Vallejo EA. Hunger diet on alternate days in the nutrition of the aged. Prensa Med Argent. 1957;44(2):119–20.
- Stunkard AJ. Nutrition, longevity and obesity. In: Rockstein M, Sussman ML, editors. Nutrition, aging and obesity. New York: Academic; 1976. p. 253–84.
- 41. Walford RL, Harris SB, Gunion MW. The calorically restricted low-fat nutrient-dense diet in biosphere 2 significantly lowers blood glucose, total leukocyte count, cholesterol, and blood pressure in humans. Proc Natl Acad Sci U S A. 1992;2389:11533–7.
- 42. Weyer C, Walford RL, Harper IT, Milner M, MacCallum T, Tataranni PA, et al. Energy metabolism after 2 y of energy restriction: the biosphere 2 experiment. Am J Clin Nutr. 2000;72(4):946–53.
- 43. Das SK, Gilhooly CH, Golden JK, Pittas AG, Fuss PJ, Cheatham RA, et al. Long-term effects of 2 energyrestricted diets differing in glycemic load on dietary adherence, body composition, and metabolism in CALERIE: a 1-y randomized controlled trial. Am J Clin Nutr. 2007;85(4):1023–30.
- 44. Racette SB, Weiss EP, Villareal DT, Arif H, Steger-May K, Schechtman KB, et al. One year of caloric restriction in humans: feasibility and effects on body composition and abdominal adipose tissue. J Gerontol A Biol Sci Med Sci. 2006;61(9):943–50.
- Redman LM, Heilbronn LK, Martin CK, Alfonso A, Smith SR, Ravussin E. Effect of calorie restriction with or without exercise on body composition and fat distribution. J Clin Endocrinol Metab. 2007;92(3):865–72.
- 46. Martin CK, Heilbronn LK, de Jonge L, Delany JP, Volaufova J, Anton SD, et al. Effect of calorie restriction on resting metabolic rate and spontaneous physical activity. Obesity (Silver Spring). 2007;15(12):2964–73.
- Martin CK, Anton SD, Han H, York-Crowe E, Redman LM, Ravussin E, et al. Examination of cognitive function during six months of calorie restriction: results of a randomized controlled trial. Rejuvenation Res. 2007;10(2): 179–90.
- Martin CK, Anton SD, York-Crowe E, Heilbronn LK, VanSkiver C, Redman LM, et al. Empirical evaluation of the ability to learn a calorie counting system and estimate portion size and food intake. Br J Nutr. 2007;98(2): 439–44.

- 49. Williamson DA, Martin CK, York-Crowe E, Anton SD, Redman LM, Han H, et al. Measurement of dietary restraint: validity tests of four questionnaires. Appetite. 2007;48(2):183–92.
- Larson-Meyer DE, Heilbronn LK, Redman LM, Newcomer BR, Frisard MI, Anton S, et al. Effect of calorie restriction with or without exercise on insulin sensitivity, beta-cell function, fat cell size, and ectopic lipid in overweight subjects. Diabetes Care. 2006;29(6):1337–44.
- Williamson DA, Martin CK, Anton SD, York-Crowe E, Han H, Redman LM, et al. Is caloric restriction associated with development of eating disorder syndromes? Results from the CALERIE trial. Health Psychol. 2008; 27(1):S32–42.
- Anton SD, Martin CK, Redman L, York-Crowe E, Heilbronn LK, Han H, et al. Psychosocial and behavioral pretreatment predictors of weight loss outcomes. Eat Weight Disord. 2008;13(1):30–7.
- Redman LM, Heilbronn LK, Martin CK, De Jonge L, Williamson DA, DeLany JP, et al. Metabolic and behavioral compensations in response to caloric restriction: implications for the maintenance of weight loss. PLoS One. 2009;4(2):e4377.
- Roth GS, Lane MA, Ingram DK, Mattison JA, Elahi D, Tobin JD, et al. Biomarkers of caloric restriction may predict longevity in humans. Science. 2002;2(5582297):811.
- 55. Rosamond W, Flegal K, Friday G, Furie K, Go A, Greenlund K, et al. Heart disease and stroke statistics—2007 update: a report from the American Heart Association Statistics Committee and Stroke Statistics Subcommittee. Circulation. 2007;115(5):e69–171.
- 56. Celermajer DS, Sorensen KE, Spiegelhalter DJ, Georgakopoulos D, Robinson J, Deanfield JE. Aging is associated with endothelial dysfunction in healthy men years before the age-related decline in women. J Am Coll Cardiol. 1994;24(2):471–6.
- Mendall MA, Patel P, Ballam L, Strachan D, Northfield TC. C reactive protein and its relation to cardiovascular risk factors: a population based cross sectional study. BMJ. 1996;312(7038):1061–5.
- DeSouza CA, Shapiro LF, Clevenger CM, Dinenno FA, Monahan KD, Tanaka H, et al. Regular aerobic exercise prevents and restores age-related declines in endothelium-dependent vasodilation in healthy men. Circulation. 2000;102(12):1351–7.
- 59. Mora S, Lee IM, Buring JE, Ridker PM. Association of physical activity and body mass index with novel and traditional cardiovascular biomarkers in women. JAMA. 2006;295(12):1412–9.
- 60. Lefevre M, Redman LM, Heilbronn LK, Smith JV, Martin CK, Rood JC, et al. Caloric restriction alone and with exercise improves CVD risk in healthy non-obese individuals. Atherosclerosis. 2009;203(1):206–13 [Randomized Controlled Trial Research Support, N.I.H., Extramural Research Support, Non-U.S. Gov't].
- Forsey RJ, Thompson JM, Ernerudh J, Hurst TL, Strindhall J, Johansson B, et al. Plasma cytokine profiles in elderly humans. Mech Ageing Dev. 2003;124(4):487–93.
- 62. Matsumoto K, Sera Y, Abe Y, Ueki Y, Tominaga T, Miyake S. Inflammation and insulin resistance are independently related to all-cause of death and cardiovascular events in Japanese patients with type 2 diabetes mellitus. Atherosclerosis. 2003;169(2):317–21.
- 63. Utzschneider KM, Carr DB, Hull RL, Kodama K, Shofer JB, Retzlaff BM, et al. Impact of intra-abdominal fat and age on insulin sensitivity and beta-cell function. Diabetes. 2004;53(11):2867–72.
- 64. Dengel DR, Pratley RE, Hagberg JM, Rogus EM, Goldberg AP. Distinct effects of aerobic exercise training and weight loss on glucose homeostasis in obese sedentary men. J Appl Physiol. 1996;81(1):318–25.
- Niskanen L, Uusitupa M, Sarlund H, Siitonen O, Paljarvi L, Laakso M. The effects of weight loss on insulin sensitivity, skeletal muscle composition and capillary density in obese non-diabetic subjects. Int J Obes Relat Metab Disord. 1996;20(2):154–60.
- 66. Sacher GA. Life table modifications and life prolongation. In: Finch CE, Hayflick L, editors. Handbook of the biology of aging. New York: van Nostrand Reinold; 1977. p. 582–638.
- 67. Vagenakis AG, Burger A, Portnary GI, Rudolph M, O'Brian JR, Azizi F, et al. Diversion of peripheral thyroxine metabolism from activating to inactivating pathways during complete fasting. J Clin Endocrinol Metab. 1975; 41(1):191–4.
- Fontana L, Klein S, Holloszy JO, Premachandra BN. Effect of long-term calorie restriction with adequate protein and micronutrients on thyroid hormones. J Clin Endocrinol Metab. 2006;91(8):3232–5.
- 69. Veldhuis JD, Erickson D, Iranmanesh A, Miles JM, Bowers CY. Sex-steroid control of the aging somatotropic axis. Endocrinol Metab Clin North Am. 2005;34(4):877–93, viii.
- 70. Smith SR. The endocrinology of obesity. Endocrinol Metab Clin N Am. 1996;25(4):921-42.
- Orentreich N, Brind JL, Rizer RL, Vogelman JH. Age changes and sex differences in serum dehydroepiandrosterone sulfate concentrations throughout adulthood. J Clin Endocrinol Metab. 1984;59(3):551–5.
- Orentreich N, Brind JL, Vogelman JH, Andres R, Baldwin H. Long-term longitudinal measurements of plasma dehydroepiandrosterone sulfate in normal men. J Clin Endocrinol Metab. 1992;75(4):1002–4.
- Lane MA, Ingram DK, Ball SS, Roth GS. Dehydroepiandrosterone sulfate: a biomarker of primate aging slowed by calorie restriction. J Clin Endocrinol Metab. 1997;82(7):2093–6.

- 74. Urbanski HF, Downs JL, Garyfallou VT, Mattison JA, Lane MA, Roth GS, et al. Effect of caloric restriction on the 24-hour plasma DHEAS and cortisol profiles of young and old male rhesus macaques. Ann N Y Acad Sci. 2004; 1019:443–7.
- 75. Lecoultre V, Ravussin E, Redman LM. The fall in leptin concentration is a major determinant of the metabolic adaptation induced by caloric restriction independently of the changes in leptin circadian rhythms. J Clin Endocrinol Metab. 2011;96(9):E1512–6 [Randomized Controlled Trial Research Support, N.I.H., Extramural Research Support, Non-U.S. Gov't].
- Leibel RL. The role of leptin in the control of body weight. Nutr Rev. 2002;60(10 Pt 2):S15–9. discussion S68–84, 5–7, [Research Support, U.S. Gov't, P.H.S. Review].
- 77. Rosenbaum M, Leibel RL. Adaptive thermogenesis in humans. Int J Obes (Lond). 2010;34 Suppl 1:S47–55 [Review].
- Keys A, Brozek J, Henschel A, Mickelson O, Taylor H. The biology of human starvation. Minneapolis, MN: University of Minnesota Press; 1950.
- Ravussin E, Burnand B, Schutz Y, Jequier E. Energy expenditure before and during energy restriction in obese patients. Am J Clin Nutr. 1985;41(4):753–9.
- Levine JA, Lanningham-Foster LM, McCrady SK, Krizan AC, Olson LR, Kane PH, et al. Interindividual variation in posture allocation: possible role in human obesity. Science. 2005;307(5709):584–6.
- Zurlo F, Ferraro RT, Fontvielle AM, Rising R, Bogardus C, Ravussin E. Spontaneous physical activity and obesity: cross-sectional and longitudinal studies in Pima Indians. Am J Physiol. 1992;263(2 Pt 1):E296–300.
- Garner DM. Psychoeducational principles in treatment. In: Garner DM, Garfinkel PE, editors. Handbook of treatment for eating disorders. New York: Guilford Press; 1997. p. 145–77.
- 83. Williamson DA. Assessment of eating disorders: obesity, anorexia, and bulimia nervosa. Elmsford, NY: Pergamon Press; 1990.
- 84. Polivy J, Herman CP. Dieting and binging. A causal analysis. Am Psychol. 1985;40(2):193-201.
- 85. Williamson DA, Martin CK. Binge eating disorder: a review of the literature after publication of DSM-IV. Eat Weight Disord. 1999;4(3):103–14.
- Vitousek KM, Gray JA, Grubbs KM. Caloric restriction for longevity: I—Paradigm, protocols and physiological findings in animal research. Eur Eat Disord Rev. 2004;12:279–99.
- Vitousek KM, Manke FP, Gray JA, Vitousek MN. Calorie restriction for longevity: II-The systematic neglect of behavioural and psychological outcomes in animal research. Eur Eat Disord Rev. 2004;12:338–60.
- Anderson DA, Williamson DA, Duchmann EG, Gleaves DH, Barbin JM. Development and validation of a multifactorial treatment outcome measure for eating disorders. Assessment. 1999;6(1):7–20.
- Stunkard AJ, Messick S. Eating inventory manual (the Psychological Corporation). San Antonio, TX: Harcourt Brace & Company; 1988.
- Cooper PJ, Taylor MJ, Cooper Z, Fairburn CG. The development and validation of the Body Shape Questionnaire. Int J Eat Disord. 1987;6:485–94.
- Flint A, Raben A, Blundell JE, Astrup A. Reproducibility, power and validity of visual analogue scales in assessment of appetite sensations in single test meal studies. Int J Obes Relat Metab Disord. 2000;24(1):38–48.
- Ware JE Jr, Kosinski M, Gandek B. SF-36 health survey: manual & interpretation guide. Lincoln, RI: QualityMetric Inc.; 1993, 2002.
- 93. Ware Jr JE, Sherbourne CD. The MOS 36-item short-form health survey (SF-36). I. Conceptual framework and item selection. Med Care. 1992;30(6):473–83.
- 94. Beck AT, Brown GK, Steer RA. Beck depression inventory-II. San Antonio, TX: Psychological Corporation; 1996.
- 95. Green MW, Rogers PJ. Impairments in working memory associated with spontaneous dieting behaviour. Psychol Med. 1998;28(5):1063–70.
- Kemps E, Tiggemann M, Marshall K. Relationship between dieting to lose weight and the functioning of the central executive. Appetite. 2005;45(3):287–94.
- Kemps E, Tiggemann M. Working memory performance and preoccupying thoughts in female dieters: evidence for a selective central executive impairment. Br J Clin Psychol. 2005;44(Pt 3):357–66.
- 98. Schmidt M. Rey auditory and verbal learning test: a handbook. Los Angeles, CA: Western Psychological Services; 1996.
- 99. Peterson LR. Short-term memory. Sci Am. 1966;215(1):90-5.
- 100. Peterson LR, Peterson MJ. Short-term retention of individual verbal items. J Exp Psychol. 1959;58:193-8.
- Sivan AB. Benton visual retention test. 5th ed. San Antonio, TX: The Psychological Corporation, Harcourt Brace & Company; 1992.
- 102. Conners CK. Conners' continuous performance test (CPT II). Toronto: Multi-Health Systems Inc.; 2000.
- 103. Merry BJ. Calorie restriction and age-related oxidative stress. Ann NY Acad Sci. 2000;908:180-98.

- Fontana L, Weiss EP, Villareal DT, Klein S, Holloszy JO. Long-term effects of calorie or protein restriction on serum IGF-1 and IGFBP-3 concentration in humans. Aging Cell. 2008;7(5):681–7.
- 105. Redman LM, Ravussin E. Could calorie restriction increase longevity in humans? Aging Health. 2007;3(1):1-4.
- Baur JA, Pearson KJ, Price NL, Jamieson HA, Lerin C, Kalra A, et al. Resveratrol improves health and survival of mice on a high-calorie diet. Nature. 2006;444(7117):337–42.
- 107. Pearson KJ, Baur JA, Lewis KN, Peshkin L, Price NL, Labinskyy N, et al. Resveratrol delays age-related deterioration and mimics transcriptional aspects of dietary restriction without extending life span. Cell Metab. 2008;8(2):157–68.

Chapter 21 High-Risk Nutrients in the Aging Population

Katherine L. Tucker

Key Points

- Healthy diet offers tremendous promise for improving health and well-being with age.
- Older adults have lower energy requirements, but higher requirements for some nutrients, making it important to focus on nutrient-dense foods.
- Many older adults fall short of recommended nutrient intakes, placing them at risk of inadequacies that may affect their health.
- Shortfall nutrients include:
 - Macronutrients: protein, n-3 fatty acids, dietary fiber.
 - Vitamins: vitamins B₆, B₁₂, D, E, and carotenoids (vitamin A precursors).
 - Minerals: calcium, magnesium, and potassium.
- Research shows that, in most cases, isolated supplements are not as effective as obtaining the natural combinations of nutrients available in whole foods.
- Exceptions where supplements are often needed include:
 - Vitamin D, which is in few foods and is commonly inadequate in older adults, as they have less exposure to and less-efficient conversion from sunlight.
 - Vitamin B₁₂, which is difficult to absorb with declining stomach acid and interference from common medications.
- Foods to emphasize include fruit, vegetables, legumes, whole grains, nuts or seeds, fish, lean meat, poultry, and low-fat fluid dairy products.

Keywords Older adults • Nutrient intake • Vitamins • Minerals • Protein • Fatty acids • Dietary fiber

K.L. Tucker, Ph.D. (🖂)

Department of Clinical Laboratory and Nutritional Sciences, University of Massachusetts, 3 Solomont Way, Suite 4, Lowell, MA 01854-3092, USA e-mail: Katherine_tucker@uml.edu

Introduction

With more and more people living longer, an increasing segment of the population has chronic health concerns, including type 2 diabetes, heart disease, osteoporosis, physical frailty, and cognitive decline—all of which require ongoing care and pose an enormous burden to the health care system. Maintenance of nutritional status can go a long way to maximize healthy years of life, while minimizing the personal and institutional burdens of these chronic conditions. As a modifiable risk factor, a focus on healthy diet offers tremendous promise for improving the health and well-being of our aging population.

Unfortunately, national surveys and other research show that we have a long way to go to improve dietary quality in older adults. Relatively recent changes in the food supply and in lifestyle have led to more meals consumed outside of the home, more use of highly processed convenience foods, and relatively lower use of fresh fruit and vegetables, legumes, whole grains, nuts, and seeds. These changes have been paralleled with a rapid increase in obesity and poor nutrient status. Changing age demographics, along with societal changes in family dynamics, place many elders at risk of food insecurity and malnutrition.

Obtaining adequate nutrient intakes in older adults is challenging on several fronts. Although total energy requirement declines with age, requirements for many nutrients actually increase. This is due to declining function in many organ systems, which may affect efficiencies in absorption, conversion to active forms, or metabolism [1]. Several commonly used medications for chronic conditions increase nutrient requirements by interacting in ways that may interfere with absorption or metabolism [2]. Together, this means that it becomes increasingly important for aging adults to consume nutrient-dense diets. Based on national surveys and observational cohort studies, it is clear that intakes of several nutrients are inadequate, relative to dietary recommendations, or with respect to physiologic biomarkers, for a large segment of the older population. These "shortfall nutrients" include macronutrients—protein, n-3 fatty acids, dietary fiber; vitamins—vitamins B_6 , B_{12} , D, E, and carotenoids (vitamin A precursors); and minerals—calcium, magnesium, and potassium.

Other nutrients tend to be consumed in excess, increasing risk of obesity, hypertension, and related chronic conditions like type 2 diabetes and heart disease. These include saturated fats (i.e., fatty meats, processed meat, full fat dairy products), *trans* fats (i.e., hydrogenated oils, margarine, shortening, many processed baked products, crackers), refined carbohydrate foods (i.e., soft drinks, fruit drinks, white bread and products with white flour, white rice), and sodium (sodium compounds in canned and other processed foods, table salt). Much has been written about the risks associated with saturated fat, *trans* fat, refined carbohydrates, and sodium, and the need to limit these in the diet is widely understood. Therefore, they will not be discussed further here. A more recently recognized risk nutrient of excess is phosphorus, and a brief discussion of this mineral is included below.

Macronutrients

With age, energy requirements go down, and nutrient density becomes extremely important. To achieve this requires limiting "empty calories" including high fat and sugar or refined carbohydratedense foods. Among the macronutrients, it is most important to focus on obtaining sufficient protein, omega-3 fatty acids, and dietary fiber, while limiting saturated and *trans* fat, added sugars, and refined grain products.

Protein

Although there is a common belief that Americans get too much protein, this is not true for many older adults. Research has shown that protein adequacy is critical to maintaining muscle, bone, and functional status with age. Aging is associated with a gradual decrease in muscle mass, called sarcopenia (see Chap. 7) [3]. It is of great importance to maintain muscle to the extent possible, as lower muscle mass leads to reduced muscle strength, lower resting metabolic rate, increased risk of falls, and increased risk of functional impairment. Studies in New Mexico reported the prevalence of sarcopenia, and of sarcopenic obesity (high ratio of body fat to muscle mass), as 15 % and 2 %, respectively, in 60–69 year olds, and 40 % and 10 %, respectively, in those over 80 years [4].

Given evidence of decreasing efficiency of protein synthesis, impaired insulin action, and the high prevalence of muscle loss with aging, some experts argue that protein requirements for older adults should be greater than those currently recommended. Results from one short-term nitrogen balance study suggested that protein intake for older men and women should be 1.0–1.25 g of protein per kilogram of body weight per day [5], rather than the current recommended dietary allowance (RDA) of 0.8 g/kg/day. The need for higher protein was also supported by the longitudinal Health, Aging, and Body Composition Study, which found that older adults with higher protein intake lost less lean body mass over time than those with lower protein intake [6].

One issue that has contributed to the limitation of protein recommendations for older adults is a belief that a high-protein diet will contribute to osteoporosis, by increasing the acid load of the diet and leading to calcium losses [7]. However, more recent studies suggest that the calciuria (loss of calcium in the urine) seen with higher protein intake in controlled studies may be compensated for by higher calcium absorption, with no net negative effect on bone [8]. Further, several observational studies have shown that high, rather than lower, protein intakes were associated with greater bone mineral density [9, 10], and lower risk of hip fracture [11]. A recent systematic review and meta-analysis of protein-feeding trials also showed positive, rather than negative, effects of protein intake on bone [12].

Given the importance of protein intake for both muscle and bone health with aging, one review went as far as to suggest that intakes as high as 1.6–1.8 g/kg/day may be beneficial [13]. Others argue that not only total daily protein, but also the distribution of protein intake across the day can affect muscle protein synthesis, and one group has suggested that 25–30 g of high-quality protein per meal may maximize muscle retention [14].

Another frequently expressed concern that has limited recommendations for higher intake is that high protein intake may increase risk of progression of impaired kidney function [15]. However, a recent international study group noted that the only individuals who should limit protein intake are those with severe kidney disease (GFR <30 mL/min/1.73 m), but who are not on dialysis [16]. For most older adults, the group concluded that daily protein intake should be in the range of 1.0-1.2 g/kg body weight/day, and that it should be higher than this (1.2-1.5 g/kg/day) for those who are exercising or who have acute or chronic disease.

Other considerations for the aging population include particular care to ensure sufficient protein for exercise, and with weight loss—to maximize retention of muscle mass. Data from the National Health and Nutrition Examination Survey (NHANES) 2009–2010 showed that, among adults aged 70 year and older, men consumed a mean 74.4 g and women, 60.1 g of protein per day (Table 21.1). Although this exceeds the currently recommended mean of 56 g/day for the 70 kg reference older man, and 46 g/day for the 58 kg reference older women [17], many aging researchers, as discussed above, believe that most adults aged \geq 70 years will benefit from higher protein intakes from healthy sources. Excellent protein sources, that maximize nutrient content and minimize saturated fat, include fish, poultry, lean fresh meats, low fat milk and yogurt, legumes, and nuts (Table 21.2).

	Recommended daily allowance (RDA) or adequate intake (AI) ^a		Mean (standard error) daily intake from the 2009–2010 National Health and Nutrition Examination Survey	
Nutrient	Men	Women	Men	Women
Macronutrients				
Protein (g)	0.8/kg/day 56 g	0.8/kg/day 46 g	74.4 (1.8)	60.1 (1.5)
α -Linolenic acid ^a , 18:3 (g)	1.6	1.1	1.48 (0.06)	1.34 (0.05)
Eicosapentaenoic acid, 20:5 (g)	0.16	0.11	0.04 (0.006)	0.02 (0.004)
Docosahexaenoic acid, 22:6 (g)			0.02 (0.003)	0.02 (0.001)
Dietary fiber ^a (g)	30	21	17.1 (0.6)	15.2 (0.4)
Vitamins				
Vitamin A, (REA)	625	500	738 (49)	612 (18)
Vitamin B_6 (mg)	1.7	1.5	2.1 (0.1)	1.7 (0.1)
Vitamin $B_{12}(\mu g)$	2.4	2.4	5.98 (0.33)	4.18 (0.24)
Vitamin D (µg)	20	20	5.8 (0.4)	4.4 (0.1)
Vitamin E (mg α -tocopherol)	15	15	8.2 (0.4)	6.3 (0.3)
Minerals				
Calcium ^a (mg)	1,200	1,200	895 (35)	813 (13)
Magnesium ^a (mg)	420	320	290 (8)	243 (5)
Potassium ^a (mg)	4,700	4,700	2,800 (7)	2,300 (3)

 Table 21.1 Recommended intakes of shortfall nutrients for older adults, and mean intakes from the 2009–2010

 National Health and Nutrition Examination Survey

Estimated Average Requirements from Dietary Reference Intake reports (IOM 1997, 1998, 2000, 2001, 2002/2005, 2010)

Intake data from NHANES 2009–2010 http://www.ars.usda.gov/SP2UserFiles/Place/12355000/pdf/0910/Table_1_NIN_GEN_09.pdf, accessed August 30, 2013

^aThese nutrients do not have an RDA, but have an AI

Increase	To improve intakes of
Fruit	Dietary fiber, potassium, carotenoids and other phytonutrients
Vegetables (particularly dark green leafy and dark orange vegetables)	Dietary fiber, carotenoids and other phytonutrients, magnesium
Legumes	Vitamin B ₆ , magnesium, protein
Low fat fluid dairy (milk and yogurt)	Protein, vitamin D, calcium, vitamin B ₁₂ , potassium, magnesium
Nuts and seeds, including almonds, sunflower seeds and walnuts	Vitamin E, α -linolenic acid, vitamin B ₆ , magnesium, protein
Fish (particularly fatty fish) and seafood	DHA and EPA, protein, vitamin B ₆ , vitamin D
Poultry and lean meats	Protein, vitamin B ₆
Whole grains (100 % whole grain breads and cereals)	Dietary fiber, vitamin B ₆ , magnesium
Fortified breakfast cereals	Vitamin B ₁₂
Canola oil	α-Linolenic acid
Limit	To reduce intake of
Soft drinks and sweetened juice drinks (particularly cola)	Added sugars (and added phosphorus in cola)
Refined grain products (white bread, baked products, white rice)	Empty calories, rapidly absorbed carbohydrates
Fast food and fried foods	Trans and saturated fats, empty calories, sodium, phosphorus
Processed meats (including basted meats)	Phosphorus compounds, saturated fat, empty calories, sodium, heme iron
Liver and red meats	Retinol (preformed vitamin A), heme iron

Table 21.2 Food groups to emphasize or limit to improve nutrient intake in older adults

Omega-3 Fatty Acids

Omega-3 fatty acids are associated with protection against several chronic conditions. Populationbased studies have shown associations between fish intake and associated DHA and EPA intakes, with reduced risk of coronary heart disease mortality [18], fatal myocardial infarction [19], stroke [20], and dementia or Alzheimer's disease [21–24]. Mechanistic studies have demonstrated effects of omega-3 fatty acids in reducing inflammation, an important factor in many chronic diseases of aging. They have been shown to reduce membrane phospholipid arachidonic acid in a variety of cells, leading to reduced production of pro-inflammatory mediators [25]; and to downregulate nuclear factor-kappa beta, which is involved in regulating gene expression in inflammatory responses [26], among other anti-inflammatory mechanisms. A recent review concluded that omega-3 fatty acids are cardioprotective through a diverse number of actions, with evidence that they can reduce risk of thrombosis, lower triglycerides, improve endothelial function, slow plaque formation, and have anti-arrhythmic effects [27].

There is also accumulating evidence that fish intake is protective against cerebrovascular events. One systematic review examined 26 prospective cohort studies of fish intake and concluded that consuming two to four fish meals per week (vs. <1) was associated with 6 % reduction in risk of cerebrovascular outcomes and further, that eating fish five or more times per week was associated with 12 % reduction in risk [28]. However, in the same review, a meta-analysis of 12 randomized trials of fish oil supplements, vs. placebo, did not show a significant protective effect of fish oil supplements for cerebrovascular outcomes.

There has also been accumulating evidence for protective effects of fish or omega-3 fatty acids on cognitive function with aging. Importantly DHA is the most concentrated fatty acid in the brain and it is important for neural function. A recent meta-analysis of ten studies with plasma concentrations of fatty acids concluded that EPA, DHA, and total omega-3 fatty acid concentrations were significantly lower in older adults with dementia, relative to similarly aged controls [29]. A recent randomized trial of fish oil supplements (2.2 g/day of long chain n-3 fatty acids) with 65 older adults for 26 weeks led to significantly improved cognitive performance, particularly executive functioning, and visible improvement in measures of brain structure, including white matter integrity and gray matter volume, compared to those taking a placebo [30]. Animal models and in vitro studies have demonstrated protective effects of DHA on learning ability in mice and against amyloid buildup and A β fibrillation in brain tissue [31].

Although the marine-based long chain omega-3 fatty acids, DHA and EPA, are thought to be the most active omega-3 fatty acids for health, recent evidence for the beneficial effects of intake the plant based α -linolenic acid, and its sources—specifically walnuts or flaxseed and associated oils—have been accumulating. As a precursor for EPA and DHA, α -linolenic acid is an essential fatty acid. However, its conversion to these other long-chain omega-3 fatty acids is not efficient. A 2008 scientific workshop to consider reference intakes for DHA and EPA concluded that there was clear and consistent evidence for an inverse association between EPA+DHA intake and risk of fatal (and possibly nonfatal) CHD, and growing evidence for protection against cognitive decline [32]. They also concluded that conversion of α -linolenic acid to these fatty acids was so low that protective tissue concentrations could only be obtained by direct consumption of DHA and EPA. It is only relatively recently that evidence for protective effects of α -linolenic acid, itself, has received attention. One review [33] concluded that α -linolenic acid might have direct roles in protecting the heart, modulating the inflammatory response, and protecting central nervous system function. More recent reports have shown reduction of inflammation in brain cells [34], asthma control [35], and reduced risk of diabetes [36].

Due to the limited number of food sources, intakes of these important fatty acids tend to be inadequate in large segments of the population. Adequate Intake (AI) recommendations have been set for α -linolenic acid as 1.6 g/day for older men and 1.1 g/day for older women. The 2009–2010 NHANES data (see Table 21.1) show mean intakes of 1.5 g/day for men and 1.3 g/day for women, aged 70 years and older, but because these mean intakes are close to the recommendation, many older adults have lower intakes. Sources of α -linolenic acid include flax seeds, walnuts, and certain oils, including canola and, to a lesser extent, soy oils. At this time, there is no AI for DHA or EPA, but individual groups have made recommendations. The workshop on this topic noted above [32] concluded that individuals should consume 0.25–0.5 g/day of DHA+EPA, independent of α -linolenic acid. The American Heart Association recommends that individuals without heart disease eat fish, preferably oily fish, two to three times per week to obtain an average of 0.5 g/day of EPA+DHA, with higher intakes recommended for patients with heart disease or high triglycerides [37]. The major source of long chain omega-3 fatty acids is from cold-water fatty fish, including salmon, mackerel, sardines, and tuna.

Dietary Fiber

Dietary fiber is important for intestinal health and protection against heart disease and metabolic syndrome, as well as adding bulk for a feeling of fullness to aid in limiting energy intake.

There are two primary classifications for dietary fiber, both of which are important for the aging population. Whole grains and the indigestible portions of vegetables contain insoluble dietary fiber, which helps to regulate intestinal transit time, and thereby avoid the common problem of constipation in older adults [38]. Fruits and grains also contain soluble fiber, which has been shown to lower serum cholesterol and reduce the risk of heart disease [39].

Fiber intakes of older men or women in the United States consistently fall well below the AI recommendation of 30 g/day for men and 21 g/day for women. The 2009–2010 NHANES data show mean intakes of 17 g/day for men and 15 g/day for women (see Table 21.1). Although fiber supplements have been widely used, it is important to recognize that fiber from foods is packaged with additional vitamins, minerals, and phytonutrients in the food matrix; therefore increasing intakes of fruit, vegetables, and whole grains is preferentially recommended to increase dietary fiber intakes [40].

Vitamins

With aging, the requirement for several vitamins increases, despite lower energy needs. This is due to declining efficiencies in absorption, metabolism, and retention of certain nutrients. Key shortfall vitamins include several B vitamins, vitamin D, vitamin E, and carotenoids—the latter are vitamin A precursors from plant sources. In contrast, high intake of preformed vitamin A—in the form of retinol—should be avoided, as the body becomes less able to excrete it with age [41].

Vitamin A and Carotenoids

Although dietary assessment reports often suggest that Vitamin A intake is inadequate in older adults, it is important to also avoid excess retinol, the preformed vitamin A found in animal foods, particularly liver, and in some supplements, including cod liver oil. In contrast to many other vitamins, where age-related decreases in efficiencies of absorption and metabolism increase requirements, there is decreased clearance of vitamin A by hepatic and other peripheral tissues with age [42]. High intakes of preformed vitamin A have been linked with bone loss [43] and increased risk of fracture [44]. A recent study also showed that brain retinol concentrations were inversely related to cognitive function, while serum carotenoid concentrations were associated with better cognitive function [45].

It is, therefore, recommended that older adults obtain most of their vitamin A from carotenoids in plant foods; β -carotene, α -carotene, and β -cryptoxantin can all be cleaved to form vitamin A, but this conversion is limited by requirement and therefore provides a natural regulatory mechanism for minimizing the risk of retinol toxicity. Beyond their vitamin A activity, there is increasing evidence that carotenoids offer other health advantages, including strong anti-oxidant activity. Carotenoids from food have been linked with lower risk of heart disease [46] and cancer [47]. Lutein and zeaxanthin (non-provitamin A carotenoids) are found in high concentration in the eye, and have been linked with prevention of cataract- and age-related macular degeneration [48]. Lycopene, another non-provitamin A carotenoid, has been shown to protect against prostate cancer [49]. Despite its importance as an antioxidant, and consistent evidence of protective effects when consumed in foods, high-profile trials providing high doses of β -carotene in the form of a supplement actually saw increased rather than the expected decreased risk of cancers [50–52]. Although the mechanisms are not fully understood, there is growing evidence that carotenoids work optimally when consumed together with other constituents in the food matrix. Therefore, intakes should be maximized from food, and supplements with carotenoids should be generally avoided [53]. The best way to ensure adequate intake of carotenoids, including the provitamin A carotenoids, is to consume a variety of colorful fruits and vegetables. The provitamin A carotenoids are most concentrated in dark orange fruits and vegetables (i.e., carrots, peaches) and leafy green vegetables (i.e., spinach, broccoli). Dark leafy greens are also a major source of lutein, corn is a source of zeaxanthin, and tomatoes and watermelon are sources of lycopene. Benefits of fruit and vegetable intake extend beyond these carotenoids to other beneficial phytochemicals, increasingly being shown to be protective against a variety of age-related health conditions [54].

B Vitamins

B vitamins tend to be plentiful in a diverse natural food supply. Historically, widespread refining of wheat and rice led to B vitamin deficiencies resulting in serious deficiency diseases, such as beriberi, which made their importance to the central nervous system very clear. To avoid these conditions, refined wheat flour and white rice have now long been enriched with vitamins B1-thiamin, B2-riboflavin, and B_3 —niacin, to add back these vitamins that are removed during milling. Until fairly recently, folate, another B vitamin, was an important shortfall nutrient for the aging population, particularly in relation to its roles in DNA methylation and in lowering homocysteine, a metabolite that has been associated with increased risk of heart disease and stroke [55]. In the 1990s, however, grains in the US food supply began to be fortified with folic acid, primarily to ensure adequate intakes for women of childbearing age (because adequate folate status has been shown to reduce the incidence of neural tube defects). Therefore, dietary inadequacy of folic acid is now relatively uncommon in all age groups. Rather, care should be taken to ensure that intakes do not increase beyond the upper level of $1,000 \mu g$ in the form of folic acid. This can easily be done with the combination of vitamin supplements, fortified breakfast cereals, fortified grains, and other foods. Despite the accomplishments of food enrichment and fortification in improving B vitamin nutriture, two B vitamins remain important shortfall nutrients for older adults, vitamin B_6 —pyridoxine, and vitamin B_{12} —cobalamin.

Vitamin B₆

Vitamin B_6 (pyridoxine) appears in several related forms in foods, including pyridoxine, pyridoxal, and pyridoxamine. The active form in the body, pyridoxal 5'-phosphate (PLP), plays a role in the function of more than 100 enzymes. As examples of its many activities, PLP is required for protein metabolism, gluconeogenesis from amino acids, the release of glucose from glycogen, synthesis of neurotransmitters, heme synthesis, conversion of tryptophan to niacin, one-carbon metabolism, and

synthesis of nucleic acids [56, 57]. In addition to folate and vitamin B_{12} , inadequate vitamin B_6 contributes to elevated homocysteine concentration, which, in turn, has been linked with increased risk of heart disease and cognitive decline [58, 59]. Several studies have linked low vitamin B_6 with heart disease [60–63]. One prospective study reported a 30 % lower incidence of coronary heart disease for those in the highest, vs. lowest, quintiles of vitamin B_6 intake [64]. In the longitudinal Veterans Administration Normative Aging Study of men, significant declines in cognitive scores were seen among those in the lowest, vs. highest, tertile of baseline plasma PLP [65]. PLP concentration has also been inversely associated with depressive symptomatology [66]. Reduced production of lymphocytes and interleukin-2 has been associated with vitamin B_6 deficiency, illustrating its importance to immune function [67]. Vitamin B_6 is also inversely correlated with markers of inflammation [68, 69], suggesting that this common condition associated with many chronic diseases may increase requirement for the vitamin [70]. Taken together, the literature on vitamin B_6 makes it clear that this nutrient is important for optimization of metabolism, and that inadequate status may have several effects on health.

Although mean intakes of vitamin B_6 in the 2009–2010 NHANES do not appear to be low in the US population (2.11 and 1.66 mg, relative to the RDA of 1.7 and 1.5 mg, for men and women aged \geq 70 years, respectively), many individuals fall below the recommended intake level [71]. Plasma PLP concentration \geq 30 nmol/L has been defined as normal status [72], but <20 nmol/L is widely used to indicate deficiency. In the Netherlands, from 10 to 45 % of adults aged 65 years and older were found to be deficient in vitamin B_6 [73]. In Massachusetts, 16 % of Hispanic and 11 % of non-Hispanic white adults, aged \geq 60 years, had PLP <20 nmol/L. Vitamin B_6 is found widely in whole foods, including liver, fish, pork, chicken, whole grains, nuts and seeds, legumes, avocados, and bananas, but tends to be low in many of the highly processed foods in the American diet.

Vitamin B₁₂

Vitamin B_{12} is a coenzyme in the pathway that converts homocysteine to methionine, required for the synthesis of *S*-adenosylmethionine, a methyl donor needed for DNA methylation. Vitamin B_{12} is also required for energy metabolism and synthesis of hemoglobin. Inadequate vitamin B_{12} leads to elevations in circulating homocysteine and methylmalonic acid, and can lead to megaloblastic anemia. Vitamin B_{12} deficiency has been linked with depression due to the importance of methylation reactions to the synthesis of neurotransmitters [74]. It is centrally important to maintaining neurological function, as it protects the myelin coating of nerves [75]. Early symptoms of deficiency include tingling and numbness in the hands and feet, and reduction in sense of vibration and position. Additional symptoms vary depending on the location of nerve damage, but may include disturbance in gait, memory loss, disorientation, insomnia, incontinence, visual disturbance, and dementia.

Detection of vitamin B_{12} deficiency has traditionally depended on follow-up after diagnosis of megaloblastic anemia, with assessment of plasma B_{12} . A clinical cutoff point of <148 pmol/L has been considered deficient. However, individuals with clinical evidence of deficiency have been documented at higher concentrations [75], and a higher clinical cutoff point, of 250 pmol/L, is now recommended to identify possible deficiency, followed by the measurement of methylmalonic acid, as a specific metabolic indicator of vitamin B_{12} inadequacy [76, 77]. Because high folic acid exposure may correct megaloblastic anemia even if vitamin B_{12} is deficient, there is concern that the fortification of cereal grains with folic acid in the United States may contribute to more undiagnosed vitamin B_{12} deficiency. Further, some studies have noted that vitamin B_{12} -deficient patients without anemia showed more severe neurological complications than those with megaloblastic anemia [78, 79], suggesting that the imbalance between folate and vitamin B_{12} may accelerate the nerve damage from B_{12} deficiency. Importantly, an analysis of post-fortification NHANES data showed that both anemia and cognitive impairment were significantly more likely in older adults with high plasma folate concentration in combination with low vitamin B_{12} concentration, relative to those without high folate, raising concern

that the recent increased exposure to folic acid may be making the problem of vitamin B_{12} deficiency more serious [80].

Unfortunately, vitamin B_{12} deficiency is relatively common among older adults. Many have difficulty absorbing vitamin B_{12} , due to low stomach acid from a progressive, and often undiagnosed, condition called atrophic gastritis. This condition has been estimated to affect as many as 25–40 % of adults aged ≥ 65 years [81]. Stomach acid is required to separate vitamin B_{12} from protein in food, and without this step, the vitamin B_{12} cannot be absorbed. Therefore, deficiency may exist despite apparently adequate intake. Another contributor to vitamin B_{12} deficiency is the widespread use of acid blocking medications, particularly proton pump inhibitors, which inhibit absorption [82]. Another common medication for diabetes, metformin, has also been shown to lower vitamin B_{12} concentration [83]. Thus, it is advisable to take vitamin B_{12} supplements when using these types of medication.

Although the mean daily intake of total vitamin B_{12} does not appear to be low for most US older adults, relative to the RDA of 2.4 μ g—NHANES 2009–2010 mean intakes for men and women, aged \geq 70 were about twice this amount (see Table 21.1)—deficiency identified by plasma concentration is common. In the Framingham Heart Study, more than 16 % of older adults had low vitamin B₁₂ concentration [84]. In the natural diet, Vitamin B_{12} is found almost exclusively in animal foods, and deficiency of this vitamin is an important concern for strict vegetarians or in populations with limited access to animal products. One study in Latin America [85] found that 51 % of men and 31 % of women had a vitamin B_{12} concentration <148 pmol/L. Because many older adults have difficulty absorbing vitamin B_{12} from food, as discussed above, the Institute of Medicine (IOM) [71] recommends that older adults obtain vitamin B_{12} in its nonbound crystalline form, in either fortified foods (some breakfast cereals) or supplements. The NHANES data now report the amount of added vitamin crystalline B₁₂ consumed; in 2009–2010, men and women, aged 70 years and older reported food intakes with 1.5 and 1.0 μ g/day, respectively, of added B₁₂. Without additional supplements, this remains short of the $2.4 \,\mu g \, RDA$ (see Table 21.1). Supplements can help, but do not guarantee adequacy. The Framingham Study found that supplement users were significantly less likely to have low vitamin B_{12} status than nonsupplement uses, but 4.0 % of supplement users were still identified as deficient [84]. Because of the importance of early treatment, older adults with concerns about vitamin B₁₂ status should have their concentration measured, and if less than 250 pmol/L (350 pg/mL), should request additional measurement of methylmalonic acid.

Vitamin D

Inadequacy of vitamin D is common throughout the population, but is a particular concern for older adults. Vitamin D is unique in that, in addition to dietary sources, it is made in the skin with exposure to UV radiation from the sun: 7-dehydrocholesterol (7-DHC) in the skin is converted to previtamin D_3 and then to vitamin D_3 . Vitamin D_2 or D_3 from the diet, as well as vitamin D_3 from the skin, is converted in the liver to 25-hydroxyvitamin D (25(OH)D), and then in the kidneys to its biologically active form 1,25-dihydroxyvitamin D (1,25(OH)₂D). Older adults are at particular risk of vitamin D deficiency for several reasons. They tend to spend less time outdoors, and therefore have less exposure than younger individuals to the sun for skin conversion. Further, even with exposure, the ability of the skin to convert pre-vitamin D_3 to vitamin D_3 with UV light declines with age, as does the efficiency of the liver and kidneys to convert it to the active form [86]. These limitations are compounded in northern latitudes, where studies have shown that sunlight in the winter months is not adequate to stimulate skin formation of vitamin D. A study in Boston, for example, showed clear seasonal declines in vitamin D status in the winter and spring, relative to the summer and fall [87]. The increased prevalence of obesity is another contributing factor to poor vitamin D status. As a fat-soluble vitamin, there is evidence that vitamin D is sequestered within adipose tissues, which limits its availability for other important processes that are detailed in the following paragraphs [88].

Vitamin D has long been recognized as centrally important for calcium absorption and metabolism, and inadequacy of vitamin D contributes to the high prevalence of low bone mass and osteoporosis, and thereby to the incidence of fracture in the older population [89]. Hip fracture is of tremendous importance as, in addition to temporary disability, it leads to loss of muscle mass and is frequently associated with long-term loss of mobility and increased risk of mortality.

More recently, vitamin D deficiency has been linked with a large number of other conditions, and vitamin D receptors have been identified in diverse tissues in the body including neural, gut, and immune cells. Of great importance is accumulating evidence for the role of vitamin D in neurologic and immunologic conditions. Multiple sclerosis, an immune-mediated disease that leads to progressive degeneration of the myelin sheath of nerves, has long been known to be more prevalent in northern climates. Accumulating evidence has recently shown a clear role for vitamin D in the etiology of multiple sclerosis [90, 91]. In addition, ongoing research is suggesting likely important roles of vitamin D in Parkinson's disease, systemic lupus erythematosus, and Alzheimer's disease [92, 93]. More generally, a recent systematic review of 28 cross-sectional and longitudinal studies supported an association between vitamin D status and cognitive decline in the aging population [94, 95]. A recent systematic review showed significant protective associations between vitamin D and cognitive test scores for 18 of 25 cross-sectional studies, and 4 of 6 longitudinal studies with 4- to 7-year follow-up [96].

In addition, vitamin D has been attracting attention in relation to heart disease and diabetes. One longitudinal study found that men in the highest tertile category of serum 25[OH]D concentration had 44 %, and women 68 %, lower risk of incident myocardial infarction relative to those in the lowest tertile [97]. Another analysis from the same study found a 37 % lower likelihood of incident type 2 diabetes across extreme vitamin D tertiles; they noticed that adjustment for inflammatory markers weakened the relationship, suggesting that subclinical inflammation may be a mediating factor [98]. In an Italian study, obese patients in the lowest (vs. highest) quartile category for serum vitamin D concentration were more than four times more likely to have metabolic syndrome [99, 100]. A recent review concluded that there is accumulating evidence that vitamin D may play a role in the pathogenesis of insulin resistance through several vitamin D-related metabolic and immune pathways [101].

The complex roles of vitamin D have also extended to cancer prevention, as vitamin D has been identified as a regulator of cell growth and differentiation. Population-based studies have shown protective effects against incidence rates of colon, breast, ovarian, renal, pancreatic, prostate, and other cancers [102]. Based on cell culture and animal model studies, a recent review described several biochemical mechanisms that explain these complex cancer protective actions, including anti-proliferation, anti-angiogenesis, pro-apoptosis, pro-differentiation and anti-inflammation activities in cells [103].

Food sources of vitamin D are limited in the US diet and the major sources—fatty fish and fortified milk—are not widely consumed by older adults. The RDA for vitamin D intake is currently 20 µg/day (800 IU/day) for men and women aged >70 years, based primarily on evidence to protect bone health [104]. However, given the growing evidence of the importance of vitamin D for a wide variety of chronic conditions, as discussed above, some scientists consider the cutoff point to be too low [105]. Still, current intakes by older adults fall well below the recommended level. The 2009–2010 NHANES reported mean intakes of only 5.8 µg/day in men and 4.4 in women, aged \geq 70 years. Plasma measures in the 2001–2006 NHANES, using the IOM definitions of <30 nmol/L as risk of deficiency and 30–49 nmol/L as risk of inadequacy, showed prevalent risk of deficiency for 7 % of men and 11 % of women over age 70 years; and of inadequacy for another 24 % of men and 27 % of women [106]. Importantly, these risks were significantly higher in African Americans and Mexican Americans than in non-Hispanic white adults.

Efforts to improve vitamin D status include increasing intake of fatty fish and fortified foods, including milk, and some yogurt and breakfast cereal brands (check the label). Certain mushrooms, exposed to UV light, contain vitamin D and are becoming available. However, it is difficult for most individuals to obtain sufficient vitamin D from diet alone. Just 10 min/day of sunlight at midday can

make a large difference in vitamin D status. However, many individuals are now using sunscreen, which interferes with this. Therefore many experts now recommend that sunscreen be used only after 10–15 min of exposure. In the winter months at high latitudes, or for individuals who cannot get out, supplements are recommended.

Vitamin E

Current recommendations for vitamin E intake are based on an increasing understanding of the importance of this potent antioxidant in the prevention of age-related declines in a variety of functions. Vitamin E has also been shown to be effective in promoting immune function to fight infection [107, 108]. There is observational evidence of protective effects of vitamin E intakes on plasma status against cardiovascular disease, cognitive decline, cataract, and all-cause mortality [109–113]. However, as with β -carotene, trials with α -tocopherol have been disappointing–showing, for the most part, no benefit for preventing heart disease events, incident cancer or mortality, in large populations of adults on the supplement for 3–10 years, vs. placebo [114–118]. Rather, one trial showed higher risk of heart failure [119], and another, an increased risk of prostate cancer [120] among groups assigned to the vitamin E supplement. A trial of vitamin E supplements to evaluate whether it could reduce risk of Alzheimer's disease also showed no benefit [121].

Vitamin E supplements in the form of α -tocopherol have been shown to reduce plasma γ -tocopherol, which may also be important for optimal health [122]. Vitamin E in nature includes differing forms of tocopherols and also tocotrienols. Although there is evidence that forms of vitamin E other than α -tocopherol are important for many aspects of health [123], including a potential role in prevention of breast cancer [124], the current RDA provides guidance only on α -tocopherol. These findings reinforce the importance of obtaining vitamin E from diet rather than from α -tocopherol supplements.

Most individuals do not come close to meeting the RDA for vitamin E, of 15 mg α -tocopherol per day [125]. Among NHANES 2009–2010 adults aged \geq 70 years, the mean vitamin E intakes were 8.2 µg/day for men and 6.3 µg/day for women. Vitamin E is another nutrient that is found in large quantities only in selected foods. The major dietary source is vegetable oil. Nuts and seeds, particularly almonds and sunflower seeds, have the highest concentrations [126]. A dietary simulation, to identify ways to improve vitamin E intake within a healthy dietary pattern, found that it would be very difficult to meet the RDA without the inclusion of nuts and seeds [127].

Minerals

As are vitamins, essential minerals are available in a diverse natural food supply. However, many of these are easily lost with processing and some are concentrated in limited food groups. Inadequacies particularly in calcium, magnesium, and potassium have important health implications for older adults and will be discussed in the following sections. First, however, potential concerns about excessive levels of two essential minerals will be mentioned. Iron, which is a shortfall nutrient throughout most of the life cycle, accumulates with age, and high serum ferritin has been associated with greater risk of coronary heart disease [128]. Thus, care should be taken by older adults to avoid too much iron in the heme form (from red and processed meats) or from supplements. Another essential mineral consumed in excess in our modern food supply is phosphorus. This mineral, which is naturally occurring in protein foods, has been added to a wide array of processed foods, including colas, basted or processed meats, and many fast and convenience foods. Excess intakes appear to be increasingly common and have been linked with adverse effects on

bone, kidney, and heart health [129]. Phosphorus intakes outside the context of natural sources may be particularly harmful, such as in cola drinks where they are not balanced by other nutrients and may have regular periodic negative effects even without large doses. For example, regular consumption of cola drinks has been linked with low bone mineral density in women [130].

Calcium

Calcium is a critically important mineral in the body as an intracellular messenger, regulating diverse cellular processes, and as a cofactor for extracellular enzymes and proteins. It is maintained in tight equilibrium in the extracellular fluid. It is stored primarily in the hydroxyapatite of bone, where it is available for release as needed to maintain plasma concentrations. Adequate calcium intake is, therefore, essential to prevent bone loss associated with aging and related risk of fracture. Bone loss is a concern for all older adults, but accelerates particularly in post-menopausal women due to effects from loss of estrogen. Declines in renal function and impaired absorption of calcium with aging further contribute to bone loss [131, 132]. In addition to the well-known importance for bone status, calcium intake is also important for blood pressure regulation [133].

Few adults meet the intake recommendation of 1,200 mg/day, and milk and low fat dairy products tend to be consumed less frequently by older adults than by younger individuals [134]. Some researchers have questioned the need for this high amount of calcium, and rather, focus on the importance of balancing vitamin D intake with somewhat lower calcium intake levels [135]. Because older adults do not meet dietary recommendations for calcium intake, physicians frequently recommend supplements, which are taken, particularly, by a large proportion of postmenopausal women. However, there is some evidence of harmful effects of high intake of calcium from supplements. For example, one study showed increased risk of heart disease and total mortality among women with calcium intakes greater than 1,400 mg/day, relative to those consuming between 600 and 1,000 mg/day, particularly if they were taking calcium supplements [136]. Other health concerns linked with calcium supplementation include kidney stones [137] and prostate cancer [138], although these remain controversial.

Taken together, the evidence suggests that it is important to balance calcium intake with other nutrients involved in its absorption and metabolism. Whole food sources of calcium, thus, are likely to have better long-term protective effects for bone status and less risk of imbalances that may have negative side effects than calcium supplements. There are many advantages to consuming calcium-rich foods, including low fat dairy products, fish (with bones, like sardines or canned salmon), and leafy green vegetables. Improved intakes of such dietary sources should be encouraged as, in addition to calcium, they also tend to include vitamin D, magnesium, potassium, and other important nutrients.

Magnesium

Numerous metabolic reactions in the body require magnesium. Among these, magnesium is essential for ATP utilization, protein and lipid metabolism, amino acid activation, DNA transcription, membrane fluidity, ion channel function, and calcium metabolism [139]. Although obvious deficiency is rare, subclinical inadequacy may contribute to risk for a variety of chronic conditions, including heart disease, sudden death from cardiac arrhythmia, osteoporosis, and type 2 diabetes [140]. A recent review noted that low magnesium intakes have also been associated with metabolic syndrome, inflammation, hypertension, vascular disease, migraine headache, asthma, and colon cancer [141]. These authors suggest that increasing calcium to magnesium ratios may also factor in the risk for these conditions.
Dietary surveys consistently show that the majority of the adult population falls short of meeting the magnesium intake recommendation [142]. Like vitamin B_6 , magnesium is found in many whole foods, but is easily lost with processing. With our increasingly processed food supply, magnesium intakes in the population have been shown to decline over the past century [143]. Importantly, the majority of older adults have low magnesium intakes. Excellent sources include whole grains, nuts, legumes, and fresh vegetables. Increasing intakes of these foods would also improve other nutrient profiles, including B vitamins and dietary fiber.

Potassium

Potassium is the major intracellular cation, and it is tightly regulated in body fluids because of its importance to cardiac rhythm and conduction, and function of skeletal muscle. It is centrally involved in cellular metabolism and transport across cell membranes. Hypokalemia can lead to cardiac arrhythmia and muscle cell necrosis. This condition is usually caused by clinical conditions, including vomiting and diarrhea, rather than by poor intake, but can also result from medications such as diuretics [144]. Subclinical inadequacy due to low potassium intake has been linked to increased risk of hypertension, cardiovascular disease, and renal disease, and to acid–base balance disturbances, glucose regulation disturbances and bone loss [145].

Potassium intake is another mineral that is particularly low in most modern diets. As with magnesium, the majority of older adults have potassium intakes that fall well below recommendations. Fruits, vegetables, and low fat dairy products are excellent sources of potassium that are inadequately consumed by most adults. Increasing intakes of these will also help to improve intakes of vitamin C, dietary fiber, carotenoids, and other antioxidant and anti-inflammatory phytonutrients, calcium, magnesium and, in the case of fortified milk or yogurt, vitamin D.

Conclusion

Evidence of the critical importance of nutrition in protecting against the development and progression of chronic conditions with aging continues to grow. Older adults require less energy, experience lessefficient absorption and utilization of many nutrients, and yet can have greater requirements for many nutrients than younger adults, due to declining organ function, chronic conditions, and medications. At the same time, they face increasing risk for health conditions that may be adversely affected by inadequate or unbalanced nutrient intakes. This poses significant challenges, as many older individuals also experience changes in taste and smell, loss of appetite, dental and chewing problems, and limitations in mobility and access to high quality fresh food.

As noted above, dietary surveys consistently show that older adults fall short in their intake of several individual nutrients. In linking nutrition with chronic disease risk and aging, most studies have focused on the role of single nutrients. This led to trials to test the efficacy of nutrient supplements for reducing risk of heart disease, cancer, osteoporosis, and other important chronic conditions. The failure of many of those trials to show benefit, along with the evidence of protective effects from whole foods, reinforces the importance of emphasizing nutrient-dense foods and varied whole foods dietary patterns, and to work toward improving access to quality foods for the older population.

Relative to food group intake recommendations, older adults tend to report inadequate intakes of fruit, vegetables, legumes, whole grains, nuts or seeds, fish, lean meat, poultry, and low-fat fluid dairy products. Because of lower energy needs, the importance of these nutrient-dense food groups cannot be overstated. Rather, most older adults have excess energy intake from refined grain products,

processed and fatty meats, fried foods, solid fats, and added sugars, which can lead to obesity without providing the nutrients critically needed to maximize health [146].

In some cases, supplements may be helpful, and these are certainly important in the case of nutrient deficiencies, or when health conditions or medications interfere with absorption or effective nutrient utilization of specific nutrients, making it difficult or impossible to obtain adequate intakes from diet alone. Examples of these, as discussed above, include vitamin B_{12} with atrophic gastritis, use of acid blocking medication, metformin or other interfering medications, and vitamin D for individuals who get inadequate sun exposure, and during the winter months in northern latitudes.

In summary, increasing intakes of nutrient-dense whole foods, while reducing energy-dense refined and processed foods, with judicious use of supplements when needed, is of critical importance for optimizing metabolism, protecting cellular and organ function and maintaining health with aging.

References

- 1. Russell RM. Micronutrient requirements of the elderly. Nutr Rev. 1992;50(12):463-6.
- 2. Chan LN. Drug-nutrient interactions. J Parenter Enteral Nutr. 2013;37(4):450-9.
- Baumgartner RN, Stauber PM, McHugh D, Koehler KM, Garry PJ. Cross-sectional age differences in body composition in persons 60+ years of age. J Gerontol A Biol Sci Med Sci. 1995;50(6):M307–16.
- 4. Baumgartner RN. Body composition in healthy aging. Ann N Y Acad Sci. 2000;904:437–48.
- 5. Evans WJ, Cyr-Campbell D. Nutrition, exercise, and healthy aging. J Am Diet Assoc. 1997;97(6):632–8.
- Houston DK, Nicklas BJ, Ding J, Harris TB, Tylavsky FA, Newman AB, et al. Dietary protein intake is associated with lean mass change in older, community-dwelling adults: the Health, Aging, and Body Composition (Health ABC) Study. Am J Clin Nutr. 2008;87(1):150–5.
- 7. Barzel US, Massey LK. Excess dietary protein can adversely affect bone. J Nutr. 1998;128(6):1051-3.
- Kerstetter JE, O'Brien KO, Insogna KL. Low protein intake: the impact on calcium and bone homeostasis in humans. J Nutr. 2003;133(3):855S–61.
- Hannan MT, Tucker KL, Dawson-Hughes B, Cupples LA, Felson DT, Kiel DP. Effect of dietary protein on bone loss in elderly men and women: the Framingham Osteoporosis Study. J Bone Miner Res. 2000;15(12):2504–12.
- Promislow JH, Goodman-Gruen D, Slymen DJ, Barrett-Connor E. Protein consumption and bone mineral density in the elderly: the Rancho Bernardo Study. Am J Epidemiol. 2002;155(7):636–44.
- Munger RG, Cerhan JR, Chiu BC. Prospective study of dietary protein intake and risk of hip fracture in postmenopausal women. Am J Clin Nutr. 1999;69(1):147–52.
- Darling AL, Millward DJ, Torgerson DJ, Hewitt CE, Lanham-New SA. Dietary protein and bone health: a systematic review and meta-analysis. Am J Clin Nutr. 2009;90(6):1674–92.
- Gaffney-Stomberg E, Insogna KL, Rodriguez NR, Kerstetter JE. Increasing dietary protein requirements in elderly people for optimal muscle and bone health. J Am Geriatr Soc. 2009;57(6):1073–9.
- Paddon-Jones D, Rasmussen BB. Dietary protein recommendations and the prevention of sarcopenia. Curr Opin Clin Nutr Metab Care. 2009;12(1):86–90.
- Paddon-Jones D, Short KR, Campbell WW, Volpi E, Wolfe RR. Role of dietary protein in the sarcopenia of aging. Am J Clin Nutr. 2008;87(5):1562S–6.
- Bauer J, Biolo G, Cederholm T, Cesari M, Cruz-Jentoft AJ, Morley JE, et al. Evidence-based recommendations for optimal dietary protein intake in older people: a position paper from the PROT-AGE Study Group. J Am Med Dir Assoc. 2013;14(8):542–59.
- Food and Nutrition Board, Institute of Medicine. Dietary reference intakes for energy, carbohydrate, fiber, fat, fatty acids, cholesterol, protein, and amino acids. Washington, DC: National Academies Press; 2002.
- He K, Song Y, Daviglus ML, Liu K, Van Horn L, Dyer AR, et al. Accumulated evidence on fish consumption and coronary heart disease mortality: a meta-analysis of cohort studies. Circulation. 2004;109(22):2705–11.
- Daviglus ML, Stamler J, Orencia AJ, Dyer AR, Liu K, Greenland P, et al. Fish consumption and the 30-year risk of fatal myocardial infarction. N Engl J Med. 1997;336(15):1046–53.
- He K, Rimm EB, Merchant A, Rosner BA, Stampfer MJ, Willett WC, et al. Fish consumption and risk of stroke in men. JAMA. 2002;288(24):3130–6.
- Kalmijn S, Launer LJ, Ott A, Witteman JC, Hofman A, Breteler MM. Dietary fat intake and the risk of incident dementia in the Rotterdam Study. Ann Neurol. 1997;42(5):776–82.
- 22. Morris MC, Evans DA, Bienias JL, Tangney CC, Bennett DA, Wilson RS, et al. Consumption of fish and n-3 fatty acids and risk of incident Alzheimer disease. Arch Neurol. 2003;60(7):940–6.

- Laurin D, Verreault R, Lindsay J, Dewailly E, Holub BJ. Omega-3 fatty acids and risk of cognitive impairment and dementia. J Alzheimers Dis. 2003;5(4):315–22.
- Schaefer EJ, Bongard V, Beiser AS, Lamon-Fava S, Robins SJ, Au R, et al. Plasma phosphatidylcholine docosahexaenoic acid content and risk of dementia and Alzheimer disease: the Framingham Heart Study. Arch Neurol. 2006;63(11):1545–50.
- Calder PC. N-3 polyunsaturated fatty acids and inflammation: from molecular biology to the clinic. Lipids. 2003;38(4):343–52.
- de Winther MP, Kanters E, Kraal G, Hofker MH. Nuclear factor kappaB signaling in atherogenesis. Arterioscler Thromb Vasc Biol. 2005;25(5):904–14.
- Kromhout D, Yasuda S, Geleijnse JM, Shimokawa H. Fish oil and omega-3 fatty acids in cardiovascular disease: do they really work? Eur Heart J. 2012;33(4):436–43.
- Chowdhury R, Stevens S, Gorman D, Pan A, Warnakula S, Chowdhury S, et al. Association between fish consumption, long chain omega 3 fatty acids, and risk of cerebrovascular disease: systematic review and meta-analysis. BMJ. 2012;345:e6698.
- Lin PY, Chiu CC, Huang SY, Su KP. A meta-analytic review of polyunsaturated fatty acid compositions in dementia. J Clin Psychiatry. 2012;73(9):1245–54.
- 30. Witte AV, Kerti L, Hermannstadter HM, Fiebach JB, Schreiber SJ, Schuchardt JP, et al. Long-chain omega-3 fatty acids improve brain function and structure in older adults. Cereb Cortex. 2013 [Epub ahead of print]
- Hashimoto M, Hossain S. Neuroprotective and ameliorative actions of polyunsaturated fatty acids against neuronal diseases: beneficial effect of docosahexaenoic acid on cognitive decline in Alzheimer's disease. J Pharmacol Sci. 2011;116(2):150–62.
- Harris WS, Mozaffarian D, Lefevre M, Toner CD, Colombo J, Cunnane SC, et al. Towards establishing dietary reference intakes for eicosapentaenoic and docosahexaenoic acids. J Nutr. 2009;139(4):804S–19.
- 33. Stark AH, Crawford MA, Reifen R. Update on alpha-linolenic acid. Nutr Rev. 2008;66(6):326-32.
- 34. Carey AN, Fisher DR, Joseph JA, Shukitt-Hale B. The ability of walnut extract and fatty acids to protect against the deleterious effects of oxidative stress and inflammation in hippocampal cells. Nutr Neurosci. 2013;16(1):13–20.
- Barros R, Moreira A, Fonseca J, Delgado L, Castel-Branco MG, Haahtela T, et al. Dietary intake of alpha-linolenic acid and low ratio of n-6:n-3 PUFA are associated with decreased exhaled NO and improved asthma control. Br J Nutr. 2011;106(3):441–50.
- Hutchins AM, Brown BD, Cunnane SC, Domitrovich SG, Adams ER, Bobowiec CE. Daily flaxseed consumption improves glycemic control in obese men and women with pre-diabetes: a randomized study. Nutr Res. 2013;33(5):367–75.
- Kris-Etherton PM, Harris WS, Appel LJ. Fish consumption, fish oil, omega-3 fatty acids, and cardiovascular disease. Circulation. 2002;106(21):2747–57.
- Sturtzel B, Mikulits C, Gisinger C, Elmadfa I. Use of fiber instead of laxative treatment in a geriatric hospital to improve the well-being of seniors. J Nutr Health Aging. 2009;13(2):136–9.
- Bazzano LA. Effects of soluble dietary fiber on low-density lipoprotein cholesterol and coronary heart disease risk. Curr Atheroscler Rep. 2008;10(6):473–7.
- Slavin J, Jacobs D. Dietary fiber: all fibers are not alike. In: Wilson T, Bray GA, Temple NJ, Struble MB, editors. Nutrition guide for physicians. New York, NY: Humana Press; 2010. p. 13–24.
- 41. Russell RM. The vitamin A spectrum: from deficiency to toxicity. Am J Clin Nutr. 2000;71(4):878-84.
- 42. Krasinski SD, Cohn JS, Schaefer EJ, Russell RM. Postprandial plasma retinyl ester response is greater in older subjects compared with younger subjects. Evidence for delayed plasma clearance of intestinal lipoproteins. J Clin Invest. 1990;85(3):883–92.
- 43. Crandall C. Vitamin A, intake and osteoporosis: a clinical review. J Womens Health (Larchmt). 2004;13(8):939–53.
- Michaelsson K, Lithell H, Vessby B, Melhus H. Serum retinol levels and the risk of fracture. N Engl J Med. 2003;348(4):287–94.
- 45. Johnson EJ, Vishwanathan R, Johnson MA, Hausman DB, Davey A, Scott TM, et al. Relationship between serum and brain carotenoids, alpha-tocopherol, and retinol concentrations and cognitive performance in the oldest old from the Georgia Centenarian Study. J Aging Res. 2013;2013:951786.
- Voutilainen S, Nurmi T, Mursu J, Rissanen TH. Carotenoids and cardiovascular health. Am J Clin Nutr. 2006;83(6):1265–71.
- 47. Mannisto S, Smith-Warner SA, Spiegelman D, Albanes D, Anderson K, van den Brandt PA, et al. Dietary carotenoids and risk of lung cancer in a pooled analysis of seven cohort studies. Cancer Epidemiol Biomarkers Prev. 2004;13(1):40–8.
- 48. Mares-Perlman JA, Millen AE, Ficek TL, Hankinson SE. The body of evidence to support a protective role for lutein and zeaxanthin in delaying chronic disease. Overview. J Nutr. 2002;132(3):518S–24.
- 49. Dahan K, Fennal M, Kumar NB. Lycopene in the prevention of prostate cancer. J Soc Integr Oncol. 2008;6(1): 29–36.

- 50. The effect of vitamin E and beta carotene on the incidence of lung cancer and other cancers in male smokers. The Alpha-Tocopherol, Beta Carotene Cancer Prevention Study Group. N Engl J Med. 1994;330(15):1029–35.
- Omenn GS, Goodman GE, Thornquist MD, Balmes J, Cullen MR, Glass A, et al. Risk factors for lung cancer and for intervention effects in CARET, the Beta-Carotene and Retinol Efficacy Trial. J Natl Cancer Inst. 1996;88(21):1550–9.
- Hennekens CH, Buring JE, Manson JE, Stampfer M, Rosner B, Cook NR, et al. Lack of effect of long-term supplementation with beta carotene on the incidence of malignant neoplasms and cardiovascular disease. N Engl J Med. 1996;334(18):1145–9.
- Kris-Etherton PM, Hecker KD, Bonanome A, Coval SM, Binkoski AE, Hilpert KF, et al. Bioactive compounds in foods: their role in the prevention of cardiovascular disease and cancer. Am J Med. 2002;113(Suppl 9B):71S–88.
- 54. Van Duyn MA, Pivonka E. Overview of the health benefits of fruit and vegetable consumption for the dietetics professional: selected literature. J Am Diet Assoc. 2000;100(12):1511–21.
- Homocysteine Studies Collaboration. Homocysteine and risk of ischemic heart disease and stroke: a meta-analysis. JAMA. 2002;288(16):2015–22.
- Leklem JE. Vitamin B6. In: Shils ME, Olson JA, Shike M, Ross AC, editors. Modern nutrition in health and disease. 9th ed. Baltimore, MD: Williams & Wilkins; 1999. p. 413–22.
- Mackey AD, Davis SR, Gregory III JF. Vitamin B6. In: Shils ME, Shike M, Ross AC, Caballero B, Cousins RJ, editors. Modern nutrition in health and disease. 3rd ed. Philadelphia, PA: Lippincott Williams & Wilkins; 2006. p. 452–61.
- Selhub J, Jacques PF, Bostom AG, D'Agostino RB, Wilson PW, Belanger AJ, et al. Relationship between plasma homocysteine, vitamin status and extracranial carotid-artery stenosis in the Framingham Study population. J Nutr. 1996;126(4 Suppl):1258S–65.
- Stampfer MJ, Malinow MR, Willett WC, Newcomer LM, Upson B, Ullmann D, et al. A prospective study of plasma homocyst(e)ine and risk of myocardial infarction in US physicians. JAMA. 1992;268(7):877–81.
- 60. Folsom AR, Nieto FJ, McGovern PG, Tsai MY, Malinow MR, Eckfeldt JH, et al. Prospective study of coronary heart disease incidence in relation to fasting total homocysteine, related genetic polymorphisms, and B vitamins: the Atherosclerosis Risk in Communities (ARIC) study. Circulation. 1998;98(3):204–10.
- Robinson K, Arheart K, Refsum H, Brattstrom L, Boers G, Ueland P, et al. Low circulating folate and vitamin B6 concentrations: risk factors for stroke, peripheral vascular disease, and coronary artery disease. European COMAC Group. Circulation. 1998;97(5):437–43.
- Robinson K, Mayer EL, Miller DP, Green R, van Lente F, Gupta A, et al. Hyperhomocysteinemia and low pyridoxal phosphate. Common and independent reversible risk factors for coronary artery disease. Circulation. 1995;92(10):2825–30.
- 63. Lin PT, Cheng CH, Liaw YP, Lee BJ, Lee TW, Huang YC. Low pyridoxal 5'-phosphate is associated with increased risk of coronary artery disease. Nutrition. 2006;22(11–12):1146–51.
- 64. Rimm EB, Willett WC, Hu FB, Sampson L, Colditz GA, Manson JE, et al. Folate and vitamin B6 from diet and supplements in relation to risk of coronary heart disease among women. JAMA. 1998;279(5):359–64.
- Tucker KL, Qiao N, Scott T, Rosenberg I, Spiro III A. High homocysteine and low B vitamins predict cognitive decline in aging men: the Veterans Affairs Normative Aging Study. Am J Clin Nutr. 2005;82(3):627–35.
- Merete C, Falcon LM, Tucker KL. Vitamin B6 is associated with depressive symptomatology in Massachusetts elders. J Am Coll Nutr. 2008;27(3):421–7.
- Meydani SN, Ribaya-Mercado JD, Russell RM, Sahyoun N, Morrow FD, Gershoff SN. Vitamin B-6 deficiency impairs interleukin 2 production and lymphocyte proliferation in elderly adults. Am J Clin Nutr. 1991;53(5): 1275–80.
- Friso S, Jacques PF, Wilson PW, Rosenberg IH, Selhub J. Low circulating vitamin B(6) is associated with elevation of the inflammation marker C-reactive protein independently of plasma homocysteine levels. Circulation. 2001;103(23):2788–91.
- Shen J, Lai CQ, Mattei J, Ordovas JM, Tucker KL. Association of vitamin B-6 status with inflammation, oxidative stress, and chronic inflammatory conditions: the Boston Puerto Rican Health Study. Am J Clin Nutr. 2010;91(2): 337–42.
- 70. Paul L, Ueland PM, Selhub J. Mechanistic perspective on the relationship between pyridoxal 5'-phosphate and inflammation. Nutr Rev. 2013;71(4):239–44.
- Food and Nutrition Board, Institute of Medicine. Dietary reference intakes for thiamin, riboflavin, niacin, vitamin B6, folate, vitamin B12, pantothenic acid, biotin, and choline. Washington, DC: National Academy Press; 1998.
- 72. Leklem JE. Vitamin B-6: a status report. J Nutr. 1990;120 Suppl 11:1503-7.
- Tolonen M, Schrijver J, Westermarck T, Halme M, Tuominen SE, Frilander A, et al. Vitamin B6 status of Finnish elderly. Comparison with Dutch younger adults and elderly. The effect of supplementation. Int J Vitam Nutr Res. 1988;58(1):73–7.
- 74. Tiemeier H, van Tuijl HR, Hofman A, Meijer J, Kiliaan AJ, Breteler MM. Vitamin B12, folate, and homocysteine in depression: the Rotterdam Study. Am J Psychiatry. 2002;159(12):2099–101.

- Lindenbaum J, Healton EB, Savage DG, Brust JC, Garrett TJ, Podell ER, et al. Neuropsychiatric disorders caused by cobalamin deficiency in the absence of anemia or macrocytosis. N Engl J Med. 1988;318(26):1720–8.
- Savage DG, Lindenbaum J, Stabler SP, Allen RH. Sensitivity of serum methylmalonic acid and total homocysteine determinations for diagnosing cobalamin and folate deficiencies. Am J Med. 1994;96(3):239–46.
- Refsum H, Smith AD, Ueland PM, Nexo E, Clarke R, McPartlin J, et al. Facts and recommendations about total homocysteine determinations: an expert opinion. Clin Chem. 2004;50(1):3–32.
- Healton EB, Savage DG, Brust JC, Garrett TJ, Lindenbaum J. Neurologic aspects of cobalamin deficiency. Medicine (Baltimore). 1991;70(4):229–45.
- Savage D, Gangaidzo I, Lindenbaum J, Kiire C, Mukiibi JM, Moyo A, et al. Vitamin B12 deficiency is the primary cause of megaloblastic anaemia in Zimbabwe. Br J Haematol. 1994;86(4):844–50.
- Selhub J, Morris MS, Jacques PF, Rosenberg IH. Folate-vitamin B-12 interaction in relation to cognitive impairment, anemia, and biochemical indicators of vitamin B-12 deficiency. Am J Clin Nutr. 2009;89(2):702S–6.
- Krasinski SD, Russell RM, Samloff IM, Jacob RA, Dallal GE, McGandy RB, et al. Fundic atrophic gastritis in an elderly population. Effect on hemoglobin and several serum nutritional indicators. J Am Geriatr Soc. 1986; 34(11):800–6.
- Dharmarajan TS, Kanagala MR, Murakonda P, Lebelt AS, Norkus EP. Do acid-lowering agents affect vitamin B12 status in older adults? J Am Med Dir Assoc. 2008;9(3):162–7.
- 83. Xu L, Huang Z, He X, Wan X, Fang D, Li Y. Adverse effect of metformin therapy on serum vitamin B12 and folate: short-term treatment causes disadvantages? Med Hypotheses. 2013;81(2):149–51.
- Tucker KL, Rich S, Rosenberg I, Jacques P, Dallal G, Wilson PW, et al. Plasma vitamin B-12 concentrations relate to intake source in the Framingham Offspring study. Am J Clin Nutr. 2000;71(2):514–22.
- Olivares M, Hertrampf E, Capurro MT, Wegner D. Prevalence of anemia in elderly subjects living at home: role of micronutrient deficiency and inflammation. Eur J Clin Nutr. 2000;54(11):834–9.
- Holick MF, Chen TC. Vitamin D deficiency: a worldwide problem with health consequences. Am J Clin Nutr. 2008;87(4):1080S–6.
- Salamone LM, Dallal GE, Zantos D, Makrauer F, Dawson-Hughes B. Contributions of vitamin D intake and seasonal sunlight exposure to plasma 25-hydroxyvitamin D concentration in elderly women. Am J Clin Nutr. 1994;59(1):80–6.
- Wortsman J, Matsuoka LY, Chen TC, Lu Z, Holick MF. Decreased bioavailability of vitamin D in obesity. Am J Clin Nutr. 2000;72(3):690–3.
- 89. Gennari C. Calcium and vitamin D nutrition and bone disease of the elderly. Public Health Nutr. 2001;4(2B): 547–59.
- Simon KC, Munger KL, Ascherio A. Vitamin D and multiple sclerosis: epidemiology, immunology, and genetics. Curr Opin Neurol. 2012;25(3):246–51.
- VanAmerongen BM, Dijkstra CD, Lips P, Polman CH. Multiple sclerosis and vitamin D: an update. Eur J Clin Nutr. 2004;58(8):1095–109.
- 92. L Ng K, Nguyen L. Role of vitamin D in Parkinson's disease. ISRN Neurol. 2012;2012:134289.
- vinh quoc Luong K, Nguyen LT. The beneficial role of vitamin D in systemic lupus erythematosus (SLE). Clin Rheumatol. 2012;31(10):1423–35.
- 94. Buell JS, Tucker KL. The value of physiologic vitamin D as a biomarker of dementia. Drugs Today (Barc). 2011;47(3):223–31.
- Soni M, Kos K, Lang IA, Jones K, Melzer D, Llewellyn DJ. Vitamin D and cognitive function. Scand J Clin Lab Invest Suppl. 2012;243:79–82.
- 96. van der Schaft J, Koek HL, Dijkstra E, Verhaar HJ, van der Schouw YT, Emmelot-Vonk MH. The association between vitamin D and cognition: a systematic review. Ageing Res Rev. 2013;12:1013–23.
- 97. Karakas M, Thorand B, Zierer A, Huth C, Meisinger C, Roden M, et al. Low levels of serum 25-hydroxyvitamin D are associated with increased risk of myocardial infarction, especially in women: results from the MONICA/ KORA Augsburg case-cohort study. J Clin Endocrinol Metab. 2013;98(1):272–80.
- Thorand B, Zierer A, Huth C, Linseisen J, Meisinger C, Roden M, et al. Effect of serum 25-hydroxyvitamin D on risk for type 2 diabetes may be partially mediated by subclinical inflammation: results from the MONICA/KORA Augsburg study. Diabetes Care. 2011;34(10):2320–2.
- Barchetta I, De Bernardinis M, Capoccia D, Baroni MG, Fontana M, Fraioli A, et al. Hypovitaminosis D is independently associated with metabolic syndrome in obese patients. PLoS One. 2013;8(7):e68689.
- 100. Niino M. Vitamin D, and its immunoregulatory role in multiple sclerosis. Drugs Today (Barc). 2010;46(4): 279–90.
- $101. \ Sung CC, Liao MT, Lu KC, Wu CC. Role of vitamin D in insulin resistance. J Biomed Biotechnol. 2012; 2012: 634195.$
- 102. Garland CF, Garland FC, Gorham ED, Lipkin M, Newmark H, Mohr SB, et al. The role of vitamin D in cancer prevention. Am J Public Health. 2006;96(2):252–61.
- 103. Chiang KC, Chen TC. The anti-cancer actions of vitamin D. Anticancer Agents Med Chem. 2013;13(1):126–39.

- Food and Nutrition Board, Institute of Medicine. Dietary reference intakes for calcium and vitamin D. Washington, DC: National Academy Press; 2011.
- 105. Holick MF. The IOM D-lemma. Public Health Nutr. 2011;14(5):939-41.
- 106. Looker AC, Johnson CL, Lacher DA, Pfeiffer CM, Schleicher RL, Sempos CT. Vitamin D status: United States, 2001–2006. NCHS Data Brief. 2011;59:1–8.
- 107. Meydani SN, Meydani M, Blumberg JB, Leka LS, Siber G, Loszewski R, et al. Vitamin E supplementation and in vivo immune response in healthy elderly subjects. A randomized controlled trial. JAMA. 1997;277(17): 1380–6.
- 108. Meydani SN, Leka LS, Fine BC, Dallal GE, Keusch GT, Singh MF, et al. Vitamin E and respiratory tract infections in elderly nursing home residents: a randomized controlled trial. JAMA. 2004;292(7):828–36.
- Stampfer MJ, Hennekens CH, Manson JE, Colditz GA, Rosner B, Willett WC. Vitamin E consumption and the risk of coronary disease in women. N Engl J Med. 1993;328(20):1444–9.
- Rimm EB, Stampfer MJ, Ascherio A, Giovannucci E, Colditz GA, Willett WC. Vitamin E consumption and the risk of coronary heart disease in men. N Engl J Med. 1993;328(20):1450–6.
- 111. Morris MC, Evans DA, Bienias JL, Tangney CC, Wilson RS. Vitamin E and cognitive decline in older persons. Arch Neurol. 2002;59(7):1125–32.
- 112. Jacques PF, Taylor A, Moeller S, Hankinson SE, Rogers G, Tung W, et al. Long-term nutrient intake and 5-year change in nuclear lens opacities. Arch Ophthalmol. 2005;123(4):517–26.
- 113. Wright ME, Lawson KA, Weinstein SJ, Pietinen P, Taylor PR, Virtamo J, et al. Higher baseline serum concentrations of vitamin E are associated with lower total and cause-specific mortality in the Alpha-Tocopherol, Beta-Carotene Cancer Prevention Study. Am J Clin Nutr. 2006;84(5):1200–7.
- 114. Dietary supplementation with n-3 polyunsaturated fatty acids and vitamin E after myocardial infarction: results of the GISSI-Prevenzione trial. Gruppo Italiano per lo Studio della Sopravvivenza nell'Infarto miocardico. Lancet. 1999;354(9177):447–55.
- 115. Yusuf S, Dagenais G, Pogue J, Bosch J, Sleight P. Vitamin E supplementation and cardiovascular events in high-risk patients. The Heart Outcomes Prevention Evaluation Study Investigators. N Engl J Med. 2000;342(3):154–60.
- 116. Lee IM, Cook NR, Gaziano JM, Gordon D, Ridker PM, Manson JE, et al. Vitamin E in the primary prevention of cardiovascular disease and cancer: the Women's Health Study: a randomized controlled trial. JAMA. 2005;294(1): 56–65.
- 117. Hercberg S, Galan P, Preziosi P, Bertrais S, Mennen L, Malvy D, et al. The SU.VI.MAX Study: a randomized, placebo-controlled trial of the health effects of antioxidant vitamins and minerals. Arch Intern Med. 2004;164(21): 2335–42.
- 118. Sesso HD, Buring JE, Christen WG, Kurth T, Belanger C, MacFadyen J, et al. Vitamins E and C in the prevention of cardiovascular disease in men: the Physicians' Health Study II randomized controlled trial. JAMA. 2008; 300(18):2123–33.
- 119. Lonn E, Bosch J, Yusuf S, Sheridan P, Pogue J, Arnold JM, et al. Effects of long-term vitamin E supplementation on cardiovascular events and cancer: a randomized controlled trial. JAMA. 2005;293(11):1338–47.
- 120. Klein EA, Thompson Jr IM, Tangen CM, Crowley JJ, Lucia MS, Goodman PJ, et al. Vitamin E and the risk of prostate cancer: the Selenium and Vitamin E Cancer Prevention Trial (SELECT). JAMA. 2011;306(14):1549–56.
- 121. Petersen RC, Thomas RG, Grundman M, Bennett D, Doody R, Ferris S, et al. Vitamin E and donepezil for the treatment of mild cognitive impairment. N Engl J Med. 2005;352(23):2379–88.
- 122. Cooney RV, Franke AA, Harwood PJ, Hatch-Pigott V, Custer LJ, Mordan LJ. Gamma-tocopherol detoxification of nitrogen dioxide: superiority to alpha-tocopherol. Proc Natl Acad Sci U S A. 1993;90(5):1771–5.
- 123. Saldeen K, Saldeen T. Importance of tocopherols beyond alpha-tocopherol: evidence from animal and human studies. Nutr Res. 2005;25:877–89.
- 124. Sylvester PW, Akl MR, Malaviya A, Parajuli P, Ananthula S, Tiwari RV, et al. Potential role of tocotrienols in the treatment and prevention of breast cancer. Biofactors. 2013;40(1):49–58.
- 125. Food and Nutrition Board, Institute of Medicine. Dietary reference intakes for vitamin C, vitamin E, selenium, and carotenoids. Washington, DC: National Academy Press; 2000.
- 126. King JC, Blumberg J, Ingwersen L, Jenab M, Tucker KL. Tree nuts and peanuts as components of a healthy diet. J Nutr. 2008;138(9):1736S–40.
- 127. Gao X, Wilde PE, Lichtenstein AH, Bermudez OI, Tucker KL. The maximal amount of dietary alpha-tocopherol intake in U.S. adults (NHANES 2001–2002). J Nutr. 2006;136(4):1021–6.
- 128. Turnlund J, Costa F, Margen S. Zinc, copper, and iron balance in elderly men. Am J Clin Nutr. 1981;34(12): 2641–7.
- 129. Calvo MS, Uribarri J. Public health impact of dietary phosphorus excess on bone and cardiovascular health in the general population. Am J Clin Nutr. 2013;98(1):6–15.
- Tucker KL, Morita K, Qiao N, Hannan MT, Cupples LA, Kiel DP. Colas, but not other carbonated beverages, are associated with low bone mineral density in older women: The Framingham Osteoporosis Study. Am J Clin Nutr. 2006;84(4):936–42.

- Nordin BE, Need AG, Steurer T, Morris HA, Chatterton BE, Horowitz M. Nutrition, osteoporosis, and aging. Ann NY Acad Sci. 1998;854:336–51.
- 132. Rizzoli R, Boonen S, Brandi ML, Burlet N, Delmas P, Reginster JY. The role of calcium and vitamin D in the management of osteoporosis. Bone. 2008;42(2):246–9.
- 133. Wang L, Manson JE, Buring JE, Lee IM, Sesso HD. Dietary intake of dairy products, calcium, and vitamin D and the risk of hypertension in middle-aged and older women. Hypertension. 2008;51(4):1073–9.
- Nicklas TA, O'Neil CE, Fulgoni III VL. The role of dairy in meeting the recommendations for shortfall nutrients in the American diet. J Am Coll Nutr. 2009;28 Suppl 1:73S–81.
- 135. Bischoff-Ferrari HA, Kiel DP, Dawson-Hughes B, Orav JE, Li R, Spiegelman D, et al. Dietary calcium and serum 25-hydroxyvitamin D status in relation to BMD among U.S. adults. J Bone Miner Res. 2009;24(5):935–42.
- Michaelsson K, Melhus H, Warensjo Lemming E, Wolk A, Byberg L. Long term calcium intake and rates of all cause and cardiovascular mortality: community based prospective longitudinal cohort study. BMJ. 2013;346:f228.
- 137. Heaney RP. Calcium supplementation and incident kidney stone risk: a systematic review. J Am Coll Nutr. 2008;27(5):519–27.
- 138. Ma RW, Chapman K. A systematic review of the effect of diet in prostate cancer prevention and treatment. J Hum Nutr Diet. 2009;22(3):187–99. quiz 200–2.
- 139. Wolf FI, Cittadini A. Chemistry and biochemistry of magnesium. Mol Aspects Med. 2003;24(1-3):3-9.
- 140. Musso CG. Magnesium metabolism in health and disease. Int Urol Nephrol. 2009;41(2):357-62.
- 141. Rosanoff A, Weaver CM, Rude RK. Suboptimal magnesium status in the United States: are the health consequences underestimated? Nutr Rev. 2012;70(3):153–64.
- 142. Ford ES, Mokdad AH. Dietary magnesium intake in a national sample of US adults. J Nutr. 2003;133(9): 2879–82.
- 143. Marier JR. Magnesium content of the food supply in the modern-day world. Magnesium. 1986;5(1):1–8.
- 144. Kjeldsen K. Hypokalemia and sudden cardiac death. Exp Clin Cardiol. 2010;15(4):e96–9.
- 145. He FJ, MacGregor GA. Beneficial effects of potassium on human health. Physiol Plant. 2008;133(4):725–35.
- 146. Ervin RB. Healthy Eating Index scores among adults, 60 years of age and over, by sociodemographic and health characteristics: United States, 1999–2002. Adv Data. 2008;20(395):1–16.

Chapter 22 Physical Activity and Exercise: Important Complements to Nutrition in Older Adults

Anne O. Brady, Alison Clune Berg, Mary Ann Johnson, and Ellen M. Evans

Key Points

- Physical activity (PA) is defined as any bodily movement that results in an increase in energy expenditure.
- Exercise (EX) is defined as a distinct subset of PA that is planned, purposeful, and intends to meet a specific health goal.
- Physical activity is essential for the prevention and management of nearly all chronic diseases and conditions, including cardiovascular diseases, cancer, osteoporosis, age-related loss of physical function, and frailty.
- Public health guidelines for physical activity include aerobic exercise, resistance training, and flexibility exercises across the lifespan, with the addition of balance exercise for older adults to decrease the risk of falls.
- Evidence-based programs specifically designed to improve both physical activity and dietary habits of older adults across a range of function are available and can be implemented in community and clinical settings, serving as a resource for clinician referrals.
- Partnerships of clinicians with community-based providers will improve access to both physical activity and nutrition programs, ultimately promoting independence and quality of life for older adults.

Keywords Physical activity • Exercise • Fitness • Cardiovascular disease • Cancer • Osteoporosis • Physical function • Frailty • Functional capacity • Evidence-based programs

A.O. Brady, B.S., M.S., Ph.D. Department of Kinesiology, University of North Carolina at Greensboro, HHP Building, Greensboro, NC 27402-6170, USA e-mail: aobrady@uncg.edu

A.C. Berg, B.S.F.C.S., M.S. Department of Foods and Nutrition, University of Georgia, 140 Barrow Hall, 115 DW Brooks Dr., Athens, GA 30602, USA e-mail: alisoncberg@gmail.com; aclune@uga.edu

M.A. Johnson, Ph.D. (⊠) Department of Foods and Nutrition, University of Georgia, Dawson Hall, Athens, GA 30602, USA e-mail: mjohnson@fcs.uga.edu; drmaryannjohnson@gmail.com

E.M. Evans, Ph.D. Department of Kinesiology, University of Georgia, 101A Ramsey Center, 330 River Road, Athens, GA 30602-6554, USA e-mail: emevans@uga.edu

Introduction

Definition of Exercise and Physical Activity

The scientific literature clearly documents that physical movement is beneficial for physical and mental health across the lifespan, especially in older adults. Public health guidelines for physical activity for all Americans have been developed [1], with a specific position stand for older adults [2]. Research is still advancing regarding how much movement (quantity) is needed and what mode(s) of movement (quality) are most effective to obtain health benefits. The ultimate answers to these questions will depend on the individual and their specific health status and the health goal to be obtained. For example, a contemporary topic of research is the influence of sedentary behavior (e.g., sitting time) on health status [3]. Importantly, it appears that leisure-time physical activity does not completely offset the adverse health effects of being completely sedentary [4]. This movement pattern would describe individuals who exercise 30–60 min per day but are sedentary the remaining 23–23.5 h of their day.

To appreciate the topics presented in this chapter, the reader will benefit from understanding the working definitions of physical activity, exercise, and physical fitness that will be utilized [1, 5]. Physical activity (PA) is defined as any bodily movement that results in an increase in energy expenditure. Alternatively, PA includes any activity that is not sedentary or that does not involve quiet sitting or lying down. PA can include leisure time PA, spanning low-intensity activities such as fishing to high-intensity activities such as playing sports and occupational or job-related PA. Exercise (EX) is defined as a distinct subset of PA that is planned, purposeful, and intends to meet a specific health goal. EX activities, especially in older adults, are conventionally completed to enhance or maintain cardiorespiratory fitness, muscular strength, and endurance, flexibility and/ or balance. However, many individuals also utilize habitual EX to enhance their psychosocial wellbeing such as for management of anxiety or mild depression, and to improve their sleep quality. For the purposes of this chapter, the term "PA/EX" will be used, and specific descriptions will be given when needed.

Many definitions of physical fitness exist typically clustering into health-related physical fitness or performance-related physical fitness [5]. Health-related physical fitness is conventionally defined as morphological (e.g., body composition, bone density), muscular, motor (e.g., balance), cardiorespiratory, and metabolic (e.g., glucose tolerance, lipid metabolism). Performance-related physical fitness is related to health-related physical fitness but applies these attributes to perform a task inclusive of competitive sports, military maneuvers, or occupational work. As older adults are primarily concerned with physical function we have chosen to define physical fitness according to the World Health Organization as "the ability to perform muscular work satisfactorily," which recognizes that many aspects of health-related physical fitness influence performance-related physical fitness in daily life. It should be recognized that these terms, although having distinctly different definitions, are often used interchangeably in the literature largely due to the fact that they are so interrelated.

Synergistic Effect of Diet and Exercise

The potential additive or synergistic effects of energy and nutrient intake with PA/EX on numerous health outcomes important for older adults have been well documented. For example, PA/EX combined with energy restriction during intentional weight loss programs promotes desirable changes in

body composition, such as reductions in fat mass, while attenuating the loss of lean mass and bone mineral mass [6]. Also, during intentional weight loss, the interactive effects of EX combined with a higher-protein-energy-restricted diet attenuates loss of lean mass and enhances loss of fat mass compared to a higher-carbohydrate-energy-restricted diet [7-9]. A classic example of the interactive effects of PA/EX with a micronutrient is calcium and vitamin D and the role of PA/EX for optimal bone mineral status [10]. Indeed, when designing a research study or clinical trial exploring bone mineral responses to either novel forms of loading (i.e., EX) or calcium/vitamin D supplementation it is necessary to carefully account for the other factor due to the established independent effects of each factor on bone status [1, 10]. Although the additive and synergistic effects of energy and nutrient intake and PA/EX on the health status of older adults have been documented, additional research is needed. As contemporary scholars and interdisciplinary teams bridge the gap between nutrition and exercise, it is anticipated that more information regarding multimodal interventions will become available. Additionally, as dietary intake and PA/EX practices are essentially behavioral choices, partnerships with behavioral scientists will exponentially advance our understanding of best practice, with the ultimate goal of enhancing the health of older adults through effective and sustainable interventions and programs.

Prevention Versus Management of Disease

PA/EX is not only instrumental in *preventing* chronic diseases and conditions, it is also a powerful strategy for *management* of chronic diseases and conditions. In the US, more than 60 % of adults over the age of 65 have at least two chronic conditions [11] and approximately 50 % of older women aged 70–80 years cannot complete general mobility tasks such as climbing a flight of stairs or completing housework [12]. This low functional status along with the high prevalence of chronic disease indicates that PA/EX may be particularly beneficial for older adults [2].

Overview and Call to Action

This chapter will summarize the literature for PA/EX for older adults regarding (1) the benefits for the main metabolic diseases, including obesity, cardiovascular diseases, diabetes mellitus, and cancer; the primary mental and psychosocial challenges, including cognitive decline and depression; and musculoskeletal challenges, including osteoporosis and physical function; (2) compliance with recommendations; (3) the current public health guidelines; (4) barriers and challenges; and (5) evidence-based programs. The goal of this chapter is to encourage the clinician to embrace the importance of PA/EX as a complement to nutrition to positively influence the health status of older adults. Moreover, if the expert in nutrition is not confident in their PA/EX knowledge, the content in this chapter might facilitate collaborations with PA/EX scholars and other professionals working in community and clinical settings.

Benefits of Physical Activity and Exercise

Engagement in PA/EX positively influences numerous chronic diseases and conditions, health-related physical fitness, and physical function to ultimately enhance quality of life for older adults. Ultimately most chronic diseases and conditions negatively influence physical functional ability, threaten independence in older adults, and impair the quality of life. Moreover, just as chronic diseases and

conditions cluster in prevalence, PA/EX has a beneficial influence on several diseases and conditions that are interrelated. For example, PA/EX expends energy (*obesity*) and simultaneously improves insulin sensitivity (*type 2 diabetes mellitus*) and the lipid profile (*cardiovascular diseases*), increases muscular strength (*physical function*), benefits psychological well-being (*depression*), and ultimately improves *quality of life*. The following section will provide a brief overview of the primary diseases and conditions that are positively influenced by habitual PA/EX in older adults.

Obesity

The relative importance of energy intake versus energy expenditure for risk of obesity remains incompletely characterized. This question will undoubtedly remain unanswered for some time due to (1) the inability to accurately measure either component of the energy balance equation over long periods of time in free-living individuals, and (2) the paradigm that "not one energy balance equation fits all" in that any given energy balance equation is highly individual, although patterns of imbalance exist for given cohorts including older adults. Additionally, the independent effects of PA/EX on health outcomes independent of nutritional factors and adiposity are an active area of interest. In other words, to what degree can an older adult maintain good health when they are "fit" but "fat" and/or have poor diet quality? Obesity is prevalent across the lifespan, including among older adults. In the US, about 35 % of adults aged 65 and over were obese in 2007–2010, which represents over eight million adults aged 65–74 years and about five million adults aged 75 and older [13]. The prevalence was similar between men and women. The forecasted increase in older adults combined with the increase in obesity in younger populations is predicted to have major impacts on our health care system. Notably, obesity is cited as a primary factor in loss of independence and nursing home admissions [14].

The public health goal for weight management is to manage the "energy gap," to control or reduce fat mass, and to maintain or enhance lean and bone mass, the latter two being very important for older adults. Current recommendations [15] indicate that a combination of restriction of energy intake with an increase in energy expenditure is of primary importance for all aspects of the weight management spectrum including weight gain prevention, weight loss, and weight loss maintenance. PA/EX is uniquely positioned for this multifaceted goal as habitual PA/EX both expends energy and assists in the preservation of lean and bone mass. PA/EX contributes to energy expenditure via several mechanisms [16]. First, the direct increased energy demand for muscle force production for movement is well recognized. Aerobic activities involving large muscle groups, often performed rhythmically, are best suited for energy expenditure with higher intensities (e.g., walking faster) and longer durations (e.g., minutes walked) producing greater energy expenditures, other factors being equal. Walking, cycling, and swimming are typical aerobic activities performed by older adults. Second, if the activity was of sufficient intensity (e.g., running) and duration (typically greater than 30 min) then energy expenditure may be influenced through excess post oxygen consumption theorized to be due to energy deficits occurring at the beginning of the exercise bout and perturbations in thermoregulatory and endocrine systems collectively resulting in an elevated metabolic rate after the activity has ceased. Third, habitual higher levels of PA/EX, including both aerobic and resistance training, potentially contribute to a higher resting energy expenditure (REE) and may attenuate the natural decline in REE that occurs with aging. The influence of PA/EX on REE is mediated primarily through lean mass, which is known to explain approximately 80 % of REE. As REE is the greatest contributor to total daily energy expenditure in most adults, especially older adults, comprising about 60-70 % of total daily energy expenditure, the preservation of REE is essential to energy balance [16].

Although long-term weight loss is achieved by both moderating energy intake and participating in habitual physical activity, the question always is how much PA/EX is enough? Current guidelines indicate that 60 min/day of moderate intensity activity is the target level to prevent weight gain,

whereas 60–90 min/day might be a more important goal to prevent weight regain following significant weight loss [17]. Resistance training does not enhance weight loss, but may increase fat-free mass and decrease fat mass and is associated with reductions in health risk and preservation of functional independence. The long-term implication of resistance exercise for weight management, especially in older adults, is not well studied. Finally, the importance of PA/EX for weight management is essential for older adults. As detailed in Chap. 10, the combined modes of endurance and strength training under conditions of dietary energy restriction are critical for the preservation of lean and bone mass in older adults undergoing weight loss [6, 18]. Due to the high risk of osteoporosis and sarcopenia in older adults, maintaining bone and lean mass with weight loss is of primary importance.

Cardiovascular Disease

It is well established that habitual PA/EX reduces the risk of cardiovascular diseases, our major public health challenge. As age is an independent risk factor for both of the primary cardiovascular diseases, coronary heart disease (CHD) and stroke [19], prevention and treatment of CHD and stroke are essential for older adults. Independent of other risk factors, PA reduces risk for CHD [19, 20], and PA/EX is a primary component of cardiac rehabilitation. PA/EX alters several modifiable risk factors for CHD, including high blood cholesterol and triglycerides, high blood pressure, diabetes mellitus, stress, and obesity. Regular participation in moderate-to-vigorous leisure time PA reduces risk for ischemic and hemorrhagic stroke by about 30 % in older adults [1]. Similar to CHD, PA/EX is a primary component of the rehabilitation from a cerebrovascular event, although the health goal differs in that physical rehabilitation is primarily used to recover physical function [21].

The mechanisms responsible for the independent effects of PA/EX on CHD are thought to be related to myocardial oxygen supply and demand mediated in large part by direct effects on the vascular wall with enhanced endothelial function and reductions in hemostatic and inflammatory biomarkers [22]. These beneficial effects are in addition to the established reduction in modifiable risk factors for CHD (e.g., lipids, blood pressure). First, regular EX lowers the oxygen demands of the heart during a given physical task of a fixed demand (i.e., intensity), and thus reduces risk of ischemia during physical work or other stress. With a chronic overload of exercise stress, cardiorespiratory fitness increases from both central and peripheral mechanisms, effects that have been demonstrated in individuals of all ages, including older adults. In addition to peripheral adaptations that occur in the trained muscle that enhances oxygen uptake and utilization, increases in plasma volume and remodeling of the left ventricle combine to enhance stroke volume, leading to a reduction in heart rate for a given bout of EX, thereby decreasing work demand on the heart. Second, risk of coronary thrombosis is also reduced by this increased plasma volume, possibly through a series of events including reduced blood viscosity, which may decrease platelet aggregation and clotting, the latter known to cause most myocardial events. Third, beneficial changes in endothelial function in response to exercise also occur. Normal endothelial function allows the appropriate vasodilation of arteries, most notably the coronary arteries, in response to increased metabolic demand. The nitric oxide system is critical in the vasodilatory response to exercise allowing the atherogenic effects of inflammation to be attenuated. Exercise stimulates nitric oxide synthase, improving endothelial function. Finally, inflammation is associated with increased blood clotting and endothelial dysfunction. The role of chronic inflammation in several age-associated chronic diseases, especially CHD, is of high contemporary research and clinical interest in older adults [23]. Importantly, PA/EX, in addition to optimal nutrition, is a key for the management of chronic inflammation [23].

Hypertension, along with being considered a primary risk factor for stroke and CHD, is also considered a widespread cardiovascular disease with approximately 70 % of individuals 65 years and older having hypertension in the US [24]. The positive effects of PA/EX and cardiorespiratory fitness on blood pressure are well documented [1]. The magnitude of the effect of habitual PA/EX on hypertension is on the order of about 7 mmHg and 6 mmHg for systolic and diastolic blood pressure, respectively, after controlling for other risk factors [25]. The mechanisms responsible for the anti-hypertensive effects of PA/EX are likely multifactorial, including the anti-inflammatory and endothelial effects described previously, along with potential beneficial changes in the sympathetic nervous system and renin-angiotensin-aldosterone system. Exercise guidelines to prevent CHD, stroke, and hypertension are all geared toward enhancing both leisure time PA and cardiorespiratory fitness. The latter requires a higher training intensity typical of EX (60–75 % compared to 40–60 % maximal capacity for leisure time PA) and continuous EX (about 30 min). Resistance training also reduces blood pressure in both normotensive and hypertensive adults [25].

Metabolic Syndrome (MetS), Type 2 Diabetes Mellitus (T2DM), and Dyslipidemia

Metabolic syndrome is a cluster of pathophysiological conditions related to insulin resistance and encompasses abdominal obesity, hypertriglyceridemia, low high-density lipoprotein cholesterol, hyperglycemia, and hypertension [26]. MetS is a major precursor to both CHD and T2DM. Although the classification of risk factors for MetS continues to be debated, what is well established is that habitual PA/EX and high cardiorespiratory fitness are protective against MetS, T2DM, and dyslipid-emia. Older adults are prime candidates for these health challenges, as over 50 % of men and women over 60 years of age have MetS [26]. Although these diseases/conditions are also influenced by weight status and dietary intake, PA/EX is known to reduce risks for MetS and T2DM independent of the effects of adiposity or dietary habits [27].

Obesity, MetS, and T2DM share the same underlying pathology described above for CHD of chronic low-level inflammation and endothelial dysfunction, along with insulin resistance. Individuals with some combination of MetS and T2DM have altered insulin function with a classic resistance pattern resulting in hyperinsulinemia and potentially hyperglycemia (classic diagnostic criteria for T2DM). Increased energy metabolism within skeletal muscle increases the demand for glucose (and other energy sources), stimulating glucose uptake even in the absence of insulin. With habitual PA/EX additional mechanisms contribute to this enhanced glucose disposal including adaptation of metabolic enzymes and mitochondrial density within skeletal muscle. Thus, PA/EX increases glucose uptake and insulin sensitivity at rest and during PA/EX [28]. Additional changes occur to enhance fat metabolism, which can benefit the serum lipid profile. LDL and cholesterol are less responsive to PA/EX compared to dietary manipulations; however, HDL cholesterol and triglycerides are positively influenced by PA/EX. Note that the two latter lipoproteins are associated with MetS. PA/EX also influences the postprandial lipemia (PPL) response, which is the surge in lipoproteins that is present after a meal that is dependent on the fat ingested within the meal. PPL is linked to the incidence of myocardial infarction and stroke through its influence on coagulation ability. It is well established that acute and chronic PA/EX decreases the length and amount of the PPL response from a meal [28].

The recommendations for EX/PA to reduce or treat MetS and T2DM are similar to that recommended for CHD in that the activity should be aerobic in nature involving large muscle group activities (e.g., walking, cycling, swimming) with an intensity being moderate to vigorous for 60 min per day, which may be divided into shorter sessions but not less than 15 min per bout and one session per week should be greater than 30 min. As the cellular adaptations that occur to enhance insulin sensitivity are transient, it is highly recommended that some PA/EX be performed daily. In addition, the energy expenditure in PA/EX should be greater than 2,000 kcal/week. Resistance exercise can be used as a supplement to the aerobic exercise. Importantly, higher intensity and longer duration EX is needed for beneficial changes in HDL cholesterol to be realized [27, 28].

Cancer

Although there is a growing interest in obesity and the link to many types of cancers, the independent benefits of PA/EX on cancer are less well-established than cardiovascular and metabolic diseases. Almost half of men and one-third of women in the US will develop some form of cancer during their lifetimes [29]. Importantly, although all cancers involve the malfunction of genes that control cell growth and replication, only about 5 % of cancer is strongly hereditary, which means that primary and secondary prevention of cancer is a high public health priority [29]. As age is an independent risk factor for cancers of all types, older adults are concerned with prevention, as well as with successful cancer-related treatment and survivorship issues. It is well established that PA/EX can positively influence cancer-related treatment and survivorship concerns in many ways including fatigue, depression, and quality of life [30]. The strongest and most consistent evidence for the beneficial effects of PA/EX relate to the prevention of cancer of the colon and breast. As evident from the protective mechanisms described below, the optimal PA/EX guidelines for cancer prevention relate to the prevention of obesity and insulin resistance as described previously.

Although the exact mechanisms whereby PA/EX prevents colon cancer have not been fully elucidated, there are several potential theories. First, the initial theory is that PA/EX results in a shortened gastrointestinal transit time, which results in less exposure of the mucosal lining of the colon to potential carcinogens, although more recent studies do not embrace this theory. Second, insulin resistance may be an important link of PA/EX with colon cancer. The protective effects of PA/EX occur over a wide range of weight statuses. The hyperinsulinemia of insulin resistance serves as an important growth factor for colon cancer cells [31]. Third, chronic low-grade inflammation may play an important role in the link between PA/EX and colon cancer. Finally, limited evidence suggests that positive changes in immune function (acquired, innate, or both) and reduction of prostaglandin levels may also contribute to the protective effects of PA/EX on colon cancer [1].

Similar to colon cancer, the exact mechanisms whereby PA/EX reduces risk for breast cancer are not known. In addition to the beneficial effects on the immune system, several theories also suggest that the beneficial effects of PA/EX on breast cancer risk are through reductions in insulin, insulin-like growth factor, and inflammation, which similar to colon cancer, are known to promote tumor cell growth [32]. As cumulative lifetime exposure to circulating ovarian hormones, especially estrogens, is a well-established risk factor for breast cancer, the influence of PA/EX on sex hormones is also an important theory. For example, strenuous exercise during adolescence can delay menarche and is also associated with longer menstrual cycles due to a longer follicular phase during the reproductive years, which may in part by driven by the reductions in adiposity that occur with physical training. PA/EX may also influence menopausal status as active women are leaner and obesity is associated with a later age at menopause thereby increasing estrogen exposure. Thus, it is difficult to ascertain if the beneficial effects of PA/EX on risk for breast cancer are due to a direct effect on the activity of ovulatory cycles or an indirect effect through diet, body composition, and caloric expenditure or some combination thereof.

Osteoporosis and Physical Functional Capacity

More so than cardiovascular or metabolic diseases, osteoporosis, frailty-related bone fractures, and loss of physical functional capacity almost exclusively occur in older adults. Of significant concern are falls [33], which can lead to traumatic brain injury, hip fracture, and other fractures. Such severe injuries in older adults can rapidly progress to loss of independence or death. It is estimated that 44 million Americans ages 50 or older, 80 % of which are women, or 55 % of our adult population, have

either osteoporosis or osteopenia [34]. Physical limitations and risk for disability are influenced by many factors, including comorbidities (e.g., cancer, cardiovascular diseases, obesity, osteoporosis), psychosocial factors (e.g., depression, self-confidence, motivation), and sensory impairments (e.g., vision), as well as muscular fitness. According to the CDC, physical limitations can be defined as difficulty performing one of any of the following activities: walking a quarter mile; walking up ten steps without resting; standing or being on your feet for about 2 h; sitting for about 2 h; stooping, bending, or kneeling; reaching up over your head; using your fingers to grasp or handle small objects; or lifting or carrying something as heavy as 10 lb [12]. Current estimates indicate that 23 % of individuals 60–69 years of age report one or more physical limitations and that the presence of physical limitations increases linearly with age [12].

Functional capacity and osteoporosis are strongly related to each other. Bone and physical functional capacity encompass the morphological, muscle, and motor components of health-related fitness and ultimately determine performance-related fitness, which in the case of an older adult is the "performance" of activities of daily living. Risk for fracture can be conceptualized as a combination of bone-dependent and bone-independent risk factors. Undeniably, proper nutrition influences both bone status and physical function, with a large literature documenting the benefits of calcium and vitamin D on bone health, as well as protein intake on muscle mass and quality. Here we highlight the most salient mechanisms whereby PA/EX positively influence bone status, muscle capacity, and physical function.

The skeleton adapts to loading, with the two primary means to load including gravity or ground reaction forces and muscle and tendon interactions or joint reaction forces, which occur during muscular contraction, ambulation, and the maintenance of posture. Although genetics largely determine bone geometric properties, environmental factors, mainly diet, hormones, and PA/EX, greatly influence bone mass, bone mineral density (BMD), and bone architecture (i.e., trabeculae) through their influence on bone resorption and formation. The two primary bone-dependent strategies to reduce risk of bone fracture are to (1) maximize peak bone mineral mass and BMD during childhood and adulthood, and (2) minimize the reduction in bone mass and BMD that occurs after the age of about 40 years [35]. The positive benefits of influence of PA/EX on bone mass and BMD across the lifespan are well documented, even in older adult populations [35]. Although not fully understood, bone adaptations are largely thought to align with Wolff's law that posits that mechanical loading impacts the microstructural architecture remodeling [36]. The biomechanical mechanism whereby mechanical strain translates into an increase in BMD is not completely characterized; however, it is thought to include piezoelectric potentials, increased blood flow, and hormonal response involving prostaglandins, nitric oxide, and growth factors [36]. Importantly, PA/EX effects on bone are maximized in the presence of optimal nutrition and the hormone milieu.

Another important component of bone fracture, especially for hip fracture, is the risk of falling, which is related to physical functional capacity. Although the aging process is known to influence muscle capacity (see Chap. 7), including muscle strength, endurance, and power (the ability to move force quickly), PA/EX can greatly attenuate the decline in muscle performance. Physical inactivity is directly linked to reduced muscle mass and quality and reductions in physical functional ability; habitual PA has been consistently associated with improvements in physical function [1, 2]. The benefits of physical activity include a prolonged time until mobility disability and loss of independence occurs in the lifespan, as well as clinically meaningful improvements in physical function testing scores in the research laboratory. Mechanisms responsible for increased muscle capacity in response to PA/EX include improved neural activation and hypertrophy of muscle fibers although hypertrophy responses are reduced compared to younger individuals. Importantly, both functional training (e.g., activities of daily living such as chair rises) and resistance training are effective training strategies in older adults, but adaptations are task-specific.

The optimal PA/EX regimen to enhance bone status involves dynamic, high peak loads that are frequent, intermittent, and novel. However, for the typical older adult, this type of loading could cause injury to osteoporotic bones, joints and/or muscles. Therefore, the safest overload strategy for the older adult is through joint reaction forces obtained from strength and/or resistance training. An EX program that is multimodal and inclusive of both resistance exercise and functional training will provide muscle capacity benefits to attenuate declines in functional capacity. High levels of PA along with cardiorespiratory and/or aerobic EX are also important to prevent other comorbidities (e.g., obesity, CHD), which also indirectly contribute to declines in physical function.

Cognition and Depression

Although the positive effects of PA/EX on the physical health status of older adults are well documented and appreciated, a physically active lifestyle also confers benefits to mental health and psychological well-being. The most salient of these conditions for the older adult are cognition and depression. Importantly, although PA/EX has been shown to positively influence cognition and depression, impairments in cognition and depression can also reduce the likelihood that an older adult will participate in PA/EX. Thus, it should be appreciated that PA/EX has multifaceted direct and indirect benefits for mental and psychological health, which are briefly highlighted below.

Literature summarizing prospective studies supports that PA delays the incidence of dementia and the onset of cognitive decline associated with aging [1]. Additional studies documenting the benefits of regular exercise and cardiorespiratory fitness on cognitive functioning suggest that the greatest effects are on executive control tasks related to goal-oriented decision-making behavior [37]. Mechanisms through which PA/EX improves cognition may include plasticity and survival of brain neurons, increased brain blood flow, and enhanced neural circuits. The most studied type of EX is aerobic cardiorespiratory in nature, which strongly suggests that these benefits are realized when changes in cardiorespiratory fitness occur.

The prevalence of depression is high among older adults and treatment is often inadequate. A rich literature base of various research designs documents that higher levels of PA are associated with lower odds of depressive symptoms in all ages, including older adults [38, 39]. Mechanisms to explain the effect of PA/EX on depressive symptoms include cognitive-related social and/or biological factors [40]. The primary neurobiological mechanisms speculated to be associated PA/EX-related reductions in depression include (1) endorphin hypothesis, (2) brain blood flow hypothesis, (3) hypothalamic-pituitary-adrenocortical (HPA) axis that regulates the body's response to stress but has also been associated with depression, (3) improved sleep, and (4) monoamine hypothesis, which focus on the neurotransmitters dopamine, norepinephrine, and serotonin. Importantly, moderate and high levels of PA reduce the odds of depression similarly. An increase in fitness is not necessary to provide the effect. Aerobic and resistance exercise training provide similar benefits for individuals diagnosed with mild to moderate depression [40].

Mortality and Quality of Life

A large body of observational studies strongly support that those with higher levels of PA/EX and higher levels of fitness have a lower incidence of premature mortality than their inactive or unfit counterparts [41]. The strength of this association is consistent among men and women and among different racial-ethnic groups, although the data on the latter are limited. As greater than 50 % of deaths

occur due to CHD, stroke, and cancer, the biological plausibility is well established. It is plausible that higher PA/EX enhances quality of life in older adults through reduction in chronic conditions as described previously, and attenuates physical functional decline thereby preserving independent living. Multimorbidity, defined as the coexistence of two or more chronic conditions, is associated with disability, poor functional status, and poor quality of life [11]. As mortality rates have declined and the population has aged, multimorbidity will undoubtedly increase in the future. Habitual PA/EX are critical throughout the lifespan for optimal physical and mental health but more so for older adults to prevent or manage chronic conditions and preserve their physical functional ability. Importantly, it is never too late to begin a PA/EX program. Position statements [1, 2] regarding PA/EX have been built on the extensive literature that documents the adaptability of the human body to PA/EX with much of the research being performed in individuals in the eighth and ninth decades of life [2].

Prevalence of Physical Activity in Older Adults

A comprehensive exercise program incorporates multiple components, including aerobic exercise, resistance training, flexibility, and balance exercises. According to the 2011 Behavioral Risk Factor Surveillance System (BRFSS), 48.9 % of older adults in the United States did not engage in the recommended amounts of aerobic exercise [42]. A larger percentage (78.1 %) did not engage in the recommended amounts of resistance training exercise, and nearly 84 % of older adults in the United States did not engage in the recommended amounts of aerobic admounts of aerobic and resistance training activity. In spite of these statistics, data from BRFSS indicate a positive shift in the number of older adults engaging in regular physical activity. In 1996, only 19.9 % of older adults obtained 30 min of physical activity five or more days per week. By 2005, this percentage doubled, such that nearly 40 % of older adults reported at least 30 min of moderate intensity physical activity five or more days per week or 20 min of vigorous activity three or more days per week [42]. Even though a well-balanced exercise program includes multiple components, there is little data regarding the percentage of older adults engaging in adequate amounts of flexibility and balance training.

Public Health Guidelines for Physical Activity in Older Adults

2008 Physical Activity Guidelines for Americans

One of the most widely recognized and recommended physical activity guidelines with recommendations specific to older adults is from the US Department of Health and Human Services (DHHS) [1]. The 2008 Physical Activity Guidelines for Americans focus on aerobic and muscle-strengthening activities across the lifespan. The DHHS PA guidelines for older adults are the same as those recommended for the general adult population, but additionally include balance training for those older adults at risk of falling. Furthermore, DHHS states that older adults with chronic conditions be as active as those conditions allow. Each type of physical activity (aerobic, muscle-strengthening, and balance) provides individuals with significant health benefits and whenever possible should all be included in a comprehensive PA/EX program.

A key recommendation from DHHS for older adults is to avoid inactivity. Any activity is better than none. The accumulation of chronic conditions may prevent an older adult from reaching the recommendations, yet an individual should be as active as their conditions allow. Specifically, in order to obtain substantial health benefits, DHHS recommends older adults acquire at least 150 min of moderate-intensity aerobic physical activity per week, or 75 min of vigorous-intensity aerobic physical activity per week, or an equivalent combination of moderate and vigorous-intensity aerobic physical activity [1]. On a scale of 0–10, with a 0 equivalent to sitting in a chair and a 10 representing a maximal effort, moderate intensity activity would be a 5 or 6. Moderate intensity activities should increase breathing and heart rate. Using the same scale, a vigorous intensity activity would be a 7 or 8 and cause large increases in both breathing and heart rate. The total amount of aerobic physical activity can be accumulated in 10 min bouts, spread throughout the week. Additional health benefits can be gained by exceeding these recommendations. The most commonly recommended aerobic activity is walking as it requires little equipment and is a common and familiar activity for most older adults. Other forms of aerobic exercise include dancing, swimming, water aerobics, bicycle riding, gardening activities (raking leaves, pushing a lawn mower), and aerobic dance classes.

Muscle-strengthening activities are also a key component to a comprehensive physical activity program. The DHHS recommends that older adults engage in muscle-strengthening activities of moderate to high intensity on at least 2 days per week. These activities should involve all major muscle groups of the body, including the muscle of the legs, hips, back, chest, abdomen, shoulders, and arms. Bands, hand weights, or resistance training machines are common examples of equipment used for muscle-strengthening. Calisthenic exercises such as wall push-ups, climbing stairs, or chair squats utilize an individual's own body weight and provide an easy way to engage in muscle-strengthening activities with limited equipment. Other alternatives include carrying groceries, digging or lifting when gardening, and some yoga and Tai chi exercises. One set of 8–12 repetitions for each of the major muscle groups is effective, but two to three sets may be more beneficial to build and maintain muscular strength.

Balance is an important component in an exercise program for older adults, because it helps reduce the risk of falling. The DHHS recommends that older adults who are at increased risk of falls or have fallen in the past, engage in balance activities. Older adults should engage in balance training three or more days per week. Examples of balance exercises include walking backwards or sideways, heel or toe walking, and standing from a sitting position. Balance exercises can be made more challenging by progressing from using a stable support while completing the activities to completing them without any support. Less is known about the type, intensity, and frequency of balance activities that are necessary to achieve health benefits.

Warm-up, cool-down, and flexibility exercises should also be included in a physical activity program for older adults. A warm-up allows the heart rate and breathing to increase gradually before beginning aerobic exercise and a cool-down allows the heart rate and breathing to decrease gradually after ending aerobic exercise. Though flexibility alone has no known benefits for prevention of the major chronic diseases (e.g., heart disease), maintaining adequate flexibility is essential to the performance of activities of daily living. Last, safety is a crucial element in an exercise program for older adults. Depending on an individual's health status, it may be recommended that exercise occur under supervision of qualified professionals.

ACSM Position Stand: Exercise and Physical Activity for Older Adults

In 2009, the American College of Sports Medicine (ACSM) published a Position Stand entitled Exercise and Physical Activity for Older Adults [2]. The purpose was to provide a summary of concerns critical to understanding the importance of physical activity in older adults, including functional changes associated with normal aging, the interaction of physical activity and aging, and the short-term and long-term benefits of physical activity on health and physical function. The document provides an in-depth review of the literature regarding these topic areas and may be a helpful resource for

practitioners seeking evidence-based research studies. The strength of evidence for each physical activity recommendation is also provided, ranging from A (overwhelming evidence from randomized controlled trials and/or observational studies, which provides a consistent pattern of findings on the basis of substantial data) to D (panel consensus judgment that the strength of evidence is insufficient to place it in categories A through C).

In general, the ACSM Position Stand recommendations for physical activity for older adults are consistent with the 2008 DHHS Physical Activity Guidelines for Americans. ACSM also recommends 150 min per week of aerobic activity for health benefits, but provides daily recommendations of at least 30-60 min of moderate-intensity aerobic exercise, 20-30 min of vigorous-intensity aerobic exercise, or an equivalent combination of moderate and vigorous activity, which can be accumulated in 10-min bouts [1, 2]. Despite some discrepancies in terminology, the recommendations for resistance training/muscle-strengthening activities are consistent across the DHHS and ACSM recommendations. The Position Stand provides more guidance regarding flexibility and balance training. According to ACSM position stand, flexibility training is recommended at least 2 days per week at moderate intensity (5–6 out of 10 on an intensity scale), including any type that maintains or increases flexibility using sustained stretches rather than ballistic movements, such as bouncing, which may cause injury. Balance training is recommended for individuals with a history of previous falls or mobility limitations. A lack of studies prevent specific recommendations regarding frequency, intensity, or type of balance exercises, but ACSM encourages progressively difficult postures that gradually reduce the base of support (e.g., two-legged stand, semitandem stand, tandem stand, one-legged stance), dynamic movements (e.g., tandem walk, circle turns), stressing the postural muscle groups (e.g., heel stands, toe stands), and reducing sensory input (e.g., standing with eyes closed). In general, there is strong evidence (evidence ratings of A or B) to support the benefits of aerobic exercise on exercise capacity, cardiovascular effects, body composition, metabolic effects, and bone health in previously sedentary individuals. Also, strong evidence (evidence ratings of A or B) exists to support the benefits of resistance training on muscular strength, muscle power, muscle quality, and bone health in previously sedentary individuals.

The National Institute on Aging provides free publications that can be viewed online, ordered in a hard copy form, or downloaded in the form of an audio book. For example, the handbook entitled *Exercise and Physical Activity: Your Everyday Guide* [43] describes the benefits of physical activity for older adults, provides example exercises with pictures and descriptions, and addresses some frequently asked questions regarding exercise for older adults.

Physical Activity Barriers and Challenges for Older Adults

Despite the numerous physical and psychosocial benefits of regular physical activity, the majority of older adults remain inactive. Inactivity may be due to actual and perceived barriers that are specific to an older adult population. Identifying and addressing the unique challenges faced by older adults may increase adherence to physical activity programs and allow more older adults to lead active lifestyles. In the general population, the most commonly cited reason for not engaging in a regular exercise program is a lack of time; however this does not seem to be the most common barrier among older adults. Older adults most commonly report limitations due to current health conditions and pain, environmental issues, lack of advice from physicians, and a general lack of knowledge regarding physical activity [44]. Other barriers identified by older adults include a lack of time and energy, fear of injury, low motivation, and safety issues.

Among older adults of all ages, current health status is the most common barrier to physical activity. In the US, 80 % of older adults have at least one chronic condition [45] and more than 60 % have at least two chronic conditions [11]. Furthermore, BRFSS data indicate that the percentage of adults reporting limitations in activities due to physical, mental, or emotional problems increases with age, such that current data indicate that over one-third of older adults report these limitations. The Centers for Disease Control and Prevention (CDC) recommend that older adults with chronic conditions speak with their physicians to find out if their health condition would in any way limit their ability to be active. If an older adult is considering beginning an exercise program, it is recommended that they assess their ability to safely engage in exercise. One commonly used tool is the PAR-Q, or physical activity readiness questionnaire. The PAR-Q contains seven questions, which require a "yes" or "no" response. If an older adult answers "yes" to any of the questions, it is recommended that they speak with their physician before beginning an exercise program. It is also important to bear in mind that the most common chronic health conditions in older adults (heart disease, diabetes, and arthritis) are improved by regular physical activity and may reduce an older adult's risk of developing other conditions. Even if an older adult cannot meet the minimum physical activity guidelines due to health problems, they are encouraged to be as active as possible, bearing in mind their chronic conditions and current health status.

An older adult's physical environment may also pose a challenge to engaging in a regular physical activity program. For example, an individual's built environment may discourage physical activity if there are poorly maintained or no sidewalks. Additionally, older adults who prefer walking outdoors are more active when they perceive that they are safe in their own neighborhood. Other environmental barriers to physical activity include geographical distance to a local recreational facility, gym, park, or swimming pool. Overcoming barriers in the built environment is challenging, but recommendations are being made by CDC, AARP, World Health Organization, and others to create environments that are safe and encourage physical activity in one's community by improving access to parks, public transportation, and walking paths.

Older adults commonly cite a lack of knowledge as a barrier to initiating a physical activity program. Individuals may lack understanding of the adequate amounts and intensities of physical activity that are necessary for health benefits. Additionally, many older adults may believe that their normal activities of daily living provide enough exercise for disease prevention. Older adults' physicians can play a key role in promoting exercise among this population. Despite the fact that older adults tend to visit physicians more often than the general population, the majority of physicians do not regularly counsel their patients about physical activity. In 2010, 41.9 % of individuals 65-74 years of age received exercise recommendations from their physician or another health professional, but this percentage decreased as age increased, 32.9 % of individuals 75-84 years of age and only 28.9 % of individuals 85 and older received exercise recommendations from physicians or other health professionals [46]. Understanding the pivotal role that physicians can play in overcoming barriers to physical activity for older adults, the American College of Sports Medicine and the American Medical Association recently introduced a health campaign, Exercise is Medicine [47]. This program encourages physicians to assess and review every patient's physical activity habits at each visit and provides physicians with tools and methods to counsel individual patients. The Exercise is Medicine website (http://www.exerciseismedicine.org) provides free guides for physicians and other healthcare providers that can be downloaded, printed, or emailed. For example, the Health Care Providers' Action Guide suggests that physicians begin by asking their patients if they currently exercise. If yes, physicians are encouraged to query regarding the type of activities, duration, intensity, and frequency and to record this information. If the answer was no, it is suggested that the physician ask why and determine if the patient is willing to begin an exercise program. Additionally, the guide includes a physical activity readiness questionnaire, exercise prescription and referral forms, and physical activity clearance forms.

In order to encourage older adults to initiate a physical activity program or improve adherence to a program, it is important to remove perceived and actual barriers. Multiple methods have been effective at removing or mitigating common barriers. For example, it is important to highlight the social support and positive interactions that can be gained through regular physical activity programs in

group settings among older adults. Exercising among peers creates a less stressful environment for older adults and involves a social component, which improves physical activity adherence in the long-term. Building self-esteem and self-efficacy is also important for older adults and has been shown to be related to older adults' physical activity patterns. Education and knowledge transfer about how to safely engage in exercise and the benefits of physical activity improve older adults' adherence and enjoyment in exercise. Finally, it is important to bear in mind that among some older adults, barriers may also act as motivators. For example, failing health, which can cause physical activity to be more challenging and is often viewed as a barrier, can serve as motivation for an older adult to become more active. Helping older adults identify their own distinct barriers and motivators towards exercise is a helpful tool used to create individualized programs that are both safe and effective. Practitioners should keep in the mind that the ultimate goal is to encourage older adults to remain active throughout their lifetime.

Conclusion: Evidence-Based Physical Activity Programs for Older Adults

This chapter concludes by translating our knowledge base about the benefits of physical activity by summarizing accessible, practical, and evidence-based programs for older adults. Clinicians must be equipped to provide knowledge, support, and encouragement for older adults to engage in physical activity. Providing basic education regarding the health benefits of physical activity, as well as encouragement is straightforward for most clinicians, but recommending specific modes of exercise appropriate for older adults with varying chronic conditions and experience levels is more complex. Fortunately, there are many evidenced-based physical activity programs that clinicians can recommend or utilize with confidence.

The National Council on Aging (NCOA) and the National Institute on Aging (NIA) at the National Institute of Health (NIH) support the use of evidence-based physical activity programs for older adults and provide resources for connecting with many program providers on their websites [48–50]. Title IIID of the Older Americans Act (OAA) authorizes funds from the Department of Health and Human Services (DHHS), Administration on Aging (AoA) specifically for use in implementing evidence-based health promotion and disease prevention programs (e.g., physical activity, smoking cessation, falls prevention) and to leverage other funds, particularly through partnerships, to support these programs [49]. AoA uses a tiered set of criteria (minimal, intermediate, and highest-level criteria) to define evidence-based programs suitable for consideration for OAA Title IIID funding. AoA, NCOA, and NIA promote the use of evidence-based physical activity programs because of their proven effectiveness in the target population and readiness for translation and implementation by a variety of types of organizations and clinicians [49]. These programs are particularly successful because they address many of the barriers to physical activity for older adults (e.g., safety, knowledge, accessibility, chronic disease management).

For example, *Healthy Moves for Aging Well*, a Title IIID highest-tier criteria qualified physical activity program, addresses safety concerns by providing chair exercises for frail individuals or those with functional limitations in addition to advanced level exercises [51, 52] (Table 22.1). Most of these NCOA, AoA, and NIA recommended programs involve instruction, demonstration, and supervision by a lay person or health professional, which can be reassuring for the apprehensive older adult. Barriers to transportation can be overcome by community-based senior centers that provide both transportation and physical activity programs, while many group living situations provide physical activity programs on-site such as in long-term care facilities, nursing homes, assisted living facilities, and independent living communities. About 90 % of community-based senior centers offer at least one evidence-based program, many of which include physical activity [48]. Moreover, some physical activity ity programs are designed for use in the home with guidance from instructional booklets and DVDs.

	ince-nased pil.	לווזאון מרוועוו		oluci auuls		
Program	Nutrition	Physical activity	Program goals	Program description	Target population	Organization and contact
Eat Better, Move More	>	>	Increase daily consumption of fruits and vegetables, fiber, and calcium	12-week group-based education sessions (≤30 min) on nutrition and physical activity lead by health professional	Older adults of any health and fitness level (with physician clearance)	Florida International University, National Resource Center on Nutrition, Physical Activity, and Aging
			Eat sensible portions for healthy weight Accumulate at least 30 min physical activity on most days of the week	Step counters used for motivation		http://nutritionandaging.fiu.edu/ You_Can/index.asp
Active Living Every Day (ALED)		\$	Integrate more physical activity into their daily lives	12-week group-based education Book with optional online companion materials	Adults interested in integrating physical activity into their daily lives	Human Kinetics www.ActiveLiving.info/ OR 800-747-4457
Healthy Eating Every Day (HEED)	>		Improve eating habits for optimal health, set realistic goals and rewards, and cope with triggers for unhealthy eating	12-week group-based education OR online individual course Book with online companion materials	Adults interested in improving nutritional health	Human Kinetics www.ActiveLiving.info/ OR 800-747-4457
Diabetes Prevention Program	`	`	Prevent or delay onset of type 2 diabetes by improving nutrition and meeting DHHS 2008 Guidelines for physical activity	1-year program Lifestyle coaches provide 22 group education sessions Training provided by DTTAC ^b	Adults interested in preventing or delaying the onset of type 2 diabetes	CDC and private partners http://www.cdc.gov/diabetes/ prevention/about.htm

Table 22.1 Evidence-based physical activity and nutrition programs for older adults^a

(continued)

	(
Program	Nutrition	Physical activity	Program goals	Program description	Target population	Organization and contact
Stanford School of Medicine Chronic	>	>	Improve chronic disease self- management	6 weeks 2.5 h per week in community centers, churches, libraries, hospitals	Adults with one or more chronic diseases	Stanford School of Medicine
Disease Self- Management Program			through education regarding nutrition, physical activity, stress-management, interpersonal communication, medication management, and behavioral concerns	Program provides information and teaches practical skills on managing chronic health problems		http://patienteducation.stanford. edu/programs/cdsmp.html
Eat Smart, Live Long	`	\$	Improve fruit and vegetable consumption (at least 3.5 cups/day)	4 interactive sessions lead by a community member	Able-bodied older adult (60-74 y) participants of USDA FNS nutrition assistance	USDA Food and Nutrition Services
			Participate in at least 30 min of moderate- intensity physical activity most days of the week	Leaders Guide and Session Guides provide self-assessment tools and education CD of reproducible participant handouts	programs (SNAP ^e , SFMNP ^d , OAA ^e Congregate Meals Providers)	http://www.arthritis.org/ resources/community- programs/walk-with-ease/
Walk With Ease		`	Reduce the pain and discomfort of arthritis	Certified instructor leads 1-h group classes 3days/week for 6 weeks	Older adults with arthritis who can be on their feet for at	Arthritis Foundation
			Increase balance, strength and walking pace	Participant guidebook provided with fitness routine, stretching exercises, heart-rate monitoring techniques	least 10 min without increased pain	http://www.arthritis.org/ resources/community- programs/walk-with-ease/
			Build confidence in ability to be physically active Improve overall health	Companion online tools, videos, and social media		OR call 1-800-283-7800

 Table 22.1 (continued)

Healthy Moves for Aging Well		`	Help maximize independence by building strength,	In-home physical activity intervention (chair bound and advanced exercises)	Frail, sedentary adults ≥65 years currently enrolled in a care	Partners in Care Foundation
			increasing flexibility, and	Exercises practiced 3-5 days/week	management program; need	http://www.picf.org/landing_ pages/22,3.html
			helping to reduce the risk of falls	One 15-min goal setting and instructional session between	assistance with 2–4 ADLs ^f)
				care manager and participant		
Project Enhance		>	Enhance Fitness:	(EF) Group physical activity	(EF) Older adults	Senior Services (Seattle, WA)
(Enhance Fitness.			Improve functional fitness of older	sessions focused on stretching and flexibility. low imnact	seeking to maintain or improve their	and University of Washington
Enhance			adults	aerobics, strength training, and	physical functioning	0
Wellness)				balance; 1 h sessions offered	5	
				3 days/week		
			Enhance Wellness	(EW) Health professionals work	(EW) Older adults with	www.projectenhance.org/
			(EW): Behavior	with participant and physician	chronic conditions	EnhanceFitness.aspx
			modification for	to develop behavioral goals and		
			those with chronic conditions	action plan		
Go4Life	>	>	Improve physical	Individual level physical activities	Adults ≥50 years	National Institute on Aging
			activity uptake			(NIA) at the National Institute of Health
				Workout to Go: Sample exercise		http://go4life.nia.nih.gov/
				routine (13 exercises) can be		- - -
				ordered or downloaded for print		
				Guidebook and video, tip sheets,		
				exercise audio book, healthy		
				aging brochures and pamphlets		
^a Adapted from DHF	IS AoA Title	: III-D Highe	est Tier Evidence-Based He	alth Promotion/Disease Prevention Pro-	grams: http://www.ncoa.org/	/improve-health/center-for-healthy-
aging/content-librar	y/Title-IIID-l	Highest-Tier	-Evidence-FINAL.pdf [52]			
^b Diabetes Training ^a	und Technical	Assistance	Center (DTTAC) at Emory	University		
'Supplemental Nutri	tion Assistar	ice Program	(SNAP)			
*Semor Farmers Ma Older Americane A	rket Nutritio	n Program (2	SFININE)			
Activities of Daily	Living (ADL					

22 Physical Activity and Exercise: Important Complements to Nutrition in Older Adults

Programs such as the *Stanford Chronic Disease Self-Management Program* encourage lifestyle changes, including increased physical activity, applicable to prevention and management of multiple chronic diseases, while other evidence-based programs are disease specific, such as the *Walk With Ease* Program that specifically targets alleviating arthritis discomfort [51, 53].

Clinicians should also contact other local organizations, such as faith communities, health systems, colleges and universities, and local fitness centers, including the YMCA, to identify existing physical activity programs and explore forming partnerships to offer these programs [54]. For example, the SilverSneakers program is offered by several Medicare health plans and provides membership with many fitness centers, including both corporate gyms and YMCAs, and access to a SilverSneakers® Program Advisor and SilverSneakers® exclusive classes [55]. The SilverSneakers program also provides personalized fitness programs for those older adults who do not have convenient access to a SilverSneakers® provider at a fitness center [55].

Table 22.1 highlights several evidence-based health promotion/disease prevention physical activity and nutrition programs, of which many are endorsed by NCOA and NIA and qualify for OAA Title IIID funding [51]. As physical activity and nutrition provide synergistic benefits for health and wellbeing, programs that target both, such as Florida International University's *Eat Better Move More* and Human Kinetics' companion programs *Active Living Every Day* and *Healthy Eating Every Day*, may be optimal for improving the health and well-being of older adults. With readily available evidence-based resources, clinicians can be instrumental in encouraging older adults to become more physically active.

References

- United States Department of Health and Human Services (DHHS). 2008 Physical activity guidelines for Americans: be active, healthy, and happy! Washington, DC. Available from: http://www.health.gov/paguidelines/guidelines/ default.aspx.
- Chodzko-Zajko WJ, Proctor DN, Fiatarone Singh MA, Minson CT, Nigg CR, Salem GJ, et al. American College of Sports Medicine position stand. Exercise and physical activity for older adults. Med Sci Sports Exerc. 2009; 41(7):1510–30.
- Proper KI, Singh AS, van Mechelen W, Chinapaw MJM. Sedentary behaviors and health outcomes among adults a systematic review of prospective studies. Am J Prev Med. 2011;40(2):174–82.
- Katzmarzyk PT, Church TS, Craig CL, Bouchard C. Sitting time and mortality from all causes, cardiovascular disease, and cancer. Med Sci Sports Exerc. 2009;41(5):998–1005.
- 5. Bouchard C, Shephard RJ, Stephens T, editors. Physical activity, fitness, and health: international proceedings and consensus statement. Champaign, IL: Human Kinetics Publishers; 1994.
- Villareal DT, Apovian CM, Kushner RF, Klein S. Obesity in older adults: technical review and position statement of the American Society for Nutrition and NAASO, the Obesity Society. Am J Clin Nutr. 2005;82(5):923–34.
- Mojtahedi MC, Thorpe MP, Karampinos DC, Johnson CL, Layman DK, Georgiadis JG, et al. The effects of a higher protein intake during energy restriction on changes in body composition and physical function in older women. J Gerontol A Biol Sci Med Sci. 2011;66(11):1218–25.
- Layman DK, Evans EM, Erickson D, Seyler J, Weber J, Bagshaw D, et al. A moderate-protein diet produces sustained weight loss and long-term changes in body composition and blood lipids in obese adults. J Nutr. 2009; 139(3):514–21.
- 9. Layman DK, Evans EM, Baum JI, Seyler J, Erickson DJ, Boileau RA. Dietary protein and exercise have additive effects on body composition during weight loss in adult women. J Nutr. 2005;135(8):1903–10.
- US Department of Agriculture (USDA) and US Department of Health and Human Services (DHHS). Dietary guidelines for Americans, 2010. 7th ed. Washington, DC: U.S. Government Printing Office; 2010.
- 11. Salive ME. Multimorbidity in older adults. Epidemiol Rev. 2013;35(1):75-83.
- Holmes J, Powell-Griner E, Lethbridge-Cejku M, Heyman K. Aging differently: physical limitations among adults aged 50 years and over: United States, 2001–2007. NCHS Data Brief. 2009;20:1–8.
- Ogden CL, Carroll MD, Kit BK, Flegal KM. Prevalence of obesity in the United States, 2009–2010. NCHS Data Brief. 2012;82:1–8.

- 14. Vincent HK, Vincent KR, Lamb KM. Obesity and mobility disability in the older adult. Obes Rev. 2010;11(8):568–79.
- 15. National Institutes of Health (NIH) National Heart, Lung, and Blood Institute (NHLBI). The practical guide: identification, evaluation, and treatment of overweight and obesity in adults. Bethesda, MD. 1998. Available from: http://catalog.nhlbi.nih.gov/catalog/product/The-Practical-Guide-Identification-Evaluation-and-Treatmentof-Overweight-and-Obesity-in-Adults/00-4084
- Kenney LW, Wilmore JH, Costill DL. Energy expenditure and fatigue. In: Physiology of sport and exercise. 5th ed. Champaign, IL: Human Kinetics; 2012.
- Donnelly JE, Blair SN, Jakicic JM, Manore MM, Rankin JW, Smith BK. Appropriate physical activity intervention strategies for weight loss and prevention of weight regain for adults. Med Sci Sports Exerc. 2009;41(2):459–71.
- Waters DL, Ward AL, Villareal DT. Weight loss in obese adults 65 years and older: a review of the controversy. Exp Gerontol. 2013;48:1054.
- Lloyd-Jones DM, Hong Y, Labarthe D, Mozaffarian D, Appel LJ, Van Horn L, et al. Defining and setting national goals for cardiovascular health promotion and disease reduction: the American Heart Association's strategic impact goal through 2020 and beyond. Circulation. 2010;121(4):586–613.
- 20. Balady GJ, Williams MA, Ades PA, Bittner V, Comoss P, Foody JA, et al. Core components of cardiac rehabilitation/secondary prevention programs: 2007 update: a scientific statement from the American Heart Association Exercise, Cardiac Rehabilitation, and Prevention Committee, the Council on Clinical Cardiology; the Councils on Cardiovascular Nursing, Epidemiology and Prevention, and Nutrition, Physical Activity, and Metabolism; and the American Association of Cardiovascular and Pulmonary Rehabilitation. J Cardiopulm Rehabil Prev. 2007; 27:121–9.
- Miller EL, Murray L, Richards L, Zorowitz RD, Bakas T, Clark P, et al. Comprehensive overview of nursing and interdisciplinary rehabilitation care of the stroke patient: a scientific statement from the American Heart Association. Stroke. 2010;41:2402–48.
- Bowles DK, Laughlin MH. Mechanism of beneficial effects of physical activity on atherosclerosis and coronary heart disease. J Appl Physiol. 2011;111(1):308–10.
- Prasad S, Sung B, Aggarwal B. Age-associated chronic diseases require age-old medicine: role of chronic inflammation. Prev Med. 2012;54:S29–37.
- Centers for Prevention and Disease Control (CDC). Vital signs: prevalence, treatment, and control of hypertensionited States, 1999–2002 and 2005–2008. MMWR Morb Mortal Wkly Rep [Internet]. 2011; 60(4):103–8. Available from: http://www.cdc.gov/mmwr/preview/mmwrhtml/mm6004a4.htm
- Pescatello LS, Franklin BA, Fagard R, Farquhar WB, Kelley GA, Ray CA. American College of Sports Medicine position stand. Exercise and hypertension. Med Sci Sports Exerc. 2004;36(3):533–53.
- 26. Ervin RB. Prevalence of metabolic syndrome among adults 20 years of age and over, by sex, age, race and ethnicity, and body mass index: United States, 2003–2006. Natl Health Stat Rep. 2009;(13):1–7
- Colberg SR, Sigal RJ, Fernhall B, Regensteiner JG, Blissmer BJ, Rubin RR, et al. Exercise and type 2 diabetes: the American College of Sports Medicine and the American Diabetes Association: joint position statement executive summary. Diabetes Care. 2010;33(12):2692–6.
- Roitman JL, LaFontaine T. Exercise and physical activity. In: The exercise professional's guide to optimizing health: strategies for preventing and reducing chronic disease. Philadelphia, PA: Lippincott Williams & Wilkins; 2012. p. 46–73.
- American Cancer Society. Cancer facts and figures 2013. Atlanta, GA: American Cancer Society; 2013 [cited 2013 April 17]; Available from: http://www.cancer.org/research/cancerfactsfigures/cancerfactsfigures/cancer-facts-figures-2013
- Schmitz KH, Courneya KS, Matthews C, Demark-Wahnefried W, Galvao DA, Pinto BM, et al. American College of Sports Medicine roundtable on exercise guidelines for cancer survivors. Med Sci Sports Exerc. 2010; 42(7):1409–26.
- Giovannucci E. Insulin, insulin-like growth factors and colon cancer: a review of the evidence. J Nutr. 2001;131(11 Suppl):3109S-20.
- Brown KA, Simpson ER. Obesity and breast cancer: mechanisms and therapeutic implications. Front Biosci (Elite Ed). 2012;4:2515–24.
- 33. Centers for Disease Control and Prevention National Center for Injury Prevention and Control, Division of Unintentional Injury Prevention. Falls among older adults: an overview 2012 [cited 2013 Apr 22]. Available from: http://www.cdc.gov/homeandrecreationalsafety/falls/adultfalls.html
- 34. National Osteoporosis Foundation. What is osteoporosis. 2013 [cited 2013 April 7]; Available from: http://www.nof.org/articles/7
- Kohrt WM, Bloomfield SA, Little KD, Nelson ME, Yingling VR. American College of Sports Medicine position stand: physical activity and bone health. Med Sci Sports Exerc. 2004;36(11):1985–96.
- Chen JH, Liu C, You L, Simmons CA. Boning up on Wolff's law: mechanical regulation of the cells that make and maintain bone. J Biomech. 2010;43(1):108–18.

- Colcombe S, Kramer AF. Fitness effects on the cognitive function of older adults: a meta-analytic study. Psychol Sci. 2003;14(2):125–30.
- 38. Bridle C, Spanjers K, Patel S, Atherton NM, Lamb SE. Effect of exercise on depression severity in older people: systematic review and meta-analysis of randomised controlled trials. Br J Psychiatry. 2012;201(3):180–5.
- Blake H, Mo P, Malik S, Thomas S. How effective are physical activity interventions for alleviating depressive symptoms in older people? A systematic review. Clin Rehabil. 2009;23(10):873–87.
- Dishman RK, Washburn RA, Heath GW. Physical activity and mental health. In: Physical activity epidemiology. Champaign, IL: Human Kinetics Publishers; 2004. p. 379–440.
- 41. Dishman RK, Washburn RA, Heath GW. All-cause mortality. In: Physical activity epidemiology. Champaign, IL: Human Kinetics Publishers; 2004. p. 77–94.
- 42. Behavioral risk factor surveillance system data. [Internet]. Atlanta, GA: Centers for Disease Control and Prevention. c 1996, 2005, 2011; Available from: http://www.cdc.gov/brfss/
- 43. National Institute on Aging (NIA). Exercise & physical activity: your everyday guide from the National Institute on Aging.Gaithersburg,MD.2009.Available from:http://www.nia.nih.gov/health/publication/exercise-physical-activityyour-everyday-guide-national-institute-aging-1
- 44. Schutzer KA, Graves BS. Barriers and motivations to exercise in older adults. Prev Med. 2004;39(5):1056-61.
- 45. Centers for Disease Control and Prevention (CDC). Healthy Aging: helping people to live long and productive lives and enjoy a good quality of life [Internet]. 2011 [cited 2013 Apr 7]; Available from: http://www.cdc.gov/chronicdisease/resources/publications/AAG/aging.htm
- 46. Barnes PM, Schoenborn CA. Trends in adults receiving a recommendation for exercise or other physical activity from a physician or other health professional, NCHS Data Brief, vol. 86. Hyattsville, MD: National Center for Health Statistics; 2012.
- 47. American College of Sports Medicine. Exercise is medicine [Internet]. 2008 [cited 2013 Apr 7]; Available from: http://exerciseismedicine.org
- National Council on Aging (NCOA) Center for Healthy Aging (US). About evidence-based programs. [Internet] Washington, DC, 2013 [cited 2013 February 2, 2013]; Available from: http://www.ncoa.org/improve-health/centerfor-healthy-aging/about-evidence-based-programs.html
- 49. (US) Administration on Aging. Disease prevention and health promotion services (oaa title iid). Washington, DC [updated 2013 Feb 5; cited 2013 February 2]; Available from: http://www.aoa.gov/AoARoot/AoA_Programs/ HPW/Title_IIID/index.aspx#purpose
- 50. National Institute on Aging (NIA) at the National Institutes of Health (NIH). Go4life: info for health professionals. [Internet] 2013 [cited 2013 February 2]; Available from: http://go4life.nia.nih.gov/about/health-professionals
- 51. National Council on Aging (NCOA) Center for Healthy Aging (US). Title iii-d highest tier evidence-based health promotion/disease prevention programs. Washington, DC: NCOA; 2012 [updated 2013 January]
- 52. Partners in Care Foundation (Partners). Healthy moves for aging well. [Internet] [cited 2013 February 2]; Available from: http://www.picf.org/landing_pages/22,3.html
- Arthritis Foundation. Walk with ease. [Internet] [cited 2013 February 2]; Available from: http://www.arthritis.org/ resources/community-programs/walk-with-ease/
- National Council on Aging (NCOA). Healthy aging programs: secrets to success. [Internet] 2012 [cited 2013 February 2]; Available from: http://www.ncoa.org/national-institute-of-senior-centers/nisc-news/healthy-agingprograms.html
- 55. Healthways Inc. SilverSneakers®? [Internet] Franklin, TN [cited 2014 September 16]; Available from: http://www.silversneakers.com/.

Chapter 23 Dietary Supplements in Older Adults

Johanna T. Dwyer, Rebecca Costello, Joyce Merkel, and Paul M. Coates

Key Points

- There are times when food alone cannot meet all the nutrient requirements of older adults, and certain dietary and medical foods can help maintain health.
- However, an established need for supplements is highly variable depending upon the nutrient, e.g., strong for vitamin D and calcium to prevent osteoporosis versus unclear for B-vitamins to protect blood vessels.
- Factors increasing the likelihood of needing supplements include osteoporosis, alcohol abuse, gastrointestinal abnormalities, renal insufficiency, cardiovascular disease, nutritional anemias, or any situation in which food intake is chronically inadequate.
- Health care providers should routinely screen elderly clients for both the need to use dietary supplements and the risks of excessive supplement intake, which may include economic difficulties as well as interactions with medications and other supplements.

Keywords Dietary supplements • Vitamins • Minerals • Botanicals • Herbals

Introduction

This chapter focuses on the rationale and evidence base for the use of dietary supplements for both maintaining health and decreasing the risks of some common chronic diseases. This review also updates the information in a review of data through 2009 on nutritional supplements for older adults [1] by summarizing evidence published in 2009–2013 from systematic reviews and meta-analyses of randomized clinical trials with disease endpoints. The chapter concludes with recommendations for assessing dietary supplement use in older persons and suggestions for counseling them. We have categorized these supplements according to their active ingredient(s), and we pay particular attention to

Note: The findings and views reported in this chapter represent those of the contributing authors and not necessarily those of the National Institutes of Health and are not intended to constitute an "authoritative statement" under the Food and Drug Administration rules and regulations.

J.T. Dwyer, D.Sc., R.D. • R. Costello, Ph.D. • J. Merkel, M.S., R.D. • P.M. Coates, Ph.D. Office of Dietary Supplements, National Institutes of Health, 6100 Executive Blvd Room 3B01, MSC7517, Bethesda, MD 20892, USA e-mail: dwyerj1@od.nih.gov; costellb@od.nih.gov; merkelj@od.nih.gov; coatesp@od.nih.gov

claims on supplement labels about these products' effects on health conditions or the health of specific parts of the body.

Sound dietary patterns are vital to maintaining health and well-being. However, there are times when food alone cannot meet all nutrient requirements, or physiological limitations due to disease alter nutrition needs. Under such circumstances, certain dietary and medical foods can help maintain health. For example, the need for supplemental vitamin D and calcium to prevent osteoporosis in older adults is well established [2–7]. In contrast, the effectiveness of the use of supplements containing B-vitamins, such as folic acid and vitamins B_6 and B_{12} , in preventing blood vessel diseases in older adults is not clear [8–10]. The reason why some dietary supplements are not effective in reducing disease risk might be that, with a few exceptions, clinical deficiency of vitamins or minerals (other than calcium and possibly vitamin D [7, 11] and iron) is uncommon in the United States, except in certain high-risk groups such as the elderly. In other cases, excessive amounts of nutrients from fortificants or supplements might have adverse effects. For example, very high doses of vitamin D that some elderly individuals take in hope of preventing osteoporosis may give rise to health problems, such as kidney stones [12].

Dietary supplements marketed today include many ingredients other than essential nutrients, such as extracts of herbs and other plants. Older Americans use these products to prevent, manage, or lessen the impact of age-associated disorders and conditions. Such use might be inappropriate for certain conditions and diseases [13]. In one recent analysis of data on adults older than 19 from the 2007–2010 National Health and Nutrition Examination Survey (NHANES), only 23 % of supplements were used in accordance with health care providers' recommendations [13]. Although approximately 25 % of all drugs used today are derived from plants [14], very few of the botanical supplement products on the market have been rigorously tested in clinical studies for efficacy and toxicity. Another concern is the risk of interactions between dietary supplements and other medications used by the elderly [15].

The Dietary Supplement Marketplace

According to the 2012 edition of *Nutrition Business Journal*, supplement sales in the United States now total more than \$30 billion each year, fueled in large part by consumers' desires for health and wellness [16]. The top five dietary supplement companies include pharmaceutical companies (Pfizer, Bayer Group, and Pharmavite) and contract manufacturers (such as Perrigo) that produce a variety of products for a variety of companies, specialty product manufacturers (such as Glanbia, Schwabe NA, and Integrative Therapeutics) that do so for a more limited group of products, and holding companies (such as the Carlyle Group).

Types of Dietary Supplements

Dietary supplements are categorized in many ways. For example, marketers may group them by source (i.e., animal or plant), nutrition researchers by their essentiality to human health or chemical composition (e.g., vitamin, mineral, or protein), epidemiologists by prevalence of use, regulators by their purported health benefits (e.g., for bone health or heart health) or manufacturing or production processes used (e.g., use of organic ingredients, production using traditional methods, or synthesis from artificial ingredients); or a combinations of these categories. However, marketers frequently promote supplements based on their purported health benefits, even if these products' efficacy has not been demonstrated.

The doses of bioactive ingredients that supplements contain can vary greatly; some supplements contain 100 % or more of the reference daily value (DV) per serving and some contain much less. "High-potency" nutrient products (containing 100 % of more of the DV per serving) and non-nutrient supplements containing large amounts of single ingredients or concentrated extracts have the potential for adverse interactions with other dietary constituents and drugs [15, 17]. These interactions can lead to health problems in elderly persons taking many medications.

Motivations for Dietary Supplement Use

The recent analysis of the 2007–2010 NHANES found that American adults use supplements to "improve" (43 %) or "maintain" (33 %) their overall health rather than to supplement intakes of nutrients from food [13]. Condition-specific use, such as for "bone health" or "heart health or to lower cholesterol" is also high. Adults older than 60 years are more likely than younger people to report motivations for using dietary supplements that are related to specific health conditions or the health of parts of the body, such as the heart, bones, joints, or eyes.

Dietary Supplement Use by Older Americans

About half of all adults and 70 % of those older than 71 years in the United States use dietary supplements [18]. Among older Americans, slightly more women (75 %) than men (66 %) use supplements. Multivitamin/multimineral (MVM) supplements are by far the most popular among older persons (46 % report using them in the past month), whereas fewer (17 %) report using botanical supplements [18]. Among older adults, prevalence of use of supplements containing at least one of the following vitamins is as follows: 41 % for vitamin C, 40 % for vitamin E, 37 % for vitamin A, 36 % for vitamin B₆, and 28 % for vitamin K. Prevalence of use for minerals in this age group is 36 % for zinc, 35 % for magnesium, 32 % for selenium, 29 % for chromium, and 16 % for iron. In addition, more older adults use supplements containing one or more vitamins and/or minerals than those containing botanical or other ingredients [19].

Adults who take more herbal and other botanical products often use these products to treat specific health conditions, such as upper respiratory tract infection or anxiety [20]. In a recent study using data from the National Health Interview Survey, individuals with more severe diabetes were more likely to use complementary and alternative medicines, including diet-based interventions and nonvitamin/ nonmineral dietary supplements, than those with less severe disease [21].

Surveys on dietary supplement use do not readily lend themselves to comparisons between studies because patterns of use change from year to year, the collection instruments or methods are different (e.g., telephone versus online surveys, 24-h recall versus food frequency questionnaires), participants might be queried about different time periods of use (e.g., over the last week versus the last month), and respondents might be confused about what types of supplements are relevant to a given question.

Current Market Trends in Supplements

The nutrient-containing supplements on the market today vary in their forms as well as the nutrient doses they provide. These products include: (1) MVM products; (2) multivitamins, such as B-complex vitamins or combinations of antioxidant vitamins; (3) calcium-containing products with

or without other minerals or vitamins, such as calcium, calcium and vitamin D, and Tums and other antacids with calcium as a primary ingredient; (4) multimineral preparations, such as magnesium and zinc; (5) single-vitamin/mineral or single-mineral supplements, such as supplements providing vitamin A or iron; (6) omega-3 and omega-6 fatty acids, such as fish oils, docosahexaenoic acid (DHA), and flaxseed; (7) products containing protein, amino acids, and nitrogenous ingredients, such as creatine, lysine, arginine, or creatinine; and (8) products with fiber as the primary ingredient or fiber used as a laxative, including over-the-counter products with dual labeling, such as Metamucil and Citrucel.

In the 1990s, the use of herbal products and high-dose vitamins became much more common [22]. During this period, sales of condition-specific and other specialty supplements, such as sports nutrition products, also grew rapidly, and many consumers switched from single-nutrient supplements to combination and condition-specific products [16, 23].

In marketing dietary supplements, manufacturers often make claims about these products' effectiveness in promoting the health of various organs. Marketers also promote combinations of supplements and prescription medications for specific conditions in drugstore chains, health food stores, grocery stores, and magazines, although such efforts are not always successful. Nevertheless, vitamin, multivitamin, and mineral supplements are among the items most frequently purchased with a heart-related prescription drug in drugstores and grocery stores [24].

The number of different dietary supplement products sold in the United States is not known, although the U.S. Food and Drug Administration (FDA) has estimated the total to be at least 55,000 [25]. Dietary supplements are produced in many different parts of the globe and are sold through many channels, including the Internet, drugstores, supermarkets, and grocery stores, health food stores, and fitness centers, as well as in some healthcare providers' offices. New supplements constantly appear on the market. The compositions, quality, and prices of supplements vary greatly, and even those with similar formulations can differ markedly from one producer to another.

Advertising themes closely track changes in population demographics, such as the greater number of older persons and emerging trends in science. Dietary supplements are marketed so imaginatively that it is easy for older Americans to infer that supplements can do far more than simply add to dietary intake, such as preventing or treating chronic diseases. Indeed, the latest compilation of the trade publication *Nutrition Business Journal's* annual statistics on the sales of supplement marketed in the United States today suggests that there are no ills or miseries of mankind for which there is not a condition-specific supplement on the market [16]. People who counsel the elderly should be aware of the current marketing appeals and be ready to discuss the actual safety with their patients and address the often quite different efficacy of these products from advertised claims.

Multivitamins are among the dietary supplements marketed today with appeals to general health. Manufacturers claim that many supplements have anti-aging effects, appealing to the desire for an enhanced appearance and youthful looks or allaying fears about memory loss. Coenzyme Q10 (CoQ10) is currently popular for anti-aging. In their marketing claims, CoQ10 manufacturers often mention telomere length (a marker of biological age) and "telomere health," which is supposedly improved by ingredients such as resveratrol; omega-3 fatty acids; and vitamins C, D, and E. However, there is little evidence that these claims are factual.

Many condition-specific supplements appeal to elders. Brain health and mental acuity also seem to have a special appeal, and nearly half the sales of products in this category are of fish and animal oils high in omega-3 fatty acids [16]. Manufacturers claim that supplement ingredients (including vitamins B, D, and E; zinc; phosphatidylserine; CoQ10; *Ginkgo biloba*; St. John's wort; SAM-e; and kava) improve cognition, reduce anxiety and depression, and slow the onset of dementia and the progression of Alzheimer disease. In addition, many supplements, such as multivitamins and 5-hydroxytryptophan, are sold for mood disorders, such as depression. For people with trouble sleeping, manufacturers claim that herbs, such as valerian, and other ingredients, such as melatonin, provide relief from stress and insomnia.

Cardiovascular/heart heath condition-specific supplement sales are also growing rapidly, particularly for omega-3-rich fish and animal oils [16]. Sales fell for folic acid and vitamin E after large clinical trials showed few benefits [26–33], but the supplement industry keeps marketing others, such as potassium, magnesium, and calcium for high blood pressure. Some products, such as CoQ10 and proathocyanidins (naturally found in dark chocolate, tea, and Concord grape juice), are being tested in clinical trials, but they are already being marketed before results are available.

For the millions of older Americans living with osteoporosis, arthritis, osteoarthritis of the knee, and chronic joint and bone pain, glucosamine/chondroitin products marketed for these conditions are widely available. For bone health, calcium is the most commonly marketed product, with vitamin D and vitamin K close behind [16]. Elderly people worry about vision and eye health, particularly advanced age-related macular degeneration that leads to vision loss and cataracts [34]. Among vision health supplements, lutein is the most popular. Sales of diabetes and metabolic syndrome condition-specific supplements are also growing rapidly, with vitamin B and biotin being the biggest sellers in this category.

Regulation of Dietary Supplements

The safety and efficacy of dietary supplements are critical to consider when dealing with the health of older Americans. Although most dietary supplements are both safe and effective, they are not tested as rigorously as prescription drugs. As with conventional foods, Congress gave the FDA the authority to regulate dietary supplements once they are on the market. However, unlike medical devices, the FDA does not have the authority to regulate dietary supplements before they are marketed. Members of the American public, including elders, are likely to believe that the FDA reviews the safety and effectiveness of dietary supplements before they are marketed. Therefore, it is important for those who counsel elders to understand what the law does and does not say. A brief review of the relevant statutes follows.

The Dietary Supplement Health and Education Act (DSHEA; Public Law [PL] 103-417) laid the foundation for the current regulatory framework for dietary supplements in the United States. This law amended the Federal Food, Drug, and Cosmetic Act of 1938 "to establish standards with respect to dietary supplements." DSHEA defines a dietary supplement as "a product (other than tobacco) intended to supplement the diet that bears or contains one or more of the following dietary ingredients: a vitamin; a mineral; an herb or other botanical; an amino acid; a dietary substance for use by man to supplement the diet by increasing the total dietary intake; or a concentrate, metabolite, constituent, extract, or combination of any [of the ingredients described above]."

Under DSHEA, the FDA regulates the safety and manufacturing of, and information about, dietary supplements, such as claims in product labels, package inserts, and accompanying literature. The FDA cannot require testing of dietary supplements prior to marketing (except for new dietary ingredients). However, although DSHEA prohibits manufacturers from selling dangerous products, it permits the FDA to remove a product from the marketplace only when the agency proves that the product is dangerous to the health of Americans. If in the labeling or marketing of a dietary supplement product a claim is made that the product can diagnose, treat, cure, or prevent disease, such as "cures cancer," the FDA determines that the product is an unapproved new drug and its sales are thus illegal.

Regulations requiring good manufacturing practices for dietary supplements and finished products have been in place since 2007 [35]. The Public Health Security and Bioterrorism Preparedness and Response Act of 2002 (PL 107-188) requires registration of manufacturers, processors, and packers of foods and dietary supplements with the FDA. The Dietary Supplement and Nonprescription Drug Consumer Protection Act (commonly known as the Adverse Event Reporting, or AER, law; PL 109-462) of 2006 requires manufacturers of dietary supplements and over-the-counter products to submit

serious adverse event reports to the FDA. The Food Safety Modernization Act of 2011 (PL 111-353) provides the FDA with mandatory recall authority for adulterated dietary supplements and foods containing undisclosed allergens.

The process for removing an unsafe dietary supplement from the market is still cumbersome, with the burden of proof falling on the FDA, even after the FDA received mandatory recall authority. Congress and the FDA continue to balance the often-conflicting goals of safety and effectiveness; access to up-to-date, complete, and unbiased information on dietary supplements; accurate reporting of adverse events; and consumer choice [36].

Health Claims for Supplements

The Nutrition Labeling and Education Act of 1990 (PL 101-535) that allowed conventional foods and dietary supplements to be labeled with health claims also permitted health claim labels for supplements that met the specified criteria. In this context, health claims are statements describing relationships between a food or food component and a disease, such as cardiovascular disease, or a health-related condition, such as hypertension, that is considered to be a surrogate marker for a disease (in this case, for cardiovascular disease). Health claims are limited to claims about disease risk reduction and cannot suggest that a product cures, mitigates, treats, or prevents a disease.

The FDA may use several ways to determine which health claims may be used on a label or in labeling for dietary supplements. The FDA website provides details on these processes (http://www.fda.gov/Food/GuidanceRegulation/). In some cases, the FDA approves health claims based on evidence that a nutrient or dietary supplement influences a biomarker in a way that is associated with reduced disease risk.

When evidence suggests that a product has a potential benefit but the FDA does not consider the evidence to be strong, it might permit a qualified health claim. Qualified health claims must be accompanied by language such as "FDA has determined that this evidence is limited and not conclusive."

The FDA has authorized 16 health-disease-related claims for foods, including 8 for cardiovascular disease, 4 for cancer, 2 for dental caries, 1 for neural tube defects, and 1 for osteoporosis-related conditions. Only three approved health claims are authorized for dietary supplements: one for osteoporosis, one for neural tube defects, and one for cardiovascular disease. The FDA has approved 17 qualified health claims for foods: 7 for cardiovascular disease, 5 for cancer, 1 for cognitive function, 1 for atopic dermatitis, 1 for hypertension, 1 for neural tube birth defects, and 1 for diabetes-related conditions. The qualified health claims for supplements include at least 2 for cardiovascular disease, 6 for cancer, 1 for cognitive function, 1 for hypertension, 1 for neural tube defects, and 1 for diabetes.

Manufacturers may make structure-function claims on the labels of conventional foods and dietary supplements to describe the role of ingredients intended to affect a normal structure or function in humans. The FDA does not review or authorize structure-function claims prior to their use. For this reason, these claims must carry a disclaimer statement.

Effectiveness and Safety of Nutrient Supplements for the Elderly

Evaluations of the associations between supplements, health, and disease risk are complicated because the formulations used vary so much from study to study. Furthermore, supplement users and nonusers have varied demographic, lifestyle, and other characteristics that affect their health. Nevertheless, some conclusions can be drawn, as we show in this section. Space does not permit a complete discussion here of all of the essential nutrients provided in dietary supplements, so we have given priority to the recent evidence for or against the efficacy of the most popular dietary supplement ingredients among older Americans. We discuss these ingredients roughly in the order of their prevalence of use.

Supplements Containing Combinations of Vitamins

MVMs

There is no regulatory definition of what nutrients or how much of certain nutrients must be included in MVMs. Many different products with various levels of nutrients and other ingredients are in the marketplace today. The most common MVMs contain vitamins that are generally at levels close to their daily values (DVs), but they contain much lower amounts than their DVs of some minerals that would otherwise add bulk to the products. The special "silver" formulations for elders usually provide the same vitamins and minerals as those marketed to all adults but in amounts that are closer to the recommended dietary allowance (RDA) levels for older individuals. High-potency preparations contain greater amounts of both vitamins and minerals than standard MVMs. Specialized "conditionspecific" supplements appealing to elder concerns, such as immune function or maintenance of memory, contain several vitamins and minerals in combination with other ingredients.

MVMs: Supplementing Nutrients in the Diet

One of the reasons why older people take MVMs is to ensure that they are obtaining adequate amounts of nutrients to maximize their health [13]. Taking MVMs does increase intakes of the nutrients these products contain. However, the nutrients in MVMs might not be those that are low or lacking in the user's diet, or the amounts present might not suffice to meet the user's nutrient needs. Also, MVM users tend to have good diets with relatively high nutrient intakes, so they might not need MVMs. Furthermore, some nutrients might already be present in plentiful or excessive amounts in users' diets. If many "high-potency" MVM supplements are taken daily, excessive intakes may result. Paradoxically, elders who are most likely to benefit from MVMs because of low micronutrient intakes are least likely to take them [18].

MVMs: Chronic Disease Prevention

Some older people take MVMs to prevent or treat chronic degenerative diseases and other ills. In 2006, the Office of Medical Applications of Research and Office of Dietary Supplements at the National Institutes of Health sponsored a systematic review of randomized controlled trials (RCTs) that assessed the risks and benefits of MVMs for many chronic diseases [37]. An expert committee evaluated the state of the science and concluded that "the present evidence is insufficient to recommend either for or against the use of MVMs by the American public to prevent chronic disease." The U.S. Preventive Services Task Force also found that the evidence was insufficient to determine whether MVMs can prevent cancer or cardiovascular disease [38]. In its 2007 report, *Food, Nutrition, Physical Activity, and the Prevention of Cancer: a Global Perspective*, the World Cancer Research Foundation and American Institute for Cancer Research recommended against using supplements to prevent cancer; instead, these organizations recommended that people meet their nutritional needs through diet alone, which would maximize the proportion of the population achieving nutritional adequacy without supplements [39].

MVMs: Mortality

Whether MVM supplements increase or decrease mortality risk in older adults continues to be debated because higher mortality rates have been reported for subgroups in some cohorts [40]. A meta-analysis of 21 randomized controlled trials (RCTs) that included approximately 91,000 individuals (average age 62 years) and 8,800 deaths found no significant effects of MVMs on all-cause mortality (relative risk [RR]: 0.98; 95 % confidence interval [CI] 0.94, 1.02) [41]. In the 13 primary prevention trials included in the analysis, MVMs had no effect on mortality due to cancer (RR: 0.96; 95 % CI: 0.88, 1.04) or vascular causes (RR: 1.01; 95 % CI: 0.93, 1.09).

MVMs: Cancer

In an analysis of cohort and case-control studies of MVMs and breast cancer, eight studies in more than 355,000 women who took MVMs for 3–10 years at least twice a week were studied [42]. Neither longer duration of use nor greater frequency of use was significantly associated with decreased risk of breast cancer in the meta-analysis of the pooled data from five cohort studies (RR: 0.10; 95 % CI: 0.60, 1.63). An analysis of data from 14 RCTs, case-control studies, and cohort studies in 230,000 men aged 40 and older also found no convincing evidence of effects of multivitamins on prostate cancer incidence, severity, or death [43]. A pooled analysis of 13 prospective cohort studies of the associations between intakes of vitamins A, C, and E (including intakes from multivitamins) and colon cancer risk included more than 676,000 adults and 5,400 colon cancer cases [44]. Intakes of the vitamins from food were not associated with colon cancer risk, even after adjustment for total folic acid intakes. However, multivitamin use was inversely associated with colon cancer risk (RR: 0.88; 95 % CI: 0.81, 0.96).

Two large RCTs have linked beta-carotene and vitamin A supplements to an increased risk of lung cancer in smokers [45]. Smokers and possibly former smokers should therefore avoid MVM products providing large amounts of these nutrients.

MVMs: Infection

Infections are major causes of morbidity among elderly people. A systematic review of 20 RCTs in more than 35,000 adults found no differences in the number of episodes of infection in people over 65 years who did or did not take MVM supplements (weighted mean difference [WMD]: 0.06; 95 % CI: -0.04, 0.16) [46]. However, the results did suggest that undernourished older people might benefit from taking supplements for at least 6 months (WMD: -0.67; 95 % CI: -1.24, -0.10).

MVMs: Cognition

A systematic review included ten RCTs that tested the effects of MVM supplements for at least 1 month on any valid cognitive outcome in 3,200 adults older than 18 with normal cognitive function [47]. MVMs enhanced immediate free-recall memory (standardized mean difference: -0.32; 95 % CI: 0.09, 0.56; p < 0.01) but had no effect on delayed free-recall memory or verbal fluency. An RCT testing daily MVMs for 12 months in 900 persons older than 65 years living in the community (450 in each treatment arm) also found no evidence of change in several domains of cognitive function, including digit span forward (an assessment of immediate memory) and verbal fluency (an assessment of executive function) [48].

MVMs: Conclusions

In summary, for some older persons whose dietary intakes are low or inadequate, taking MVMs containing approximate DV levels of the micronutrients that their diets do not provide in sufficient quantities might be helpful. The evidence is still weak on whether taking MVMs prevents chronic disease or decreases mortality risk among older persons. More high-quality studies are needed before MVM use can be recommended for these purposes.

Supplements Containing Calcium and Vitamin D

Many calcium-containing supplements with or without other vitamins and minerals are marketed to older people. In its 2011 report on the dietary reference intakes (DRIs) for calcium and vitamin D, the Food and Nutrition Board of the Institute of Medicine summarized the evidence for the association of vitamin D and calcium with various chronic degenerative disease outcomes [49]. The 2011 DRIs are based on the amounts of calcium and vitamin D required to decrease the risks of osteoporosis and fractures.

Calcium-containing supplements with and without vitamin D are extremely popular among older Americans, in part because many elders do not obtain enough of these nutrients from food alone [50]. The RDA is 1,200 mg for calcium and 800 IU (20 μ g) for vitamin D for women and men over age 71. According to an analysis of 2010 NHANES data, the prevalence of use of calcium supplements was 65 % among women aged 71 years or older and 56 % among men [50]. These intakes contributed a mean of 608 mg/day to intakes in women and 372 mg/day in men. The prevalence of use of vitamin D supplements in elders aged 71 years or older was 49 %, and these supplements contributed about 11 μ g/day. Total food and supplement intakes for both of these nutrients were lower than the adequate intake (AI) for this age group, but intakes in users of calcium and vitamin D supplements were closer to the AI than in nonusers.

Calcium and Vitamin D: Osteoporosis

More than 99 % of total body calcium is stored in bones and teeth to support their structure [51]. Consuming adequate amounts of calcium and vitamin D throughout infancy, childhood, and adolescence and engaging in weight-bearing exercise maximizes bone strength and bone density, which helps prevent osteoporosis and fractures later in life. In 1993, the FDA authorized a health claim for food labels about calcium and osteoporosis in response to scientific evidence indicating that an inadequate calcium intake contributes to low peak bone mass and is a risk factor for osteoporosis [52]. The claim states that "adequate calcium intake throughout life, as part of a well-balanced diet, may reduce the risk of osteoporosis." Manufacturers may also make health claims about calcium and osteoporosis for dietary supplements. An example of these claim statements is, "Adequate calcium and vitamin D, as part of a well-balanced diet, along with physical activity, may reduce the risk of osteoporosis."

More than 1.5 million fractures per year in the United States are linked to osteoporosis, and they generate high healthcare costs [53]. The U.S. Surgeon General's report on bone health and treatment noted the importance of calcium and vitamin D supplementation to reduce bone loss and fracture risk in some populations [54]. However, data on the efficacy of calcium supplements for preventing bone fractures in healthy postmenopausal women remain equivocal. An analysis of data from the Women's Health Initiative, which followed more than 36,000 postmenopausal women aged 50–79 years,
showed that, compared to placebo, daily supplementation with calcium carbonate (1,000 mg) and vitamin D₃ (400 IU) for up to 10 years resulted in a small but significant improvement in hip bone density (1.06 %; p < 0.01) but did not significantly reduce the number of hip fractures (hazard ratio [HR]: 0.88; 95 % CI: 0.72, 1.08) [4].

In its clinical practice guideline for managing osteoporosis in men, the Endocrine Society suggested testing men aged 70 and older and younger men who have many risk factors for osteoporosis [55]. The authors also recommended that men with osteoporosis or who were at risk for this condition consume 1,000–1,200 mg/day calcium and, in those with low vitamin D levels, vitamin D supplements.

Calcium and Vitamin D: Bone Mineral Density

Since the publication of the 2011 DRIs, additional information has become available that is relevant to older persons. One of the better surrogate measures of fracture risk is bone mineral density (BMD). A review of 32 RCTs in 3,200 postmenopausal women from Australia found that daily calcium supplements (700–2,000 mg/day) for at least a year slowed bone loss but only when supplement doses were higher than 700 mg/day [56]. The effects became weaker after 4 years of treatment. Doses higher than 1,000 mg had no significant beneficial effects compared to doses of 1,000 mg/day. However, a review of 15 RCTs of calcium supplementation (750–1,200 mg/day) with or without vitamin D (400–800 IU/day) in more than 47,000 mostly postmenopausal women showed that greater increases in BMD at the lumbar spine were not necessarily associated with larger reductions in fracture risk [57].

Calcium and Vitamin D: Fractures

A 2011 meta-analysis that included 11 RCTs found that combined supplementation with vitamin D (300-1,100 IU/day) and calcium (500-1,200 mg/day) reduced fracture risk (pooled RR: 0.88; 95 % CI: 0.78, 0.99) in 53,000 adults, of which 69 % were postmenopausal women from the Women's Health Initiative [12]. However, the effects differed considerably by the setting, being greatest in institutional settings (RR: 0.71; 95 % CI: 0.57, 0.88) and somewhat lower among community-dwelling elders (RR: 0.89; 95 % CI: 0.76, 1.04), probably due to adherence differences among settings. Older women occasionally experienced adverse events, including increased risk of kidney and urinary tract stones.

Calcium and Vitamin D: Mortality and Other Outcomes

A recent study in Canada examined the association between total calcium and vitamin D intakes and mortality in a population-based longitudinal cohort of 9,033 men and women who were at least 25 years old [58]. The associations between higher calcium intakes and lower mortality risk were inconclusive for both men and women. Use of calcium supplements was associated with reduced mortality risk (HR: 0.78; 95 % CI: 0.66, 0.92) in female supplement users compared to nonusers but not in male users.

In summary, for older persons whose dietary intakes from food and fortified foods are low, vitamin D supplements, either as vitamin D₂ (ergocalciferol) or D₃ (cholecalciferol) may help to achieve intake levels approximating the RDA of 600 IU (15 μ g) per day for those 51–70 years, and 800 IU (20 μ g) per day for those greater than 70 years. Usual intakes over 4,000 IU (100 μ g) per day from any source should be avoided because of risks of toxicity.

Some physicians use a blood measure of vitamin D status, specifically serum levels of 25-hydroxyvitamin D, and base their recommendations on this measure. Levels are stated either in nanomoles per liter (nmol/L) or nanograms per milliliter (ng/mL), where 1 nmol/L=0.4 ng/mL. In general, levels below 30 nmol/L (12 ng/mL) are too low for bone or overall health, and when these are present vitamin D supplements may be prescribed. Serum levels of 50 nmol/L (20 ng/mL) are sufficient for most of the population. There is no known benefit of levels above about 50 nmol/L (20 ng/mL) for bone or other health outcomes. Serum levels above 125 nmol/L (50 ng/mL) are probably too high and should be avoided since emerging evidence links potential adverse effect to such high levels, particularly when they exceed 150 nmol/L (60 ng/mL).

Older women 51–70 years of age should aim to consume at least 1,200 mg calcium per day and males at least 1,000 mg/day. Individuals of both sexes who are more than 70 years of age should consume at least 1,200 mg/day from all sources. Dietary supplements may be helpful for those whose intakes of food and fortified food do not reach these levels. More studies are needed on the effects of very large amounts of calcium; the current UL should not be exceeded. It is also important to ensure that vitamin D intakes are adequate, and that usual intakes of 2,000 mg/day or more be avoided because of possible adverse effects such as kidney stones from excessive calcium.

Combinations of Vitamins, Minerals, and Other Nutrients

Prostate Cancer

Blood levels of prostate-specific antigen (PSA) are often elevated in men with prostate cancer. PSA is also used as a marker of cancer progression because increases in PSA after treatment sometimes indicate that the cancer has returned. Some supplements might decrease PSA levels, but there is no evidence that they decrease cancer progression. A 2013 systematic review of RCTs on the use of nonherbal dietary supplements and vitamins found that in two RCTs, a combination of dietary supplements (including isoflavones, lycopene, minerals, phytoestrogens, and vitamins) significantly decreased PSA levels compared to placebo [59]. However, six other studies found that supplements with various ingredients had no significant effects on PSA. Therefore, use of supplements such as these cannot be recommended as effective treatments for patients with prostate cancer.

Eye Health

Supplements containing combinations of vitamins, minerals, and other nutrients are widely marketed to the elderly for "eye health," but there is little evidence that dietary supplements can prevent or treat the eye diseases that many elderly people experience. The Age-Related Eye Disease Study (AREDS) examined the clinical course of age-related macular degeneration (AMD) and lens opacities and included an RCT to test the use of dietary supplements for 5 years to reduce the progression of these diseases in 3,540 participants aged 55–80 years [60]. The supplements contained the antioxidants vitamin C (500 mg), vitamin E (400 IU), and beta carotene (15 mg); zinc (80 mg as zinc oxide) and copper (2 mg as cupric oxide); antioxidants plus zinc; or placebo. Participants who were at high risk of developing advanced AMD, had intermediate AMD, or had advanced AMD in one eye only at baseline benefited from the supplements. Among these participants, those taking antioxidants plus zinc had the lowest risk (20 %, compared to 22 % for those taking zinc only, 23 % for those taking antioxidants only, and 28 % for those taking placebo) of advanced-stage AMD and its accompanying vision loss at 5 years. However, these dietary supplements had no significant effect on development or progression of age-related cataracts. The AREDS formula was the first evidence-based treatment for people at high risk of AMD.

The Blue Mountains Eye Study was an Australian population-based cohort study of 3,654 participants aged 49 years or older at baseline who were followed for 10 years [61]. This study found that higher zinc intakes are associated with a lower risk of AMD (RR for top decile of intake compared to the rest of the population: 0.56; 95 % CI: 0.32, 0.97). However, unlike AREDS, the study did not show that a combination of high doses of zinc, beta-carotene, and vitamins C and E prevented advanced-stage AMD.

Preliminary animal studies suggested that omega-3 fatty acids or other antioxidants might prevent cellular damage in the retina by reacting with free radicals produced in the process of light absorption. Therefore, experts hoped that supplementation with omega-3 fatty acids or other antioxidants might prevent or slow the progression of AMD. AREDS 2 was a multicenter, Phase III RCT of 4,200 participants aged 50–85 years who were at risk of advanced AMD to assess the effects of oral supplementation with lutein (10 mg/day), zeaxanthin (2 mg/day), and the omega-3 long-chain polyunsaturated fatty acids docosahexaenoic acid (DHA, 350 mg) and eicosapentaenoic acid (EPA, 650 mg/day) on the risk of progression to advanced AMD [62]. Adding lutein and zeaxanthin, DHA and EPA, or all of these ingredients to the AREDS formula did not reduce the risk of advanced AMD more than the original AREDS formula.

Some more recent trials of combinations of vitamins to treat or prevent AMD have also had negative results. A 2012 Cochrane review of antioxidant vitamin and mineral supplements for slowing AMD progression that included four RCTs in 62,520 healthy people found that vitamin E and betacarotene supplements alone or in combination did not decrease the incidence of acute AMD (pooled RR: 0.98; 95 % CI: 0.89, 1.08) or advanced AMD (RR: 1.05; 95 % CI: 0.80, 1.39) [63]. A systematic review and meta-analysis of six longitudinal studies found that dietary intakes of lutein and zeaxanthin from foods had little association with risk of early-stage AMD but were associated with a reduced risk of late-stage AMD (RR: 0.74; 95 % CI: 0.57, 0.97) and of neovascular AMD (RR: 0.68; 95 % CI: 0.51, 0.92) [64]. More research is needed on the use of lutein and perhaps other carotenoids to prevent late-stage AMD.

Vitamins

B-Complex Vitamins

Manufacturers often market supplements containing B-complex vitamins with claims that these supplements improve cardiovascular health, prevent cancer, improve cognitive function, and decrease mortality risk.

In the United States, B-vitamin deficiency is rarely caused by a low dietary intake but, rather, by diseases, such as pernicious anemia or atrophic gastritis, bariatric surgery, or other gastrointestinal factors that affect absorption. These illnesses are more common with aging. Furthermore, proton pump inhibitors are commonly used by elderly people, and these medications also decrease gastric acidity and thereby lower vitamin B_{12} absorption from food. The Institute of Medicine recommends that people older than 50 take vitamin B_{12} in unbound form either in fortified foods or as oral supplements in DV amounts. Such oral supplements are efficacious for treating vitamin B_{12} deficiency and can prevent the need for B_{12} injections to prevent deficiency in healthy elders [65, 66].

Folic Acid: Mortality

A meta-analysis of cause-specific mortality in eight RCTs included more than 37,000 individuals at increased risk of cardiovascular disease (CVD) who were treated with folic acid supplements for a

median of 5 years [67]. Folic acid supplements (0.8–40.0 mg/day) had no effects on cancer mortality (rate ratio: 1.00; 95 % CI: 0.85, 1.18) or all-cause mortality (rate ratio: 1.02; 95 % CI: 0.97, 1.08) during the treatment period or in the years following treatment.

Folic Acid: CVD

There is contradictory information about whether supplements containing B-vitamins can reduce the risk of CVD by lowering levels of homocysteine, an amino acid that might act as a vascular toxin when present at high levels [68–72]. For example, in the meta-analysis of cause-specific mortality in eight RCTs discussed above, treatment with 0.8–40.0 mg/day folic acid had no effect on vascular outcomes, including major vascular events (rate ratio: 1.01; 95 % CI: 0.97, 1.05), major coronary events (rate ratio: 1.03; 95 % CI: 0.97, 1.10), and stroke (rate ratio: 0.96; 95 % CI: 0.87, 1.06), even though supplementation reduced homocysteine levels by 25 % [67]. A systematic review and metaanalyses of 16 RCTs also found that although folic acid supplementation (0.8-40 mg/day) could lower homocysteine levels, the supplements had no effects on rates of major cardiovascular events, myocardial infarction, all-cause mortality, or stroke [73]. In another systematic review, folic acid supplementation did not affect the risks of primary cardiovascular clinical endpoints (RR: 1.02; 95 % CI: 0.93, 1.13) or stroke (RR: 0.95; 95 % CI: 0.84, 1.08) [74]. However, a meta-analysis that included ten RCTs involving more than 43,000 individuals living in areas with little or no folic acid fortification of the food supply found that folic acid supplementation (0.4-0.8 mg/day) reduced stroke risk by 11 % (RR: 0.89; 95 % CI: 0.82, 0.97); the effects were even greater in populations with lower use of statins [75].

Although supplementation might have had a small effect on CVD risk, this effect was unlikely to be dramatic, especially in countries (such as the United States) where folic acid fortification of flour and use of MVM supplements containing folic acid are common and blood levels of folic acid are already high. Also in such countries, use of statins is very common, making it difficult to detect any small additional effects of folic acid.

In summary, a healthy diet and, if needed, other measures are probably more appropriate to prevent CVD and hypercholesterolemia in the United States than supplementation with folic acid or other B-vitamins.

Folic Acid: Possible Risks of Excess Amounts

The relationships between diet, dietary supplements, and health are complex. There are concerns about the increased folic acid blood levels resulting from folic acid fortification of enriched, uncooked cereal grains in the United States and Canada since 1998 and the increased use of folic acid-containing dietary supplements. Data from NHANES show that in 2003–2006, total folate and folic acid intakes from foods and supplements were highest in those aged 50 years or older, and about 5 % of those aged 50–71 years had total intakes exceeding the UL's [76]. Folic acid fortification has achieved its goal of reducing the number of births complicated by neural tube defects. However, whether the hoped-for cardiovascular benefits have occurred is unclear, and concerns have been raised about cancer risk.

Folate is important for DNA synthesis, is a known cofactor in one-carbon transfer reactions, and might modulate colorectal cancer development [77]. Epidemiologic studies have suggested an inverse relationship between folate intake and incidence of colorectal cancer [78]. However, animal studies have suggested that the timing and dose of folic acid supplementation might influence its effect on development or inhibition of colorectal cancer [77]. When folate fortification began in the United States and Canada in the mid-1990s, the downward trend in colorectal cancer incidence was reversed before it began to decline again a few years later, causing some researchers to speculate that the

phenomenon had a cause-and-effect association [79]; this association, however, has not been confirmed. In normal epithelial tissue, folate deficiency might promote neoplastic transformation, whereas moderate folate supplementation might suppress tumor development [77]. But the opposite could be true in established tumors or microscopic neoplasms. Animal studies suggest that folate deficiency might inhibit, and supplementation might promote, progression of established colorectal neoplasms.

The Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial, which collected observational data on 25,400 women aged 55–74 years at baseline, showed that high folate intakes could increase the risk of breast cancer [80]. Women who ingested at least 400 μ g/day of supplemental folic acid had a 19 % greater risk of developing breast cancer than those who did not consume supplemental folic acid (95 % CI: 1.01, 1.41). Participants in the highest folate quintile for total folate intakes had a 32 % greater risk of breast cancer than those in the lowest quintile (95 % CI: 1.04, 1.68). These data suggest that folate has dual modulatory effects on cancer development. Although the data are not definitive, they do warrant continued population-based monitoring of free folic acid levels in blood, as the NHANES survey does.

B-Complex Vitamins: Cognition

B-complex vitamins are advertised as promoting "brain health" and cognition. However, many confounding factors are associated with cognitive decline in later life. These factors include current tobacco use, presence of certain genotypes (e.g., apolipoprotein E epsilon-4 genotype), multi-infarct dementia, and some medical conditions. However, a systematic review of 127 observational studies, 22 RCTs, and 16 systematic reviews found little clear evidence that nutritional factors play a role in maintaining cognitive function [81].

Vitamin B_{12} deficiency has well-known adverse effects on cognitive function, including difficulties with concentration, disorientation, and dementia [82]. Folic acid supplementation can mask the presence of vitamin B_{12} deficiency and, by delaying treatment for this deficiency, can allow progression of the associated neurological symptoms, including mood and cognitive disorders [83]. For this reason, there is concern about high-dose folic acid supplements in elders who might be vitamin B_{12} -deficient.

Folic acid supplements decrease homocysteine levels somewhat, and some evidence indicates that high homocysteine levels are associated with an increased risk of dementia and Alzheimer's disease [83]. But there is little evidence that the other B-vitamins affect cognition or that folic acid supplements with or without vitamin B_{12} are useful in preventing dementia in healthy elders or treating those suffering from dementia. For example, an analysis of data from the Framingham study on 1,092 participants (mean age 76 years) without dementia found that those who had homocysteine levels greater than 14 mmol/L had almost twice the risk of developing Alzheimer disease 8 years later as people with lower homocysteine levels (95 % CI: 1.3, 2.5); for each 5 mmol/L elevation, risk their increased by 40 % (95 % CI: 1.1, 1.9) [84]. An observational cohort study in 816 dementia-free adults (mean age 74 years) assessed rates of newly diagnosed dementia and Alzheimer disease [85]. After 4 years, the risk of dementia and Alzheimer disease was higher in those who had baseline levels of homocysteine higher than 15 µmol/L (HR for Alzheimer disease: 2.11; 95 % CI: 1.19, 3.76; HR for dementia: 2.08; 95 % CI: 1.31, 3.30) and those who had folate levels of 11.8 mmol/L or lower (HR for Alzheimer disease: 1.98; 95 % CI: 1.15, 3.40; HR for dementia: 1.87; 95 % CI: 1.21, 2.89). However, whether hyper-homocysteinemia was a risk marker or risk factor for dementia was not clear.

If a high level of homocysteine is a risk factor, reducing this level by increasing intake of folic acid or vitamin B_{12} might decrease the risk of dementia. However, if hyper-homocystinemia is a marker for dementia, lowering homocysteine levels is not likely to affect risk of this disease. One RCT found that supplementation with vitamin B_{12} (1,000 µg/day) with or without folic acid (400 mg/day) for 24 weeks in 195 elders with mild vitamin B_{12} deficiency had no effect on cognitive function [86].

Another RCT compared the use of folic acid supplements (800 µg/day) to placebo for 2 years in 818 participants aged 50–70 years with high plasma homocysteine and normal serum vitamin B₁₂ levels [87]. The supplements improved global functioning (weighted mean difference [WMD]: 0.05; 95 % CI: 0.004, 0.096; p=0.033), memory (WMD: 0.132; 95 % CI: 0.032, 0.233; p=0.06), and information-processing speed (WMD: 0.087; 95 % CI: 0.016, 0.158; p=0.016). In both studies, the supplements decreased homocysteine levels.

The authors of a 2009 Cochrane review of the use of folic acid supplements (400 μ g to 15 mg/day) with or without vitamin B_{12} (1,000 µg to 1 mg/day) for preventing or treating dementia concluded, based on eight RCTs, that the evidence showing that such supplements can prevent declines in cognitive function or mood in healthy elderly people is inadequate [88]. Of the four trials reviewed that tested supplementation in people with cognitive impairment, one found that folic acid supplements (1 mg/day) significantly improved responses to cholinesterase inhibitors, typically given to patients with Alzheimer's disease (odds ratio: 4.06; 95 % CI: 1.22, 13.53; p=0.02), and scores of instrumental activities of daily living and social behavior measured by a subscale of the Nurse's Observation Scale for Geriatric patients. The other three trials found no benefit of folic acid supplements with or without vitamin B_{12} on cognitive function. A 2012 meta-analysis of 19 RCTs also found that supplements containing vitamin B_{12} (0.5–1 mg/day), vitamin B_6 (25 mg/day), and folic acid (0.8–15 mg/day) alone or in combination for 60 days to 24 months did not improve cognitive function in 5,398 people aged 50 years or older who were healthy or who had mild cognitive impairments [89]. The results did not seem to vary by folic acid status or by study duration or size. A systematic review of 22 RCTs that tested the effects of supplements with at least one kind of vitamin, mineral, or omega-3 fatty acid on various cognitive tests in 3,400 persons aged 65 or older concluded that B-vitamin supplements had no significant effects on global cognitive functions [90]. The authors found no significant effects of B-vitamins on global cognitive function.

Vitamin D

The effects of calcium and vitamin D on bone and on mortality are discussed above. The evidence on the effects of vitamin D supplements on health issues that are particularly relevant to older people is reviewed in this section.

A 2007 Agency for Healthcare Research and Quality (AHRQ) review included 167 studies (112 RCTs, 19 prospective cohort studies, 30 case-control studies, and 6 before-and-after studies) on the association between circulating 25-hydroxyvitamin D [25(OH)D] concentrations and bone health outcomes [91]. The authors found that in 22 RCTs whose investigators reported whether vitamin D supplements had toxic side effects (including 17 RCTs in older adults), intakes of vitamin D that were higher than current DRIs were generally well tolerated, although the most relevant trials were not adequately designed to assess the long-term risks of adverse events. Although some studies found that high intakes of vitamin D were associated with an increased risk of hypercalcemia and hypercalciuria, these effects were not clinically significant.

Vitamin D: Bone Health

The 2007 ARHQ review [91] and a 2006 National Institutes of Health (NIH) conference entitled "Vitamin D and Health in the 21st Century: An Update" [87] revealed that the effects of vitamin D supplements alone, independent of calcium intake, could not be evaluated in most of the trials conducted to date. In most trials reviewed in the AHRQ report, daily supplementation with 700 IU vitamin D_3 and calcium compared to placebo had a small beneficial effect on BMD and reduced the risk of fractures and falls, although the benefit was sometimes confined to specific subgroups [92].

The authors of the AHRQ report also noted that determining which blood levels of markers for vitamin D status are optimal for bone health is difficult. One challenge is that current methods, which measure serum 25(OH)D levels, have yielded highly inconsistent results in many studies, in part because of the lack of commercially available standard reference preparations for calibrating measurements between laboratories.

A 2011 meta-analysis of 13 RCTs of vitamin D supplementation without exercise interventions in adults aged 60 years and older found that supplements decreased postural sway (standardized mean difference: -0.20; 95 % CI: -0.39, -0.01), decreased time to complete the Timed Up and Go Test (-0.19; 95 % CI, -0.11, 0.20), and improved lower extremity strength (0.05; 95 % CI: 0.11, 0.20), but had no effect on gait [93]. All of the studies in which daily vitamin D doses were 800 IU or more showed that the supplements improved balance and muscle strength.

A systematic review and meta-analysis of 26 RCTs that enrolled more than 45,700 participants, most of whom were elderly women, who had a median risk of falling of 50 %, found that the use of supplements containing vitamin D (200–1,100 IU/day, 10,000 IU weekly, 100,000 IU every 10 weeks, or 100,000–600,000 IU once) reduced the risk of falls (odds ratio [OR] of at least one fall: 0.86; 95 % CI: 0.77, 0.96) [94]. The effect was especially striking in patients who had lower levels of 25(OH)D at baseline and those who took both calcium and vitamin D supplements.

Several other reviews have found that vitamin D supplements can reduce the risk of falling in older adults. For example, a systematic review and meta-analysis of ten RCTs found that vitamin D supplementation (200–1,000 IU) for 1–36 months reduced the number of falls by 14 % (RR: 0.86, 95 % CI: 0.79, 0.93) compared to calcium supplements alone (200–1,200 mg/day) or placebo [95]. The effects were statistically significant, especially for those who lived in the community, were younger than 80 years, took adjunctive calcium supplements, had no history of fractures of falls, used the supplements for longer than 6 months, or received a dose of 800 IU vitamin D or more. A Cochrane review of 60 RCTs that tested interventions to reduce falls in care facilities and hospitals found, based on five trials in 4,603 participants, that vitamin D supplementation (200–1,100 IU/day or 10,000 IU weekly) for 3–5 months reduced the rate of falls (rate ratio: 0.63; 95 % CI: 0.46, 0.86) but not the risk of falling (rate ratio: 0.99; 95 % CI 0.90, 1.08) [96].

However, other studies have not found vitamin D supplementation to be effective for preventing falls. A Cochrane review of trials of interventions to prevent falls in community-dwelling older people found that vitamin D supplementation (200–2,000 IU/day or 300,000 IU once) did not appear to reduce the number of falls (rate ratio: 1.00, 95 % CI: 0.90, 1.11 based on 7 trials in more than 9,300 participants) or the risk of falls (rate ratio: 0.96; 95 % CI: 0.89, 1.03 based on 13 trials in more than 26,700 participants), except possibly among individuals who had lower intakes of vitamin D before treatment. Supplements did not reduce the risk of falling (rate ratio: 0.96; 95 % CI: 0.89, 1.03) [97].

Some evidence indicates that vitamin D supplementation can improve muscle strength. For example, in a systematic review of 17 RCTs involving 5,072 participants (who were mostly older than 60 years), supplementation with vitamin D (400–1,000 IU/day vitamin D₃, 60,000 IU/week vitamin D₃, 150,000 IU/month vitamin D₃, or 300,000 IU once vitamin D₃, 2,000–9,000 IU/day vitamin D₂, or 600,000 IU once vitamin D₂) did not have a significant effect on muscle strength (as measured by grip strength or proximal lower limb strength) in adults whose baseline levels of 25(OH)D were higher than 25 nmol/L, but supplements in people with lower levels of 25(OH)D improved hip muscle strength [98].

The evidence on the use of vitamin D supplements to reduce the risk of fractures is mixed. A 2012 pooled analysis included participant-level data from 11 RCTs that compared oral vitamin D supplements (199–846 IU/day) with or without calcium (84–830 mg/day) to calcium alone or placebo for preventing fractures in 31,022 mostly female adults aged 65 years or older [99]. Virtually all of the women who were randomly assigned to receive vitamin D compared to those assigned to control groups experienced a nonsignificant (10 %) reduction in risk of hip fracture (HR: 0.90; 95 % CI: 0.80, 1.01) and a 7 % lower risk of nonvertebral fracture (HR: 0.93; 95 % CI: 0.87, 0.99). Risks of fracture

were only evident at the highest levels of intake (equivalent to a median of 800 IU/day). People with higher baseline D levels might not require such high doses of supplements to prevent fractures. The optimal intake level probably depends on the individual's baseline nutritional vitamin D status, which varies greatly from country to country.

Vitamin D: Cancer

A meta-analysis of studies that evaluated the use of vitamin D supplements with or without calcium to prevent cancer or fractures included three RCTs lasting 4–7 years in healthy postmenopausal women or in men and women aged 71 years or older and 28 observational studies focused on cancer prevention [12]. Limited data from the RCTs suggested that 1,000 IU/day vitamin D supplements might reduce the risk of cancer, but the data from the observational studies suggested that higher blood 25(OH)D concentrations might be associated with an increased cancer risk. A mixed-effects, dose-response meta-analysis in this analysis showed that each 10 nmol/L increase in blood 25(OH)D concentrations was associated with a 6 % reduced risk of colorectal cancer (95 % CI: 0.03, 0.09), but no significant dose-response relationship was evident for breast or prostate cancer. A meta-analysis of prospective studies that evaluated the association between vitamin D intakes from foods and supplements in 6,466 individuals (nine studies) or blood levels of 25(OH)D in 2,767 cases and 3,948 controls and colorectal cancer risk found an inverse association between vitamin D intake (pooled RR: 0.88; 95 % CI: 0.80, 0.96) and blood 25(OH)D levels (pooled RR: 0.67; 95 % CI: 0.54, 0.80) and risk of colorectal cancers [100].

Vitamin D: Cognition and Dementia

A review of 37 studies (21 cross-sectional studies, 10 case-control studies, 1 before-and-after study with a comparison group, 2 prospective cohort studies, and 3 RCTs) that evaluated the association of vitamin D concentrations with cognition and dementia in adults older than 18 years found that lower vitamin D concentrations were associated with poorer cognitive function and a higher risk of Alzheimer disease [101]. However, the authors suggested that the results might have been due to association rather than causation because most of the studies were prospective. In both a systematic review and meta-analysis of two longitudinal and five cross-sectional studies on the association between vitamin D deficiency and cognitive impairment in 7,688 elderly individuals, people with low vitamin D levels had an increased risk of cognitive impairment compared to those with normal vitamin D levels [102]. However, whether vitamin D deficiency caused the impairment could not be determined.

Vitamin D: Other Chronic Diseases

Many other associations between vitamin D levels and chronic diseases have been suggested but not proven. For example, in a systematic review and meta-analysis of 51 RCTs that assessed the associations between vitamin D supplements (200–3,332 IU/day, 8–10 µg/day, 10,000–100,000 IU weekly, 50,000 IU twice weekly, 12,000 IU every 2 weeks, 100,000 IU every 4 months, or single injections of 100,000–300,000 IU) with or without calcium (500–1,000 mg/day) and/or other nutrients and cardio-vascular outcomes in adults, the supplements had no significant effects on risk of death (RR: 0.96; 95 % CI: 0.93, 1.00; p=0.08), myocardial infarction (RR: 10.02; 95 % CI: 0.93, 1.13; p=0.64), or stroke (RR: 1.05; 95 % CI: 0.88, 1.25; p=0.59) [103]. A meta-analysis of 12 RCTs that assessed the effects of vitamin supplements on blood lipid levels in 1,346 individuals found that vitamin D

supplements (800 IU/day or 100,000–200,000 IU once) had small effects (e.g., 1–6 mg/dL) on lowdensity lipoprotein (LDL) cholesterol levels [104]. A systematic review and meta-analysis of 15 RCTs that assessed the effects of vitamin D supplements on glycemic control and insulin resistance in adults who were healthy or had type 2 diabetes or insulin resistance found that supplements had no significant effects on fasting glucose, hemoglobin A1c levels, or insulin resistance, and, in patients with diabetes or impaired glucose tolerance, only small effects on fasting glucose (-0.32 mmol/L; 95 % CI: -0.57, -0.07) and insulin resistance (standard mean difference [SMD]: -0.25; 95 % CI: -0.48, -0.03) [105]. The effects of vitamin D supplements (200–600 IU/day) on systolic blood pressure were also quite small (2.44 mmHg; WMD: -2.44; 95 % CI: -4.86, -0.02) in a review of four RCTs in 429 adults (mean age 64 years) who had normal or high blood pressure [106]. A systematic review and metaanalysis of eight RCTs in hypertensive patients in western Europe found that although vitamin D supplements or ultraviolet radiation reduced systolic blood pressure compared to placebo, the results were not significant (-3.6 mmHg; 95 % CI: -9.0, 0.7) [107].

Vitamin E

Vitamin E is a popular vitamin that elders often taken in very large doses in the hope that it will improve their cardiovascular health. Various forms of vitamin E are available in supplements; alpha-tocopherol is considered the most active form. Unlike other vitamins, the form of alpha-tocopherol made in the laboratory and found in many supplements (all Rac-alpha-tocopherol) is not identical to the natural form (RRR-alpha-tocopherol) and is less active than the natural form. The vitamin E RDA of 15 mg equals 22 IU of natural and 33 IU of synthetic vitamin E. Studies suggest that daily doses of vitamin E ranging from 200 to 1,200 IU are safe when taken by healthy adults for up to 4 months [108–110].

Vitamin E: Cardiovascular Health

Reactive oxygen species promote atherogenesis. As an antioxidant with inflammatory properties, vitamin E might mediate free radical reactions that oxidize LDL cholesterol and initiate or promote atherosclerosis [111, 112].

The Women's Health Study indicated that taking 600 IU vitamin E every other day for approximately 10 years did not decrease the risk of major cardiovascular events in healthy women aged 45 or older (RR: 0.93; 95 % CI: 0.82, 1.05; p=0.26), but the supplements were associated with a 24 % reduction in the incidence of cardiovascular deaths (primarily sudden death in women aged 65 years or older; RR: 0.76; 95 % CI: 0.59, 0.98; p=0.03) [28]. The Heart Outcomes Prevention Evaluation (HOPE) trial, which examined the effect of vitamin E supplementation on major cardiovascular events in 9,541 people with vascular disease or diabetes aged 55 years or older, demonstrated that vitamin E supplementation (400 IU/day for 4.5 years, on average) did not reduce the risk of a composite measure of myocardial infarction, stroke, and cardiovascular death (RR, 1.05; 95 % CI: 0.95, 1.16; p=0.33) and was not associated with any adverse effects [113]. The results of a 7-year passive followup of the HOPE trial in 738 participants suggested that vitamin E supplementation failed to prevent major cardiovascular events (RR: 1.04; 95 % CI: 0.96, 1.14; p=0.34) [27].

The results of clinical trials have not supported a role for vitamin E in either the primary or secondary prevention of heart disease. The weakness of evidence supporting a treatment role for vitamin E for CVD led the FDA to reject health claims for vitamin E. It might be too optimistic to expect a single-vitamin supplement to overcome the effects of poor dietary habits and a sedentary lifestyle on known risk factors, such as hypertension and hypercholesterolemia, for heart disease [112]. Additional studies are needed to further explore the benefit of vitamin E in treating and/or preventing heart disease. Older persons at high risk of developing CVD or who have CVD should discuss the benefits of taking a supplement containing vitamin E at a dose that not exceed the UL of 1,000 IU/day with their cardiologist or primary physician. This amount of vitamin E is not available in most multivitamin preparations and usually requires a vitamin E supplement.

Vitamin E: Cancer Prevention

The evidence on the benefits of vitamin E supplements in cancer prevention is also uncertain, at best. Vitamin E supplements (600 IU every other day for 10 years) did not prevent cancer (RR: 1.01; 95 % CI: 0.94, 1.08; p=0.87) in the Women's Health Study, an RCT of 39,876 healthy female health professionals (median age 54.6 years) [28]. However, the follow-up phase of the Alpha-Tocopherol Beta-Carotene study, which randomly assigned 29,133 50-69-year-old male smokers from Finland to 50 mg alpha-tocopherol, 20 mg beta-carotene, both, or a placebo for 5–8 years, demonstrated a 32 %decrease in incidence (95 % CI: -0.47, -0.12) and a 41 % decrease in mortality (95 % CI: -0.65, -0.01) from prostate cancer among participants who received alpha-tocopherol supplements [114]. Six years after the trial ended, follow-up data continued to indicate that alpha-tocopherol supplements were associated with a lower risk of prostate cancer (RR: 0.88; 95 % CI: 0.76, 1.03) [115]. Interestingly, an analysis of 19 years of follow-up data on intakes found that men in the higher quintiles of baseline serum alpha-tocopherol levels had a lower risk of death due to cancer (RR: 0.79; 95 % CI: 0.72, 0.86) than those in the lowest quintile, suggesting that higher circulating concentrations of alpha-tocopherol within the normal range had health benefits [116]. The authors of a pooled analysis of eight prospective studies of intakes from North America and Europe examining correlations between intakes of vitamin antioxidants, multivitamins, and lung cancer concluded that regardless of smoking habits and lung cancer cell type, the available data do not support an association between intakes of vitamin E and lung cancer risk (pooled multivariate RR: 0.86; 95 % CI: 0.76, 0.99) [117].

One of the largest recent studies of vitamin E and cancer was the Selenium and Vitamin E Cancer Prevention Trial (SELECT), which included 35,533 men with a PSA level of up to 4.0 ng/mL who were aged 50 or older for black men and 55 or older for other men in the United States, Canada, and Puerto Rico [118]. The men were randomly assigned to supplementation with vitamin E (400 IU/ day), selenium (200 µg/day), both, or placebo. After 7–12 years, not only did vitamin E supplements not reduce prostate cancer incidence, but they were associated with an increased risk of prostate cancer compared to placebo (HR for vitamin E supplements alone: 1.17; 99 % CI: 1.004, 1.36; p=0.008).

At present, there does not appear to be evidence that vitamin E supplements can help prevent prostate or other cancers. In their recommendations regarding nutrition and cancer, the American Cancer Society [119] and World Cancer Research Federation/American Institute of Cancer Research [39] do not recommend that older people take vitamin E supplements. At this time, there is insufficient evidence to recommend vitamin E supplements for older adults.

Minerals

Potassium

In the 2010 Dietary Guidelines for Americans, potassium is described as a nutrient of public health concern because very few Americans consume the AI amount of 4,700 mg/day [120].

Potassium: CVD

The Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure notes that diets that are rich in potassium can reduce blood pressure, and the American Heart Association recommends increasing potassium intakes to 4,700 mg/day to lower blood pressure [121]. Combining an overall healthy diet with weight loss, lower salt intake, and higher potassium intake is thought to prevent and treat hypertension on the basis of some studies [122]. Healthy people who eat eight to ten servings of fruits and vegetables each day can achieve an intake of 4,700 mg/day, but this intake might be too high for those with impaired kidney function or congestive heart failure and not in line with many peoples' food choices. A diet containing 75 mEq (i.e., approximately 3,500 mg of elemental potassium) daily might help reduce the risk of stroke. However, potassium supplements are neither necessary nor recommended for the general population. Potassium supplements at high doses are not sold over the counter, and those containing more than the small doses included in a MVM preparation should only be taken under the guidance and instruction of a physician.

Selenium

Selenium supplements are popular among older men who take them to prevent prostate cancer. Selenium is available in supplements as selenomethionine and as yeast grown in a selenium-rich medium (selenized yeast). This yeast provides up to $1,000-2,000 \mu g/day$ of selenium, the vast majority of which is in the form of selenomethionine. The RDA for selenium is only 55 $\mu g/day$, and mean intakes from foods and supplements appear to be above that level [123, 124]. Older adults whose dietary selenium intake is inadequate might benefit from MVM supplements providing 100 % of the RDA for selenium. The UL for selenium is 400 $\mu g/day$.

Selenium: Cancer

The FDA approved a qualified health claim recognizing the anticarcinogenic potential of selenium [125]. However, selenium supplements provide no anticarcinogenic benefit in people whose diets provide adequate amounts of selenium, and people whose total intakes from food and supplements are excessive could be harmed by taking selenium supplements. Older adults who are unable or unwilling to increase their dietary selenium intake to RDA levels might benefit from an MVM supplement providing 100 % of the RDA/DV for selenium.

Selenium is incorporated in approximately 25 selenoproteins. Two selenoproteins, thioredoxin reductase and glutathione peroxidase, are antioxidant enzymes that provide protection from free radicals. Researchers have hypothesized that selenium compounds and selenoproteins might prevent cancer by modulating free radical metabolism, among other biologic mechanisms [126]. Furthermore, the incidence of some cancers has been lower among people with higher blood levels or intakes of selenium in some studies [127, 128].

In the early 2000s, there was great hope that selenium and vitamin E either alone or in combination might decrease the incidence of prostate cancer. Experts believed that intakes higher than the RDA level might be needed to maximize the selenium content of all selenium-containing proteins based primarily on the known antioxidant activity of selenium and vitamin E, secondary analyses of large-scale chemoprevention trials, and human observational studies showing that these compounds might decrease cancer risk. SELECT, described in the "Vitamin E" section above, was designed to test this hypothesis [118]. Unfortunately, after 7 years, supplements containing selenium alone (200 μ g/day) or selenium and vitamin E (400 IU/day) were not associated with reduced prostate cancer incidence in high-risk men [129]. In retrospect, although the supplement doses, supplement formulations, study

design, or cohort characteristics might have been faulty, the most likely explanation of the SELECT results is that any anticarcinogenic effects of selenium supplements might be confined to certain subpopulations of men [130]. The challenge is identifying these subpopulations, including their baseline selenium status, age, and perhaps genotype of specific selenoproteins to maximize results.

At present, there is little support for taking high-dose selenium supplements for prostate health, and doing so might have some adverse effects.

Selenium: Excessive Amounts

The threshold between adequate and excessive selenium amounts is small, and NHANES III crosssectional data suggest that high levels of serum selenium are positively associated with diabetes risk [131]. In addition, two secondary analyses of data from the Nutritional Prevention of Cancer trial found that supplementation for an average of 7 years with 200 μ g/day of selenium did not prevent CVD [132] and might have increased the risk of type 2 diabetes mellitus [133].

Omega-3 and -6 Fatty Acids

Should older Americans take omega-3 fatty acid supplements in addition to eating a healthful diet? Marketers claim that these supplements improve cardiovascular health, protect people from vision loss, improve cognition, and prevent cancers. The polyunsaturated fatty acids include alpha-linolenic acid (ALA), an omega-3 fatty acid, and linoleic acid, an omega-6 fatty acid. Linoleic acid is converted in the body to arachidonic acid. ALA is converted in the body to the fatty acids eicosapentaenoic acid (EPA) and docosahexaenoic acid (DHA), and these are all metabolized along the same biochemical pathways. The diet must supply these nutrients because the body cannot make them. Most American diets provide more than ten times as much omega-6 as omega-3 fatty acids.

AHRQ conducted an extensive systematic evidence-based review of many putative benefits of omega-3 fatty acids in 2004 and 2005 [134–144]. The review identified little evidence that omega-3 fatty acids had beneficial effects for most of the indications, other than possible cardiovascular benefits.

Omega-3 Fatty Acids: CVD

Essential omega-6 lipids promote development of pro-inflammatory eicosanoids, and essential omega-3 lipids promote development of anti-inflammatory eicosanoids. The anti-inflammatory effects of omega-3 fats might partially explain their cardioprotective benefit because incorporation of these fats in cell membranes promotes vasodilation, stimulates antiarrhythmic effects, and promotes vascular patency [145, 146].

Several large diet studies in the 1990s supported the cardiovascular benefits of omega-3 fats. The strongest evidence for a benefit was in secondary prevention of CVD. By far the largest secondary prevention trial was the GISSI-Prevenzione Study that enrolled 11,324 persons in Italy (of which about half were aged 60 years or older) who had survived a myocardial infarction for a median of 16 days [147]. Researchers examined the effects of daily supplementation for 3.5 years with 850–882 mg EPA and DHA (average EPA-DHA ratio of 1:2), and 300 mg vitamin E using a factorial design. Treatment with EPA and DHA reduced total mortality by 20 %, predominantly by reducing the incidence of sudden cardiac death by 45 %.

Few studies have focused on the effects of omega-3 fatty acid supplements for primary prevention of CVD. The Japan EPA Lipid Intervention Study, a primary prevention trial, randomly assigned 18,645 patients (mean age 61 years) with hyperlipidemia to receive 1.8 g/day EPA and a statin or a statin only [148]. At baseline, 20 % of the participants had a history of coronary artery disease. After an average of 4.6 years, the incidence of major coronary events was significantly reduced by 19 % in the EPA group (p=0.011). A 2006 meta-analysis of 48 RCTs, most of limited size and duration, in 36,913 participants followed for 6 months to 6 years found that supplementation with omega-3 fatty acids (0.4–7 g/day) was associated with smaller, nonsignificant reductions in total mortality (13 %; 95 % CI: 0.73, 1.03) and total CVD events (5 %; 95 % CI: 0.82, 1.12) [149]. A meta-analysis of 16 RCTs in 901 participants (baseline mean age 9–66 years) found that omega-3 supplements (0.45–4.5 g/day for a median of 56 days) lowered blood triglyceride levels in a dose-dependent manner, which might have reduced atherosclerosis risk [150]. In a meta-analysis of 11 RCTs and 7 cohort studies, supplementation with fish oils (1–3 g/day) or dietary consumption of fish for 5 days to 18 years had no impact on risk of atrial fibrillation (pooled odds ratio [OR]: 0.79; 95 % CI: 0.56, 1.12; p=0.19 for RCTs; pooled OR 0.83; 95 % CI: 0.59, 1.16; p=0.27 for cohort studies) [151]. In the AHRQ review of 123 studies that assessed the effects of omega-3 fatty acids on cardiovascular risk factors, omega-3 supplements reduced blood pressure by approximately 2 mmHg [142].

The most recent review of RCTs on the use of omega-3 fatty acids to prevent CVD included 20 primary and secondary prevention trials in 68,680 patients treated for more than 1 year with either omega-3-rich foods or supplements [152]. The omega-3 fatty acid supplements (0.24–3.4 g/day) had no statistically significant effects on all-cause mortality (RR: 0.96; 95 % CI: 0.91, 1.02), cardiac death (RR: 0.91; 95 % CI: 0.85, 0.98), sudden death (RR: 0.87, 95 % CI: 0.75, 1.01), myocardial infarction (RR: 0.80, 95 % CI: 0.76, 1.04), or stroke (RR: 1.05, 95 % CI: 0.93, 1.18) when all the trials were analyzed together. Moreover, factors such as prevention setting, the presence of implantable cardioverterdefibrillators in patients, or dose of omega-3 fatty acids did not appear to have affected the results.

Another meta-analysis was problematic. It included 14 secondary prevention RCTs in 20,485 adults (mean age 63 years) with a history of CVD who were followed for approximately 2 years or less [153]. Omega-3 fatty acid supplements (0.4–4.8 g/day) for 1–4.7 years had no significant effects on the overall incidence of cardiovascular events (RR: 0.99; 95 % CI: 0.89, 1.09), all-cause mortality (RR: 0.96; 95 % CI: 0.90, 1.02), sudden cardiac death (RR: 0.92; 95 % CI: 0.66, 1.30), myocardial infarction (RR: 0.81; 95 % CI: 0.65, 1.01), congestive heart failure (RR: 0.92; 95 % CI: 0.71, 1.17), or transient ischemic attack and stroke (RR: 1.13; 95 % CI: 0.77, 1.66). However, the authors found a small reduction in the number of cardiovascular deaths (RR: 0.91; 95 % CI: 0.84, 0.99) that disappeared when they excluded an open-label study without a placebo. Subgroup analyses also showed no significant associations by history of CVD, use of concomitant medications, trial quality, treatment duration, EPA or DHA dose, use of fish oil supplementation only, country, or area within a country. Although the authors found no association between dose of omega-3 fatty acids and clinical outcomes, the doses in most of the studies reviewed were very high (more than 1 g/day). The findings of this meta-analysis were later criticized because it included studies that were small, used an inappropriate placebo, lacked sufficient statistical power, or were of poor quality and excluded large studies that were not placebo controlled.

The 2004–2005 AHRQ review found only weak evidence that omega-3 fatty acids might have beneficial effects in tertiary prevention, such as on coronary artery re-stenosis after angioplasty and improved exercise capacity in patients with coronary atherosclerosis [142]. More high-quality, well-powered, well-controlled RCTs would be helpful for addressing these issues.

The FDA has approved a prescription form of omega-3 fatty acids (Lovaza, GlaxoSmithKline) for treatment of very high triglyceride levels [154]. The prescription product contains 0.84 g EPA and DHA in each 1 g capsule. Approximately 2–4 g EPA and DHA per day are required to lower triglyceride levels. EPA/DHA capsules are also available without a prescription, but the American Heart Association advises people with hypertriglyceridemia to undergo therapy with EPA and DHA only under a physician's care [155].

The American Heart Association and American College of Cardiology recommend an intake of 1 g/day of omega-3 fatty acids in people with atherosclerotic vascular disease to manage lipid levels [156]. However, the National Heart, Lung, and Blood Institute (NHLBI), American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines, American College of Physicians, American Association for Thoracic Surgery, Preventive Cardiovascular Nurses Association, Society for Cardiovascular Angiography and Interventions, and Society of Thoracic Surgeons do not recommend omega-3 supplements for people with or at risk of CVD, although NHLBI suggests eating at least two meals that include fish each week for people with high cholesterol levels [157, 158].

If older adults wish to use omega-3 fatty acid supplements, they should discuss this option with a qualified healthcare provider. Together, patients and providers should review the advantages and disadvantages of various treatment options, individual needs, and potential interactions (such as with blood thinners, including warfarin).

Omega-3 Fatty Acids: Cancers

Researchers have suggested that omega-3 fatty acids might be useful in cancer prevention, but the evidence for such an effect remains weak. The rationale is that omega-3 fatty acids suppress biosynthesis of arachidonic acid-derived eicosanoids that influence angiogenesis, apoptosis, cell proliferation, inflammation, immune cell function, and other mechanisms [159]. Some preclinical research has suggested that a nutrition intervention with omega-3 fatty acids might improve cell sensitivity to chemotherapy [160].

However, available epidemiologic data on omega-3 fatty acids and the risk of cancer continues to be inconsistent. In a review of 38 reports examining the effects of dietary consumption of omega-3 fatty acids on cancer risk in 20 prospective cohorts that included more than 700,000 participants, the results did not provide evidence for a significant association [141]. The authors concluded that dietary supplementation with omega-3 fatty acids is unlikely to prevent cancer.

In one large meta-analysis of seven studies in 489,465 adults on dietary consumption of omega-3 fatty acids and colon cancer prevention, overall effects were null but the results of a subgroup analysis showed a trend toward reduced risk of colorectal cancer among men (RR: 0.87; 95 % CI: 0.75–1.00) in four cohort studies [161]. The results of meta-analyses of studies on the effects of fish consumption or dietary alpha-linoleic acid intakes on prostate and gastric cancer risk were also negative [162–164]. Based on an ongoing systematic review, the World Cancer Research Foundation and American Institute of Cancer Research do not recommend omega-3 fatty acids for cancer prevention [39].

Omega-3 Fatty Acids: Cognition

Many elderly people take omega-3 fatty acid supplements to improve or prevent declines in cognitive function, but the evidence that this nutrient has these effects is inconclusive. Little has changed since the 2004–2005 AHRQ review to clarify its beneficial effects in the last few years [138].

A meta-analysis of ten RCTs on the effects of omega-3 fatty acid supplements (20-1,670 mg/day EPA and/or 40-1,550 mg/day DHA) on various domains of memory among healthy people, individuals with mild forms of cognitive impairment with no dementia, and those with Alzheimer disease found no effects when all of these patients were pooled and analyzed together (Hedge's g effect size: 0.04; 95 % CI: -0.06, -0.14) [165]. However, among people with cognitive impairment but without dementia, the results showed a small but statistically significant effect on immediate recall (Hedge's g effect size: 0.16; 95 % CI: 0.01, 0.31). In contrast, no benefit was evident in either healthy elders or

those who already had Alzheimer disease. Omega-3 fatty acid supplements in patients with Alzheimer disease did not improve scores on the Mini Mental State Examination or Alzheimer Disease Assessment Scale-Cognitive Subscale. In a Cochrane review of three RCTs lasting at least 6 months of omega-3 fatty acid supplements (200–1,093 mg/day EPA and 160–847 mg/day DHA and, in one study, 2 g ALA) in 3,536 participants, omega-3 fatty acid supplements had no effect on incident dementia [166]. However, many of these studies were small, of short duration, and of poor quality.

Although larger studies of longer duration might show effects, the quantity and strength of evidence is inadequate today to conclude that omega-3 fatty acid supplements protect cognitive function with aging or decrease the incidence or clinical progression of dementia, including Alzheimer disease. However, omega-3 fatty acid supplements at the doses studied appear to be safe, with few reported adverse events.

Effectiveness and Safety of Non-nutrient Supplements for the Elderly

Recent meta-analyses of the evidence supporting the use of a select number of these supplements that are popular are summarized below. These examples demonstrate the uncertainties that remain and the difficulties in making evidence-based recommendations regarding the use of these products.

Herbals and Botanicals

Botanical dietary supplements, which are made from plants, are sold in many forms, including fresh, dried, liquid, or solid extracts; tablets; capsules; powders; and tea bags. These products contain one or more botanical or herbal ingredients, such as *Gingko biloba* and St. John's wort, with or without vitamins and minerals. Experts do not understand how many botanical preparations work or what their active components are. The presence of adulterants or undeclared drugs also might be toxic.

Soy Protein and Soy Isoflavones

Soy contains three aglycones, or hydrolyzed forms of isoflavones—genistein, daidzein, and glycitein and their respective glycoside forms esterified to either malonic or acetic acid. Isoflavones in American foods are primarily present as glycoside conjugates. The levels of aglycones in supplements and foods tend to be very low unless the food has been fermented. The term "aglycone isoflavone equivalents" best describes the bioactive form of isoflavones. Hence, drawing generalizable conclusions about the health effects of soy from soy studies is very difficult without a complete description of the intervention [167]. This shortcoming plagued many of the early studies of the health effects of soy.

Soy Protein and Soy Isoflavones: CVD

One of the first health claims that the FDA permitted on food packages, in 1999, was that foods containing at least 25 g of soy protein could prevent coronary heart disease (although the validity of this claim has been challenged in recent years). However, later studies summarized below showed that these foods did not lower LDL levels.

In 2006, a science advisory panel of the American Heart Association determined, on the basis of 22 RCTs, that soy (25–135 g/day for soy proteins and 40–318 mg/day for soy isoflavones) reduced

LDL levels to a very small extent (about 3 %, on average), even when people ate very large amounts of soy protein (50 g, equivalent to about half an individual's usual protein intake). Moreover, soy protein had no significant effects on high-density lipoprotein cholesterol (HDL) cholesterol, triglycerides, lipoproteins, or blood pressure [168]. The 19 soy isoflavone studies evaluated in this review showed that supplementation with soy isoflavones had no effects on lipid risk factors or hot flushes in menopausal women, and the supplements had only equivocal effects on postmenopausal bone loss and unclear effects on cancer risk. There was little evidence that soy isoflavones alone are responsible for soy protein's cholesterol-lowering effects. The panel concluded that there was little evidence that soy protein had clinically important benefits compared to other proteins and did not recommend the use of isoflavone supplements in food or pills.

There is still much disagreement about the size of the effects of supplements containing soy or soy isoflavones on LDL. In a meta-analysis of 43 RCTs, median intakes of 30 g/day for 9.2 weeks, on average, of soy protein were associated with net decreases in serum lipoproteins of 5.5 % (95 % CI: -0.28, -0.18) in parallel studies and of 4.2 % (95 % CI: -0.22, -0.11) in crossover studies along with increases in serum HDL (3.2 %) in parallel studies [169]. In another meta-analysis of 30 RCTs with 2,913 participants to assess a United Kingdom health claim for foods containing soy, intakes of at least 25 g soy protein were associated with significant standard mean differences in mean LDL of 6.0 % or 0.23 mmol/L (95 % CI: -0.160, -0.306; p < 0.0001), total cholesterol of 3.7 % or 0.22 mmol/L (95 % CI: -0.142, -0.291; p < 0.0001), and the calculated average total cholesterol:HDL ratio was reduced by 10 %, although the results showed no dose-response effects at intakes of 15–40 g [170], which are the levels of isoflavones usually present in these foods. However, many different soy foods were included in the studies evaluated in this meta-analysis, and the quality of the included studies varied greatly. There is also continuing disagreement about whether the effects of soy on LDL and total cholesterol are due to soy isoflavones alone or to extracted soy isoflavones plus protein. A meta-analysis of ten systematic reviews and nine RCTs of the effects of consumption of an average of 70 mg/day soy isoflavones as aglycones without soy protein for up to 3 months found no measureable effects on either total (0.01 mmol/L, 95 % CI: -0.12, 0.14; p=0.86) or LDL cholesterol levels (0.03 mmol/L; 95 % CI: -0.011, 0.16; p = 0.71) in menopausal women with normal cholesterol levels [171].

The Women's Isoflavone Soy Health trial (WISH), a double-blind RCT sponsored by NIH, included 350 postmenopausal women who did not have clinically evident diabetes or CVD at baseline [172]. They received either two daily doses of 25 g soy protein containing 91 mg aglycone isoflavone equivalents or placebo for 2.7 years. The researchers monitored their carotid artery intima-media thickness, a sign of subclinical atherosclerosis. Atherosclerosis progressed in both groups (4.77 μ m/year, 95 % CI: 3.39, 6.16 in the treatment groups; and 5.68 μ m; 95 % CI: 4.30, 7.06 in the placebo groups), and the rate of progression did not differ statistically between the two groups. However, in the subgroup of healthy younger women who had undergone menopause within the previous 5 years, those receiving isoflavone soy protein supplements had a lower progression rate than placebo participants and this difference was statistically significant (2.16 μ m/year; 95 % CI: -1.10 versus 6.79 μ m/year; 95 % CI: 3.56, 10.01; respectively).

An additional analysis of results from WISH revealed that the soy protein supplement had no effect on global cognition, a composite score of the weighted sums of 14 neuropsychological test scores, after more than 2 years [173]. Mean standardized improvements from baseline in global cognition score were 0.42 in the soy group and 0.31 in the placebo group, a difference of 0.11 points (95 % CI: -0.13, -0.35). Visual memory was better in the soy group (mean standardized difference: 0.33 points; 95 % CI: 0.06, 0.60), but three other cognitive factors and individuals test scores showed no changes. Furthermore, cognitive factor change scores did not improve significantly in younger postmenopausal women. The authors concluded that long-term supplementation with soy isoflavone supplements in doses comparable to those in traditional Asian diets had no effects on global cognition in healthy postmenopausal women, although the supplements might have improved visual memory. The emerging evidence on soy protein and soy isoflavones demonstrates the importance for health care providers of keeping up with the literature on supplement safety and efficacy. For the same reason, health care providers need to continue to update their recommendations to patients because these recommendations might need to change over time.

Soy Protein and Soy Isoflavones: Osteoporosis

Osteoporosis is of great concern to many older women and to older men who have decreased bone density after the climacteric. A majority of the most effective treatments for decreased bone density in women have adverse side effects. Concerns have been raised that estrogen alone is associated with an increased stroke risk, and estrogen-progestin combinations are associated with a greater risk of both stroke and breast cancer [174, 175]. Bisphosphonates have been linked to osteonecrosis of the jaw and gastrointestinal problems, and other agents used to retard bone loss might have estrogen-related effects, including an increased risk of blood clots. Thus, there is much interest in the possible benefits of soy isoflavones and soy protein for bone health.

One meta-analysis of ten RCTs was based on data from 896 women who took 87 mg/day, on average, soy isoflavones for at least a year [176]. The results showed no effects on BMD at the lumbar spine, femoral neck, or total hip. A later and more complete systematic review included data on the effects of supplements containing soy isoflavone extracts or isolated soy protein on BMD and turnover markers in approximately twice as many menopausal women in Asia and Western countries [177]. This systematic review included three meta-analyses on the effects of soy isoflavones on lumbar spine, total hip, femoral neck, and trochanter BMD. Soy isoflavones improved lumbar spine BMD moderately but did not affect total hip, femoral neck, or trochanter BMD. A treatment period of 6 months appeared to be enough to exert beneficial effects at doses higher than 75 mg/day. Urine deoxypyridinoline, a bone resorption marker, decreased, but serum alkaline phosphatase and osteocalcin, two bone accretion markers, were not affected.

A more recent systematic review included 19 small RCTs in 1,442 women in Asia, Europe, and the United States who took supplements containing 22.7–126 mg soy isoflavones for 1 month to 2.5 years [178]. Bone-mineral density increased by 54 % (95 % CI: 0.13, 0.94; p < 0.001), and bone resorption, as measured by urine deoxypyridinoline, decreased by 23 % (95 % CI: -0.44, -0.02). However, serum alkaline phosphatase, a marker of bone formation, did not change. Also, there was substantial variation in results depending on the type of supplement, the dose of isoflavones, and the length of the intervention.

In contrast, an analysis of 12 RCTs in 1,433 women in Western countries who were consuming much lower total doses of soy isoflavones (52–120 mg/day for 24 weeks to 3 years) than the baseline levels of intake in the meta-analyses that included Asian women and in other older studies found few effects on bone loss [179]. The lack of effect might have been due to the lower isoflavone doses, the low baseline levels of isoflavone intakes, or other differences in the supplements that were used.

For women whose baseline levels of intake are already high, taking isoflavone supplements decreases bone loss at the lumbar spine (hip) to a moderate extent but not at other sites. However, the same isoflavone supplements might not benefit older American women whose baseline isoflavone levels are much lower. More studies are needed to verify the size of these effects in postmenopausal Americans women. Also, because some women eat soy foods and take supplements containing soy isoflavones and isolated soy proteins as well as anti-osteoporosis drugs, possible adverse interactions should be assessed. In a 2010 statement, the North American Menopause Society stated that data suggesting any benefit of dietary isoflavones in the prevention or treatment of postmenopausal osteoporosis, regardless of the isoflavone source, are relatively weak [180].

Soy Protein and Soy Isoflavones: Menopausal Symptoms

Hot flushes, also called hot flashes, are thermoregulatory disturbances, such as a sudden feeling of heat in the face, neck, and chest; night sweats; and other vasomotor symptoms that are common in perimenopausal and menopausal women. Recommendations regarding the use of hormone replacement therapy with estrogens alone or with progestins to relieve hot flushes, bone loss, or cardiovascular risk in menopausal women were revised and updated with newer long-term outcome data because of concerns that these treatments (especially estrogen-progestin therapy) could increase the risk of stroke, gallbladder disease, and breast cancer [181]. Health care providers counseled women to use hormone replacement therapy, if at all, at the lowest possible doses for only a short time. This advice led many women to seek alternative therapies for perimenopausal symptoms. These alternative therapies include dietary soy, isoflavone supplements, and red clover (*Trifolium pretense*, an isoflavone-rich medicinal herb) supplements as well other preparations that have been suggested to relieve these symptoms.

Soy isoflavones and soy protein have some effects on circulating estrogens and other hormones in pre- and postmenopausal women, but the clinical significance of these effects is equivocal. In a recent meta-analysis of 47 randomized or crossover studies, 35 studies in 1,165 postmenopausal women showed that taking supplements containing soy isoflavones and soy protein for 4–104 weeks had no statistically significant effects on estradiol, estrone, sex hormone-binding globulin, follicle-stimulating hormone, or luteinizing hormone levels [182]. The results did show a small, nonsignificant increase in total estradiol levels in women taking soy or isoflavones (approximately a 14 % SMD; p=0.07, in 21 studies). Therefore, although isoflavone-rich soy products might increase estradiol levels in postmenopausal women, the clinical implications of the modest hormonal changes in follicle-stimulating hormone and luteinizing hormone remain to be determined.

The existing studies of the effects of isoflavones and soy products have used various doses and preparations (such as soy food supplements, soy extracts, or isoflavone concentrates of genistein or daidzein), and their duration of use and other factors have also varied, making firm conclusions about the efficacy of isoflavones and soy products to treat perimenopausal symptoms difficult [183]. One meta-analysis of 19 RCTs of soy isoflavones for at least 3 months in postmenopausal women found an overall SMD in favor of soy of -0.39 (95 % CI: -0.53, -0.25), but the data were heterogeneous, particularly for women taking extracts and supplements [184]. This study illustrates the difficulty of combining many different types of products with many different forms of isoflavones in a single analysis.

The largest and most recent review focused only on the efficacy of extracted or synthesized soybean isoflavones on hot flashes [185]. Ingestion of soy isoflavones (median dose 54 mg aglycone equivalents) for 6 weeks to 12 months reduced the frequency of hot flashes by 20.6 % (95 % CI: -28.38, -12.86) compared to placebo. The soybean isoflavones also reduced hot flash severity by 26.2 % (95 % CI: -42.23, -10.15). Moreover, at higher doses (above the median of 19 mg genistein), their potency for reducing hot flash frequency was greater. Therefore, these well-characterized soy isoflavone supplements seemed to be somewhat better than placebos in reducing the frequency and severity of hot flashes, but the clinical effects might not have been great enough to help some women.

Women undergoing treatment for breast cancer with anti-estrogenic therapies should consult their physicians before using concentrated isoflavone supplements, although there is little cause for concern about eating foods containing soy.

Black Cohosh (Cimicifuga racemosa)

Black cohosh has been used in Germany since the mid-1950s to manage menopausal symptoms [186]. The pharmacologically active ingredients in black cohosh are prepared from the plant's rhizome and root [186].

The mechanism of action of black cohosh has not been defined. Previously, experts thought that black cohosh might exert its beneficial effects through estrogen receptors to moderate menopausal symptoms and hot flashes. More recently, research has shown that black cohosh extracts possess sero-tonergic activity [187]. Much of the published evidence on the safety and efficacy of black cohosh is based on the use of a commercial extract, Remifemin, which has regulatory approval for use in Germany. Black cohosh has been used in clinical studies lasting up to 12 months [188, 189]. The quality of black cohosh products in American markets varies.

Although some of the data are contradictory, most research findings suggest that black cohosh extract has a modest effect on menopausal hot flashes, night sweats, anxiety, and insomnia. The Herbal Alternatives for Menopause Trial (HALT) included 351 pre- and postmenopausal women aged 45–55 years [190]. Participants in this 12-month, 5-arm trial were randomly assigned to treatment with: (1) black cohosh (160 mg/day); (2) multiple botanicals (including 200 mg/day black cohosh); (3) multiple botanicals plus counseling about dietary soy; (4) hormone therapy (0.625 mg conjugated equine estrogen daily); or (5) placebo. Hormone therapy significantly reduced vasomotor symptoms compared with the other four treatments, and no differences were detected between the other four treatment arms at any point in the trial. In general, women who had undergone menopause more recently seemed to have a better response to black cohosh than those who had undergone menopause earlier. Because HALT included a placebo group and an estrogen replacement group, 12-month follow-up data collection, and a 92 % trial completion rate, HALT provides strong evidence that black cohosh has, at best, limited efficacy for the treatment of menopausal symptoms.

There is some evidence that black cohosh preparations can improve menopausal vasomotor symptoms, but results vary from study to study [191]. Moreover, an NIH-sponsored RCT in 89 healthy pre- and postmenopausal women did not show that supplementation with black cohosh (128 mg/day standardized to 7.27 mg triterpene glycosides) or red clover (398 mg/day standardized to 120 mg isoflavones) for 12 months reduced the number of vasomotor symptoms compared to placebos [189]. Another RCT conducted in 92 healthy women aged 45–65 years who reported at least six vasomotor symptoms per 24 h in Australia also failed to show that taking a formula containing Chinese herbs and black cohosh (350 mg/day) for 16 weeks reduced vasomotor symptoms [192]. A Cochrane review included 16 RCTs that compared orally administered preparations of black cohosh (either *Cimicifuga racemosa* or *Actaea racemosa*, median dose 40 mg/day for 23 weeks, on average) or placebo for treating menopausal symptoms in 2,027 perimenopausal and postmenopausal women [193]. The authors found no significant effects on the frequency of hot flushes (mean difference: 0.07 flushes per day; 95 % CI: 0.43, 0.56) or on menopausal symptom scores (SMD: –0.10; 95 % CI: –0.32, 0.11). In addition, the herb (64 mg/day for up to 12 weeks) had little effect on anxiety compared to placebo in another study in 28 peri- or postmenopausal women [194].

At present, there is insufficient evidence to support the use of black cohosh to treat menopausal symptoms. However, black cohosh in chemically and biologically standardized extract form has a relatively good safety profile [189]. Because of case reports of liver toxicity associated with the herb and because some black cohosh preparations on the market are poorly characterized, health authorities from several countries require a cautionary label on black cohosh products. Women who have liver disorders or who develop symptoms of liver trouble (such as abdominal pain, dark urine, or jaundice) should discontinue use of the herb and consult a health care practitioner.

Saw Palmetto (Serenoa repens)

Saw palmetto has been popular for many years as an alternative therapy for treating the symptoms associated with benign prostatic hyperplasia (BPH), a nonmalignant enlargement of the prostate that can lead to obstruction and irritation in the lower urinary tract. The ripe fruit of saw palmetto is used in several forms, including ground and dried fruit or whole berries. Saw palmetto is available as a

liquid extract, tablets, capsules, and as an infusion or a tea. Modern saw palmetto preparations contain lipids extracted from the powdered berries. The primary ingredients include saturated and unsaturated fatty acids as well as free and conjugated plant sterols [195]. Many saw palmetto products are standardized based on their fatty acid content.

Saw palmetto has been shown to have anti-androgenic, anti-proliferative, and anti-inflammatory properties that seem to be responsible for relieving BPH symptoms [196]. Saw palmetto does not seem to affect overall prostate size, but experts believe that it shrinks the inner prostatic epithelium [196]. The fruit of saw palmetto has been shown in vitro to inhibit 5-alpha-reductase and aromatase and prevent the conversion of testosterone to dihydrotestosterone, which might play a role in the development of BPH [197].

In a number of double-blind RCTs, some lasting up to a year, saw palmetto improved urinary flow rate and most other measures of prostate disease compared to placebo [198]. However, in a well-designed clinical trial, 225 men with moderate-to-severe BPH had no improvement after taking 320 mg daily saw palmetto for 1 year compared to placebo [199]. To evaluate these findings further, NIH sponsored a large RCT of the effects of increasing doses of saw palmetto extract (one, two, or three doses of 320 mg/day for 24 weeks at each dose) in 369 men aged 45 years or older [200]. At baseline, the men had peak urine flow rates of 4 mL/s or more and a score on the American Urological Association's Symptom Index of between 8 and 24 for lower urinary tract symptoms. At 72 weeks, symptom scores decreased slightly from 14.4 to 12.2 in the treatment group (-2.20 points; 95 % CI: -3.04, -0.36), but symptom scores also declined in the placebo group, from 14.7 to 11.7 (-2.99 points; 95 % CI: -3.81, -2.17). The mean difference in symptom score change from baseline to 72 weeks between the saw palmetto extract and placebo groups was 0.79 points, but this difference favored the placebo.

The latest systematic review of saw palmetto for BPH, a Cochrane review that included the trials discussed above, evaluated data on 5,666 men who had symptomatic BPH for at least a month [198]. The 32 RCTs reviewed compared the effects of saw palmetto (100–960 mg/day for 29.2 weeks, on average) and placebo on outcomes such as urologic symptoms and urodynamic measurements. In the high-quality, longer-term trials, the herb reduced lower urinary tract symptoms to the same extent as placebo when measured using the Symptom Score Index (mean difference: 0.25 points; 95 % CI: -0.58, 1.07). However, the effects of saw palmetto and placebo on other measures, such as nocturia, peak urine flow, and prostate size were not significantly different. Therefore, at this time, there is not enough evidence to support the use of saw palmetto for reducing the size of an enlarged prostate.

Saw palmetto is well tolerated, is essentially nontoxic, and has no known drug interactions. Adverse effects such as abdominal pain, diarrhea, nausea, fatigue, headache, decreased libido, and rhinitis are rare and reversible upon cessation of use of the herb [201, 202]. An advantage of saw palmetto, unlike the standard therapy of Proscar, is that it does not appear to affect PSA readings, allowing clinicians to more accurately screen patients with BPH for prostate cancer.

Ginkgo biloba

Ginkgo biloba is one of the most popular medicines in Germany and France, where physicians prescribe it for memory lapses, dizziness, anxiety, headaches, tinnitus, and other problems. Ginkgo leaf and *Ginkgo biloba* extract (GBE) contain flavonoids, terpenoids, and organic acids. Each of these chemicals should exist in a specific amount in clinical-quality. Many of gingko's constituents have intrinsic pharmacological effects, and it is possible that the constituents work synergistically to produce more potent pharmacological effects than any individual constituent. The authors of two early independent meta-analyses concluded that 120–240 mg/day GBE for 3–12 months significantly improved objective measures of cognitive function compared to placebo [203, 204]. NIH sponsored a 5-year, multicenter RCT, the Ginkgo Evaluation of Memory (GEM) study, to evaluate the safety and efficacy to prevent dementia or Alzheimer disease of 240 mg daily *Ginkgo biloba* (EGb 761) for 6.1 years compared to placebo in 3,069 participants who were older than 75 years [205]. The primary outcome was the incidence of all-cause dementia; secondary outcomes included rates of cognitive and functional decline, cardiovascular and cerebrovascular events, and mortality. The overall dementia rate was 3.3 per 100 person-years in the gingko group and 2.9 per 100 person-years in the placebo group. Neither the HRs for *Gingko biloba* compared with placebo for all-cause dementia (1.12; 95 % CI: 0.94, 1.33) nor for Alzheimer disease (1.16; 95 % CI: 0.97, 1.39) was statistically significant. Gingko also had no significant effect on the rate of progression to dementia in participants with mild cognitive impairment (HR: 1.13; 95 % CI: 0.85, 1.50). Nearly a third of the GEM cohort reported use of some type of nonvitamin/nonmineral dietary supplement at entry into study [206]. Nearly 10 % of these participants were taking ginkgo and were unwilling to give up their current ginkgo supplements or would not accept assignment to placebo.

The GuidAge study, a 5-year, double-blind RCT of the efficacy of 240 mg/day of a standardized gingko extract, EGb 761, for the prevention of Alzheimer disease in 2,854 patients over age 70 with a memory complaint, was the largest study carried out in Europe on the prevention of Alzheimer disease [207]. Unfortunately, long-term use of the standardized GBE in that study did not reduce the risk of progression to Alzheimer disease compared with placebo. By 5 years, the rates of cases were 1.2/100 person-years for the ginkgo group and 1.4/100 person-years for the placebo group for probable Alzheimer disease (HR: 0.84; 95 % CI: 0.60, 1.18) [208]. The incidence of adverse events, stroke, and other hemorrhagic or CVD events was not different between the two groups.

A Cochrane review of 36 RCTs that evaluated the use of gingko (80–600 mg/day) for 3–52 weeks to treat dementia or cognitive decline showed only inconsistent and unreliable results [209]. Most of the studies used the same standardized gingko extract, EGb 761. Many of the studies were small, with possible publication bias, although few adverse events were observed. In a meta-analysis of 13 RCTs on the use of gingko to improve memory, executive function, and attention in 1,145 healthy adults, the use of Ginkgo biloba had no positive effects [210]. Another meta-analysis of six RCTs on the use of a standardized ginkgo preparation (120-240 mg/day) for at least 6 months to treat dementia in 1,838 adults found a small but significant SMD in favor of the herb compared to placebo in cognitive function (-0.89; 95 % CI: -1.82, 0.04) and change from baseline in cognitive score (-2.65; 95 % CI: -4.53, -0.76) [211]. A larger analysis of nine RCTs in 2,372 participants with vascular dementia, mixed dementia, or Alzheimer disease who received a standardized gingko extract (120-240 mg/day) for 12-52 weeks found that for all participants, cognition change scores favored the standardized gingko extract compared to placebo (SMD: 0.58; 95 % CI: -1.14, -0.01) [212]. However, differences were not significant for activities of daily living. A review of all the RCTs on treatment of memory disorder with gingko and/or drugs in patients with multiple sclerosis and mild memory impairment found that gingko was not associated with any serious adverse events, although the data on outcomes was not included in detail in the review [213]. Thus, at present, there is no evidence to support the use of ginkgo to treat memory disorders.

Gingko is generally safe, although it has the potential to cause bleeding, especially when used in combination with anticoagulants [15]. This adverse effect was not apparent in a study of ginkgo (100 mg/day) and CoQ10 (100 mg/day) in 24 patients (median age 64.5 years) on stable warfarin therapy; the mean dosage of warfarin did not change during either 4-week treatment period [214]. Similarly, in a double blind, crossover RCT in 50 healthy men aged 20–44 years, ginkgo (240 mg/day) combined with aspirin (500 mg/day) for 7 days did not alter bleeding times or other coagulation parameters beyond the effects of aspirin alone [215]. The authors of a meta-analysis of 18 RCTs in 1,895 adults (of which 87 % had dementia, peripheral artery disease, or diabetes mellitus; the remainder were healthy) concluded that the bleeding risk of standardized *Ginkgo biloba* extract therapy was no higher than that of placebo [216]. However, patients who are taking anticoagulants, have bleeding disorders, or are scheduled for surgery or dental extractions should be counseled not to use gingko until they consult a health care practitioner.

Non-nutrient, Non-botanical Supplements

Non-nutrient supplements include non-botanical products, such as melatonin and CoQ10, and many "condition-specific" supplements, such as glucosamine/chondroitin, that are marketed to prevent or cure many ills.

Glucosamine/Chondroitin

Glucosamine hydrochloride is an amino sugar compound in cartilage and other connective tissues. Chondroitin sulfate is a complex glucopolysaccharide that helps cartilage retain water. Both are marketed to elders with claims that these products help maintain healthy joints, protect tendons, and decrease inflammation. In 2004, the FDA denied a health claim that glucosamine, chondroitin, and glucosamine and chondroitin reduce the risk of osteoarthritis, joint degeneration, and cartilage deterioration.

A Cochrane review of 25 studies of the use of glucosamine (oral dose of 1,500 mg/day or parenteral dose of 800–2,800 mg/week) for 25.5 weeks, on average, to treat osteoarthritis in 4,963 patients failed to show any reductions in pain as measured by the Western Ontario and McMaster Universities Arthritis Index (WOMAC) [217]. Some brands appeared to be more efficacious than others, and one proprietary brand appeared to be more effective than placebo in treating pain and gait impairments.

In response to these and other findings, NIH funded the Glucosamine/Chondroitin Arthritis Intervention Trial (GAIT) to determine whether a combination of glucosamine (1,500 mg/day), chondroitin (1,200 mg/day), celecoxib (200 mg/day), a combination of all three treatments, or placebo for 24 months would help people with moderate-to-severe pain from knee osteoarthritis [218]. An analysis of 24 months of data on 662 patients enrolled in 9 of the 16 GAIT sites revealed that the odds of achieving a 20 % decrease in WOMAC pain subscale score was 1.21 for celecoxib, 1.16 for glucosamine, 0.83 for glucosamine and chondroitin sulfate, and 0.69 for chondroitin sulfate alone. The confidence intervals for all treatments overlapped substantially.

Many other studies have been carried out since GAIT. A Norwegian group, for example, studied the effects of 1,500 mg glucosamine for 6 months on pain-related disability in 125 adults with chronic low back pain and degenerative lumber osteoarthritis in an RCT [219]. The glucosamine treatment did not reduce pain-related disability at the end of the intervention period or 6 months later. A systematic evidence-based review of the use of glucosamine (1,500–1,800 mg/day), chondroitin (1,200 mg/day), and methylsulfonylmethane (another supplement) for 2–4 months to treat degenerative joint disease and spinal osteoarthritis found only two high-quality articles that included 124 adults [220]. The authors of both of these articles reported negative results.

A systematic review included ten large RCTs in 3,803 patients on the use of glucosamine (1,500 mg/ day), chondroitin (800–1,200 mg/day), both glucosamine and chondroitin, or placebo for up to 3 years to treat osteoarthritis of the hip or knee that assessed differences in pain intensity on a 10 cm visual analogue scale [221]. The results were 0.4 cm for glucosamine (95 % CI: -0.7, -0.1), -0.3 cm for chondroitin (95 % CI: -0.7, 0.0), and 0.5 cm for the combination (95 % CI; -0.9, 0.0). None of these differences were considered to be clinically important. The differences were larger in industry-funded than in independently conducted trials. The authors concluded that none of these products reduced joint pain or narrowed joint space and that the treatment was not useful.

Although other studies have found that the use of glucosamine or chondroitin sulfate for longer than 3 years might delay radiological progression of osteoarthritis of the knee [222], their effects on pain have not been measured. And even if glucosamine or chondroitin are effective, their use has not been shown to be cost-effective [223].

A review of evidence by the Osteoarthritis Research Society International on therapies to treat hip and knee osteoarthritis identified 1 systematic review and 20 RCTs that evaluated the efficacy of glucosamine [224]. A meta-analysis of these RCTs found an effect size of 0.46 (95 % CI: 0.23, 0.69), but outcomes were heterogeneous and there was substantial evidence of publication bias. The review also identified five meta-analyses of chondroitin studies. The most recent, which included 20 trials in 3,846 patients, found an effect size of 0.75 (95 % CI: 0.50, 0.99), but these reports demonstrated publication bias and the results were also highly heterogeneous. Another expert group that reviewed different treatments for osteoarthritis of the knee recommended against using glucosamine or chondroitin [225].

In summary, the evidence supporting the use of glucosamine and chondroitin to prevent or treat joint pain and osteoarthritis is not compelling.

CoQ10

CoQ10, also known as ubiquinone, is involved in oxidative phosphorylation and generation of adenosine tri phosphate. In addition, CoQ10 is a free radical scavenger and membrane stabilizer. Japanese scientists first reported therapeutic properties of CoQ10 in the 1960s [196]. Some evidence suggested that CoQ10 might improve the efficiency of energy production in heart tissue and thus assist the heart during physical and/or oxidative stress. Typical doses of CoQ10 range from 100 to 200 mg in two to three divided doses a day [196].

CoQ10: Cardiovascular Health

Most of the consistent evidence for positive outcomes of CoQ10 therapy comes from people with congestive heart failure and not healthy people. Dozens of controlled trials have evaluated the effects of CoQ10 on CVD. Early studies summarized in the prior edition of this volume showed that CoQ10 has beneficial subjective (quality of life) and objective (increased left ventricular ejection fraction, improved stroke index scores, decreased hospitalizations) effects, but more recent studies have been less supportive [226].

The authors of a meta-analysis of nine RCTs in 824 patients of the use of CoQ10 (100–200 mg/day or 2 mg/kg) for 2–12 weeks to treat heart failure identified nonsignificant trends toward increased ejection fraction and reduced mortality [227]. However, the numbers of patients were insufficient to yield meaningful results. An updated meta-analysis that included 11 clinical trials in 319 patients of CoQ10 in doses ranging from 60 to 200 mg/day for 1–6 months found a net 3.7 % improvement in ejection fraction (95 % CI: 1.59, 5.77), and the effect was greater (6.74 %; 95 % CI: 2.63, 10.86) in patients who were not treated with angiotensin-converting enzyme inhibitors [228].

The most recent meta-analysis focused on 13 RCTs in 395 patients with heart failure in which the primary outcome was change in the ejection fraction or New York Heart Association functional class. CoQ10 supplementation (60–300 mg/day) for 4–28 weeks resulted in an improved pooled mean net change of 3.67 % (95 % CI: 1.60, 5.74) in ejection fraction and a decrease of -0.30 (95 % CI: -0.66, 0.06) in functional class [229]. Moreover, CoQ10 had positive effects in crossover trials, trials of short treatment duration (12 weeks), older studies (published before 1994), studies using a dose of 100 mg/ day or less, and studies whose participants had less severe congestive heart failure at study entry. However the numbers of patients in the studies were small, although the results appear promising.

Reduced plasma or serum levels of CoQ10 have been documented in observational studies and a number of RCTs in patients on statin therapies. The largest trial included 1,049 patients with moderate hypercholesterolemia and showed reductions in plasma CoQ10 levels of about a third after 1 year of treatment with atorvastatin [230]. The decrease in blood CoQ10 levels with statin treatment might be related to reduced synthesis of CoQ10 as well as decreases in circulating levels of CoQ10 because LDL cholesterol carries CoQ10 in the blood. Supplementation with CoQ10 (100 mg/day for 30 days) decreased muscle pain by 40 % (p<0.001) compared to vitamin E (400 IU/day) in 32 patients with statin-associated myopathy symptoms [231]. A recent systematic review of 18 studies on the role of CoQ10 in statin-associated myopathy confirmed that statin treatment (5–80 mg/day for 1–18 months) reduced circulating levels of CoQ10 and that CoQ10 supplementation could raise circulating levels of CoQ10 [232]. However, data on the effects of CoQ10 supplementation on myopathy symptoms remain scarce and contradictory. The review's authors concluded that the evidence is insufficient to prove that CoQ10 deficiency has a causal role in statin-associated myopathy and that more large, well-designed clinical trials were required to address this issue. In another study, 76 patients with myalgias presumed to be caused by statins were randomly assigned to therapy with 120 mg CoQ10 a day or placebo while continuing statin therapy [233]. CoQ10 did not reduce patient-reported pain to a greater extent than placebo.

At this time, therefore, routine use of CoQ10 to prevent myalgia due to statins is of questionable value.

CoQ10 has also been used to lower blood pressure. The authors of a Cochrane review of three RCTs evaluating the use of 100–120 mg/day CoQ10 for at least 3 weeks to reduce blood pressure in 96 patients with primary hypertension found that CoQ10 had no clear-cut effects [234]. Another meta-analysis of five RCTGs in 194 patients with and without CVD focused on CoQ10's effects on vascular endothelial function [235]. CoQ10 supplements (150–300 mg/day for 1–3 months) significantly improved flow-mediated dilation in a random effects model (SMD: 1.70; 95 % CI: 1.0, 2.4). However, endothelial function assessed by nitrate-mediated arterial dilatation did not improve when the authors used a fixed-effects model (SMD: -0.19; 95 % CI: -1.75, -1.38). An RCT that assessed the use of CoQ10 as adjunctive therapy in patients with hypertension and metabolic syndrome showed that it did not reduce systolic or diastolic blood pressure [236]. In an RCT of 56 patients with left ventricular systolic dysfunction and ischemic heart disease, CoQ10 supplementation (300 mg/day for 8 weeks) improved flow-mediated dilation by 1.51 % (p=0.03) compared to placebo [237]. Thus, the effects of CoQ10 supplements on CVD-related outcomes are not impressive at present.

In general, CoQ10 supplements appear to be safe. No significant side effects have been found, even in studies that lasted a year. CoQ10 chemically resembles vitamin K. Because vitamin K counters the anticoagulant effect of warfarin, anecdotal case reports have associated CoQ10 therapy with decreased international normalized ratio (INR), a measure of the time required for blood to clot compared to the average time in patients on warfarin therapy. In a randomized, double-blind, placebo-controlled, crossover trial, 100 mg CoQ10 daily for 4 weeks had no effect on INR in 21 patients on warfarin [214]. Patients taking both CoQ10 and warfarin should be aware that CoQ10 could decrease the effectiveness of warfarin and lower clotting time.

CoQ10: Neurological Health

There is great interest in investigating the potential usefulness of CoQ10 for treating neurodegenerative diseases.

CoQ10 has shown some promise for slowing the progression of Parkinson's disease. In an NIHsponsored RCT, 80 people with early Parkinson's disease were given 300, 600, or 1,200 mg daily CoQ10 or placebo for 16 months [238]. The results suggested that CoQ10, especially at the highest dose, might have slowed disease progression. In a pilot trial, doses of up to 3,000 mg/day of CoQ10 were well tolerated [239]. In a more recent trial conducted in Germany, researchers randomly assigned 131 patients with mid-stage Parkinson disease who were taking medications for Parkinson disease to 300 mg of CoQ10 daily or placebo for 3 months [240]. The results did not show significant differences between the groups, although both the treatment and placebo groups had significant improvement in Unified Parkinson's Disease Rating Scale scores. The investigators concluded that the study did not support the hypothesis that restoring the impaired energy metabolism of dopaminergic neurons leads to beneficial effects in patients undergoing treatment for mid-stage Parkinson's disease. A Cochrane review of four RCTs that assessed the safety and efficacy of CoQ10 in 452 patients with early-stage or mid-stage primary Parkinson's disease found that up to 1,200 mg/day for 16 months CoQ10 was well tolerated, with few adverse effects other than mild pharyngitis and diarrhea [241]. The treatment was associated with improvements in activities of daily living (weighted mean difference: -3.12; 95 % CI: -5.880, -0.36) and other tests.

In summary, the small neuroprotection trials of CoQ10 in Parkinson's disease conducted to date have been somewhat encouraging. However, further evidence is required before CoQ10 supplements can be recommended for routine use.

Guidelines for Safe Use of Dietary Supplements by Elders

Dietary supplements influence drug therapy within and outside of acute care facilities. Because so many elderly adults take medications and polypharmacy is very common, health care providers need to recognize the potential effects of dietary supplements on medical status and treatment. To that end, the Joint Commission has included vitamins, herbals, and "nutraceuticals" in their scope of medications addressed by safety standards. The Joint Commission expects healthcare facilities to establish standardized processes to compare current medication orders to medications usually taken and thus eliminate medication errors, such as omissions, drug interactions, duplications, and dosing discrepancies in diverse care settings (including admission, surgery, and discharge) and all transitions of care. For this reason, healthcare staff need to ask their patients about their use of dietary supplements. Registered dietitians, in particular, should become familiar with dietary supplements. Nurses also need to become familiar with dietary supplements because they do most admission assessment interviews. Shared accountability encourages processes that use input from all disciplines.

Some manufacturers use standardization to ensure the batch-to-batch consistency of their products. This process often involves the use of markers that help identify a consistent product and provide a measure of quality control. Because no legal or regulatory definition of standardization exists for dietary supplements, standardization statements on dietary supplement packages have no association with quality. Some hospitals have restricted or banned patient use of dietary supplements because of drug–nutrient interactions, inconsistent manufacturing standards, and concerns that admission assessments of such substances cannot be accurately conducted [242, 243].

Preventing Interactions with Dietary Supplements

The potential for drug/supplement interactions is particularly likely in the elderly due to their high use of medications. The Natural Medicines Comprehensive Database identifies more than 1,900 drug–supplement interactions [196]. Approximately 15 % of these interactions are likely to occur and be of high severity when they occur, including 74 % that are probable and of moderate severity and 11 % that are unlikely to occur and insignificant. Many of these interactions are documented only by a few case reports, underscoring the need for better adverse event reporting. Of these potential interactions, 25–45 % are pharmacokinetic (depending on their absorption, distribution, metabolism, and excretion patterns) and 45–55 % are pharmacodynamic (due to their metabolic effects). But most of these estimates are speculative because most of the interactions are not documented in an easily retrievable clinical format.

Botanical and herbal products, unlike most conventional drugs, consist of complex mixtures of bioactive entities that might or might not have therapeutic activity. The active ingredients are frequently not known and complete characterization of all the chemical constituents is lacking, as

Criterion	Drugs	Herbal and botanical supplements
History of use	Typically a short tradition of use	Typically a long tradition of use for the whole plant but not necessarily for extracts or other preparations
Ingredient characteristics	Active ingredient known	Active ingredient often not known, and marker compounds often used as standards
	Pure compound available	Pure compounds not available; many compounds often present
	Composition constant	Composition variable due to season, temperature, and other harvest conditions
Mechanism of action	Often known	Often unknown
Safety	Side effects often known	Side effects often not studied and adverse effects might have been evaluated for some but not all marketed forms
	Frequently a narrow therapeutic window	Usually a wide therapeutic window
	Frequent adverse effects	Rare or unknown and poorly characterized adverse effects
Regulatory oversight in the United States	Requires prior approval from Food and Drug Administration for marketing	Does not require premarket approval from Food and Drug Administration for marketing

Table 23.1 Distinctions between herbal and botanical supplements and drugs

described in Table 23.1. As with conventional drugs, many herbal products are therapeutic at one dose and toxic at another, particularly if they are highly concentrated. Concurrent use of herbs can mimic, magnify, or oppose the effects of drugs. The importance of unrecognized interactions between herbs and conventional drugs is particularly important in cardiology because many cardiovascular drugs, such as warfarin and digoxin, have narrow therapeutic windows.

Additional references for herb/botanical-drug interactions are listed in Table 23.2. Some herbal products affect clinical laboratory test results by direct interference, most commonly with immunoassays; exert physiologic effects through toxicity or enzyme induction; or produce interference by contaminants.

Needs Assessment for Dietary Supplements

Given the prevalence of inadequate dietary intakes of many vitamins and minerals as well as the potential for inappropriate supplement use among older adults, health care providers should routinely screen elderly clients for both the need to use dietary supplements and the risk of excessive supplement intake. The following questions can identify conditions, symptoms, or situations that suggest a need to offer counseling on the use of fortified foods and/or dietary supplements in addition to advising older adults to make appropriate dietary changes.

- Has the older client been diagnosed with physical conditions or does he/she display symptoms that could indicate a need for fortified foods and/or dietary supplements? Examples of conditions that could indicate a need for dietary supplements and supplements that might benefit patients with these conditions are as follows:
 - · Osteoporosis: calcium, vitamin D, and magnesium
 - Alcohol abuse: B-vitamins and magnesium
 - Gastrointestinal abnormalities, including acid reflux, diarrhea, fat malabsorption, and atrophic gastritis: vitamin B₁₂, fat-soluble vitamins (vitamins A, D, E, and K), and magnesium

	Title	Additional information
Monographs	American Herbal Pharmacopoeia® monographs	Available from: American Herbal Pharmacopoeia PO Box 66809 Scotts Valley, CA 95067 www.herbal-ahp.org ahp@herbal-ahp.net Phone: 831-461-6318
	Blumenthal M, Goldberg A, Brinkmann J, editors. <i>Herbal Medicine: Expanded Commission E Monographs</i> . Austin (TX): Integrative Medicine Communications; 2000.	Available to American Botanical Council members from: American Botanical Council 6200 Manor Road Austin, TX 78723 www.herbalgram.org abc@herbalgram.org Phone: 512-926-4900 Also available from online booksellers
-	 World Health Organization. WHO Monographs on Selected Medicinal Plants. Vol. 1. Geneva: World Health Organization: 1999. (28 monographs) World Health Organization. WHO Monographs on Selected Medicinal Plants. Vol. 2. Ravello-Salerno (Italy): World Health Organization; 1999. (30 monographs) World Health Organization. WHO Monographs on Selected Medicinal Plants. Vol. 3. Ottawa (Canada): World Health Organization; 2001. (31 monographs) World Health Organization. WHO Monographs on Selected Medicinal Plants. Vol. 4. Station of Health Organization; 2001. (31 monographs) World Health Organization. WHO Monographs on Selected Medicinal Plants. Vol. 4. Station-Paestum (Italy): World Health Organization; 2005. (28 monographs) 	Provide scientific information on the safety, efficacy, and quality control of widely used medicinal plants; available at: Volume 1: http://apps.who.int/medicinedocs/pdffs2200e/s2200e/s2200e.pdf Volume 2: http://apps.who.int/medicinedocs/documents/s14213e/ s14213e.pdf Volume 4: http://apps.who.int/medicinedocs/documents/s17534en/ s17534en.pdf
Dooks Dooks	 Barrett M, eutor. <i>Ine Handbook of Cumcalty Lested Herbal Kemedies</i>. New York: Routledge; 2013. Bone K, Milles S. <i>Principles and Practice of Phytotherapy: Modern Herbal Medicine</i>. 2nd ed. London: Churchill Livingstone; 2013. Coates PM, Betz JM, Blackman MR, Cragg GM, Levine M, Moss J, White JD, editors. <i>Encyclopedia of Dietary Supplements</i>. 2nd ed. New York: Informa Healthcare; 2010. Fugh-Berman A. <i>The 5-Minute Herb and Dietary Supplement Consult</i>. Philadelphia: Lippincott Williams & Wilkins; 2003. Gardner Z, McGuffin M, editors. <i>American Herbal Products Association's Botanical Safety Handbook</i>. 2nd ed. Boca Raton (FL): CRC Press; 2013. Hendler S, Rorvik D. <i>PDR® for Nutritional Supplements</i>. 2nd ed. Montvale (NJ): PDR Network; 2008. 	

Table 23.2 References on dietary supplements and herb-drug interactions

	Natural Standard. Natural Standard Herb & Supplement Guide: An Evidence-Based Reference, 1st ed. Mosby, St. Louis (MO); 2010.Otten JJ, Hellwig JP, Meyers LD, editors. Dietary Reference Intakes: The Essential Guide to Nutrient Requirements. Washington: National Academics Press; 2006.Pharmaceutical Press Editorial. Herbal Medicines. 4th ed. London: The Pharmaceutical Press; 2013.	
	Schulz V,Hänsel R, Blumenthal M, Tyler VE. Rational Phytotherapy: A Reference Guide for Physicians and Pharmacists. 5th ed. New York: Springer-Verlag; 2004.	
	I nomson reauncare. <i>FDK⁻ for rieroat meatcines</i> . 4th ed. Montvale (NJ): 1nomson Reuters; 2007.	
Databases	Integrative Medicine Service, Memorial Sloan-Kettering Cancer Center. About Herbs, Botanicals & Other Products.	Evidence-based information about herbs, botanicals, and dietary supplements
	Available from: www.mskcc.org/cancer-care/integrative-medicine/ about-herbs-botanicals-other-products.	
	ConsumerLab. Available at: www.consumerlab.com.	Evaluations of the composition, potency, purity, bioavailability, and consistency of commercially available dietary supplements; annual subscriptions available for a fee
	Office of Dietary Supplements and National Library of Medicine, National Institutes of Health. Dietary Supplement Label Database. Available at: www.dsld.nlm.nih.gov/dsld/	Data on information from labels of some dietary supplements marketed in the United States
	Health Canada. Natural Health Products Ingredients Database. Available at: webprod. hc-sc.gc.ca/nhpid-bdipsn/	Information on acceptable medicinal and nonmedicinal ingredients used in natural health products in Canada
	MedLine Plus, National Library of Medicine. Herbal Medicine. Available at: http://www.nlm.nih.gov/medlineplus/herbalmedicine.html	Up-to-date, high-quality health care information on herbs and herbal medicine
	MedLine Plus, National Library of Medicine. Dietary Supplements. Available at: http://www.nlm.nih.gov/medlineplus/dietarysupplements.html.	Links to information sources, including the latest research, on dietary supplements
	Natural Medicines Comprehensive Database. Available to subscribers at: www.naturaldatabase.com	Searchable database of monographs with extensive information pm common uses, evidence of efficacy and safety, mechanisms,
		interactions, and dosages of dietary supplements that is extensively referenced and updated daily; requires a subscription fee
	Natural Standard. Food, Herbs & Supplements. Available at:	Searchable database of information on herbs and supplements, inclusing analysis of anidozoon accurates a concentration for
	http://www.naturaistandaru.com/uatabascs/netbssupplements/au/au/ National Cantar for Commlementary and Alternative Medicine National Institutes of	Including quanty of evidence; requires a subscription rec Brief fact chease with basic information shout energific harks and
	Institutional Center for Comprementary and Augmente internative, inauronal manumes of Health. Herbs at a Glance. Available at: nccam.nih.gov/health/herbsataglance.htm	brief lact sneets with basic information about specific fields and botanicals, including common names, scientific evidence,

23 Dietary Supplements in Older Adults

411

(continued)

potential side effects, and information resources

(continued)	
Table 23.2	

	Title	Additional information
	Office of Dietary Supplements. PubMed Dietary Supplement Subset. Available at: ods.od.nih.gov/Research/PubMed_Dietary_Supplement_Subset.aspx.	Filter designed to limit search results of the PubMed database to citations from a broad spectrum of dietary supplement literature, including publications on vitamin, mineral, phytochemical, ergogenic, botanical, and herbal supplements in human nutrition and animal models; includes links to many full-text articles on journal websites and other related Web resources
Mobile applications	Office of Dietary Supplements. MyDS. Available at: http://ods.od.nih.gov/ HealthInformation/mobile/AboutMyDS.aspx	A free mobile application that helps consumers track the vitamins, minerals, herbs, and other supplements they take and provides access to reliable, science-based information on dietary supplements
	European Scientific Operative on Phytotherapy. ESCOP Herb Reference iPhone [®] App. Available at: http://www.escop.com	Lists more than 100 herbs and their uses for a range of health conditions

23 Dietary Supplements in Older Adults

- Renal insufficiency: vitamin D
- Cardiovascular disease: vitamin E and folic acid
- Nutritional anemia: iron, vitamin B₁₂, and folic acid
- Weight loss, anorexia, nausea, or other symptom indicating inadequate intake: general MVM supplement or medical nutritional supplement
- 2. Does the older client have eating or supplement use behaviors that place him/her at high risk of nutritional deficiency or excess? These behaviors could include the following:
 - Eats fewer than two meals per day
 - Drinks more than one (women) or two (men) alcoholic beverages per day
 - · Eats less than two servings of vegetables per day
 - Eats less than two servings of fruits per day
 - Takes many dietary supplements every day
 - Takes doses of vitamin and mineral supplements that greatly exceed the RDA (unless a physician has recommended these doses), especially if intakes are greater than the UL
 - Does not know the doses of supplements that he or she takes
 - · Consumes a high volume of grapefruit juice that could interact with medications
 - Uses proton-pump inhibitors, which could cause significant magnesium depletion and the patient therefore require periodic checks of serum magnesium levels
- 3. Do body composition changes or physical limitations suggest that the patient might have nutrient deficiencies? These changes and limitations include the following:
 - Has a body mass index of less than 21
 - Has unintentionally lost or gained 10 lb in the last 6 months
 - Has difficulty chewing or swallowing
 - Has physical disabilities that limit his or her ability to shop for and/or prepare food
- 4. Does the patient have lifestyle habits or conditions that could impair normal nutrient intakes? These lifestyle habits or conditions include the following:
 - Is housebound
 - Has clinical evidence of depressive illness
 - Needs assistance with self-care
 - Demonstrates mental or cognitive impairment
- 5. Does the client have a disease or medical condition that contraindicates the use of dietary supplements? For example, does the patient have medical problem(s), such as impaired renal function, that limit excretion of supplements such as magnesium? Does the patient take any medications that could adversely interact with dietary supplements?

Every day, clinicians must interpret the evidence, both established and emerging, to provide the best care for their patients. Recommendations occur on many levels [226]. Level I recommendations are influenced by the physician's personal experience with the disease and capacity to deal with risk. When making such decisions, physicians might ask themselves, "Would I have this done for myself or for someone else in my immediate family?" A clinician making Level II recommendations for a patient is also influenced by prior experience, but the strength of the scientific evidence might play greater role. Lastly, a Level III recommendation can be viewed as an across-the-board public health recommendation for a population. Such recommendations must be based on rigorous assessments of the scientific evidence. As the data on dietary supplements become more robust and more evidence becomes available, clinicians will feel more confident in making recommendations about the use of dietary supplements for health maintenance and risk reduction at both the individual and group levels.

Dietary Supplement Databases and Other Resources

In addition to the print resources listed in Table 23.2, the Office of Dietary Supplements (ODS), in collaboration with the National Library of Medicine of NIH recently developed a dietary supplement label database that is available at http://dsld.nlm.nih.gov. Other useful information on dietary supplements is available on the ODS website, at http://www.ods.od.nih.gov, and the National Center for Complementary and Alternative Medicine's website, at http://www.nccam.nih.gov. The FDA website, http://www.fda.gov/Food/DietarySupplements, provides up-to date information on dietary supplement recalls and health claims.

Conclusion: Continuing Concerns

Older people, particularly those in fragile health, need to describe and discuss their use of dietary supplements with their health care providers. Providers and patients must talk frankly with each other about these products. Open discussions about the reasons elders use dietary supplements sometimes help health care providers identify health problems, such as memory difficulties, that otherwise might go undetected. These discussions also permit providers and patients to work together to choose appropriate supplements when supplements are necessary, avoid adverse interactions with the many medications that patients take, and avoid falling prey to misleading and deceptive advertising. Dietary supplements are often costly and can be difficult for elders on fixed incomes to afford. In addition, because elders often suffer from many chronic diseases that require multiple medications, they have a higher risk than younger people of adverse reactions. For all of these reasons, frank and forthright discussions about the safety and efficacy of dietary supplements should be a part of interactions between older patients and health care providers.

Acknowledgements The authors thank their colleagues in the Office of Dietary Supplements, especially Leila Saldanha, Ph.D., R.D., for their helpful comments.

References

- 1. Buhr G, Bales CW. Nutritional supplements for older adults: review and recommendations-part I. J Nutr Elder. 2009;28(1):5–29.
- Dawson-Hughes B, Harris SS, Krall EA, Dallal GE, Falconer G, Green CL. Rates of bone loss in postmenopausal women randomly assigned to one of two dosages of vitamin D. Am J Clin Nutr. 1995;61:1140–5.
- Chapuy MC, Arlot ME, Duboeuf F, Brun J, Crouzet B, Arnaud S, et al. Vitamin D3 and calcium to prevent hip fractures in the elderly women. N Engl J Med. 1992;327(23):1637–42.
- Jackson RD, LaCroix AZ, Gass M, Wallace RB, Robbins J, Lewis CE, et al. Calcium plus vitamin D supplementation and the risk of fractures. N Engl J Med. 2006;354(7):669–83.
- Grant AM, Avenell A, Campbell MK, McDonald AM, MacLennan GS, McPherson GC, et al. Oral vitamin D3 and calcium for secondary prevention of low-trauma fractures in elderly people (Randomised Evaluation of Calcium Or vitamin D, RECORD): a randomised placebo-controlled trial. Lancet. 2005;365(9471):1621–8.
- O'Brien KO. Combined calcium and vitamin D supplementation reduces bone loss and fracture incidence in older men and women. Nutr Rev. 1998;56:148–58.
- 7. Institute of Medicine. Dietary reference intakes for calcium and vitamin D. Washington, DC: National Academies Press; 2011.
- Picciano MF, Yetley EA, Coates PM. Folate and health. CAB Rev Perspect Agric Vet Sci Nutr Nat Resour [Internet]. 2007 June 20, 2013; 2:[No. 018 p.]. Available from: http://www.cabi.org/cabreviews/?loadmodule=rev iew&page=4051&reviewid=29483&site=167.

- Refsum H, Ueland PM, Nygard O, Vollset SE. Homocysteine and cardiovascular disease. Annu Rev Med. 1998; 49:31–62.
- Tawakol A, Migrino RQ, Aziz KS, Waitkowska J, Holmvang G, Alpert NM, et al. High-dose folic acid acutely improves coronary vasodilator function in patients with coronary artery disease. J Am Coll Cardiol. 2005;45:158–64.
- 11. Holick MF. Vitamin D, deficiency. N Engl J Med. 2007;357(3):266-81.
- Chung M, Lee J, Terasawa T, Lau J, Trikalinos TA. Vitamin D with or without calcium supplementation for prevention of cancer and fractures: an updated meta-analysis for the U.S. Preventive Services Task Force. Ann Intern Med. 2011;155(12):827–38.
- Bailey RL, Gahche JJ, Miller PE, Thomas PR, Dwyer JT. Why US adults use dietary supplements. JAMA Intern Med. 2013;173(5):355–61.
- Schmidt BM, Ribnicky DM, Lipsky PE, Raskin I. Revisiting the ancient concept of botanical therapeutics. Nat Chem Biol. 2007;3(7):360–6.
- 15. Fugh-Berman A. Herb-drug interactions. Lancet. 2000;355:134-8.
- Journal NB. Supplement business report 2012. San Diego, CA: New Hope Natural Media, Penton Media Inc.; 2012.
- Rovati LC, Girolami F, Persiani S. Crystalline glucosamine sulfate in the management of knee osteoarthritis: efficacy, safety, and pharmacokinetic properties. Ther Adv Musculoskelet Dis. 2012;4(3):167–80.
- Bailey RL, Gahche JJ, Lentino CV, Dwyer JT, Engel JS, Thomas PR, et al. Dietary supplement use in the United States, 2003–2006. J Nutr. 2011;141(2):261–6.
- Kishiyama SS, Leahy MJ, Zitzelberger TA, Guariglia R, Zajdel DP, Calvert Jr JF, et al. Patterns of dietary supplement usage in demographically diverse older people. Altern Ther Health Med. 2005;11(3):48–53.
- Bardia A, Nisly NL, Zimmerman BM, Gryzlak BM, Wallace RB. Use of herbs among adults based on evidencebased indications: findings from the National Health Interview Survey. Mayo Clin Proc. 2007;82(5):561–6.
- Nahin RL, Byrd-Clark D, Stussman BJ, Kalyanaraman N. Disease severity is associated with the use of complementary medicine to treat or manage type-2 diabetes: data from the 2002 and 2007 National Health Interview Survey. BMC Complement Altern Med. 2012;12:193.
- Eisenberg DM, Davis RB, Ettner SL, Appel S, Wilkey S, Rompay MV, et al. Trends in alternative medicine use in the United States, 1990–1997: results of a follow-up national survey. JAMA. 1998;280:1569–75.
- 23. Sloan E. Why people use vitamin and mineral supplements. Nutr Today. 2007;42(2):55-61.
- Health Focus International. The 2005 HealthFocus trend report: a national study of public attitudes and actions toward shopping and eating. St. Petersburg, FL: HealthFocus International; 2005.
- 25. U.S. Food and Drug Administration. Draft guidance for industry: dietary supplements: new dietary ingredient notifications and related issues 2011 [August 7, 2013]. Available from: http://www.fda.gov/Food/Guidance Regulation/GuidanceDocumentsRegulatoryInformation/DietarySupplements/ucm257563.htm.
- Jialal I, Devaraj S. Vitamin E supplementation and cardiovascular events in high-risk patients. N Engl J Med. 2000;342(25):1917–8.
- Lonn E, Bosch J, Yusuf S, Sheridan P, Pogue J, Arnold JM, et al. Effects of long-term vitamin E supplementation on cardiovascular events and cancer: a randomized controlled trial. JAMA. 2005;293(11):1338–47.
- Lee IM, Cook NR, Gaziano JM, Gordon D, Ridker PM, Manson JE, et al. Vitamin E in the primary prevention of cardiovascular disease and cancer: the Women's Health Study: a randomized controlled trial. JAMA. 2005;294(1): 56–65.
- Klein EA, Thompson Jr IM, Tangen CM, Crowley JJ, Lucia MS, Goodman PJ, et al. Vitamin E and the risk of prostate cancer: the Selenium and Vitamin E Cancer Prevention Trial (SELECT). JAMA. 2011;306(14):1549–56.
- Ebbing M, Bonaa KH, Nygard O, Arnesen E, Ueland PM, Nordrehaug JE, et al. Cancer incidence and mortality after treatment with folic acid and vitamin B12. JAMA. 2009;302(19):2119–26.
- 31. Andreeva VA, Touvier M, Kesse-Guyot E, Julia C, Galan P, Hercberg S. B vitamin and/or omega-3 fatty acid supplementation and cancer: ancillary findings from the supplementation with folate, vitamins B6 and B12, and/ or omega-3 fatty acids (SU.FOL.OM3) randomized trial. Arch Intern Med. 2012;172(7):540–7.
- 32. Song Y, Manson JE, Lee IM, Cook NR, Paul L, Selhub J, et al. Effect of combined folic acid, vitamin B(6), and vitamin B(12) on colorectal adenoma. J Natl Cancer Inst. 2012;104(20):1562–75.
- 33. Gibson TM, Weinstein SJ, Pfeiffer RM, Hollenbeck AR, Subar AF, Schatzkin A, et al. Pre- and postfortification intake of folate and risk of colorectal cancer in a large prospective cohort study in the United States. Am J Clin Nutr. 2011;94(4):1053–62.
- 34. National Eye Institute. Eye health needs of older adults literature review [August 7, 2013]. Available from: http:// www.nei.nih.gov/nehep/research/The_Eye_Health_needs_of_Older_Adults_Literature_Review.pdf.
- 35. U.S. Food and Drug Administration. Current good manufacturing practice in manufacturing, packaging, labeling, or holding operations for dietary supplements; Final Rule. Fed Regis 72(121):34751–958 Washington, DC: Food and Drug Administration; 2007 [cited 2013 June 20]. Available from: http://www.fda.gov/Food/GuidanceRegulation/GuidanceDocumentsRegulatoryInformation/DietarySupplements/ucm238182.htm.

- Corby-Edwards AK. Regulation of dietary supplements. Washington, DC: Congressional Research Service; 2013. p. 21.
- Coates PM, Dwyer JT, Thurn AL. Introduction to state-of-the-science conference: multivitamin/mineral supplements and chronic disease prevention. Am J Clin Nutr. 2007;85(1):2558–6.
- U.S. Preventive Services Task Force. Routine vitamin supplementation to prevent cancer and cardiovascular disease: recommendations and rationale. Ann Intern Med. 2003;139(1):51–5.
- 39. World Cancer Research Fund, American Institute for Cancer Prevention. Food, nutrition, physical activity, and the prevention of cancer: a global perspective. Washington, DC: American Institute for Cancer Research; 2007.
- 40. Mursu J, Robien K, Harnack LJ, Park K, Jacobs Jr DR. Dietary supplements and mortality rate in older women: the Iowa Women's Health Study. Arch Intern Med. 2011;171(18):1625–33.
- 41. Macpherson H, Pipingas A, Pase MP. Multivitamin-multimineral supplementation and mortality: a meta-analysis of randomized controlled trials. Am J Clin Nutr. 2013;97(2):437–44.
- 42. Chan AL, Leung HW, Wang SF. Multivitamin supplement use and risk of breast cancer: a meta-analysis. Ann Pharmacother. 2011;45(4):476–84.
- 43. Stratton J, Godwin M. The effect of supplemental vitamins and minerals on the development of prostate cancer: a systematic review and meta-analysis. Fam Pract. 2011;28(3):243–52.
- 44. Park Y, Spiegelman D, Hunter DJ, Albanes D, Bergkvist L, Buring JE, et al. Intakes of vitamins A, C, and E and use of multiple vitamin supplements and risk of colon cancer: a pooled analysis of prospective cohort studies. Cancer Causes Control. 2010;21(11):1745–57.
- Prentice RL. Clinical trials and observational studies to assess the chronic disease benefits and risks of multivitaminmultimineral supplements. Am J Clin Nutr. 2007;85(1):308S–13.
- 46. Stephen AI, Avenell A. A systematic review of multivitamin and multimineral supplementation for infection. J Hum Nutr Diet. 2006;19(3):179–90.
- Grima NA, Pase MP, Macpherson H, Pipingas A. The effects of multivitamins on cognitive performance: a systematic review and meta-analysis. J Alzheimers Dis. 2012;29(3):561–9.
- 48. McNeill G, Avenell A, Campbell MK, Cook JA, Hannaford PC, Kilonzo MM, et al. Effect of multivitamin and multimineral supplementation on cognitive function in men and women aged 65 years and over: a randomised controlled trial. Nutr J. 2007;6:10.
- 49. Institute of Medicine. Dietary reference intakes: water, potassium, sodium, chloride, and sulfate. Washington, DC: National Academies Press; 2004.
- Bailey RL, Dodd KW, Goldman JA, Gahche JJ, Dwyer JT, Moshfegh AJ, et al. Estimation of total usual calcium and vitamin D intakes in the United States. J Nutr. 2010;140(4):817–22.
- Food and Nutrition Board, Institute of Medicine. Dietary reference intakes: applications in dietary assessment. Washington, DC: National Academies Press; 2000.
- 52. U.S. Food and Drug Administration. Guidance for industry: a food labeling guide—health claims: calcium and osteoporosis. 21 CFR 101.72. Washington, DC: Food and Drug Administration; 1993 [cited 2013 June 20]. Available from: http://www.fda.gov/Food/GuidanceRegulation/GuidanceDocumentsRegulatoryInformation/LabelingNutrition/ucm064919.htm.
- 53. Gass M, Dawson-Hughes B. Preventing osteoporosis-related fractures: an overview. Am J Med. 2006;119(4): S3–11.
- 54. Office of the Surgeon General. Bone health and osteoporosis: a report of the Surgeon General. Rockville, MD: Office of the Surgeon General, Department of Health and Human Services; 2004.
- Watts NB, Adler RA, Bilezikian JP, Drake MT, Eastell R, Orwoll ES, et al. Osteoporosis in men: an Endocrine Society clinical practice guideline. J Clin Endocrinol Metab. 2012;97(6):1802–22.
- Nordin BE. The effect of calcium supplementation on bone loss in 32 controlled trials in postmenopausal women. Osteoporos Int. 2009;20(12):2135–43.
- Rabenda V, Bruyere O, Reginster JY. Relationship between bone mineral density changes and risk of fractures among patients receiving calcium with or without vitamin D supplementation: a meta-regression. Osteoporos Int. 2011;22(3):893–901.
- Langsetmo L, Berger C, Kreiger N, Kovacs CS, Hanley DA, Jamal SA, et al. Calcium and vitamin D intake and mortality: results from the Canadian Multicentre Osteoporosis Study (CaMos). J Clin Endocrinol Metab. 2013;98:3010–8.
- 59. Posadzki P, Lee MS, Onakpoya I, Lee HW, Ko BS, Ernst E. Dietary supplements and prostate cancer: a systematic review of double-blind, placebo-controlled randomised clinical trials. Maturitas. 2013;75(2):125–30.
- 60. Age-Related Eye Disease Study Research Group. A randomized, placebo-controlled, clinical trial of high-dose supplementation with vitamins C and E, beta carotene, and zinc for age-related macular degeneration and vision loss: AREDS report no. 8. Arch Ophthalmol. 2001;119(10):1417–36.
- Tan JSL, Wang JJ, Flood V, Rochtchina E, Smith W, Mitchell P. Dietary antioxidants and the long-term incidence of age-related macular degeneration: the Blue Mountains Eye Study. Am Acad Ophthalmol. 2008;115(2):334–41.

- 62. Age-Related Eye Disease Study Group. Lutein+zeaxanthin and omega-3 fatty acids for age-related macular degeneration: the Age-Related Eye Disease Study 2 (AREDS2) randomized clinical trial. JAMA. 2013;309(19): 2005–15.
- Evans JR, Lawrenson JG. Antioxidant vitamin and mineral supplements for preventing age-related macular degeneration. Cochrane Database Syst Rev. 2012;6, CD000253.
- 64. Ma L, Dou HL, Wu YQ, Huang YM, Huang YB, Xu XR, et al. Lutein and zeaxanthin intake and the risk of agerelated macular degeneration: a systematic review and meta-analysis. Br J Nutr. 2012;107(3):350–9.
- Andres E, Fothergill H, Mecili M. Efficacy of oral cobalamin (vitamin B12) therapy. Expert Opin Pharmacother. 2010;11(2):249–56.
- Andres E, Vogel T, Federici L, Zimmer J, Kaltenbach G. Update on oral cyanocobalamin (vitamin B12) treatment in elderly patients. Drugs Aging. 2008;25(11):927–32.
- 67. Clarke R, Halsey J, Lewington S, Lonn E, Armitage J, Manson JE, et al. Effects of lowering homocysteine levels with B vitamins on cardiovascular disease, cancer, and cause-specific mortality: meta-analysis of 8 randomized trials involving 37,485 individuals. Arch Intern Med. 2010;170(18):1622–31.
- Bronstrup A, Hages M, Pietrzik K. Lowering of homocysteine concentrations in elderly men and women. Int J Vitam Nutr Res. 1999;69(3):187–93.
- Brouwer IA, van Dusseldorp M, Thomas CM, Duran M, Hautvast J, Eskes T, et al. Low-dose folic acid supplementation decreases plasma homocysteine concentrations: a randomized trial. Am J Clin Nutr. 1999;69:99–104.
- Selhub J, Jacques PF, Wilson PF, Rush D, Rosenberg IH. Vitamin status and intake as primary determinants of homocysteinemia in an elderly population. JAMA. 1993;270:2693–8.
- Lonn E, Yusuf S, Arnold MJ, Sheridan P, Pogue J, Micks M, et al. Homocysteine lowering with folic acid and B vitamins in vascular disease. N Engl J Med. 2006;354(15):1567–77.
- Bønaa KH, Njolstad I, Ueland PM, Schirmer H, Tverdal A, Steigen T, et al. Homocysteine lowering and cardiovascular events after acute myocardial infarction. N Engl J Med. 2006;354(15):1578–88.
- 73. Zhou YH, Tang JY, Wu MJ, Lu J, Wei X, Qin YY, et al. Effect of folic acid supplementation on cardiovascular outcomes: a systematic review and meta-analysis. PLoS One. 2011;6(9):e25142.
- 74. Miller III ER, Juraschek S, Pastor-Barriuso R, Bazzano LA, Appel LJ, Guallar E. Meta-analysis of folic acid supplementation trials on risk of cardiovascular disease and risk interaction with baseline homocysteine levels. Am J Cardiol. 2010;106(4):517–27.
- Huo Y, Qin X, Wang J, Sun N, Zeng Q, Xu X, et al. Efficacy of folic acid supplementation in stroke prevention: new insight from a meta-analysis. Int J Clin Pract. 2012;66(6):544–51.
- 76. Bailey RL, Dodd KW, Gahche JJ, Dwyer JT, McDowell MA, Yetley EA, et al. Total folate and folic acid intake from foods and dietary supplements in the United States: 2003–2006. Am J Clin Nutr. 2010;91(1):231–7.
- Kim YI. Folate, colorectal carcinogenesis, and DNA methylation: lessons from animal studies. Environ Mol Mutagen. 2004;44(1):10–25.
- Giovannucci E. Epidemiologic studies of folate and colorectal neoplasia: a review. J Nutr. 2002;132(8 Suppl): 2350S–5.
- Mason JB, Dickstein A, Jacques PF, Haggarty P, Selhub J, Dallal G, et al. A temporal association between folic acid fortification and an increase in colorectal cancer rates may be illuminating important biological principles: a hypothesis. Cancer Epidemiol Biomarkers Prev. 2007;16(7):1325–9.
- Stolzenberg-Solomon RZ, Chang SC, Leitzmann MF, Johnson KA, Johnson C, Buys SS, et al. Folate intake, alcohol use, and postmenopausal breast cancer risk in the Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial. Am J Clin Nutr. 2006;83(4):895–904.
- Plassman BL, Williams Jr JW, Burke JR, Holsinger T, Benjamin S. Systematic review: factors associated with risk for and possible prevention of cognitive decline in later life. Ann Intern Med. 2010;153(3):182–93.
- Institute of Medicine. Dietary reference intakes for thiamin, riboflavin, niacin, vitamin B6, folate, vitamin B12, pantothenic acid, biotin, and choline. Washington, DC: National Academies Press; 1998.
- Schneider JA, Tangney CC, Morris MC. Folic acid and cognition in older persons. Expert Opin Drug Saf. 2006;5(4):511–22.
- 84. Seshadri S, Beiser A, Selhub J, Jaques PF, Rosenberg IH, D'Agostino RB, et al. Plasma homocysteine as a risk factor for dementia and Alzheimer's disease. N Engl J Med. 2002;346:476–83.
- Ravaglia G, Forti P, Maioli F, Martelli M, Servadei L, Brunetti N, et al. Homocysteine and folate as risk factors for dementia and Alzheimer disease. Am J Clin Nutr. 2005;82(3):636–43.
- Eussen SJ, de Groot LC, Joosten LW, Bloo RJ, Clarke R, Ueland PM, et al. Effect of oral vitamin B-12 with or without folic acid on cognitive function in older people with mild vitamin B-12 deficiency: a randomized, placebocontrolled trial. Am J Clin Nutr. 2006;84(2):361–70.
- Durga J, van Boxtel MPJ, Schouten EG, Kok FJ, Jolles J, Katan MB, et al. Effect of 3-year folic acid supplementation on cognitive function in older adults in the FACIT trial: a randomised, double blind, controlled trial. Lancet. 2007;369(9557):208–16.

- Malouf R, Grimley Evans J. Folic acid with or without vitamin B12 for the prevention and treatment of healthy elderly and demented people. Cochrane Database Syst Rev. 2008;4, CD004514.
- Ford AH, Almeida OP. Effect of homocysteine lowering treatment on cognitive function: a systematic review and meta-analysis of randomized controlled trials. J Alzheimers Dis. 2012;29(1):133–49.
- 90. Jia X, McNeill G, Avenell A. Does taking vitamin, mineral and fatty acid supplements prevent cognitive decline? A systematic review of randomized controlled trials. J Hum Nutr Diet. 2008;21(4):317–36.
- Cranney A, Horsley T, O'Donnell S, Weiler H, Puil L, Ooi D, et al. Effectiveness and safety of vitamin D in relation to bone health (prepared by University of Ottawa Evidence-based Practice Center, under contract No. 290-02-0021). Evidence Report/Technology Assessment No 158. Rockville, MD: Agency for Healthcare Research and Quality; 2007
- Cranney A, Horsley T, O'Donnell S, Weiler H, Puil L, Ooi D, et al. Effectiveness and safety of vitamin D in relation to bone health. Evid Rep Technol Assess (Full Rep). 2007;(158):1–235.
- Muir SW, Montero-Odasso M. Effect of vitamin D supplementation on muscle strength, gait and balance in older adults: a systematic review and meta-analysis. J Am Geriatr Soc. 2011;59(12):2291–300.
- 94. Murad MH, Elamin KB, Abu Elnour NO, Elamin MB, Alkatib AA, Fatourechi MM, et al. Clinical review: The effect of vitamin D on falls: a systematic review and meta-analysis. J Clin Endocrinol Metab. 2011;96(10): 2997–3006.
- Kalyani RR, Stein B, Valiyil R, Manno R, Maynard JW, Crews DC. Vitamin D treatment for the prevention of falls in older adults: systematic review and meta-analysis. J Am Geriatr Soc. 2010;58(7):1299–310.
- 96. Cameron ID, Gillespie LD, Robertson MC, Murray GR, Hill KD, Cumming RG, et al. Interventions for preventing falls in older people in care facilities and hospitals. Cochrane Database Syst Rev. 2012;12, CD005465.
- Gillespie LD, Robertson MC, Gillespie WJ, Sherrington C, Gates S, Clemson LM, et al. Interventions for preventing falls in older people living in the community. Cochrane Database Syst Rev. 2012;9, CD007146.
- Stockton KA, Mengersen K, Paratz JD, Kandiah D, Bennell KL. Effect of vitamin D supplementation on muscle strength: a systematic review and meta-analysis. Osteoporos Int. 2011;22(3):859–71.
- Bischoff-Ferrari HA, Willett WC, Orav EJ, Lips P, Meunier PJ, Lyons RA, et al. A pooled analysis of vitamin D dose requirements for fracture prevention. N Engl J Med. 2012;367(1):40–9.
- 100. Ma Y, Zhang P, Wang F, Yang J, Liu Z, Qin H. Association between vitamin D and risk of colorectal cancer: a systematic review of prospective studies. J Clin Oncol. 2011;29(28):3775–82.
- Balion C, Griffith LE, Strifler L, Henderson M, Patterson C, Heckman G, et al. Vitamin D, cognition, and dementia: a systematic review and meta-analysis. Neurology. 2012;79(13):1397–405.
- Etgen T, Sander D, Bickel H, Sander K, Forstl H. Vitamin D deficiency, cognitive impairment and dementia: a systematic review and meta-analysis. Dement Geriatr Cogn Disord. 2012;33(5):297–305.
- 103. Elamin MB, Abu Elnour NO, Elamin KB, Fatourechi MM, Alkatib AA, Almandoz JP, et al. Vitamin D and cardiovascular outcomes: a systematic review and meta-analysis. J Clin Endocrinol Metab. 2011;96(7):1931–42.
- 104. Wang H, Xia N, Yang Y, Peng DQ. Influence of vitamin D supplementation on plasma lipid profiles: a metaanalysis of randomized controlled trials. Lipids Health Dis. 2012;11:42.
- 105. George PS, Pearson ER, Witham MD. Effect of vitamin D supplementation on glycaemic control and insulin resistance: a systematic review and meta-analysis. Diabet Med. 2012;29(8):e142–50.
- 106. Wu SH, Ho SC, Zhong L. Effects of vitamin D supplementation on blood pressure. South Med J. 2010;103(8): 729–37.
- Witham MD, Nadir MA, Struthers AD. Effect of vitamin D on blood pressure: a systematic review and metaanalysis. J Hypertens. 2009;27(10):1948–54.
- 108. Dereska NH, McLemore EC, Tessier DJ, Bash DS, Brophy CM. Short-term, moderate dosage vitamin E supplementation may have no effect on platelet aggregation, coagulation profile, and bleeding time in healthy individuals. J Surg Res. 2006;132(1):121–9.
- 109. Meydani SN, Meydani M, Blumberg JB, Leka LS, Pedrosa M, Diamond R, et al. Assessment of the safety of supplementation with different amounts of vitamin E in older adults. Am J Clin Nutr. 1998;68:311–8.
- 110. Morinobu T, Ban R, Yoshikawa S, Murata T, Tamai H. The safety of high-dose vitamin E supplementation in healthy Japanese male adults. J Nutr Sci Vitaminol (Tokyo). 2002;48(1):6–9.
- 111. Blumberg JB. An update: vitamin E supplementation and heart disease. Nutr Clin Care. 2002;5(2):50-5.
- 112. Traber MG. Heart disease and single-vitamin supplementation. Am J Clin Nutr. 2007;85(1):293S-9.
- 113. Yusuf S, Dagenais G, Pogue J, Bosch J, Sleight P. Vitamin E supplementation and cardiovascular events in high-risk patients. The Heart Outcomes Prevention Evaluation Study Investigators. N Engl J Med. 2000;342(3):154–60.
- 114. Heinonen OP, Albanes D, Virtamo J, Taylor PR, Huttunen JK, Hartman AM, et al. Prostate cancer and supplementation with alpha-tocopherol and beta-carotene: incidence and mortality in a controlled trial. J Natl Cancer Inst. 1998;18:440–6.
- 115. Virtamo J, Pietinen P, Huttunen JK, Korhonen P, Malila N, Virtanen MJ, et al. Incidence of cancer and mortality following alpha-tocopherol and beta-carotene supplementation: a postintervention follow-up. JAMA. 2003; 290(4):476–85.

- 116. Wright ME, Lawson KA, Weinstein SJ, Pietinen P, Taylor PR, Virtamo J, et al. Higher baseline serum concentrations of vitamin E are associated with lower total and cause-specific mortality in the Alpha-Tocopherol, Beta-Carotene Cancer Prevention Study. Am J Clin Nutr. 2006;84(5):1200–7.
- 117. Cho E, Hunter DJ, Spiegelman D, Albanes D, Beeson WL, van den Brandt PA, et al. Intakes of vitamins A, C and E and folate and multivitamins and lung cancer: a pooled analysis of 8 prospective studies. Int J Cancer. 2006;118(4):970–8.
- 118. Klein EA, Thompson IM, Lippman SM, Goodman PJ, Albanes D, Taylor PR, et al. SELECT: the next prostate cancer prevention trial. Selenium and Vitamin E Cancer Prevention Trial. J Urol. 2001;166:1311–5.
- 119. Kushi LH, Doyle C, McCullough M, Rock CL, Demark-Wahnefried W, Bandera EV, et al. American Cancer Society Guidelines on nutrition and physical activity for cancer prevention: reducing the risk of cancer with healthy food choices and physical activity. CA Cancer J Clin. 2012;62(1):30–67.
- 120. U.S. Department of Agriculture and U.S. Department of Health and Human Services. Dietary guidelines for Americans, 2010 7th Edition. Washington, DC: 2010 December 2010. US Government Printing Office.
- 121. Lichtenstein AH, Appel LJ, Brands M, Carnethon M, Daniels S, Franch HA, et al. Diet and lifestyle recommendations revision 2006: a scientific statement from the American Heart Association Nutrition Committee. Circulation. 2006;114:82–96.
- 122. Appel LJ, Brands MW, Daniels SR, Karanja N, Elmer PJ, Sacks FM. Dietary approaches to prevent and treat hypertension: a scientific statement from the American Heart Association. Hypertension. 2006;47(2):296–308.
- 123. Institute of Medicine. Dietary reference intakes for vitamin C, vitamin E, selenium, and carotenoids. Washington, DC: National Academies Press; 2000.
- 124. U.S. Department of Agriculture, Agricultural Research Service. What we eat in America, 2009–2010 [updated July 16, 2012]. Available from: http://www.ars.usda.gov/Services/docs.htm?docid=18349.
- 125. U.S. Food and Drug Administration. Qualified health claims subject to enforcement discretion: selenium & cancer, Docket No. 02P-0457. Washington, DC: Food and Drug Administration; 2003 [cited 2013 June 20]. Available from: http://www.fda.gov/Food/IngredientsPackagingLabeling/LabelingNutrition/ucm073992.htm#selenium.
- Piekutowski K, Makarewicz R, Zachara BA. The antioxidative role of selenium in pathogenesis of cancer of the female reproductive system. Neoplasma. 2007;54:374–8.
- 127. Chen Y, Hall M, Graziano JH, Slavkovich V, van Geen A, Parvez F, et al. A prospective study of blood selenium levels and the risk of arsenic-related premalignant skin lesions. Cancer Epidemiol Biomarkers Prev. 2007;16(2):207–13.
- 128. Brinkman M, Reulen RC, Kellen E, Buntinx F, Zeeers M. Are men with low selenium levels at increased risk of prostate cancer? Eur J Cancer. 2006;42:2463–71.
- 129. Lippman SM, Klein EA, Goodman PJ, Lucia MS, Thompson IM, Ford LG, et al. Effect of selenium and vitamin E on risk of prostate cancer and other cancers: the Selenium and Vitamin E Cancer Prevention Trial (SELECT). JAMA. 2009;301(1):39–51.
- Nicastro HL, Dunn BK. Selenium and prostate cancer prevention: insights from the selenium and vitamin E cancer prevention trial (SELECT). Nutrients. 2013;5(4):1122–48.
- 131. Bleys J, Navas-Acien A, Guallar E. Serum selenium and diabetes in U.S. adults. Diabetes Care. 2007;30(4):829–34.
- 132. Stranges S, Marshall JR, Trevisan M, Natarajan R, Donahue RP, Combs GF, et al. Effects of selenium supplementation on cardiovascular disease incidence and mortality: secondary analyses in a randomized clinical trial. Am J Epidemiol. 2006;163:694–9.
- 133. Stranges S, Marshall JR, Natarajan R, Donahue RP, Trevisan M, Combs GF, et al. Effects of long-term selenium supplementation on the incidence of type 2 diabetes: a randomized trial. Ann Intern Med. 2007;147:217–23.
- 134. Hodge W, Barnes D, Schachter H, Pan Y, Lowcock E, Zhang L, et al. Effects of omega-3 fatty acids on eye health. Rockville, MD: Agency for Healthcare Research and Quality; 2005. Contract No.: AHRQ Publication No. 04-E012-2.
- 135. Lewin GA, Schachter HM, Yuen D, Merchant P, Mamaladze V, Tsertsvadze A. Effects of omega-3 fatty acids on child and maternal health. Rockville, MD: Agency for Healthcare Research and Quality; 2005. Contract No.: AHRQ Publication No. 05-E025-2.
- 136. Schachter H, Kourad K, Merali Z, Lumb A, Tran K, Miguelez M, et al. Effects of omega-3 fatty acids on mental health. Rockville, MD: Agency for Healthcare Research and Quality; 2005. Contract No.: AHRQ Publication No. 05-E022-2.
- 137. Schachter H, Reisman J, Tran K, Dales B, Kourad K, Barnes D, et al. Health effects of omega-3 fatty acids on asthma. Rockville, MD: 2004 Contract No.: AHRQ Publication No. 04-E013-2.
- 138. Maclean CH, Issa AM, Newberry SJ, Mojica WA, Morton SC, Garland RH, et al. Effects of omega-3 fatty acids on cognitive function with aging, dementia, and neurological diseases. Evid Rep Technol Assess (Summ). 2005;(114):1–3.
- 139. MacLean CH, Mojica WA, Morton SC, Pencharz J, Hasenfeld Garland R, Tu W, et al. Effects of omega-3 fatty acids on lipids and glycemic control in type II diabetes and the metabolic syndrome and on inflammatory bowel disease, rheumatoid arthritis, renal disease, systemic lupus erythematosus, and osteoporosis. Rockville, MD: Agency for Healthcare Research and Quality; 2004. Contract No.: AHRQ Publication No. 04-E012-2.
- 140. MacLean CH, Newberry SJ, Mojica WA, Issa A, Khanna P, Lim YW, et al. Effects of omega-3 fatty acids on cancer. Rockville, MD: Agency for Healthcare Research and Quality; 2005. Contract No.: AHRQ Publication No. 05-E010-2.
- 141. MacLean CH, Newberry SJ, Mojica WA, Khanna P, Issa AM, Suttorp MJ, et al. Effects of omega-3 fatty acids on cancer risk: a systematic review. JAMA. 2006;295(4):403–15.
- 142. Balk E, Chung M, Lichtenstein A, Chew P, Kupelnick B, Lawrence A, et al. Effects of omega-3 fatty acids on cardiovascular risk factors and intermediate markers of cardiovascular disease. Rockville, MD: Agency for Healthcare Research and Quality; 2004. Contract No.: AHRQ Publication No. 04-E010-2.
- 143. Jordan H, Matthan N, Chung M, Balk E, Chew P, Kupelnick B, et al. Effects of omega-3 fatty acids on arrhythmogenic mechanisms in animal and isolated organ/cell culture studies. Rockville, MD: Agency for Healthcare Research and Quality; 2004. Contract No.: AHRQ Publication No 04-E011-2.
- 144. Wang C, Chung M, Lichtenstein A, Balk E, Kupelnick B, DeVine D, et al. Effects of omega-3 fatty acids on cardiovascular disease. Rockville, MD: Agency for Healthcare Research and Quality; 2004. Contract No.: AHRQ Publication No. 04-E009-2.
- 145. Calder PC. Polyunsaturated fatty acids and inflammation. Prostaglandins Leukot Essent Fatty Acids. 2006;75(3): 197–202.
- de Lorgeril M. Essential polyunsaturated fatty acids, inflammation, atherosclerosis and cardiovascular diseases. Subcell Biochem. 2007;42:283–97.
- 147. Gruppo Italiano per lo Studio della Sopravvivenza nell'Infarto miocardico. Dietary supplementation with n-3 polyunsaturated fatty acids and vitamin E after myocardial infarction: results of the GISSI-Prevenzione trial. Gruppo Italiano per lo Studio della Sopravvivenza nell'Infarto miocardico. Lancet. 1999;354:447–55.
- 148. Yokoyama M, Origasa H, Matsuzaki M, Matsuzawa Y, Saito Y, Ishikawa Y, et al. Effects of eicosapentaenoic acid on major coronary events in hypercholesterolaemic patients (JELIS): a randomised open-label, blinded endpoint analysis. Lancet. 2007;369:1090–8.
- 149. Hooper L, Thompson RL, Harrison RA, Summerbell CD, Ness AR, Moore HJ, et al. Risks and benefits of omega 3 fats for mortality, cardiovascular disease, and cancer: systematic review. BMJ. 2006;332:752–60.
- 150. Wang Q, Liang X, Wang L, Lu X, Huang J, Cao J, et al. Effect of omega-3 fatty acids supplementation on endothelial function: a meta-analysis of randomized controlled trials. Atherosclerosis. 2012;221(2):536–43.
- 151. Khawaja O, Gaziano JM, Djousse L. A meta-analysis of omega-3 fatty acids and incidence of atrial fibrillation. J Am Coll Nutr. 2012;31(1):4–13.
- 152. Rizos EC, Ntzani EE, Bika E, Kostapanos MS, Elisaf MS. Association between omega-3 fatty acid supplementation and risk of major cardiovascular disease events: a systematic review and meta-analysis. JAMA. 2012;308(10):1024–33.
- 153. Kwak SM, Myung SK, Lee YJ, Seo HG. Efficacy of omega-3 fatty acid supplements (eicosapentaenoic acid and docosahexaenoic acid) in the secondary prevention of cardiovascular disease: a meta-analysis of randomized, double-blind, placebo-controlled trials. Arch Intern Med. 2012;172(9):686–94.
- 154. McKenney JM, Sica D. Role of prescription omega-3 fatty acids in the treatment of hypertriglyceridemia. Pharmacotherapy. 2007;5:715–28.
- Kris-Etherton PM, Harris WS, Appel LJ. Fish consumption, fish oil, omega-3 fatty acids, and cardiovascular disease. Arterioscler Thromb Vasc Biol. 2003;23(2):e20–30.
- 156. Smith Jr SC, Allen J, Blair SN, Bonow RO, Brass LM, Fonarow GC, et al. AHA/ACC guidelines for secondary prevention for patients with coronary and other atherosclerotic vascular disease: 2006 update: endorsed by the National Heart, Lung, and Blood Institute. Circulation. 2006;113(19):2363–72.
- 157. National Heart, Lung, and Blood Institute. Your guide to lowering Your cholesterol with TLC: National Heart, Lung, and Blood Institute; 2006 [cited 2013 June 20]. Available from: http://www.nhlbi.nih.gov/health/public/heart/chol/chol_tlc.pdf.
- 158. Fihn SD, Gardin JM, Abrams J, Berra K, Blankenship JC, Dallas AP, et al. ACCF/AHA/ACP/AATS/PCNA/SCAI/ STS guideline for the diagnosis and management of patients with stable ischemic heart disease: a report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines, and the American College of Physicians, American Association for Thoracic Surgery, Preventive Cardiovascular Nurses Association, Society for Cardiovascular Angiography and Interventions, and Society of Thoracic Surgeons. J Am Coll Cardiol. 2012;60(24):e44–164.
- Larsson SC, Kumlin M, Ingelman-Sundberg J, Wolk A. Dietary long-chain n-3 fatty acids for the prevention of cancer: a review of potential mechanisms. Am J Clin Nutr. 2004;79(6):935–45.
- 160. Pardini RS. Nutritional intervention with omega-3 fatty acids enhances tumor response to anti-neoplastic agents. Chem Biol Interact. 2006;162(2):89–105.
- 161. Shen XJ, Zhou JD, Dong JY, Ding WQ, Wu JC. Dietary intake of n-3 fatty acids and colorectal cancer risk: a metaanalysis of data from 489,000 individuals. Br J Nutr. 2012;108(9):1550–6.
- 162. Szymanski KM, Wheeler DC, Mucci LA. Fish consumption and prostate cancer risk: a review and meta-analysis. Am J Clin Nutr. 2010;92(5):1223–33.

- 163. Carayol M, Grosclaude P, Delpierre C. Prospective studies of dietary alpha-linolenic acid intake and prostate cancer risk: a meta-analysis. Cancer Causes Control. 2010;21(3):347–55.
- 164. Wu S, Liang J, Zhang L, Zhu X, Liu X, Miao D. Fish consumption and the risk of gastric cancer: systematic review and meta-analysis. BMC Cancer. 2011;11:26.
- 165. Mazereeuw G, Lanctot KL, Chau SA, Swardfager W, Herrmann N. Effects of omega-3 fatty acids on cognitive performance: a meta-analysis. Neurobiol Aging. 2012;33(7):1482 e17–29.
- 166. Sydenham E, Dangour AD, Lim WS. Omega 3 fatty acid for the prevention of cognitive decline and dementia. Cochrane Database Syst Rev. 2012;6, CD005379.
- 167. Klein MA, Nahin RL, Messina MJ, Rader JI, Thompson LU, Badger TM, et al. Guidance from an NIH workshop on designing, implementing, and reporting clinical studies of soy interventions. J Nutr. 2010;140(6):1192S–204.
- 168. Sacks FM, Lichtenstein A, Van Horn L, Harris W, Kris-Etherton P, Winston M. Soy protein, isoflavones, and cardiovascular health: an American Heart Association Science Advisory for professionals from the Nutrition Committee. Circulation. 2006;113(7):1034–44.
- 169. Anderson JW, Bush HM. Soy protein effects on serum lipoproteins: a quality assessment and meta-analysis of randomized, controlled studies. J Am Coll Nutr. 2011;30(2):79–91.
- 170. Harland JI, Haffner TA. Systematic review, meta-analysis and regression of randomised controlled trials reporting an association between an intake of circa 25 g soya protein per day and blood cholesterol. Atherosclerosis. 2008;200(1):13–27.
- 171. Taku K, Umegaki K, Ishimi Y, Watanabe S. Effects of extracted soy isoflavones alone on blood total and LDL cholesterol: meta-analysis of randomized controlled trials. Ther Clin Risk Manag. 2008;4(5):1097–103.
- 172. Hodis HN, Mack WJ, Kono N, Azen SP, Shoupe D, Hwang-Levine J, Petitti D, Whitfield-Maxwell L, Yan M, Franke AA, Selzer RH for the WISH Research Group. Isoflavone soy protein supplementation and atherosclerosis progression in healthy postmenopausal women: a randomized controlled trial. Stroke. 2011;42(11):3168–75.
- 173. Henderson VW, St John JA, Hodis HN, Kono N, McCleary CA, Franke AA, Mack WJ for the WISH Research Group. Long term soy isoflavone supplementation and cognition in women: a randomized controlled trial. Neurology. 2012;78:1841–8.
- 174. Hendrix SL, Wassertheil-Smoller S, Johnson KC, Howard BV, Kooperberg C, Rossouw JE, et al. Effects of conjugated equine estrogen on stroke in the Women's Health Initiative. Circulation. 2006;113(20):2425–34.
- 175. Stefanick ML, Anderson GL, Margolis KL, Hendrix SL, Rodabough RJ, Paskett ED, et al. Effects of conjugated equine estrogens on breast cancer and mammography screening in postmenopausal women with hysterectomy. JAMA. 2006;295(14):1647–57.
- 176. Liu J, Ho SC, Su YX, Chen WQ, Zhang CX, Chen YM. Effect of long-term intervention of soy isoflavones on bone mineral density in women: a meta-analysis of randomized controlled trials. Bone. 2009;44(5):948–53.
- 177. Taku K, Melby MK, Nishi N, Omori T, Kurzer MS. Soy isoflavones for osteoporosis: an evidence-based approach. Maturitas. 2011;70(4):333–8.
- 178. Wei P, Liu M, Chen Y, Chen DC. Systematic review of soy isoflavone supplements on osteoporosis in women. Asian Pac J Trop Med. 2012;5(3):243–8.
- 179. Ricci E, Cipriani S, Chiaffarino F, Malvezzi M, Parazzini F. Soy isoflavones and bone mineral density in perimenopausal and postmenopausal Western women: a systematic review and meta-analysis of randomized controlled trials. J Womens Health (Larchmt). 2010;19(9):1609–17.
- North American Menopause Society. Management of osteoporosis in postmenopausal women: 2010 position statement of The North American Menopause Society. Menopause. 2010;17(1):25–54. quiz 5–6.
- North American Menopause Society. The 2012 hormone therapy position statement of: The North American Menopause Society. Menopause. 2012;19(3):257–71.
- 182. Hooper L, Ryder JJ, Kurzer MS, Lampe JW, Messina MJ, Phipps WR, et al. Effects of soy protein and isoflavones on circulating hormone concentrations in pre- and post-menopausal women: a systematic review and metaanalysis. Hum Reprod Update. 2009;15(4):423–40.
- 183. Jacobs A, Wegewitz U, Sommerfeld C, Grossklaus R, Lampen A. Efficacy of isoflavones in relieving vasomotor menopausal symptoms—a systematic review. Mol Nutr Food Res. 2009;53(9):1084–97.
- Bolanos R, Del Castillo A, Francia J. Soy isoflavones versus placebo in the treatment of climacteric vasomotor symptoms: systematic review and meta-analysis. Menopause. 2010;17(3):660–6.
- 185. Taku K, Melby MK, Kronenberg F, Kurzer MS, Messina M. Extracted or synthesized soybean isoflavones reduce menopausal hot flash frequency and severity: systematic review and meta-analysis of randomized controlled trials. Menopause. 2012;19(7):776–90.
- 186. Blumenthal M, Busse WR, Goldberg A, Gruenwald J, Hall T, Riggins CW, et al., editors. Complete German Commission E monographs: therapeutic guide to herbal medicines. Austin, TX: American Botanical Council; 1998.
- 187. Burdette JE, Liu J, Chen SN, Fabricant DS, Piersen CE, Barker EL, et al. Black cohosh acts as a mixed competitive ligand and partial agonist of the serotonin receptor. J Agric Food Chem. 2003;51(19):5661–70.

- 188. Maki PM, Rubin LH, Fornelli D, Drogos L, Banuvar S, Shulman LP, et al. Effects of botanicals and combined hormone therapy on cognition in postmenopausal women. Menopause. 2009;16(6):1167–77.
- 189. Geller SE, Shulman LP, van Breemen RB, Banuvar S, Zhou Y, Epstein G, et al. Safety and efficacy of black cohosh and red clover for the management of vasomotor symptoms: a randomized controlled trial. Menopause. 2009;16(6):1156–66.
- Newton KM, Reed SD, LaCroix AZ, Grothaus LC, Ehrlich K, Guiltinan J. Treatment of vasomotor symptoms of menopause with black cohosh, multibotanicals, soy, hormone therapy, or placebo: a randomized trial. Ann Intern Med. 2006;145(12):869–79.
- 191. Shams T, Setia MS, Hemmings R, McCusker J, Sewitch M, Ciampi A. Efficacy of black cohosh-containing preparations on menopausal symptoms: a meta-analysis. Altern Ther Health Med. 2010;16(1):36–44.
- 192. van der Sluijs CP, Bensoussan A, Chang S, Baber R. A randomized placebo-controlled trial on the effectiveness of an herbal formula to alleviate menopausal vasomotor symptoms. Menopause. 2009;16(2):336–44.
- 193. Leach MJ, Moore V. Black cohosh (Cimicifuga spp.) for menopausal symptoms. Cochrane Database Syst Rev. 2012;9, CD007244.
- 194. Amsterdam JD, Yao Y, Mao JJ, Soeller I, Rockwell K, Shults J. Randomized, double-blind, placebo-controlled trial of Cimicifuga racemosa (black cohosh) in women with anxiety disorder due to menopause. J Clin Psychopharmacol. 2009;29(5):478–83.
- 195. Schulz V, Hansel R, Tyler VE. Rational phytotherapy: a physician's guide to herbal medicine. 4th ed. Berlin: Springer; 2001.
- Therapeutic Research Faculty. Natural Medicines Comprehensive Database [Internet]. Stockton, CA: Therapeutic Research Center; 1995–2013 [cited 2013 June 20]. Available from: http://www.naturaldatabase.com. Subscription required.
- 197. Koch E, Biber A. Pharmacological effects of sabal and urtica extracts as basis for a rational drug therapy in benign prostate hyperplasia. Urologe. 1994;34(2):90–5.
- Tacklind J, Macdonald R, Rutks I, Stanke JU, Wilt TJ. Serenoa repens for benign prostatic hyperplasia. Cochrane Database Syst Rev. 2012;12, CD001423.
- Bent S, Kane C, Shinohara K, Neuhaus J, Hudes ES, Goldberg H, et al. Saw palmetto for benign prostatic hyperplasia. N Engl J Med. 2006;354(6):557–66.
- 200. Barry MJ, Meleth S, Lee JY, Kreder KJ, Avins AL, Nickel JC, et al. Effect of increasing doses of saw palmetto extract on lower urinary tract symptoms: a randomized trial. JAMA. 2011;306(12):1344–51.
- 201. Agbabiaka TB, Pittler MH, Wider B, Ernst E. Serenoa repens (saw palmetto): a systematic review of adverse events. Drug Saf. 2009;32(8):637–47.
- 202. Avins AL, Bent S, Staccone S, Badua E, Padula A, Goldberg H, et al. A detailed safety assessment of a saw palmetto extract. Complement Ther Med. 2008;16(3):147–54.
- Oken BS, Storzbach DM, Kaye JA. The efficacy of Ginkgo biloba on cognitive function in Alzheimer's disease. Arch Neurol. 1998;55:1409–15.
- 204. Ernst E, Pittler MH. Ginkgo biloba for dementia: a systematic review of double-blind placebo controlled trials. Clin Drug Invest. 1999;17:301–8.
- 205. DeKosky ST, Williamson JD, Fitzpatrick AL, Kronmal RA, Ives DG, Saxton JA, et al. Ginkgo biloba for prevention of dementia: a randomized controlled trial. JAMA. 2008;300(19):2253–62.
- 206. Nahin RL, Fitzpatrick AL, Williamson JD, Burke GL, DeKosky ST, Furberg C, et al. Use of herbal medicine and other dietary supplements in community-dwelling older people: baseline data from the ginkgo evaluation of memory study. J Am Geriatr Soc. 2006;54:1725–35.
- 207. Vellas B, Andrieu S, Ousset PJ, Ouzid M, Mathiex-Fortunet H, GuidAge Study Group. The GuidAge study: methodological issues. A 5-year double-blind randomized trial of the efficacy of EGb 761 for prevention of Alzheimer disease in patients over 70 with a memory complaint. Neurology. 2006;67(9 Suppl 3):S6–11.
- 208. Vellas B, Coley N, Ousset PJ, Berrut G, Dartigues JF, Dubois B, et al. Long-term use of standardised Ginkgo biloba extract for the prevention of Alzheimer's disease (GuidAge): a randomised placebo-controlled trial. Lancet Neurol. 2012;11(10):851–9.
- 209. Birks J, Evans Grimley J. Ginkgo biloba for cognitive impairment and dementia. Cochrane Database Syst Rev. 2009;1, CD003120.
- Laws KR, Sweetnam H, Kondel TK. Is Ginkgo biloba a cognitive enhancer in healthy individuals? A metaanalysis. Hum Psychopharmacol. 2012;27(6):527–33.
- 211. Wang BS, Wang H, Song YY, Qi H, Rong ZX, Wang BS, et al. Effectiveness of standardized Ginkgo biloba extract on cognitive symptoms of dementia with a six-month treatment: a bivariate random effect meta-analysis. Pharmacopsychiatry. 2010;43(3):86–91.
- 212. Weinmann S, Roll S, Schwarzbach C, Vauth C, Willich SN. Effects of Ginkgo biloba in dementia: systematic review and meta-analysis. BMC Geriatr. 2010;10:14.
- He D, Zhou H, Guo D, Hao Z, Wu B. Pharmacologic treatment for memory disorder in multiple sclerosis. Cochrane Database Syst Rev. 2011;10, CD008876.

- Engelsen J, Nielsen JD, Winther K. Effect of coenzyme Q10 and Ginkgo biloba on warfarin dosage in stable, longterm warfarin treated outpatients. A randomised, double blind, placebo-crossover trial. Thromb Haemost. 2002;87(6):1075–6.
- 215. Wolf HR. Does Ginkgo biloba special extract EGb 761 provide additional effects on coagulation and bleeding when added to acetylsalicylic acid 500 mg daily? Drugs R D. 2006;7(3):163–72.
- Kellermann AJ, Kloft C. Is there a risk of bleeding associated with standardized Ginkgo biloba extract therapy? A systematic review and meta-analysis. Pharmacotherapy. 2011;31(5):490–502.
- 217. Towheed TE, Maxwell L, Anastassiades TP, Shea B, Houpt J, Welch V, et al. Glucosamine therapy for treating osteoarthritis. Cochrane Database Syst Rev. 2009;4, CD002946. doi:10.1002/14651858.CD002946.pub2.
- 218. Sawitzke AD, Shi H, Finco MF, Dunlop DD, Harris CL, Singer NG, et al. Clinical efficacy and safety of glucosamine, chondroitin sulphate, their combination, celecoxib or placebo taken to treat osteoarthritis of the knee: 2-year results from GAIT. Ann Rheum Dis. 2010;69(8):1459–64.
- Wilkens P, Scheel IB, Grundnes O, Hellum C, Storheim K. Effect of glucosamine on pain-related disability in patients with chronic low back pain and degenerative lumbar osteoarthritis: a randomized controlled trial. JAMA. 2010;304(1):45–52.
- Stuber K, Sajko S, Kristmanson K. Efficacy of glucosamine, chondroitin, and methylsulfonylmethane for spinal degenerative joint disease and degenerative disc disease: a systematic review. J Can Chiropr Assoc. 2011;55(1): 47–55.
- 221. Wandel S, Juni P, Tendal B, Nuesch E, Villiger PM, Welton NJ, et al. Effects of glucosamine, chondroitin, or placebo in patients with osteoarthritis of hip or knee: network meta-analysis. BMJ. 2010;341:c4675.
- 222. Lee YH, Woo JH, Choi SJ, Ji JD, Song GG. Effect of glucosamine or chondroitin sulfate on the osteoarthritis progression: a meta-analysis. Rheumatol Int. 2010;30(3):357–63.
- 223. Black C, Clar C, Henderson R, MacEachern C, McNamee P, Quayyum Z, et al. The clinical effectiveness of glucosamine and chondroitin supplements in slowing or arresting progression of osteoarthritis of the knee: a systematic review and economic evaluation. Health Technol Assess. 2009;13(52):1–148.
- 224. Zhang W, Nuki G, Moskowitz RW, Abramson S, Altman RD, Arden NK, et al. OARSI recommendations for the management of hip and knee osteoarthritis: part III: Changes in evidence following systematic cumulative update of research published through January 2009. Osteoarthritis Cartilage. 2010;18(4):476–99.
- Richmond J, Hunter D, Irrgang J, Jones MH, Levy B, Marx R, et al. Treatment of osteoarthritis of the knee (nonarthroplasty). J Am Acad Orthop Surg. 2009;17(9):591–600.
- 226. Costello RB, Leser M, Coates PM. Dietary supplements: current knowledge and future frontiers. In: Bales CW, Ritchie CS, editors. Handbook of clinical nutrition and aging. 2nd ed. New York, NY: Humana Press; 2009. p. 553–633.
- Rosenfeldt F, Hilton D, Pepe S, Krum H. Systematic review of effect of coenzyme Q10 in physical exercise, hypertension and heart failure. Biofactors. 2003;18(1–4):91–100.
- 228. Sander S, Coleman CI, Patel AA, Kluger J, White CM. The impact of coenzyme Q10 on systolic function in patients with chronic heart failure. J Card Fail. 2006;12(6):464–72.
- Fotino AD, Thompson-Paul AM, Bazzano LA. Effect of coenzyme Q10 supplementation on heart failure: a metaanalysis. Am J Clin Nutr. 2013;97(2):268–75.
- 230. Davidson M, McKenney J, Stein E, Schrott H, Bakker-Arkena R, Fayyad R, et al. Comparison of one-year efficacy and safety of atorvastatin versus lovastatin in primary hypercholesterolemia. Am J Cardiol. 1997;79(11): 1475–81.
- 231. Caso G, Kelly P, McNurlan MA, Lawson WE. Effect of coenzyme Q10 on myopathic symptoms in patients treated with statins. Am J Cardiol. 2007;99:1409–12.
- 232. Marcoff L, Thompson PD. The role of coenzyme Q10 in statin-associated myopathy: a systematic review. J Am Coll Cardiol. 2007;49:2231–7.
- Bookstaver DA, Burkhalter NA, Hatzigeorgiou C. Effect of coenzyme Q10 supplementation on statin-induced myalgias. Am J Cardiol. 2012;110(4):526–9.
- 234. Ho MJ, Bellusci A, Wright JM. Blood pressure lowering efficacy of coenzyme Q10 for primary hypertension. Cochrane Database Syst Rev. 2009;4, CD007435.
- 235. Gao L, Mao Q, Cao J, Wang Y, Zhou X, Fan L. Effects of coenzyme Q10 on vascular endothelial function in humans: a meta-analysis of randomized controlled trials. Atherosclerosis. 2012;221(2):311–6.
- 236. Young JM, Florkowski CM, Molyneux SL, McEwan RG, Frampton CM, Nicholls MG, et al. A randomized, double-blind, placebo-controlled crossover study of coenzyme Q10 therapy in hypertensive patients with the metabolic syndrome. Am J Hypertens. 2012;25(2):261–70.
- 237. Dai YL, Luk TH, Yiu KH, Wang M, Yip PM, Lee SW, et al. Reversal of mitochondrial dysfunction by coenzyme Q10 supplement improves endothelial function in patients with ischaemic left ventricular systolic dysfunction: a randomized controlled trial. Atherosclerosis. 2011;216(2):395–401.
- Shults CW, Oakes D, Kieburtz K, Beal MF, Haas R, Plumb S, et al. Effects of coenzyme Q10 in early Parkinson disease: evidence of slowing of the functional decline. Arch Neurol. 2002;59(10):1541–50.

- 239. Shults CW, Flint Beal M, Song D, Fontaine D. Pilot trial of high dosages of coenzyme Q10 in patients with Parkinson's disease. Exp Neurol. 2004;188(2):491–4.
- 240. Storch A, Jost WH, Vieregge P, Spiegel J, Greulich W, Durner J, et al. Randomized, double-blind, placebo-controlled trial on symptomatic effects of coenzyme Q(10) in Parkinson disease. Arch Neurol. 2007;64(7):938–44.
- 241. Liu J, Wang L, Zhan SY, Xia Y. Coenzyme Q10 for Parkinson's disease. Cochrane Database Syst Rev. 2011;12, CD008150.
- 242. Cohen MH, Hrbek A, Davis RB, Schachter SC, Eisenberg DM. Emerging credentialing practices, malpractice liability policies, and guidelines governing complementary and alternative medical practices and dietary supplement recommendations: a descriptive study of 19 integrative health care centers in the United States. Arch Intern Med. 2005;165(3):289–95.
- Bazzie KL, Witmer DR, Pinto B, Bush C, Clark J, Deffenbaugh Jr J. National survey of dietary supplement policies in acute care facilities. Am J Health Syst Pharm. 2006;63(1):65–70.

A

Abusabha, R., 9 Acid-base balance, diet acidic environment bone, negative effects, 280 muscles, negative effects, 280 for bone development and preservation, 274 metabolic acidosis, 280 NAE values, 280-281 potassium bicarbonate, 281 ACSM Position Stand: Exercise and Physical Activity for Older Adults, 365-366 Activity of daily living (ADL), 44, 288 Administration on Aging (AoA), 368 Adverse Event Reporting (AER), 379, 408 Aerobic exercise, 106, 119, 316, 319, 360, 364-366 Agency for Healthcare Research and Quality (AHRQ) review, 22-24, 389-390, 396-397 Age-Related Eye Disease Study (AREDS), 64, 65, 67, 70-72, 385-386 Age-related macular degeneration (AMD) abnormal blood vessels, 58 blindness, 58 and cataracts, 58-60 components, 58 early and intermediate stages, 58 for eye health, 341, 379, 385 and lens opacities, 385 lutein and zeaxanthin, 65-66 modifiable risk factors, 58 nutrient supplement effects AREDS study, 70-71 epidemiologic studies, 71 lutein and zeaxanthin, 72 vitamin C, 71 vitamin E, 71 zinc, 72 omega-3 fatty acids, 66-68 prevalence, 58 restore vision, 60 retina (macula), 59

retinal pigment epithelium (RPE), 59 vitamin C, 64, 65 vitamin E. 64, 65 zinc, 67-68 Aging AD. 133 anti-aging effects (see Calorie restriction (CR)) biomarkers, 131, 321-322 cognitive function, 126-127 high-risk nutrients (see Nutrient intake) knowledge and procedural memory, 125 mediterranean diet, 136 monounsaturated and polyunsaturated fatty acids, 135 and polyphenols, 133 successful agers, 13 usual. 13 vitamin D, 131, 277 Aglycone isoflavone equivalents, 398 Allman, R., 239 Alpha-Tocopherol and Beta-Carotene trial, 71 Alzheimer's disease (AD) aging, 133 APOE £4, 129 B-complex vitamins, 128, 388-389 dementia patients, 192, 288, 306-307 FAST, 306 mediterranean diet, 137 omega-3 fatty acids, 339, 397-398 polyphenols, 132 primary neurological symptom, 136 vitamin D, 131, 391 vitamin E supplements, 345 Amella, E., 9 Amella, E.J., 287-299 American Association of Clinical Endocrinologists (AACE), 186 American College of Sports Medicine (ACSM), 104, 365-367 American Diabetes Association (ADA), 181, 183-187, 194 American Geriatric Society (AGS), 183 Amyotrophic lateral sclerosis (ALS), 307

C.W. Bales et al. (eds.), *Handbook of Clinical Nutrition and Aging*, Nutrition and Health, DOI 10.1007/978-1-4939-1929-1, © Springer Science+Business Media New York 2015

Analytic framework. See also Systematic reviews dietary reference intakes, 22, 23 organizational structure, 22 vitamin D and calcium, 23, 24 Annweiler, C., 131, 132 Anthony, P.S., 239 Anthropometrics, 40-41, 228, 238, 290 Apolipoprotein E (APOE) ɛ4 genotype, 129 Arnarsson, A., 63 Arthritis, 167, 204 Artificial nutrition/hydration (ANH) terminal cancer, 305-306 terminal dementia alternative approaches, 307 comfort feeding only, 307 costs and benefits, 306-307 Aselage, M.B., 287-299 Atrophic gastritis, 343 Auditory Consonant Trigram (ACT), 328

B

Baldwin, C., 254 Bales, C.W., 287-299 Banks, M., 168 Barrett, M., 410 Bartholomew, L.K., 6 Basal metabolic rate (BMR), 220, 221 Baumgartner, R.N., 101, 103 B-complex vitamins cognition, 388-389 CVD, 387 mortality, 386-387 risks of excess amounts, 387-388 Beatty, S., 66 Beddhu, S., 261-268 Behavioral Risk Factor Surveillance System (BRFSS), 364.366 Behavior modification, obesity, 172-173 Benign prostatic hyperplasia (BPH), 402–403 Bennett, D.A., 239 Benton Visual Retention Test (BVRT), 328 Berg, A.C., 354-372 Bergenstal, R.M., 184 Bergstrom, N., 239 Berlowitz, D.R., 239 Bernstein, M., 8 Bernstein, P.S., 66 Betz, J.M., 410 Bioavailability, 28, 130 Bioequivalence, 29 Blackman, M.R., 410 Blanc, S., 316 Blood glucose, DM cognitive impairment, 192 food and insulin intake, 191 ICU, 186 management, 182 and metabolic abnormalities, 181 metabolic syndrome, 188 micro and macrovascular disease, 182

non-Hispanic whites, 180 nutrition care plans, 187 postprandial, 183 SMBG, 189 type 1 diabetes, 189 Blood nutrient status, 90-91 Blueberries, 133 Blue Mountains Eye Study, 63, 386 Blumenthal, M., 410, 411 BMD. See Bone mineral density (BMD) Body mass index (BMI) frailty, 169 overweight and obesity, 165, 166, 223, 263-264 periodontitis, 86 recommendations, older adults, 42 unintentional weight loss, 41, 238 uremia, 265 Body Shape Questionnaire (BSQ), 327 Bolland, M.J., 276 Bone, K., 410 Bone mass, 274, 277–279 Bone mineral density (BMD) age-related bone loss, 274 calcium and vitamin D, combinations, 384 fractures and bone mass, vitamin D, 278 increased body weight, 169 osteoporosis, nutritional concerns, 274, 400 PA/EX effects, 362 protein intakes, 337 weight loss in older adults, 170, 171 Bone, R.A., 66 Botanicals. See Herbals and botanicals Bouchard, D.R., 99-108 Bourdel-Marchasson, I., 239 Bozzetti, F., 257 Brady, A.O., 354-372 Brandeis, 239 Breslow, R.A., 239, 240 Bridge to heart transplantation (BTT), 218 Brief Inventory for Mental State (BIMS), 293 Brinkmann, J., 410 Briscoe, V.J., 195 Brown, B., 129 Brown, L., 64 Bruera, E., 255 B-type natriuretic peptide (BNP), 221 Buse, J.B., 184 Buys, D.R., 147-157 **B** vitamins Alzheimer's disease, 128 and APOE £4 genotype, 129 and cognitive function, 128 deficiencies, 127 homocysteine, 127-128 MTHFR genotype, 129-130 one-carbon metabolism cycle, 127 SAM, 127 supplementation, 128 vitamin B₆, 341-342 vitamin B₁₂, 342-343

С

Cachexia cancer. 305-306 cardiac (see Cardiac cachexia) cytokine-induced, 247 inflammatory-mediated, 245, 247 interventions, 247 Calcium minerals, 346 osteoporosis intake requirements, 275 physiology, 274-275 recommendations, 276 safety, 276 phosphorus, 267-268 and vitamin D, combinations bone mineral density, 384 dietary reference intakes (DRIs), 383 fractures, 384 mortality and other outcomes, 384-385 osteoporosis, 383-384 CALERIE (Comprehensive Assessment of the Long-term Effect of Reducing Intake of Energy), 318-321, 323, 326-328 Calorie restriction (CR) cardiovascular disease, 316-317 centenarians from Okinawa, 317-318 increased longevity in humans, 328-329 insulin resistance, 316-317 and lifespan, 315-316 physiological effects body composition, 321 cardiovascular and diabetes risk factors, 322-323 DHEA-S, 324 leptin, 324 longevity, biomarkers, 321-322 metabolic adaptation and oxidative stress, 323 physical activity, 324-326 somatotropic axis, 324 thyroid function, 323 psychological and behavioral effects cognitive function and performance, 328 eating disorder symptoms, 327 hunger, subjective feelings, 327 outcomes, 326 QOL and mood, 327-328 rate of living and oxidative stress, 316 RCT, nonobese humans, 318 type 2 diabetes mellitus, 316-317 unexpected CR, Biosphere 2, 318, 319 Vallejo study, 318 Calorie Restriction with Optimal Nutrition (CRON) diet, 328 Cancer cachexia, 305-306 exercise, benefits, 361 head and neck, 258 incurable, enteral nutrition, 254-255

MVM supplements, 382 obesity, 167 omega-3 fatty acids, 397 oral (see Oral cancer (OC)) PA/EX, effects, 361 physical activity, 361 preoperative, 257 prevention, vitamin E, 393 prostate, 385 selenium, 394-395 terminal, ANH, 305–306 treatable, malnourished cancer patients, 256-257 vitamin D, 391 Cardiac cachexia BNP. 221 characterization, 220 definition, 220 endotoxin, 221 ghrelin, 221 leptin, 221 metabolic disturbances, 220 starvation, 220 TNF-a, 220-221 Cardiac event prevention AHA diet and lifestyle recommendations, 205 alcohol, 210 comorbidities, 204-205 description, 203-204 and dietary intake, 205-209 dietary patterns, 210-212 dyslipidemia, 205 heart-healthy diet, 213 hypertension, 205 lifestyle behavioral counseling, 205 nutrition counseling, 205 physical changes, 204 PROSPER study, 205 socio-economic change, 204 sodium, 210 stanols and sterols, 209 statin therapy, 205 Cardiac function abnormalities, 216 oral administration, 226 and oxygen consumption, 222 and symptoms, 225 Cardiovascular disease (CVD) B-complex vitamins, 387 CR, anti-aging effects, 316-317 diabetes-related comorbidities, 190, 192 dynapenic obesity, 104 and MetS, 188 omega-3 fatty acids, 395-397 PA/EX, effects, 359-360 potassium, 394 soy protein and soy isoflavones, 398-400 type 2 diabetes, 180, 316-317 vitamin C supplementation, 225 vitamin D, 130

Caregiving team, interdependent roles consultant therapists occupational therapy, 292 physical therapy, 292 SLP. 291 social worker, 291 dietitians and dietary managers, 290-291 family caregivers, 292 interdisciplinary team approach, 292 nursing, 291 physician/nurse practitioner/physician assistant, 290 Carotenoids in Age-Related Eye Disease Study (CAREDS), 65, 66 Cataract cortical, 59 definition, 59 human eye, 58, 59 lens, 58, 59 lutein and zeaxanthin, 62-64 nuclear, 59 nutrient supplement effects LINXIAN trial, 70 REACT. 70 vitamin C, 69 vitamin E, 69-70 omega-3 fatty acids, 63 PSC, 59 vitamin C, 60, 61 vitamin E, 60-63 Certified nursing assistants (CNAs), 289, 291, 295 Chasen-Taber, L., 64, 69 Chen, X., 261-268 Chernoff, R.S., 240 Chocolate, 134, 135, 379 Christen, W.G., 64, 71 Chronic kidney disease (CKD) calcium-phosphorus, 267 diet and obesity, role insulin resistance and risk, 263-264 protein intake and kidney function, 262-263 salt intake, hypertension and risk, 262 dietary recommendations, 268 nutritional status, role, 262 parathyroid hormone, 267 uremia, nutritional issues malnutrition, pathophysiology, 265-267 obesity paradox, 265 vascular calcification, 267-268 Chung, M., 23, 24 Chylack, L.T. Jr., 61 Clark, B.C., 102 Clark, N., 195 Coates, P.M., 375-414 Coenzyme Q10 (CoQ10) for anti-aging, 378 cardiovascular health, 406-407 neurological health, 407-408 Cognitive function B vitamins, 127–130 cross-sectional studies, 126

epidemiological studies, 126 FFO. 126 mediterranean diet, 136-137 Mini-Mental State Examination, 127 monounsaturated and polyunsaturated fatty acids, 135-136 number of domains, 126 nutrient intake, 126 polyphenols, 132-135 prospective studies, 126 RCTs, 126-127 score ranges, 127 vitamin D, 130-132 Conners' Continuous Performance Test-II (CPT-II), 328 Costello, R., 375-414 Cragg, G.M., 410 Cresci, G., 259 CR mimetics, 326, 329 CRONIES, 323-324, 328-329 Cumming, R.G., 63 CVD. See Cardiovascular disease (CVD) Cytokines in apoptosis, 219 induced cachexia, 247 inflammatory, 221, 246-247 mediated anorexia and weight loss, 246 pressure ulcers, 246-247

D

Daly, J.M., 257 D'Anci, K.E., 125-138 Dawson-Hughes, B., 273-281 de Koning, L., 205 Delcourt, C., 63, 64 Delmi, M., 239 Dementia caregiving team, interdependent roles consultant therapists, 291-292 dietitians and dietary managers, 290-291 family caregivers, 292 interdisciplinary team approach, 292 nursing, 291 physician/nurse practitioner/physician assistant, 290 mealtime difficulties, assessment causes, 289–290 compromised meal behaviors, 293 health status, 293-295 neurologic/cognitive, 293 personal factors, 295-296 physical and social environment, 296-297 nutritional care, PWD careful hand feeding, PEG feeding, 298 PEG feeding, risks/benefits, 297-298 Dental caries artificial sweeteners, 84 characterized, 84 cheese consumption, 84 chewing sugar-free gum, 84

demineralization/remineralization process, 84 diet. 84 fluorides. 84 oral conditions, 82 saliva, 84 sugar intake, 84 Depression and cognition, 363 and dementia, 180 DM, 191-192 GDS, 294 QOL, 194 symptoms, 326 weight loss in LTC, 295 Desneves, K.J., 241 DETERMINE checklist, 38, 39, 48 Devore, E.E., 133 Dherani, M., 61 Dhesi, J.K., 132 Diabetes Control and Complications Trial (DCCT) assay, 181-183, 189 Diabetes mellitus (DM) acute illness, 186 ADA, 183 AGS, 183 balancing diet and medication, 185 blood glucose, 182 blood pressure reduction, 183 cardiovascular morbidity and mortality, 183 cognitive dysfunction, 192-193 comorbidities, 190 DCCT. 182-183 depression, 191-192 diagnosis, 181 dietary habits, 191 dietary intake, 193-194 enteral and parenteral nutrition, 186, 187 glycemic goals, 183 health consequences, 180 hospitalization, 186-187 imparting dietary information, 196 long-term care facilities, 188 macronutrient recommendations, 184-185 management of hyperglycemia, 183, 184 nutritional recommendations, 184 overweight and obesity, 185-186 quality of life, 194-196 self-management behaviors, 190-191 treatment, 181 type 1 diabetes, 181-182 type 2 diabetes, 182 **UKPDS**, 183 underweight and malnutrition, 186 Diabetes prevention program (DPP), 188, 196 Diabetes self-management education and training (DSME/T), 190, 192 Diamant, M., 184 Diastolic heart failure, 217, 218, 224

Diet eating and healthful, 4 healthy and nutrient-dense, 5 low quality diets types, 12 multivitamins, 31 older adults, 6, 10 and related behaviors, 13 restrictive, 4 Dietary Approaches to Stop Hypertension (DASH) Sodium Study, 172, 206, 208, 210, 262 Dietary counseling, 91, 218, 257-258 Dietary fats benefits vs. risks of eating fish, 207 omega-3 and omega-6 fats, 207 saturated and trans-fats, 206-207 Dietary fiber, 28, 90, 208, 336, 338, 340, 347 Dietary intake and blood levels, nutrients and eye disease, 60-68 DM and attitudes, 193 and self-care, ethnic/cultural issues, 193-194 and heart disease management added sugars, 209 dairy products, 208 dietary fats, 206-207 fruits and vegetables, 206 whole grains and starches, 208 nutrition assessment, 44 Dietary patterns OC, 88-89 preventive cardiology caloric intakes, 211-212 low-carbohydrate diet, 210 randomized clinical trial, 210 weight loss, 210 Dietary reference intakes (DRIs), 23, 31, 224-226, 383-384, 389 Dietary Supplement and Nonprescription Drug Consumer Protection Act, 379 Dietary Supplement Health and Education Act (DSHEA), 379 Dietary supplements bicarbonate supplements, 106 cardiovascular/heart heath condition-specific supplement, 379 combination and condition-specific products, 378 CoO10. 378 definition, DSHEA, 379 food-based interventions, 29 glucosamine/chondroitin products, 379 guidelines for safe use assessment, 409, 413 databases and other resources, 414 preventing interactions, 408-409 health claims FDA website, 380 Nutrition Labeling and Education Act, 380 health concerns and nutrient, 32

Dietary supplements (cont.) and herb-drug interactions, 410-412 hormones, 106 marketplace, 376 multivitamins, 378 non-nutrient supplements, effectiveness and safety herbals and botanicals, 308-404 non-nutrient, non-botanical supplements, 405-408 nutrient and herbal, 28 nutrient-containing supplements, 377-378 nutrient supplements, effectiveness and safety calcium and vitamin D combinations, 383-385 minerals, 393-395 omega-3 and-6 fatty acids, 395-398 vitamins, 381-393 omega-3 fatty acid, 106 protein supplements, 105-106 regulation, 379-380 types, 376-377 use, Older Americans, 377 Diet therapy education distribution, 196-197 older adult weight-loss therapy, 174 weight management intervention, 172 DiMaria-Ghalili, R.A., 9, 35-48 Diuretic-induced hyponatremia, 224 DM. See Diabetes mellitus (DM) Docosahexaenoic acid (DHA), 24, 29, 58, 86, 378, 386, 395 Dual-energy X-ray absorptiometry (DEXA), 103, 104 Dwyer, J.T., 375-414 Dynapenia consequences functional limitation and physical disability, 102 metabolic health, 103 mortality, 103 description, 102 obesity, 103-104, 106-107 prevalence, 102 prevention and treatment resistance training, 104-105 whole-body vibration, 105 Dyslipidemia added sugars, 209 dietary behaviors, 204 and hypertension, 205 low-density lipoprotein cholesterol levels, 205 MetS and T2DM, 360

E

Eat Better Move More, 372 Eating Inventory, 327 Ecological model aging, 4 nutrition interventions, 8 and social support models, 7 Edentulism and obesity, 90 tooth loss, 82–83

weight loss, 90 Edinburgh Feeding Evaluation in Dementia Scale (EdFED), 293 Eicosapentaenoic acid (EPA), 24, 29, 58, 67, 135, 386, 395 Electrolytes, 224, 227, 228, 230, 304 End-stage renal disease (ESRD), 262, 264 Enteral nutrition ALS, 307 diabetes mellitus, 186-187 at the end of life, 291, 308-309 head and neck cancer patients, 257 immunonutrition, 259 in incurable cancer patients, 254-255, 306 medical treatment, 305, 310 nutritional counseling, 258 tube feedings, 240, 304 Esmarck, B., 105 Estimated average requirement (EAR), 24, 275 Evans, E.M., 354-372 Evidence-based programs Administration on Aging (AoA), 368 National Council on Aging (NCOA), 368, 372 National Institute on Aging (NIA), 368 physical activity and nutrition programs, 368-371 SilverSneakers program, 372 Title IIID of the Older Americans Act (OAA), 368 Walk With Ease Program, 372 Evidence tables, 25-26 Exercise (EX) advantages, 118 aerobic, 106, 170 basal muscle protein synthesis rates, 118 benefits cancer. 361 cardiovascular disease, 359-360 cognition and depression, 363 dyslipidemia, 360 MetS. 360 mortality and QOL, 363-364 obesity, 358-359 osteoporosis, 361-363 physical functional capacity, 361-363 T2DM. 360 BMD, 171 capacity, 173 definition, 356 and diet, synergistic effect, 356-357 endurance exercise, 119 fast twitch/type II muscle fibers, 118 intracellular stress, 118 low-intensity activities, 119 motor unit denervation, 118 and nutrition, 120 for older adults, 357 physical activity, 172 prevention vs. management of disease, 357 resistance, 119, 170 specialist, 173 and weight loss therapy, 170, 173 Exercise and Physical Activity: Your Everyday Guide, 366 Eye disease AMD (*see* Age-related macular degeneration (AMD)) AREDS, 385 cataract (*see* Cataract) treatment guidelines, 72–74

F

Falls balance activities, 365 magnesium intake recommendation, 347 malnutrition, 36, 154 proteins, 337 unintentional weight loss, 41 vitamin D supplementation, 277-278, 280, 281, 389-390 Fan, S.T., 256-257 Fat-free mass (FFM), 101, 103, 164, 165, 168-170, 323 Fatty acids monounsaturated, 59, 135-136 n-3, 336 omega-3, 24, 58, 63, 66–67, 72, 86, 206–207, 336, 339-340, 378, 386, 395-397 omega-6, 378, 395-397 polyunsaturated, 59, 135-136, 222 triglyceride and ethyl esters, 29 Federal Food, Drug, and Cosmetic Act, 379 Feeding assistance, PWD, 289 behavior, changes, 45 careful hand, 297-298 comfort feeding only, 304, 307 difficulties, 131 direct hand, 291, 298, 299 enteral, 240, 257 hand over hand, 298, 299 hand under hand, 298, 299 hypercaloric, 245, 247 non-oral, 304 PEG, 245, 297-298 recreational, 304, 307 resistive or aversive, 288, 293 self-feeding, 293, 297 tube, 187, 242, 245, 291, 304-307 Feldman, J.J., 13 Ferrannini, E., 184 Ferrell. 239 Ferrigno, L., 61, 63 Fitness cardiorespiratory, 359-360, 363 centers, 372, 378 definition, WHO, 356 muscular, 362 physical, health-related, 356-357, 362 Flavanols, 134, 135 Florez, H., 195 Folic acid B vitamins, 341-342 CVD. 387 deficiencies, 150, 226 folate, 29

homocysteine levels, 388 mortality, 386-387 risks of excess amounts, 387-388 Food assistance food insecurity, 148 older adults, 155-156 Food insecurity distribution geographic variation, 153 individual and household characteristics, 151 - 152middle and older adults, 152-153 health implications diabetes, 153 malnutrition, 154-155 older adults, 154 poor physical and mental health status, 153 risk factors, 153 interventions, 155-156 linchpin of healthful living, 148 and malnutrition, 336 overnutrition and obesity, 151 in US CFSM, 148, 149 communities, 149 FSS. 148 high food security, 148 households, 148-149 low food security, 148 LSRO, 148 marginal food security, 148 USDA, 148 very low food security, 148 WHO definition, 148 Food intake and diets, 148 FFO, 126 healthcare services, 6 interpersonal and individual factors, 5 neighborhood environment, 6 policy-related factors, 6 social factors, 5 Foods bolus, swallowing, 92 carbohydrates, 84 carcinogenesis, 87 dietary patterns, 88-89 dry and sticky, 92 hard, 90 high-fat, 89 and hydration at end of life, 308 insecurity, 147-157 intakes of dairy, 85 lactic acid, 85 lutein/zeaxanthin, 73 and nutrients, 88, 338 physical characteristics, 84 plaque, 84 softer. 91 vitamin C, 73 vitamin E, 73

Food Safety Modernization Act, 380 Fractures. See also Osteoporosis and bone mass, vitamin D BMD, 278 hip fracture risk, 279 multiple-dose anti-fracture efficacy trial, 279 subject-level meta-analysis, 278-279 calcium intake requirements, 275, 346, 383-384, 389-390 femoral neck, 239, 244, 246 hip, 156, 169, 240, 273-275, 278, 337 vitamin D, 277-280, 344 Frailty BMI. 169 bone fractures, 361 physical activity, 172 quality of life, 170-171 sarcopenia, 114, 168, 182 sarcopenic-obese, 168 unintentional weight loss, 41 Fruits citrus, 134 polyphenol-rich foods, 136 randomized controlled trials, 133 total fruit, 134 and vegetables, 87, 132, 133, 137, 206 Fugh-Berman, A., 410 Functional capacity factors influencing, 361-363 loss, older adults, 361-362 and osteoporosis, 362

G

Gale, C.R., 64 Gardner, Z., 410 Geriatric Depression Scale (GDS), 294 Giles, G.E., 125-138 Gilmore, L.A., 314-329 Ginkgo biloba extract (GBE), 133, 403-404 Ginkgo Evaluation of Memory (GEM) study, 404 Glomerular filtration rate (GFR), 263 Glucosamine/Chondroitin Arthritis Intervention Trial (GAIT), 405 Goldberg, A., 410 Golden, S.D., 9 Goodwin, J.S., 128 Gorse, 239 Grapes, 133, 138 Green tea. 134 Gritz, D.C., 69 Growth hormone (GH), 106, 114, 118, 165, 221, 324 Guralnik, J.M., 239

Н

Haas, L.B., 195 Halfdanarson, T.R., 258 Halter, J.B., 195 Hankinson, S.E., 61, 62, 64, 69, 70 Hänsel, R., 411 Harris-Benedict equation, 241-242 Harris, T.B., 13 Hartgrink, H.H., 239 Health behavior change and adherence, 13 and education. 8 inform and guide interventions, 7 theories and models, 7 Health claims FDA website, 380 Nutrition Labeling and Education Act, 380 Health promotion information technology, 12-13 interventions, 9 older adults, 9 Healthy Moves for Aging Well, 368 Heart failure age-related changes, 216 age-specific nutritional issues, 227-228 caloric intake, fat and protein, 221-222 cardiac cachexia, 220-221 cardinal symptoms, 217 description, 216 device-specific nutritional therapies, 228 etiology, 216-217 medication effects, 227 metabolic syndrome, 219-220 nutrients (see Nutrient(s)) and obesity, 222-223 pathophysiology, 217 prognosis, 217-218 recommendations caffeine. 230 fluid intake, 229 nutritional guidelines, 228, 229 sodium restriction, 228 well-balanced diet rich, 230 treatment ACE inhibitors, 218 ARBs, 218 diuretics, 218 LVADs, 218-219 metoprolol and carvedilol, 218 nonpharmacological measures, 218 pharmacotherapy, 218, 219 spironolactone reduces mortality, 218 Heart failure with preserved ejection fraction (HFPEF), 217 - 218Heart failure with reduced ejection fraction (HFREF), 217 - 218Heart Outcomes Prevention Evaluation (HOPE) trial, 225-226, 392 Hellwig, J.P., 411 Hemodialysis, 265-267 Henderson, C.T., 240 Hendler, S., 410 Herbal Alternatives for Menopause Trial (HALT), 402

Herbals and botanicals black cohosh (Cimicifuga racemosa), 401-402 Ginkgo biloba, 403-404 herb-drug interactions, 410-412 saw palmetto (Serenoa repens), 402-403 soy protein and soy isoflavones CVD, 398-400 menopausal symptoms, 401 osteoporosis, 400 supplements and drugs, 409 Holley, C.L., 215-231 Hormones, 41, 106, 108, 114, 187, 321, 323, 361-362, 401 Hot flushes (Hot flashes), 399, 401 Houwing, R.H., 239 Humans CR (see Calorie restriction (CR)) longevity, 316, 328-329 nonobese, RCT, 318 obese, 322-323 protein-calorie undernutrition, 237 vitamin D deficiency, 277 Hunger CR. 327 food insecurity, 148 and malnutrition, 149 older adults, 150 U.S. Household Food Security Scale, 149 Hunt, C.D., 275 Hypertension added sugars, 209 cardiovascular disease, 359-360 cardiovascular risk factors, 205 CHD, 259 CKD, 262 CoQ10, 407 dietary behaviors, 204 and dyslipidemias, 205 heart failure and obesity, 222 and hypercholesterolemia, 392 salt intake and CKD, 262 Hypoglycemia, DM blood glucose-lowering agents, 189 DCCT, 189 drugs, 189 malnourished, 189 management of diabetes, 188 self-monitoring and dietary treatment, 189 unawareness and treatment, 189-190 Hypomagnesemia, 224

I

Immunonutrition, 259 Inman, 239 Instrumental activities of daily living (IADLs), 4, 44, 156, 190, 293, 389 Insulin-like growth factor-1 (IGF-1), 114–118, 317, 324 Intervention emerging, 12–13 heart failure interventions and age, 227–228 inflammatory-mediated cachexia, 247 nutrition (*see* Nutrition interventions) nutritional, 239–241 persons with diabetes, 186–188 weight management intervention, 172 Inzucchi, S.E., 184

J

Jacobs, D.R. Jr., 208 Jacques, P.F., 61, 62, 69 Jalal, D., 61 Janssen, I., 99–108 Jatoi, A., 253–259 Johnson, D., 10 Johnson, E., 275 Johnson, E.J., 57–74 Johnson, M.A., 354–372 Joshipura, K.J., 81–93

K

Kahn, R.L., 13 Kalmijn, S., 126 Kanarek, R.B., 125-138 Keller, H.H., 291 Key questions components, 25 limitations, 31 nutrients, 29 systematic reviews, 23 vitamin D and calcium, 23 King, A.C., 3-14 Kirkman, S.M., 195 Klein, S., 168 Knekt, P., 63 Knight, H.M., 179-197 Krall, E.A., 85 Kraus, W.E., 203-213 Kristal, A., 8

L

LaRowe, T.L., 66 Lee, J.S., 154 Lee, S.K., 241 Left ventricular assist devices (LVADs), 218, 219, 228 Leigh, B., 241 Leske, M.C., 60-63, 69, 70 Levine, M., 410 Licensed Practical Nurses (LPNs), 291 Lichtenstein, A.H., 21-32 Lindquist, L.W., 91 Lipid-modified diet, 205 Locher, J.L., 147-157 Longevity biomarkers, 321-322 CR, 322 DHEA-S, 324 in humans, 328-329

Long-term care (LTC). *See* Dementia Lutein and zeaxanthin AMD, 65–66 cataracts, 62–64 content of foods, 72, 73 dietary intakes, 386 Lyle, B.J., 64

М

Macready, A.L., 127 Macronutrients dietary fiber, 340 omega-3 fatty acids, 339-340 protein, 337-338 Macular degeneration. See Age-related macular degeneration (AMD) Malnutrition adult, 46-47 ASPEN, 46 definition, 46 ESPEN, 46 and health, 154-155 and hunger, 149 inflammation, 46 older adults, 150 and underweight, 186 in uremia, 265-267 Manini, T.M., 102 Mares-Perlman, J.A., 60-66 McCann, R.M., 255 McCay, C.M., 315 McGeer, A.J., 254 McGuffin, M., 410 Mealtime difficulties, dementia causes, PWD, 289-290 compromised meal behaviors, 293 health status meal intake and bowel movement review, 294-295 psychosocial factors, 295 vital signs/lab work and weight patterns, review, 294 impact, 288 neurologic/cognitive, 293 personal factors emotional factors, 295 functional performance, 296 physiological determinants of intake, 295 physical and social environment dining ambience, 297 light level, noise level, and/or temperature, 296 recommendations, 296 staff stability and staff mix, 297 precipitators, 289 Measured resting energy expenditure (mREE), 241 Medical nutrition therapy (MNT), 12, 187 Mediterranean diet AD, 137 and cognitive aging, 136 components, 137

food groups, 136 healthy eating index, 137 Mini-Mental State Examination, 137 monounsaturated fats and fish, 137 polyphenol, 136 vascular disease, 137 Mente, A., 205 Merchant, A.T., 86 Merkel, J., 375-414 Messier, V., 103 Meta-analysis calcium balance, 280 omega-3 fatty acids, 66-67 **RCTs**, 382 subject-level, 279 systematic reviews, 27 vitamin D supplements, 391 Metabolic acidosis, 265-267, 280 Metabolic syndrome (MetS) and CVD, 188 description, 188 DPP, 188 heart failure, 219–220 insulin resistance, 188 NCEP-ATP III, 188 T2DM and dyslipidemia, 360 type 2 diabetes, 188 Metabolism calcium, 85, 344, 346 carbohydrate, 221, 227 energy, 323, 325, 342, 407 lipid, 227, 346, 356 muscle, 117, 120 protein, 114, 220-222, 346 MetS. See Metabolic syndrome (MetS) Meyers, L.D., 411 Miller, M.D., 101 Milles, S., 410 Minerals bone mineral density, 171 calcium, 346 heart failure, 225 magnesium, 346-347 potassium CVD. 394 2010 Dietary Guidelines for Americans, 393 high risk nutrients, 347 pressure ulcers, 244-245 selenium cancer. 394-395 excessive amounts, 395 and vitamins, 244-245 Mini-Mental State Examination scores (MMSE), 127, 132, 134, 135, 137, 293, 398 Minimum Data Set (MDS), 38, 292, 293 Mini-nutrition assessment-short form (MNA-SF), 39 Moayyeri, A., 274 Modification of Diet in Renal Diseases (MDRD) Study, 263 Moeller, S.M., 66

Mohan, M., 61 Mokshagundam, S.P.L., 179-197 Monounsaturated fatty acids (MUFA), 135-136, 187 Moolten, 239 Moreland, J.D., 102 Moss, J., 410 MTHFR genotype, 129-130 Multifactorial Assessment of Eating Disorder Symptoms (MAEDS), 327 Multivitamin/multimineral (MVM) supplements cancer. 382 chronic disease prevention, 381 cognition, 382 CoQ10, 378 dietary intakes, 383 infection, 382 mortality, 382 nutrients in diet, 381 products, 377 Muscle mass and IGF-1/Akt-1 signaling, 116 and muscle strength, 99-108 older adults, 114-117 sarcopenia, 114, 115 signaling hormones, 114, 115 skeletal muscle, 114 STARS/SRF signaling, 116-117 Muscle metabolism muscle mass, regulation (see Muscle mass) and nutrition, 117-118 strength (see Strength and function, muscle)

Ν

Nadalin, G., 69, 70 Nass, R.M., 106 National Cholesterol Education Program-Adult Treatment Panel III (NCEP-ATP III), 188 National Council on Aging (NCOA), 368 National Health and Nutrition Examination Survey (NHANES), 262, 337-338, 376 National Health Interview Survey, 377 National Institute on Aging (NIA), 368 Nauck, M., 184 Need-Driven Dementia Compromised (NDB) Model, 290 Net acid excretion (NAE) values, 280-281 Newsome, D.A., 67, 71 Nixon, D.W., 254 Norris, J.R., 241 Nourmohammadi, I., 63 Nutrient(s) antioxidant, 60, 70, 72, 74 bioavailability, 28 bioequivalence, 29 biological stores, 29 and eye disease, 60-68 heart failure carnitine, 226 coenzyme Q_{10} , 226 creatine phosphate, 226

electrolytes, 224 minerals, 225 multinutrient therapy, 227 propionyl L-carnitine, 226 vitamins, 225-226 water and sodium, 223-224 and oral health, 81-93 status, 28 Nutrient intake food groups, 338 macronutrients dietary fiber, 340 omega-3 fatty acids, 339-340 protein, 337-338 minerals calcium, 346 magnesium, 346-347 potassium, 347 NHANES, 338 vitamins B vitamins, 341-343 vitamin A and carotenoids, 340-341 vitamin D, 343-345 vitamin E, 345 Nutrition assessment (see Nutrition assessment) heart failure caloric intake, fat and protein, 221-222 cardiac cachexia, 220-221 metabolic syndrome, 219-220 and obesity, 222-223 malnutrition. 36 and muscle metabolism amino acids and BCAA, 117 anabolic resistance, 117 inflammation and oxidative stress, 118 mTOR, 117 nutritional status, 36 osteoporosis, nutritional concerns (see Osteoporosis) overnutrition, 36 screening (see Nutrition screening) systematic reviews, 21-32 Nutritional supplements, pressure ulcers amino acids, 243-244 energy, 241-242 mode of delivery, 245 protein, 242-243 vitamins and minerals, 244-245 Nutrition assessment anthropometrics, 40-41 body composition, 41-42 comorbid conditions, 45-46 description, 40 dietary intake, 44 functional status, 44 health professions, 47-48 medications, 44, 45 physical exam, 42-43 psychosocial and economic factors, 44-45 visceral proteins, 43

Nutrition interventions. See also Older adults aging. 4 emerging intervention, 12-13 evaluation and assessment, 10-11 financial resources, 12 food assistance programs, 12 health-promoting information technology, 12-13 **MNT. 12** obesity, 14 persons with diabetes acute illness, 186 enteral and parenteral nutrition, 187 hospitalization, 186-187 long-term care, 188 racial and ethnic minorities, 11 in rural areas, 12 Nutrition Labeling and Education Act, 380 Nutrition screening and assessment, 36, 37 definition. 36 DETERMINE checklist, 39 health professions, 47-48 MNA-SF, 39 programs, 39 regulations community, 38 homecare, 38 hospital/acute care settings, 36 long-term care, 36, 38 primary care, 38 Nutrition support in advanced, incurable cancer patients acknowledging exceptional circumstances, 255-256 alternative palliative strategies, 255 enteral nutrition, 254-255 parenteral nutrition, 254 decision-making families and surrogate decision maker, 309 legal, religious, and ethical precedents, 304-305 and treatment goals, 308 definitions, 304 at end of life ANH, advanced illness, 305-308 enteral/parental nutrition and hydration, 308-309 food and hydration, 308 solid tumor cancer patients head and neck cancer patients, 258 immunonutrition, 259 nasogastric vs. PEG, 257 nutritional counseling, 258 treatable cancers, 256-257

0

Obesity adverse effects arthritis, 167 cancer, 167 comorbid disease, 166

functional impairment and QOL, 168-169 metabolic abnormalities, 166-167 mortality, 166 pulmonary abnormalities, 167 urinary incontinence, 167 and diet, CKD insulin resistance and risk, 263-264 protein intake and kidney function, 262-263 salt intake, hypertension and risk, 262 dynapenia aerobic exercise, 106 caloric restriction, 107 consequences, 104 definition and prevalence, 103-104 and heart failure, 222-223 in older adults aging, 164 beneficial effects, 169 body fat, 164 definition. 163 fat-free mass (FFM), 164 fat mass (FM), 164 NHANES, 164 and overnutrition, 151 and overweight, 165, 166, 185-186 paradox in uremia, 265 pharmacotherapy, 173 prevalence of obesity, 164 and undernutrition, 149 weight loss, 170-171, 173 weight management intervention, 163, 172-173 PA/EX, effects, 358-359 sarcopenia aerobic exercise, 106 caloric restriction, 107 consequences, 103 definition and prevalence, 103 Office of Dietary Supplements (ODS), 410-412, 414 Older adults DM (see Diabetes mellitus (DM)) exercise, 118-120 food assistance, 155-156 with heart failure (see Heart failure) high-risk nutrients (see Nutrient intake) hypoglycemia, 188-190 muscle metabolism, 114-118 nutrition interventions community-based and institutional, 9 components and delivery channels, 7-8 ecological perspective, 8 food intake, 5-6 framework, 6-7 interpersonal and intrapersonal, 8-9 multilevel and policy, 9-10 nutritional status, 4-5 planning and design, 6 obesity (see Obesity) oral health (see Oral health) PA/EX (see Exercise (EX); Physical activity (PA)) systematic reviews, 31-32

weight-loss therapy bariatric surgery, 175 behavior therapy, 174 diet therapy, 174 exercise therapy, 174 initial assessment, 174 Ollberding, N., 8 Omega-3 fatty acids AMD AREDS, 67 Beaver Dam Eye Study, 67 Blue Mountain Eye Study, 67 Dietary Ancillary Study of the Eye Disease Case-Control Study, 67 and late ARM, 67, 68 meta-analysis, 66 cancers, 397 cataracts, 63 cognition, 397-398 CVD. 395-397 fish oils, 72, 74 Omega-6 fatty acids, 395-398 Oral cancer (OC) anti-oxidants and micronutrients, 87-88 dietary patterns, 88-89 disparities, 83 eating and swallowing, 92 fiber and whole grains, 88 fruits and vegetables, 87 and pharyngeal, 83 problem management, 92 squamous cell carcinoma, 83 tobacco and alcohol use. 86 Oral health dental caries, 81, 82 nutritional status chronic periodontitis, 85-86 dental caries, 84 dentate, blood, 90-91 OC (see Oral cancer (OC)) plaque and calculus formation, 83-84 tooth loss, 89-90 tooth replacement strategies, 91 xerostomia, 92 periodontal disease, 82 periodontitis, 81 quality of life, 81 root caries, 82 tooth loss, 82-83 Osteoarthritis (OA), 167, 405-406 Osteoporosis acid-base balance, diet, 280-281 BMD assessments, 274 calcium intake requirements, 275 physiology, 274-275 recommendations, 276 safety, 276 supplements, 383-384 definition, 273

heritability of bone loss, 274 and physical functional capacity, 361–363 vitamin D bone mass and fractures, 278–279 muscle performance, balance, and falls, 277–278 physiology and interactions, aging process, 277 recommendations, 280 supplements, 383–384 Otten, J.J., 411

Р

Paddon-Jones, D., 113-120 Palacios, C., 81-93 Palliative treatment, 308 Parathyroid hormone (PTH), 267, 274 Parenteral nutrition in advanced, incurable cancer patients, 254 diabetes mellitus, 186 palliative effects, 255-256 treatable cancers, 256-257 Parkinson's disease, 308, 344, 407-408 Patel, Krishna A., 253–259 Peneau, S., 133 Pennings, B., 106 Percutaneous endoscopic gastrostomy (PEG) careful hand feeding, 298 placement, 307-308 pressure ulcer, 245 risks/benefits, 297-298 Periodontal disease, 82, 85, 86, 91-93 Periodontitis bacterial plaque, 85 BMI, 86 dairy foods, 85 diet and nutrients, 85 genetic risk factors, 85 immune-modulating nutrients, 86 NHANES III, 85, 86 and obesity, 86 omega 3 fatty acids, 86 PPD, 86 and tooth loss, 85 VDR polymorphisms, 85 vitamin C, 86 vitamin D, 85, 86 waist circumference (WC), 86 Persons with dementia (PWD). See also Dementia careful hand feeding, PEG feeding, 298 mealtime difficulty, 289-290 PEG feeding, risks/benefits, 297-298 Pharmacotherapy left ventricular systolic dysfunction, 219 obesity in older adults, 173 osteoporosis, 276 systolic heart failure, 218 Physical activity (PA) benefits cancer. 361 cardiovascular disease, 359-360

Physical activity (PA) (cont.) cognition and depression, 363 dyslipidemia, 360 MetS, 360 mortality and QOL, 363-364 obesity, 358-359 osteoporosis, 361-363 physical functional capacity, 361-363 T2DM, 360 CR, 324-326 definition. 356 environmental barriers, 367 evidence-based programs, 368, 372 Exercise is Medicine, 367 for older adults, 357 PAR-Q, 367 2008 Physical Activity Guidelines for Americans, 364-365 prevalence, 364 prevention vs. management of disease, 357 self-esteem and self-efficacy, 368 2008 Physical Activity Guidelines for Americans aerobic exercise, 365 balance exercises, 365 DHHS PA guidelines, 364 muscle-strengthening activities, 365 warm-up, cool-down, and flexibility exercises, 365 Physical activity readiness questionnaire (PAR-Q), 367 Physical function dynapenia, 102 functional capacity, 361-363 and quality of life, 170-171 sarcopenia, 101 Pinchcofsky-Devin, G.D., 239 Plaque and calculus formation, 83-84 Podrabsky, M., 205 Polyphenols berries, 133 chocolate, 134 description, 132 executive function, 135 flavanoid intake, 134 fruits and vegetables, 132, 133 grapes, 133 green tea, 134 language and verbal abilities, 134-135 n-3 PUFA, 138 plant food consumption and performance, 132 polyphenol-rich foods, 133 questionable dementia, 132 total polyphenol intake, 134-135 Polyunsaturated fatty acids (PUFA), 135-138, 222 Posterior subcapsular cataract (PSC), 59, 60, 62, 63, 70 Potassium bicarbonate, 281 Preoperative cancer patients, 257 Pressure ulcers cachexia, 245 cytokines, 246-247 epidemiological associations, 238, 239 healing, trials, 240-241

inflammatory-mediated cachexia, interventions, 247 nutritional supplements amino acids, 243-244 energy, 241-242 mode of delivery, 245 protein, 242-243 vitamins and minerals, 244-245 prevention enteral tube feedings, 240 nutritional interventions, 238, 239 oral nutrition supplements, effects, 238-240 serum proteins, measures, 245-246 and undernutrition, 237-238 weight loss and undernutrition, 245 Pressure Ulcer Scale for Healing (PUSH) score, 242-244 Prostate cancer, 341, 345-346, 382, 385, 393-394, 403 Prostate, Lung, Colorectal, and Ovarian Cancer Screening Trial, 388 Prostate-specific antigen (PSA), 385 Protein high-risk nutrients, 337-338 intake with CKD, 263 and nutritional status, uremia, 265-266 with renal function, 262-263 metabolism exercise, 18-20 nutrition (see Nutrition) PLP, 341 Pruitt, J.D., 203-213 Przybelski, R., 132 Public Health Security and Bioterrorism Preparedness and Response Act of 2002, 379

Q

Quality of life (QOL) DM and diabetes diet, 194–195 and weight loss, 195–196 and functional impairment, 168–169 malnutrition, 36 and mood, 327–328 and mortality, 363–364 and patient definitions, 193 and physical function, 170–171 physical functioning and vitality, 328

R

Rautiainen, S., 69 Ravasco, P., 258 Ravussin, E., 314–329 Recommended dietary allowance (RDA), 275, 337 Redman, L.M., 314–329 Registered Dietitian (RD), 290–291 Remifemin, 402 Remig, V., 205 Renal disease, 262–263 Resistance exercise

and aerobic exercise, 360, 363 blood flow, 105 muscle mass and strength in older populations, 119 muscle strengthening activities, 104-105 physical activity, 104 protein supplements, 105 treatment strategy, sarcopenia and dynapenia, 108 Resting energy expenditure (REE), 241, 316, 358 Resting metabolic rate (RMR), 164, 241, 319, 323-324, 337 Retina AMD, 58 human eye, 59 and lens, 58 lutein and zeaxanthin, 65 oxidative stress, 59 photoreceptors, 60 RPE, 59 Rey Auditory and Verbal Learning Test (RAVLT), 328 Rich, M.W., 215-231 Ritchie, C.S., 303-310 Robertson, J.M., 69, 70 Root caries, 82, 84 Rorvik, D., 410 Rosenberg, I., 99 Rose, S.B., 287-299 Rouhiainen, P., 63 Rowe, J.W., 13 Russell, A.P., 113-120

\mathbf{S}

Sahyoun, N.R., 8 Sanders, T.A.B., 65, 307 Sarcopenia adults at risk, 113 age-related skeletal muscle loss, 99, 100 body composition, 170 combat, 114 consequences functional limitation and physical disability, 101 metabolic health, 102 mortality, 101 definition and prevalence, 100-101, 114 FFM. 168 and frailty, 182 muscle size, 100 obesity, 103, 106-107 physical frailty, 114 prevalence, 168 prevention and treatment resistance training, 104-105 strategies, 114 whole-body vibration, 105 progression, 114 in uremia, 266 Schulz V, 411 Schweitzer, D., 66 Sebring, N.G., 91 Seddon, J.M., 65, 66, 71

Selenium and Vitamin E Cancer Prevention Trial (SELECT), 393 Self-monitoring of blood glucose levels (SMBG), 189, 190 Selman, C., 316 Senechal, M., 104, 107 Shah, K., 163-175 Shea, B., 275 Sheats, J.L., 3-14 Siener, C., 168 SilverSneakers program, 372 Silver tsunami, 4 Simon, J.A., 61 Sinacore, D.R., 168 Skeletal muscle (SM) aging human, 116 molecular signaling pathways, 115 muscle protein synthesis and muscle protein, 114 resistance training + protein supplements, 105 sarcopenia, 100-102 size and strength, 104 Snellen, E.L., 66 Sodium DASH-sodium trial, 210, 262 food preparation, 12 lowering blood pressure, 210 restriction. 223-224 and water, 223-224 Speech language pathologist (SLP), 291 Stanford Chronic Disease Self-Management Program, 372 Stanols, 209 Starches, 89, 208 Stein, M.S., 132 Sterols, 209, 403 Stetson, B., 179–197 Strength and function, muscle and aerobic, 364 and functional capacity, 114 and muscle mass (see Muscle mass) vitamin D supplementation, 390 Sugars and calories content, 12 and carbohydrates, 83, 84 classification, 209 controlling, sugar consumption, 209 empty calories, 336 food and beverages, 209 glucosamine/chondroitin, 405 Sunshine vitamin, 130 Symons, T.B., 105 Systematic reviews analytic framework, 22-24 extract and summarize data, 25-26 field of medicine, 22 field of nutrition, 22 hypovitaminosis D, 131 impartial and complete assessment, 22 inclusion/exclusion criteria, 24 internet-mediated interventions, 196 literature selection, 25, 26

Systematic reviews (cont.) meta-analysis, 27 nutrition availability, biological stores, 29 baseline exposure, 27-28 bioavailability, 28 bioequivalence, 29 DHA. 29 dose-response relationships, 30 EPA, 29 food-based interventions, 29 nutrient status, 28 older adults, 31-32 research (key) questions, 23 review team, 22 RTCs, 127, 381, 382, 385, 389 search terms, 25 Systolic heart failure, 217, 218, 222, 229

Т

Tan, A.G., 61 Tan, J.S.L., 66 Tavani, A., 61, 62, 69 Taylor, H.R., 71 Taylor, T.V., 241 Tea black/oolong, 134 green, 134 sweeten coffee and breakfast cereal, 209 Technology-based interventions, 12-13 Teikari, J.M., 71 Terminal illness cancer, 305-306 dementia alternative approaches, 307 costs and benefits, 306-307 terminal stages, neurological diseases, 307-308 ter Riet, G., 241 Theory free radical theory of aging, 316 health behavior change, 7 intervention methods and practical strategies, 7 planned behavior, 7 rate of living theory, 316, 323 social cognitive theory, 7 Thiamine, 225, 227 Thomas, D.R., 237-248 Tooth loss health surveys in US, 82 non-Hispanic black and white adults, 82 oral health body weight status, 90 nutritional status, 89-90 public health problems, 83 Tooth replacement strategies, 91 Total daily energy expenditure (TDEE), 325 Trumbore, L.S., 239 Tucker, K.L., 128, 335-348 Tyler, V.E., 411

Type 1 diabetes, 181–182, 185, 189 Type 2 diabetes mellitus (T2DM) calorie restriction, 316–317 cardiovascular disease, 316–317 characterization, 182 dynapenic obesity, 104 insulin resistance, 316–317 mechanism of insulin resistance, 182 metabolic syndrome, 188 overweight and obesity, 185

U

Ubiquinone. See Coenzyme Q10 (CoQ10) Undernutrition biochemical and anthropometric variables, 238 categories, 245 hunger, 150 malnutrition and health, 154-155 and obesity, 149 older adults, 150-151 and overnutrition, 156 and pressure ulcers, 245 protein-calorie undernutrition, 231 serum albumin, 242 and weight loss, 245 Unintentional weight loss AD, 45 adverse effects, 41 comorbid conditions, 45-46 intentional vs., 41 and malnutrition, 36, 44, 305 and nutritional status, 36 pressure ulcers, 238 undernutrition, 150-151 United Kingdom Prospective Diabetes Study (UKPDS), 183 Uremia malnutrition, pathophysiology inflammation and nutritional status, 267 metabolic acidosis and nutritional status, 266-267 protein intake and nutritional status, 265-266 obesity paradox, 265 sarcopenia, 265 US Department of Health and Human Services (DHHS), 364

V

Vader, J.M., 215–231 Valero, M.P., 61 van Anholt, R.D., 241 Van Belle, T.L., 182 Villareal, D.T., 163–175 Vision loss AMD (*see* Age-related macular degeneration (AMD)) cataract (*see* Cataract) lens implantation, 57 Visser, M., 102 Visual Analogue Scales (VAS), 327

Vitale, S., 61, 63, 69 Vitamin A and carotenoids, 340-341 Vitamin C AMD, 64, 65 cataracts antioxidant status, 60, 61 epidemiologic studies, 60, 61 NHANES II. 60 plasma ascorbic acid concentrations and prevalence, 60 content of foods, 72, 73 fruits and vegetables, 87 recommended daily allowance (RDA), 244 Vitamin D administration, 131 age-related diseases, 130 aging population, 343-345 Alzheimer's disease, 131 antioxidant and anti-inflammatory properties, 131 bioavailability, 130 bone health, 389-391 building and maintaining strong bones, 130 and calcium BMD. 384 fractures, 384 homeostasis, 226 mortality and outcomes, 384-385 osteoporosis, 383-384 cancer. 391 in children, 130 chronic diseases, 391-392 cognition and dementia, 391 dietary supplements, 130 intranasal insulin, 132 low serum levels, 131 MCI, 131 and mental functioning, 132 nervous system, 130, 131 neurotransmitters, 131 neurotrophic factors, 131 oral administration, 132 osteoporosis bone mass and fractures, 278-279 muscle performance/balance/falls, 277-278 physiology and interactions, aging process, 277 recommendations, 280 RCT. 132 serum level of 25(OH)D, 130, 131 type 1 diabetes, 181-182 type 2 diabetes, 182 VDR, 130-131 Vitamin D receptor (VDR), 85, 130, 131, 275, 277 Vitamin E aging population, 345 AMD, 64, 65 anti-oxidant properties, 225 cancer prevention, 393 cardiovascular health, 392-393 cataracts epidemiologic studies, 60, 62 high and low vitamin E intake, 61-62

lower prevalence, 60-61 plasma vitamin E, 62, 63 content of foods, 72, 73 fruits and vegetables, 87 Vitamins B-complex vitamins cognition, 388-389 CVD. 387 mortality, 386-387 risks of excess amounts, 387-388 combinations eye health, 385-386 MVMs, 381-383 prostate cancer, 385 folic acid, 226 high-dose niacin, 226 homocysteine, 226 HOPE-TOO, 225-226 low beta-carotene intake, 226 vitamin A and carotenoids, 340-341 vitamin B₆, 226, 341-342 vitamin B₁₂, 226, 342-343 vitamin B₁ (thiamine) deficiency, 225 vitamin C supplementation, 225 vitamin D aging population, 343-345 bone health, 389-391 cancer, 391 cognition and dementia, 391 other chronic diseases, 391-392 vitamin E aging population, 345 cancer prevention, 393 cardiovascular health, 392-393 Vogiatzoglou, A., 128 Vu, H.T.V., 64

W

Waist circumference (WC), 11, 86, 165-167, 188, 208 Walford, Dr.R., 318 Walk With Ease Program, 372 Weight loss aerobic exercise, 106 anorexia/cachexia syndrome, 245 anthropometrics, 40-41 caloric restriction, 107 CMS. 38 comorbid conditions, 45, 46 and cytokine-mediated anorexia, 246 definition, MDS, 292 geriatric patients, 38, 44 health status, 293-294 heart failure, 221 intentional vs. unintentional, 41 measurement, 42 MNA-SF, 39 obesity BMD. 171 body composition, 170 medical complications, 170

Weight loss (cont.) mortality, 171 physical function and QOL, 170-171 physiological determinants of intake, 295 and QoL in older adults with diabetes, 195-196 surgery, 173 unintentional (see Unintentional weight loss) Weight management intervention behavior modification, 172-173 diet therapy, 172 physical activity, 172 Western Ontario and McMaster Universities Arthritis Index (WOMAC), 405 West, S., 65 White, J.D., 410 Whole-body vibration apparatus, 105 Whole grains dietary fiber, 340 and starches, 208 Williamson, D.A., 327 Winter, S.J., 3-14

Wolff's law, 362 Women's Health Initiative (WHI), 276 Women's Isoflavone Soy Health trial (WISH), 399 Wound healing, pressure ulcers, 240–241

Х

Xerostomia, 89, 92, 93

Y

Yamada, Y., 316 Yukawa, M., 35–48, 303–310

Z

Zinc AMD, 67–68 eye health, 385 supplementation, 72 urinary excretion, 225